# SHAW, PITTMAN, POTTS & TROWBRIDGE

A PARTNERSHIP INCLUDING PROFESSIONAL CORPORATIONS

2300 N STREET, N.W. WASHINGTON, D.C. 20037-1128 (202) 663-8000 FACSIMILE (202) 663-8007

April 30, 1997

JOHN H. O'NEILL, JR. P.C. (202) 663-8148

(Pr

**By Hand Delivery** 

U.S. Nuclear Regulatory Commission Document Control Desk One White Flint North 11555 Rockville Pike Rockville, MD 20852

> Re: **Application for Transfers of Control Regarding Operating** Licenses Nos. DPR-44 and DPR-56 for the Peach Bottom Atomic Power Station (Docket Nos. 50-277 and 50-278) and Licenses Nos. DPR-70 and DPR-75 for the Salem Nuclear Generating Station (Docket Nos. 50-272 and 50-311)

Ladies and Gentlemen:

Pursuant to Section 184 of the Atomic Energy Act, as amended, 10 C.F.R. § 50.80, and NRC Administrative Letter 96-02, Atlantic City Electric Company t/a Atlantic Electric ("ACE") and Delmarva Power & Light Company ("DP&L") hereby submit an application for transfer of control of their interests in Operating Licenses Nos. DPR-44 and 56 for the Peach Bottom Atomic Power Station ("Peach Bottom") and Nos. DPR-70 and 75 for the Salem Nuclear Generating Station ("Salem"). The Application requests the consent of the Nuclear Regulatory Commission ("NRC") to the indirect transfers of control of ACE and DP&L's possessory interests in the Peach Bottom and Salem licenses that will occur under a proposed merger of Atlantic Energy, Inc. (the parent holding company of ACE) and DP&L, resulting in the formation of a new holding company, Conectiv, Inc. ("Conectiv"). PECO Energy Company, the licensed operator of Peach Bottom, and Public Service Electric & Gas Company, the licensed operator of Salem, are not involved in the merger.

The merger will achieve significant cost savings and efficiencies and create a stronger, more competitive enterprise. It is expected to generate about \$500 million in net savings during the first ten years of Conectiv's operation, thereby enhancing the applicants' financial resources to pio061/1 possess their respective minority ownership interests in the Peach Bottom and Salem units.

010036

PDR

12

9705020075 970430 PDR ADDCK 05000272

#### SHAW, PITTMAN, POTTS & TROWBRIDGE A PARTNERSHIP INCLUDING PROFESSIONAL CORPORATIONS

U.S. Nuclear Regulatory Commission April 30, 1997 Page 2

In order to maximize the significant cost savings and efficiencies upon consummation of the merger, the applicants request expeditious review of this Application by the NRC and the issuance of a 10 C.F.R § 50.80 consent. The applicants believe that the NRC should be able to consent to the indirect transfers of control occasioned by the proposed merger, since the merger will have no adverse affect on either the technical management or operation of Peach Bottom or Salem, will not require a specific demonstration of financial qualifications since the applicants' status as utilities under 10 C.F.R § 50.2 will remain unchanged, and will not require any NRC antitrust review because both Peach Bottom and Salem were licensed under Section 104(b) of the Atomic Energy Act.

The proposed merger requires approval of other regulatory agencies in addition to consent from the NRC. The NRC Project Managers for Peach Bottom and Salem will be kept informed of the progress made by the other regulatory agencies. The merger is expected to be completed by December 31, 1997, enabling Conectiv to come into corporate existence on that date. Accordingly, it is requested that the NRC consent to the indirect transfer of control of the Peach Bottom and Salem licenses by September 30, 1997.

If you have any questions or require additional information, please contact me at the telephone number below or Paul Gaukler at (202) 663-8304.

truly yours

John H. O'Neill, Jr. SHAW PITTMAN POTTS & TROWBRIDGE 2300 N Street, N.W. Washington, D.C. 20037 (202) 663-8148

Enclosure

cc (w/ encl.): Regional Administrator, NRC Region I Resident Inspector, Peach Bottom Resident Inspector, Salem Joseph W. Shea, Project Manager, Peach Bottom Leonard N. Olshan, Project Manager, Salem

# APPLICATION FOR TRANSFER OF CONTROL REGARDING OPERATING LICENSE NOS. DPR-70 AND DPR-75 FOR THE SALEM NUCLEAR GENERATING STATION AND OPERATING LICENSE NOS. DPR-44 AND DPR-56 FOR THE PEACH BOTTOM ATOMIC POWER STATION

# EXHIBIT A

# JOINT PROXY STATEMENT/PROSPECTUS



December 26, 1996

Dear Fellow Stockholder:

I am pleased to send you the enclosed Joint Proxy Statement/Prospectus describing the details of Atlantic Energy's Agreement and Plan of Merger with Delmarva Power & Light Company.

After carefully reviewing the enclosed material, I urge you to cast your vote by marking, signing and dating the enclosed proxy and returning it promptly in the accompanying envelope.

I also invite you to attend a special meeting of Atlantic Energy stockholders on Thursday, January 30, 1997, at 3:00 p.m. at the Frank Guaracini, Jr. Fine & Performing Arts Center at Cumberland County College, Sherman Avenue and College Drive, Vineland, New Jersey 08360. At this important meeting, our stockholders will be asked to approve the Agreement and Plan of Merger. You will also be asked to approve the Conectiv, Inc. Incentive Compensation Plan (the "Company Plan"), which upon consummation of the Mergers will replace Atlantic's Equity Incentive Plan and an existing plan of Delmarva's, subject to approval by stockholders. Votes cast by mail will be tallied at the meeting. The Joint Proxy Statement/ Prospectus provides a detailed account of the matters that will be acted upon at the meeting. Even if you plan to attend, we suggest that you mail in your proxy to expedite the voting process.

OUR BOARD OF DIRECTORS HAS CAREFULLY CONSIDERED THE TERMS OF THE MERGER AGREEMENT AND THE COMPANY PLAN AND BELIEVES THEY ARE IN THE BEST INTEREST OF ATLANTIC ENERGY AND ITS STOCKHOLDERS. OUR BOARD, BY UNANI-MOUS VOTE, APPROVED THE MERGER AGREEMENT AND, BY UNANIMOUS VOTE OF ALL DIRECTORS PRESENT, THE COMPANY PLAN AND RECOMMENDS THAT OUR STOCKHOLD-ERS VOTE FOR APPROVAL OF THE MERGER AGREEMENT AND THE COMPANY PLAN.

# Why We Support The Merger With Delmarva Power

Simply stated, the merger will place us in a stronger position to compete in the energy services industry.

As described in greater detail in the Joint Proxy Statement/Prospectus, Atlantic Energy and Delmarva have remarkably similar visions for the kind of companies we want to be and the strategic paths we must follow to achieve success. Both companies believe that in order to succeed we must have competitive rates, offer innovative service choices, have large customer bases and have access to even larger consumer markets. Our proposed merger with Delmarva will significantly strengthen our position in all of these critical areas while providing benefits to our stockholders, customers, employees and communities. Finally, we strongly believe that combining forces now is a natural fit and a wise strategic move for our respective companies.

# **Key Terms Of The Transaction**

- As a result of the merger contemplated by the Merger Agreement, all of Atlantic Energy's subsidiaries, including Atlantic City Electric Company ("ACE"), and Delmarva Power and all of its subsidiaries will become subsidiaries of a new holding company.
- For each full share of Atlantic Energy Common Stock you own, you will receive 0.75 shares of Common Stock of the new holding company (par value \$0.01 per share) and 0.125 shares of Class A Common Stock of the new holding company (par value \$0.01 per share).
- Based on the stock conversion ratios established in the merger agreement and the capitalization of the two companies as of September 30, 1996, Atlantic Energy stockholders would hold 39.4% of the total shares outstanding of the new holding company Common Stock if the merger were consummated on such date. This new holding company Common Stock will represent the equity interest in the new holding company represented by the business of Delmarva, the nonregulated businesses of ACE, the first \$40 million of the annual earnings attributable to the current regulated business of ACE and 70% of such earnings in excess of the first \$40 million.

- Atlantic Energy's common stockholders will receive all of the outstanding shares of the Class A Common Stock representing 30% of the earnings of the new holding company attributable to the current regulated utility business of ACE in excess of \$40 million per year, and Atlantic Energy and Delmarva have determined that dividends on the Class A stock will be based on this amount after the "Initial Period" described below. Such amounts for the fiscal year ended December 31, 1995 and the 12-month period ended September 30, 1996 are set forth in the accompanying Joint Proxy Statement/Prospectus under "Description of the Company's Capital Stock." Accordingly, the Class A stock is intended to apportion a greater share of the business results, growth prospects and the risks associated with the regulated utility business of ACE to Atlantic Energy's stockholders. Together with the interest in such earnings represented by the Company Common Stock, Atlantic Energy stockholders will retain more than half of the benefits and risks related to earnings of the current regulated utility business of ACE in excess of \$40 million per year. See the Joint Proxy Statement/Prospectus under "Description of the Company's Capital Stock." The Class A stock will not, however, have any specific rights or claims against the businesses, assets or liabilities of ACE. Both classes of common stock of the new holding company are expected to be listed on the New York Stock Exchange.
- We anticipate that the initial annual dividend for the new holding company Common Stock will be \$1.54 per share, the same annual dividend both companies now pay. The Merger Agreement provides that, subject to declaration by the Board of the new holding company and its obligation to react to the financial condition and regulatory environment of the new holding company and its results, the new holding company will pay an annual dividend of \$3.20 per share on the Class A Common Stock until the earlier of July 1, 2001 or the end of the twelfth calendar quarter following the calendar quarter in which the mergers are consummated (the "Initial Period"). Accordingly, during the Initial Period it is expected that the current Atlantic stockholders will receive an aggregate dividend of \$1.555 for each Atlantic share they currently hold. See "Description of the Company's Capital Stock—Common Stock—Dividends" and "The Company Following the Mergers—Dividend Policies" in the accompanying Joint Proxy Statement/Prospectus."

Please refer to the accompanying Joint Proxy Statement/Prospectus for the complete details of the Class A Common Stock, the new holding company Common Stock and the proposed merger.

Once again I encourage you to promptly complete and return your proxy. Your vote is important no matter how many shares you own. For the Merger Agreement to be approved, it must receive the affirmative vote of at least a majority of all votes cast by holders of Atlantic Energy Common Stock. For the Company Plan to be approved, it must receive the affirmative vote of a majority of all votes cast by holders of Atlantic Energy Common Stock. For the Company Plan to be approved, it must receive the affirmative vote of a majority of all votes cast by holders of Atlantic Energy Common Stock. Please be aware that if you should change your mind after casting your proxy, you may revoke it in writing any time or choose to vote your shares personally upon request if you attend the meeting. The accompanying Joint Proxy Statement/Prospectus sets forth the voting rights of holders of Atlantic common stock with respect to these matters. Stockholders are urged to review carefully the attached Joint Proxy Statement/Prospectus, which contains a detailed description of the Merger Agreement, the terms and conditions thereof and the transactions contemplated thereby and of the Company Plan.

In addition to receiving formal approval of the merger from our respective stockholders, as described in the accompanying Joint Proxy Statement/Prospectus, Atlantic Energy and Delmarva must secure numerous consents from state, federal and regulatory bodies. We anticipate that this approval process will take approximately 12 to 18 months to successfully complete from our August 1996 announcement date. Promptly following the completion of the mergers, we will mail you a formal notice. The notice will include instructions on how to exchange your Atlantic Energy Common Stock Certificates for Company Common Stock certificates and Class A Common Stock certificates. PLEASE DO NOT SEND YOUR CERTIFI-CATES WITH THE ENCLOSED PROXY CARD.

On behalf of Atlantic Energy's Board of Directors, I thank you for taking the time to consider and vote upon this important issue.

Very truly yours,

Jewold L. Jacobs



# ATLANTIC ENERGY, INC. 6801 Black Horse Pike Egg Harbor Township, NJ 08234-4130 (609) 645-4420

# NOTICE OF SPECIAL MEETING OF STOCKHOLDERS To Be Held on January 30, 1997

December 26, 1996

To the Stockholders of Atlantic Energy, Inc.

A Special Meeting (the "Special Meeting") of holders of common stock of Atlantic Energy, Inc., a New Jersey corporation ("Atlantic"), will be held on Thursday, January 30, 1997 at 3:00 p.m. at The Frank Guaracini, Jr. Fine & Performing Arts Center at Cumberland County College, Sherman Avenue and College Drive, Vineland, New Jersey 08360 for the following purposes.

1. To consider and vote upon a proposal to approve an Agreement and Plan of Merger, dated as of August 9, 1996 as amended and restated as of December 26, 1996 (the "Merger Agreement") among Atlantic, Delmarva Power & Light Company, a Delaware and Virginia corporation ("Delmarva"), Conectiv, Inc., a newly formed Delaware corporation (the "Company"), and DS Sub, Inc., a newly formed Delaware corporation and wholly owned transitory subsidiary of the Company established solely to effectuate a merger with and into Delmarva ("DS Sub"). The Merger Agreement provides for the merger of Atlantic with and into the Company (the "Atlantic Merger") and the merger of DS Sub with and into Delmarva (the "Delmarva Merger") (the Atlantic and the Delmarva Merger, together the "Mergers"), and whereby with certain limitations, (a) each issued and outstanding share of common stock, no par value, of Atlantic (the "Atlantic Common Stock") will be converted into the right to receive 0.75 of one share of Common Stock, par value \$.01 per share, of the Company (the "Class A Common Stock"); (b) each issued and outstanding share of the common stock, par value \$.01 per share, of the Company (the "Class A Common Stock"); (b) each issued and outstanding share of the common Stock, par value \$.01 per share, of the Company (the "Class A Common Stock"); (b) each issued and outstanding share of the common stock, par value \$.02 per share, of Delmarva (the "Delmarva Common Stock"); (b) each issued and outstanding share of the common stock, par value \$.25 per share, of Delmarva (the "Delmarva Common Stock") will be converted into the right to receive one share of Company Common Stock; and (c) the common stock issues of Delmarva and Atlantic will become common stockholders of the Company, all as more fully described stock.

accompanying Joint Proxy Statement/Prospectus.

į,

2. To consider and vote upon a proposal to approve the Company Incentive Compensation Plan.

3. To transact such other business related to matters incident to the conduct of the Special Meeting as may properly come before the Special Meeting or any adjournment or adjournments thereof.

Based on the conversion ratios set forth in the Merger Agreement and the capitalization of Atlantic and Delmarva as of September 30, 1996, (i) the holders of Atlantic Common Stock and Delmarva Common Stock would have held approximately 39.4% and 60.6%, respectively, of the Company Common Stock and (ii) the holders of the Atlantic Common Stock would have held 100% of the Class A Common Stock representing 30% of the earnings of the Company attributable to the current regulated utility business of Atlantic City Electric Company in excess of \$40 million per year that would have been outstanding, had the Mergers been consummated as of such date.

Stockholders of record at the close of business on December 18, 1996 will be entitled to notice of and to vote at the Special Meeting or at any adjournment or adjournments thereof. Even if you now expect to attend the Special Meeting, you are requested as promptly as possible to mark, sign, date and return the accompanying proxy in the enclosed addressed, postage-paid envelope.

If you do attend the Special Meeting, you may vote in person if you so desire.

By order of the Board of Directors,

ames E. Franklin I

James E. Franklin II Secretary

Whether or not you expect to be present at the Special Meeting, please sign, date and return the accompanying proxy promptly so that your shares may be represented and voted at the Meeting. You may revoke your proxy in writing if you so desire at any time before it is voted. A return envelope, which requires no postage if mailed in the United States, is enclosed for your convenience. If you have any questions or need assistance in voting your proxy, please call our proxy solicitor, Georgeson & Company Inc. at (800) 223-2064.

#### JOINT PROXY STATEMENT 1.51 OF

# **DELMARVA POWER & LIGHT COMPANY** AND **ATLANTIC ENERGY. INC.**

i de la composición de

States & States

S. AND MARKS STARS

Personal Report

# PROSPECTUS

11.1.1.1.12

a green to be at

61 - 1. 39L -

the and states

# reasonation a range of OF sector as the range been CONECTIV, INC.

# COMMON STOCK which is handwide a reflective AND and Article as the

# CLASS & COMMON STOCK

Special Meeting of Stockholders of Delmarva Power & Light Company to be Held on January 30, 1997

#### Special Meeting of Stockholders of Atlantic Energy, Inc. to be Held on January 30, 1997 $q \in q$

Conectiv, Inc., a Delaware corporation (the "Company"), has filed a registration statement on Form S-4 (together with all amendments, schedules and exhibits thereto, the "Registration Statement") with the Securities and Exchange Commission (the "SEC") pursuant to the Securities Act of 1933, as amended (the "Securities Act"), covering the shares of Common Stock, par value \$.01 per share ("Company Common Stock"), of the Company and Class A Common Stock, par value \$.01 per share ("Class A Common Stock"), of the Company to be issued (in the case of the Company Common Stock) in connection with the merger (the "Delmarva Merger") of DS Sub, Inc., a Delaware corporation and a wholly owned transitory subsidiary of the Company established solely to effectuate such merger ("DS Sub"), with and into Delmarva Power & Light Company, a Delaware and Virginia corporation ("Delmarva"), and (in the case of the Company Common Stock and the Class A Common Stock) in connection with the merger (the "Atlantic Merger") of Atlantic Energy, Inc., a New Jersey corporation ("Atlantic"), with and into the Company (the Atlantic Merger and the Delmarva Merger are referred to herein, together, as the "Mergers"), all as more fully described below. The Company will be a registered public utility holding company under the Public Utility Holding Company Act of 1935, as amended (the "1935 Act"). This Joint Proxy Statement/Prospectus is being furnished in connection with the Special Meeting of Stockholders of Delmarva and the Special Meeting of Stockholders of Atlantic. This Joint Proxy Statement/Prospectus also constitutes the prospectus of the Company filed as a part of the Registration Statement. See "Available Information." The second free of the : A. A. A. Sec. 5 (1) ater a color se

All information herein with respect to Delmarva has been furnished by Delmarva, all information herein with respect to Atlantic has been furnished by Atlantic, and all information herein with respect to the Company has been furnished by the Company.

See "Risk Factors" commencing on page 14 for information that should be considered by stockholders in connection with the Mergers.

THESE SECURITIES HAVE NOT BEEN APPROVED OR DISAPPROVED BY THE SECURITIES AND EXCHANGE COMMISSION OR ANY STATE SECURITIES COMMISSION NOR HAS THE SECURITIES AND EXCHANGE COMMISSION OR ANY STATE SECURITIES COMMISSION PASSED UPON THE ACCURACY OR ADEQUACY OF THIS PROSPECTUS. ANY REPRESENTATION TO THE **CONTRARY IS A CRIMINAL OFFENSE.** 

The date of this Joint Proxy Statement/Prospectus is December 26, 1996. This Joint Proxy Statement/ Prospectus is first being mailed to the stockholders of Delmarva and Atlantic on or about December 27, 1996.

# Matters Relating to the Mergers

Stockholder Meetings. At the Delmarva Special Meeting (the "Delmarva Meeting"), scheduled to be held on January 30, 1997, and the Atlantic Special Meeting (the "Atlantic Meeting"), scheduled to be held on January 30, 1997, common stockholders of Delmarva and Atlantic, respectively, will consider and vote upon a proposal to approve an Agreement and Plan of Merger dated as of August 9, 1996 as amended and restated as of December 26, 1996 (the "Merger Agreement"), among Delmarva, Atlantic, the Company and DS Sub and the transactions contemplated thereby, pursuant to which, among other things: (i) DS Sub will be merged with and into Delmarva in the Delmarva Merger, with Delmarva to be the surviving corporation; (ii) Atlantic will be merged with and into the Company in the Atlantic Merger, with the Company to be the surviving corporation; and (iii) Delmarva and Atlantic City Electric Company, a wholly owned subsidiary of Atlantic ("ACE"), will become wholly owned subsidiaries of the Company. In addition, at the Delmarva Meeting and the Atlantic Meeting, common stockholders of Delmarva and Atlantic, respectively, will consider and vote upon a proposal to approve the Company Incentive Compensation Plan (the "Company Plan").

Conversion and Cancellation of Shares Upon Consummation of the Mergers. Upon consummation of the Mergers, pursuant to the Merger Agreement:

• Each issued and outstanding share of Delmarva Common Stock, par value \$2.25 per share ("Delmarva Common Stock") (other than shares of Delmarva Common Stock that are owned by Delmarva as treasury stock or by Atlantic or by any wholly owned subsidiary of Delmarva or Atlantic), will be converted into the right to receive one fully paid and nonassessable share of Company Common Stock. Each share of Delmarva Common Stock that is owned by Delmarva as treasury stock or by Atlantic or any wholly owned subsidiary of Delmarva as treasury stock or by Atlantic or any wholly owned subsidiary of Delmarva or Atlantic will be cancelled and will cease to exist and no consideration will be delivered in exchange therefor.

• Each issued and outstanding share of Atlantic common stock, no par value per share ("Atlantic Common Stock") (other than shares of Atlantic Common Stock that are owned by Atlantic as treasury stock or by Delmarva or by any wholly owned subsidiary of Atlantic or Delmarva), will be converted into the right to receive 0.75 fully paid and nonassessable shares of Company Common Stock and 0.125 fully paid and nonassessable shares of the Class A Common Stock. Each share of Atlantic Common Stock or by Delmarva or any wholly owned subsidiary of Atlantic or any wholly owned subsidiary of Atlantic or Delmarva will be cancelled and will cease to exist and no consideration will be delivered in exchange therefor.

• All shares of capital stock of the Company issued and outstanding immediately prior to the Mergers will be cancelled without consideration and will cease to exist.

Upon consummation of the Mergers, each certificate representing Delmarva Common Stock and Atlantic Common Stock issued and outstanding prior to the Mergers, other than any shares that will not be converted as described above, will represent instead the right to receive the shares of Company Common Stock and, in the case of Atlantic Common Stock, Class A Common Stock and cash in lieu of fractional shares into which those issued and outstanding shares will be converted. Upon conversion, all such shares of Delmarva Common Stock and Atlantic Common Stock will be cancelled without consideration and cease to exist, and each holder of a certificate representing any such shares will cease to have any rights with respect thereto, except the right to receive, upon the surrender of the certificate, without interest, the shares of Company Common Stock and, in the case of Atlantic Common Stock; Class A Common Stock. Each share of Preferred Stock (par value \$100 per share) and each share of Preferred Stock—\$25 Par of Delmarva (together, the "Delmarva Preferred Stock") outstanding at the time of the consummation of the Mergers will remain outstanding preferred stock of Delmarva and the rights and preferences of the Delmarva Preferred Stock will not be affected by the Mergers.



No person is authorized to give any information or to make any representation concerning matters covered by this Joint Proxy Statement/Prospectus other than those contained or incorporated by reference in this Joint Proxy Statement/Prospectus, and if given or made, such information or representation should not be relied upon as having been authorized. This Joint Proxy Statement/Prospectus does not constitute an offer to sell, or a solicitation of an offer to purchase, the securities offered by this Joint Proxy Statement/Prospectus or the solicitation of a proxy, in any jurisdiction, to or from any person to whom or from whom it is unlawful to make such offer, solicitation of an offer or proxy solicitation in such jurisdiction. Neither the delivery of this Joint Proxy Statement/Prospectus nor any distribution of securities pursuant to this Joint Proxy Statement/Prospectus shall, under any circumstances, create an implication that there has been no change in the information set forth herein since the date of this Joint Proxy us. AVAILABLE INFORMATION Statement/Prospectus. د. مریکه و اکس

Each of Delmarva, Atlantic and ACE is subject to the informational requirements of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and accordingly files reports, proxy statements and other information with the SEC. Such reports, proxy statements and other information filed with the SEC are available for inspection and copying at the public reference facilities maintained by the SEC at Room 1024, Judiciary Plaza, 450 Fifth Street, N.W., Washington, D.C. 20549 and at the SEC's Regional Offices at 7 World Trade Center, Suite 1300, New York, New York 10048. Copies of such documents may also be obtained from the Public Reference Room of the SEC at Judiciary Plaza, 450 Fifth Street, N.W., Washington, D.C. 20549, at prescribed rates. In addition, any such material and other information concerning both Delmarva and Atlantic may be inspected at the New York Stock Exchange, Inc. (the "NYSE"), 20 Broad Street, New York, New York 10005 and the Philadelphia Stock Exchange, Inc. (the "Philadelphia Exchange"), 1900 Market Street, Philadelphia, Pennsylvania 19103 and, in the case of Atlantic, the Pacific Stock Exchange, Inc. (the "Pacific Exchange"), 301 Pine Street, San Francisco, California 94104. Certain of such reports, proxy statements and other information filed by Delmarva, Atlantic or ACE are also available on the Internet at the SEC's World Wide Web site at http:// and the second attendance which is a straight for ام مانین www.sec.gov.

The Company has filed with the SEC under the Securities Act a Registration Statement with respect to the shares of Company Common Stock and the Class A Common Stock issuable in the Mergers. This Joint Proxy Statement/Prospectus does not contain all of the information set forth in the Registration Statement, certain parts of which are omitted in accordance with the rules and regulations of the SEC. The Registration Statement, including any amendments, schedules and exhibits thereto, is available for inspection and copying as set forth above. Summaries of the contracts or documents referred to herein are summaries of the material provisions thereof and are not necessarily complete and in each instance reference is made to the copy of such contract or other document filed as an exhibit to the Registration Statement, each such summary being qualified in all respects by such reference.

and the second second

and the second second

iii

# **INCORPORATION BY REFERENCE**

This Joint Proxy Statement/Prospectus incorporates documents by reference that are not presented herein or delivered herewith. These documents are available upon request from, in the case of documents relating to Delmarva, Donald P. Connelly, Secretary, Delmarva Power & Light Company, 800 King Street, Wilmington, Delaware 19899 (Tel: (302) 429-3011), and, in the case of documents relating to Atlantic and ACE, James E. Franklin II, Secretary, Atlantic Energy, Inc., 6801 Black Horse Pike, Egg Harbor Township, New Jersey 08234-4130 (Tel: (609) 645-4420). In order to ensure timely delivery of the documents, any request should be made by five business days before the date of the meeting.

Delmarva and Atlantic hereby undertake to provide without charge to each person, including any beneficial owner of Delmarva Common Stock or Atlantic Common Stock to whom a copy of this Joint Proxy Statement/Prospectus has been delivered, upon the written or oral request of such person, a copy (without exhibits, except those specifically incorporated by reference) of any and all of the documents referred to below that have been or may be incorporated in this Joint Proxy Statement/Prospectus by reference. Requests for such documents should be directed to the persons indicated above.

The following documents, previously filed with the SEC pursuant to the Exchange Act, are hereby incorporated by reference:

1. Delmarva Annual Report on Form 10-K for the year ended December 31, 1995 (File No. 1-1405) (the "Delmarva 1995 Form 10-K");

2. Delmarva Quarterly Reports on Form 10-Q for the quarters ended March 31, 1996, June 30, 1996 and September 30, 1996 (File No. 1-1405) (the "Delmarva 1996 Forms 10-Q");

3. Delmarva Current Reports on Form 8-K dated February 22, 1996, May 29, 1996, July 23, 1996 and August 9, 1996 (File No. 1-1405);

4. Delmarva Proxy Statement dated April 26, 1996, for the 1996 Annual Meeting of Stockholders held on May 30, 1996;

5. Atlantic Annual Report on Form 10-K for the year ended December 31, 1995 (File No. 1-9760) (the "Atlantic 1995 Form 10-K");

6. Atlantic Quarterly Reports on Form 10-Q for the quarters ended March 31, 1996, June 30, 1996 and September 30, 1996 and on Form 10-Q/A for the quarter ended September 30, 1996 (File No. 1-9760) (the "Atlantic 1996 Forms 10-Q");

7. Atlantic Current Reports on Form 8-K dated February 23, 1996, May 29, 1996, June 26, 1996, July 25, 1996, August 13, 1996 and October 23, 1996 (File No. 1-9760); and

8. Atlantic Proxy Statement dated March 15, 1996, for the 1996 Annual Meeting of Stockholders held on April 24, 1996.

9. ACE Annual Report on Form 10-K for the year ended December 31, 1995 (File No. 1-3559) (the "ACE 1995 Form 10-K");

10. ACE Quarterly Reports on Form 10-Q for the quarters ended March 31, 1996, June 30, 1996 and September 30, 1996 and on Form 10-Q/A for the quarter ended September 30, 1996 (File No. 1-3559) (the "ACE 1996 Forms 10-Q");

All documents filed by Delmarva, Atlantic and ACE pursuant to Section 13(a), 13(c), 14 or 15(d) of the Exchange Act after the date hereof and prior to the date of the Delmarva Meeting on January 30, 1997, and any adjournment or adjournments thereof, or the Atlantic Meeting on January 30, 1997, and any adjournment or adjournments thereof, shall be deemed to be incorporated by reference herein and to be a part hereof from the date of filing of such documents.

Any statement contained in a document incorporated or deemed to be incorporated by reference herein shall be deemed to be modified or superseded for purposes of this Joint Proxy Statement/ Prospectus to the extent that a statement contained herein or in any other subsequently filed document which also is or is deemed to be incorporated by reference herein modifies or supersedes such statement. Any such statement so modified or superseded shall not be deemed, except as so modified or superseded, to constitute a part of this Joint Proxy Statement/Prospectus.

# TABLE OF CONTENTS

.

	Page
AVAILABLE INFORMATION	iii
INCORPORATION BY REFERENCE	iv
TABLE OF CONTENTS	v
INDEX OF DEFINED TERMS	ix
INDEX OF DEFINED TERMS	1
The Parties	· 1
The Company	1
Delmarva	1
Atlantic	· 1
Risk Factors	1
Stockholder Meetings	1
Delmarva	. 1
Atlantic	· 2
Required Vote	· 2 3
The Mergers Exchange of Stock Certificates	3
Exchange of Stock Certificates	. 4
Conversion Ratios for Delmarva Common Stock and Atlantic Common Stock	5 5 5 5 6
Company Incentive Compensation Plan	5
Background of the Mergers	· 5
Reasons for the Mergers	5
Recommendations of the Boards of Directors	. 6
Delmarva	6
Atlantic	6
Opinions of Investment Bankers	6
Delmarva	6
Atlantic	6
Conflicts of Interest of Certain Persons in the Mergers	. 7
Directorships	7
Severance Agreements	7
Employee Plans	7
Indemnification	7
Certain Federal Income Tax Consequences	7
Accounting Treatment	8
Appraisal Rights	. 8
Regulatory Matters	8
Conditions to the Mergers	9
Rights to Terminate, Amend or Waive Conditions	9
Common Equity of the Company	10
Dividends	12
Comparison of Corporate Charters and Rights of Security Holders	13
Selected Information Concerning Delmarva and Atlantic	13
The Company Following the Mergers	. 13
RISK FACTORS Risk Factors Associated with the Dual Class Common Equity Structure	14
Risk Factors Associated with the Dual Class Common Equity Structure	14
Risk Factors Associated with Each Class of Common Stock	18
Risk Factors Associated with the Class A Common Stock	19
Risk Factors Associated with the Company Common Stock	21
SELECTED HISTORICAL AND PRO FORMA FINANCIAL DATA	23
Selected Historical Financial Data	23
Selected Unaudited Pro Forma Combined Financial Data and Comparative Per Share Data .	25
Comparative Earnings, Dividends and Book Values Per Share of Common Stock and Class A	
Common Stock	26
Comparative Per Share Prices of Delmarva Common Stock and Atlantic Common Stock	28

MEETINGS, VOTING AND PROXIES		
Delmarva Meeting		
Atlantic Meeting		
THE MERGERS		• • • •
Background of the Mergers		
Recommendations of the Boards of Director	S	
	••••••	
	• • • • • • • • • • • • • • • • • • • •	
	•••••••••••••••••••••••••••••••••••••••	
Conflicts of Interest of Certain Persons in th	e Mergers	• • • •
Employee Plans, Severance Arrangements and	d Agreements	• • • •
The Company Benefit Plans		
Certain Federal Income Tax Consequences .		
	• • • • • • • • • • • • • • • • • • • •	
Stock Exchange Listing of Company Capital	Stock	
	·····	
		• • • •
REGULATORY MATTERS		
Antitrust Considerations		
Federal Power Act.		
Public Utility Holding Company Act of 1935		
Nuclear Regulatory Commission	• • • • • • • • • • • • • • • • • • • •	
Delaware Public Service Commission		
Virginia State Corporation Commission		
Pennsylvania Public Utility Commission	• • • • • • • • • • • • • • • • • • • •	
Maryland Public Service Commission		• • • •
Effects of Certain Degulatory Trands	· · · · · · · · · · · · · · · · · · ·	· • • •
Other Degulatory Metters	• • • • • • • • • • • • • • • • • • • •	••••
	••••••••••••••••••••••••••••••••••••	
THE MERGER AGREEMENT		
The Mergers	•••••••••••••••••••••••••••••••••••••••	
Consummation of the Mergers		
Representations and Warranties	• • • • • • • • • • • • • • • • • • • •	
Certain Covenants		. <b></b>
No Solicitation of Transactions		
	•••••••••••••••••••••••••••••••••••••••	
	••••••	
		• • • •
Iermination Fees		
Expenses	•••••••••••••••••••••••••••••••••••••••	••••

- .

.

	Page
DESCRIPTION OF THE COMPANY'S CAPITAL STOCK	75
General	75
Preferred Stock	75
Common Stock	75
Intergroup Interest	79
Conversion and Redemption of the Class A Common Stock	82
Notice and Other Provisions with Respect to Conversion and Redemption of the Class A	
Common Stock	85
Allocation of Consideration Between the Company Common Stock and the Class A	
Common Stock in a Subsequent Merger or Consolidation	87
Voting	87
Liquidation Rights	88
Determinations by the Company Board	88
No Preemptive Rights	88
Certain Definitions	88
Stock Exchange Listing	96
Anti-Takeover Provisions	96
COMPARISON OF CORPORATE CHARTERS AND RIGHTS OF SECURITY HOLDERS.	98
Delmarva	98
Atlantic	103
APPROVAL OF THE COMPANY INCENTIVE COMPENSATION PLAN	108
Objective	108
Shares Available Under the Plan	108
Termination	108
Administration	108
Participation	109 109
Section 162(m) Compliance	109
Awards	110
Tax Withholding	112
Federal Income Tax Consequences	112
New Plan Benefits	112
UNAUDITED PRO FORMA COMBINED FINANCIAL STATEMENTS	115
CONECTIV, INC. CONSOLIDATED BALANCE SHEET	139
SELECTED INFORMATION CONCERNING DELMARVA AND ATLANTIC	141
Business of Delmarva	141
Business of Atlantic	141
Prior Relationship Between Delmarva and Atlantic	142
Salem Nuclear Generating Station	142
THE COMPANY FOLLOWING THE MERGERS	145
Management of the Company	145
Operations of the Company	147
Dividend Policies	148
Financial Statements	148
Name of the Company	148
EXPERTS	149
LEGAL MATTERS	149
STOCKHOLDER PROPOSALS	149

,

		Page
ANNEX I	MERGER AGREEMENT	I-1
ANNEX II	FAIRNESS OPINION OF MERRILL LYNCH, PIERCE, FENNER & SMITH	II-1
	INCORPORATED	
ANNEX III	FAIRNESS OPINION OF MORGAN STANLEY & CO. INCORPORATED	III-1
ANNEX IV	THE COMPANY'S RESTATED CERTIFICATE OF INCORPORATION	IV-1
ANNEX V	THE COMPANY'S RESTATED BYLAWS	<b>V-1</b>
ANNEX VI	ILLUSTRATION OF CERTAIN CLASS A COMMON STOCK TERMS	VI-1
ANNEX VII	COMPANY INCENTIVE COMPENSATION PLAN	VII-1

viii

# **INDEX OF DEFINED TERMS**

.

,

	Page
1935 Act	i
1992 Act	
Abernathy	
ACE	
ACE 1995 Form 10-K	
ACE 1996 Forms 10-Q	
ACE Charter	
ACE Common Stock	
ACE Preferred Stock	
AEE	
AEII	
Antitrust Division	
Atlantic	
Atlantic 1995 Form 10-K	-
Atlantic 1996 Forms 10-Q	
Atlantic Board	
Atlantic Bylaws	
Atlantic Charter	-
Atlantic Common Stock	
Atlantic Conversion Ratio	
Atlantic Meeting	•
Atlantic Merger	,
Atlantic Record Date	
Atlantic Utility Group	
Atlantic Utility Group Available Dividend Amount	
Atomic Energy Act	
Class A Common Stock	
Code	
Common Shares Trust	
Common Stock	
Company	
Company Bylaws	
Company Charter	
Company Common Stock	
Company Net Income (Loss) Attributable to the Atlantic Utility Group	
Company Net Income (Loss) Attributable to the Residual Group	
Company Plan	
Convertible Securities	
D&T Consulting Group	
Delmarva	
Delmarva 1995 Form 10-K	
Delmarva 1996 Forms 10-Q	
Delmarva Board	
Delmarva Bylaws	
Delmarva Charter	
Delmarva Common Stock	
Delmarva Conversion Ratio	
Delmarva LTIP	29



-

	Page
Delmarva Meeting	ii
Delmarva Merger	i
Delmarva Preferred Stock	ii
Delmarva Record Date	· 2
DGCL	3
Disposition	91
DPSC	62
DS Sub	i
Effective Date	91
Effective Time	3
Exchange Act	iii
Exchange Agent	4
Fair Value	91
FERC.	8
Final Order	71
Fraction	80
FTC	61
Group	14
HSR Act	8
Indemnified Party	70
Initial Period	19
Intergroup Interest	80
Intergroup Interest Fraction	91
IRS	8
ISO	110
LeBoeuf/NJ	35
LeBoeuf/NY	34
Market Capitalization	92
Market Value	92
Market Value Ratio of the Class A Common Stock to the Company Common Stock	92
Merger Agreement	ii
Mergers	i
Merrill Lynch	6
Morgan Stanley	6
MPSC	63
Net Proceeds	93
NJBCA	3
NJBPU	- 11
NorthBridge	36
Notional Fixed Charge	10
NRC	8
Number of Shares Issuable with Respect to the Intergroup Interest	93
NYSE	iii
Out-of-Pocket Expenses	73
Outstanding Atlantic Utility Fraction	94
Pacific Exchange	iii
Philadelphia Exchange	iii
Potter Anderson	35
PPUC.	63
Preferred Stock	75

	Page
PSE&G	142
Publicly Traded	94
Registration Statement	i
Related Business Transaction	94
Residual Group	94
Residual Group Available Dividend Amount	95
Salem	142
SEC	i
Securities Act	i
Simpson Thacher	34
SAR	110
Target Party	74
Time-Weighted Market Price	96
Trading Day	96
VSCA	3
VSCC	3

# SUMMARY OF JOINT PROXY STATEMENT/PROSPECTUS

The following is a summary of the material terms and conditions of the Mergers and related information. This summary does not purport to be complete and is qualified in its entirety by reference to the more detailed information appearing in this Joint Proxy Statement/Prospectus and the Annexes. Stockholders are urged to read this Joint Proxy Statement/Prospectus and the Annexes in their entirety.

# The Parties

The Company. The Company is a Delaware corporation that was created to become a holding company for Delmarva, ACE and the other subsidiaries of Delmarva and Atlantic following the Mergers. The Company will be a public utility holding company registered under the Public Utility Holding Company Act of 1935, as amended (the "1935 Act"). See "The Company Following the Mergers." It has, and prior to the consummation of the Mergers will have, no operations except as contemplated by the Merger Agreement. See "The Merger Agreement." As of the date of this Joint Proxy Statement/ Prospectus, 50% of the outstanding capital stock of the Company is owned by each of Delmarva and Atlantic. The principal executive offices of the Company are located at 800 King Street, Wilmington, Delaware 19899. Upon the consummation of the Mergers, the name of the Company will be changed to "Conectiv."

*Delmarva*. Delmarva is an investor owned public utility that provides electric and gas service. Delmarva provides electric service to approximately 437,500 retail and wholesale customers in Delaware, ten primarily Eastern Shore counties in Maryland, and the Eastern Shore area of Virginia in an area consisting of about 6,000 square miles with a population of approximately 1.1 million. Delmarva provides gas service to approximately 98,000 retail and transportation customers in an area consisting of about 275 square miles with a population of approximately 470,000 in northern Delaware, including the City of Wilmington. The principal executive offices of Delmarva are located at 800 King Street, Wilmington, Delaware 19899 and its telephone number is (302) 429-3011.

Atlantic. Atlantic is a public utility holding company as defined in the 1935 Act and has claimed an exemption from substantially all of the provisions of the 1935 Act pursuant to Section 3(a) of the 1935 Act. Atlantic is the parent of Atlantic City Electric Company, a New Jersey corporation ("ACE"), which is a public utility primarily engaged in the generation, transmission, distribution and sale of electric energy. ACE provides electric service to approximately 473,000 customers in a service territory encompassing approximately 2,700 square miles within the southern one-third of New Jersey with the majority of customers being residential and commercial. The principal executive offices of Atlantic are located at 6801 Black Horse Pike, Egg Harbor Township, New Jersey 08234 and its telephone number is (609) 645-4500.

#### **Risk Factors**

For a description of the risk factors to be considered by stockholders in connection with the Mergers, see "Risk Factors."

#### **Stockholder Meetings**

*Delmarva*. At the Delmarva Meeting and any adjournment or adjournments thereof, the holders of Delmarva Common Stock will be asked to consider and vote upon (i) a proposal to approve the Merger Agreement and the transactions contemplated thereby and (ii) a proposal to approve the Company Incentive Compensation Plan (the "Company Plan," the provisions of which are set forth in Annex VII to this Joint Proxy Statement/Prospectus), approval of which has been recommended by the Delmarva Board and the Company Board. See "Meetings, Voting and Proxies—Delmarva Meeting."

The Delmarva Meeting is scheduled to be held at 10:00 a.m., local time, on January 30, 1997, at the Holiday Inn, 800 King Street, Wilmington, Delaware 19899. The Board of Directors of Delmarva (the "Delmarva Board") has fixed the close of business on December 18, 1996 as the record date (the "Delmarva Record Date") for the determination of holders of Delmarva Common Stock entitled to notice of, and to vote at, the Delmarva Meeting.

The Delmarva Board, by a unanimous vote of all of the directors, has approved the Merger Agreement and the Company Plan and the transactions contemplated therein and recommends that holders of Delmarva Common Stock vote FOR approval of the Merger Agreement and the Company Plan. Certain members of the Delmarva Board will become directors and/or employees of the Company following consummation of the Mergers and/or may become entitled to severance benefits as a result of the Mergers. See "The Mergers—Conflicts of Interest of Certain Persons in the Mergers" and "The Mergers—Employee Plans, Severance Arrangements and Agreements."

*Atlantic.* At the Atlantic Meeting and any adjournment or adjournments thereof, the holders of Atlantic Common Stock will be asked to consider and vote upon (i) a proposal to approve the Merger Agreement and the transactions contemplated thereby and (ii) a proposal to approve the Company Plan, approval of which has been recommended by the Atlantic Board and the Company Board. See "Meetings, Voting and Proxies—Atlantic Meeting."

The Atlantic Meeting is scheduled to be held at 3:00 p.m., local time, on January 30, 1997, at the Frank Guaracini, Jr. Fine & Performing Arts Center at Cumberland County College, Sherman Avenue and College Drive, Vineland, New Jersey 08360. The Board of Directors of Atlantic (the "Atlantic Board") has fixed the close of business on December 18, 1996 as the record date (the "Atlantic Record Date") for the determination of holders of Atlantic Common Stock entitled to notice of, and to vote at, the Atlantic Meeting.

The Atlantic Board, by unanimous vote, has approved the Merger Agreement and, by unanimous vote of all of the directors present, the Company Plan and the transactions contemplated therein and recommends that holders of Atlantic Common Stock vote FOR approval of the Merger Agreement and the Company Plan. Certain members of the Atlantic Board will become directors and/or employees of the Company following the consummation of the Mergers and/or may be entitled to severance benefits as a result of the Mergers. See "The Mergers—Conflicts of Interest of Certain Persons in the Mergers" and "The Mergers—Employee Plans, Severance Arrangements and Agreements."

# **Required Vote**

The affirmative vote of the holders of more than two-thirds of all votes entitled to be cast by all holders of Delmarva Common Stock is required for the approval of the Merger Agreement and the transactions contemplated thereby. The affirmative vote of the holders of a majority of Delmarva Common Stock present in person or by proxy and entitled to vote is required to approve the Company Plan. On the Delmarva Record Date, 60,754,568 shares of Delmarva Common Stock were outstanding and entitled to vote. As of the Delmarva Record Date, directors, executive officers and their affiliates owned less than 1% of the issued and outstanding shares of Delmarva Common Stock. See "Meetings, Voting and Proxies—Delmarva Meeting."

The affirmative vote of a majority of all votes cast by the holders of Atlantic Common Stock is required for the approval of the Merger Agreement and the transactions contemplated thereby. The affirmative vote of at least a majority of all votes cast by all holders of Atlantic Common Stock entitled to vote is required to approve the Company Plan. On the Atlantic Record Date, 52,704,052 shares of Atlantic Common Stock were outstanding and entitled to vote. As of the Atlantic Record Date, directors, executive officers and their affiliates owned less than 1% of the issued and outstanding shares of Atlantic Common Stock. See "Meetings, Voting and Proxies—Atlantic Meeting."

2

# The Mergers

The Merger Agreement provides for: (i) the merger of DS Sub with and into Delmarva in the Delmarva Merger; (ii) the merger of Atlantic with and into the Company in the Atlantic Merger; (iii) the cancellation of the Delmarva Common Stock owned by Delmarva, any subsidiary of Delmarva, Atlantic or any subsidiary of Atlantic; (iv) the cancellation of the Atlantic Common Stock owned by Atlantic, any subsidiary of Atlantic, Delmarva or any subsidiary of Delmarva; (v) the conversion of shares of Delmarva Common Stock issued and outstanding immediately prior to the consummation of the Delmarva Merger (except shares that are cancelled) into shares of Company Common Stock; (vi) the conversion of all issued and outstanding shares of Atlantic Common Stock (except shares that are cancelled) into shares of Company Common Stock, Class A Common Stock and cash in lieu of fractional shares; (vii) the conversion of all issued and outstanding shares of common stock of DS Sub into shares of Delmarva Common Stock; and (viii) the cancellation of all shares of capital stock of the Company that are issued and outstanding immediately prior to the Mergers. As a result of the Mergers, holders of Delmarva Common Stock and Atlantic Common Stock will become holders of Company Common Stock and, in the case of Atlantic Common Stock, Class A Common Stock. Each share of Delmarva Preferred Stock outstanding at the time of the consummation of the Mergers will remain outstanding as preferred stock of Delmarva and the rights and preferences of the Delmarva Preferred Stock will not be affected by the Mergers. See "The Merger Agreement-The Mergers."

Pursuant to the Merger Agreement: (a) in connection with the Delmarva Merger, (i) a certificate of merger complying with the General Corporation Law of the State of Delaware (the "DGCL") will be executed by Delmarva and filed with the Secretary of State of the State of Delaware and (ii) articles of merger complying with the Virginia Stock Corporation Act ("VSCA") will be executed by Delmarva and filed with the Secretary of Corporation Act ("VSCA") will be executed by Delmarva and filed with the Virginia State Corporation Commission (the "VSCC"); and (b) in connection with the Atlantic Merger, (i) a certificate of merger complying with the DGCL will be executed by the Company and filed with the Secretary of State of the State of Delaware and (ii) a certificate of merger complying with the New Jersey Business Corporation Act (the "NJBCA") will be executed by Atlantic and the Company and filed with the Secretary of State of the State of New Jersey, each on the second business day immediately following the satisfaction or waiver of all conditions to the Mergers, or at such other time as both Delmarva and Atlantic will agree. The Mergers will become effective simultaneously and at the time specified in the respective certificates and articles of merger for the Mergers. The "Effective Time" means the time and date that the Delmarva Merger and the Atlantic Merger become effective. See "The Merger Agreement—The Mergers." The Company will become a holding company required to be registered under the 1935 Act. See "Regulatory Matters—Public Utility Holding Company Act of 1935."

The relationship among the Company, Delmarva, Atlantic and the direct subsidiaries of Delmarva and Atlantic before and after the Mergers are illustrated, respectively, in the following diagrams: **BEFORE THE MERGERS** Delmarva Power & Light Company Atlantic Energy, Inc. 50% \$50% ê Delmarva Delmarva Delmarva Atlantic City Service Delmarva Atlantic Atlantic Confidence Capital Energy Industries Services Electric Energy Enterprises Energy Investments International The to be merged Company DS Sub to be merged FOLLOWING THE MERGERS The Company Atlantic City Atlantic Atlantic Delmarva Power & Light Company Electric Energy Enterprises Energy International Service Delmarva Delmarva Delmarva Delmarva Confidence Energy Capital Industries Services Investments

As illustrated in the diagrams, upon consummation of the Mergers (i) Delmarva will become a direct, wholly owned subsidiary of the Company and Delmarva's subsidiaries will become indirect subsidiaries of the Company and (ii) Atlantic will cease to exist and Atlantic's direct subsidiaries, including ACE, will become direct subsidiaries of the Company.

# **Exchange of Stock Certificates**

On or before the consummation of the Mergers, the Company will deposit with a bank, trust company or other agent selected by Delmarva and Atlantic (the "Exchange Agent") certificates representing shares of Company Common Stock and Class A Common Stock to effect the conversion of Delmarva Common Stock or Atlantic Common Stock, as the case may be, into Company Common Stock and, in the case of Atlantic Common Stock, Class A Common Stock.

As soon as practicable after the consummation of the Mergers, the Exchange Agent will mail, to each holder of record of Delmarva Common Stock or Atlantic Common Stock eligible for exchange at the consummation of the Mergers, transmittal instructions advising the holder of the procedure for surrendering Delmarva Common Stock certificates or Atlantic Common Stock certificates for Company Common Stock certificates and, in the case of Atlantic Common Stock, Class A Common Stock certificates. Holders of certificates that prior to the consummation of the Mergers represented shares of Delmarva Common Stock or Atlantic Common Stock will not be entitled to vote or receive any payment of dividends or other distributions on or payment for any fractional shares of Company Common Stock or Class A Common Stock until such certificates are delivered for exchange. See "The Merger Agreement—The Mergers."

# Conversion Ratios for Delmarva Common Stock and Atlantic Common Stock

Each share of Delmarva Common Stock outstanding immediately prior to the Effective Time will, upon consummation of the Delmarva Merger, be converted into the right to receive one share of Company Common Stock. Each share of Atlantic Common Stock outstanding immediately prior to the Effective Time will, upon consummation of the Atlantic Merger, be converted into the right to receive 0.75 shares of Company Common Stock and 0.125 shares of Class A Common Stock. Fractional shares of Company Common Stock and Class A Common Stock will not be issued. Holders of Atlantic Common Stock eligible for exchange will receive cash in lieu of fractional shares. See "The Merger Agreement—Consummation of the Mergers."

#### **Company Incentive Compensation Plan**

Subsequent to the execution of the Merger Agreement, Delmarva and Atlantic, as the stockholders of the Company, determined that it would be in the best interests of the Company for the Company to adopt the Company Plan, which upon consummation of the Mergers will replace the Delmarva Long-Term Incentive Plan (the "Delmarva LTIP") and Atlantic's Equity Incentive Plan, subject to approval by stockholders. The Delmarva Board and the Atlantic Board each approved the Company Plan as of December 12, 1996, which by its terms is subject to the approval of the stockholders of Delmarva and Atlantic. The Company Plan is a comprehensive cash and stock compensation plan providing for the grant of annual incentive awards, as well as long-term incentive awards such as restricted stock, stock options, stock appreciation rights, performance units and dividend equivalents, and any other types of awards as the committee of the Board which will administer the Company Plan (the "Company Plan Committee") deems appropriate. The maximum number of shares of Company Common Stock available for issuance under the Company Plan is 5,000,000. The Delmarva Board and the Atlantic Board determined to recommend that the respective stockholders of Delmarva and Atlantic vote FOR approval of the Company Plan. For a description of the Company Plan, see "The Mergers—The Company Benefit Plans" and "Approval of the Company Incentive Compensation Plan."

# **Background of the Mergers**

For a description of the background of the Mergers, see "The Mergers-Background of the Mergers."

#### **Reasons** for the Mergers

Delmarva and Atlantic share a common vision of the future as well as a similar view of the strategic path necessary to succeed in the competitive marketplace and believe that the Mergers will provide opportunities to achieve benefits for their respective stockholders, customers, employees and communities that would not be available if they were to remain separate companies. The benefits to be achieved through the Mergers include: increased scale; cost savings; competitive prices and services; and a more balanced customer base. In addition, the combined companies will have increased financial flexibility and greater access to the regional market. For a more detailed description of the reasons for the Mergers, see "The Mergers—Reasons for the Mergers."

# **Recommendations of the Boards of Directors**

Delmarva. The Delmarva Board unanimously approved the Merger Agreement and the transactions contemplated thereby and determined to recommend that the stockholders of Delmarva vote FOR approval of the Merger Agreement. The Delmarva Board approved the Merger Agreement after consideration of a number of factors, which are described under the heading "The Mergers—Recommendations of the Boards of Directors—Delmarva." Certain of the members of the Delmarva Board will become directors and/or employees of the Company following consummation of the Mergers and/or may become entitled to severance benefits as a result of the Mergers. Therefore, such directors have interests in the Merger Agreement that are in addition to the interests of stockholders of Delmarva generally and that could potentially represent conflicts of interest. The Delmarva Board was aware of these interests and considered them, among other matters, in approving the Merger Agreement. See "The Mergers—Conflicts of Interest of Certain Persons in the Mergers" and "The Mergers—Employee Plans, Severance Arrangements and Agreements."

Atlantic. The Atlantic Board unanimously approved the Merger Agreement and the transactions contemplated thereby and determined to recommend that the stockholders of Atlantic vote FOR approval of the Merger Agreement. The Atlantic Board approved the Merger Agreement after consideration of a number of factors, which are described under the heading "The Mergers—Recommendations of the Boards of Directors—Atlantic." Certain of the members of the Atlantic Board will become directors and/or employees of the Company following consummation of the Mergers and/or may become entitled to severance benefits as a result of the Mergers. Therefore, such directors have interests in the Merger Agreement that are in addition to the interests of stockholders of Atlantic generally and that could potentially represent conflicts of interest. The Atlantic Board was aware of these interests and considered them, among other matters, in approving the Merger Agreement. See "The Mergers—Conflicts of Interest of Certain Persons in the Mergers" and "The Mergers—Employee Plans, Severance Arrangements and Agreements."

# **Opinions of Investment Bankers**

Delmarva. Merrill Lynch, Pierce, Fenner & Smith Incorporated ("Merrill Lynch") delivered to the Delmarva Board its written opinions dated August 9, 1996 and December 26, 1996, each to the effect that, as of the date of each such opinion, the Delmarva Conversion Ratio (as defined herein under "The Merger Agreement—Consummation of the Mergers") contemplated by the Merger Agreement is fair to Delmarva and its stockholders from a financial point of view. The written opinion of Merrill Lynch dated December 26, 1996 is attached to this Joint Proxy Statement/Prospectus as Annex II and should be read in its entirety. For a description of the matters considered and assumptions made by Merrill Lynch in reaching its opinions and the fees received and to be received by Merrill Lynch, see "The Mergers—Opinion of Financial Advisor to Delmarva" and Annex II.

Atlantic. Morgan Stanley & Co. Incorporated ("Morgan Stanley") delivered to the Atlantic Board its oral opinion on August 9, 1996, which was confirmed in written opinions dated August 9, 1996 and December 26, 1996, to the effect that, as of the respective dates of such opinions and based upon the procedures and subject to the assumptions described therein, the Atlantic Conversion Ratio (as defined herein under "The Merger Agreement—Consummation of the Mergers") taking into account the Delmarva Conversion Ratio is fair to the holders of Atlantic Common Stock from a financial point of view. The written opinion of Morgan Stanley dated December 26, 1996 is attached to this Joint Proxy Statement/ Prospectus as Annex III and should be read in its entirety. For a description of matters considered and

assumptions made by Morgan Stanley in reaching its opinions and the fees received and to be received by Morgan Stanley, see "The Mergers—Opinion of Financial Advisor to Atlantic" and Annex III.

# **Conflicts of Interest of Certain Persons in the Mergers**

*Directorships.* The Merger Agreement provides that the Delmarva Board will be entitled to nominate ten members to the Company Board and the Atlantic Board will be entitled to nominate eight members to the Company Board. Each member of the Delmarva Board and the Atlantic Board serving in such capacity immediately prior to the Effective Time will be given the opportunity to serve on the Company Board. See "The Mergers—Conflicts of Interest of Certain Persons in the Mergers—Board of Directors."

Severance Agreements. The Delmarva Board and the Atlantic Board have each adopted agreements providing for severance benefits to the respective Delmarva and Atlantic employees designated as parties to such agreements. The agreements will be binding on the Company. A total of 48 Delmarva employees and 13 Atlantic employees are parties to such agreements. If payments under such agreements to all eligible employees were required to be made on the date of this Joint Proxy Statement/Prospectus, the aggregate cost to Delmarva would be approximately \$16.6 million, and the aggregate cost to Atlantic (including the cost of early vesting under employee plans) would be approximately \$22.4 million. See "The Mergers—Employee Plans, Severance Arrangements and Agreements."

*Employee Plans.* Under certain benefit plans, severance arrangements and other employee agreements maintained, or entered into, by Delmarva and Atlantic, certain benefits may become vested, and certain payments may become payable, in connection with the Mergers. The cost resulting from early vesting of benefits of Delmarva and Atlantic employees who have severance agreements is included in the severance costs described in the preceding paragraph. See "The Mergers—Employee Plans, Severance Arrangements and Agreements."

*Indemnification.* From and after the Effective Time, the Company will, to the fullest extent not prohibited by applicable law, indemnify, defend and hold harmless the present and former officers, directors and management employees of Delmarva and Atlantic against all losses, expenses (including reasonable attorneys' fees), claims, damages or liabilities or, subject to certain restrictions, amounts paid in settlement arising out of actions or omissions occurring at or prior to or after the Effective Time that are in whole or in part based on, or arising out of, the fact that such person is or was a director, officer or management employee of such party or arising out of or pertaining to the transactions contemplated by the Merger Agreement. See "The Merger Agreement—Indemnification."

# **Certain Federal Income Tax Consequences**

A condition precedent to consummation of the Mergers is the receipt of opinions of counsel substantially to the effect that the Mergers will be treated as a tax-free reorganization under Section 368(a) of the Internal Revenue Code of 1986, as amended (the "Code") or an exchange under Section 351 of the Code. Assuming the Mergers so qualify, then for federal income tax purposes (i) no gain or loss will be recognized by Atlantic, Delmarva or the Company as a result of the Mergers, (ii) holders of Delmarva Common Stock whose shares are converted into Company Common Stock and holders of Atlantic Common Stock whose shares are converted into Company Common Stock and Class A Common Stock in the Mergers will recognize gain or loss as a result of the extent they receive cash in lieu of fractional shares) and (iii) the holding period and basis applicable to shares of Company Common Stock and Class A Common Stock or Atlantic Common Stock that was converted into Company Common Stock and basis attributable to the Delmarva Common Stock in the Mergers will be the same as the holding period and basis attributable to the Delmarva Common Stock in the Mergers (reduced by any amount allocable to a fractional share interest in Company Common Stock and Class A Common Stock in the Mergers (reduced by any amount allocable to a fractional share interest in Company Common Stock and Class A Common Stock in the Mergers (reduced by any amount allocable to a fractional share interest in Company Common Stock and Class A Common Stock for which cash is received). Although it is the

opinion of Simpson Thacher & Bartlett that the Class A Common Stock will be treated as stock of the Company for federal income tax purposes, there are no federal income tax regulations, court decisions or published Internal Revenue Service ("IRS") rulings bearing directly on this issue. If the Class A Common Stock were determined not to be stock of the Company, the receipt of the Class A Common Stock in the Atlantic Merger would be taxable to the Atlantic Stockholders. See "The Mergers—Certain Federal Income Tax Consequences."

EACH DELMARVA AND ATLANTIC STOCKHOLDER IS URGED TO CONSULT HIS OR HER OWN TAX ADVISOR TO DETERMINE THE PARTICULAR TAX CONSEQUENCES TO HIM OR HER OF THE MERGERS, INCLUDING THE APPLICATION AND EFFECT OF FEDERAL, STATE, LOCAL AND FOREIGN TAX LAWS.

# **Accounting Treatment**

The Mergers will be accounted for under the purchase method of accounting for financial reporting purposes. See "The Mergers—Accounting Treatment."

# Appraisal Rights

Dissenters' or other appraisal rights are not available to holders of Delmarva Common Stock, Delmarva Preferred Stock or Atlantic Common Stock in connection with the transactions contemplated by the Merger Agreement. See "The Mergers—Appraisal Rights."

# **Regulatory Matters**

The approvals of the SEC under the 1935 Act, the Nuclear Regulatory Commission (the "NRC") under the Atomic Energy Act of 1954, as amended (the "Atomic Energy Act"), the Federal Energy Regulatory Commission (the "FERC") under the Federal Power Act, as well as the approvals of the Delaware, Virginia, Maryland, New Jersey and Pennsylvania utility commissions under applicable state laws and the expiration or termination of the applicable waiting period under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "HSR Act"), are required to consummate the Mergers. See "Regulatory Matters."

Upon consummation of the Mergers, the Company must register as a holding company under the 1935 Act. The 1935 Act imposes restrictions on the operations of registered holding company systems. Among these are requirements that securities issuances, sales and acquisitions of utility assets or of securities of utility companies and acquisitions of interests in any other business be approved by the SEC. The 1935 Act also limits the ability of registered holding companies to engage in nonutility ventures and regulates holding company system service companies and the rendering of services by holding company affiliates to the system's utilities. Delmarva, Atlantic and the Company recognize that the divestiture of the existing gas operations of Delmarva and certain nonutility operations of Delmarva and Atlantic is a possibility under the new registered holding company structure, but the Company will request in its 1935 Act application that it be allowed to retain the gas utility operations of Delmarva and the nonutility operations of Delmarva and Atlantic or, in the alternative, that the question of divestiture be deferred. If divestiture is ultimately required, the SEC historically has allowed companies sufficient time to accomplish divestiture in a manner that protects stockholder value. See "Regulatory Matters—Public Utility Holding Company Act of 1935."

For a discussion of certain regulatory trends and potential effects on Delmarva, ACE and the Company, see "Regulatory Matters-Effects of Certain Regulatory Trends."

# **Conditions to the Mergers**

The obligations of Delmarva and Atlantic to consummate the Mergers are subject to the satisfaction of certain conditions, including: the approval of the stockholders of each of Delmarva and Atlantic; the absence of any restraining order, injunction or applicable law or regulation preventing or prohibiting the Mergers; the effectiveness of the Registration Statement; the approval of the listing on the NYSE of the Company Common Stock and the Class A Common Stock upon official notice of issuance; the receipt of all material statutory approvals as final orders, without such approvals having a material adverse effect on Delmarva or Atlantic; the expiration or termination of the applicable waiting period under the HSR Act; the performance of the agreements and covenants required to be performed under the Merger Agreement in all material respects; the accuracy of the representations and warranties set forth in the Merger Agreement in all material respects; the absence of any material adverse effect on the business, operations or results of operations of Delmarva or Atlantic; the receipt of material third party consents; the receipt by each of Delmarva and Atlantic of an officer's certificate from the other relating to the satisfaction of conditions in the Merger Agreement; the receipt of tax opinions; and the receipt of certain agreements from affiliates within the meaning of SEC Rule 145, if any, of Delmarva and Atlantic. See "The Merger Agreement—Conditions to the Mergers."

# **Rights to Terminate, Amend or Waive Conditions**

The Merger Agreement may be terminated under certain circumstances, including: by mutual written consent of the Delmarva Board and the Atlantic Board; by Delmarva or Atlantic if the Mergers are not consummated by 18 months after the signing of the Merger Agreement (or by 30 months after the signing of the Merger Agreement under certain conditions described under "The Merger Agreement-Termination"); by Delmarva or Atlantic if the requisite stockholder approvals are not obtained; by Delmarva or Atlantic if a law, order, rule or regulation is issued or adopted that has the effect of prohibiting the Mergers, or if any final and nonappealable action by a court of competent jurisdiction prohibits either of the Mergers; in connection with a third-party acquisition proposal by the recipient of the third-party acquisition proposal, if the Board of Directors of the party receiving the acquisition proposal determines in good faith that acceptance of the acquisition proposal is necessary for the Board of Directors to act consistent with its fiduciary duties; by Delmarva or Atlantic if there occurs a material breach of the Merger Agreement by the other party that is not cured within 20 days after receipt of notice; or by Delmarva or Atlantic if the other party shall not have complied with certain agreements and covenants or if its Board of Directors withdraws or modifies in an adverse manner or does not reaffirm upon request its recommendation to its stockholders in favor of the Merger Agreement and the transactions contemplated thereby or recommends a different acquisition proposal than contemplated by the parties; or by Delmarva or Atlantic if a third party acquires greater than 50% of the voting power of the other party or certain changes in the composition of the board of the other party occur.

The Merger Agreement requires certain fees to be paid upon termination of the Merger Agreement under certain circumstances. The aggregate termination fees under these provisions may not exceed \$40 million (including out-of-pocket expenses of up to \$10 million). See "The Merger Agreement—Termination Fees."

The Merger Agreement may be amended by Delmarva or Atlantic pursuant to action by the Delmarva Board and the Atlantic Board, respectively, at any time before or after its approval by the holders of Delmarva Common Stock and Atlantic Common Stock, respectively, but after any such approval no amendment may be made that alters or changes (i) the amount or kind of shares, securities or cash to be received or exchanged for or on the conversion of shares of Delmarva or Atlantic, (ii) the terms or conditions of the Merger Agreement if such alterations or changes, alone or in the aggregate, would adversely affect the rights of the holders of any class or series of stock of Delmarva or Atlantic or (iii) the certificate of incorporation of the Company or Delmarva, except for alterations or changes that could otherwise be adopted by the Company Board without the further approval of such stockholders. See "The Merger Agreement—Amendment and Waiver."

At any time prior to the Effective Time, the time of performance of any obligation or other acts may be extended or any inaccuracies in the representations and warranties or conditions to a party's obligation to consummate the Mergers under the Merger Agreement may be waived by the other parties thereto.

# **Common Equity of the Company**

The common equity of the Company will be divided into two classes: the Company Common Stock and the Class A Common Stock. The use of two classes of common stock is designed to address the difference in Delmarva's and Atlantic's evaluations of the growth prospects of, and uncertainties associated with deregulation of, the regulated electric utility business of Atlantic. Upon the consummation of the Mergers, the Company Common Stock will be issued both to the holders of the Delmarva Common Stock and to the holders of the Atlantic Common Stock and the Class A Common Stock will be issued to the holders of the Atlantic Common Stock, thereby giving the current holders of Atlantic Common Stock a proportionately greater opportunity to share in the growth prospects of, and a proportionately greater exposure to the uncertainties associated with deregulation of, the regulated electric utility business of Atlantic.

The principal advantage of this structure is that it addresses the concerns of the managements of both Delmarva and Atlantic and allows the respective stockholders of Delmarva and Atlantic to gain the level of exposure to the growth prospects of, and uncertainties associated with deregulation of, the regulated electric utility business of Atlantic that the respective managements have deemed advisable. The level of such exposure is reflected primarily by (i) the amount of the Notional Fixed Charge (defined below) (\$40 million), (ii) the percentage of equity interest in the Company attributable to the Atlantic Utility Group represented by the outstanding Class A Common Stock (30%) and (iii) the percentages of Company Common Stock and Class A Common Stock issued to the Atlantic stockholders (39.4% and 100%) and to the Delmarva stockholders (60.6% and 0%). Through the ownership of 39.4% of the Company Common Stock, the Atlantic stockholders will have the benefit of 27.6% (39.4% of 70%) of the interest in the earnings attributable to the current regulated utility business of ACE in excess of \$40 million per year as well as a 39.4% of the interest in the first \$40 million of such earnings. Accordingly, together with the 30% interest in such earnings represented by the Class A Common Stock, Atlantic stockholders will retain more than half the benefits and risks related to such business. Such exposure will be realized by the respective stockholders to the extent that dividends on the Class A Common Stock will actually be based on, and to the extent that the market price of the Class A Common Stock will actually reflect, the earnings of the Atlantic Utility Group in excess of the Notional Fixed Charge, which are subject to significant uncertainties. The principal disadvantage of this structure is that it creates a complex capital structure for the Company which could have an adverse effect on the trading market for the shares of both the Class A Common Stock and the Company Common Stock. See "Risk Factors-Risk Factors Associated with the Class A Common Stock-No Assurance as to Market Price of Class A Common Stock" and "-Risk Factors Associated with the Company Common Stock-No Assurance as to Market Price of Company Common Stock." For a discussion of risk factors associated with this dual class capital structure, see "Risk Factors-Risk Factors Associated with the Dual Class Common Equity Structure."

The Class A Common Stock will at the time of the consummation of the Mergers represent 30% of the interest in the Company attributable to the Atlantic Utility Group (as defined below), subject to a \$40 million per annum charge (the "Notional Fixed Charge") described below. This is primarily due to the fact that the Company intends to base the amount of dividends on the Class A Common Stock on the earnings of the Atlantic Utility Group in excess of the Notional Fixed Charge, although the Company Board generally has discretion to examine all relevant considerations, including factors affecting the Company overall, in determining such amount. Delmarva and Atlantic have determined that certain businesses, assets and liabilities of ACE described below will be "attributed to," or included in, the Atlantic Utility Group for purposes of determining such Group's earnings in excess of the Notional Fixed Charge. However, holders of the Class A Common Stock will, like the holders of Company Common Stock, be common stockholders of the Company, will not have any specific rights or claims against such businesses, assets and liabilities, including upon liquidation of the Company, other than as common stockholders of the Company and will be subject to risks associated with an investment in the Company and all of its businesses, assets and liabilities. In addition, funds representing the earnings of the Atlantic Utility Group will not necessarily be held in any segregated accounts of such Group. See "Risk Factors—Risk Factors Associated with the Dual Class Common Equity Structure—Stockholders of One Company; Financial Effects on One Business Could Affect Other Businesses."

The Atlantic Utility Group will include the interest of the Company at the Effective Time in ACE, and solely those lines of business in which ACE was engaged as of August 9, 1996, and the assets and liabilities attributable to those lines of business, and which as of August 9, 1996 were: price regulated by the New Jersey Board of Public Utilities (the "NJBPU"); directly related to the supply of electricity (generation and purchase of electricity) or the delivery of electricity (transmission and distribution of electricity); and a line of business for which ACE has a franchise; all of the foregoing as further qualified by the Restated Certificate of Incorporation of the Company (the "Company Charter"), the provisions of which are set forth in Annex IV to this Joint Proxy Statement/Prospectus.

The Company Common Stock will represent the interest in the Company that is not represented by the Class A Common Stock. This will include (i) the interest in the businesses, assets and liabilities of the Company other than those of the Atlantic Utility Group and (ii) the interest in the Company attributable to the Atlantic Utility Group that is not represented by the outstanding shares of Class A Common Stock but is retained by the Company through its "Intergroup Interest" in the Atlantic Utility Group and through the Notional Fixed Charge.

Upon consummation of the Mergers, the Intergroup Interest will represent 70% of the interest in the Company attributable to the Atlantic Utility Group. Following the Mergers the Intergroup Interest will be subject to increase or decrease under the circumstances specified in the Company Charter. Should the 70% Intergroup Interest increase or decrease, the Class A Common Stock's 30% proportionate interest in the interest in the Company attributable to the Atlantic Utility Group will correspondingly decrease or increase accordingly.

Dividends on the Company Common Stock will be based on (i) the earnings of the Company other than those of the Atlantic Utility Group, (ii) earnings of the Atlantic Utility Group in proportion to the Intergroup Interest and (iii) earnings of the Atlantic Utility Group up to the amount of the Notional Fixed Charge.

The market prices of the Company Common Stock and the Class A Common Stock after the consummation of the Mergers would be determined in the trading markets and could be influenced by many factors, including the consolidated results of the Company, as well as the respective performances of the Residual Group (as defined under "Description of the Company's Capital Stock—Certain Definitions") and the Atlantic Utility Group (each a "Group" and, together, the "Groups"), investors' expectations for the Groups, trading volumes, the regulatory environment and general economic and market conditions. There can be no assurance as to the extent to which investors would assign values to the Company Common Stock and the Class A Common Stock based on the reported financial results or other measures of performance or prospects of the relevant businesses. There is no present intention on the part of Delmarva and Atlantic for the Company to issue any shares of Class A Common Stock following consummation of the Mergers.

See "Description of the Company's Capital Stock—Common Stock," "—The Class A Common Stock, the Atlantic Utility Group and the Notional Fixed Charge," "—Dividends," "—Intergroup Interest," "—Liquidation Rights," "—Certain Definitions—Atlantic Utility Group" and Annex IV to this Joint Proxy Statement/Prospectus.

It is expected that the Company Common Stock and the Class A Common Stock will be listed on the NYSE.

#### Dividends

It is anticipated that following the Mergers the Company will initially pay dividends on the Company Common Stock at the rate of \$1.54 per annum, subject to evaluation from time to time by the Company Board based on the Company's results of operations, financial condition, capital requirements and other relevant considerations. However, no assurance can be given that such dividend rate will be in effect or will remain unchanged, and the Company reserves the right to increase or decrease the dividend on the Company Common Stock as may be required by law or contract or as may be determined by the Company Board, in its discretion, to be advisable. For a description of certain restrictions on the Company's ability to pay dividends on Company Common Stock, see "Description of the Company's Capital Stock."

Subject to declaration by the Company Board and the obligation of the Company Board to consider the financial condition and regulatory environment of the Company and the results of its operations, the dividends declared and paid on the Class A Common Stock will be maintained at a level of \$3.20 per share per annum until the earlier of July 1, 2001, or the end of the twelfth calendar quarter following the calendar quarter in which the Effective Time occurs (the "Initial Period"). Thereafter, it is the intention of the Company, subject to declaration by the Company Board and the obligation of the Company Board to consider the financial condition and regulatory environment of the Company and the results of its operations, to pay annual dividends on the Class A Common Stock in an amount (the amount of such dividends to include the amount credited to the Intergroup Interest as described under "Description of the Company's Capital Stock-Intergroup Interest-Adjustments in Connection with Various Transactions—Dividends") equal to 90% of the Company Net Income Attributable to the Atlantic Utility Group, subject to the fact that if the annual dividends paid on the Class A Common Stock during the Initial Period (including the amount credited to the Intergroup Interest as aforesaid) shall have exceeded 100% of the Company Net Income Attributable to the Atlantic Utility Group during such period, the Company Board may consider such fact in determining the amount of future dividends, if any, on the Class A Common Stock. There can be no assurance that the Company's earnings attributable to the Atlantic Utility Group will be sufficient to cover dividends on the Class A Common Stock during the Initial Period. Dividends on the Class A Common Stock will not be cumulative.

The Company Net Income Attributable to the Atlantic Utility Group for the fiscal year ended December 31, 1995 and the 12-month period ended September 30, 1996 would have been \$44.1 million and \$31.8 million (*i.e.*, \$84.1 million and \$71.8 million minus the \$40 million Notional Fixed Charge), respectively. The Company Net Income Attributable to the Atlantic Utility Group for these same periods would have been \$57.6 million and \$49.1 million (*i.e.*, \$97.6 million and \$89.1 million minus the \$40 million Notional Fixed Charge), respectively, not taking into account the higher than expected costs associated with the Salem Nuclear Generating Station described under "Selected Information Concerning Delmarva and Atlantic—Salem Nuclear Generating Station" and the effects of foregone recovery of certain fuel costs under the Southern New Jersey Economic Initiative, which are not expected to recur, and the effects of abnormal weather.

The dividend policies with respect to the Class A Common Stock and the Company Common Stock are and will be subject to evaluation from time to time by the Company Board based on the Company's results of operations, financial condition and capital requirements and other relevant considerations, including regulatory considerations. See "Description of the Company's Capital Stock—Common Stock—Dividends—Dividend Policies."

# **Comparison of Corporate Charters and Rights of Security Holders**

As a result of the Mergers, holders of Delmarva Common Stock and Atlantic Common Stock who receive shares of Company Common Stock and, in the case of holders of Atlantic Common Stock, Class A Common Stock will become stockholders of the Company and will have certain rights as stockholders of the Company different from those they had as stockholders of Delmarva or Atlantic. For comparisons of the DGCL to the NJBCA and the VSCA and of the provisions in the articles or certificate of incorporation and bylaws of Delmarva, Atlantic and the Company governing the rights of stockholders of Delmarva, Atlantic and the Company of Corporate Charters and Rights of Security Holders."

# Selected Information Concerning Delmarva and Atlantic

For recent developments with respect to the Salem Nuclear Generating Station and other information concerning Delmarva and Atlantic, see "Selected Information Concerning Delmarva and Atlantic."

# The Company Following the Mergers

Following the consummation of the Mergers, the Company will maintain its corporate headquarters and principal executive offices in Wilmington, Delaware, and the Company will maintain a significant presence in New Jersey. After the consummation of the Mergers, the Company will provide charitable contributions and community support within the service areas of Delmarva and Atlantic and each of their subsidiaries at levels substantially comparable to historical levels of charitable contributions and community support provided by Delmarva and Atlantic prior to the consummation of the Mergers. The business of the Company will be to operate as a holding company for its utility subsidiaries and various nonutility subsidiaries. See "The Company Following the Mergers."

Following the consummation of the Mergers, the Company and its subsidiaries will honor all prior contracts, agreements, collective bargaining agreements and commitments with current or former employees or current or former directors of Delmarva or Atlantic and their respective subsidiaries, in accordance with the respective terms of such contracts, agreements and commitments, subject to the Company's right to enforce them in accordance with their terms (including any reserved right to amend, modify, suspend, revoke or terminate them). The Company will take all action necessary so that, after the consummation of the Mergers, the Dividend Reinvestment and Common Share Purchase Plan, the Savings and Thrift Plan and the Long-Term Incentive Plan of Delmarva and the Equity Incentive Plan, the Directors' Restricted Stock Plan, the Employee Stock Purchase Plan and the Dividend Reinvestment Plan of Atlantic will be terminated, replaced or amended to provide for the issue and sale of Company Common Stock in place of Delmarva Common Stock or Atlantic Common Stock, as the case may be, under such plans. See "The Merger Agreement—Benefit Plans."

Both the holders of Company Common Stock and the holders of Class A Common Stock will receive the consolidated financial statements of the Company. Since upon consummation of the Mergers the financial results of ACE will be substantially identical to the financial results for the Atlantic Utility Group, the notes to the consolidated financial statements of the Company will at such time include condensed financial information of ACE, including a reconciliation of ACE's Income Available to Common Stockholders to the Earnings Applicable for Class A Common Stock. Complete financial statements of ACE will continue to be filed under the Exchange Act and will be available to stockholders upon request.

13

# **RISK FACTORS**

Stockholders should carefully consider the following factors, in addition to the other information contained elsewhere in this Joint Proxy Statement/Prospectus, the Annexes hereto and the documents incorporated by reference or otherwise referred to herein, in connection with the Mergers.

# Risk Factors Associated with the Dual Class Common Equity Structure

Stockholders of One Company; Financial Effects on One Business Could Affect Other Businesses. Notwithstanding the attribution of assets and liabilities, equity and items of income and expense between the Atlantic Utility Group and the Residual Group (each as defined under "Description of the Company's Capital Stock—Certain Definitions"; each a "Group" and, together, the "Groups"), the capital structure of the Company contemplated by the Merger Agreement will not affect legal title to such assets or responsibility for such liabilities of the Company or any of its subsidiaries. Holders of Company Common Stock and of Class A Common Stock will be common stockholders of the Company and will be subject to risks associated with an investment in the Company and all of its businesses, assets and liabilities. Financial effects arising from either Group that affect the Company's consolidated results of operations or financial condition could, if significant, affect the combined results of operations or financial condition of the other Group and the market price of both the Company Common Stock and the Class A Common Stock.

The financial condition of each Group will principally reflect the financial condition of the businesses included therein. However, the Residual Group's financial condition will reflect the fact that the Residual Group will hold an undivided interest in the individual businesses, assets and liabilities of the Atlantic Utility Group equal to the Intergroup Interest Fraction, which at the consummation of the Mergers will equal 70%. See "Description of the Company's Capital Stock—Intergroup Interest."

Limited Separate Stockholder Rights; Effects on Voting Power. From and after the consummation of the Mergers, holders of Company Common Stock and Class A Common Stock would not be provided any rights specifically related to their corresponding Groups or have any right to vote on matters as a separate class, other than (i) as set forth in the Company Charter and described under "Description of the Company's Capital Stock—Common Stock" and (ii) separate voting rights in limited circumstances as required by the DGCL. Separate meetings for the holders of Company Common Stock or Class A Common Stock would not be held.

From and after the consummation of the Mergers, holders of Company Common Stock, Class A Common Stock and any preferred stock, par value \$.01 per share, of the Company (the "Preferred Stock") having general voting rights would vote as one class on all matters coming before any meeting of stockholders and would not have any separate class voting rights except in limited circumstances as required by the DGCL and as provided in the Company Charter in the case of any Preferred Stock. Under the DGCL, the holders of the outstanding shares of a class are entitled to vote as a class upon a proposed amendment to a corporation's certificate of incorporation, whether or not entitled to vote on such amendment by the certificate of incorporation, if the amendment would alter or change the powers, preferences or special rights of such class so as to affect them adversely. The DGCL does not provide for any other separate voting rights of a class of capital stock (other than with respect to a change in par value or an increase or decrease in the authorized shares of such class).

Certain matters on which holders of Company Common Stock and of Class A Common Stock would vote together as a single class could involve a divergence or the appearance of a divergence of interests between the holders of Company Common Stock and the holders of Class A Common Stock. For example, the Company Charter does not require that a merger or consolidation of the Company requiring the approval of the Company's stockholders be approved by a separate vote of holders of either class of Common Stock. As a result, if holders of any one or more classes of Common Stock that possess the requisite voting power vote to approve the merger or consolidation, the merger or consolidation could be consummated even if the holders of a majority of some other class of Common Stock vote against the merger or consolidation. See "—Potential Divergence of Interests and Requirement to Act on Behalf of Company and All Stockholders" and "—Management and Allocation Policies Not Formulated; Policies Subject to Revision." The holders of Common Stock will not be entitled to cumulative votes for the election of directors. Except as may otherwise be required by the laws of the State of Delaware, the holders of Company Common Stock and the holders of Class A Common Stock will vote as one class for all purposes. Neither the holders of Company Common Stock nor the holders of Class A Common Stock will have any rights to vote as a separate class on any matters submitted to a vote at a meeting of stockholders except with respect to certain limited class voting rights provided under the DGCL.

If the Merger Agreement is approved by the stockholders of Delmarva and Atlantic and the Mergers are consummated, the Class A Common Stock will have one vote per share and the Company Common Stock will have one vote per share. Based on the shares of Delmarva Common Stock and Atlantic Common Stock outstanding as of the close of business on September 30, 1996, the Company Common Stock would represent approximately 94.2%, and the Class A Common Stock would represent approximately 5.8%, of the voting power of all classes entitled to vote on matters presented to stockholders. See "Description of Company's Capital Stock—Voting."

Potential Divergence of Interests and Requirement to Act on Behalf of Company and All Stockholders. In certain instances where the interests of the holders of Company Common Stock and the holders of Class A Common Stock may diverge or appear to diverge, the Company Board will be required to act on behalf of the Company and its stockholders taken as a whole. As further discussed below, examples include determinations by the Company Board to (i) convert each outstanding share of Class A Common Stock into shares of Company Common Stock, (ii) approve the disposition of all or substantially all of the properties and assets of the Atlantic Utility Group, (iii) allocate resources and financial support to or pursue business opportunities or operational strategies through one Group instead of the other Group, (iv) allocate the proceeds of issuances of Class A Common Stock either to the Residual Group in reduction in the Intergroup Interest or to the Atlantic Utility Group, (v) pay or omit to pay dividends on Company Common Stock or Class A Common Stock or (vi) approve transactions involving the transfer of funds or assets from one Group to the other Group or make other operational or financial decisions with respect to one Group that could be considered to be detrimental to the other Group. When making decisions with regard to matters with respect to which the holders of the Company Common Stock and the holders of the Class A Common Stock would have potentially divergent interests, the Company Board would act in accordance with the terms of the Company Charter, its fiduciary duties and, to the extent applicable and consistent with its fiduciary duties, the management and accounting policies, if any, adopted by the Company Board as discussed under "The Company Following the Mergers-Management of the Company." See "-Fiduciary Duties of the Company Board Are to All Stockholders Regardless of Class."

Allocation of Resources and Financial Support; Pursuit of Operational Strategies. The Company Board could from time to time allocate resources and financial support to or pursue operational strategies through one Group instead of the other Group. The decision to allocate resources and financial support to one Group may adversely affect the ability of the other Group to obtain funds sufficient to implement its business strategies. Any such decision may favor one Group at the expense of the other. For example, the decision to obtain funds for one Group may adversely affect the ability of the other Group to obtain funds sufficient to implement its growth strategies. All such decisions will be made by the Company Board in its good faith business judgment and in accordance with procedures and policies adopted by the Company Board from time to time, including the policies described under "The Company Following the Mergers—Certain Management Policies," to ensure that such decisions will be made in a manner consistent with the best interests of the Company and all of its stockholders, including both the holders of the Class A Common Stock and the holders of the Company Common Stock. Following the Effective Time, the Audit Committee of the Company Board will be charged with the responsibility of advising the Company Board with respect to certain intercompany transactions and other fiduciary matters that may relate to the Class A Common Stock. See "The Company Following the Mergers—Management of the Company."



: . .

Fiduciary Duties of the Company Board Are to All Stockholders Regardless of Class. Under Delaware law, the Company Board has a duty to act with due care and in the best interests of the Company as a whole and the entire body of stockholders, including the holders of Company Common Stock and Class A Common Stock. Although the Company is not aware of any precedent concerning the manner in which principles of Delaware law would be applied in the context of a capital structure involving multiple classes or series of capital stock the rights of which include terms designed to reflect the separate performance of specified businesses, principles of Delaware law provide that a board of directors must act in accordance with its good faith business judgment of the corporation's best interests, taking into consideration the interests of the entire body of stockholders regardless of class or series. Under these principles of Delaware law and the "business judgment rule," a good faith determination made by a disinterested and adequately informed Company Board with respect to any matter having a disparate impact upon the holders of Company Common Stock and the holders of Class A Common Stock would be a defense to any challenge to such a determination. Nevertheless, a Delaware court hearing a case involving such a challenge may decide to apply principles of Delaware law other than those discussed above or may fashion new principles of Delaware law, in order to decide such a case, which would be a case of first impression. There may arise circumstances involving a divergence of interests in which the Company Board is held to have properly discharged its responsibilities to act with due care and in the best interests of the Company and the entire body of stockholders but in which holders of either the Company Common Stock or the Class A Common Stock consider themselves to be disadvantaged relative to the other class. In such a case, such holders would not have any remedy under Delaware law with respect to the circumstances giving rise to the divergence of interests.

Disproportionate ownership interests of members of the Company Board in the Company Common Stock and the Class A Common Stock could create or appear to create potential conflicts of interest when directors are faced with decisions that could have different implications for different series. See "— Potential Divergence of Interests." The existing benefit plans of Delmarva and Atlantic will be converted to plans of the Company that will provide for the issuance of Company Common Stock and not Class A Common Stock. See "The Company Following the Mergers—Operations of the Company" and "Approval of the Company Incentive Compensation Plan." Nevertheless, the Company believes that a director would be able to discharge his or her fiduciary responsibilities even if his or her interests in shares of Company Common Stock and Class A Common Stock were disproportionate or had disparate values.

Management and Allocation Policies Not Formulated; Policies Subject to Revision. Except for the provision contained in the Merger Agreement pursuant to which the Audit Committee of the Company Board will advise the Company Board with respect to certain intercompany transactions and other fiduciary matters that may be related to the Class A Common Stock (see "The Company Following the Mergers—Management of the Company"), Delmarva and Atlantic have determined not to adopt specific management and allocation policies with respect to cash management, corporate expenses, allocation of assets and liabilities and inter-Group transactions at the present time. Rather than develop policies at this time, the Company Board intends to exercise from time to time its judgment, as to how best to obtain information regarding the divergence (or potential divergence) of interests, under what circumstances to seek the assistance of outside advisers, and how to assess which available alternative is in the best interests of the Company and all of its stockholders. Delmarva and Atlantic believe the advantages of retaining board flexibility to address circumstances as they arise outweigh any perceived advantages that may derive from the adoption of policies at the present time.

Any management and allocation policies with respect to cash management, corporate expenses, allocation of assets and inter-Group transactions adopted by the Company Board will be required to conform to federal and state regulations, which extensively regulate the operations of the Company and its affiliates and which may impose procedural, substantive, record keeping, accounting and other requirements on the Company in connection with such matters. See "The Company Following the Mergers—Management of the Company."

After management and allocation policies, if any, are adopted, any and all of them could be modified or rescinded by the Company Board, in its sole discretion, without the approval of stockholders. Any determination to modify or rescind such policies, or to adopt other policies in their place, including any such decision that could have disparate effects upon holders of Company Common Stock or Class A Common Stock, would be made by the Company Board as set forth under "— Fiduciary Duties of the Company Board Are to All Stockholders Regardless of Class." See "The Company Following the Mergers—Management of the Company."

Limitations on Potential Acquisition of a Group. If each Group were a separate publicly held corporation, any person interested in acquiring such corporation without negotiation with management could seek control of the outstanding stock of such corporation by means of a tender offer or proxy contest. Although approval and consummation of the Mergers would authorize issuance of the Class A Common Stock that is intended to reflect the performance of the Atlantic Utility Group and the Company Common Stock that is intended to reflect the performance of the Residual Group, a person interested in acquiring only one Group without negotiation with the Company's management would still be required to seek control of the voting power represented by all of the outstanding capital stock of the Company, including the Company Common Stock and the Class A Common Stock. See "Description of the Company's Capital Stock—Voting."

Absence of Approval Rights with Respect to Future Issuances of Authorized Shares. Following consummation of the Mergers, the authorized but unissued shares of capital stock would be available for issuance from time to time by the Company at the sole discretion of the Company Board for any proper corporate purpose. Such issuances could include shares of Company Common Stock or (although there is no present intention to issue any such shares) shares of Class A Common Stock, as well as the issuance of such shares upon the conversion or exercise of securities of the Company that are convertible into or exercisable or exchangeable for such shares. The approval of the stockholders of the Company will not be sought by the Company for the issuance of authorized but unissued shares of Company Common Stock or Class A Common Stock (or the reissuance of previously issued shares that have been reacquired by the Company) or securities of the Company that are convertible into or exchangeable for such shares, unless deemed advisable by the Company Board or required by applicable law, regulation or stock exchange requirements. The authorized capital stock of the Company, as of the time of the consummation of the Mergers, will consist of (i) 160 million shares of common stock, of which 150 million shares will be designated as Company Common Stock and 10 million shares will be designated as Class A Common Stock, and (ii) 20 million shares of Preferred Stock.

Anti-Takeover Considerations. The Company Charter and the bylaws of the Company (the "Company Bylaws"), the existence of the Class A Common Stock, the DGCL, various state public utility regulatory statutes and the Public Utility Holding Company Act of 1935, as amended (the "1935 Act") may serve to discourage or make more difficult a change in control of the Company without the support of the Company Board or without meeting various other conditions. Charter and bylaw provisions that may discourage or make more difficult a change in control of the Company include the authorization of the Company Board to issue additional shares of Preferred Stock in one or more series and to fix and state the designations, powers, preferences, qualifications, limitations, restrictions and relative rights of the shares of each such series without further action by the Company's stockholders, certain procedures required in connection with the nomination of directors of the Company and the other provisions described under "Description of the Company's Capital Stock-Anti-Takeover Provisions." The existence of the Class A Common Stock would present complexities and could in certain circumstances pose obstacles, financial and otherwise, to an acquiring person. For example, a potential acquiror would have to take into consideration that holders of different series of Common Stock might be more or less receptive to the acquiror's proposal, that a tender offer would have to be structured so as to take into account different prices at which shares of the different series might be acquired, that a merger would require allocation of consideration among the different classes of Common Stock and the effects of actions the Company might

take such as causing a conversion of the Class A Common Stock. In addition, to the extent the relative market values of the Company Common Stock and the Class A Common Stock fluctuate, a potential acquiror may be able to acquire relatively more or less voting power for the same consideration depending on which class of Common Stock is purchased. The Company is subject to Section 203 of the DGCL, which, in general, prohibits a "business combination" between a corporation and an "interested stockholder" unless certain conditions are met. In addition, certain acquisitions of outstanding voting shares would require prior approval of state regulatory authorities pursuant to applicable state public utility regulatory statutes and of the SEC pursuant to the 1935 Act. See "Regulatory Matters." The provisions of the Company Charter and the Company Bylaws, the existence of the Class A Common Stock and provisions of the DGCL, various state utility regulatory statutes and the 1935 Act could, under certain circumstances, prevent stockholders from profiting from an increase in the market value of their shares as a result of a change in control of the Company by delaying or preventing such change in control. See "Description of the Company's Capital Stock—Anti-Takeover Provisions."

# **Risk Factors Associated with Each Class of Common Stock**

No Assurance of Payment of Dividends on the Common Stock. Although the Merger Agreement sets forth a dividend policy with respect to the Class A Common Stock, and Delmarva and Atlantic have announced a dividend policy for the Company Common Stock at the time of the consummation of the Mergers, such dividend policies are and will be subject to evaluation from time to time by the Company Board based on the Company's results of operations, financial condition and capital requirements and other relevant considerations, including regulatory considerations. Such policies, moreover, are not contained within the Company Charter, which is the instrument that establishes the terms of the Common Stock. See "Description of the Company's Capital Stock-Common Stock-Dividends-Dividend Policies" and "The Company Following the Mergers-Dividend Policies." Any dividends on the Company Common Stock and the Class A Common Stock that may be declared by the Company Board will be payable out of the lesser of (i) the funds of the Company legally available for such purpose, which are determined on the basis of the entire Company, and (ii) the Residual Group Available Dividend Amount and the Atlantic Utility Group Available Dividend Amount, respectively. See "Description of the Company's Capital Stock-Common Stock-Dividends-Available Amounts." The Company's ability to pay dividends will depend primarily upon the ability of its subsidiaries, including Delmarva and ACE, to pay dividends or otherwise transfer funds to it. Financing arrangements, charter provisions and regulatory requirements may impose restrictions on the ability of the Company's subsidiaries to transfer funds to the Company in the form of cash dividends, loans or advances. Such charter provisions include the provisions in the Restated Certificate and Articles of Incorporation, as amended, of Delmarva (the "Delmarva Charter") and in the Certificate of Incorporation, as amended, of ACE (the "ACE Charter") that, respectively, preclude the payment of dividends on Delmarva Common Stock and the common stock, \$3.00 par value, of ACE (the "ACE Common Stock") if there are any arrearages in payment of dividends on the Delmarva Preferred Stock or on the ACE preferred stock (the "ACE Preferred Stock"). Under the 1935 Act, the SEC has the power to preclude the payment of dividends by Delmarva and ACE to the Company or to preclude the payment of dividends by the Company under certain circumstances. Any net losses of the Company (without regard to whether such losses arose from any specific Group), any dividends or distributions on, or repurchases of, the Company Common Stock or the Class A Common Stock, and any dividends on, and certain repurchases of Company Preferred Stock, will reduce the funds of the Company legally available for payment of dividends on both the Company Common Stock and the Class A Common Stock. Subject to limitations of the DGCL and the Company Charter, the Company Board may declare and pay dividends on Company Common Stock and Class A Common Stock in equal or unequal amounts, or may decide not to declare and pay such dividends, notwithstanding the relationship between the Residual Group Available Dividend Amount and the Atlantic Utility Group Available Dividend Amount, the respective amounts of prior dividends paid on, or liquidation rights of, the Company Common Stock or the Class A Common Stock or any other factor. See "Description of the Company's Capital Stock-Common Stock-Dividends-Dividend Policies."

Disposition of Group Assets Without Stockholder Approval. As long as the assets of a Group represent less than substantially all of the properties and assets of the Company, the Company Board may approve sales and other dispositions of any amount of the properties and assets of such Group without stockholder approval, since under the DGCL and the Company Charter stockholder approval is required only for a sale or other disposition of all or substantially all of the properties and assets of the Company as a whole. The proceeds from any such disposition would be assets attributed to such Group and used for its benefit.

# **Risk Factors Associated with the Class A Common Stock**

No Assurance of Payment of Dividends on the Class A Common Stock. The Merger Agreement provides, subject to declaration by the Company Board and the obligation of the Company Board to react to the financial condition and regulatory environment of the Company and its results of operations, that the dividends declared and paid on the Class A Common Stock will be maintained at a level of \$3.20 per share per annum until the earlier of July 1, 2001 or the end of the twelfth calendar quarter following the calendar quarter in which the Mergers are consummated (the "Initial Period"). It is the intention of the Company after the Initial Period to pay dividends on Class A Common Stock (the amount of such dividends to include the amount credited to the Intergroup Interest as described below under "Description of the Company's Capital Stock-Intergroup Interest-Adjustments in Connection with Various Transactions—Dividends") at a rate equal to 90% of the Company's earnings attributable to the Atlantic Utility Group in excess of an amount equal to \$40,000,000 per annum (the "Notional Fixed Charge"). The anticipated dividend rate for the Initial Period of \$3.20 per annum would imply, at a payout ratio of 90%, that Company Net Income Attributable to the Atlantic Utility Group exceeded \$3.56 per outstanding share of Class A Common Stock. There can be no assurance that the Atlantic Utility Group will have sufficient earnings after the Initial Period to maintain a \$3.20 per share dividend rate for the Class A Common Stock. Furthermore, the Merger Agreement provides that if and to the extent that the annual dividends paid on the Class A Common Stock during the Initial Period (including the amount credited to the Intergroup Interest as aforesaid) shall have exceeded 100% of Company Net Income Attributable to the Atlantic Utility Group during such period, the Company Board may consider such fact in determining the appropriate annual dividend rate on the Class A Common Stock following the Initial Period. Dividends on the Class A Common Stock will not be cumulative. See "Description of the Company's Capital Stock-Common Stock-Dividends-Dividend Policies."

Conversion of Class A Common Stock into Company Common Stock at the Option of the Company. The Company Board could, in its sole discretion, determine to convert each outstanding share of Class A Common Stock into shares of Company Common Stock (i) at any time at a premium that will decline over four years from 25% to 10% or (ii) following a dividend or partial redemption undertaken in connection with a disposition of all or substantially all of the properties and assets of the Atlantic Utility Group. Such determination could be made at a time when either or both the Company Common Stock and the Class A Common Stock may be considered to be overvalued or undervalued. In addition, any such conversion at any such premium would dilute the interests in the Company of the holders of Company Common Stock and could preclude holders of both classes of Common Stock from retaining their investment in a security that is intended to reflect separately the performance of the relevant business group. In determining whether to convert Class A Common Stock into the Company Common Stock, the Company Board would act in accordance with its good faith business judgment that any such conversion is in the best interests of the Company and all of its stockholders, including both the holders of the Class A Common Stock and the holders of the Company Common Stock. The Company cannot predict the impact on the market prices of the Company Common Stock or the Class A Common Stock of its ability to effect any such conversion or the effect, if any, that the exercise by the Company of such conversion right would have on the market price of the Company Common Stock or the Class A Common Stock prevailing at such time. See "-Risk Factors Associated with the Dual Class Common Equity Structure-Potential Divergence of Interests and Requirement to Act on Behalf of Company and All Stockholders" and "Description of the Company's Capital Stock-Conversion and Redemption of the Class A Common Stock."

No Assurance as to Market Price of Class A Common Stock. Because there has been no prior market for the Class A Common Stock of the Company, there can be no assurance as to the market price of the Class A Common Stock following the consummation of the Mergers. Accordingly, there can be no assurance that the combined market values of the Company Common Stock and the Class A Common Stock held by an Atlantic stockholder after the consummation of the Mergers will equal or exceed the market value of the Atlantic Common Stock held by such stockholder prior to such time. See "Selected Historical and Unaudited Pro Forma Financial Data—Comparative Per Share Prices of Delmarva Common Stock and Atlantic Common Stock."

The market prices of the Class A Common Stock after the consummation of the Mergers would be determined in the trading markets and could be influenced by many factors, including the results of the Atlantic Utility Group, investors' expectations for the Atlantic Utility Group, trading volumes, regulatory environment and general economic and market conditions. There can be no assurance as to the extent to which investors would assign values to the Class A Common Stock based on the reported financial results or other measures of performance or prospects of the relevant businesses. There is no present intention on the part of Delmarva and Atlantic for the Company to issue any shares of Class A Common Stock following consummation of the Mergers. Financial effects of the Atlantic Utility Group that affect the Company's consolidated results of operations or financial condition could affect the market prices of the Class A Common Stock on their market price, such as (i) basing the consideration to be paid, if all or substantially all of the properties and assets of the Atlantic Utility Group are sold in a Disposition, on the Net Proceeds of such Disposition, (ii) the ability of the Company Board to make various determinations with respect to the Atlantic Utility Group and the Class A Common Stock into

The possibility that the Company's earnings attributable to the Atlantic Utility Group will not be sufficient to maintain the dividend rate that the Company intends to pay during the Initial Period, and therefore, the possibility that the dividend rate payable on the Class A Common Stock will be reduced after the Initial Period by a material amount, may have an adverse impact on the market price of the Class A Common Stock. In addition, because the earnings of the Company attributable to the Atlantic Utility Group out of which dividends on the Class A Common Stock will be paid are calculated after deduction of the Notional Fixed Charge, a small change in the earnings of the Company attributable to the Atlantic Utility Group may have a disproportionately large impact, on a percentage basis, on the earnings available to pay dividends on the Class A Common Stock. As a result, the market price of the Class A Common Stock may be more volatile than the market price of the Company Common Stock.

The earnings of the Company attributable to the Atlantic Utility Group are dependent on a number of factors, many of which are beyond the control of the Company. These factors include the growth in demand for electric power in the areas served by the Atlantic Utility Group and the scope and timing of regulatory changes that would allow other suppliers of electric power to serve customers in the areas served by the Atlantic Utility Group's average cost of power is above-market, changes in the regulatory environment that allow lower cost suppliers of power to compete with the Atlantic Utility Group without a corresponding non-bypassable charge having been allowed by the regulatory authorities may have an adverse impact on the Company's earnings attributable to the Atlantic Utility Group.

In connection with payments of dividends on the Class A Common Stock (other than a dividend payable in shares of Class A Common Stock or in securities of the Company attributed to the Atlantic Utility Group), the Residual Group will be credited and the Atlantic Utility Group will be charged an amount reflecting the Intergroup Interest. See "Description of the Company's Capital Stock—Intergroup Interest" and "—Dividends." In the event the businesses attributable to the Atlantic Utility Group do not make a cash payment to the Residual Group in the amount of such credit and charge at the time a dividend is paid on the Class A Common Stock, the Company may account for the credit and charge in a number of possible ways. For instance, the Company could treat the amount effectively as a loan from the

Residual Group to the Atlantic Utility Group, on which interest would be payable. Alternatively, the Company could also adjust for the credit and the charge by reducing the Outstanding Atlantic Utility Fraction and increasing the Intergroup Interest Fraction.

Potential Adverse Effects of Possible Disposition of Assets Attributable to the Atlantic Utility Group. The terms of the Class A Common Stock provide that if the Company and/or its subsidiaries were to dispose of all or substantially all of the properties and assets attributable to the Atlantic Utility Group, other than in a transaction in which the Company receives primarily equity securities of an entity engaged or proposing to engage primarily in a similar or complementary business and other than in connection with the disposition of all or substantially all of the assets of the Company, the Company would be required, at its option, either to (i) distribute to holders of Class A Common Stock an amount equal to their proportionate interest in the Net Proceeds of such Disposition, either by special dividend or by redemption of all or part of the outstanding shares of Class A Common Stock or (ii) convert outstanding shares of Class A Common Stock into shares of the corresponding series of Company Common Stock at a conversion ratio based on 110% of the average daily ratio of the Market Value of a share of Class A Common Stock to the Market Value of a share of Company Common Stock over a specified period following such Disposition. In addition, the Company Charter will not require the Company Board to select the option that would result in the distribution with the highest value to the holders of the Class A Common Stock or with the smallest effect on the Company Common Stock. The Company Board would select an option based upon its good faith business judgment that such option is in the best interests of the Company and all of its stockholders, including both the holders of the Class A Common Stock and the holders of the Company Common Stock. "Net Proceeds" means the proceeds of such Disposition after payment of or provision for certain specified costs, including taxes to be paid by the Company in respect of the Disposition or such dividend or redemption, transaction costs and liabilities and other obligations (including obligations in respect of preferred stock) attributed to the Atlantic Utility Group. If the Atlantic Utility Group were a separate independent company and its shares were acquired by another person, certain of those costs, including corporate level taxes, might not be payable in connection with such an acquisition. As a result, the consideration that would be received by stockholders of such a separate independent company in connection with such a stock acquisition might be greater than the Fair Value of the Net Proceeds that would be received by holders of Class A Common Stock if all or substantially all of the properties and assets of the Atlantic Utility Group were sold. In addition, no assurance can be given that the Net Proceeds per share of Class A Common Stock to be received in connection with a Disposition of all or substantially all of the properties and assets of the Atlantic Utility Group will be equal to or more than the market value per share of Class A Common Stock prior to or after announcement of such Disposition. See "---Conversion of Class A Common Stock into Company Common Stock at the Option of the Company" and "-No Assurance as to Market Price of Class A Common Stock" and "Description of the Company's Capital Stock-Conversion and Redemption of the Class A Common Stock."

# **Risk Factors Associated with the Company Common Stock**

No Assurance as to Market Price of Company Common Stock. Because there has been no prior market for the Company Common Stock, there can be no assurance as to the market price of the Company Common Stock following the consummation of the Mergers. Accordingly, there can be no assurance that the combined market values of the Company Common Stock and the Class A Common Stock held by an Atlantic stockholder after the consummation of the Mergers will equal or exceed the market value of the Atlantic Common Stock held by such stockholder prior to such time or that the market value of the Company Common Stock held by a Delmarva stockholder after the consummation of the Mergers will equal or exceed the market value of the Delmarva Common Stock held by such stockholder prior to such time. See "Selected Historical and Unaudited Pro Forma Financial Data—Comparative Per Share Prices of Delmarva Common Stock."

The market price of the Company Common Stock after the consummation of the Mergers would be determined in the trading markets and could be influenced by many factors, including the consolidated results of the Company, as well as the respective performances of the Groups, investors' expectations for the Groups, trading volumes, regulatory environment and general economic and market conditions. There can be no assurance as to the extent to which investors would assign values to the Company Common Stock based on the reported financial results or other measures of performance or prospects of the relevant businesses. Financial effects of the Groups that affect the Company's consolidated results of operations or financial condition could affect the market prices of the Company Common Stock. In addition, the Company cannot predict the impact of certain terms of the Company Common Stock and the Class A Common Stock on the market price of the Company Stock such as (i) basing the consideration to be paid, if all or substantially all of the properties and assets of the Atlantic Utility Group are sold in a Disposition, on the Net Proceeds of such Disposition, (ii) the ability of the Company Board to make various determinations with respect to the Groups and the Company Common Stock and Class A Common Stock.

No Company Charter Provision Governing Disposition of All or Substantially All of the Residual Group's Properties and Assets. The Company Charter does not contain provisions relating to a disposition of all or substantially all of the properties and assets of the Residual Group comparable to those relating to a Disposition of all or substantially all of the properties and assets of the Atlantic Utility Group discussed above under "—Risk Factors Associated with the Class A Common Stock—Potential Adverse Effects of Possible Disposition of Assets of the Atlantic Utility Group." See "Description of Company's Capital Stock—Conversion and Redemption of the Class A Common Stock." The appropriate disposition of proceeds, if any, in the case of a disposition of all or substantially all of the properties and assets of the Company Board in accordance with the Company Charter and in the exercise of its fiduciary duties. See "—Risk Factors Associated with the Dual Class Common Equity Structure—Fiduciary Duties of the Company Board Are to All Stockholders Regardless of Class."

## SELECTED HISTORICAL AND PRO FORMA FINANCIAL DATA

The following tables contain the (1) selected historical consolidated financial data, including certain historical per share data, for Delmarva and Atlantic and (2) selected unaudited pro forma combined financial data, including pro forma combined per share data, for the Company. The selected historical financial data should be read in conjunction with the historical consolidated financial statements and related notes thereto of Delmarva and Atlantic as presented in their respective 1995 Forms 10-K and 1996 Forms 10-Q, which are incorporated by reference herein. See "Available Information" and "Incorporation by Reference." The selected unaudited pro forma combined financial data should be read in conjunction with the unaudited pro forma combined financial statements and related notes thereto of the Company, which are set forth elsewhere in this Joint Proxy Statement/Prospectus.

## Selected Historical Financial Data

The selected historical consolidated financial data for Delmarva and Atlantic for each year in the fiveyear period ended December 31, 1995 have been derived from the audited financial statements of Delmarva and Atlantic, respectively. The selected historical financial data for Delmarva and Atlantic for the nine months ended September 30, 1996 and 1995, have been derived from the unaudited financial statements of Delmarva and Atlantic, respectively, which, in the opinion of the management of both companies, reflect all adjustments, including only normal recurring adjustments, necessary for a fair presentation of the financial results for the interim periods. The results for such interim periods do not necessarily indicate the results for the full fiscal year.

## DELMARVA POWER & LIGHT COMPANY (Dollars in Thousands, Except Per Share Amounts)

	Nine Months Ended September 30,						
·	1996	1995	1995	1994 ·	1993	1992 .	1991
Income Statement Data			6			· .	
Operating revenues	\$834,579	\$753,893	\$995,103	\$991,021	\$970,607	\$864,044	\$855,821
Income from continuing			-			· ·	•
operations	94,503	97,566	117,488	108,310(a)	111,076	· 98,526(b)	80,506
Net income	94,503	97,566	117,488	108,310(a)	111,076	98,526(b)	93,236(c)
Earnings applicable to			,	, ()		, , ,	
common stock	87,210	90,089	107,546	98,940(a)	101,074	90,177(b)	85,259(c)
Earnings per average share	,		,	, , ,			
of common stock from							
continuing operations	\$1.44	\$1.50	\$1.79	\$1.67(a)	\$1.76	\$1.69(b)	\$1.44
Earnings per average share							·
of common stock	\$1.44	\$1.50	\$1.79	\$1.67(a)	\$1.76	\$1.69(b)	\$1.69(c)
Dividends declared per share						,	(-)
of common stock	\$1.155	\$1.155	\$1.54	\$1.54	\$1.54	\$1.54	\$1.54
	September 30,		December 31,				
	1996		1995	1994	1993	1992	1991
Balance Sheet Data							
Total assets	\$2,898,111		\$2,866,685	\$2,669,785	\$2,592,479	\$2,374,793	\$2,263,718
Long-term debt and capital							
lease obligations	848,953(d)		874,672(d)	794,218(d)	759,703(d)	813,468(d)	799,483(d)
Book value per common			,			, , , ,	
share outstanding	\$15.48		\$15.20	\$14.85	\$14.66	\$13.77	\$13.42

(a) The 1994 early retirement offer reduced net income and earnings per share by \$10.7 million and \$0.18, respectively.

(b) The 1992 settlement of the Peach Bottom lawsuit increased net income and earnings per share by \$11.4 million and \$0.21, respectively.

(c) The 1991 one-time cumulative effect of a change in accounting for unbilled revenues increased net income and earnings per share by \$12.7 million and \$0.25, respectively.

(d) Excludes variable rate demand bonds classified as current liabilities and amounts due within one year.

	Nine Months Ended September 30,			Year Ended December 31,					
	1996	1995	1995	1994	1993	1992	1991		
Income Statement Data							,		
Operating revenues	\$752,968	3 \$727,543	\$953,137	\$913,039	\$865,675	\$816,825	\$808,374		
Net income	58,352	2 70,782	81,768	76,113(a)	95,297(b)	86,210(c)	85,635		
Earnings per average share of									
common stock	\$1.1	1 \$1.34	\$1.55	\$1.41(a)	\$1.80(b)	\$1.67(c)	\$1.75		
Dividends declared per share of									
common stock	\$1.15	5 \$1.155	\$1.54	\$1.54	\$1.535	\$1.515	\$1.495		
				•					
	September 30,		December 31,						
· · ·		1996	1995	1994	1993	1992	1991		
Balance Sheet Data	• • •								
Total assets		\$2,679,681	\$2,620,896	\$2,545,555	\$2,487,508	\$2,219,338	\$2,151,416		
Preferred stock subject to mandate									
redemption	-	90,000(d) <sup>.</sup>	114,750(d)	) 149,250(d)	173,750(d)	190,250(d)	191,300(0		
Long-term debt and capital lease									
obligations		866,805(d)	870,083(d)	) 819,399(d)	810,508(d)	680,085(d)	617,900(0		
Book value per common share									

# ATLANTIC ENERGY, INC. (Dollars in Thousands, Except Per Share Amounts)

(a) The 1994 employee separation costs reduced net income and earnings per share by \$17.3 million and \$0.32, respectively.

(b) The 1993 nonrecurring charges for reorganization activities reduced net income and earnings per share by \$5.4 million and \$0.10, respectively.

(c) The 1992 settlement of the Peach Bottom lawsuit increased net income and earnings per share by \$7.7 million and \$0.15, respectively.

(d) Excludes amounts due within one year. • • •

.

#### Selected Unaudited Pro Forma Combined Financial Data and Comparative Per Share Data

The following selected unaudited pro forma combined financial data combine the historical consolidated balance sheets and statements of operations of Delmarva and Atlantic for the same periods, after giving effect to the Mergers under the purchase method of accounting and assuming that the Mergers had been effective for all periods presented. Pro forma per share data for Company Common Stock give effect to the conversion of each share of Delmarva Common Stock into one share of Company Common Stock and the conversion of each share of Atlantic Common Stock into 0.75 of one share of Company Common Stock and 0.125 of one share of Class A Common Stock. This data does not reflect any cost savings or other synergies anticipated by management as a result of the Mergers. See "The Merger Agreement—The Mergers." The selected unaudited pro forma combined financial data are not necessarily indicative of the actual operating outcomes or financial position that would have resulted had the Mergers been consummated on the dates for which the Mergers are being given effect and should not be construed as necessarily indicative of future operating results or financial position. The following information is based on the unaudited pro forma combined statements of income and balance sheet presented below in this Joint Proxy Statement/Prospectus (see "Unaudited Pro Forma Combined Financial Statements").

## THE COMPANY

## (Dollars in Thousands, Except Per Share Amounts)

· · · · ·	Pro Forma		
· · ·	Nine Months Ended September 30, 1996	Year Ended December 31, 1995	
Income Statement Data			
Operating revenues	\$1,587,547	\$1,948,240	
Net income	144,407	187,567	
Earnings applicable to common stock	134,828	173,736	
Earnings applicable to Class A common stock	9,579	13,831	
Earnings per average share of common stock	\$1.35	\$1.74	
Earnings per average share of Class A common stock	\$1.45	\$2.10	
Dividends declared per share of common stock	\$1.155	\$1.54	
Dividends declared per share of Class A common stock	\$2.40	\$3.20	
		September 30, 1996	
Balance Sheet Data	. *		
Total assets		\$5,774,342	
Preferred stock of subsidiaries subject to mandatory redemption		183,950	
Long-term debt and capital lease obligations, net of current maturi		1,715,758	
Book value per share outstanding		\$17.53	

Comparative Earnings, Dividends and Book Values Per Share of Common Stock and Class A Common Stock

	Nine Months Ended September 30, 1996	Year Ended December 31, 1995
Earnings Per Average Share of Common Stock		
The Čompany Pro forma (a)	\$ 1.35	\$ 1.74
Delmarva Historical	1.35	1.79 1.74
Atlantic         Historical         Equivalent pro forma (c)	$\begin{array}{c} 1.11\\ 1.01\end{array}$	1.55 1.31
Earnings Per Average Share of Class A Common Stock	e de la companya de l	
The Company Pro forma (a) Atlantic	1.45	2.10
Equivalent pro forma (d)	1.45	2.10
Equivalent Pro Forma Earnings Per Share to be Received by Atlantic Common Stockholders for each Share of Atlantic		
Common Stock (e) Dividends Per Share of Common Stock	1.19	1.57
The Company Pro forma (a) Delmarva	1.155	1.54
Historical	1.155 1.155	1.54 1.54
Atlantic         Historical         Equivalent pro forma (c)	1.155 0.866	1.54 1.155
Dividends Per Share of Class A Common Stock		1
The Company Pro forma (a) Atlantic	. 2.40	3.20
Equivalent pro forma (d)	2.40	3.20
Equivalent Pro Forma Dividends Per Share to be Received by Atlantic Common Stockholders for each Share of Atlantic	· · ·	· 'u
Common Stock (e)	1.166	1.555
	<i>,</i> .	September 30, 1996
Book Value Per Share Outstanding		
The Company Pro forma (a)		\$17.53
		ψ11.JJ
Historical		15.48 17.53
Atlantic Historical Equivalent pro forma	· · · ·	15.42 17.53

(a) The Company pro forma financial data are based on the unaudited pro forma combined statements of income and balance sheet presented below in this Joint Proxy Statement/Prospectus (see "Unaudited Pro Forma Combined Financial Statements").

(b) Delmarva equivalent pro forma earnings and dividends per share of common stock are equal to the Company equivalent combined amounts because each share of Delmarva Common Stock outstanding immediately prior to the Effective Time shall be converted into the right to receive one share of Company Common Stock.

(c) Atlantic equivalent pro forma earnings and dividends per share of common stock are based upon the Company equivalent combined amount multiplied by the exchange ratio of 0.75 share of Company Common Stock for each one share of Atlantic Common Stock outstanding immediately prior to the Effective Time.

(d) Atlantic equivalent pro forma earnings and dividends per share of Class A Common Stock are equal to the Company equivalent combined amounts because the Class A Common Stock is to be exchanged solely for Atlantic Common Stock outstanding immediately prior to the Effective Time.

(e) Equivalent pro forma earnings and dividends per share of Common Stock and Class A Common Stock to be received by Atlantic common stockholders in exchange for each share of Atlantic Common Stock are computed based on the conversion ratios as follows:

	Nine Months Ended September 30, 1996	Year Ended December 31, 1995
Earnings Per Average Share:		
Earnings per average share of Common Stock		
The Company Pro forma	\$ 1.35	\$ 1.74
Common Stock	0.75	0.75
Subtotal	\$ 1.01	\$ 1.31
Earnings per average share of Class A Common Stock The Company		
Pro forma	\$ 1.45 0.125	$2.10 \\ 0.125$
Subtotal	\$ 0.18	\$ 0.26
Equivalent pro forma earnings per share of Common Stock and	φ 0.10 	φ 0.20 
Class A Common Stock to be received by Atlantic common stockholders for each share of Atlantic Common Stock	\$ 1.19	<u>\$ 1.57</u>
Dividends Per Average Share:	· · ·	
Dividends per average share of Common Stock The Company		• • •
Pro forma	\$1.155	\$ 1.54
Common Stock	0.75	0.75
Subtotal	\$0.866	\$1.155
Dividends per average share of Class A Common Stock	2	
The Company Pro forma	\$ 2.40	\$ 3.20
Atlantic's conversion ratio for Class A Common Stock	0.125	0.125
Subtotal	\$ 0.30	\$ 0.40
Equivalent pro forma dividends per share of Common Stock and Class A Common Stock to be received by Atlantic common		
stockholders for each share of Atlantic Common Stock	\$1.166	<u>\$1.555</u>
	· · ·	1



## Comparative Per Share Prices of Delmarva Common Stock and Atlantic Common Stock

The Delmarva and Atlantic Common Stocks are listed on the NYSE and the Philadelphia Stock Exchange, and the Atlantic Common Stock is also listed on the Pacific Stock Exchange. The following table presents, for the calendar quarters indicated, the dividends paid and the high and low sales prices of Delmarva Common Stock and Atlantic Common Stock as reported on the NYSE—Composite Transactions, in each case based on published financial sources.

	Delmarva			Atlantic		
	High	Low	Dividends	High	Low	Dividends
1994						
First Quarter	\$235/8	\$201⁄2	\$0.381/2	\$213⁄4	\$197⁄8	\$0.381/2
Second Quarter	21	161/8	0.38½	211/2	16¾	0.381/2
Third Quarter	20	17¾	0.381/2	195⁄8	161/8	0.381/2
Fourth Quarter	19¼	175⁄8	0.381⁄2	18¼	16	0.381/2
1995						
First Quarter	20	171/8	0.381/2	19	17¾	0.381/2
Second Quarter	21¼	191⁄8	0.381/2	195/8	171⁄8	0.381/2
Third Quarter	23	191⁄2	0.381/2	197⁄8	181⁄8	0.381/2
Fourth Quarter	23 <sup>5</sup> /8;	211/8	0.381/2	201/8	19	0.381/2
1996			i			•
First Quarter	235/8	21	0.381/2	20	16%	0.381/2
Second Quarter	213⁄8	19½	0.381/2	18¾	16	0.381/2
Third Quarter	211⁄4	20	0.381/2	18½	17 -	0.381/2
Fourth Quarter (a)	21¼	19%	0.381/2	18¼	173/8	0.381⁄2

(a) Through the close of business on December 20, 1996.

On August 9, 1996, the last full trading day before the public announcement of the execution and delivery of the Merger Agreement, the closing price per share was \$20% for the Delmarva Common Stock and \$17% for the Atlantic Common Stock, as reported on the NYSE—Composite Transactions.

On December 20, 1996, the closing price per share was \$20 for the Delmarva Common Stock and \$17% for the Atlantic Common Stock, as reported on the NYSE—Composite Transactions.

The market prices of Delmarva Common Stock and Atlantic Common Stock are subject to fluctuation. As a result, Delmarva and Atlantic stockholders are urged to obtain current market quotations for Delmarva Common Stock and Atlantic Common Stock.

The equivalent market value per share for Atlantic Common Stock is based on the market value of the securities of the Company received for each share of Atlantic Common Stock outstanding immediately prior to the consummation of the Mergers. The portion of the equivalent market value related to the exchange of each share of Atlantic Common Stock for 0.75 of one share of Company Common Stock is \$15.47 based on the August 9, 1996 closing price per share of Delmarva Common Stock of \$20%. The portion of the equivalent market value related to the exchange of each share of Class A Common Stock is not determinable since there will not be a market for the Class A Common Stock until it is issued in conjunction with the Mergers.

### **MEETINGS, VOTING AND PROXIES**

This Joint Proxy Statement/Prospectus is being furnished to (i) the stockholders of Delmarva in connection with the solicitation of proxies by the Delmarva Board from the holders of Delmarva Common Stock for use at the Delmarva Meeting and (ii) the stockholders of Atlantic in connection with the solicitation of proxies by the Atlantic Board from the holders of Atlantic Common Stock for use at the Atlantic Board from the holders of Atlantic Common Stock for use at the Atlantic Meeting, each to consider and vote on proposals to approve the Merger Agreement and the transactions contemplated thereby and to adopt the Company Plan.

#### **Delmarva Meeting**

Purpose of Delmarva Meeting. The purpose of the Delmarva Meeting is to consider and vote upon the proposal to approve the Merger Agreement, pursuant to which the holders of Delmarva Common Stock and the holders of Atlantic Common Stock will become holders of Company Common Stock and, in the case of holders of Atlantic Common Stock, Class A Common Stock upon the completion of the Delmarva Merger and the Atlantic Merger, respectively, and to vote on the Company Incentive Compensation Plan (the "Company Plan"), which upon consummation of the Mergers will replace the Delmarva Power & Light Company Long-Term Incentive Plan (the "Delmarva LTIP") and Atlantic's Equity Incentive Plan, subject to approval by stockholders. The Delmarva Board is not aware, as of the date of this Joint Proxy Statement/Prospectus, of any other matters that may properly come before the Delmarva Meeting. The bylaws of Delmarva, as amended (the "Delmarva Bylaws"), provide that only business within the purposes described in the notice of the Delmarva Meeting may be conducted at the Delmarva Meeting. The enclosed form of proxy authorizes the voting of shares represented by proxy on all other matters that may properly come before the Delmarva Meeting and any adjournment or adjournments thereof. If any other matter incident to the conduct of the Delmarva Meeting properly came before the Delmarva Meeting or any adjournment or adjournments thereof, it is the intention of the persons named in the proxy to vote the proxies in accordance with their best judgment.

THE DELMARVA BOARD, BY UNANIMOUS VOTE OF THE DIRECTORS, HAS APPROVED THE MERGER AGREEMENT AND THE COMPANY PLAN AND RECOMMENDS THAT HOLDERS OF DELMARVA COMMON STOCK VOTE FOR APPROVAL OF THE MERGER AGREEMENT AND THE COMPANY PLAN.

In considering the recommendation of the Delmarva Board with respect to the Merger Agreement, stockholders should be aware that certain members of the Delmarva Board will become directors and/or employees of the Company following consummation of the Mergers and/or be or become entitled to severance benefits as a result of the Mergers and therefore, such directors may have interests in the Mergers that are in addition to the interests of stockholders of Delmarva generally and that could potentially represent conflicts of interest. The Delmarva Board was aware of these interests and considered them, among other matters, in approving the Merger Agreement. See "The Mergers—Conflicts of Interest of Certain Persons in the Mergers."

Date, Place and Time; Record Date. The Delmarva Meeting is scheduled to be held on January 30, 1997 at 10:00 a.m., local time, at the Holiday Inn, 800 King Street, Wilmington, Delaware 19899. Holders of record of Delmarva Common Stock at the close of business on the Delmarva Record Date (December 18, 1996) will be entitled to notice of and to vote at the Delmarva Meeting or any adjournment or adjournments thereof. A list of stockholders of record entitled to vote at the Delmarva's principal business office at 800 King Street, Wilmington, Delaware 19899, prior to the Delmarva Meeting. The list will also be available on the day of the Delmarva Meeting at the meeting site. Holders of record of shares of Delmarva Common Stock at the close of business on the Delmarva Meeting. The list will also be available on the day of the Delmarva Meeting at the meeting site. Holders of record of shares of Delmarva Common Stock at the close of business on the Delmarva Record Date will also be entitled to vote at the Delmarva Meeting on the proposal to adopt the Company Plan.



Voting Rights. Each share of Delmarva Common Stock entitles its holder to one vote with respect to the Merger Agreement and the Company Plan.

A majority of the shares issued and outstanding and entitled to vote, present in person or represented by proxy, will constitute a quorum for the transaction of business at the Delmarva Meeting. Abstentions and broker nonvotes are counted as present for establishing a quorum. The affirmative vote of the holders of more than two-thirds of all votes entitled to be cast by all holders of Delmarva Common Stock is required for approval of the Merger Agreement and the transactions contemplated thereby. The affirmative vote of the holders of a majority of Delmarva Common Stock present in person or by proxy and entitled to vote is required for approval of the Company Plan. Under the DGCL and the VSCA, in determining whether the Merger Agreement has received the requisite number of affirmative votes, abstentions and broker nonvotes will have the same effect as a vote against approval of the Merger Agreement; abstentions will have the same effect as a vote against approval of the Company Plan, and brokers who do not receive specific instructions are entitled to vote on the Company Plan.

On the Delmarva Record Date, 60,754,568 shares of Delmarva Common Stock were outstanding and entitled to vote. As of the Delmarva Record Date, directors, executive officers and their affiliates owned less than 1% of the issued and outstanding shares of Delmarva Common Stock.

*Proxies.* Holders of Delmarva Common Stock may vote either in person or by properly executed proxy. By completing and returning the form of proxy, the Delmarva stockholder authorizes the persons named therein to vote all of such Delmarva stockholder's shares on his or her behalf. All completed Delmarva proxies returned will be voted in accordance with the instructions indicated in the proxies. If no contrary instructions are given, the Delmarva proxies will be voted FOR approval of the Merger Agreement and the transactions contemplated thereby and FOR the Company Plan. Any proxy given pursuant to this solicitation may be revoked by the person giving it at any time before it is voted (i) by delivery to the Secretary of Delmarva at 800 King Street, Wilmington, Delaware 19899 on or before the taking of the vote at the Delmarva Meeting, a written notice of revocation bearing a later date than the proxy or a later dated proxy relating to the same shares of Delmarva Common Stock or (ii) by attending the Delmarva Meeting and voting in person. Attendance at the Delmarva Meeting will not in itself constitute a revocation of a proxy.

Delmarva will bear the cost of the solicitation of proxies for the Delmarva Meeting, except that Delmarva and Atlantic will share equally expenses incurred in connection with printing and filing this Joint Proxy Statement/Prospectus. See "The Merger Agreement—Expenses." Proxies may be solicited by certain officers and employees of Delmarva or its subsidiaries by mail, telephone, personally or by other communications without compensation apart from their normal salaries. Delmarva has retained Georgeson & Company Inc. to assist in soliciting proxies from Delmarva stockholders, including brokers accounts, at a fee for such services of approximately \$50,000 and reasonable out-of-pocket expenses.

The Delmarva Meeting may be adjourned one or more times to another date and/or place for any purpose (including, without limitation, for the purpose of soliciting additional proxies).

#### Atlantic Meeting

*Purpose of Atlantic Meeting.* The purpose of the Atlantic Meeting is to consider and vote upon the proposal to approve the Merger Agreement, pursuant to which the holders of Atlantic Common Stock and the holders of Delmarva Common Stock will become holders of Company Common Stock and, in the case of holders of Atlantic Common Stock, Class A Common Stock upon the completion of the Atlantic Merger and the Delmarva Merger, respectively, and to vote on the Company Plan, which upon consummation of the Mergers will replace Atlantic's Equity Incentive Plan and the Delmarva LTIP, subject to approval by stockholders. The Atlantic Board is not aware, as of the date of this Joint Proxy Statement/Prospectus, of any other matters that may properly come before the Atlantic Meeting. The bylaws of Atlantic (the "Atlantic Bylaws") provide that only business within the purposes described in the notice of the Atlantic

Meeting may be conducted at the Atlantic Meeting. The enclosed form of proxy authorizes the voting of shares represented by proxy on all other matters that may properly come before the Atlantic Meeting and any adjournment or adjournments thereof. If any other matter incident to the conduct of the Atlantic Meeting properly came before the Atlantic Meeting or any adjournment or adjournments thereof, it is the intention of the persons named in the proxy to vote the proxies in accordance with their best judgment.

THE ATLANTIC BOARD, BY UNANIMOUS VOTE OF THE DIRECTORS, HAS APPROVED THE MERGER AGREEMENT AND THE COMPANY PLAN AND RECOMMENDS THAT HOLD-ERS OF ATLANTIC COMMON STOCK VOTE FOR APPROVAL OF THE MERGER AGREE-MENT AND THE COMPANY PLAN.

In considering the recommendation of the Atlantic Board with respect to the Merger Agreement, stockholders should be aware that certain members of the Atlantic Board will become directors and/or employees of the Company following consummation of the Mergers and/or be or become entitled to severance benefits as a result of the Mergers and therefore, such directors may have interests in the Mergers that are in addition to the interests of stockholders of Atlantic generally and that could potentially represent conflicts of interest. The Atlantic Board was aware of these interests and considered them, among other matters, in approving the Merger Agreement. See "The Mergers—Conflicts of Interest of Certain Persons in the Mergers."

Date, Place and Time; Record Date. The Atlantic Meeting is scheduled to be held on January 30, 1997 at 3:00 p.m., local time, at the Frank Guaracini, Jr. Fine & Performing Arts Center at Cumberland County College, Sherman Avenue and College Drive, Vineland, New Jersey 08360. Holders of record of Atlantic Common Stock at the close of business on the Atlantic Record Date (December 18, 1996) will be entitled to notice of and to vote at the Atlantic Meeting or any adjournment or adjournments thereof. A list of stockholders of record entitled to vote at the Atlantic Meeting will be available for inspection by holders of Atlantic Common Stock at Atlantic's principal business office at 6801 Black Horse Pike, Egg Harbor Township, New Jersey 08234, prior to the Atlantic Meeting. The list will also be available on the day of the Atlantic Meeting at the meeting site. Holders of record of shares of Atlantic Common Stock at the close of Date will also be entitled to vote at the Atlantic Meeting on the proposal to adopt the Company Plan.

Voting Rights. Each share of Atlantic Common Stock entitles its holder to one vote with respect to the Merger Agreement and the Company Plan.

A majority of the shares issued and outstanding and entitled to vote, present in person or represented by proxy, will constitute a quorum for the transaction of business at the Atlantic Meeting. Abstentions and broker nonvotes are counted as present for establishing a quorum. The affirmative vote of the holders of at least a majority of all votes cast by all holders of Atlantic Common Stock entitled to vote is required for approval of the Merger Agreement and the transactions contemplated thereby. The affirmative vote of at least a majority of all votes cast by all holders of Atlantic Common Stock entitled to vote is required for approval of the Company Plan. Abstentions and broker nonvotes are not counted in determining approval of the Merger Agreement; abstentions are not counted in determining approval of the Merger Agreement; abstentions are not counted in determining approval of the Company Plan, and brokers who do not receive specific instructions are entitled to vote on the Company Plan.

On the Atlantic Record Date, 52,704,052 shares of Atlantic Common Stock were outstanding and entitled to vote. As of the Atlantic Record Date, directors, executive officers and their affiliates owned less than 1% of the issued and outstanding shares of Atlantic Common Stock.

*Proxies.* Holders of Atlantic Common Stock may vote either in person or by properly executed proxy. By completing and returning the form of proxy, the Atlantic stockholder authorizes the persons named therein to vote all of such Atlantic stockholder's shares on his or her behalf. All completed Atlantic proxies returned will be voted in accordance with the instructions indicated in the proxies. If no contrary instructions are given, the Atlantic proxies will be voted FOR approval of the Merger Agreement and the



transactions contemplated thereby and FOR the Company Plan. Any proxy given pursuant to this solicitation may be revoked by the person giving it at any time before it is voted (i) by delivery to the Secretary of Atlantic at 6801 Black Horse Pike, Egg Harbor Township, New Jersey 08234 on or before the taking of the vote at the Atlantic Meeting, a written notice of revocation bearing a later date than the proxy or a later dated proxy relating to the same shares of Atlantic Common Stock or (ii) by attending the Atlantic Meeting and voting in person. Attendance at the Atlantic Meeting will not in itself constitute a revocation of a proxy.

Atlantic will bear the cost of the solicitation of proxies for the Atlantic Meeting, except that Atlantic and Delmarva will share equally expenses incurred in connection with printing and filing this Joint Proxy Statement/Prospectus. See "The Merger Agreement—Expenses." Proxies may be solicited by certain officers and employees of Atlantic or its subsidiaries by mail, telephone, personally or by other communications without compensation apart from their normal salaries. Atlantic has retained Georgeson & Company Inc. to assist in soliciting proxies from Atlantic stockholders, including brokers accounts, at a fee for such services of approximately \$30,000 and reasonable out-of-pocket expenses.

The Atlantic Meeting may be adjourned one or more times to another date and/or place for any purpose (including, without limitation, for the purpose of soliciting additional proxies).

#### THE MERGERS

## **Background of the Mergers**

In this section, there are references to the following groups (comprising the persons listed):

#### Delmarva Working Group:

Between March 28 and April 26, 1996—Barbara S. Graham, Senior Vice President, Treasurer and Chief Financial Officer (lead); Paul Gerritsen, Vice President; Philip S. Reese, General Manager— Marketing (elected Vice President effective May 30, 1996); Thomas S. Shaw, Senior Vice President; and David M. Velazquez, Manager of Strategic Planning.

Between April 26 and July 3, 1996—During this period, the Delmarva working group was expanded to include: Randall V. Griffin, Senior Counsel; Daniel G. Tavani, Engineer, Business Unit Planning/Analyst; and Gary M. Zibinski, Manager of Financial Analysis.

Between July 3 and August 9, 1996—During this period, the Delmarva working group was expanded to include: Donald E. Cain, Vice President; Peter F. Clark, Assistant General Counsel; Joseph W. Ford, Senior Vice President; James P. Lavin, Comptroller and Chief Accounting Officer; Michael Ratchford, General Manager, Communications and Community Relations; and Dale G. Stoodley, Vice President and General Counsel.

#### Atlantic Executive Committee:

Jerrold L. Jacobs, Chairman of the Board and Chief Executive Officer; Michael J. Chesser, President and Chief Operating Officer (effective July 1, 1996), Senior Vice President (until July 1, 1996); Michael J. Barron, Vice President and Chief Financial Officer; James E. Franklin II, Vice President, Secretary and General Counsel; Meredith I. Harlacher, Jr., Vice President—Power System; Henry K. Levari, Vice President—External Affairs; Marilyn T. Powell, Vice President—Marketing; Scott B. Ungerer, Vice President—Enterprise; Ernest L. Jolly, Vice President—Human Resources and Transformation; and Robert L. Aveyard, Director, Information Technology and Communications Services.

### Atlantic Working Group:

Between April 3 and July 3, 1996—Michael J. Barron, Vice President and Chief Financial Officer (lead); James E. Franklin II, Vice President, Secretary and General Counsel; Meredith I. Harlacher, Jr., Vice President; Henry K. Levari, Vice President; Christopher Moschella, Manager of Business Planning; and Louis M. Walters, Treasurer.

Between July 3 and August 9, 1996—During this period, the Atlantic working group was expanded to include: Robert L. Aveyard, Director, Information Technology and Communications Services; Joseph M. Castaldi, Manager of External Affairs; Neely D. Crowley, Supervisor of Communications/Public Relations; Nancy J. Cunningham, Manager of Corporate Development; Louis A. DeCicco, Manager of Bulk Power Marketing; Frank E. DiCola, Vice President (AEE); Gary L. Hanson, Controller (ACE); Ernest L. Jolly, Vice President; Robert K. Marshall, Manager of Finance and Treasury Operations; Marilyn T. Powell, Vice President; and Deborah L. Turner-Fox, Manager of Human Resources.

Atlantic and Delmarva are neighboring utilities that have had a variety of working relationships on a wide range of matters over many years. These included joint minority ownership in a number of electric production facilities and co-membership in the Pennsylvania-New Jersey-Maryland Interconnection Association, a regional power pool.

The Energy Policy Act of 1992 (the "1992 Act"), which enhanced the authority of the Federal Energy Regulatory Commission (the "FERC") to order electric utilities to provide transmission service, has prompted new developments in the electric utility industry. The 1992 Act also created a new class of power producers, exempt wholesale generators, which are exempt from regulation under the 1935 Act. This exemption has increased the number of entrants into the wholesale electric generation market and so

increased competition in the wholesale segment of the electric utility industry. Pursuant to its authority under the 1992 Act, the FERC issued a number of orders in specific cases commencing in December 1993 directing utilities to provide transmission services. The FERC's actions have increased the availability of transmission services, thus creating significant competition in the wholesale power market. Other developments have resulted from policies at the SEC, which has liberalized its interpretation and administration of the 1935 Act in ways that have made mergers between utility companies less burdensome, thereby facilitating the creation of larger industry competitors. Moreover, regulatory bodies in a number of states have initiated proceedings to review the basic structure of the industry, and such activity has led industry observers to anticipate the advent of retail competition on a significant scale within the electric utility industry.

In the fall of 1995, following a number of general discussions between Atlantic's senior management and its financial advisors and legal counsel, among others, regarding the potential strategic value of acquisitions, alliances and mergers in the restructuring utility and energy services industry, Atlantic began investigations of strategic alternatives. Atlantic's long-term advisors, corporate counsel at Simpson Thacher & Bartlett ("Simpson Thacher"), which had previously advised Atlantic and briefed the Atlantic Board on its fiduciary duties in the context of a strategic transaction, and financial advisors at Morgan Stanley & Co. Incorporated ("Morgan Stanley"), were alerted to Atlantic's interest in pursuing discussions with individual target companies.

During 1995, Delmarva's senior management team participated in a series of retreats focused on the future direction of the industry and their implications for the company. Over the course of the last 12-18 months Delmarva consulted with various advisors, including its long-term legal advisor, LeBoeuf, Lamb, Greene & MacRae, L.L.P. ("LeBoeuf/NY"), regarding strategic opportunities including, among other things, alliances, joint ventures and acquisitions, and in June of 1995 acquired Conowingo Power Company from PECO Energy Company.

Over the course of their long business relationship, Mr. Howard E. Cosgrove, Chairman, President and Chief Executive Officer of Delmarva, and Mr. Jerrold L. Jacobs, Chairman of the Board and Chief Executive Officer of Atlantic, regularly met to discuss industry issues. At one such meeting, on February 21, 1996, Mr. Cosgrove raised the possibility of a merger of the two companies. At the time, Mr. Jacobs declined to pursue the discussions, primarily because Atlantic was in the process of investigating other alternatives. Later, Atlantic decided not to continue to consider these alternatives.

On March 4, 1996, Mr. Jacobs called Mr. Cosgrove, indicating his interest in commencing discussions that could lead to a merger or other business combination of the two companies. They met on March 7, 1996 to conduct exploratory discussions.

At a regularly scheduled Atlantic Board meeting on March 14, 1996, Mr. Jacobs advised the Atlantic Board of the possibility of a merger or other business combination with Delmarva.

On March 15, 1996, Mr. Jacobs and Mr. Michael J. Barron, Vice President and Chief Financial Officer of Atlantic, held a teleconference with representatives of Morgan Stanley to provide Morgan Stanley with an overview of discussions to date and to go over the due diligence process that would be involved in determining whether to effect a business combination with Delmarva.

At a regularly scheduled Delmarva Board meeting on March 28, 1996, Mr. Cosgrove advised the Delmarva Board of his discussions with Mr. Jacobs and interest in pursuing a possible merger or other business combination. Later that evening, Mr. Cosgrove and Mrs. Barbara S. Graham, Senior Vice President, Treasurer and Chief Financial Officer of Delmarva, met with representatives of Merrill Lynch, Pierce, Fenner & Smith Incorporated ("Merrill Lynch") to provide Merrill Lynch with an overview of discussions to date and to go over the due diligence process that would be involved in determining whether to effect a business combination with Atlantic. This meeting was followed by meetings on March 29 and April 1, 1996 between representatives of Delmarva and Merrill Lynch, as background for a meeting on April 4 that would initiate the first phase of the merger discussions.

On April 3, 1996, Mr. Jacobs met with Mr. Michael J. Chesser, then Senior Vice President, now President and Chief Operating Officer of Atlantic, and the Atlantic working group. On April 4, 1996, Messrs. Jacobs and Cosgrove met with the joint working groups and Merrill Lynch and Morgan Stanley to commence Phase I of the merger discussions. This phase was to include preliminary discussions of benefits at a conceptual level and the identification of issues that would need to be resolved before proceeding with a merger of the two companies. Subgroups consisting of representatives of Delmarva and Atlantic were established to address areas of interest and report back to a joint working group meeting within a few days.

On April 9, 1996, representatives of Delmarva and Atlantic met with Merrill Lynch and Morgan Stanley to assess the potential reaction of the capital markets to a combination of the companies. On April 10, 1996, Atlantic and Delmarva entered into a confidentiality agreement pursuant to which the parties agreed to provide nonpublic information to one another. A joint meeting was held on April 10, 1996 with Atlantic's long-term regulatory counsel at LeBoeuf, Lamb, Greene & MacRae, L.L.P. ("LeBoeuf/NJ"), to discuss potential legal regulatory issues. On April 11, 1996, members of management from both companies made presentations to a meeting at which representatives of Delmarva, Atlantic, Merrill Lynch and Morgan Stanley were present, covering the business plans of each company, potential benefits and other issues related to a potential business combination of the two companies. A teleconference was held on April 15, 1996 among members of the working group, Merrill Lynch, Morgan Stanley, LeBoeuf/NY and Potter Anderson & Corroon ("Potter Anderson"), Delmarva's long-term legal advisor and co-counsel to LeBoeuf/NY, and Simpson Thacher to discuss updates of the evaluations of synergies and legal and regulatory issues. At a meeting on April 16, 1996 of members of the Delmarva and Atlantic working groups at which Merrill Lynch and Morgan Stanley were present, the joint working group discussed a framework for analyzing potential merger synergies.

On April 17, 1996, another meeting of the working groups was held to discuss regulatory strategy and other matters, which Merrill Lynch, Morgan Stanley, LeBoeuf/NY, Potter Anderson and Simpson Thacher attended. Preliminary conclusions of Phase I were discussed. A preliminary report by Mr. Louis M. Walters, Treasurer of Atlantic, identifying potential merger synergies, was discussed. There was general agreement about the desire to move the discussions forward, but additional analysis was sought on potential obstacles to a business combination between Delmarva and Atlantic. On April 18, 1996 a teleconference was held among members of the working groups, Merrill Lynch and Morgan Stanley during which regulatory issues were discussed. At the conclusion of the teleconference there was a consensus that discussions of a potential business combination between Delmarva and Atlantic should be continued but that there was need for further study of issues requiring resolution.

On April 19, 1996, representatives of Delmarva and Atlantic met to prepare for a Phase I presentation to Messrs. Cosgrove, Jacobs and Chesser. This presentation was made by Mrs. Graham and Mr. Barron at a meeting held on April 22, 1996 at which Merrill Lynch and Morgan Stanley were present. It ended with a consensus to proceed with Phase IIa. This phase was to involve a focused investigation of the identified issues needing resolution in order to proceed with the transaction. These included the emerging regulatory environment and general valuation issues.

At a regularly scheduled Atlantic Board meeting on April 24, 1996, Morgan Stanley made a presentation which included an update on the potential transaction, a description of other comparable transactions and the results of their preliminary valuation analysis. Mr. Barron briefed the Atlantic Board on the status of the discussions with Delmarva. Mr. Walters gave a presentation on potential synergies that could be realized from a combination of the two companies based on a high-level preliminary review. A recommendation was made to proceed with Phase IIa. The Atlantic Board encouraged management to proceed with discussions.

On April 25, 1996, at a regularly scheduled Delmarva Board meeting, Mr. Cosgrove, Mrs. Graham and Mr. Thomas S. Shaw, Senior Vice President of Delmarva, provided the Delmarva Board with a strategic overview of the proposed merger. Representatives from Merrill Lynch and LeBoeuf/NY also commented on the proposed merger. A recommendation was made to proceed with Phase IIa. On April 26, 1996, directly following the Delmarva and Atlantic board meetings, Messrs. Jacobs and Cosgrove met to discuss the reactions of their boards and their expectations for going forward with discussions in Phase IIa.

A joint regulatory subgroup of the Delmarva and Atlantic working groups held a meeting on April 30, 1996 at which LeBoeuf/NJ provided a briefing on New Jersey regulation. On May 2, 1996, this subgroup also heard a presentation from The NorthBridge Group ("NorthBridge"), an economic consulting firm specializing in the utility industry, about the scope of a stranded cost review. Merrill Lynch and Morgan Stanley were also present at this meeting. The companies decided after the presentation to have their counsel jointly engage NorthBridge to do an evaluation of potential stranded costs arising in each of the companies. NorthBridge presented to the joint working group on May 15, 1996 its preliminary stranded costs review.

NorthBridge analyzed the potential for differential or asymmetric stranded cost exposure for Delmarva and Atlantic as a result of a transition to retail competition in the electric utility industry. Relative exposure was assessed within each of four regulatory scenarios involving less than full recovery of stranded costs: (i) administratively determined stranded costs, which are based on regulators' forecasts of market prices (assumed to be higher than actual market prices); (ii) a rate cap approach, in which rates are frozen for a period of time and then permitted to fall to market levels; (iii) the "California approach," in which stranded costs would be permitted to earn a lower return on equity than currently permitted; and (iv) staged retail wheeling, in which retail wheeling is made available to different classes of customers at different times. In scenarios (i), (ii) and (iv), determinations were made under two subscenarios, one with and one without the assumption of full recovery under purchased power contracts over the transition period; in scenario (iii), such recovery is assumed by definition. As a reference point for the study, an assessment was made using current rates and market prices assuming an immediate and full loss of all stranded costs.

NorthBridge used market price forecasts provided by Delmarva and Atlantic. NorthBridge found these prices to be within a reasonable range and used average figures for its base case (considering as well high and low cases at equal points above and below these averages). It also relied on the companies' information concerning production-related revenue requirements and forecasts of regulatory asset revenue requirements, purchased power obligations and load. Under each of the scenarios and subscenarios NorthBridge identified a potential for greater proportional exposure for Atlantic, except under the administrative scenario assuming that purchased power contracts were exempted from the transition, in which case the exposure was marginally higher for Delmarva. NorthBridge noted that the relative proportionalities were maintained over the full range of future market prices considered but that different results could occur where different jurisdictions adopted different approaches to deregulation. The information was used by the working groups as one point of reference in reflecting the potential impact of deregulation on the respective values of the companies.

NorthBridge was selected by Delmarva and Atlantic based on its experience, including its familiarity with trends in the electric utility industry. The fees of NorthBridge incurred in connection with its assignment were based on time spent plus expenses and are estimated at \$50,000.

The joint legal subgroup had held a meeting on May 8, 1996 to address a number of legal matters at which LeBoeuf/NJ was present.

Following this period of intense review of the potential obstacles to a merger of Atlantic and Delmarva, representatives of the two companies met with Merrill Lynch and Morgan Stanley on May 29, 1996. Discussions were held on the status of the regulatory analysis, the analysis of general stand-alone valuation issues and the likely reaction of the capital markets to an announcement of a combination of the two companies. The companies' working groups and advisors laid out a number of options, including having as a component of the merger consideration a "second security" (*i.e.*, a security in addition to the conventional common stock of the new company) that would be distributed to the stockholders of Atlantic to reflect the growth prospects of, and uncertainties associated with deregulation of, the regulated electric

utility business of Atlantic. The parties were considering the use of such a second security as a mechanism to address the difference in Delmarva's and Atlantic's evaluations of the overall impact of these growth prospects and uncertainties on the regulated electric utility business of Atlantic. The parties considered that the second security could take the form either of a "letter stock," *i.e.*, a common stock to be issued by the holding company that, following the Mergers, would own the businesses of both Delmarva and Atlantic but the performance of which would be tied in some manner to that of the regulated New Jersey electric utility business of Atlantic, or of a preferred stock that was in some way tied to the performance of such business.

At a regularly scheduled Delmarva Board meeting on May 30, 1996, Mrs. Graham provided a status report on Phase IIa of the discussions, including an overview of the preliminary NorthBridge study and an update on a review of Atlantic's business plan. The concept of a second security was also reviewed with the Delmarva Board. It was recommended, based on management's financial review of the proposed transaction to date, that Delmarva proceed with the next phase of discussions and analysis, assuming inclusion of a second security.

On May 31, 1996, the Atlantic Executive Committee met to discuss the status of the merger discussions, the business plan and related issues. On June 4, 1996, these individuals met again. At the conclusion of the June 4 session, the Atlantic Executive Committee determined that discussions should continue and that nearly two days of briefings on topics related to a potential business combination with Delmarva be presented to the Atlantic Board at its scheduled Board Retreat on June 12-13, 1996.

On June 7, 1996, Messrs. Cosgrove and Jacobs met to discuss the status of management structure issues and of the possible second security. This was the beginning of a series of negotiation sessions between the two chief executive officers.

At the June 12-13 Atlantic Board Retreat, Messrs. Jacobs and Barron, along with other members of Atlantic's management, reported on the status of negotiations, including those relating to a second security, such as a preferred or letter stock. The status of negotiations was reviewed against the backdrop of an in-depth discussion of Atlantic's long range business plans, including regulatory issues.

On June 26, 1996, the Delmarva Board was provided with an update of discussions with Atlantic. Representatives of LeBoeuf/NY and Potter Anderson reviewed the structural implications of being a registered holding company under the 1935 Act and Delmarva Board responsibilities in a merger of equals transaction. On June 27, 1996, a representative from NorthBridge presented to the Delmarva Board a preliminary review of stranded costs. This presentation was followed by an overview of a proposed structure for a second security by Merrill Lynch and an update by Delmarva management on its review of financial aspects of the proposed transaction. The meeting was concluded with an overview of the remaining areas of due diligence to be completed and a proposed timeline for completion of Phase II.

On July 2, 1996, Messrs. Jacobs and Cosgrove met and agreed to start Phase IIb discussions. Both companies had at this point agreed generally on the use of a second security, which could be either a common or preferred equity security. Phase IIb was to include due diligence investigations, detailed evaluation of synergies, formal pricing discussions and the development of a communications plan. Later that day, Mr. Jacobs met with Mr. Chesser and members of the Atlantic working group to commence Phase IIb.

On July 3, 1996, members of both working groups and Morgan Stanley, LeBoeuf/NY and Potter Anderson held a teleconference. Teams were formed to address a range of due diligence issues: business plans and valuation; regulatory issues; accounting, tax and financial systems; asset evaluation and operations; communication and information systems; human resources; marketing, communications and public relations; litigation; corporate documents; and environmental and real estate. During the July 3, 1996 teleconference, a decision was made to have counsel for Delmarva and Atlantic jointly engage Deloitte & Touche Consulting Group ("D&T Consulting Group"), a nationally recognized consulting firm with experience in utility mergers and acquisitions, to assist Delmarva and Atlantic management in identifying and quantifying the potential cost savings that could result from a business combination between the two

companies. D&T Consulting Group is a division of Deloitte & Touche LLP. Following the teleconference, the parties identified the additional representatives from each company that would be included in the working group for Phase IIb.

During July and in early August, intensive due diligence activities, including the exchange of documents between Delmarva and Atlantic and a series of meetings, were conducted by Delmarva and Atlantic. On July 19 and July 25, 1996, each Atlantic due diligence team presented internal due diligence reports to Mr. James E. Franklin II, Atlantic's General Counsel, identifying areas for continuing evaluation. These topics, in addition to any others identified, were worked to closure through the final stages of negotiation.

Delmarva held a series of meetings throughout Phase II in connection with completion of due diligence. On July 18, Delmarva met with Merrill Lynch, LeBoeuf/NY and Potter Anderson to discuss due diligence and the draft merger agreement. On July 19, 1996, representatives of Delmarva met with Potter Anderson to go over the status of due diligence.

Through a series of conference calls held July 15 through July 18, 1996 that included representatives of Delmarva and Atlantic and representatives of Merrill Lynch, Morgan Stanley, LeBoeuf/NY, Potter Anderson and Simpson Thacher, agreement was reached that the second security would take the form of a letter stock, *i.e.*, a common equity security, rather than a preferred stock.

During a joint meeting of the communications subgroups of the Delmarva and Atlantic teams on July 16, 1996, a decision was made that it was timely to engage a communications advisor knowledgeable in merger-related communications.

A meeting of Atlantic and Delmarva together with Merrill Lynch, Morgan Stanley, LeBoeuf/NY, Potter Anderson and Simpson Thacher was held on July 23, 1996. A number of smaller sessions were held among subgroups. There were detailed discussions about the terms of a draft merger agreement and the terms of the letter stock. In addition, members of Delmarva and Atlantic managements and D&T Consulting Group discussed the status of the potential synergies identification effort. At a communications subgroup session, at which Abernathy MacGregor & Associates ("Abernathy") discussed with management of Delmarva and Atlantic the necessary components of a communications plan, the parties agreed to jointly engage Abernathy to assist them in the development of a communication plan and in the preparation of communication materials in connection with the potential transaction.

On July 25, 1996, Messrs. Jacobs and Chesser of Atlantic were invited to a segment of the Delmarva Board meeting at which D&T Consulting Group, as a part of its assistance to the joint working group, discussed the joint analysis of potential synergies with the Delmarva Board, including the basic structure, process and content of a synergy analysis, generally described the type of synergies identified in other mergers, then explained the results to date of the joint synergies analysis. The evaluation included preliminary estimates of synergies, net of costs to achieve them, in excess of \$500 million over a 10-year period that might be obtained from a business combination of the two companies. After Messrs. Jacobs and Chesser and D&T Consulting Group left, the Delmarva Board was updated on the status of negotiations with Atlantic. In addition, Merrill Lynch gave a preliminary review of financial aspects of the proposed transaction, and LeBoeuf/NY and Potter Anderson provided the Delmarva Board with a summary of merger agreement issues. Updates were also provided on regulatory strategy, the second security and the status of Delmarva's review of financial aspects of the proposed transaction.

On July 26, 1996, Messrs. Jacobs and Barron of Atlantic and Mr. Cosgrove and Mrs. Graham of Delmarva met to conclude the negotiation of management structure issues and to begin to make progress on the parameters of the potential transaction, including the extent to which the merger consideration distributed to Atlantic's stockholders would include letter stock.

On July 30, 1996, the Atlantic Board was furnished with information regarding the proposed transaction including the proposed general parameters for the letter stock, in preparation for its August 5 board meeting.

At a meeting of the Delmarva Board on August 1, 1996, management provided an update on the status of discussions with Atlantic. A report was given on due diligence, which, although essentially complete, would continue through the final stages of negotiations. Additional presentations were made by LeBoeuf/NY on the draft merger agreement and the second security, followed by presentations by management and Merrill Lynch on financial aspects of the proposed transaction and a presentation by management on the analysis of potential synergies. In addition, the Delmarva Board was provided with an overview of communications plans.

On August 2, 1996, members of the Delmarva and Atlantic working groups met with D&T Consulting Group to review the final results of the analysis prepared by Delmarva and Atlantic with the assistance of D&T Consulting Group on potential synergies that could result in connection with a business combination of Delmarva and Atlantic.

Discussions continued over the next several days among members of management of each company, their financial advisors and their legal counsel regarding the completion of due diligence, valuation issues, separate and combined financial forecasts, regulatory issues, terms of the proposed merger agreement, human resources issues and management responsibilities.

During discussions regarding the proposed merger at the August 5, 1996 Atlantic Board meeting, D&T Consulting Group, as a part of its assistance to the joint working group, discussed the joint analysis of potential synergies with the Atlantic Board. Morgan Stanley and Simpson Thacher were present at this meeting and provided information on the results of due diligence, letter stock and examples of its use, and a number of other merger-related topics. Simpson Thacher also briefed the Atlantic Board on its responsibilities and obligations in a merger of equals. Mr. Cosgrove met with the Atlantic Board, following which Messrs. Cosgrove and Shaw and Mrs. Graham met informally with the Atlantic Board at a recess in the proceedings.

On August 7, 1996, Messrs. Jacobs and Barron of Atlantic met with Mr. Cosgrove and Mrs. Graham of Delmarva to engage in negotiations regarding pricing, dividend policy and the proposed name of the new company. There was still no agreement on material terms, but the parties agreed to proceed to inform their respective boards of directors.

At the Atlantic Board meeting on August 8, the Atlantic Board was briefed on the status of the negotiations and considered final presentations from management on the rationale for a business combination of Delmarva and Atlantic, including the potential benefits and the similarity of vision and strategy between the two companies. Morgan Stanley made a presentation which included a description of the letter stock and the results of their valuation analysis. Simpson Thacher was present at this meeting and made presentations to the Atlantic Board on a number of merger-related issues.

Prior to a meeting of the Atlantic Board on August 9, 1996, Mr. Cosgrove and Mrs. Graham and Messrs. Jacobs and Barron held a teleconference and continued their negotiations, in the end agreeing to take the status of the negotiations to their respective boards for discussion.

At the Atlantic Board meeting of August 9, 1996, detailed presentations were made by Morgan Stanley and management on the status of pricing negotiations. Simpson Thacher reviewed in detail with the Atlantic Board the terms of the Merger Agreement. The joint communication plan that would be put in place upon an approved merger was presented to the Atlantic Board by management and a representative of Abernathy. During the balance of the day additional telephone discussions were held between Messrs. Jacobs and Barron of Atlantic and Mr. Cosgrove and Mrs. Graham of Delmarva, and the parties agreed upon conversion ratios to take to their respective boards. Simpson Thacher made presentations to the Atlantic Board on a number of merger-related issues. Morgan Stanley made a presentation which included a summary of the terms of the transaction, a further description of the letter stock and the results of their valuation analysis. Morgan Stanley rendered to the Atlantic Board its oral opinion, which was subsequently confirmed in writing, to the effect that as of the date of such meeting the Atlantic Conversion Ratio taking into account the Delmarva Conversion Ratio was fair from a financial point of view to the

holders of Atlantic Common Stock. The Atlantic Board then approved the terms of the Merger Agreement, which was subsequently executed.

At the Delmarva Board meeting on the same day, management noted that due diligence had been concluded and that no issues had been identified that would preclude management's recommending that Delmarva proceed with the proposed merger; management further noted that the synergies analysis was finalized. Counsel to Delmarva reviewed in detail the transaction terms and representatives of Merrill Lynch reviewed various financial and other information and rendered to the Delmarva Board its opinion that, as of such date and based upon and subject to the matters discussed therein, the Delmarva Conversion Ratio was fair to Delmarva and its stockholders from a financial point of view. The Delmarva Board discussed with Merrill Lynch and with counsel the information and advice it had received at this and previous meetings. After such discussions, the Delmarva Board approved the terms of the Merger Agreement and the Merger Agreement was subsequently executed.

## **Reasons for the Mergers**

The primary rationale for the Mergers is that Delmarva and Atlantic share a common vision of the future as well as the strategic path necessary to succeed in the increasingly competitive utility and energy services marketplace.

Delmarva and Atlantic see the electric industry splitting into a commodity-like wholesale power generation market and a retail side where customers have service needs that extend beyond kilowatt hour supply. The companies also see the convergence of energy, energy-related services, electric utility services, and ultimately other utility services such as telecommunications, water and wastewater. The Mergers will better position the combined company to successfully implement strategies in the following ways:

- **Increased Scale**—As competition intensifies within the industry, Atlantic and Delmarva believe scale will be one parameter that will contribute to overall business success. Scale has importance in many areas, including utility operations, product development, advertising and corporate services. The Mergers are expected to improve the profitability of the combined company by roughly doubling the customer base and providing increased economies of scale in all of these areas.
- Cost Savings—Through the elimination of duplicative activities, increased scale and improved purchasing power, the companies expect to capture merger-related savings net of transaction costs in excess of \$500 million over the next ten years, based on preliminary estimates by the managements of Delmarva and Atlantic with the assistance of D&T Consulting Group. These estimated cost savings would be attributable to the Mergers and would not include other types of savings that might be achieved without a business combination of Delmarva and Atlantic. The estimated cost savings reflect the creation of cost reduction or cost avoidance opportunities through the ability to consolidate separate, stand-alone operations into a single entity. In addition, the Company will continue efforts already underway in both Delmarva and Atlantic to increase productivity and reduce costs by redesigning or re-engineering key business processes.
- Competitive Prices and Services—Sales to industrial, large commercial and wholesale customers are considered to be at greatest near-term risk as a result of increased competition in the electric utility industry. The Mergers will enable the Company to meet the challenges of the increased competition and will create operating efficiencies through which the Company will be able to provide more competitive prices to customers.
- More Balanced Customer Base—The Mergers will create a larger company with less reliance on the chemical and financial services industries, from Delmarva's perspective, and on casino gaming, tourism and recreation, from Atlantic's perspective. The combined service territories of Delmarva and Atlantic will be more diverse than their individual service territories, reducing the Company's exposure to adverse changes in any sector's economic and competitive conditions.

• **Financial Flexibility**—By roughly doubling the market capitalization of the Company compared with the individual companies, the Mergers should improve the Company's overall credit quality and liquidity of the securities issued by the holding company and therefore improve the Company's ability to fund continued growth.

• **Regional Platform for Growth**—The combination of Atlantic and Delmarva will create a regional platform in the mid-Atlantic corridor. The corridor is experiencing economic growth that is led by the casino gaming industry in South Jersey and the expansion of the financial services industry in Delaware. The Company plans to expand relationships with existing customers and to develop relationships with new customers in the region. The Company will use its combined distribution channels to market a portfolio of energy-related products throughout the region and will follow regional relationships to other geographical areas.

The estimated cost savings resulting from the Mergers referred to above are expected to be allocated among stockholders and customers. This allocation will depend upon the results of regulatory proceedings in the various jurisdictions in which Delmarva and Atlantic operate their utility businesses. See "Regulatory Matters."

The analyses employed in order to develop the preliminary estimates of potential cost savings resulting from the Mergers referred to above include forward-looking statements within the meaning of Section 21E of the Exchange Act. These forward-looking statements reflect numerous assumptions, which involve judgments with respect to a number of risks and uncertainties. Among the factors that could cause actual results to differ materially from those set forth above are: electric load and customer growth; abnormal weather conditions; available sources and cost of fuel and generating capacity; the speed and degree to which competition enters the power generation, wholesale and retail sectors of the electric utility industry; state and federal regulatory initiatives that increase competition, threaten cost and investment recovery and impact rate structures; the ability of the Company to reduce successfully its cost structure; operating performance of the nuclear generating facilities, decommissioning costs associated with such facilities and impact on future operational and financial condition associated with the uncertain status of the Salem Nuclear Generating Station; the degree to which the Company develops nonregulated business ventures; the economic climate and growth in the service territories of Delmarva and Atlantic following the Mergers; economies generated by the Mergers; inflationary trends and interest rates; and other risks detailed from time to time in the reports filed with the SEC by Delmarva and Atlantic. Accordingly, while Delmarva and Atlantic believe that such assumptions are reasonable for purposes of the development of estimates of potential savings, there can be no assurance that such assumptions will approximate actual experience or that all such savings will be realized.

#### **Recommendations of the Boards of Directors**

.

Delmarva. THE DELMARVA BOARD HAS UNANIMOUSLY APPROVED THE MERGER AGREEMENT AND DETERMINED TO RECOMMEND THE DELMARVA MERGER TO HOLD-ERS OF DELMARVA COMMON STOCK. THE DELMARVA BOARD UNANIMOUSLY RECOM-MENDS THAT THE HOLDERS OF DELMARVA COMMON STOCK VOTE FOR APPROVAL OF THE MERGER AGREEMENT AND THE TRANSACTIONS CONTEMPLATED THEREBY.

In considering the recommendation of the Delmarva Board with respect to the Merger Agreement, stockholders should be aware that certain members of the Delmarva Board will become directors and/or employees of the Company following consummation of the Mergers and/or may become entitled to severance benefits as a result of the Mergers. Therefore, such directors have interests in the Merger Agreement that are in addition to the interests of stockholders of Delmarva generally and that could potentially represent conflicts of interest. The Delmarva Board was aware of these interests and considered them, among other matters, in approving the Merger Agreement. See "—Conflicts of Interest of Certain Persons in the Mergers."

The Delmarva Board believes that the Mergers will provide strategic and operational opportunities and will enable Delmarva and its stockholders to participate in a company which, through the pooling of common stock equity, management, manpower and technical expertise and increased coordination in the use of the facilities of Delmarva and Atlantic, will be better able to meet the competitive environment for the delivery of energy and services. In addition, the Delmarva Board believes that the combined entity will be able, in the long term, to achieve benefits of increased financial stability and strength, improved and unified management, efficiencies of operations, better use of facilities for the benefit of customers and reduced or deferred requirements for future additional generating capacity. In addition to the benefits expected to be derived from the Mergers, the Delmarva Board considered the additional regulatory oversight that would result, including regulation under the 1935 Act and the possibility that the Company may be required to divest itself of its natural gas operations and/or certain nonutility enterprises, and the problems inherent in merging the operations of two large companies, including composition of the Company Board, but concluded that the benefits outweighed these disadvantages.

In reaching its conclusions, the Delmarva Board considered: (i) the amount of Company Common Stock to be received by holders of each share of Delmarva Common Stock and the inclusion of Class A Common Stock as a second security to be issued to the holders of Atlantic Common Stock pursuant to the Mergers, as more fully discussed below; (ii) the financial performance, condition, business operations and prospects of each of Delmarva and Atlantic and the fact that the Company will likely have greater financial stability and strength due to its participation in the economic climate and growth of both Delmarya and Atlantic's service territories, its increased economies of scale, the diversification of its customer base and the impact of the potential operating efficiencies and merger-related savings realized through the consolidation of Delmarva's and Atlantic's stand-alone operations, as more fully discussed above; (iii) the improved credit quality and liquidity of the Company as compared to Delmarva and Atlantic individually, which will enhance the Company's access to capital and ability to fund growth opportunities; (iv) current industry, economic, market and regulatory conditions that encourage consolidation to compete effectively in an increasingly deregulated environment (as described under "Background of the Mergers"); (y) the anticipated positive effects of the Mergers on stockholders through their ownership of stock in a stronger company as a result of the reasons described in (ii), (iii) and (iv) above and on customers through more competitive rates; (vi) the terms of the Merger Agreement, which provide for reciprocal representations and warranties, conditions to closing and rights to termination, balanced rights and obligations and protection for employees of Delmarva (as discussed under "The Merger Agreement"); (vii) the recent incidence of merger transactions involving electric utility companies in surrounding markets, which is part of a wider trend in the utility industry towards consolidation and strategic partnerships in order to create larger, stronger companies made to face an increasingly competitive environment; (viii) the impact of regulation under various state and federal laws (as described under "Background of the Mergers"); and (ix) the opinion of Merrill Lynch, described below, that the Delmarva Conversion Ratio is fair to Delmarva and its stockholders from a financial point of view. With regard to item (i) above, the Class A Common Stock was determined by the Delmarva Board to be necessary to bridge a difference in view between Delmarva and Atlantic on the appropriate conversion ratio for a business combination between the two companies. From Delmarva's perspective, this difference in view relates primarily to the impact of deregulation on the regulated utility business of Atlantic given its average cost of power relative to other potential suppliers in the region. The second security allocates proportionately more of the risks associated

with Atlantic's regulated utility business to Atlantic's current stockholders and, at the same time, provides them with the opportunity to participate in proportionately more of the growth prospects of Atlantic's regulated utility business. In approving the Merger Agreement and recommending that the common stockholders of Delmarva approve the Merger Agreement, the Delmarva Board considered the above factors as a whole and did not assign specific or relative weights to any one factor or group of factors.

Atlantic. THE ATLANTIC BOARD HAS UNANIMOUSLY APPROVED THE MERGER AGREEMENT AND DETERMINED TO RECOMMEND THE ATLANTIC MERGER TO HOLD-ERS OF ATLANTIC COMMON STOCK. THE ATLANTIC BOARD UNANIMOUSLY RECOM-MENDS THAT THE HOLDERS OF ATLANTIC COMMON STOCK VOTE FOR APPROVAL OF THE MERGER AGREEMENT AND THE TRANSACTIONS CONTEMPLATED THEREBY.

In considering the recommendation of the Atlantic Board with respect to the Merger Agreement, stockholders should be aware that certain members of the Atlantic Board will become directors and/or employees of the Company following consummation of the Mergers and/or may become entitled to severance benefits as a result of the Mergers. Therefore, such directors have interests in the Merger Agreement that are in addition to the interests of stockholders of Atlantic generally and that could potentially represent conflicts of interest. The Atlantic Board was aware of these interests and considered them, among other matters, in approving the Merger Agreement. See "—Conflicts of Interest of Certain Persons in the Mergers."

The Atlantic Board believes that the Mergers will provide strategic and operational opportunities and will enable Atlantic and its stockholders to participate in a company which, through the pooling of common stock equity, management, manpower and technical expertise and increased coordination in the use of the facilities of Atlantic and Delmarva, will be better able to meet the competitive environment for the delivery of energy and services. In addition, the Atlantic Board believes that the combined entity will be able, in the long term, to achieve benefits of increased financial stability and strength, improved and unified management, efficiencies of operations, better use of facilities for the benefit of customers and reduced or deferred requirements for future additional generating capacity. In additional regulatory oversight that would result, including regulation under the 1935 Act and the possibility that the Company may be required to divest itself of Delmarva's natural gas operations and/or certain nonutility enterprises, and the problems inherent in merging the operations of two large companies, including composition of the Company Board, but concluded that the benefits outweighed these disadvantages.

In reaching its conclusions, the Atlantic Board considered: (i) the amount of Company Common Stock and Class A Common Stock to be received by holders of each share of Atlantic Common Stock, as discussed more fully below; (ii) the fact that the inclusion of the Class A Common Stock permitted Atlantic and Delmarva to proceed with a mutually beneficial transaction despite the two companies' different evaluations of the growth prospects of, and uncertainties associated with deregulation of, the regulated electric utility business of Atlantic; (iii) the financial performance, condition, business operations and prospects of each of Atlantic and Delmarva and the fact that the Company will likely have greater financial stability and strength due to its participation in the economic climate and growth of both Atlantic and Delmarva's service territories, its increased economies of scale, the diversification of its customer base and the impact of the potential operating efficiencies and merger-related savings realized through the consolidation of Atlantic's and Delmarva's stand-alone operations, as more fully discussed above; (iv) the improved credit quality and liquidity of the Company as compared to Atlantic and Delmarva individually, which will enhance the Company's access to capital and ability to fund growth opportunities; (v) current industry, economic, market and regulatory conditions, which encourage consolidation to compete effectively in an increasingly deregulated environment (as described under "Background of the Mergers"); (vi) the anticipated positive effects of the Mergers on stockholders through their ownership of stock in a stronger company for the reasons described in (iii), (iv) and (v) above and on customers through more competitive rates; (vii) the terms of the Merger Agreement, which provide for reciprocal representations

and warranties, conditions to closing and rights to termination, balanced rights and obligations and protection for employees of Atlantic (as discussed under "The Merger Agreement"); (viii) the recent wave of merger activity involving electric utility companies in surrounding markets, which is part of a wider trend in the utility industry towards consolidation and strategic partnerships in order to create larger, stronger companies ready to face an increasingly competitive environment; (ix) the impact of regulation under various state and federal laws (as described under "Background of the Mergers"); and (x) the opinion of Morgan Stanley, described below, that the Atlantic Conversion Ratio, taking into account the Delmarva Conversion Ratio, is fair from a financial point of view to the holders of the Atlantic Common Stock. With regard to item (i) above, the Class A Common Stock was determined by the Atlantic Board to be necessary to bridge a difference in view between Atlantic and Delmarva on the appropriate conversion ratio for a business combination between the two companies. The second security allocates proportionately more of the risks associated with Atlantic's regulated utility business to Atlantic's current stockholders and, at the same time, provides them with the opportunity to participate in proportionately more of the growth prospects of Atlantic's regulated utility business. In approving the Merger Agreement and recommending that the stockholders of Atlantic approve the Merger Agreement, the Atlantic Board considered the above factors as a whole and did not assign specific or relative weights to any one factor or group of factors.

# Opinion of Financial Advisor to Delmarva

#### Delmarva

Merrill Lynch has delivered its written opinions to the Delmarva Board, dated August 9, 1996 and the date of this Joint Proxy Statement Prospectus, to the effect that, as of the dates of such opinions, and based upon the assumptions made, matters considered and limits of review, as set forth in such opinions, the Delmarva Conversion Ratio (as defined herein) contemplated by the Merger Agreement is fair to Delmarva and its stockholders from a financial point of view. References herein to the "Merrill Lynch Opinion" refer to the written opinion of Merrill Lynch dated August 9, 1996.

A copy of the Merrill Lynch opinion dated the date of this Joint Proxy Statement/Prospectus, which sets forth the assumptions made, matters considered and certain limitations on the scope of review undertaken by Merrill Lynch, is attached as Annex II to this Joint Proxy Statement/Prospectus. Delmarva stockholders are urged to read this opinion in its entirety. The Merrill Lynch opinions are directed only to the fairness of the Delmarva Conversion Ratio from a financial point of view and do not constitute a recommendation to any Delmarva stockholder as to how such stockholder should vote at the Delmarva Meeting. The summary of the Merrill Lynch opinion set forth in this Joint Proxy Statement/Prospectus is qualified in its entirety by reference to the full text of the Merrill Lynch opinion attached as Annex II hereto.

The Merrill Lynch opinion dated the date of this Joint Proxy Statement/Prospectus is substantially similar to the Merrill Lynch Opinion dated August 9, 1996.

In arriving at the Merrill Lynch Opinion, Merrill Lynch among other things (i) reviewed Delmarva's and Atlantic's Annual Reports, Forms 10-K and related financial information for the three fiscal years ended December 31, 1995, and Forms 10-Q and related unaudited financial information for the quarterly periods ended March 31, 1996; (ii) reviewed certain other filings with the SEC made by Delmarva and Atlantic, including proxy statements, Forms 8-K, and registration statements, during the last three years; (iii) reviewed certain information, including financial forecasts, relating to the business, earnings, dividends, cash flow, assets and prospects of Delmarva and Atlantic, furnished to Merrill Lynch by Delmarva and Atlantic concerning their respective businesses, regulatory environments, prospects and strategic objectives and possible operating, administrative and capital synergies which might be realized for the combined companies following the Mergers; (v) reviewed the historical market prices and trading activity for Delmarva Common Stock and Atlantic Common Stock; (vi) compared the results of operations of Delmarva and Atlantic with those of certain companies deemed by Merrill Lynch to be reasonably similar

to Delmarva and Atlantic, respectively; (vii) compared the proposed financial terms of the Mergers with the financial terms of certain other mergers and acquisitions which Merrill Lynch deemed to be relevant; (viii) considered the pro forma effect of the Mergers, in terms of net income available to common stockholders, dividends per common share, book value per common share and capitalization, on the Delmarva Common Stock; (ix) reviewed a draft of the Merger Agreement dated August 7, 1996; (x) reviewed an analysis prepared by Atlantic and Delmarva with the assistance of a third-party consultant regarding estimates of the amount and timing of the cost savings expected to be derived from the Mergers; and (xi) reviewed such other financial studies and analyses and made such other inquiry and took into account such other matters deemed necessary or appropriate by Merrill Lynch for purposes of the Merrill Lynch Opinion.

In preparing the Merrill Lynch opinions, Merrill Lynch relied on the accuracy and completeness of all information supplied or otherwise made available to it by Delmarva and Atlantic, and did not independently verify such information or any underlying assumptions. Merrill Lynch did not undertake an independent appraisal or physical inspection of the assets or liabilities (contingent or otherwise) of Delmarva or Atlantic. Merrill Lynch also assumed that the financial forecasts and estimated synergies furnished to it by Delmarva and Atlantic were reasonably prepared and reflected the best currently available estimates and judgments of Delmarva's and Atlantic's management as to the expected future financial performance of Delmarva and Atlantic, respectively, and as to the expected future projected outcomes of various legal, regulatory and other contingencies. Merrill Lynch also assumed that the estimates relating to the impact of regulatory change and competition, furnished by Delmarva and Atlantic, have been reasonably prepared and reflect the best currently available estimates and judgment of Delmarva's and Atlantic's management as to the expected future projected outcomes relating to such matters. Merrill Lynch also assumed that the Mergers will be free of federal tax to Delmarva, the Company, DS Sub, Atlantic and the respective holders of Delmarva Common Stock and Atlantic Common Stock. Merrill Lynch's opinions are based upon general economic, market, monetary and other conditions as they existed and could be evaluated, and the information made available to it, as of the respective dates of such opinions. The Merrill Lynch opinions are directed to the Delmarva Board and do not constitute a recommendation to any Delmarva stockholder as to how such stockholder should vote at the Delmarva Meeting.

The matters considered by Merrill Lynch in arriving at the Merrill Lynch opinions are based on numerous macroeconomic, operating and financial assumptions with respect to industry performance, general business and economic conditions, many of which are beyond the control of Delmarva and Atlantic, and involve the application of complex methodologies and educated judgment. Any estimates incorporated in the analyses performed by Merrill Lynch are not necessarily indicative of actual past or future results or values, which may be significantly more or less favorable than such estimates. Estimated values do not purport to be appraisals and do not necessarily reflect the prices at which businesses or companies may be sold in the future. The Merrill Lynch opinions do not present a discussion of the relative merits of the Mergers as compared to any other business plan or opportunity that might be presented to Delmarva, or the effect of any other arrangement in which Delmarva might engage.

The following is a summary of certain financial and comparative analyses performed by Merrill Lynch in arriving at its August 9, 1996 opinion.

Trading Ratio Analysis. Merrill Lynch reviewed the performance of the per share market price of Delmarva Common Stock and Atlantic Common Stock over the five year period ended August 2, 1996. Merrill Lynch also calculated the ratio of the per share market price of Atlantic Common Stock to the per share market price of Delmarva Common Stock from August 2, 1991 to August 2, 1996, August 2, 1993 to August 2, 1996, and August 2, 1995 to August 2, 1996. This analysis showed that over the five year period, the per share market price of Atlantic Common Stock compared to the per share price of Delmarva Common Stock, traded at an average ratio of 0.950. Over the three year period this analysis showed that the per share market price of Atlantic Common Stock compared to the per share price of Delmarva

Common Stock, traded at an average ratio of 0.920. Over the one year period this analysis showed that the per share market price of Atlantic Common Stock compared to the per share price of Delmarva Common Stock, traded at an average ratio of 0.853. Based on the August 8, 1996 closing prices, the trading ratio of Atlantic Common Stock was 0.831.

*Contribution Analysis.* In order to determine an implied exchange ratio range based upon contribution analysis, Merrill Lynch calculated the contribution of each of Delmarva and Atlantic to the pro forma combined Company with respect to (i) earnings per share of common stock, (ii) dividends per share of common stock, and (iii) common equity per share of common stock, for the years 1993 through 1995 and, using certain projections provided by the respective managements of Delmarva and Atlantic, for the years 1996 through 2000. The analysis of earnings per share of common stock for the years 1993 through 2000, yielded a range of implied exchange ratios for Atlantic Common Stock to Delmarva Common Stock of 0.82 to 1.03. The analysis of dividends per share of common stock for the years 1993 through 2000, yielded an implied exchange ratio for Atlantic Common Stock to Delmarva Common Stock of 1.00. The analysis of common equity per share of common stock to Delmarva Common Stock of 1.00. The analysis of common equity per share of common stock to Delmarva Common Stock of 1.00. The analysis of common equity per share of common stock to Delmarva Common Stock of 1.00. The analysis of effective exchange ratios for Atlantic Common Stock to Delmarva Common Stock of 0.99 to 1.04.

Comparable Publicly Traded Company Analysis. Using publicly available information, Merrill Lynch compared certain financial and operating information and ratios (described below) for Delmarva and Atlantic, respectively, with the corresponding financial and operating information and ratios for separate groups of publicly traded companies that Merrill Lynch deemed to be reasonably comparable to Delmarva and Atlantic, respectively. The companies included in the Delmarva comparable company analyses were: Baltimore Gas and Electric Company, GPU, Inc., MidAmerican Energy Company, PECO Energy Company, Potomac Electric Power Company, PP&L Resources, Inc., Public Service Enterprise Group Incorporated and WPS Resources Corporation (collectively, the "Delmarva Comparables"). The companies included in the Atlantic comparable company analyses were: Baltimore Gas and Electric Company, Boston Edison, General Public Utilities, Nevada Power Company, PECO Energy Company, Potomac Electric Power Company, PP&L Resources, Inc. and Public Service Enterprise Group Incorporated (collectively, the "Atlantic Comparables"). Merrill Lynch selected the companies in the Delmarva Comparables and Atlantic Comparables, respectively, from the universe of possible comparable utility companies based upon Merrill Lynch's views as to the comparability of financial and operating characteristics of these companies to Delmarva and Atlantic, respectively. ana ji tu a

In order to determine an implied effective exchange ratio range based upon comparable publicly traded company analysis, Merrill Lynch compared the market value of Delmarva Common Stock and Atlantic Common Stock as a multiple of (a) estimated 1997 earnings per share ("EPS"), which estimates were obtained from management forecasts (the "1997 EPS Ratio"), and (b) book value of common equity as of March 31, 1996, the most recently available fiscal quarter (the "Common Equity Ratio"), to the corresponding ratios for each of the Delmarva Comparables and Atlantic Comparables. The results of the foregoing were; (i) the 1997 EPS Ratio resulted in a range of implied effective exchange ratios for Atlantic Common Stock to Delmarva Common Stock of 0.77 to 1.10, and (ii) the Common Equity Ratio resulted in a range of implied effective exchange ratios for each of 0.86 to 1.15.

Merrill Lynch then calculated the implied effective exchange ratio of Atlantic Common Stock to Delmarva Common Stock using varying sensitivities relating to the estimated impact of regulatory change and competition furnished by the managements of Delmarva and Atlantic, and with and without certain synergies estimated by the managements of Delmarva and Atlantic. Utilizing comparable publicly traded company analysis and the varying sensitivities mentioned, Merrill Lynch calculated implied effective exchange ratio ranges for Atlantic Common Stock to Delmarva Common Stock of 0.44 to 1.34.

Discounted Cash Flow Analysis. In order to determine an implied exchange ratio range based upon discounted cash flow ("DCF") analysis, Merrill Lynch performed unlevered DCF analyses for the distinct

5 . 1 . 1 . I

businesses of Delmarva and Atlantic using the same management projections, and calculated ranges of value for Delmarva Common Stock and Atlantic Common Stock.

Merrill Lynch performed separate DCF analyses for the Delmarva utility business and the Delmarva unregulated businesses. The Delmarva utility business DCF was based upon the discount to present value, assuming discount rates ranging from 8.0% to 9.0%, of (i) its projected unlevered free cash flow for the years 1996 through 2000, and (ii) its 2000 value based upon a range of multiples from 11.5 to 12.5 times its projected 2000 net income, and 1.3 to 1.4 times its projected 2000 book value, plus in each case, assumed debt and preferred stock at year-end 2000. Merrill Lynch calculated a range of value for the Delmarva unregulated businesses using a variety of DCF and other valuation analyses. Based on its DCF analyses, Merrill Lynch calculated a range of value for \$1,376 million.

Merrill Lynch performed separate DCF analyses for the Atlantic utility business and the Atlantic unregulated businesses. The Atlantic utility business DCF was based upon the discount to present value, assuming discount rates ranging from 8.0% to 9.0%, of (i) its projected unlevered free cash flow for the years 1996 through 2000, and (ii) its 2000 value based upon a range of multiples from 11.5 to 12.5 times its projected 2000 net income, and 1.3 to 1.4 times its projected 2000 book value, plus in each case, assumed debt and preferred stock at year-end 2000. Merrill Lynch calculated a range of value for the Atlantic unregulated businesses using a variety of DCF and other valuation analyses. Based on its DCF analyses, Merrill Lynch calculated a range of value for Atlantic Common Stock of \$935 million to \$1,212 million.

Merrill Lynch then calculated the implied effective exchange ratio of Atlantic Common Stock to Delmarva Common Stock using varying sensitivities relating to the estimated impact of regulatory change and competition furnished by the managements of Delmarva and Atlantic, and with and without certain synergies estimated by the managements of Delmarva and Atlantic. Utilizing DCF analysis and the varying sensitivities mentioned, Merrill Lynch calculated implied effective exchange ratio ranges for Atlantic Common Stock to Delmarva Common Stock of 0.43 to 1.38.

Comparable Merger Transactions Analysis. Using publicly available information, Merrill Lynch reviewed 11 transactions announced between March 16, 1990 and January 22, 1996, involving the acquisition of selected electric utility companies (the "Comparable Merger Transactions"). The Comparable Merger Transactions and the date the transaction was announced were as follows: Kansas City Power & Light Co.'s merger with UtiliCorp United Inc. (January 1996), WPL Holdings, Inc.'s merger with IES Industries Inc. and Interstate Power Company (November 1995), Baltimore Gas and Electric Company's merger with Potomac Electric Power Company (September 1995), Public Service Company of Colorado's merger with Southwestern Public Service Company (August 1995), Northern States Power Company's merger with Wisconsin Energy Corporation (May 1995), Midwest Resources Inc.'s merger with Iowa-Illinois Gas & Electric Company (July 1994), Washington Water Power Company's merger with Sierra Pacific Resources (June 1994), Cincinnati Gas & Electric Company's merger with PSI Resources, Inc. (August 1993), IE Industries, Inc.'s merger with Iowa Southern, Inc. (February 1991), Kansas Power & Light Company's merger with Kansas Gas & Electric Company (October 1990), and Iowa Resources, Inc.'s merger with Midwest Energy Company (March 1990).

In order to determine an implied exchange ratio range based on Comparable Merger Transactions analysis, Merrill Lynch (i) compared the offer value in each of the Comparable Merger Transactions as a multiple of the then publicly available (a) latest 12 months ("LTM") net income available to common stock (the "Net Income Multiple"), and (b) book value of common equity for the most recently available fiscal quarter preceding such transaction (the "Book Value Multiple") to the corresponding multiples for Delmarva Common Stock and Atlantic Common Stock. The results of the foregoing were: (i) the Net Income Multiple resulted in a range of implied effective exchange ratios for Atlantic Common Stock to Delmarva Common Stock of 0.70 to 1.09, and (ii) the Book Value Multiple resulted in a range of implied effective exchange ratios of 0.82 to 1.21.



Merrill Lynch then calculated the implied effective exchange ratio of Atlantic Common Stock to Delmarva Common Stock using varying sensitivities relating to the estimated impact of regulatory change and competition furnished by the managements of Delmarva and Atlantic, and with and without certain synergies estimated by the managements of Delmarva and Atlantic. Utilizing the Comparable Merger Transactions analysis and the varying sensitivities mentioned, Merrill Lynch calculated implied effective exchange ratio ranges for Atlantic Common Stock to Delmarva Common Stock of 0.40 to 1.39.

*Pro Forma Analysis.* Merrill Lynch also analyzed certain pro forma effects resulting from the Mergers, including the potential impact to earnings per share of Delmarva Common Stock. Using the projected earnings for the years 1998 through 2000 provided by the respective managements of Delmarva and Atlantic, Merrill Lynch compared the projected earnings per share of Delmarva on a stand-alone basis assuming the Mergers do not occur, to the earnings per share of Company Common Stock assuming varying sensitivities to certain estimated synergies provided by the respective managements of Delmarva and Atlantic. The analysis indicated that the Mergers would be accretive to the projected earnings per share of a Delmarva stockholder in each of 1998, 1999, and 2000.

The summary set forth above does not purport to be a complete description of the analyses performed by Merrill Lynch in arriving at the Merrill Lynch Opinion. The preparation of a fairness opinion is a complex process not necessarily susceptible to partial or summary description. Merrill Lynch believes that its analyses must be considered as a whole and that selecting portions of its analyses and of the factors considered by it, without considering all such factors and analyses, could create a misleading view of the process underlying its analyses set forth in the Merrill Lynch Opinion. No company in the Delmarva Comparables or the Atlantic Comparables is identical to Delmarva or Atlantic, respectively, and none of the Comparable Merger Transactions is identical to the Mergers. Accordingly, an analysis of comparable publicly traded companies and comparable merger transactions is not mathematical; rather, it involves complex considerations and judgments concerning differences in financial and operating characteristics of the comparable companies and other factors that could affect the public trading value of the comparable comparable companies or the value being compared.

The Delmarva Board selected Merrill Lynch to render a fairness opinion because Merrill Lynch is an internationally recognized investment banking firm with substantial experience in transactions similar to the Mergers and because it is familiar with Delmarva and its business. Merrill Lynch has from time to time rendered investment banking, financial advisory and other services to Delmarva and Atlantic, for which it has received customary compensation. As part of its investment banking business, Merrill Lynch is continually engaged in the valuation of businesses and their securities in connection with mergers and acquisitions.

Pursuant to the terms of an engagement letter dated March 28, 1996, Delmarva paid to Merrill Lynch a \$150,000 retainer fee and an additional fee of \$1,125,000 upon the execution of the Merger Agreement. In addition, Delmarva has agreed to pay Merrill Lynch a fee of \$1,125,000 upon the approval of the Mergers by the stockholders of Delmarva, and a fee of \$2,250,000 upon consummation of the Mergers, to which the \$150,000 retainer fee already paid will be credited. Delmarva has also agreed to reimburse Merrill Lynch for its reasonable out-of-pocket expenses, including all reasonable fees and disbursements of its legal counsel, and to indemnify Merrill Lynch and certain related persons against certain liabilities in connection with its engagement, including certain liabilities under the federal securities laws.

In the ordinary course of Merrill Lynch's business, Merrill Lynch may actively trade the securities of Delmarva and Atlantic for its own account and for the accounts of its customers and, accordingly, may at any time hold a long or short position in such securities.

## **Opinion of Financial Advisor to Atlantic**

In April, 1996, Morgan Stanley was retained by Atlantic to act as its financial advisor in connection with the Merger. Morgan Stanley is an internationally recognized investment banking firm and was

selected by Atlantic based on Morgan Stanley's experience and expertise. In connection with Morgan Stanley's engagement Atlantic requested that Morgan Stanley evaluate the fairness of the Atlantic Conversion Ratio, taking into account the Delmarva Conversion Ratio, from a financial point of view to the holders of Atlantic Common Stock. On August 9, 1996, Morgan Stanley rendered to Atlantic's Board of Directors an oral opinion, which was subsequently confirmed in writing, to the effect that, as of such date and based on and subject to certain matters stated therein, the Atlantic Conversion Ratio, taking into account the Delmarva Conversion Ratio account the Delmarva Conversion Ratio, was fair from a financial point of view to the holders of the Atlantic Common Stock. Morgan Stanley subsequently confirmed its opinion of August 9, 1996 by delivery of a written opinion dated the date hereof.

The full text of Morgan Stanley's written opinion dated the date hereof, which sets forth the assumptions made, matters considered with limitations on the review undertaken, is attached as Annex III of this Joint Proxy Statement/Prospectus and is incorporated herein by reference. Holders of Atlantic Common Stock are urged to, and should, read this opinion carefully in its entirety. Morgan Stanley's opinion addresses only the fairness of the Atlantic Conversion Ratio, taking into account the Delmarva Conversion Ratio, from a financial point of view to the holders of Atlantic Common Stock, and it does not address any other aspect of the Mergers nor does it constitute a recommendation to any holder of Atlantic Common Stock as to how to vote at the Atlantic Meeting. The summary of the opinion of Morgan Stanley set forth in this Joint Proxy Statement/Prospectus is qualified in its entirety by reference to the full text of such opinion.

In arriving at this opinion, Morgan Stanley: (i) analyzed certain publicly available financial statements and other information of Atlantic and Delmarva, respectively; (ii) analyzed certain internal financial statements and other financial and operating data concerning Atlantic and Delmarva prepared by their respective managements; (iii) analyzed certain financial projections of Atlantic and Delmarva prepared by their respective managements; (iv) discussed the past and current operations and financial condition and the prospects of Atlantic and Delmarva with senior executives of Atlantic and Delmarva, respectively; (v) reviewed the reported prices and trading activity of both Atlantic Common Stock and Delmarva Common Stock; (vi) compared the financial performance of Atlantic and Delmarva and the prices and trading activity of Atlantic Common Stock and Delmarva Common Stock with that of certain other comparable publicly traded companies and their securities; (vii) reviewed the financial terms, to the extent publicly available, of certain comparable merger or acquisition transactions; (viii) analyzed the pro forma financial impact of the Mergers on Atlantic; (ix) participated in discussions and negotiations among representatives of Atlantic and Delmarva and their respective financial and legal advisors; (x) reviewed the Merger Agreement and certain related documents; (xi) reviewed and discussed with Atlantic and Delmarva an analysis prepared by Atlantic and Delmarva with the assistance of a third-party consultant regarding estimates of the amount and timing of the cost savings estimated to be derived from the Mergers; and (xii) performed such other analyses and examinations and considered such other factors as Morgan Stanley deemed appropriate.

In rendering its opinion, Morgan Stanley assumed and relied upon without independent verification the accuracy and completeness of the information reviewed by Morgan Stanley for the purposes of its opinion. With respect to the financial projections and estimates of the cost savings expected to be derived in the Mergers, Morgan Stanley assumed that they were reasonably prepared on bases reflecting the best currently available estimates and judgments of the future financial performance of the Company, Atlantic and Delmarva, respectively. Morgan Stanley did not make any independent valuation or appraisal of the assets or liabilities of Atlantic or Delmarva. In addition, Morgan Stanley assumed that the Mergers will be consummated in accordance with the terms set forth in the Merger Agreement, including, among other things, that the Mergers will be treated as a tax-free reorganization and/or exchange, each pursuant to the Internal Revenue Code of 1986, as amended (the "Code"). Morgan Stanley's opinion is necessarily based on economic, market and other conditions as in effect on, and the information made available to it as of, the date of its opinion.

In arriving at its opinion, Morgan Stanley assumed that in connection with the receipt of all the necessary regulatory and governmental approvals for the proposed Mergers, no restriction will be imposed that would have a material adverse effect on the contemplated benefits expected to be derived in the proposed Mergers. In addition, Morgan Stanley was not authorized to solicit, and did not solicit, interest from any party with respect to a merger with or other business combination transaction involving Atlantic or any of its assets, nor did Morgan Stanley have discussions or negotiations with any parties, other than Delmarva, in connection with the Mergers.

The following is a brief summary of certain analyses performed by Morgan Stanley and reviewed with the Atlantic Board on August 9, 1996 in connection with Morgan Stanley's presentation and opinion to the Atlantic Board on such date:

Comparable Publicly Traded Company Analysis. As part of its analysis, Morgan Stanley compared certain financial information of Atlantic with that of a group of publicly traded electric utility companies, including Public Service Enterprise, General Public Utilities, Orange & Rockland Utilities, Rochester Gas & Electric, and United Illuminating (collectively, the "Atlantic Comparable Companies") and also compared certain financial information of Delmarva with that of a group of publicly traded electric utility companies, including Baltimore Gas and Electric Company, PP&L Resources, Public Service Enterprise, and Dominion Resources (collectively, the "Delmarva Comparable Companies"). Such financial information included price to LTM ended March 31, 1996, forecasted 1996 and forecasted 1997 earnings multiples, price to book value multiple, price to LTM operating cash flow multiple and dividend yield. In particular, such analyses indicated that as of August 7, 1996 and based on a compilation of earnings projections by securities research analysts as of July 27, 1996, Atlantic and Delmarva traded at 10.2 and 11.5 times historical LTM earnings, respectively, 10.3 and 11.8 times forecasted earnings for the calendar year 1996, respectively, 10.2 and 11.2 times forecasted earnings for the calendar year 1997, respectively, 1.21 and 1.39 times book value as of the quarter ended March 31, 1996, respectively, 5.0 times historical LTM operating cash flow for each company, and a 8.0% and a 7.1% dividend yield, respectively. Morgan Stanley noted that, based on a compilation of earnings projections by securities research analysts as of July 27, 1996, the Atlantic Comparable Companies and Delmarva Comparable Companies traded in a range of 8.8 to 10.7 times and 10.8 to 11.9 times historical LTM earnings, respectively, 9.8 to 10.7 times and 10.9 to 12.1 times 1996 forecasted earnings, respectively, 9.4 to 10.2 times and 10.7 to 11.6 times 1997 forecasted earnings, respectively, 1.15 to 1.26 times and 1.28 to 1.39 times book value as of the quarter ended March 31, 1996, respectively, and had a 7.3% to 8.0% and a 6.5% to 7.7% dividend yield, respectively.

*Trading Ratio Analysis.* Morgan Stanley also reviewed the ratio of the Atlantic Common Stock to Delmarva Common Stock trading prices over varying intervals of time over the latest five years. Based on the closing price of Atlantic Common Stock and Delmarva Common Stock on August 7, 1996 of \$17.50 and \$20.75, respectively, the ratio was 0.84.

Discounted Cash Flow Analysis. Morgan Stanley performed DCF analyses of Atlantic and Delmarva based on certain financial projections provided by the respective managements of each company for the period 1996 through 2000. Unlevered free cash flows of each company were calculated as net income available to common stockholders plus the aggregate of preferred stock dividends, depreciation and amortization, deferred taxes, and other noncash expenses and after-tax net interest expense less the sum of capital expenditures and investment in noncash working capital. Morgan Stanley calculated terminal values by applying a range of multiples to the net income in fiscal 2000 and the cash-flow streams and terminal values were then discounted to the present using a range of discount rates representing an estimated range of the weighted average cost of capital for each of Atlantic ranging from \$14.05 to \$20.09 and for Delmarva ranging from \$22.54 to \$26.52.

Discounted Dividend Analysis. Morgan Stanley performed discounted dividend analyses of Atlantic and Delmarva based on certain financial projections provided by the respective managements of each company for the period 1996 through 2000. Morgan Stanley calculated terminal values by applying a range of multiples to the EPS in fiscal year 2000 and the dividend streams and terminal values were then discounted to the present using a range of discount rates representing an estimated range of the cost of equity for each of Atlantic and Delmarva. Based on this analysis, Morgan Stanley calculated median per share values for Atlantic ranging from \$20.26 to \$22.37 and for Delmarva ranging from \$21.60 to \$24.66.

Analysis of Selected Precedent Transactions. Using publicly available information, Morgan Stanley reviewed the following proposed or completed transactions constituting mergers of equals in the electric utility industry: UtiliCorp United and Kansas Power & Light Co., IES Industries and WPL Holdings, Potomac Electric Power and Baltimore Gas and Electric Company, Southwestern Public Service Co. and Public Service Co. of Colorado, Northern States Power Co. and Wisconsin Energy Corp., and Iowa-Illinois Gas & Electric Co. and Midwest Resources, Inc. (collectively, the "Electric Utility MOE Transactions"). Morgan Stanley compared certain financial and market statistics of the Electric Utility MOE Transactions. The premium to unaffected market price (*i.e.*, the market price one month prior to the announcement of the transaction) ranged from 0.2% to 14.3%, the price to book value multiple ranged from 1.4 to 1.7 times, the LTM price to earnings multiple ranged from 12.0 to 14.6 times and the LTM operating cash flow multiple ranged from 4.6 to 6.8 times. Based on this analysis, Morgan Stanley calculated per share values for Atlantic ranging from \$12.70 to \$26.09.

Class A Common Stock. Morgan Stanley utilized several approaches to estimate the value of the Class A Common Stock which would be received in the Mergers for each outstanding share of Atlantic Common Stock. Based on an assumed price to 1997 earnings multiple of 7.0 to 8.5 times, and an estimated 1997 earnings level for the Class A Common Stock based on the securities research analyst consensus estimate for Atlantic (adjusted to reflect the \$40 million fixed charge), Morgan Stanley calculated the estimated value of the Class A Common Stock per share of Atlantic Common Stock ranging from \$1.98 to \$2.40; based on a required dividend yield of 10.0% to 12.5%, an estimated 1997 earnings level as derived above and an assumption of a 90% dividend payout (without giving effect to the initial dividend level on the Class A Common Stock which may exceed this payout ratio), Morgan Stanley calculated the estimated value of the Class A Common Stock per share of Atlantic Common Stock ranging from \$2.03 to \$2.54; and based on a discounted dividend valuation model, an earnings level based on the securities research analyst consensus as derived above and a 90% payout ratio (without giving effect to the initial dividend level on the Class A Common Stock which may exceed this payout ratio), Morgan Stanley calculated the estimated value of the Class A Common Stock per share of Atlantic Common Stock ranging from \$2.22 to \$2.61. The above analyses indicate a value range for the Class A Common Stock per share of Atlantic Common Stock of approximately \$2.00 to \$2.50, and based on the exchange ratio of .75 share of DS Common Stock per share of Atlantic Common Stock and the price of Delmarva Common Stock as of August 7, 1996, Morgan Stanley calculated the total consideration per share of Atlantic Common Stock ranging from \$17.56 to \$18.06, which represented a premium of 0.3% to 3.2% over the Atlantic Common Stock closing price of \$17.50 on August 7, 1996.

Pro Forma Analysis of the Mergers. Morgan Stanley analyzed the pro forma impact of the Mergers on Atlantic's EPS and dividends per share for the fiscal years ended 1997 through 2000. The analysis was performed utilizing stand-alone earnings estimated for the fiscal years ended 1997 through 2000 for Atlantic and Delmarva based on certain financial projections prepared by the respective managements of each company taking into account the cost savings expected to be derived from the Mergers as estimated by Atlantic and Delmarva. Based on such analysis, the Mergers would be accretive to Atlantic's stockholders for the period 1997 through 2000 on an EPS basis in the range of 11.8% in 1997 to 1.2% in 2000 and on a dividends per share basis by approximately 1.0%.

*Pro Forma Ownership.* Morgan Stanley compared the aggregate percentage share ownership of the combined company assuming the Mergers were consummated on August 9, 1996 at the proposed Exchange Ratio. Based on this analysis, the holders of Atlantic Common Stock would own 40.0% of the

combined company and the holders of Delmarva Common Stock would own 60.0% of the combined company.

The preparation of a fairness opinion is a complex process and is not necessarily susceptible to a partial analysis or summary description. Morgan Stanley believes that its analyses must be considered as a whole and that selecting portions of its analyses, without considering all analyses, would create an incomplete view of the process underlying its opinion. In addition, Morgan Stanley may have given various analyses more or less weight than other analyses, and may have deemed various assumptions more or less probable than other assumptions, so that the ranges of valuations resulting for any particular analysis described above should not be taken to be Morgan Stanley's view of the actual value of Atlantic and Delmarva.

In performing its analyses, Morgan Stanley made numerous assumptions with respect to industry performance, general business and economic conditions and other matters, many of which are beyond the control of Atlantic and Delmarva. The analyses performed by Morgan Stanley are not necessarily indicative of actual value, which may be significantly more or less favorable than suggested by such analyses. Such analyses were prepared solely as part of Morgan Stanley's analysis of the fairness of the Atlantic Conversion Ratio, taking into account the Delmarva Conversion Ratio, from a financial point of view to the holders of Atlantic Common Stock and were provided to the Atlantic Board in connection with the delivery of Morgan Stanley's written opinion dated the date hereof confirming its oral opinion of August 9, 1996. The analyses do not purport to be appraisals or to reflect the prices at which Atlantic and Delmarva might actually be sold. Because such estimates are inherently subject to uncertainty, none of Atlantic, Morgan Stanley or any other person assumes responsibility for their accuracy. In addition, as described above, Morgan Stanley's opinion and presentation to the Atlantic Board was one of many factors taken into consideration by the Atlantic Board in making its determination to approve the Mergers. Consequently, the Morgan Stanley analyses described above should not be viewed as determinative of the opinion of the Atlantic Board or the view of the management of Delmarva with respect to the value of Delmarva and Atlantic or of whether the Atlantic Board or the management of Delmarva would have been willing to agree to a different exchange ratio.

As part of its investment banking business, Morgan Stanley is regularly engaged in the valuation of businesses and securities in connection with mergers and acquisitions, negotiated underwritings, competitive biddings, secondary distributions of listed and unlisted securities, private placements and valuation for estate, corporate and other purposes. In the ordinary course of its business, Morgan Stanley and its affiliates may actively trade the debt and equity securities of Atlantic and Delmarva for their own account and for the accounts of customers and, accordingly, may at any time hold a long or short position in such securities. In the past, Morgan Stanley has provided financial advisory and financing services to Atlantic and Delmarva, for which services Morgan Stanley has received customary fees.

Morgan Stanley has been retained by Atlantic to act as financial advisor to Atlantic with respect to the Mergers. Pursuant to a letter agreement dated August 8, 1996 between Atlantic and Morgan Stanley, Morgan Stanley is entitled to (i) an advisory fee for its time and efforts expended in connection with the engagement, which is estimated to be between \$150,000 and \$250,000 and which is payable in the event the transaction is not consummated, (ii) an announcement fee of \$1,000,000, which has been paid, and (iii) a Merger fee equal to approximately \$4,230,000, which is payable upon consummation of the transaction. Any amounts paid or payable to Morgan Stanley as advisory or announcement fees will be credited against the transaction fee. Atlantic has also agreed to reimburse Morgan Stanley for its expenses of its counsel, and to indemnify Morgan Stanley and its affiliates against certain liabilities and expenses, including liabilities under federal securities laws.

#### Conflicts of Interest of Certain: Persons in the Mergers

In considering the recommendations of the Delmarva Board and the Atlantic Board with respect to the Mergers, stockholders should be aware that certain members of the Delmarva and Atlantic management, the Delmarva Board and the Atlantic Board have certain interests in the Mergers in addition to the interests of stockholders of Delmarva and Atlantic generally. The Delmarva Board and the Atlantic Board were aware of these interests and considered them, among other matters, in approving the Merger Ágreement and the transactions contemplated thereby.

*Employee Plans, Severance Arrangements and Agreements.* Under certain benefit plans, severance arrangements and other employee agreements maintained, or entered into, by Delmarva and Atlantic, certain benefits may become vested, and certain payments may become payable, in connection with the Mergers. These benefit plans, severance arrangements and agreements and the nature of such payments thereunder are described in greater detail below, under "—Employee Plans, Severance Arrangements and Agreements."

Board of Directors. As provided in the Merger Agreement, at the Effective Time, the Company Board will consist of 18 directors, ten of whom will be nominated by the Delmarva Board and eight of whom will be nominated by the Atlantic Board. Each member of the Delmarva Board and Atlantic Board serving in such capacity immediately prior to the consummation of the Mergers, will be given an opportunity to serve on the Company Board. See "The Company Following the Mergers—Management of the Company." The directors nominated by the Delmarva Board and the Atlantic Board will be divided between the classes of directors of the Company so that each class, to the extent possible, has the same proportion of directors nominated by each of the Delmarva Board and the Atlantic Board. At the consummation of the Mergers, the Audit Committee of the Company Board will consist of an equal number of directors nominated by the Delmarva Board and the Atlantic Board.

Indemnification. From and after the consummation of the Mergers, the Company will, to the fullest extent not prohibited by applicable law, indemnify, defend and hold harmless the present and former officers and directors of Delmarva and Atlantic against all losses, expenses (including reasonable attorneys' fees), claims, damages or liabilities or, subject to certain restrictions, amounts paid in settlement arising out of actions or omissions occurring at or before or after the consummation of the Mergers, that are in whole or in part based on, or arising out of, the fact that such person is or was a director, officer or management employee of such party or arising out of or pertaining to the transactions contemplated by the Merger Agreement. See "The Merger Agreement—Indemnification."

and the second second

# **Employee Plans, Severance Arrangements and Agreements**

Delmarva Severance Agreements. Delmarva has severance agreements (the "Delmarva Severance Agreements") with 48 of its officers and senior managers providing benefits upon a Change in Control (as defined in the Delmarva Severance Agreements). The Delmarva Severance Agreements provide a payment to the employee if the employee's employment is terminated by the employee for Good Reason (as defined in the Delmarva Severance Agreements), or by Delmarva for any reason other than retirement, death, disability, or cause at any time during the period that begins on the date of the Change in Control and ends 24 months thereafter. The payment will equal 2.99 times the employee's Base Amount (as that term is defined in Code Section 280G) less certain deductions. These deductions include the amount of any other payment (or the value of any other benefit) received or to be received by the employee in connection with a Change in Control if the payment or benefit is considered a parachute payment (as defined in Code Section 280G(b)(2)) which together with other parachute payments would result in disallowance of deductions by reason of Code Section 280G. The employee will receive group life, disability, medical and dental insurance benefits substantially similar to those received prior to employment termination for the 24 months following employment termination, reduced by comparable benefits the employee actually receives from other sources during that period. In addition, the employee will

receive a lump sum payment equal to the actuarial value of the excess of (i) the benefits payable under Delmarva's Retirement Plan and, for the most senior officers who participate in the Supplemental Executive Retirement Plan, the Supplemental Executive Retirement Plan (the "Retirement Plans") assuming 24 additional months service and compensation equal to the highest annual rate during the 12 month period preceding employment termination over (ii) the actual benefit payable under the Retirement Plans. "Change in Control" is defined in the Delmarva Severance Agreements as, among other things, stockholder approval of a merger of Delmarva other than a merger where Delmarva's voting securities outstanding immediately prior to the merger continue to represent at least 80% of the combined voting power of the surviving entity outstanding immediately after the merger. Stockholder approval of the Mergers will constitute a Change in Control under the Delmarva Severance Agreements. "Good Reason" is defined under the Delmarva Severance Agreements as, among other things, (i) assignment to the employee of duties inconsistent with the employee's status or position with Delmarva; (ii) a reduction by Delmarva in the employee's base salary (except for across-the-board salary reductions similarly affecting all of Delmarva employees and employees of any person in control of Delmarva); (iii) relocation of Delmarva's principal offices to a location more than 50 miles from the location prior to the Change in Control, or the closing thereof, or relocation of the employee's office location following the Change in Control; (iv) failure to pay current compensation; (v) failure to provide compensation plans and benefits otherwise provided to similarly situated employees; and (vi) the failure of Delmarva to obtain the agreement by a successor company to expressly assume the severance agreement. If the employment of all executives with Delmarva Severance Agreements had been terminated as of the date of this Joint Proxy Statement/Prospectus, the aggregate cost to Delmarva would not exceed approximately \$16.6 million.

Delmarva Power & Light Company Management Incentive Compensation Plan. The Delmarva Power & Light Company Management Incentive Compensation Plan (the "Compensation Plan") is a short-term annual performance award program provided to a certain level of management employees. The Compensation Plan provides payments based upon Delmarva and individual employee performance. Award amounts for a year are paid in a lump sum after the end of the year in which they were earned. Except for termination of employment due to retirement, death or disability, an employee must be employed on the last day of the year to be entitled to a payment. In the event of certain mergers or a change in the ownership of at least 25% of Delmarva's voting securities, all amounts that have been earned but not paid will be immediately paid. The consummation of the Mergers would be a transaction contemplated by that provision in the Compensation Plan.

Delmarva Power & Light Company Long-Term Incentive Plan. The Delmarva LTIP provides long-term compensation in the form of stock options (both nonqualified stock options and incentive stock options within the meaning of Code Section 422) restricted stock, stock appreciation rights, performance awards or such other forms as the Delmarva LTIP committee (a committee comprised of outside directors) may in its discretion deem appropriate. Persons eligible to participate in the Delmarva LTIP are officers and senior managers of Delmarva and its subsidiaries who, in the opinion of the Delmarva LTIP committee are in the position to significantly affect the growth and financial success of Delmarva. Delmarva directors who are not otherwise officers or employees of Delmarva are not eligible to participate in the Delmarva LTIP. Upon a Change in Control (as defined in the Delmarva LTIP) all options and stock appreciation rights that are outstanding at the date of the Change in Control become fully exercisable and vested; all incomplete performance periods for performance awards that are outstanding on the date of the Change in Control shall be deemed to have been completed, the maximum level of performance set forth shall be deemed to have been obtained and the pro-rata portion (based on the number of full and partial months that have lapsed with respect to each such performance period) of each outstanding award will become payable immediately in cash to each participant with the remaining of each such outstanding award being cancelled for no value; and any conditions to the vesting of restricted stock (whether performance objectives or periods of employment) are deemed to have been satisfied and all restrictions lapse and these shares become fully vested and nonforfeitable. Currently, Delmarva has outstanding two forms of long-term compensation: stock options, all of which are vested, and restricted stock, which would vest without restriction at the consummation of the Mergers. For purposes of the Delmarva LTIP, a Change in Control includes, among other things, consummation of a merger of Delmarva other than a merger where Delmarva's voting securities outstanding immediately prior to the merger continue to represent at least 80% of the combined voting power of the voting securities of the surviving entity outstanding immediately after the merger. Consummation of the Mergers will constitute a Change in Control under the Delmarva LTIP.

Delmarva Management Life Insurance Plan. Under the Delmarva Management Life Insurance Plan (the "Delmarva Life Insurance Plan") Delmarva offers a life insurance benefit to its officers and senior managers. The amount of life insurance is approximately three times annual salary. Benefits are available through both a term and permanent life insurance. Delmarva pays the premium for term coverage and Delmarva and the employee both pay portions of the premium for permanent coverage. Upon a change in the ownership of Delmarva including a merger where Delmarva is not the surviving entity, the surviving entity must assume the Delmarva Life Insurance Plan and prepay premiums for life insurance in force. The Mergers would result in such a change in ownership. Delmarva intends to amend the Delmarva Life Insurance Plan prior to the Mergers to eliminate any requirement for prepayment of any premiums and will retain the existing provision that the surviving entity shall assume Delmarva's rights and obligations under this plan.

Delmarva Power & Light Company Deferred Compensation Plan. The Delmarva Power & Light Company Deferred Compensation Plan (the "Delmarva Deferred Compensation Plan") provides officers and senior managers with the opportunity to defer a portion of their compensation and provides certain key executives the opportunity to receive matching contributions that are not otherwise payable under the terms of the Delmarva Power & Light Company Savings and Thrift Plan due to certain restrictions. In the event of a Change in Control (as defined in the Delmarva Deferred Compensation Plan) and the termination of an employee's employment, the timing of the payment of amounts to such employee under the Delmarva Deferred Compensation Plan may be accelerated at the discretion of the Delmarva Deferred Compensation Plan committee comprised of outside directors. For purposes of the Delmarva Deferred Compensation Plan a "Change of Control" would include a merger that would result in Delmarva's voting securities outstanding immediately prior to such merger being less than 75% of the combined voting power of the voting securities of the surviving entity immediately following such merger. The consummation of the Mergers will constitute a Change in Control for purposes of the Delmarva Deferred Compensation Plan.

Delmarva Supplemental Executive Retirement Plan. The Delmarva Power & Light Company Supplemental Executive Retirement Plan (the "SERP") provides supplemental retirement income to executives who are vice president or above and other managers or executives selected by Delmarva's Chief Executive Officer and approved by the Compensation Committee of the Delmarva Board. The supplemental benefits are benefits that would be provided under the Delmarva Power & Light Company Retirement Plan but for certain statutory limits on qualified plan benefits and benefits for certain employees recruited for or into key executive positions. Upon a merger of Delmarva in which Delmarva is not the surviving corporation or in which the holders of Delmarva stock receive securities in another corporation, Delmarva will prepay all premiums to any life insurance or annuity contracts intended to fund obligations under the SERP. Currently, only benefits for two former executives are provided through arrangements of this type. In addition, SERP participants will become immediately vested. The consummation of the Mergers would be a transaction contemplated by that provision in the SERP.

Atlantic Severance Agreements. Atlantic has previously entered into employment agreements (the "Agreements") with 13 executive officers (the "Executives"). For purposes of the Agreements, the Mergers will constitute a "Change in Control." Therefore, in the event an Executive's employment is terminated by Atlantic without "Cause" (as defined in the Agreements) or by the Executive for "Good Reason" (as defined in the Agreements) following the Mergers, the Executive will receive (a) the executive officer's full

base salary through the date of termination, (b) a cash amount from Atlantic equal to either three or two times the sum of (x) the Executive's annual base salary and (y) the higher of the bonus paid to the Executive for the most recent fiscal year or the target bonus for the current fiscal year (the "Minimum Bonus Amount"), (c) the prorated portion of the Executive's unpaid Minimum Bonus Amount, (d) any other amounts otherwise payable in respect of Atlantic's otherwise applicable long-term incentive compensation and equity plans and programs, and (e) all vested amounts or benefits owing to the Executive under Atlantic's otherwise applicable employee benefit plans and programs. In addition, the Executive will be entitled to continue to participate in Atlantic's employee and Executive pension, welfare and fringe benefit plans excluding supplemental retirement benefits. For purposes of calculating the Executive's retirement benefit, additional years will be added to both the Executive's age and service with Atlantic. The Agreements further provide that if the payments described above constitute "excess parachute payments" under applicable provisions of the Internal Revenue Code and related regulations, Atlantic will pay the Executive an additional amount sufficient to place the Executive in the same after-tax financial position the Executive would have been in if the Executive had not incurred the excise tax imposed under Section 4999 of the Internal Revenue Code in respect of excess parachute payments. If the employment of all Executives with Agreements had been terminated as of the date of this Joint Proxy Statement/Prospectus, the aggregate cost to Atlantic would not exceed approximately \$22.4 million.

*Consulting Arrangement.* The agreement with J.L. Jacobs establishes an opportunity at the election of J.L. Jacobs to enter into a consulting arrangement with Atlantic for a term of two years with compensation at \$130,000 annually to commence on the date following his retirement, but not before December 1, 1996.

Atlantic Executive Severance Policy. The Atlantic Executive Severance Policy covers approximately 24 executives and provides for severance in the event of a termination by Atlantic without "Cause" (as defined in the Atlantic Executive Severance Policy) or by the Executive for "Good Reason" (as defined in the Atlantic Executive Severance Policy) following a "Change in Control" (*i.e.*, the Mergers) in an amount equal to 1.5 times an executive's base salary and continuation of benefits for a period of one year following termination. In the event any of the payments due to an executive in connection with a Change in Control constitute excess parachute payments, amounts due such executive under the policy shall be reduced so that no payment results in the imposition of an excise tax under Code Section 4999.

Atlantic Employee Stock Purchase Plan. Upon a Change of Control, the right of employees participating in the Employee Stock Purchase Plan to acquire one share of Atlantic Common Stock will be converted into the right to receive .75 shares of Company Common Stock and .125 shares of Class A Common Stock.

Atlantic Equity Incentive Plan, Atlantic Supplemental Executive Retirement Plan and Atlantic Supplemental Executive Retirement Plan II. At the Effective Time, all stock options granted under Atlantic's Equity Incentive Plan will, absent action to the contrary at the discretion of the Personnel & Benefits Committee of the Atlantic Board, if the option has any value, be cashed out pursuant to the terms of the Equity Incentive Plan. The option price of all outstanding options granted pursuant to the Equity Incentive Plan exceeded the closing price as reported on the NYSE on December 5, 1996. In addition, at the Effective Time the restrictions applicable to any awards of restricted stock or other stock-based awards lapse and such stock and awards will be deemed fully vested. Benefits previously accrued under Atlantic's Supplemental Executive Retirement Plan ("SERP") and Atlantic Supplemental Executive Retirement Plan II ("SERP II") become nonforfeitable after the consummation of the Mergers. Under the terms of Atlantic's SERP, upon the Effective Time a participating executive with less than five years of service will be deemed to have five years of service and as a result will become fully vested in such executive's accrued benefits. In addition, under the terms of Atlantic's SERP II, upon the Effective Time a participating executive with less than five years of service with Atlantic will be deemed to have five years of service and as a result will become vested in 50% of such executive's accrued benefit.



Atlantic Director Retirement Plan. With respect to its directors, Atlantic intends to amend its Director Retirement Plan to provide for lump sum payouts at the Effective Time, equal to an Atlantic director's annual benefit under such plan (not actuarially reduced), multiplied by the number of years remaining in such director's life expectancy.

Atlantic Directors Restricted Stock Plan. Under the terms of the Atlantic Directors Restricted Stock Plan, at the Effective Time all restrictions on all shares that have not yet vested will be eliminated and the shares will become fully vested. Certain directors are also participants in Atlantic's Equity Incentive Plan described above.

## The Company Benefit Plans

Following the consummation of the Mergers, the Company and its subsidiaries will honor all prior contracts, agreements, collective bargaining agreements and commitments with current or former employees or current or former directors of Delmarva or Atlantic and their respective subsidiaries, in accordance with the respective terms of such contracts, agreements and commitments, subject to the Company's right to enforce them in accordance with their terms (including any reserved right to amend, modify, suspend, revoke or terminate them). The Company will take all action necessary so that after the consummation of the Mergers, the Dividend Reinvestment and Common Share Purchase Plan and the Savings and Thrift Plan of Delmarva and the Directors' Restricted Stock Plan, the Employee Stock Purchase Plan and the Dividend Reinvestment Plan of Atlantic will be terminated, replaced or amended to provide for the issuance and sale of Company Common Stock in place of Delmarva Common Stock or Atlantic Common Stock, as the case may be, under such plans.

Subsequent to the execution of the Merger Agreement, Delmarva and Atlantic, as the stockholders of the Company, determined that it would be in the best interests of the Company for the Company to adopt the Company Plan described below to replace the Delmarva LTIP and Atlantic's Equity Incentive Plan, subject to stockholder approval of the Company Plan at the Delmarva Meeting and the Atlantic Meeting. The Company Plan will become effective as of the consummation of the Mergers. The Company Plan is a comprehensive cash and stock compensation plan designed to provide the Company with the ability to provide incentives supporting the growth, development and financial success of the Company. The Company Plan provides for the grant of annual incentive awards and long-term incentive awards such as restricted stock, stock options (including incentive stock options and nonqualified stock options), stock appreciation rights ("SARs"), performance units and dividend equivalents, and any other types of awards as the Company Plan Committee deems appropriate. The maximum number of shares of Company Common Stock available for issuance under the Company Plan is 5,000,000. The Company Plan Committee, which will be comprised solely of nonemployee directors, will administer the Company Plan and make awards thereunder, and will have broad authority to fix the terms and conditions of individual grants and agreements with participants. The Company Plan is being submitted to stockholders of Delmarva and Atlantic for approval, and is described in greater detail under "Approval of the Company Incentive Compensation Plan" elsewhere in this Joint Proxy Statement/Prospectus. A copy of the Company Plan is attached to this Joint Proxy Statement/Prospectus as Annex VII. Following implementation of the Company Plan, no further awards will be made under the Delmarva LTIP or Atlantic's Equity Incentive Plan.

Delmarva currently has outstanding two forms of awards under the Delmarva LTIP: stock options, all of which are exercisable, and performance-based restricted stock, as to which all conditions to vesting would lapse at the consummation of the Mergers and would be exchanged for shares of Company Common Stock at the Delmarva Conversion Ratio, without restriction. It is anticipated that any further awards under the Delmarva LTIP prior to the consummation of the Mergers will be in the form of performance-based restricted stock. Such awards will be granted with the condition that the consummation of the Mergers does not constitute a Change in Control which causes the restrictions to lapse (the "New Delmarva Awards"). After the consummation of the Mergers, the New Delmarva Awards either (a) will continue to be held in the Delmarva LTIP (but in the form of Company Common Stock rather than Delmarva Common Stock) until the performance period for such New Delmarva Awards expires and the New Delmarva Awards vest at 100% (or, if performance criteria are exceeded, at up to 150%, such additional shares to be issued in Company Common Stock, without restriction) or are forfeited in part or in whole and, in such case, the Delmarva LTIP will continue in effect until the performance periods for such New Delmarva Awards has expired, or (b) the Company will honor the outstanding New Delmarva Awards and the terms and conditions of such New Delmarva Awards will otherwise remain the same; except that, if the performance cycle with respect to such New Delmarva Awards extends beyond the time at which the Mergers are consummated, the performance criteria will be modified to take into consideration the performance of the Company, as well as the performance of Delmarva prior to such time and, in such case, the Delmarva LTIP will be terminated by Delmarva as soon as practicable after such time.

Atlantic currently has outstanding two forms of awards under its Equity Incentive Plan: stock options that first become exercisable beginning in 1997 and performance-based restricted stock, as to which all conditions to vesting would lapse at the consummation of the Mergers and absent any action by the Personnel & Benefits Committee of the Atlantic Board (the "Committee") would accelerate and be cashed out in accordance with terms of the Equity Incentive Plan. It is anticipated that any further awards under Atlantic's Equity Incentive Plan prior to the consummation of the Mergers will be in the form of stock options and performance-based restricted stock (the "New Atlantic Awards"). With the exception of the rights granted to the thirteen executive officers covered by employment agreements, the Committee has the broad discretion, not inconsistent with the terms of the Equity Incentive Plan, with respect to whether and to what extent employees may be granted New Atlantic Awards and to whether and what extent performance based restricted stock granted under the Equity Incentive Plan would be exchanged for shares of Company Common Stock and Class A Common Stock at the Atlantic Conversion Ratio, without restriction. In exercising its discretion, the Committee may (a) accelerate vesting of all or a part of the New Atlantic Awards or waive forfeiture restrictions regarding any New Atlantic Awards or (b) determine that any New Atlantic Awards either (i) will continue to be held in the Equity Incentive Plan (but in the form of Company Common Stock and Class A Common Stock rather than Atlantic Common Stock) until the performance period for such New Atlantic Awards expires and the New Atlantic Awards will yest at 100% or are forfeited in part or in whole, and, in such case, the Equity Incentive Plan will continue in effect until the performance periods for such New Atlantic Awards has expired, or (ii) the Company will honor the outstanding New Atlantic Awards and the terms and conditions of such New Atlantic Awards will otherwise remain the same; except that, if the performance cycle with respect to such New Atlantic Awards extends beyond the consummation of the Mergers, the performance criteria will be modified to take into consideration the performance of the Company, as well as the performance of Atlantic prior to the consummation of the Mergers and, in such case, the Equity Incentive Plan will be terminated by Atlantic as soon as practicable after the consummation of the Mergers.

## **Certain Federal Income Tax Consequences**

The Mergers are designed to qualify as a tax-free reorganization or exchange under the Code. A condition precedent to Atlantic's obligation to consummate the Atlantic Merger is the receipt by Atlantic of an opinion of Simpson Thacher & Bartlett substantially to the effect that the Atlantic Merger will be treated as a reorganization under Section 368(a) of the Code. A condition to Delmarva's obligation to consummate the Delmarva Merger is the receipt by Delmarva of an opinion by LeBoeuf, Lamb, Greene & MacRae, L.L.P. substantially to the effect that the Delmarva Merger, taken together with the Atlantic Merger, will be treated as an exchange under Section 351 of the Code and/or as a reorganization under Section 368(a) of the Code. Assuming the Mergers so qualify, then for federal income tax purposes (i) no gain or loss will be recognized by Delmarva, Atlantic or the Company as a result of the Mergers, (ii) holders of Delmarva Common Stock whose shares are converted into Company Common Stock and Class A Common Stock in the Mergers will recognize no gain or loss as a result of the conversion (except, with

respect to Atlantic stockholders, who will recognize gain or loss to the extent that they receive cash in lieu of fractional shares), (iii) the holding period and basis applicable to shares of Company Common Stock and Class A Common Stock received in the Mergers will be the same as the holding period and basis attributable to the Delmarva Common Stock or Atlantic Common Stock, as the case may be, that was converted into Company Common Stock and Class A Common Stock in the Mergers (reduced by any amount allocable to a fractional share interest in Company Common Stock and Class A Common Stock for which cash is received). A holder of shares of Atlantic Common Stock will allocate his or her basis in such stock among the Company Common Stock and the Class A Common Stock received based upon the relative fair market values of the Company Common Stock and the Class A Common Stock. A holder of shares of Atlantic Common Stock who receives cash in lieu of a fractional share interest in Company Common Stock and Class A Common Stock will recognize gain or loss measured by the difference between the amount of cash received and the amount of the stockholder's aggregate basis allocated to the fractional share interest. Any gain an Atlantic stockholder recognizes will be taxed either as a dividend or as a capital gain. The Internal Revenue Service has published a ruling holding that, in the case of a minority stockholder whose relative stock interest in the surviving corporation is minimal, who exercises no control over the surviving corporation's affairs, and whose relative ownership interest in the surviving corporation has been reduced by a minimal amount as a result of the receipt of cash in lieu of fractional shares, any gain or loss such stockholder recognizes will be a capital gain or loss. If an Atlantic stockholder has held his or her shares of Atlantic Common Stock as a capital asset, any capital gain or loss recognized will be long-term if the stockholder held his or her shares for more than one year at the time of the Mergers.

THE ABOVE-DESCRIBED CONDITIONS PRECEDENT TO THE CONSUMMATION OF THE ATLANTIC MERGER AND THE DELMARVA MERGER ARE WAIVABLE BY ATLANTIC AND DELMARVA, RESPECTIVELY. Delmarva and Atlantic each expect it will receive a tax opinion from its counsel prior to the consummation of the Mergers and that it will not waive the tax opinion condition applicable to it. In the event of such a waiver, however, Atlantic and Delmarva will furnish a supplement to this Joint Proxy Statement/Prospectus disclosing such waiver and all related material matters, including risks to investors, and will resolicit proxies in favor of the Mergers and the Company Plan following the furnishing of such supplement.

Although it is the opinion of Simpson Thacher & Bartlett that the Class A Common Stock will be treated as stock of the Company for federal income tax purposes, there are no federal income tax regulations, court decisions or published IRS rulings bearing directly on this issue. In addition, the IRS announced during 1987 that it was studying the federal income tax consequences of stock that has certain voting and liquidation rights in an issuing corporation but whose dividend rights are determined by reference to the earnings of a segregated portion of the issuing corporation's assets, including assets held by its subsidiary, and that it would not issue any advance rulings regarding such stock. During 1995, the IRS withdrew such stock from its list of matters under consideration and reiterated that it would not issue advance rulings regarding such stock. In the absence of such a ruling, there is a risk that the IRS could assert that the Class A Common Stock were determined not to be stock of the Company, the receipt of the Class A Common Stock in the Atlantic Merger would be taxable to the Atlantic stockholders.

The discussion above represents a summary of the opinions of counsel that Delmarva and Atlantic expect to receive prior to (and as a waivable condition to) the consummation of the Mergers. Neither Delmarva nor Atlantic has received an opinion as of the date of this Joint Proxy Statement/Prospectus. At such time as Delmarva and Atlantic are in receipt of such opinions, they will be filed as exhibits, by post-effective amendment, to the Registration Statement. THE DISCUSSION ABOVE IS INCLUDED FOR GENERAL INFORMATION ONLY. IT DOES NOT ADDRESS THE STATE, LOCAL OR FOREIGN TAX ASPECTS OF THE MERGERS. THE DISCUSSION IS BASED ON CURRENTLY EXISTING PROVISIONS OF THE CODE, EXISTING AND PROPOSED TREASURY REGULATIONS

THEREUNDER AND CURRENT ADMINISTRATIVE RULINGS AND COURT DECISIONS. ALL OF THE FOREGOING ARE SUBJECT TO CHANGE AND ANY SUCH CHANGE COULD AFFECT THE CONTINUING VALIDITY OF THE DISCUSSION. EACH DELMARVA AND ATLANTIC STOCKHOLDER SHOULD CONSULT HIS OR HER OWN TAX ADVISOR AS TO THE SPECIFIC TAX CONSEQUENCES OF THE MERGERS TO HIM OR HER, INCLUDING THE APPLICATION AND EFFECT OF FEDERAL, STATE, LOCAL AND FOREIGN TAX LAWS.

#### **Accounting Treatment**

The Mergers will be accounted for under the purchase method of accounting for financial reporting purposes.

## Stock Exchange Listing of Company Capital Stock

The Company will apply for the listing of Company Common Stock on the NYSE. Approval of the listing on the NYSE of the shares of Common Stock issuable in the Mergers, upon official notice of issuance, is a condition precedent to the consummation of the Mergers. The Company may also apply for listings on other exchanges. If Delmarva and Atlantic continue to meet the requirements of the NYSE and the Philadelphia Exchange and, in the case of Atlantic the Pacific Exchange, Delmarva Common Stock and Atlantic Common Stock, respectively, will continue to be listed on those exchanges until the Mergers have been consummated.

#### **Federal Securities Law Consequences**

All shares of Common Stock received by Delmarva and Atlantic stockholders in the Mergers will be freely transferable, except that shares of the Common Stock received by persons who are deemed to be "affiliates" (as defined under the Securities Act) of Delmarva or Atlantic prior to or upon consummation of the Mergers may be resold by them only in transactions permitted by the resale provisions of Rule 145 promulgated under the Securities Act (or Rule 144 or after registration in the case of such persons who become affiliates of the Company upon consummation of the Mergers) or as otherwise permitted under the Securities Act. Persons who may be deemed to be affiliates of Delmarva, Atlantic or the Company generally include individuals or entities that control, are controlled by, or are under common control with, such party and may include certain officers and directors of such party as well as principal stockholders of such party. The Merger Agreement requires each of Delmarva and Atlantic to use its best efforts to cause each of its affiliates to execute a written agreement to the effect that the affiliate will not offer or sell or otherwise dispose of any shares of Company Common Stock issued to the affiliate in or pursuant to the Mergers in violation of the Securities Act or the rules and regulations promulgated by the SEC thereunder.

The Company will comply with all disclosure requirements under the Exchange Act with respect to each of the Company Common Stock and the Class A Common Stock as a separate class. In addition, holders of Company Common Stock and Class A Common Stock will have to comply with Schedules 13D and 13G under the Exchange Act separately for each class.

## Appraisal Rights

Delaware law does not provide appraisal rights to holders of Delmarva Common Stock or Delmarva Preferred Stock because all of such stock is listed on a national securities exchange. Virginia law does not provide dissenters' or other appraisal rights to holders of any series of Delmarva Preferred Stock because the holders of such stock are not entitled to vote on the Delmarva Merger. Virginia law does not provide dissenters' or other appraisal rights to the holders of Delmarva Common Stock because all of such stock is listed on a national securities exchange.

New Jersey law does not provide dissenters' or other appraisal rights to holders of Atlantic Common Stock because all of such stock is listed on a national securities exchange. Atlantic has no shares of preferred stock outstanding. The rights and preferences of the ACE Preferred Stock will not be affected by the Mergers, such stock will remain outstanding following the Mergers and the holders of such stock will have no appraisal rights.

## **REGULATORY MATTERS**

Set forth below is a summary, based on advice of counsel for Delmarva and Atlantic, of the regulatory requirements affecting the Mergers. Failure to obtain any necessary regulatory approval or any adverse conditions that are imposed with respect to any necessary regulatory approval, including the failure to obtain favorable ratemaking treatment, may affect the consummation of the Mergers.

## **Antitrust Considerations**

The HSR Act and the rules and regulations thereunder provide that certain transactions (including the Mergers) may not be consummated until certain information has been submitted to the Antitrust Division of the Department of Justice (the "Antitrust Division") and the Federal Trade Commission (the "FTC") and the specified HSR Act waiting period requirements have been satisfied. Delmarva and Atlantic will provide their respective Premerger Notifications pursuant to the HSR Act. The expiration or earlier termination of the HSR Act waiting period would not preclude the Antitrust Division or the FTC from challenging the Mergers on antitrust grounds. Neither Delmarva nor Atlantic believes that the Mergers will violate federal antitrust laws. If the Mergers are not consummated within 12 months after the expiration or earlier termination of the initial HSR Act waiting period, Delmarva and Atlantic must submit new information to the Antitrust Division and the FTC, and a new HSR Act waiting period must expire or be earlier terminated before the Mergers may be consummated.

# Federal Power Act

Section 203 of the Federal Power Act provides that no public utility shall sell or otherwise dispose of its jurisdictional facilities or directly or indirectly merge or consolidate its facilities with those of any other person or acquire any security of any other public utility without first having obtained authorization from the FERC. The approval of the FERC is required to consummate the Mergers. Under Section 203 of the Federal Power Act, the FERC will approve a merger if it finds the merger to be "consistent with the public interest." Delmarva and Atlantic filed a joint application before the FERC seeking approval of the Mergers on November 27, 1996.

# **Public Utility Holding Company Act of 1935**

The Company is required to obtain SEC approval under Section 9(a)(2) of the 1935 Act in connection with the Mergers. An application for approval of the Mergers will be filed by the Company.

Under the applicable standards of the 1935 Act, the SEC is directed to approve the proposed Mergers unless it finds that (i) the Mergers would tend towards detrimental interlocking relations or a detrimental concentration of control, (ii) the consideration to be paid in connection with the Mergers is not reasonable, (iii) the Mergers would unduly complicate the capital structure of the applicant's holding company system or would be detrimental to the proper functioning of the applicant's holding company system or (iv) the Mergers would violate applicable state law. To approve the proposed Mergers, the SEC must also find that the Mergers would tend towards the development of an integrated public utility system and would otherwise conform to the 1935 Act's integration and corporate simplification standards.

Upon consummation of the Mergers as currently structured, the Company must register as a holding company under the 1935 Act because it will not qualify for any exemptions available under such Act. The 1935 Act imposes restrictions on the operations of registered holding company systems. Among these are requirements that certain securities issuances as well as sales and acquisitions of utility assets or of securities of utility companies and acquisitions of interests in any other business be approved by the SEC. The 1935 Act also limits the ability of registered holding companies to engage in nonutility ventures and regulates holding company system service companies and the rendering of services by holding company affiliates to the system's utilities. The SEC has authority under the 1935 Act to preclude the payment of dividends by Delmarva and Atlantic to the Company or to preclude the payment of dividends by the Company under certain circumstances. Delmarva, Atlantic and the Company recognize that the divestiture

of the existing gas operations of Delmarva and certain nonutility operations of Delmarva and Atlantic is a possibility under the new registered holding company structure but the Company will request in its 1935 Act application that it be allowed to retain the gas utility operations of Delmarva and the nonutility operations of Delmarva and Atlantic or, in the alternative, that the question of divestiture be deferred. If divestiture is ultimately required, the SEC historically has allowed companies sufficient time to accomplish divestiture in a manner that protects stockholder value.

On June 20, 1995, the SEC issued a series of proposed regulations that are designed, among other things, to liberalize the restrictions on and regulation of the activities of registered holding companies. At the same time, the SEC's Division of Investment Management (the "Division") issued a report of legislative recommendations, including the Division's preferred recommendation that Congress repeal the 1935 Act, subject to the transfer of certain authority over audits, books and records and affiliate transactions of registered holding companies to state utility commissions and to the FERC. The report also recommended liberalizing the interpretation of the SEC's regulations to permit registered holding companies to own both electric and gas utility systems if state commissions agree. There is no assurance that the regulations proposed by the SEC will be implemented or that the suggestions in the Division's report will be adopted. In addition, on October 12, 1995, a bill was introduced in the U.S. Senate to repeal the 1935 Act and enact the Public Utility Holding Company Act of 1995 that contains significantly fewer regulatory restrictions than the 1935 Act and would be administered by the FERC. To the extent that some or all of the regulations and recommendations are implemented or the 1935 Act is repealed, restrictions on and regulation of the Company's activities may be liberalized, and the Company's ability to retain ownership of the gas utility properties and nonutility ventures currently operated by Delmarva and Atlantic would be enhanced.

#### **Nuclear Regulatory Commission**

Delmarva and Atlantic each own a 7.41% interest in the Salem Nuclear Generating Station, which consists of two nuclear units, and a 7.51% interest in the Peach Bottom Nuclear Generating Station, which consists of two nuclear units. In addition, Atlantic owns a 5% interest in the Hope Creek Nuclear Generating Station, which consists of one nuclear unit. Delmarva and Atlantic hold NRC licenses with respect to their ownership interests in these nuclear units. The Atomic Energy Act currently provides that licenses may not be transferred or in any manner disposed of, directly or indirectly, to any person unless the NRC finds that such transfer is in accordance with the Atomic Energy Act and consents to the transfer. Pursuant to the Atomic Energy Act, Delmarva and Atlantic will seek approval from the NRC to the full extent required to reflect that after the Mergers, Delmarva and Atlantic, although continuing to own the identical pre-Merger shares of the nuclear units, will become operating subsidiaries of the Company.

# **Delaware Public Service Commission**

Delmarva is incorporated in Delaware and subject to the jurisdiction of the Delaware Public Service Commission (the "DPSC"). Pursuant to Section 215 of the Public Utilities Act, Delmarva must obtain the approval of the DPSC in order to directly or indirectly merge or consolidate with any other person or company. Section 215 also provides that no other entity shall acquire control, either directly or indirectly, of any public utility doing business within Delaware without the prior approval of the DPSC. The DPSC is required to approve the proposed merger and acquisition when it finds that the same is to be made in accordance with law, for a proper purpose and is in the public interest. Delmarva and the Company will seek the approvals of the DPSC consistent with these requirements.

#### Virginia State Corporation Commission

Delmarva is incorporated in Virginia and subject to the jurisdiction of the VSCC. Pursuant to the Utility Transfers Act, no person, whether acting alone or in concert with others, shall, directly or indirectly, acquire control of a public utility without the prior approval of the VSCC and it is unlawful for any public

utility, directly or indirectly, to dispose of any utility assets situated within Virginia unless authorized by the VSCC. Upon application, the VSCC will approve a proposed transaction if satisfied that adequate service to the public at just and reasonable rates will not be impaired or jeopardized by granting the application.

Except to the extent preempted by the Securities Exchange Commission, the VSCC, pursuant to statutory provisions under which the VSCC regulates relations with affiliated interests, must approve certain contracts or arrangements for certain services, purchases, sales, leases or exchanges, loans and guarantees between a public service company and affiliates.

Delmarva and its affiliates will seek the approvals of the VSCC consistent with these requirements.

# New Jersey Board of Public Utilities

As the parent company of Atlantic City Electric Company, the transfer of the ownership or control, or the merger of, Atlantic is subject to the jurisdiction of the New Jersey Board of Public Utilities (the "NJBPU"). Pursuant to Title 48 of the New Jersey Statutes Annotated, no person shall acquire or seek to acquire control of a public utility directly or indirectly through the medium of an affiliated or parent corporation without requesting and receiving the written approval of the NJBPU. In addition, any transfer of stock to another public utility, or a transfer that vests another corporation with a majority interest in the stock of a public utility, must be authorized by the NJBPU.

The NJBPU, pursuant to statutory provisions under which the NJBPU regulates relations between public utilities and affiliated interests, must approve certain contracts or arrangements for certain services, purchases or loans between a public utility and affiliates.

Atlantic and its affiliates will seek the approvals of the NJBPU consistent with these requirements.

# Pennsylvania Public Utility Commission

Delmarva and Atlantic own fractional interests in the Keystone, Conemaugh and Peach Bottom electric generating stations and related transmission lines located in Pennsylvania. Pursuant to Pennsylvania statute, the transfer to any person or corporation of the stock, including a transfer by merger, of a public utility must be approved by the Pennsylvania Public Utility Commission ("PPUC"). The PPUC will approve such transfers upon a showing that the merger will affirmatively promote the service, accommodation, convenience or safety of the public in some substantial way. Delmarva and Atlantic will seek the approvals of the PPUC consistent with these requirements.

#### Maryland Public Service Commission

The Maryland Public Service Commission (the "MPSC") is granted general authority to supervise and regulate public utilities with operations in the State of Maryland. Delmarva has advised the MPSC of the transactions contemplated by the Merger Agreement and that it does not believe that the approval of the MPSC of the Mergers is required. However, the MPSC has ruled that it has jurisdiction over the Mergers to determine whether the Mergers will have an adverse effect on the conduct of Delmarva's Maryland franchises and any other matters that properly come before the MPSC at such hearing. Delmarva will seek to show that the Mergers will not have an adverse effect on the conduct of its Maryland franchises.

1

#### **Effects of Certain Regulatory Trends**

SFAS 71—Accounting for the Effects of Certain Types of Regulation in general recognizes that accounting for rate-regulated enterprises should reflect the relationship of costs and revenues. As a result, a regulated utility may defer recognition of cost (a regulatory asset) or recognize an obligation (a regulatory liability) if it is probable that, through the ratemaking process, there will be a corresponding increase or decrease in revenues. Accordingly, Delmarva and Atlantic have deferred costs that will be amortized over various periods. To the extent that collection of such costs or payment of such liabilities is no longer probable as a result of changes in regulation and/or the competitive position of Delmarva or

Atlantic, as the case may be, the associated regulatory asset or liability will be reversed with a charge or credit to income. If Delmarva or Atlantic were to discontinue the application of SFAS 71 the accounting impact would be an extraordinary noncash charge to operations that could be material to the financial position and results of operations of Delmarva or Atlantic, as the case may be.

Competition exists and is expected to increase for certain electric and gas energy markets historically served exclusively by regulated utilities. In recent years, changing laws and governmental regulations permitting competition from other utilities as well as nonregulated energy suppliers have prompted some customers to use self-generation or alternative sources to meet their electric and gas needs. The transition from strictly regulated to competitive resale and retail markets is changing the structure of the utility industry and the way in which it conducts business. As the electric utility industry transitions from a regulated to a competitive industry, utilities may not be able to recover certain costs. These costs, which are known as "stranded" costs, could result from the shift from cost of service based pricing to market based pricing and from customers choosing different energy suppliers than Delmarva or Atlantic. Potential types of stranded costs could be (i) above-market costs associated with generation facilities or long term power purchase agreements and (ii) regulatory assets, which are expenses deferred and expected to be recovered from customers in the future.

Several states are presently evaluating the most appropriate method to address the rapidly emerging competitive energy marketplace and allow consumers to participate in that market.

Delmarva has proposed to the Maryland and Delaware state regulatory commissions that a forum be established to address changes in the regulation of the electric utility industry, the objective being to develop cooperatively a blueprint to move toward increased customer choice. The forums will address issues such as open access to transmission and distribution lines by retail customers (retail wheeling), stranded investment, rate redesign and alternative forms of regulation, such as performance-based regulation.

To date, the NJBPU has made no formal pronouncement regarding deregulation or recovery of stranded commitments. The NJBPU is currently examining changes to the electric power industry. Specific issues are an evaluation of actions needed to establish a wholesale electric generation marketplace, an analysis to determine if divestiture of utility generating assets is necessary, a determination of the appropriateness and necessity of retail wheeling, given the achievement of a competitive wholesale market, and the definition and equitable treatment of stranded costs. Working groups representing various interested parties have been established to review these issues. In a preliminary status report issued in May 1996, certain recommendations were made, including a proposed 3-7 year transition to a competitive retail market. A formal recommendation by the NJBPU is expected to be submitted in early 1997 to the New Jersey governor and legislature, which is expected to include a specific timetable, interim pilot programs and recommended changes to legislation.

Delmarva and ACE have not filed for accelerated depreciation of any capital assets or special rate plans applicable to particular classes of customers. However, ACE has entered into contracts with selected customers which provide for special rates in conjunction with NJBPU approved guidelines. To date, the aggregate amount of such discounts is not material to ACE's financial condition.

ACE has significant long term contract commitments to purchase capacity and energy from nonutility sources at above-market costs. Recovery of amounts associated with these contracts is through ACE's Levelized Energy Clause, for rates are subject to approval by the NJBPU annually.

Delmarva and Atlantic are unable to predict the outcome of the various regulatory proceedings that are presently underway with respect to deregulation. However, if changes in the regulatory environment ultimately require a recognition of any amounts considered to be stranded costs, Delmarva or Atlantic, as the case may be, would be required to write down asset values, and such writedowns could be material. Given the uncertainty with respect to the timing of any regulatory changes, the resulting deregulated market prices for energy and/or capacity and the extent to which regulatory commissions allow for recovery

of any previously incurred costs, it is not possible to predict the level of stranded costs, if any, which would result.

The effect of competition on the Company's equity from reductions in profit margins or extraordinary charges against income would reduce the amount of common equity in the capital structure and could result in lowered credit ratings on existing debt securities and higher corresponding financing costs. To the extent that additional equity capital is required, issuances of common stock may be necessary. To the extent that additional equity capital is required, the effect would be to reduce reported earnings per share, the amount of which Delmarva and Atlantic cannot presently determine.

# **Other Regulatory Matters**

Each of the states in which Delmarva and Atlantic operate regulates the rates charged by Delmarva or Atlantic, as the case may be, to its utility customers in that state. In approving rates, each state may take into account the effect of the Mergers.

Delmarva, Atlantic and the Company have agreed in the Merger Agreement to use all commercially reasonable efforts to obtain regulatory approvals, but there can be no assurance as to when or if such approvals will be obtained or that such approvals will be obtained on terms or conditions that do not have a material adverse effect on the business, operations, properties, assets, condition, prospects or results of the Company following the Mergers.





65

#### THE MERGER AGREEMENT

The following is a brief summary of the material provisions of the Merger Agreement, which is attached as Annex I and is incorporated herein by reference. This summary is qualified in its entirety by reference to the Merger Agreement.

## The Mergers

The Merger Agreement provides that, following its approval by the stockholders of Delmarva and Atlantic and the satisfaction or waiver of the other conditions to the Mergers, (i) DS Sub will be merged with and into Delmarva in the Delmarva Merger and (ii) Atlantic will be merged with and into the Company in the Atlantic Merger.

If the Merger Agreement is approved by the stockholders of Delmarva and Atlantic and the other conditions to the Mergers are satisfied or waived, the Mergers will become effective after the filing (i) by Delmarva of the certificate of merger with the Secretary of State of the State of Delaware and articles of merger with the VSCC and (ii) by the Company of a certificate of merger with the Secretary of State of the State of Delaware and by Atlantic and the Company of a certificate of merger with the Secretary of State of the State of New Jersey, at the time specified by Delmarva and Atlantic in the respective certificates and articles of merger. The date and time specified will be the same with respect to both the Delmarva Merger and the Atlantic Merger.

## **Consummation of the Mergers**

At the Effective Time, pursuant to the Merger Agreement:

Each share of (i) Delmarva Common Stock issued and outstanding immediately prior to the Effective Time (other than shares of Delmarva Common Stock that are owned by Delmarva as treasury stock or by Atlantic or by any wholly owned subsidiary of Delmarva or Atlantic) will be converted into the right to receive one fully paid and nonassessable share of Company Common Stock (the "Delmarva Conversion, Ratio"), (ii) Atlantic Common Stock, issued and outstanding immediately prior to the Effective Time (other than shares of Atlantic Common Stock that are owned by Atlantic as treasury stock or by Delmarva or a wholly owned subsidiary of Atlantic or Delmarva), will be converted into the right to receive 0.75 fully paid and nonassessable shares of Company Common Stock and 0.125 fully paid and nonassessable shares of Class A Common Stock (the "Atlantic Conversion Ratio").

Each share of (i) Delmarva Common Stock that is owned by Delmarva as treasury stock or by Atlantic or any wholly owned subsidiary of Delmarva or Atlantic, (ii) Atlantic Common Stock that is owned by Atlantic as treasury stock or by Delmarva or by any wholly owned subsidiary of Atlantic or Delmarva and (iii) Company Common Stock that is owned by Delmarva, Atlantic or any wholly owned subsidiary of Delmarva or Atlantic in each case will be cancelled and cease to exist, and no consideration will be delivered in exchange therefor.

Each share of DS Sub Common Stock, par value \$.01 per share, issued and outstanding immediately prior to the Effective Time will be converted into and become one share of Common Stock, par value \$2.25 per share, of the surviving corporation in the Delmarva Merger.

Each share of Delmarva Preferred Stock outstanding at the time of the consummation of the Mergers will remain as outstanding preferred stock of Delmarva and the rights and designations of the Delmarva Preferred Stock will not be affected by the Mergers. The shares of Delmarva Preferred Stock currently listed on a national securities exchange will continue to be so listed following consummation of the Mergers.

On or before the Effective Time, the Company will deposit with the Exchange Agent certificates representing shares of Company Common Stock and Class A Common Stock required to effect the conversion of Delmarva Common Stock and Atlantic Common Stock into Company Common Stock and, in the case of Atlantic Common Stock, Class A Common Stock.

As soon as practicable after the Effective Time, the Exchange Agent will mail to each holder of record of Delmarva Common Stock or Atlantic Common Stock at the Effective Time, transmittal instructions advising such holder of the procedure for surrendering Delmarva Common Stock or Atlantic Common Stock certificates for Company Common Stock and Class A Common Stock certificates. Delivery will be effected, and risk of loss and title to the certificates will pass, only upon actual delivery to the Exchange Agent.

After the Effective Time, each certificate evidencing Delmarva Common Stock or Atlantic Common Stock until so delivered will, for all purposes, represent only the right to receive upon such delivery the certificate representing Company Common Stock or Class A Common Stock and if applicable, the right to receive any cash payment in lieu of a fractional share of Company Common Stock or Class A Common Stock or Class A Common Stock without interest. The holder of the unexchanged certificate will not be entitled to vote or to receive dividends or other distributions payable by the Company until the certificate is delivered, at which time the holder will be entitled to receive all dividends or other distributions accrued and unpaid from the Effective Time until the delivery.

No certificates or scrip representing fractional shares of Company Common Stock or Class A Common Stock will be issued upon the delivery for exchange of certificates and such fractional share interests will not entitle the owner thereof to vote or to any rights of a stockholder of the Company. As promptly as practicable following the Effective Time, the Exchange Agent will determine the excess of (x) the number of full shares of Company Common Stock and Class A Common Stock delivered to the Exchange Agent by the Company over (y) the aggregate number of full shares of Company Common Stock and Class A Common Stock to be distributed to holders of Delmarva Common Stock and Atlantic Common Stock (such excess being herein called the "Excess Shares"). As soon as practicable after the Effective Time, the Exchange Agent will sell the Excess Shares on the NYSE through one or more member firms of the NYSE, in round lots to the extent practicable. Until the proceeds of such sale or sales have been distributed to the holders of Atlantic Common Stock, the Exchange Agent will hold such proceeds in trust (the "Common Shares Trust"). The Company will pay all commissions, transfer taxes and other outof-pocket transaction costs, including the expenses of compensation of the Exchange Agent incurred in connection with such sale of the Excess Shares. The Exchange Agent will determine the portion of the proceeds of the Common Shares Trust to which each holder of Atlantic Common Stock will be entitled, if any, by multiplying the amount of the aggregate proceeds comprising the Common Shares Trust by a fraction of the numerator of which is the amount of the fractional share interest to which such holder of Atlantic Common Stock is entitled and the denominator of which is the aggregate amount of fractional share interests to which all holders of Atlantic Common Stock are entitled. As soon as practicable after the sale of the Excess Shares described above, the Exchange Agent will distribute such amounts to holders of Atlantic Common Stock who have delivered certificates for exchange.

Any certificates representing Company Common Stock and Class A Common Stock deposited with the Exchange Agent and not exchanged within one year after the Effective Time will be returned to the Company, which will thereafter act as Exchange Agent.

#### **Representations and Warranties**

The Merger Agreement contains customary representations and warranties by each of Delmarva and Atlantic relating to, among other things: (i) their respective organization and qualification, the organization and qualification of their respective subsidiaries and similar corporate matters; (ii) their respective capital structures; (iii) authorization, execution, delivery, performance and enforceability of the Merger Agreement and related matters; (iv) regulatory and statutory approvals; (v) compliance with applicable laws and agreements; (vi) reports and financial statements filed with governmental authorities and the accuracy of information contained therein; (vii) absence of material adverse changes and undisclosed liabilities; (viii) litigation; (ix) the accuracy of information supplied by Delmarva and Atlantic for use in the Registration Statement and Joint Proxy Statement/Prospectus filed by the Company in connection with the issuance of Company Common Stock and Class A Common Stock; (x) certain tax matters; (xi) employee

67

matters and ERISA; (xii) environmental matters; (xiii) the utility regulatory status of Delmarva and Atlantic and their respective subsidiaries; (xiv) the stockholder vote of Delmarva and Atlantic required to approve the Merger Agreement; (xv) fairness opinions of Merrill Lynch, Pierce, Fenner & Smith Incorporated and Morgan Stanley & Co. Incorporated; (xvi) insurance; (xvii) applicability of certain laws; (xviii) ownership by Delmarva and Atlantic, respectively, of the other's common stock; and (xix) operations of nuclear facilities.

# Certain Covenants

Pursuant to the Merger Agreement, Delmarva and Atlantic have each agreed that, during the period from the date of the Merger Agreement until the Effective Time or earlier termination of the Merger Agreement, except (i) as permitted under the Merger Agreement or (ii) as otherwise consented to in writing by the other parties, each will (and each of its subsidiaries will) among other things: (a) carry on its business in the ordinary course consistent with past practice and use all commercially reasonable efforts to preserve certain arrangements so that goodwill and ongoing businesses are not materially impaired at the Effective Time; (b) not (i) enter into any new line of business, except as permitted under the Merger Agreement, involving material investment of assets or resources or material exposure to liability or loss to such party and its subsidiaries taken as a whole, (ii) make investments in the permitted new businesses in excess of \$250 million or (iii) make any investment in a Listed Activity other than an investment by Atlantic in thermal services, in excess of \$7.5 million per year without consulting the other party; (c) not declare or pay any dividends on or make other distributions in respect of any of its capital stock, other than (i) to such party or its wholly owned subsidiaries, (ii) stated dividends on Delmarva Preferred Stock or on ACE Preferred Stock, (iii) regular quarterly dividends on the Delmarva Common Stock and the Atlantic Common Stock with usual record and payment dates at an annual rate not to exceed \$1.54 per share; (d) not split, combine or reclassify any capital stock or repurchase or otherwise acquire capital stock, other than (i) in the ordinary course of business including (A) in connection with employee plans, (B) in connection with repurchases, redemptions and other acquisitions required by terms of any series of Delmarva Preferred Stock or ACE Preferred Stock, (C) in connection with the refunding of Delmarva Preferred Stock or ACE Preferred Stock or indebtedness, at a lower cost of funds or through the issuance of long-term indebtedness and (D) open market purchases used to fund up to \$50 million in any fiscal year of the cost of any acquisition permitted under the Merger Agreement or (ii) intercompany acquisitions of capital stock; (e) not issue capital stock, warrants, rights, options or convertible or similar securities other than (i) the issuance of common stock or stock appreciation or similar rights pursuant to (A) the Dividend Reinvestment and Common Share Purchase Plan, the Savings and Thrift Plan and the Long-Term Incentive Plan of Delmarva or (B) the Employee Incentive Plan, the Directors' Restricted Stock Plan, the Employee Stock Purchase Plan and the Dividend Reinvestment Plan of Atlantic, in each case consistent in kind and amount with past practice and in the ordinary course of business under such plans substantially in accordance with their present terms, (ii) the issuance by a wholly owned subsidiary of shares of its capital stock to its parent and (iii) the issuance of common stock by Delmarva or Atlantic utilized to fund up to \$50 million per year of the cost of certain permitted acquisitions; (f) not amend its certificate or articles of incorporation or its bylaws, except as necessary to provide for the issuance of securities as permitted under the Merger Agreement or as required by law; (g) not engage in material acquisitions subject to certain exceptions; (h) not engage in material dispositions, subject to certain exceptions; (i) not incur or guarantee any indebtedness other than (i) short-term indebtedness incurred in the ordinary course of business consistent with past practice, (ii) long-term indebtedness in connection with the refinancing of existing indebtedness either at its stated maturity or at a lower cost of funds, (iii) long-term indebtedness in connection with the refunding of Delmarva Preferred Stock or ACE Preferred Stock at a lower cost of funds and (iv) additional indebtedness in any fiscal year not exceeding \$25 million more than provided in the Delmarva Budget or the Atlantic Budget (as defined in the Merger Agreement); (j) not make any capital expenditures other than capital expenditures (i) in connection with the construction of new facilities, (ii) to replace or repair facilities destroyed or damaged due to casualty or accident and (iii) additional capital expenditures in any year not exceeding 10% of the amount provided in the

Delmarva Budget or the Atlantic Budget as the case may be; (k) subject to certain exceptions, not enter into, adopt or amend (except as required by law), or increase the amount or accelerate the payment or vesting of any benefit or amount payable under, any employee benefit plan or other contract, agreement, commitment or arrangement to increase in any manner the compensation or fringe benefits, or otherwise to extend, expand or enhance the engagement, employment or any related rights of any director, officer or other employee of such party or any of its subsidiaries, except pursuant to binding legal commitment and except for normal (including incentive) increases in the ordinary course of business consistent with past practice that, in the aggregate, do not result in a material increase in benefits or compensation expense to such party and its subsidiaries taken as whole, provided that Atlantic will terminate its retirement plan for nonemployee directors and pay in cash all accrued benefits thereunder prior to the Effective Time, and each of Delmarva and Atlantic may, with the agreement of the other, modify the benefits payable under its severance arrangements with management employees in a manner that makes the aggregate level of severance benefits received by such management employees comparable to the aggregate level of severance benefits that are, on the date of the Merger Agreement, provided to comparable employees of the other; (1) except as contemplated by the Merger Agreement, not engage in any activity that would cause a change in its status under the 1935 Act or impair the ability of Atlantic to claim an exemption pursuant to Rule 2 of the 1935 Act; (m) not make any changes in their accounting methods other than required by law or in accordance with GAAP; (n) not take any action that would adversely affect the status of the Mergers as a tax free transaction under Section 368(a) of the Code and/or Section 351 of the Code; (o) not pay, discharge or satisfy any material claims, liabilities or obligations, other than the payment in the ordinary course of business consistent with past practice or in accordance with their terms, of liabilities reflected in, reserved against in, or contemplated by the most recently filed financial sheets filed with the SEC, as part of or pursuant to any settlement of any rate filings before the public utility commission of any state or the FERC pending on the date of the Merger Agreement, or as otherwise contemplated in the Merger Agreement; (p) cooperate with the other parties and notify the other parties of any significant change, including by providing copies of any filings with a governmental agency; (q) discuss with the other parties any proposed changes in its rates or charges, standards of service or accounting; (r) use all commercially reasonable efforts to obtain third-party consents to the Mergers; (s) not take any action that is likely to result in the material breach of any provision of the Merger Agreement or result in any of its representations and warranties becoming untrue; (t) not take any action that is likely to jeopardize the qualification of outstanding revenue bonds issued for the benefit of Delmarva or Atlantic as tax-exempt industrial revenue bonds; (u) create and maintain a special transition management task force headed by Howard E. Cosgrove and Michael J. Chesser; (v) maintain insurance with financially responsible insurance companies, in such amounts and against such risks and losses as are customary for companies in the utility industry; (w) maintain in effect all existing permits pursuant to which such party or its subsidiaries operate; and (x) not modify, amend, terminate, renew or fail to use reasonable business efforts to renew any material contract or agreement to which it or any of its subsidiaries is a party, or waive, release or assign any material rights or claims.

## No Solicitation of Transactions

The Merger Agreement provides that the parties thereto and their respective subsidiaries will not and will not authorize or permit any of their respective representatives to, directly or indirectly, solicit, initiate or encourage, or take any other action to facilitate knowingly any inquiries or the making of any proposal that constitutes or may reasonably be expected to lead to an Acquisition Proposal (as defined below) from any person, or engage in any discussion or negotiations relating thereto or accept any Acquisition Proposal. Notwithstanding anything to the contrary, Delmarva or Atlantic may (i) at any time prior to the Delmarva stockholders' approval or the Atlantic stockholders' approval, as the case may be, having been obtained, negotiate or discuss with a third party who seeks to initiate such discussions or negotiations and may furnish such third party with information concerning itself if and to the extent that (A) (x) such third party shall have first made an Acquisition Proposal that is financially superior to the Mergers and has demonstrated that financing is reasonably likely to be obtained and (y) the Board of Directors of such

party shall have concluded in good faith on the basis of a written opinion of outside counsel that such action is necessary to act in a manner consistent with its fiduciary duties and (B) prior to furnishing such information or entering into such discussions or negotiations, such party (x) provides prompt notice to the other party and (y) obtains an executed confidentiality agreement in customary form in terms no more favorable to the third party than the confidentiality agreement entered into between Delmarva and Atlantic, (ii) comply with Rule 14e-2 promulgated under the Exchange Act in connection with a tender offer and (iii) accept an Acquisition Proposal from a third party; provided that such party terminates the Merger Agreement pursuant to clause (f) under "The Merger Agreement-Termination." Each party will notify the other party orally and in writing of any such inquiries, offers or proposals (including without limitation the terms and conditions of any such proposal and the identity of the person making it) within 24 hours of the receipt thereof and will give the other party five days' advance notice of any agreement to be entered into with or any information to be supplied to any person making such inquiry, offer or proposal. As used in the Merger Agreement, "Acquisition Proposal" means a proposal or offer (other than by another party to the Merger Agreement) for a tender or exchange offer, merger, consolidation or other business combination involving a party or any material subsidiary of such party or any proposal to acquire a substantial equity interest in or a substantial portion of the assets of such party or any material subsidiary of such party.

#### **Company Board of Directors**

The Merger Agreement provides that the Delmarva Board will be entitled to nominate 10 members and the Atlantic Board will be entitled to nominate eight members to serve on the Company Board at the Effective Time and will take all necessary action to cause each director of Delmarva and Atlantic serving as such immediately prior to the Effective Time to have the opportunity to serve as a director of the Company. The directors nominated by the Delmarva Board and the Atlantic Board will be divided between the classes of directors of the Company so that each class, to the extent possible, has the same proportion of directors nominated by each of the Delmarva Board and the Atlantic Board. At the Effective Time, the Audit Committee of the Company Board will consist of an equal number of directors nominated by the Delmarva Board and the Atlantic Board.

## Indemnification

The Merger Agreement provides that from and after the Effective Time, the Company will, to the fullest extent not prohibited by applicable law, indemnify, defend and hold harmless the present and former officers, directors and management employees of Delmarva and Atlantic and their respective subsidiaries (each an "Indemnified Party" and, collectively, the "Indemnified Parties") against all losses, expenses (including reasonable attorneys' fees and expenses), claims, damages, costs, liabilities, judgments or, subject to the proviso of the next succeeding sentence, amounts paid in settlement arising out of actions or omissions occurring at or prior to or after the Effective Time that are in whole or in part based on, or arising out of, the fact that such person is or was a director, officer or management employee of Delmarva or Atlantic and arising out of or pertaining to the transactions contemplated by the Merger Agreement. In the event of any such loss, expense, claim, damage or liability (whether or not arising prior to the Effective Time), (i) the Company will pay the reasonable fees and expenses of counsel selected by the Indemnified Parties, which counsel shall be reasonably satisfactory to the Company, promptly after statements therefor are received and otherwise advance to such Indemnified Parties upon request reimbursement of documented expenses reasonably incurred, in either case to the extent not prohibited by applicable law, (ii) the Company will cooperate in the defense of any such matter and (iii) any determination required to be made with respect to whether an Indemnified Party's conduct complies with the standards under applicable law of the State of Delaware and the Company Charter or Company Bylaws will be made by independent counsel mutually acceptable to the Company and the Indemnified Party; provided that the Company will not be liable for any settlement effected without its written consent. The Indemnified Parties as a group may retain only one law firm (other than local counsel) with respect to each related matter except to the extent there is, in the sole opinion of counsel to an Indemnified Party under applicable standards of professional conduct, a conflict on any significant issue between positions of any two or more Indemnified Parties. If an Indemnified Party is required to bring any action to enforce any of its rights under the Merger Agreement and is successful, the Company will reimburse such Indemnified Party for all of its expenses in bringing and pursuing such action. In addition, each Indemnified Party is entitled to the advancement of expenses in connection with any such action.

The Merger Agreement provides that, for a period of six years after the Effective Time, the Company will cause to be maintained in effect the policies of directors' and officers' liability insurance maintained by Delmarva and Atlantic to the extent such liability insurance can be maintained annually at a cost to the Company not greater than 200% of the respective current annual premiums for their directors' and officers' liability insurance; provided that the Company may substitute therefor policies of at least the same coverage containing terms that are no less advantageous with respect to matters occurring prior to the Effective Time; provided that if such insurance cannot be so maintained or obtained at such cost, the Company will maintain or obtain as much of such insurance for each of Delmarva and Atlantic as can be so maintained or obtained at a cost equal to 200% of the respective current annual premiums of each of Delmarva and Atlantic for their directors' and officers' liability insurance for their directors' and agreements.

The Merger Agreement also provides that in the event that the Company or any of its successors or assigns (i) consolidates with or merges into any other person and will not be the continuing or surviving corporation or entity of such consolidation or merger or (ii) transfers all or substantially all of its properties and assets to any person, then and in either such case, proper provision will be made so that the successors and assigns of the Company will assume the obligations set forth above.

#### Conditions to the Mergers

The respective obligations of Delmarva and Atlantic to effect the Mergers are subject to the following conditions, among others: (i) the required approvals of the Delmarva stockholders and the Atlantic stockholders specified in the Merger Agreement; (ii) the absence of any temporary restraining order, preliminary or permanent injunction or other order or any applicable federal or state law or regulation preventing the consummation of the Mergers and the transactions contemplated by the Merger Agreement; (iii) the effectiveness of this Registration Statement and the absence of any stop order suspending its effectiveness; (iv) the approval for listing on the NYSE, upon official notice of issuance of the shares of Company Common Stock and Class A Common Stock issuable in the Mergers; (v) the receipt of all statutory approvals as Final Orders (as defined below) that Delmarva and Atlantic are required to obtain pursuant to the Merger Agreement, without any such Final Order imposing terms or conditions having or reasonably likely to have, a material adverse affect on business, operations, properties, assets, condition (financial or otherwise), prospects or results of operations of Delmarva or Atlantic, in each case as if it were organized as a separate division of the Company; (vi) the expiration or termination of the applicable waiting periods under the HSR Act; (vii) the agreements and covenants required to be performed under the Merger Agreement having been performed in all material respects; (viii) the representations and warranties set forth in the Merger Agreement being true in all material respects; (ix) the absence of any material adverse effect on the business, operations, properties, assets, conditions (financial or otherwise), prospects or results of operations of Delmarva or Atlantic and their subsidiaries and there being no fact or circumstance that is reasonably likely to have such an effect; (x) the receipt of material third party consents; (xi) the receipt by Delmarva and Atlantic of officers' certificates from the other stating that the conditions set forth in the Merger Agreement have been satisfied; (xii) the receipt by Delmarva and Atlantic from their respective special tax counsel opinions to the effect that, in the case of Atlantic, the Atlantic Merger will be treated as a reorganization described in Section 368(a) of the Code and in the case of Delmarva, the Delmarva Merger, taken together with the Atlantic Merger, will be treated as a nontaxable exchange described in Section 351 of the Code; and (xiii) the receipt by Delmarva and Atlantic of certificates of certain affiliates of Delmarva and Atlantic. As defined in the Merger Agreement, a "Final Order" means action by the relevant regulatory authority that has not been reversed, stayed, enjoined, set aside, annulled or suspended, with respect to which any waiting period prescribed by law before the

transactions contemplated by the Merger Agreement may be consummated has expired, and as to which all conditions to the consummation of such transactions prescribed by law, regulation or order have been satisfied, and as to which all opportunities for rehearing are exhausted (whether or not any appeal thereof is pending).

#### **Benefit Plans**

Following the Effective Time, the Company and its subsidiaries will honor, without modification, all prior contracts, agreements, collective bargaining agreements and commitments with current or former employees or current or former directors of Delmarva and Atlantic and their respective subsidiaries, subject to any reserved right to amend, modify, suspend, revoke or terminate.

Each of the Delmarva Benefit Plans and the Atlantic Benefit Plans, as defined in the Merger Agreement, in effect on the date of signing of the Merger Agreement will be maintained in effect with respect to the employees or former employees of Delmarva and Atlantic and any of their respective subsidiaries, who are covered by such plans immediately prior to the closing date until the Company determines otherwise; provided that nothing shall limit any reserved right to amend, modify, suspend, revoke or terminate any such plan. Each participant in any Delmarva Benefit Plan or Atlantic Benefit Plan will receive credit for purposes of eligibility to participate, vesting and eligibility to receive benefits under any benefit plan of the Company or any subsidiaries or affiliates for service credited for the corresponding purpose under any such benefit plan. The Company will take all action necessary so that after the Effective Time, the Dividend Reinvestment and Common Share Purchase Plan, the Savings and Thrift Plan and the Long-Term Incentive Plan of Delmarva and the Executive Incentive Plan, the Directors' Restricted Stock Plan, the Employee Stock Purchase Plan and the Dividend Reinvestment Plan of Atlantic will be terminated, replaced or amended to provide for the issue and sale of Company Common Stock in place of Delmarva Common Stock or Atlantic Common Stock, as the case may be, under such plans.

## Termination

The Merger Agreement may be terminated at any time prior to the closing, whether before or after approval by the stockholders of Delmarva or Atlantic: (a) by mutual written consent of the Delmarva Board and the Atlantic Board; (b) by Delmarva or Atlantic, by written notice to the other, if the Effective Time shall not have occurred on or before 18 months from the signing of the Merger Agreement; provided that such date will automatically be extended to 30 months from signing of the Merger Agreement, if on the date 18 months from signing of the Merger Agreement (i) the condition of obtaining the required statutory approvals described above in clause (v) under "Conditions to the Mergers" has not been satisfied or waived, (ii) the other conditions to the consummation of the transactions contemplated by the Merger Agreement are then capable of being satisfied and (iii) any required statutory approvals that have not yet been obtained are being pursued with diligence; and provided that the right to terminate the Merger Agreement under the provision described in this clause (b) will not be available to any party whose failure to fulfill any obligation under the Merger Agreement has been the cause of, or resulted in, the failure of the Effective Time to occur on or before the termination date; (c) by Delmarva or Atlantic, by written notice to the other party, if any required stockholder approval shall not have been obtained at a duly held meeting of stockholders or at any adjournment or adjournments thereof; (d) by Delmarva or Atlantic, if any state or federal law, order, rule or regulation is adopted or issued, which has the effect, as supported by the written opinion of outside counsel for such party, of prohibiting either or both of the Mergers, or if any court of competent jurisdiction in the United States or any State shall have issued an order, judgment or decree permanently restraining, enjoining or otherwise prohibiting either or both of the Mergers and such order, judgment or decree shall have become final and nonappealable; (e) by Delmarva or Atlantic, upon five days' prior notice to the other if, as a result of an acquisition proposal by a party other than Delmarva or Atlantic, as the case may be, or any of their respective affiliates, the Delmarva Board or the Atlantic Board, as the case may be, determines in good faith on the basis of a written opinion of outside counsel, that acceptance of the acquisition proposal is necessary for the Board of Directors to act consistent with its



fiduciary duties; provided that (i) the Delmarva Board or the Atlantic Board, as the case may be, shall have been advised by outside counsel that, notwithstanding a binding commitment to consummate an agreement of the nature of the Merger Agreement entered into in the proper exercise of its applicable fiduciary duties, and notwithstanding all concessions that may be offered in negotiations entered into pursuant to clause (ii), such fiduciary duties would also require the directors to reconsider such commitment as a result of such acquisition proposal and (ii) prior to any such termination, Delmarva or Atlantic, as the case may be, will and will cause its respective financial and legal advisors to negotiate with the other to make such adjustments in the terms and conditions of the Merger Agreement as would enable Delmarva or Atlantic, as the case may be, to proceed with the transactions contemplated therein; provided that Delmarva and Atlantic acknowledge and affirm that notwithstanding anything in the provisions described in this clause (e) to the contrary, Delmarva and Atlantic intend the Merger Agreement to be an exclusive agreement and, accordingly, nothing in the Merger Agreement is intended to constitute a solicitation of an acquisition proposal, it being acknowledged and agreed that any such proposal would interfere with the strategic advantages and benefits that Delmarva and Atlantic expect to derive from the Mergers and other transactions contemplated thereby; (f) by Delmarva or Atlantic (the "Terminating Party"), by written notice to the other (the "Breaching Party"), if (i) there shall have been any material breach of any representation or warranty or any material breach of any covenant or agreement of the Breaching Party made in the Merger Agreement that is not remedied within 20 days after receipt by the Breaching Party of notice in writing from the Terminating Party specifying the nature of such breach and requesting that it be remedied, or (ii) the Board of Directors of the Breaching Party (A) shall withdraw or modify in any manner adverse to the Terminating Party its approval of the Merger Agreement and the transactions contemplated thereby or its recommendation to its stockholders regarding the approval of the Merger Agreement, (B) shall fail to reaffirm such approval or recommendation upon the request of the Terminating Party, (C) shall approve or recommend the acquisition by a third party of the Breaching Party or of a material portion of its assets or any tender offer for the capital stock of the Breaching Party, or (D) shall resolve to take any of the actions specified in clauses (A), (B) or (C) above; or (g) by either party by written notice to the other party if (i) a third party acquires securities representing more than 50% of the voting securities of the other party or (ii) individuals who as of the date of the Merger Agreement constitute the Delmarva Board or the Atlantic Board cease for any reason to constitute a majority of the Delmarva Board or the Atlantic Board, as the case may be, then in office.

## **Termination Fees**

If the Merger Agreement is terminated pursuant to the provision described above in clause (e) under "Termination" ("clause (f)(i)"), pursuant to the provision described above in clause (f)(i) under "Termination" ("clause (f)(i)"), pursuant to the provision described above in clause (f)(ii) under "Termination" ("clause (f)(ii)", together with clause f(i), "clause f") or pursuant to the provision described in clause (g) under "Termination" ("clause (g)"), then the Breaching Party or the party whose Board of Directors has exercised its fiduciary out or changed its recommendation or the Party whose stock has been acquired or whose Board of Directors has changed, as the case may be, shall promptly (but not later than five business days after notice of the amount due is received) pay to the Terminating Party as liquidated damages an amount equal to the out-of-pocket expenses and fees incurred by the Terminating Party (including, without limitation, fees and expenses arising out of, in connection with or related to the Mergers or the transactions contemplated by the Merger Agreement) not to exceed \$10 million in the aggregate ("Out-of-Pocket Expenses"); provided that if the Merger Agreement is terminated as a result of willful breach or failure to perform or comply with any agreements and covenants therein, the nonbreaching party may pursue any remedies available at law or equity in addition to the Out-of-Pocket Expenses.

If the Merger Agreement (i) is terminated by Delmarva or Atlantic pursuant to the provision described above in clause (b) under "Termination," clause (f), clause (e), pursuant to clause (c) under "Termination" ("clause (c)"), the provision described above in clause (d) under "Termination," or as a result of a party's failure to take specified action with respect to obtaining approval of the Merger



Agreement from its stockholders, (ii) at the time of such termination (or prior to the meeting of such party's stockholders, in the case of a termination resulting from a failure to take action with respect to obtaining stockholder approval) there shall have been a third-party tender offer for shares of, or a third-party offer or proposal with respect to a acquisition proposal involving, Delmarva or Atlantic, as the case may be (the "Target Party") or its affiliates that, at the time of such termination or of the meeting of such party's stockholders, shall not have been (x) rejected by the Target Party and its Board of Directors and (y) withdrawn by the third party and (iii) within two and one-half years of any such termination, the Target Party or any of its affiliates becomes a subsidiary of such offeror or a subsidiary of an affiliate of such offeror or accepts a written offer or enters into a written agreement to consummate or consummates an acquisition proposal with such offeror or an affiliate thereof, then such Target Party, upon the signing of definitive agreement or, if no agreement is signed, at the closing (as a condition of closing) of the Target Party becoming a subsidiary or of such acquisition proposal, then the Target Party will pay to the other party \$30 million plus Out-of-Pocket Expenses. If the Merger Agreement is terminated pursuant to clause (g) then the nonterminating party will pay the terminating party \$30 million plus Out-of-Pocket Expenses.

In the event that termination fees are payable pursuant to the termination provisions contained in the Merger Agreement, the aggregate amount payable to either Delmarva or Atlantic and each of their respective affiliates will not exceed \$40 million (including Out-of-Pocket Expenses).

#### Expenses

In the Merger Agreement, the parties thereto agree that the agreements described under "Termination Fees" are an integral part of the transactions contemplated by the Merger Agreement and constitute liquidated damages and not a penalty. If one party fails to pay promptly to the other any fees due thereunder, the defaulting party will pay the costs and expenses (including legal fees and expenses) in connection with any action, including the filing of any lawsuit or other legal action, taken to collect payment, together with interest on the amount of any unpaid fee at the publicly announced prime rate of Citibank, N.A. from the date such fees are required to be paid.

Except as set forth above, all fees and expenses incurred in connection with the Merger Agreement and the transactions contemplated thereby will be paid by the party incurring such expenses, except that the expenses in connection with printing and filing of this Joint Proxy Statement/Prospectus will be shared equally by Delmarva and Atlantic.

#### **Amendment and Waiver**

The Merger Agreement may be amended by agreement of the respective Boards of Directors of the parties thereto, at any time before or after the approval thereof by the holders of Delmarva Common Stock and Atlantic Common Stock, respectively, and prior to the Effective Time, but after such approvals, no such amendment may be made that will (i) alter or change the amount or kind of shares, securities or cash to be received or exchanged for or on conversion of any class or series of capital stock of either Delmarva or Atlantic, (ii) alter or change any of the terms or conditions of the Merger Agreement if any of the alterations or changes, alone or in the aggregate, would adversely affect the rights of holders of the holders of any class or series of capital stock of Delmarva or Atlantic, or (iii) alter or change any term of the Company Charter or the Delmarva Charter, except for alterations or changes that could otherwise be adopted by the Company Board without the further approval of such stockholders. At any time prior to the Effective Time, the parties to the Merger Agreement may extend the time for the performance of any of the obligations or other acts of the other parties thereto, waive any inaccuracies in the representations and warranties contained therein or in any document delivered pursuant thereto and waive compliance with any agreements or conditions to a party's obligation to consummate the Mergers contained in the Merger Agreement.



# DESCRIPTION OF THE COMPANY'S CAPITAL STOCK

# THE FOLLOWING DESCRIPTION IS QUALIFIED BY REFERENCE TO ANNEX IV TO THIS JOINT PROXY STATEMENT/PROSPECTUS, WHICH CONTAINS THE FULL TEXT OF THE COMPANY CHARTER, AND ANNEX V TO THIS JOINT PROXY STATEMENT/PROSPECTUS, WHICH CONTAINS THE FULL TEXT OF THE COMPANY BYLAWS.

## General

The authorized capital stock of the Company, as of the time of the consummation of the Mergers, will consist of (i) 160 million shares of common stock, of which 150 million shares will be designated as Company Common Stock and 10 million shares will be designated as Class A Common Stock (the Company Common Stock and the Class A Common Stock are referred to collectively herein as "Common Stock"), and (ii) 20 million shares of preferred stock, par value \$.01 per share ("Preferred Stock").

# **Preferred Stock**

Although the Company Charter authorizes the issuance of Preferred Stock, the Company has no present intention to issue any Preferred Stock, and no Preferred Stock will be outstanding at the time of the consummation of the Mergers. Subject to any approval of the SEC that may be required under the 1935 Act, the Company Board is authorized to divide any Preferred Stock issued by the Company into series and, within the limitations set forth in the Company Charter or prescribed by law, to fix and determine the relative rights and preferences of the shares of any series so established. Such rights and preferences include the maximum number of shares in a series, preferences as to dividends and upon liquidation, dividend rate or rates, redemption prices and terms, sinking fund provisions, if any, conversion rights, voting rights, restrictions on the creation of indebtedness of such series and any other rights or preferences as to which the DGCL permits variations between different series of Preferred Stock.

## Common Stock

The authorized and unissued shares of Common Stock will be available for issuance by the Company from time to time, as determined by the Company Board, for any proper corporate purpose, which could include raising capital, acquiring other companies or making investments and providing compensation or benefits to employees. The issuance of such shares would not be subject to the approval of the stockholders of the Company unless deemed advisable by the Company Board or required by applicable law, regulation or stock exchange requirements. Although the Company Charter authorizes the issuance of Class A Common Stock in addition to the shares of Class A Common Stock to be issued pursuant to the Atlantic Merger, the Company has no present intention to issue any shares of Class A Common Stock following the consummation of the Mergers, and no shares of Class A Common Stock will be outstanding at the time of the consummation of the Mergers other than those to be issued to the holders of Atlantic Common Stock (including those to be sold by the Exchange Agent with the proceeds being distributed to such holders) pursuant to the Merger Agreement.

The Class A Common Stock, the Atlantic Utility Group and the Notional Fixed Charge. The Class A Common Stock is intended to reflect the performance of the Atlantic Utility Group above the Notional Fixed Charge, as defined below, since it is the intention of the Company to base the dividends on the Class A Common Stock on such factor as well as all other relevant considerations (subject to, and taking into consideration, the provisions relating to dividend policy set forth in the Merger Agreement). The Company Charter defines "Atlantic Utility Group" to include the interest of the Company at the Effective Time in ACE, and solely those lines of business in which ACE was engaged as of August 9, 1996, and the assets and liabilities attributable to those lines of business, and which as of August 9, 1996 were: (a) price regulated by the NJBPU, including without limitation off-tariff agreements; (b) directly related to the supply of

electricity (generation and purchase of electricity) or the delivery of electricity (transmission and distribution of electricity); and (c) a line of business for which ACE has a franchise; provided that the Deepwater Operating Company, a wholly owned subsidiary of ACE that operates certain generating facilities, is specifically included in the Atlantic Utility Group. The Atlantic Utility Group will also include certain assets and liabilities, including but not limited to those associated with the included businesses and liabilities under the \$40 million per annum Notional Fixed Charge. Specifically excluded without limitation from the Atlantic Utility Group are ACE's present unregulated businesses. However, holders of the Class A Common Stock will, like the holders of Company Common Stock, be common stockholders of the Atlantic Utility Group, including upon liquidation of the Company, other than as common stockholders of the Company and will be subject to risks associated with an investment in the Company and all of its businesses, assets and liabilities. See "—Certain Definitions."

The term "Company Net Income (Loss) Attributable to the Atlantic Utility Group," as defined in the Company Charter, means the net income or loss of the Atlantic Utility Group determined in accordance with GAAP minus the Notional Fixed Charge plus the amount of amortization of goodwill arising from the Atlantic Merger. See "—Certain Definitions." It is anticipated that the Class A Common Stock will, following the Initial Period (as defined below), pay dividends based upon the Company Net Income (Loss) Attributable to the Atlantic Utility Group, which would include the adjustments for the Notional Fixed Charge and amortization of goodwill arising from the Mergers. See "—Dividends—Dividend Policies." Accordingly, the Class A Common Stock reflects the intent of Delmarva and Atlantic to have a portion of the consideration distributable to the holders of Atlantic Common Stock pursuant to the Atlantic Merger provide a financial return to such holders based upon the earnings in excess of \$40 million annually (after the inclusion of amortized goodwill) of the presently regulated businesses of Atlantic. See "Background of the Mergers." Funds representing such earnings will not necessarily be held in any segregated accounts of the Atlantic Utility Group.

The principal advantage of this structure is that it addresses the concerns of the managements of both Delmarva and Atlantic and allows the respective stockholders of Delmarva and Atlantic to gain the level of exposure to the growth prospects of, and uncertainties associated with deregulation of, the regulated electric utility business of Atlantic that the respective managements have deemed advisable. The level of such exposure is reflected primarily by (i) the amount of the Notional Fixed Charge (defined below) (\$40 million), (ii) the percentage of equity interest in the Company attributable to the Atlantic Utility Group represented by the outstanding Class A Common Stock (30%) and (iii) the percentages of Company Common Stock and Class A Common Stock issued to the Atlantic stockholders (39.4% and 100%) and to the Delmarva stockholders (60.6% and 0%). Through the ownership of 39.4% of the Company Common Stock, the Atlantic stockholders will have the benefit of 27.6% (39.4% of 70%) of the interest in the earnings attributable to the current regulated utility business of ACE in excess of \$40 million per year as well as a 39.4% of the interest in the first \$40 million of such earnings. Accordingly, together with the 30% interest in such earnings represented by the Class A Common Stock, Atlantic stockholders will retain more than half the benefits and risks related to such business. Such exposure will be realized by the respective stockholders to the extent that dividends on the Class A Common Stock will actually be based on, and to the extent that the market price of the Class A Common Stock will actually reflect, the earnings of the Atlantic Utility Group in excess of the Notional Fixed Charge, which are subject to significant uncertainties. The principal disadvantage of this structure is that it creates a complex capital structure for the Company which could have an adverse effect on the trading market for the shares of both the Class A Common Stock and the Company Common Stock. See "Risk Factors-Risk Factors Associated with the Class A Common Stock-No Assurance as to Market Price of Class A Common Stock" and "-Risk Factors Associated with the Company Common Stock-No Assurance as to Market Price of Company Common Stock." For a discussion of risk factors associated with this dual class capital structure, see "Risk Factors-Risk Factors Associated with the Dual Class Common Equity Structure."

The market prices of the Class A Common Stock after the consummation of the Mergers would be determined in the trading markets and could be influenced by many factors, including the results of the Atlantic Utility Group, investors' expectations for the Atlantic Utility Group, trading volumes, regulatory environment and general economic and market conditions. There can be no assurance as to the extent to which investors would assign values to the Class A Common Stock based on the reported financial results or other measures of performance or prospects of the relevant businesses. Financial effects of the Atlantic Utility Group that affect the Company's consolidated results of operations or financial condition could affect the market prices of the Class A Common Stock.

The Company Common Stock. The Company Common Stock will represent the equity value of the Company that is not represented by the Class A Common Stock. This will include (i) the value attributable to the businesses, assets and liabilities of the Company other than those of the Atlantic Utility Group and (ii) the value attributable to the Atlantic Utility Group that is not represented by the outstanding shares of Class A Common Stock but is retained by the Company through its Intergroup Interest in the Atlantic Utility Group, as described below under "—Intergroup Interest."

Dividends—General. The holders of Common Stock will be entitled to receive such dividends as the Company Board may from time to time declare, subject to the rights and preferences of holders of Preferred Stock, if any. The Company's ability to pay dividends will depend primarily upon the ability of its subsidiaries, including Delmarva and ACE, to pay dividends or otherwise transfer funds to it. Financing arrangements, charter provisions and regulatory requirements may impose restrictions on the ability of the Company's subsidiaries to transfer funds to the Company in the form of cash dividends, loans or advances. Such charter provisions include the provisions in the Delmarva Charter and in the ACE Charter that, respectively, preclude the payment of dividends on Delmarva Common Stock and on ACE Common Stock if there are any arrearages in payment of dividends on the Delmarva Preferred Stock or on the ACE Preferred Stock, respectively. Under the 1935 Act, the SEC has the power to preclude the payment of dividends by Delmarva and ACE to the Company or to preclude the payment of dividends by the Company inder certain circumstances. Revenues of Delmarva's and ACE's electric utility businesses are dependent upon rates determined in proceedings before regulatory agencies with jurisdiction over the companies' respective electric utility rate structures. These rates are presently set at levels that cover the utilities' cost of service to their customers, including an equity return on the capital used in providing such service. The weighted average allowed equity rate of return for Delmarva's jurisdictions is approximately 11.5%. The allowed equity rate of return for Atlantic is approximately 12.5%. Changes in regulation could result in certain revenues being based on market rates or on other forms of price regulation. See "Regulatory Matters-Effects of Certain Regulatory Trends." Restrictions on electric utility rates affecting the ability of Delmarva or ACE to pay dividends will affect the Company's ability to pay dividends on both the Company Common Stock and the Class A Common Stock, since such restrictions will apply to actual subsidiaries of the Company rather than to the notional Groups. To the extent specifically applicable to the electric utility rates of ACE, such restrictions will affect the revenues available to the Atlantic Utility Group.

Available Amounts. Dividends on the Common Stock will be limited to funds of the Company legally available under the DGCL. The funds of the Company legally available for dividends on both the Company Common Stock and the Class A Common Stock will be determined on the basis of the entire Company and not on the basis of, respectively, the Residual Group and the Atlantic Utility Group. Consequently, the amount of funds legally available will be reduced to reflect any net losses of either Group and any dividends on, or repurchases of, Company Common Stock or Class A Common Stock.

The Company Charter further limits (i) dividends on the Company Common Stock to an amount not to exceed the Residual Group Available Dividend Amount and (ii) dividends on the Class A Common Stock to an amount not to exceed the Atlantic Utility Group Available Dividend Amount. The Residual Group Available Dividend Amount is intended to reflect the amount that would be legally available for the payment of dividends on the Company Common Stock under the DGCL if the Residual Group were a

77

Delaware corporation distinct from the Atlantic Utility Group subject to the adjustments for the Notional Fixed Charge and amortization of goodwill arising from the Mergers (to the extent but only to the extent such goodwill would otherwise decrease the Company Net Income Attributable to the Atlantic Utility Group or increase the Company Net Loss Attributable to the Atlantic Utility Group). The Atlantic Utility Group Available Dividend Amount is intended to reflect the amount that would be legally available for the payment of dividends on the Class A Common Stock under the DGCL if the Atlantic Utility Group were a Delaware corporation distinct from the Residual Group subject to the adjustments for the Notional Fixed Charge and amortization of goodwill arising from the Mergers (to the extent but only to the extent such goodwill would otherwise decrease the Company Net Income Attributable to the Atlantic Utility Group or increase the Company Net Income Attributable to the Atlantic Utility Group or increase the Company Net Income Attributable to the Atlantic Utility Group or increase the Company Net Income Attributable to the Atlantic Utility Group or increase the Company Net Income Attributable to the Atlantic Utility Group or increase the Company Net Loss Attributable to the Atlantic Utility Group).

Limitations on Share Distributions. If at any time after the consummation of the Mergers a distribution is made in Common Stock or any other securities convertible into Common Stock or Preferred Stock (a "share distribution"), such share distribution will be declared and made only as follows:

(i) a share distribution consisting of shares of Company Common Stock (or securities convertible into shares of Company Common Stock) to holders of shares of Company Common Stock or shares of Preferred Stock attributed to the Residual Group;

(ii) a share distribution consisting of shares of Class A Common Stock (or securities convertible into shares of Class A Common Stock) to holders of shares of Class A Common Stock or shares of Preferred Stock attributed to the Atlantic Utility Group; and

(iii) a share distribution consisting of shares of Class A Common Stock (or securities convertible into shares of Class A Common Stock) to holders of shares of Company Common Stock or shares of Preferred Stock attributed to the Residual Group, provided that the sum of (1) the number of shares of Class A Common Stock to be so issued (or the number of such shares that would be issuable upon conversion of any Convertible Securities to be so issued) and (2) the number of such shares issuable upon conversion of any Convertible Securities then outstanding that are attributed to the Residual Group, is less than or equal to the Number of Shares Issuable with Respect to the Intergroup Interest

Dividend Policies. The Merger Agreement provides, subject to declaration by the Company Board and the obligation of the Company Board to react to the financial condition and regulatory environment of the Company and its results of operations, that the dividends declared and paid on the Class A Common Stock will be maintained at a level of \$3.20 per share per annum through the Initial Period. Thereafter, it is the intention of the Company, subject to declaration by the Company Board and the obligation of the Company Board to consider the financial condition and regulatory environment of the Company and the results of its operations, to pay annual dividends on the Class A Common Stock in an aggregate amount (the amount of such dividends to include the amount credited to the Intergroup Interest as described under "-Intergroup Interest-Adjustments in Connection with Various Transactions-Dividends") equal to 90% of Company Net Income Attributable to the Atlantic Utility Group. The Company Net Income Attributable to the Atlantic Utility Group for the fiscal year ended December 31, 1995 and the 12-month period ended September 30, 1996 would have been \$44.1 million and \$31.8 million (i.e., \$84.1 million and \$71.8 million minus the \$40 million Notional Fixed Charge), respectively. The Company Net Income Attributable to the Atlantic Utility Group for these same periods would have been \$57.6 million and \$49.1 million (i.e., \$97.6 million and \$89.1 million minus the \$40 million Notional Fixed Charge), respectively, not taking into account the higher than expected costs associated with the Salem Nuclear Generating Station described under "Selected Information Concerning Delmarva and Atlantic-Salem Nuclear Generating Station" and the effects of foregone recovery of certain fuel costs under the Southern New Jersey Economic Initiative, which are not expected to recur, and the effect of abnormal weather. The Merger Agreement further provides that notwithstanding the aforesaid intention with respect to dividends on the Class A Common Stock following the Initial Period, if and to the extent that the annual dividends paid on the Class A Common Stock during the Initial Period (including the amount credited to the Intergroup Interest as aforesaid) shall have exceeded 100% of Company Net Income Attributable to the

Atlantic Utility Group during such period, the Company Board may consider such fact in determining the appropriate annual dividend rate on the Class A Common Stock following the Initial Period. There can be no assurance that the Company's earnings attributable to the Atlantic Utility Group will be sufficient to cover dividends on the Class A Common Stock during the Initial Period. **Dividends on the Class A Common Stock will not be cumulative.** In addition, Delmarva and Atlantic presently expect the Company to pay an annual dividend of \$1.54 per share on the Company Common Stock at the time of the consummation of the Mergers.

As an example, if in any year the Company Board determines to pay dividends on the Class A Stock equal to 90% of the Company Net Income Attributable to the Atlantic Utility Group, assuming Net Income Attributable to the Atlantic Utility Group of \$100 million, the aggregate amount of dividends payable to holders of the outstanding Class A Common Stock would equal \$16.2 million, calculated as follows:

	(in millions)
Net Income of Atlantic Utility Group (Assumed):	\$100.0
Notional Fixed Charge	(40.0)
Company Net Income Attributable to the Atlantic Utility Group	\$ 60.0
After 70% Intergroup Interest	\$ 18.0
Dividends Payable to Holders of Class A Common Stock	\$ 16.2

The financial results of the Atlantic Utility Group will be substantially identical to the financial results of ACE. Since ACE will remain a registrant following the Mergers, because of securities of ACE held by the public, goodwill resulting from the Mergers will not be pushed down to ACE's balance sheet and amortization of goodwill will not reduce ACE's net income. Therefore, amortization of goodwill resulting from the Mergers of Atlantic Utility Group above. See Note (o) to the Unaudited Pro Forma Combined Financial Statements for examples of the computation of Earnings Applicable to Class A Common Stock.

The policies described above have been agreed to by Delmarva and Atlantic, but are not contained within the Company Charter, which is the instrument that establishes the terms of the Common Stock. Pursuant to the Merger Agreement, following the Effective Time the Audit Committee of the Company Board will be charged with the responsibility of advising the Company Board with respect to certain intercompany transactions and other fiduciary matters that may relate to the Class A Common Stock. Subject to the limitations described under "-General," "-Available Amounts" and "-Limitations on Share Distributions," the Company Charter provides to the Company Board the authority and discretion to declare and pay dividends exclusively on the Company Common Stock, exclusively on the Class A Common Stock or on both classes of Common Stock notwithstanding the relative amounts of the Residual Group Available Dividend Amount and the Atlantic Utility Group Available Dividend Amount, the amount of dividends previously declared on each class, the respective voting or liquidation rights of each class, the aforesaid dividend policies or any other factor. The Company Charter does not require the declaration or payment of any dividends on the Company Common Stock or on the Class A Common Stock and establishes no priority or preference in favor of either security with respect to the other or with respect to any other security the Company may issue. In addition, any net losses of either Group, dividends or distributions on, or repurchases of, the Company Common Stock or the Class A Common Stock, and dividends on, or certain repurchases of, preferred stock, will reduce funds of the Company legally available for the payment of dividends on both the Company Common Stock and the Class A Common Stock.

## **Intergroup Interest**

The Outstanding Atlantic Utility Fraction and the Intergroup Interest Fraction. The Class A Common Stock to be issued to the holders of Atlantic Common Stock upon the consummation of the Mergers will at the time of such consummation represent 30% of the equity value attributable to the Atlantic Utility Group, including a proportionate undivided interest in the individual businesses, assets and liabilities

thereof. This proportion is expressed in the "Outstanding Atlantic Utility Fraction," which is defined to mean the percentage interest in the equity value of the Company attributable to the Atlantic Utility Group that is represented at any time by the outstanding shares of Class A Common Stock. The remainder of the equity value of the Company attributable to the Atlantic Utility Group will be retained by the Company through its "Intergroup Interest" in the Atlantic Utility Group, which upon consummation of the Mergers will represent the balance, or 70%, of the equity value of the Company so attributable. This proportion is expressed in the "Intergroup Interest Fraction," which is defined to mean the percentage interest in the equity value of the Company so attributable that is attributed to the Company. Following the Mergers the Intergroup Interest will be subject to increase or decrease under the circumstances specified in the Company Charter for adjusting the Number of Shares Issuable with Respect to the Intergroup Interest. The sum of the Outstanding Atlantic Utility Fraction and the Intergroup Interest Fraction will always equal 100%. See "—Certain Definitions."

Number of Shares Issuable with Respect to the Intergroup Interest. The term "Number of Shares Issuable with Respect to the Intergroup Interest" refers to the number of shares of Class A Common Stock that could be issued or sold by the Company for the account of the Company in respect of the Intergroup Interest. See "—Certain Definitions." The Number of Shares Issuable with Respect to the Intergroup Interest will not be represented by outstanding shares of Class A Common Stock and accordingly, the Company will not be entitled to any voting rights in respect of the Intergroup Interest. In addition, outstanding shares of Class A Common Stock that are held by majority-owned subsidiaries of the Company (as to which the Company owns a majority of the shares entitled to vote in the election of directors) will not be entitled to vote on matters presented to stockholders or be counted for quorum purposes under the DGCL. Accordingly, the outcome of any vote of the Class A Common Stock will be determined by the holders of the outstanding shares of Class A Common Stock (excluding any shares held by such majority-owned subsidiaries of the Company).

For financial reporting purposes, shares of Class A Common Stock acquired by consolidated subsidiaries of the Company that remain outstanding following such acquisition would be combined with the Number of Shares Issuable with Respect to the Intergroup Interest and reported as the Company's investment in the Atlantic Utility Group. Any differences between such reported investment and the then existing Intergroup Interest would be reconcilable by adding to the then existing Intergroup Interest the number of outstanding shares of Class A Common Stock held by consolidated subsidiaries of the Company. Since these shares would still be outstanding for purposes of the receipt of dividends and payment of redemption or liquidation amounts, the Company would obtain substantially the same economic benefits from such outstanding shares as it would have received had such shares been retired or otherwise ceased to be outstanding following their purchase and added to the Number of Shares Issuable with Respect to the Intergroup Interest.

Adjustments in Connection with Various Transactions. Following the consummation of the Mergers, adjustments will be made in the Outstanding Atlantic Utility Fraction, the Intergroup Interest Fraction, the financial statements for the Groups, and/or the Number of Shares Issuable with Respect to the Intergroup Interest in a variety of transactional contexts. The following transactions are illustrative only and not indicative of any plans with respect to the Company following the consummation of the Mergers.

Dividends. Except as described below, from and after the payment of any dividend or other distribution to the holders of outstanding shares of Class A Common Stock, the Residual Group will be credited, and the Atlantic Utility Group will be charged (in addition to the charge to the Atlantic Utility Group for the amounts distributed to such holders), an amount equal to the product of (i) the aggregate amount distributed in respect of the outstanding Class A Common Stock times (ii) a fraction (for purposes of this and the following paragraph, the "Fraction") the numerator of which is the Intergroup Interest Fraction and the denominator of which is the Outstanding Atlantic Utility Fraction. At the consummation of the Mergers, the Fraction will equal 2<sup>1</sup>/<sub>3</sub>. As the Fraction increases or decreases as a consequence of an increase or decrease in the Intergroup Interest Fraction, the proportionate credit to the Residual Group

and proportionate charge to the Atlantic Utility Group for an amount distributed to the holders of outstanding shares of Class A Common Stock will be increased or decreased accordingly. In the event the Atlantic Utility Group did not effect an actual cash payment to the Company in the amount of such credit and charge when a dividend is paid on the Class A Common Stock, the Company may account for the credit and charge in a number of possible manners. For instance, the Company could reduce the capital of the Atlantic Utility Group to the extent of the amount of such credit and charge. Alternatively, the Company could treat the amount effectively as a loan from the Residual Group to the Atlantic Utility Group, on which interest would be payable. The Company could also adjust for the credit and the charge by reducing the Outstanding Atlantic Utility Fraction and increasing the Intergroup Interest Fraction.

In the case of a distribution payable in shares of Class A Common Stock, an appropriate adjustment, as determined by the Company Board, will be made in the Number of Shares Issuable with Respect to the Intergroup Interest. In the case of a distribution payable in securities of the Company attributed to the Atlantic Utility Group, the Residual Group will be deemed to hold an interest in the Atlantic Utility Group in the form of such securities in an amount equal to the product of the number of securities so distributed times the Fraction. For certain further adjustments with respect to distributed securities, see the Company Charter's definition of "Residual Group" as set forth under "—Certain Definitions."

See Annex VI for illustrations of the effect of dividends payable in Class A Common Stock and of the calculation of the proportionate credits and charges to the Residual Group and the Atlantic Utility Group, respectively, to be made in connection with the payment of dividends and other distributions on outstanding shares of Class A Common Stock.

Contributions to Atlantic Utility Group. The Company Board could, in its sole discretion, determine from time to time to contribute cash or other property of the Residual Group as additional equity to the Atlantic Utility Group and increase the Number of Shares Issuable with Respect to the Intergroup Interest by the number determined by dividing the amount of such cash or the fair value (as determined by the Company Board) of such property by the Market Value of one share of Class A Common Stock as of the late of such contribution. In such event, the Intergroup Interest Fraction would increase and the Dutstanding Atlantic Utility Fraction would decrease accordingly. The Company Board could, in its sole discretion, also determine from time to time to transfer cash or other property of the Atlantic Utility Group from the Atlantic Utility Group to the Residual Group in respect of a reduction in the Intergroup Interest, and the Number of Shares Issuable with Respect to the Intergroup Interest would be decreased by the amount of such cash or the fair value (as determined by the Company Board) of such property divided by the Market Value of one share of Class A Common Stock as of the date of such contribution. In such event, the Intergroup Interest Fraction would decrease and the Outstanding Atlantic Utility Fraction would increase accordingly. The Company Board of such contribution. In such event, the Intergroup Interest Fraction would decrease and the Outstanding Atlantic Utility Fraction would increase accordingly. The Company Board could, in its sole discretion, determine to make contributions or other transfers referred to in this paragraph.

Repurchases of Class A Common Stock. If shares of Class A Common Stock were retired or otherwise ceased to be outstanding following their repurchase with funds attributed to the Residual Group the Number of Shares Issuable with Respect to the Intergroup Interest would increase on a share-for-share basis and the Intergroup Interest Fraction would increase and the Outstanding Atlantic Utility Fraction would decrease accordingly. If the purchase of shares of Class A Common Stock were attributed to the Atlantic Utility Group, the Number of Shares Issuable with Respect to the Intergroup Interest would remain the same but the Intergroup Interest Fraction would increase and the Outstanding Atlantic Utility Fraction would decrease. The Company Board would, in its sole discretion, determine whether purchases of Class A Common Stock should be made with consideration attributed to the Residual Group or the Atlantic Utility Group on the basis of such factors as it deems relevant.

Issuances of Class A Common Stock. Although the Company has no present intention to issue any shares of Class A Common Stock following the issuance of shares of Class A Common Stock to the holders of Atlantic Common Stock pursuant to the Atlantic Merger, the authorized shares of Class A Common Stock in excess of the total number of shares outstanding will be available for issuance or sale without



further approval by the stockholders of the Company and may be issued at any time at prices that could dilute the value of the outstanding shares of Class A Common Stock. In the event that shares of Class A Common Stock are issued or sold by the Company following the consummation of the Mergers, the Company will identify (i) the number of shares of Class A Common Stock issued or sold for the account of the Residual Group in respect of a reduction in the Intergroup Interest, the net proceeds from the sale of which would be reflected in the combined financial statements of the Residual Group, and (ii) the number of such shares issued or sold for the account of the Atlantic Utility Group as an additional equity interest in the Atlantic Utility Group, the net proceeds of which will be reflected in the combined financial statements of the Atlantic Utility Group. The Company Board would make such determination, in its sole discretion, after consideration of such factors as it deems relevant. In the event that, following the consummation of the Mergers, additional shares of Class A Common Stock were issued or sold by the Company for the account of the Residual Group in respect of a reduction in the Intergroup Interest, the Number of Shares Issuable with Respect to the Intergroup Interest would decrease on a share-for-share basis, and the Intergroup Interest Fraction would decrease and the Outstanding Atlantic Utility Fraction increase accordingly. If the Number of Shares Issuable with Respect to the Intergroup Interest were reduced to zero as a result of such issuances, shares of Class A Common Stock could no longer be issued or sold by the Company for the account of the Residual Group unless an Intergroup Interest were again subsequently created. If the net proceeds of any issuance or sale by the Company of Class A Common Stock were allocated to the Atlantic Utility Group, the Number of Shares Issuable with respect to the Intergroup Interest would not be reduced, but the Intergroup Interest Fraction would decrease and the Outstanding Atlantic Utility Fraction would increase accordingly.

Although the Company has no present intention to do so, if the Company Board determined to issue or deliver shares of Class A Common Stock as a distribution on the Company Common Stock, such distribution would be treated as a distribution of shares issuable with respect to the Intergroup Interest and, as a result, the Number of Shares Issuable with Respect to the Intergroup Interest would decrease by the number of shares distributed to the holders of Company Common Stock, resulting in a proportionate decrease in the Intergroup Interest Fraction and increase in the Outstanding Atlantic Utility Fraction.

See Annex VI for illustrations of the calculation of the Outstanding Atlantic Utility Fraction, the Intergroup Interest Fraction and the Number of Shares Issuable with Respect to the Intergroup Interest in connection with the aforesaid transactions.

# **Conversion and Redemption of the Class A Common Stock**

...

Conversion at the Option of the Company. The Company may at any time convert each share of Class A Common Stock into the number of shares of Company Common Stock equal to the applicable percentage, on the date of such conversion, set forth below of the Market Value Ratio (as defined in the Company Charter; see "—Certain Definitions") of the Class A Common Stock to the Company Common Stock as of the fifth Trading Day prior to the date of the notice of such conversion:

12-Month Period Prior to the Applicable Anniversary of the Effective Date	· •	۰.	· 1		Percentage of Market Value Ratio
First			· · · · · · · · · · · ·		125%
Second					120%
Third					115%
Fourth and T	hereafter .	• • • • • • •		••••••	110%

Any such conversion would dilute the interests of holders of Company Common Stock and would preclude holders of Class A Common Stock from retaining their interest in a security reflecting separately the business of the Atlantic Utility Group.

Redemption or Conversion Following a Tender or Exchange Offer. If the Company consummates (i) a tender offer for all of the Class A Common Stock at an all cash price of at least 110% of the Time-Weighted Market Price (as defined in the Company Charter; see "-Certain Definitions") of a share of Class A Common Stock as of the Trading Day immediately preceding the date of such offer or (ii) an offer to exchange each share of Class A Common Stock into a number of shares of Company Common Stock equal to at least 110% of the Market Value Ratio of the Class A Common Stock to the Company Common Stock as of the Trading Day immediately preceding the date of such offer, that, in either case, is accepted by the holders of more than 50% of the Class A Common Stock, the Company may either (x) (if funds are legally available therefor) redeem each share of Class A Common Stock remaining outstanding in exchange for cash in an amount equal to the highest cash price paid per share by the Company pursuant to such tender offer or to the product of the highest number of shares of Company Common Stock per share issued in exchange for any share of Class A Common Stock pursuant to such exchange offer and the Time-Weighted Market Price of a share of Company Common Stock as of the Trading Day immediately preceding the date of such exchange offer, as the case may be, or (y) convert each share of Class A Common Stock remaining outstanding into a number of shares of Company Common Stock equal to (1) the quotient of (A) the highest cash price paid per share by the Company pursuant to such tender offer and (B) the Time-Weighted Market Price of a share of Company Common Stock as of the Trading Day immediately preceding the date of the notice of such conversion or to (2) the highest number of shares of Company Common Stock per share issued in exchange for any share of Class A Common Stock pursuant to any such exchange offer.

If any person (including the Company) consummates a tender offer for all of the outstanding shares of Company Common Stock at an all cash price that is accepted by the holders of more than 50% of the Company Common Stock, the Company may either (x) redeem each share of Class A Common Stock for cash in an amount equal to the product of (1) the highest cash price paid per share by such person pursuant to such tender offer and (2) the Market Value Ratio of the Class A Common Stock to the Company Common Stock as of the fifth Trading Day prior to the date of such tender offer or (y) convert each share of Class A Common Stock into a number of shares of Company Common Stock equal to the quotient of (A) the highest cash price paid per share by such person pursuant to such tender offer and the Time-Weighted Market Price of a share of Company Common Stock as of the Trading Day immediately preceding the date of the notice of such conversion.

Participation in a Tender Offer. If any person (including the Company) makes a tender offer to purchase shares of Company Common Stock for cash, property or other securities, any holder of Class A Common Stock may elect to convert each and any of such holder's shares of Class A Common Stock into the number of shares of Company Common Stock equal to 100% of the Market Value Ratio of the Class A Common Stock to the Company Common Stock as of the Trading Day immediately preceding the date of such tender offer; provided that (x) any such election may be revoked by notice to the Company prior to the consummation of such tender offer, and (y) such conversion will only be effective with respect to such shares of Company Common Stock issuable upon such conversion that are accepted for purchase pursuant to such tender offer.

Dividend, Redemption or Conversion upon Disposition of All or Substantially All Assets of the Atlantic Utility Group. Upon the Disposition, in one transaction or a series of related transactions, by the Company and/or its subsidiaries of all or substantially all of the properties and assets of the Atlantic Utility Group to one or more entities (other than (i) in connection with the liquidation, dissolution or winding up of the Company, (ii) the distribution of the properties and assets of the Atlantic Utility Group to all holders of shares of Class A Common Stock and to the Company or subsidiaries thereof, divided among such holders and the Company or subsidiaries thereof on a pro rata basis in accordance with the number of shares of Class A Common Stock outstanding and the Number of Shares Issuable with Respect to the Intergroup Interest, (iii) to any entity controlled by the Company or (iv) in connection with a Related

Business Transaction (as defined in the Company Charter; see "-Certain Definitions")), the Company will, within 85 Trading Days following such Disposition either:

(I) (a) if there are assets of the Company legally available therefor and the Atlantic Utility Group Available Dividend Amount is sufficient therefor, pay to the holders of Class A Common Stock a dividend in cash and/or in securities (other than Common Stock) or other property having an aggregate Fair Value (as defined in the Company Charter; see "—Certain Definitions") as of the date of such Disposition equal to the product of (x) the Outstanding Atlantic Utility Fraction as of the record date for determining holders entitled to receive such dividend times (y) the Fair Value of the Net Proceeds of such Disposition; or

(b) subject to the Atlantic Utility Group Available Dividend Amount and other limitations as set forth in the Company Charter, if such Disposition involves all (and not merely substantially all) of the properties and assets attributed to the Atlantic Utility Group, redeem all of the Class A Common Stock for cash and/or for securities (other than Common Stock) or other property having a Fair Value as of the date of such Disposition equal to the product of (x) the Outstanding Atlantic Utility Fraction as of the date of such redemption times (y) the Fair Value of the Net Proceeds of such Disposition; or

(c) subject to the Atlantic Utility Group Available Dividend Amount and other limitations as set forth in the Company Charter, if such Disposition involves substantially all (but not all) of the properties and assets attributed to the Atlantic Utility Group, redeem the number of whole shares of Class A Common Stock (which may be all of such shares outstanding) whose aggregate average Market Value during the ten Trading Days beginning on the 16th Trading Day immediately succeeding such Disposition is closest to the product of (x) the Outstanding Atlantic Utility Fraction as of the date such shares are selected for redemption times (y) the Fair Value, as of the date of such Disposition, of the Net Proceeds of such Disposition in consideration for cash and/or securities (other than Company Common Stock) or other property having a Fair Value in the aggregate equal to such product; or

(II) convert each share of Class A Common Stock into a number of shares of Company Common Stock equal to 110% of the ratio of the average Market Value of one share of Class A Common Stock over the ten Trading Days beginning on the 16th Trading Day following the date of such Disposition to the average Market Value of one share of Company Common Stock over the same ten Trading Day period.

The Company Charter defines "substantially all of the properties and assets of the Atlantic Utility Group" to mean a portion of such properties and assets (x) that represents at least 80% of the Fair Value of the properties and assets attributed to the Atlantic Utility Group as of such date or (y) from which were derived at least 80% of the aggregate revenues for the immediately preceding twelve fiscal quarterly periods of the Company (calculated on a pro forma basis to include revenues derived from any of such properties and assets acquired during such period) derived from the properties and assets of the Atlantic Utility Group as of such date.

In the event of a dividend or redemption as described in the preceding clause (I), the Company Board may convert the remaining Class A Common Stock prior to the first anniversary of such dividend or such redemption, as the case may be, into a number of shares of Company Common Stock equal to 110% of the Market Value Ratio of the Class A Common Stock to the Company Common Stock as of the 5th Trading Day prior to the date of the notice of such conversion.

See Annex VI for illustrations of certain calculations in connection with a Disposition of all or substantially all of the properties and assets of the Atlantic Utility Group.

Notice and Other Provisions with Respect to Conversion and Redemption of the Class A Common Stock

Notice Following an Optional Conversion. If the Company determines to convert the Class A Common Stock into Company Common Stock pursuant to the provisions described under "—Conversion and Redemption of the Class A Common Stock—Conversion at the Option of the Company" or "—Conversion and Redemption of the Class A Common Stock—Redemption or Conversion Following a Tender or Exchange Offer" or pursuant to the provisions described in clause (II) under "—Dividend, Redemption or Conversion upon Disposition of All or Substantially All Assets of the Atlantic Utility Group," the Company, will within 35 to 45 Trading Days prior to the conversion date, furnish certain information to each holder of Class A Common Stock and of securities conversion after a Disposition, shall not be more than 85 Trading Days following the consummation of such Disposition and, in the case of a conversion following a tender or exchange offer, shall not be less than 35 or more than 85 Trading Days following the consummation of such Offer), the number of shares of Class A Common Stock into which each share of Class A Common Stock will be converted and the number of shares of Class A Common Stock outstanding and covered by outstanding Convertible Securities.

- Notice Following a Disposition of All or Substantially All of the Atlantic Utility Group Assets. The Company Charter provides that not later than the 10th Trading Day following a Disposition referred to above under "-Conversion and Redemption of the Class A Common Stock-Dividend, Redemption or Conversion upon Disposition of All or Substantially All Assets of the Atlantic Utility Group," the Company will publicly announce (i) the Net Proceeds of such Disposition, (ii) the number of outstanding shares of Class A Common Stock, (iii) the number of outstanding shares of Class A Common Stock into or for which Convertible Securities are then convertible and the conversion price thereof and (iv) the Outstanding Atlantic Utility Fraction on the date of such notice. Not earlier than the 26th Trading Day following such Disposition, the Company will announce publicly by press release which of the actions described under "---Conversion and Redemption of the Class A Common Stock-Dividend, Redemption or Conversion upon Disposition of All or Substantially All Assets of the Atlantic Utility Group" it has determined to take. If the Company determines to pay a dividend on shares of Class A Common Stock, the Company will, within 30 Trading Days following such Disposition, furnish certain information to each holder of Class A Common Stock and each holder of securities convertible into shares of Class A Common Stock, including the record date and anticipated payment date for such dividend (which shall not be more than 85 Trading Days following the consummation of such Disposition), the type of property to be distributed, the Net Proceeds of the Disposition, the applicable Outstanding Atlantic Utility Fraction and the number of shares of Class A Common Stock outstanding and covered by outstanding Convertible Securities. If the Company determines to undertake a redemption of shares of Class A Common Stock, the Company will, between 35 and 45 Trading Days prior to the date of redemption, furnish certain information to each holder of Class A Common Stock and each holder of securities convertible into shares of Class A Common Stock, including notice of the redemption, the anticipated redemption date (which shall not be more than 85 Trading Days following the consummation of such Disposition), the type of property in which the redemption price is to be paid, the Net Proceeds of the Disposition, the applicable Outstanding Atlantic Utility Fraction and the number of shares of Class A Common Stock outstanding and covered by outstanding Convertible Securities. If the Company determines to undertake a redemption of shares of Class A Common Stock following a Disposition of substantially all (but not all) of the properties and assets of the Atlantic Utility Group, the Company will within 30 Trading Days following such Disposition furnish certain information to each holder of Class A Common Stock and each holder of securities convertible into shares of Class A Common Stock, including the anticipated redemption date (which shall not be more than 85 Trading Days following the consummation of such Disposition), the type of property in which the redemption price is to be paid, the Net Proceeds of such Disposition, the

applicable Outstanding Atlantic Utility Fraction and the number of shares of Class A Common Stock outstanding and covered by outstanding Convertible Securities.

Notice Following Consummation of a Tender or Exchange Offer. If the Company determines to redeem Class A Common Stock following consummation of a tender or exchange offer as described under "—Conversion and Redemption of Class A Common Stock—Redemption or Conversion Following a Tender or Exchange Offer," the Company will within 35 to 45 Trading Days prior to the redemption date furnish certain information to each holder of shares of Class A Common Stock and of securities convertible into shares of Class A Common Stock, including the redemption date (which shall not be less than 35 or more than 85 Trading Days following the consummation of the tender or exchange offer), the redemption price and the number of shares of Class A Common Stock outstanding and covered by outstanding convertible securities.

Validity Not Affected. Neither the failure to mail any notice to any particular holder of Class A Common Stock or of convertible securities nor any defect therein will affect the sufficiency thereof with respect to any other holder of outstanding shares of Class A Common Stock or of convertible securities or the validity of any conversion or redemption.

No Fractional Shares Required. The Company will not be required to issue or deliver fractional shares of any class of capital stock or any other securities to any holder of Class A Common Stock upon any conversion, redemption, dividend or other distribution. If more than one share of Class A Common Stock will be held at the same time by the same holder, the Company may aggregate the number of shares of any capital stock that shall be issuable or any other securities or property distributable to such holder upon any conversion, redemption, dividend or other distribution (including any fractional shares). If fractional shares of any capital stock or of any other securities are otherwise required to be issued or distributed to holders of Class A Common Stock, the Company will, if such fractional shares are not issued or distributed to the holder, pay cash in respect of such fractional shares in an amount equal to the Fair Value thereof on the 5th Trading Day prior to the date such payment is to be made (without interest).

Selection of Securities in a Partial Redemption. If less than all of the outstanding shares of Class A Common Stock are to be redeemed in connection with a Disposition of substantially all (but less than all) of the assets of Atlantic Utility Group, the shares to be redeemed by the Company will be selected on a pro rata basis or by lot or by such other method as the Company Board may determine to be equitable.

No Dividend Adjustments. No adjustments in respect of dividends may be made upon the conversion or redemption of any shares of Class A Common Stock, but if the date of any such conversion or redemption is subsequent to the record date for the payment of a dividend or other distribution, the holders of shares of Class A Common Stock at the close of business on such record date will be entitled to receive such distribution without interest, notwithstanding the subsequent conversion or redemption of such shares.

Effect of Conversion or Redemption on Rights. From and after any applicable conversion or redemption date, all rights of a holder of shares of Class A Common Stock that were converted or redeemed will cease except for the right, upon surrender of the certificates representing such shares, to receive cash and/ or certificates or instruments representing shares of the kind of capital stock and/or other securities or property for which such shares were converted or redeemed, together with any payment in respect of fractional shares and rights to dividends, in each case without interest. No holder of a certificate that immediately prior to the applicable conversion date represented shares of Class A Common Stock will be entitled to receive any distribution or interest payment with respect to shares of any kind of capital stock or other security or instrument for which Class A Common Stock was converted until the surrender of such certificate in exchange for a certificate or instrument representing such capital stock or other security. Upon such surrender, there will be paid to the holder the amount of any distributions (without interest) that theretofore became payable on any class of capital stock of the Company as of a record date after the conversion date, but that were not paid by reason of the foregoing, with respect to the number of whole shares of the kind of capital stock represented by the certificate or certificates issued upon such surrender. From and after a conversion date, the Company will, however, be entitled to treat the certificates for Class A Common Stock that have not yet been surrendered for conversion as evidencing the ownership of the number of whole shares of the kind or kinds of capital stock of the Company for which the shares of Class A Common Stock represented by such certificates shall have been converted, notwithstanding the failure to surrender such certificates.

Documentary and Stamp Taxes. The Company will pay all documentary, stamp or similar issue or transfer taxes in respect of the issuance or delivery of any shares of capital stock and/or other securities upon conversion or redemption of shares of Class A Common Stock. The Company will not, however, be required to pay any tax in a name other than that in which the shares of Class A Common Stock so converted or redeemed were registered, and no issuance or delivery will be made unless the person requesting such issuance or delivery has paid to the Company the amount of any such tax or has established to the satisfaction of the Company that such tax has been paid.

*Board Discretion.* The Company Board may establish such rules and requirements to facilitate the effectuation of the aforementioned transactions as it determines to be appropriate.

# Allocation of Consideration Between the Company Common Stock and the Class A Common Stock in a Subsequent Merger or Consolidation

In the event of a merger or consolidation to which the Company is a party and pursuant to which the holders of Common Stock are entitled to receive consideration or shares of Common Stock are converted into securities or other consideration, the holders of outstanding shares of Class A Common Stock will be entitled to receive in exchange for each such share, or to have each such share converted into, the consideration attributable pursuant to such merger or consolidation to the number of shares of Company Common Stock equal to the Market Value Ratio of the Class A Common Stock to the Company Common Stock as of the Trading Day immediately preceding the date of the first public announcement of such merger or consolidation subject to adjustments for stock splits and combinations and stock dividends.

# Voting

Holders of Company Common Stock and holders of Class A Common Stock will each be entitled to one vote per share on all matters submitted to a vote at a meeting of stockholders, subject to the rights, if any, of holders of Preferred Stock to vote on a matter as a class or series. Based on the shares of Delmarva Common Stock and Atlantic Common Stock outstanding as of the close of business on September 30, 1996, the Company Common Stock would represent approximately 94.2%, and the Class A Common Stock would represent approximately 5.8%, of the voting power of all classes entitled to vote on matters presented to stockholders. The holders of Common Stock will not be entitled to cumulative votes for the election of directors. Except as may otherwise be required by the laws of the State of Delaware, the holders of Company Common Stock and the holders of Class A Common Stock will vote as one class for all purposes. Neither the holders of Company Common Stock nor the holders of Class A Common Stock will have any rights to vote as a separate class on any matters submitted to a vote at a meeting of stockholders except with respect to certain limited class voting rights provided under the DGCL. Under the DGCL, the approval of the holders of a majority of the outstanding shares of any class of capital stock of a corporation, voting separately as a class, is required to approve any amendment to the certificate of incorporation of the corporation that would alter or change the powers, preferences or special rights of the class so as to affect it adversely. The DGCL does not provide for any other separate voting rights of a class of capital stock (other than with respect to a change in par value or an increase or decrease in the authorized shares of such class).

87

# **Liquidation Rights**

In the event of a liquidation, dissolution of winding up of the Company, whether voluntary or involuntary, after payment or provision for payment of the debts and other liabilities of the Company and subject to the prior payment in full of the preferential amounts to which the Preferred Stock, if any, is entitled, the holders of the outstanding shares of Common Stock will be entitled to receive the remaining assets of the Company, regardless of the Group to which such assets are attributed, divided among the holders of Common Stock in accordance with the per share "Liquidation Units" attributable to each class of Common Stock. Each share of Company Common Stock will be attributed one Liquidation Unit and each share of Class A Common Stock will be attributed one Liquidation Unit, in the case of each such class of Common Stock subject to adjustment as determined by the Company Board to be appropriate to reflect any subdivision (by stock split or otherwise) or combination (by reverse stock split or otherwise) of such class of Common Stock or any dividend or other distribution of shares of such class of Common Stock to holders of shares of such class of Common Stock. Neither the merger nor consolidation of the Company into or with any other company, nor the merger or consolidation of any other company into or with the Company, nor a sale, transfer or lease of all or any part of the assets of the Company, will be deemed a liquidation or winding up of the Company, or cause the dissolution of the Company, for liquidation rights purposes.

#### **Determinations by the Company Board**

The Company Charter will provide that any determination made by the Company Board under any provisions of the Company Charter governing the capital stock of the Company, and any determinations so made with respect to either Group or the rights of the holders of any class of Common Stock made pursuant to or in furtherance of such provisions, will be final and binding on all stockholders. Such a determination is subject to applicable law and the fiduciary duty of the Company Board. See "Risk Factors—Fiduciary Duties of the Company Board Are to All Stockholders Regardless of Class."

#### **No Preemptive Rights**

The holders of the Common Stock will not have any preemptive rights to subscribe for any additional shares of Common Stock or Preferred Stock or other obligations convertible into shares of Common Stock or Preferred Stock that may be issued following the consummation of the Mergers.

# **Certain Definitions**

"Atlantic Utility Group" means, as of any date from and after the Effective Date:

(A) the interest of the Company on such date in ACE, and any successor companies, and solely those lines of business in which ACE was engaged as of August 9, 1996, and the assets and liabilities attributable to those lines of business, and which meet all of the criteria listed in clause (i) below, as of August 9, 1996, specifically excluding those businesses, lines of business, and products and services listed in clause (ii) below, and specifically including those businesses, lines of business and products and services listed in clause (iii) below:

(i) only those businesses meeting all of the following criteria as of August 9, 1996 shall be included in the Atlantic Utility Group: (a) price regulated by the NJBPU, including, without limitation, off-tariff agreements; (b) directly related to the supply of electricity (generation and purchase of electricity) or the delivery of electricity (transmission and distribution of electricity); and (c) a line of business for which ACE has a franchise;

(ii) specifically excluded from the Atlantic Utility Group are the following businesses, lines of business and products and services, which list is not intended to be inclusive: (a) Appliance Shield program, (b) lighting upgrade programs, (c) water heater service business, (d) thermostat sales, (e) telecommunications business and investments, (f) energy services and consulting, (g) utility services and consulting (*e.g.*, plant services and electrical equipment O&M services), (h) Atlantic Energy International, and (i) Atlantic Energy Enterprises and its subsidiaries;

(iii) specifically included in the Atlantic Utility Group is the Deepwater Operating Company.

(B) all assets and liabilities of the Company and its subsidiaries (other than capital stock of a subsidiary) and liabilities relating to the notional obligation of the Atlantic Utility Group to the Residual Group described in clause (iii) of the definition of Company Net Income (Loss) Attributable to the Atlantic Utility Group on such date attributed by the Company Board to ACE or the businesses thereof, whether or not such assets or liabilities are or were also assets and liabilities of ACE, including, without limitation, the assets and liabilities as of the Effective Date specified in the schedules filed with the records of the actions of the Company Board (a copy of which shall be made available to any stockholder of the Company upon written request therefor);

(C) all properties and assets transferred to the Atlantic Utility Group from the Residual Group (other than a transaction pursuant to paragraph (D) of this definition) after the Effective Date pursuant to transactions in the ordinary course of business of both the Residual Group and the Atlantic Utility Group or otherwise as the Company Board may have directed as permitted by the applicable provisions of the Company Charter;

(D) all properties and assets transferred to the Atlantic Utility Group from the Residual Group in connection with an increase in the Number of Shares Issuable with Respect to the Intergroup Interest; and

(E) the interest of the Company or any of its subsidiaries in any business or asset acquired and any liabilities assumed by the Company or any of its subsidiaries outside of the ordinary course of business and attributed to the Atlantic Utility Group, as determined by the Company Board;

provided that (1) from and after the payment date of any dividend or other distribution with respect to shares of Class A Common Stock (other than a dividend or other distribution payable in shares of Class A Common Stock, with respect to which adjustment shall be made as provided in the Company Charter, or in securities of the Company attributed to the Atlantic Utility Group, for which provision shall be made as set forth in clause (2) of this proviso), the Atlantic Utility Group shall no longer include an amount of assets or properties previously attributed to the Atlantic Utility Group of the same kind as so paid in such dividend or other distribution with respect of shares of Class A Common Stock as have a Fair Value on the record date for such dividend or distribution equal to the product of (a) the Fair Value on such record date of the aggregate of such dividend or distribution to holders of shares of Class A Common Stock declared times (b) a fraction the numerator of which is equal to the Intergroup Interest Fraction in effect on the record date for such dividend or distribution and the denominator of which is equal to the Outstanding Atlantic Utility Fraction in effect on the record date for such dividend or distribution, (2) if the Company shall pay a dividend or make some other distribution with respect to shares of Class A Common Stock payable in securities of the Company that are attributed to the Atlantic Utility Group for purposes of the applicable provisions of the Company Charter (other than Class A Common Stock), there shall be excluded from the Atlantic Utility Group an interest in the Atlantic Utility Group equivalent to the number or amount of such securities that is equal to the product of the number or amount of securities so distributed to holders of Class A Common Stock times the fraction specified in clause 1(b) of this proviso (determined as of the record date for such distribution) (and such interest in the Atlantic Utility Group shall be attributed to the Residual Group) and, to the extent interest is, or dividends are, paid on the securities so distributed, the Atlantic Utility Group shall no longer include a corresponding ratable amount of the kind of assets paid as such interest or dividends as would have been paid in respect of the securities equivalent to such interest in the Atlantic Utility Group deemed held by the Residual Group if the securities equivalent to such interest were outstanding (and in such eventuality such assets as are no longer

included in the Atlantic Utility Group shall be attributed to the Residual Group) and (3) from and after any transfer of any assets or properties from the Atlantic Utility Group to the Residual Group, the Atlantic Utility Group shall no longer include such assets or properties so contributed or transferred. The Company may also, to the extent a dividend or distribution on the Class A Common Stock has been paid in Convertible Securities convertible into for Class A Common Stock, cause such Convertible Securities as are deemed to be held by the Residual Group in accordance with the third to last sentence of the definition of "Residual Group" and clause (2) of the proviso to the immediately preceding sentence to be deemed to be converted, as provided in the penultimate sentence of the definition of "Residual Group," in which case such Convertible Securities shall no longer be deemed to be held by the Residual Group.

"Atlantic Utility Group Available Dividend Amount," on any date, shall mean either:

(i) (x) the amount equal to the product of (1) the Outstanding Atlantic Utility Fraction as of such date multiplied by (2) an amount equal to the fair market value of the total assets attributed to the Atlantic Utility Group less the total amount of the liabilities attributed to the Atlantic Utility Group (provided that Preferred Stock shall not be treated as a liability), in each case as of such date and determined on a basis consistent with that applied in determining Company Net Income (Loss) Attributable to the Atlantic Utility Group, minus (y) the aggregate par value of, or any greater amount determined to be capital in respect of, all outstanding shares of Class A Common Stock and shares of each class or series of Preferred Stock attributed to the Atlantic Utility Group, plus (z) the amount, as of such date, of amortization of goodwill during the period from the Effective Date through such date arising from the Mergers with respect to the Atlantic Utility Group (determined as set forth in clause (ii) of the definition of Company Net Income (Loss) Attributable to the Atlantic Utility Group, or

(ii) in case the total amount calculated pursuant to clause (i) above is not a positive number, an amount equal to Company Net Income (Loss) Attributable to the Atlantic Utility Group (if positive) for the fiscal year in which the dividend is declared and/or the preceding fiscal year.

Notwithstanding the foregoing provisions of this definition, and consistent with the definition of Convertible Securities, at any time when there are not outstanding both (i) one or more shares of Company Common Stock or Convertible Securities convertible into Company Common Stock and (ii) one or more shares of Class A Common Stock or Convertible Securities convertible into Class A Common Stock, the "Atlantic Utility Group Available Dividend Amount," on any calculation date during such time period, with respect to the Company Common Stock or the Class A Common Stock, as the case may be (depending on which of such classes of Common Stock or Convertible Securities convertible into such class of Common Stock is outstanding), shall mean the amount available for the payment of dividends on such Common Stock in accordance with law.

"Company Net Income (Loss) Attributable to the Atlantic Utility Group," for any period through any date, shall mean (i) the net income or loss of the Atlantic Utility Group for such period (or in respect of the fiscal periods of the Company commencing prior to the Effective Date, the pro forma net income or loss of the Atlantic Utility Group for such period as if the Effective Date had been the first day of such period) determined in accordance with generally accepted accounting principles in effect at such time, reflecting income and expense of the Company attributed to the Atlantic Utility Group on a basis substantially consistent with attributions of income and expense made in the calculation of Company Net Income (Loss) Attributable to the Residual Group, including, without limitation, corporate administrative costs, net interest and other financial costs and income taxes, increased by (ii) the amount of amortization of goodwill arising from the Mergers with respect to the Atlantic Utility Group to the extent but only to the extent such goodwill would otherwise decrease the Company Net Income Attributable to the Atlantic Utility Group or increase the Company Net Loss Attributable to the Atlantic Utility Group (such amount calculated for fiscal periods of the Company commencing prior to the Effective Date on a pro forma basis as if the Effective Date had been the first day of the relevant period), determined in accordance with generally accepted accounting principles in effect at such time applied on a basis substantially consistent with that applied in determining Company Net Income (Loss) Attributable to the Atlantic Utility Group and reduced by (iii) an amount equal to \$40 million per fiscal year.

"Company Net Income (Loss) Attributable to the Residual Group," for any period through any date, shall mean (i) the net income or loss of the Residual Group for such period (or in respect of fiscal periods of the Company commencing prior to the Effective Date, the pro forma net income or loss of the Residual Group for such period as if the Effective Date had been the first day of such period) determined in accordance with generally accepted accounting principles in effect at such time, reflecting income and expense of the Company attributed to the Residual Group on a basis substantially consistent with attributions of income and expense made in the calculation of Company Net Income (Loss) Attributable to the Atlantic Utility Group, including, without limitation, corporate administrative costs, net interest and other financial costs and income taxes, reduced by (ii) the amount of amortization of goodwill arising from the Mergers with respect to the Atlantic Utility Group to the extent but only to the extent such goodwill results in a decrease (increase) in the Company Net Income (Loss) Attributable to the Atlantic Utility Group pursuant to clause (ii) of the definition thereof (such amount calculated for fiscal periods of the Company commencing prior to the Effective Date on a pro forma basis as if the Effective Date had been the first day of the relevant period), determined in accordance with generally accepted accounting principles in effect at such time applied on a basis substantially consistent with that applied in determining Company Net Income (Loss) Attributable to the Atlantic Utility Group (excluding the portion thereof, if any, already applied to reduce net income or increase net loss of the Residual Group for such period by virtue of the Intergroup Interest) and increased by (iii) an amount equal to \$40 million per fiscal year.

"Convertible Securities" at any time shall mean any securities of the Company or of any subsidiary thereof (other than shares of Common Stock), including warrants and options, outstanding at such time that by their terms are convertible into or exchangeable or exercisable for or evidence the right to acquire any shares of any class of Common Stock, whether convertible, exchangeable or exercisable at such time or later time or only upon the occurrence of certain events, but in respect of antidilution provisions of such securities only upon the effectiveness thereof.

"Disposition" shall mean a sale, transfer, assignment or other disposition (whether by merger, consolidation, sale or contribution of assets or stock or otherwise) of properties or assets (including stock, other securities and goodwill).

"Effective Date" shall mean the date on which the Mergers shall become effective.

*"Fair Value"* shall mean, in the case of equity securities or debt securities of a class or series that has previously been Publicly Traded for a period of at least 15 months, the Market Value thereof (if such value, as so defined, can be determined) or, in the case of an equity security or debt security that has not been Publicly Traded for at least such period, shall mean the fair value per share of stock or per other unit of such other security, on a fully distributed basis, as determined by an independent investment banking firm experienced in the valuation of securities selected in good faith by the Company Board, or, if no such investment banking firm is, as determined in the good faith judgment of the Company Board, available to make such determination, in good faith by the Company Board; provided that in the case of property other than securities, the "Fair Value" thereof shall be determined in good faith by the Company Board shall in good faith determine to be appropriate in accordance with good business practice. Any such determination of Fair Value shall be described in a statement filed with the records of the actions of the Company Board.

"Intergroup Interest Fraction" as of any date shall mean a fraction the numerator of which shall be the Number of Shares Issuable with Respect to the Intergroup Interest on such date and the denominator of which shall be the sum of (A) such Number of Shares Issuable with Respect to the Intergroup Interest and (B) the aggregate number of shares of Class A Common Stock outstanding on such date.

. .

91

"Market Capitalization" of any class or series of common stock on any date shall mean the product of (i) the Market Value of one share of such class or series of common stock on such date and (ii) the number of shares of such class or series of common stock outstanding on such date.

"Market Value" of a share of any class or series of capital stock of the Company on any day shall mean the average of the high and low reported sales prices regular way of a share of such class or series on such Trading Day or, in case no such reported sale takes place on such Trading Day, the average of the reported closing bid and asked prices regular way of a share of such class or series on such Trading Day, in either case as reported on the New York Stock Exchange Composite Tape or, if the shares of such class or series are not listed or admitted to trading on such Exchange on such Trading Day, on the principal national securities exchange in the United States on which the shares of such class or series are listed or admitted to trading or, if not listed or admitted to trading on any national securities exchange on such Trading Day, on the NASDAQ National Market or, if the shares of such class or series are not listed or admitted to trading on any national securities exchange or quoted on such National Market System on such Trading Day, the average of the closing bid and asked prices of a share of such class or series in the over-the-counter market on such Trading Day as furnished by any New York Stock Exchange member firm selected from time to time by the Company or, if such closing bid and asked prices are not made available by any such New York Stock Exchange member firm on such Trading Day, the Fair Value of a share of such class or series; provided that, for purposes of determining the market value of a share of any class or series of capital stock for any period, (i) the "Market Value" of a share of capital stock on any day prior to any "ex-dividend" date or any similar date occurring during such period for any dividend or distribution (other than any dividend or distribution contemplated by clause (ii)(B) of this sentence) paid or to be paid with respect to such capital stock shall be reduced by the Fair Value of the per share amount of such dividend or distribution and (ii) the "Market Value" of any share of capital stock on any day prior to (A) the effective date of any subdivision (by stock split or otherwise) or combination (by reverse stock split or otherwise) of outstanding shares of such class or series of capital stock occurring during such period or (B) any "exdividend" date or any similar date occurring during such period for any dividend or distribution with respect to such capital stock to be made in shares of such class or series of capital stock or Convertibl Securities that are convertible, exchangeable or exercisable for such class or series of capital stock shall be appropriately adjusted, as determined by the Company Board, to reflect such subdivision, combination, dividend or distribution.

"Market Value Ratio of the Class A Common Stock to the Company Common Stock" as of any date shall mean the fraction (which may be greater than 1/1), expressed as a decimal (rounded to the nearest five decimal places), of a share of Company Common Stock to be issued in respect of a share of Class A Common Stock upon a conversion of Class A Common Stock into Company Common Stock in accordance with the aforesaid terms, based on the ratio of the market value of a share of Class A Common Stock to the market value of a share of Company Common Stock (or such other common stock) as of such date, determined by the fraction the numerator of which shall be the sum of (A) four times the average Market Value of one share of Class A Common Stock over the period of five consecutive Trading Days ending on such date, (B) three times the average Market Value of one share of Class A Common Stock over the period of five consecutive Trading Days ending on the fifth Trading Day prior to such date, (C) two times the average Market Value of one share of Class A Common Stock over the period of five consecutive Trading Days ending on the tenth Trading Day prior to such date and (D) the average Market Value of one share of Class A Common Stock over the period of five consecutive Trading Days ending on the fifteenth Trading Day prior to such date and the denominator of which shall be the sum of (A) four times the average Market Value of one share of Company Common Stock (or such other common stock) over the period of five consecutive Trading Days ending on such date, (B) three times the average Market Value of one share of Company Common Stock (or such other common stock) over the period of five consecutive Trading Days ending on the fifth Trading Day prior to such date, (C) two times the average Market Value of one share of Company Common Stock (or such other common stock) over the period of five consecutive Trading Days ending on the tenth Trading Day prior to such date and (D) the average Market Value of one share of Company Common Stock (or such other common stock) over the period of five consecutive Trading Days ending on the fifteenth Trading Day prior to such date.

"Net Proceeds" shall mean, as of any date with respect to any Disposition of any of the properties and assets attributed to the Atlantic Utility Group an amount, if any, equal to what remains of the gross proceeds of such Disposition after payment of, or reasonable provision is made as determined by the Company Board for, (A) any taxes payable by the Company (or which would have been payable but for the utilization of tax benefits attributable to the Residual Group) in respect of such Disposition or in respect of any resulting dividend or redemption pursuant to clause (I) under "-Conversion and Redemption of the Class A Common Stock—Dividend, Redemption or Conversion upon Disposition of All or Substantially All Assets of the Atlantic Utility Group," (B) any transaction costs, including, without limitation, any legal, investment banking and accounting fees and expenses, (C) any liabilities (contingent or otherwise) of or attributed to the Atlantic Utility Group, including, without limitation, any liabilities for deferred taxes or any indemnity or guarantee obligations of the Company incurred in connection with the Disposition or otherwise, and any liabilities for future purchase price adjustments and any preferential amounts plus any accumulated and unpaid dividends in respect of Preferred Stock attributed to the Atlantic Utility Group and (D) a capitalized amount (as determined by the Company Board) of the notional obligation of the Atlantic Utility Group to the Residual Group described in clause (iii) of the definition of Company Net Income (Loss) Attributable to the Atlantic Utility Group. For purposes of this definition, any properties and assets attributed to the Atlantic Utility Group remaining after such Disposition shall constitute "reasonable provision" for such amount of taxes, costs, liabilities (contingent or otherwise) and capitalized amount as the Company Board determines can be expected to be supported by such properties and assets.

"Number of Shares Issuable with Respect to the Intergroup Interest" shall as of the Effective Date be the number equal to 7/3 times the number of Class A Common Stock to be issued pursuant to the Mergers, such number to be inserted prior to the Effective Time; provided that such number shall from time to time thereafter be:

(A) adjusted as determined by the Company Board to be appropriate to reflect any subdivision (by stock split or otherwise) or combination (by reverse stock split or otherwise) of the Class A Common Stock or any dividend or other distribution of shares of Class A Common Stock to holders of shares of Class A Common Stock or any reclassification of Class A Common Stock;

(B) decreased (but to not less than zero) by action of the Company Board by (1) the number of shares of Class A Common Stock issued or sold by the Company that, immediately prior to such issuance or sale, were included (as determined by the Company Board pursuant to paragraph (C) of this definition) in the Number of Shares Issuable with Respect to the Intergroup Interest, (2) the number of shares of Class A Common Stock issued upon conversion of Convertible Securities that, immediately prior to the issuance or sale of such Convertible Securities, were included in the Number of Shares Issuable with Respect to the Intergroup Interest, (3) the number of shares of Class A Common Stock issued by the Company as a dividend or other distribution (including in connection with any reclassification or exchange of shares) to holders of Company Common Stock, (4) the number of shares of Class A Common Stock issued upon the conversion of any Convertible Securities issued by the Company as a dividend or other distribution (including in connection with any reclassification or exchange of shares) to holders of Company Common Stock, or (5) the number (rounded, if necessary, to the nearest whole number) equal to the quotient of (a) the aggregate Fair Value as of the date of contribution of properties or assets (including cash) transferred from the Atlantic Utility Group to the Residual Group in consideration for a reduction in the Number of Shares Issuable with Respect to the Intergroup Interest divided by (b) the Market Value of one share of Class A Common Stock as of the date of such transfer; and

(C) increased by (1) the number of outstanding shares of Class A Common Stock repurchased by the Company for consideration that is attributed as provided by the definition of "Residual Group" to

the Residual Group and (2) the number (rounded, if necessary, to the nearest whole number) equal to the quotient of (a) the Fair Value of properties or assets (including cash) theretofore attributed as provided by the definition of "Residual Group" to the Residual Group that are contributed to the Atlantic Utility Group in consideration of an increase in the Number of Shares Issuable with Respect to the Intergroup Interest, divided by (b) the Market Value of one share of Class A Common Stock as of the date of such contribution and (3) the number of shares of Class A Common Stock into or for which Convertible Securities are deemed converted pursuant to the penultimate sentence of the definition of "Residual Group."

"Outstanding Atlantic Utility Fraction," as of any date, means the fraction (which may simplify to 1/1) the numerator of which shall be the number of shares of Class A Common Stock outstanding on such date and the denominator of which shall be the sum of the number of shares of Class A Common Stock outstanding on such date and the Number of Shares Issuable with Respect to the Intergroup Interest on such date.

"Publicly Traded" with respect to any security shall mean (i) registered under Section 12 of the Exchange Act (or any successor provision of law), and (ii) listed for trading on the NYSE or the American Stock Exchange (or any national securities exchange registered under Section 7 of the Exchange Act (or any successor provision of law), that is the successor to either such exchange or quoted in the National Association of Securities Dealers Automation Quotation System (or any successor system).

"Related Business Transaction" means any Disposition of all or substantially all the properties and assets attributed to the Atlantic Utility Group in a transaction or series of related transactions that results in the Company receiving in consideration of such properties and assets primarily equity securities (including, without limitation, capital stock, debt securities convertible into or exchangeable for equity securities or interests in a general or limited partnership or limited liability company, without regard to the voting power or other management or governance rights associated therewith) of any entity which (i) acquires such properties or assets or succeeds (by merger, formation of a joint venture or otherwise) to the business conducted with such properties or assets or controls such acquiror or successor and (ii) i primarily engaged or proposes to engage primarily in one or more businesses similar or complementary to the businesses conducted by the Atlantic Utility Group prior to such Disposition, as determined by the Company Board.

"Residual Group" shall mean, as of any date from and as of the Effective Date:

(A) the interest of the Company or any of its subsidiaries on such date in all of the assets, liabilities and businesses of the Company or any of its subsidiaries (and any successor companies), other than any assets, liabilities and businesses attributed in accordance with the applicable provisions of the Company Charter to the Atlantic Utility Group;

(B) a proportionate undivided interest in each and every business, asset and liability attributed to the Atlantic Utility Group equal to the Intergroup Interest Fraction as of such date;

(C) all properties and assets transferred to the Residual Group from the Atlantic Utility Group (other than pursuant to paragraph (D) of this definition) after the Effective Date pursuant to transactions in the ordinary course of business of both the Residual Group and the Atlantic Utility Group or otherwise as the Company Board may have directed as permitted by the applicable provisions of the Company Charter;

(D) all properties and assets transferred to the Residual Group from the Atlantic Utility Group in connection with a reduction of the Number of Shares Issuable with Respect to the Intergroup Interest; (E) the interest of the Company or any of its subsidiaries in any business or asset acquired and any liabilities assumed by the Company or any of its subsidiaries outside the ordinary course of business and attributed to the Residual Group, as determined by the Company Board; and

(F) from and after the payment date of any dividend or other distribution with respect to shares of Class A Common Stock (other than a dividend or other distribution payable in shares of Class A Common Stock, with respect to which adjustment shall be made as provided in paragraph (A) of the definition of "Number of Shares Issuable with Respect to the Intergroup Interest," or in securities of the Company attributed to the Atlantic Utility Group, for which provision shall be made as set forth in the third to last sentence of this definition), an amount of assets or properties previously attributed to the Atlantic Utility Group of the same kind as were paid in such dividend or other distribution with respect to shares of Class A Common Stock as have a Fair Value on the record date for such dividend or distribution equal to the product of (1) the Fair Value on such record date of the aggregate of such dividend or distribution to holders of shares of Class A Common Stock declared multiplied by (2) a Fraction the numerator of which is equal to the Intergroup Interest Fraction in effect on the record date for such dividend or distribution and the denominator of which is equal to the Outstanding Atlantic Utility Fraction in effect on the record date for such dividend or distribution;

provided that from and after any transfer of any assets or properties from the Residual Group to the Atlantic Utility Group, the Residual Group shall no longer include such assets or properties so transferred (other than as reflected in respect of such a transfer by the Intergroup Interest Fraction, as provided by paragraph (B) of this definition).

If the Company shall pay a dividend or make some other distribution with respect to shares of Class A Common Stock payable in securities of the Company that are attributed to the Atlantic Utility Group (other than Class A Common Stock), the Residual Group shall be deemed to hold an interest in the Atlantic Utility Group equivalent to the number or amount of such securities that is equal to the product of the number or amount of securities so distributed to holders of Class A Common Stock multiplied by he Fraction specified in clause (2) of paragraph (F) of this definition (determined as of the record date for such distribution) and, to the extent interest is or dividends are paid on the securities so distributed, the Residual Group shall include, and there shall be transferred thereto out of the Atlantic Utility Group, a corresponding ratable amount of the kind of assets paid as such interest or dividends as would have been paid in respect of such securities so deemed to be held by the Residual Group if such securities were outstanding. The Company may also, to the extent the securities so paid as a dividend or other distribution to the holders of Class A Common Stock are Convertible Securities and at the time are convertible into shares of Class A Common Stock, treat such Convertible Securities as are so deemed to be held by the Residual Group to be deemed to be converted and shall do so to the extent such Convertible Securities are mandatorily converted (and to the extent the terms of such Convertible Securities require payment of consideration for such conversion the Residual Group shall then no longer include an amount of the kind of properties or assets required to be paid as such consideration for the amount of Convertible Securities deemed converted (and the Atlantic Utility Group shall be attributed such properties or assets)), in which case, from and after such time, the securities into or for which such Convertible Securities so deemed to be held by the Residual Group were so considered converted shall be deemed held by the Residual Group (as provided in clause (3) of paragraph (C) of the definition of "Number of Shares Issuable with Respect to the Intergroup Interest") and such Convertible Securities shall no longer be deemed to be held by the Residual Group.

### "Residual Group Available Dividend Amount," on any date, shall mean either

(i) (x) the amount equal to the fair market value of the total assets attributed to the Residual Group less the total amount of the liabilities attributed to the Residual Group (provided that Preferred Stock shall not be treated as a liability), in each case as of such date and determined on a basis consistent with that applied in determining Company Net Income (Loss) Attributable to the

Residual Group, minus (y) the aggregate par value of, or any greater amount determined to be capital in respect of, all outstanding shares of Company Common Stock and shares of each class or series of Preferred Stock attributed to the Residual Group, minus (z) the amount, as of such date, of amortization of goodwill during the period from the Effective Date through such date arising from the Mergers with respect to the Atlantic Utility Group (determined as set forth in clause (ii) of the definition of "Company Net Income (Loss) Attributable to the Residual Group" and, as in such clause (ii), excluding the portion thereof, if any, already applied to reduce net income or increase net loss of the Residual Group for such period by virtue of its Intergroup Interest), or er dat et

(ii) in case the total amount calculated pursuant to clause (i) above is not a positive number, an amount equal to Company Net Income (Loss) Attributable to the Residual Group (if positive) for the fiscal year in which the dividend is declared and/or the preceding fiscal year.

Notwithstanding the foregoing provisions of this definition, at any time when there are not outstanding both (i) one or more shares of Company Common Stock or Convertible Securities convertible into Company Common Stock and (ii) one or more shares of Class A Common Stock or Convertible Securities convertible into Class A Common Stock, the "Residual Group Available Dividend Amount," on any calculation date during such time period, with respect to the Company Common Stock or the Class A Common Stock, as the case may be (depending on which of such classes of Common Stock or Convertible Securities convertible into such class of Common Stock is outstanding), shall mean the amount available for the payment of dividends on such Common Stock in accordance with law.

"Time-Weighted Market Price" as of any date with respect to any class of Common Stock shall mean an amount equal to (i) the sum of (A) four times the average Market Value of one share of such class of Common Stock over the period of five consecutive Trading Days ending on such date, (B) three times the average Market Value of one share of such class of Common Stock over the period of five consecutive Trading Days ending on the fifth Trading Day prior to such date, (C) two times the average Market Value of one share of such class of Common Stock over the period of five consecutive Trading Days ending on the tenth Trading Day prior to such date and (D) the average Market Value of one share of such class of Common Stock over the period of five consecutive Trading Days ending on the fifteenth Trading Day prior to such date, divided by (ii) ten. · · . 1 4 . . C 1.1.1

"Trading Day" shall mean each weekday other than any day on which the relevant class of common stock of the Company is not traded on any national securities exchange or quoted in the NASDAQ National Market or in the over-the-counter market.

### **Stock Exchange Listing**

It is expected that the Company Common Stock and the Class A Common Stock will be listed on the and the set of the set NYSE. 

and the second second second second second

#### **Anti-Takeover Provisions**

. . .

à"...\*

The Company Charter and the Company Bylaws contain provisions that may have the effect of discouraging persons from acquiring large blocks of capital stock of the Company or of delaying or preventing a change in control of the Company. Material provisions that may have the effect of discouraging persons from acquiring large blocks of capital stock of the Company are (i) provisions providing that the Company Board will be divided into three classes of directors with the term of only the directors in one class expiring each year and permitting removal of directors only for cause and only at a special meeting of stockholders called for that purpose, (ii) authorization for the Company Board to issue Preferred Stock in series and to fix rights and preferences of the series, (iii) provisions limiting the right to call a special meeting of stockholders to the Company Board and limiting the business at a special meeting of stockholders to proposals brought before the meeting by the Company Board, (iv) advance notice

procedures with respect to nominations of directors or proposals other than those adopted or recommended by the Company Board, (v) a provision requiring action by stockholders only at a meeting and (vi) provisions permitting amendment of certain provisions contained in the Company Charter only by the vote of at least 80% of stockholders entitled to vote.

The existence of the Class A Common Stock would present complexities and could in certain circumstances pose obstacles for an acquiring person. For example, it would be necessary for a potential acquiror to take into consideration that holders of the Class A Common Stock might be more or less receptive to the acquiror's proposal, that a tender offer must be structured to take into account different prices at which shares of different classes might be acquired, and the effects of actions the Company might take, such as causing a conversion of the Class A Common Stock. In addition, to the extent the relative market values of the Company Common Stock and the Class A Common Stock fluctuate, a potential acquiror may be able to acquire relatively more or less voting power for the same consideration depending on which class of Common Stock is purchased.

In addition, the DGCL contains business combination and supermajority voting provisions that would be applicable to certain mergers, share exchanges, asset sales and other transactions involving the Company or a subsidiary of the Company and a significant stockholder. These provisions could have the effect of substantially increasing the cost of an acquisition to an acquiror and thereby discouraging any such transaction.

The provisions of the DGCL, the Company Charter and the Company Bylaws and the existence of the Class A Common Stock could, under certain circumstances, prevent stockholders from profiting from an increase in the market value of their shares as a result of a change in control of the Company by delaying or preventing such change in control.

1.1.1



#### COMPARISON OF CORPORATE CHARTERS AND RIGHTS OF SECURITY HOLDERS

### Delmarva

If the Delmarva Merger is consummated, the holders of Delmarva Common Stock will become holders of Company Common Stock and their rights will be governed by the Company Charter, the Company Bylaws, and the DGCL. The material differences between the rights of holders of Company Common Stock and the holders of Delmarva Common Stock are set forth below. This summary of the material differences between rights of stockholders does not purport to be an exhaustive list or a detailed description of the provisions discussed and is qualified in its entirety by reference to the full text of the Company Charter and the Company Bylaws attached as Annexes IV and V, respectively. See also "Available Information."

*Voting Power.* If the Delmarva Merger is approved, based on the number of shares outstanding as of September 30, 1996, holders of Delmarva Common Stock would hold approximately 60.8 million of the approximately 100.3 million shares of Company Common Stock that would have been outstanding at the consummation of the Mergers had the Mergers been consummated on September 30, 1996. Following the Mergers, Delmarva stockholders will therefore not possess the same relative voting power on matters put to a vote to the stockholders of the Company as they possessed prior to the Mergers. Neither the Company Charter nor the Delmarva Charter permits cumulative voting in the election of directors.

*Capitalization.* The Company Charter authorizes the issuance of up to 150 million shares of Company Common Stock, 10 million shares of Class A Common Stock of the Company and 20 million shares of Preferred Stock. The Company Board is authorized to issue Preferred Stock in series and to determine the relative rights and preferences of the shares of any series of Preferred Stock, which may rank prior to shares of Company Common Stock or Class A Common Stock for the payment of dividends or upon the dissolution, liquidation or winding up of the Company.

Special Stockholder Meetings; Stockholder Action Without a Meeting. The Delmarva Bylaws provide that special meetings of stockholders may be called by the Chairman/Chairwoman of the Board, the President or the Board of Directors. The Company Charter provides that special meetings of its stockholders may be called only by the Chairman/Chairwoman of the Board or by a majority of the Company Board.

The VSCA permits actions that could be taken at a stockholder meeting to be taken by the unanimous written consent of the stockholders entitled to vote on such action. The DGCL permits actions that could be taken at a stockholder meeting to be taken by written consent of the holders of outstanding stock having not less than the minimum number of votes that would be necessary to take such action at a meeting at which all shares entitled to vote thereon were present and voted, unless the corporation's certificate of incorporation provides otherwise. The Delmarva Charter permits action by written consent only by the unanimous written consent of stockholders entitled to vote on the particular matter. The Company Charter requires stockholder action to be effected at a meeting of stockholders and prohibits stockholder action by written consent.

*Board of Directors.* The Delmarva Charter provides for a Board of Directors consisting of not fewer than three and not more than 15 directors, the exact number of directors to be determined from time to time by resolution of the Delmarva Board or by resolution of two-thirds of the shares, represented by proxy or in person, entitled to vote at a meeting at which a quorum is present. The Delmarva Charter provides that directors shall be divided into three classes (designated Class I, Class II and Class III) to provide for staggered terms, with each class consisting, as nearly as may be possible, of one-third of the total number of directors. The directors in each class are elected for three year terms.

The Company Charter provides for a board of not fewer than nine and not more than 18 directors, as determined from time to time by resolution of the Board of Directors. The Company Charter provides that the directors shall be divided into three classes (designated Class I, Class II, and Class III) to provide for

staggered terms, with each class consisting, as nearly as possible, of one-third of the total number of directors. The terms of the initial Class I directors will expire on the date of the first annual meeting following the Effective Time of the Mergers, the terms of the initial Class II directors will expire on the date of the second annual meeting following the consummation of the Mergers, and the terms of the initial Class III directors will expire on the date of the second annual meeting following the consummation of the Mergers, and the terms of the initial Class III directors will expire on the date of the third annual meeting following the consummation of the Mergers, and thereafter the directors in each class will be elected for a term of three years.

*Removal of Directors.* Under the VSCA, unless the articles of incorporation provide that directors may be removed only for cause, stockholders may remove any director with or without cause if the number of votes cast to remove a director constitutes a majority of the votes entitled to be cast at an election of directors of the voting group or voting groups by which the director was elected, unless the articles of incorporation require a greater vote. The DGCL provides that unless a corporation's certificate of incorporation otherwise provides, directors with staggered terms may be removed only for cause and requires that the removal of directors be approved by a majority vote of the shares then entitled to vote at an election of directors. The Delmarva Charter provides that directors may be removed only for cause, at an annual meeting or a special meeting of the stockholders called for that purpose, by a majority vote of not less than a majority of votes entitled to be cast by the holders of all the then outstanding shares of capital stock entitled to vote in the election of directors generally.

Vacancies on the Board and Newly Created Directorships. The Delmarva Charter provides that newly created directorships and vacancies in the Delmarva Board may be filled by the Delmarva Board and if the directors remaining in office constitute fewer than a quorum, by a majority of the directors remaining in office, or at an annual meeting of stockholders by stockholders entitled to vote on the election of directors. The Company Charter provides that newly created directorships and vacancies in the Company Board of Directors shall be filled by a majority of the directors then in office, even if less than a quorum, or by the pole remaining director.

Stockholder Proposals; Nomination of Directors by Stockholders. The Delmarva Bylaws provide that any stockholder proposal or nomination for election to the Board of Directors brought before Delmarva's annual meeting must be received by Delmarva not fewer than 50 days and, in the case of a stockholder proposal, not more than 75 days, or, in the case of a nomination for election to the Delmarva Board, not more than 90 days prior to the meeting; provided that in the event that fewer than 65 days' notice or prior public disclosure of the date of the meeting is given or made to stockholders, proposals or nominations by stockholders must be received not later than the close of business on the fifteenth day following the day on which notice of the annual meeting was mailed or such public disclosure was made, whichever first occurs.

The Company Charter provides that any stockholder proposal or nomination for election to the Board brought before the Company's annual meeting must be received by the Company not less than 60 days nor more than 90 days prior to the first anniversary of the preceding year's annual meeting. If the date of such annual meeting is advanced by more than 30 days or delayed by more than 60 days, or in the case of the Company's first annual meeting after the consummation of the Mergers, notice by the stockholder to be timely must be received not earlier than the ninetieth day prior to such annual meeting and not later than the close of business on the later of (i) the sixtieth day prior to such annual meeting or (ii) the tenth day following the date on which notice of the date of the annual meeting was given. The Company Charter also provides that no business may be brought before a special meeting by stockholders.

*Mergers.* Under the VSCA, a plan of merger or share exchange must be approved by more than twothirds of all the votes entitled to be cast thereon by each voting group, provided that the board of directors may require a greater vote and the articles of incorporation may provide for a greater or lesser vote, so long as the vote provided for is not less than a majority of all votes cast by each voting group at a meeting at which a quorum exists. The DGCL provides that a merger or consolidation may be approved by a majority vote of the shares entitled to vote thereon, unless a greater percentage is required by the corporation's certificate of incorporation.

Inspection Rights. Under the VSCA, a stockholder who has been a stockholder of record for at least six months or who is the holder of record of at least five percent of all of a corporation's outstanding shares is entitled to inspect certain records of the corporation if he or she gives the corporation written notice of his or her demand, the demand is made in good faith and for a proper purpose, the demand describes with reasonable particularity the purpose of the demand and the records to be inspected, and the records are directly connected with such purpose. The DGCL provides that any stockholder shall have the right to inspect the corporation's stock ledger, a list of its stockholders and its other books and records upon written demand under oath stating the purpose thereof if the demand for inspection is made for a proper purpose.

Dividends and Repurchases of Stock. The VSCA provides that, subject to any restriction in a corporation's articles of incorporation, a board of directors may authorize and the corporation may pay distributions to the corporation's stockholders, unless after such distribution the corporation would not be able to pay its debts as they come due or unless its total assets after the distribution would be less than the sum of its total liabilities plus (unless the articles of incorporation permits otherwise) the amount that would be needed, if the corporation were to be dissolved at the time of the distribution, to satisfy upon dissolution the preferential rights of stockholders whose preferential rights are superior to those receiving the distribution.

Under the DGCL, the board of directors may declare and pay dividends upon shares of the corporation's capital stock either (1) out of the corporation's surplus (*i.e.*, the excess, if any, of the net assets of the corporation over the capital of the corporation) or (2) in the case there shall be no such surplus, out of its net profits for the fiscal year in which the dividend is declared and/or the preceding fiscal year, so long as the declaration and payment of such dividends do not violate the corporation's certificate of incorporation. Under the DGCL, a corporation generally may repurchase shares of its stock out o surplus so long as such repurchase does not violate the corporation's certificate of incorporation, but the DGCL permits a corporation to repurchase shares out of capital under certain circumstances.

Certain Business Combinations and Share Purchases. The VSCA contains a provision that prohibits a corporation from engaging in an "affiliated transaction" with a 10% stockholder for three years following the acquisition of the 10% stake unless (i) the 10% stockholder became such with the approval of the corporation's disinterested directors, (ii) the transaction is approved by a majority of the corporation's disinterested directors and the holders of two-thirds of the shares not held by the 10% stockholder or (iii) the corporation has elected not to be bound by this provision. The VSCA also contains a "control share" provision whereby the voting rights of a stockholder are eliminated with respect to shares, the acquisition of which causes the percentage of shares beneficially owned by such stockholder to exceed one of several percentage thresholds, unless a resolution granting voting power is approved by a majority of all votes that could be cast in an election of directors, excluding certain shares including those of the acquiring stockholder, or unless the corporation has elected not to be bound by this provision. Delmarva pursuant to the VSCA has elected to opt out of this provision.

Under Section 203 of the DGCL, a corporation such as the Company may not engage in any business combination (defined to include mergers, share exchanges, and asset sales) with any interested stockholder (defined as a 15% stockholder) for a period of three years following the date that such stockholder became an interested stockholder, unless (i) prior to the time that such person became an interested stockholder, the board of directors of the corporation approved either the business combination or the transaction in which the stockholder became an interested stockholder, the interested stockholder became an interested stockholder, (ii) upon consummation of the transaction by which the stockholder became an interested stockholder, the interested stockholder owned at least 85% of the voting stock of the corporation (excluding certain shares owned by officers and directors of the corporation and by employee stock plans) or (iii) the business combination is approved by both the board

of directors of the corporation and the affirmative vote of at least two-thirds of the outstanding voting stock of the corporation that is not owned by the interested stockholder.

Amendment to Articles or Certificate of Incorporation. The VSCA provides that a board of directors may condition its submission of a proposed charter amendment to the stockholders on any basis. To be adopted, the amendment must be approved by a more than two-thirds vote of each voting group entitled to vote thereon, provided that the directors may require a greater vote or the articles of incorporation may provide for a greater or lesser vote, so long as the vote provided for is not less than a majority of those votes cast at a meeting at which a quorum exists. The DGCL provides that the certificate of incorporation of a corporation may be amended upon an affirmative vote of a majority of the outstanding shares of capital stock entitled to vote thereon and a majority of the outstanding stock of each class entitled to vote thereon as a class, unless the corporation's certificate of incorporation requires a larger percentage. Under the Delmarva Charter, the vote of the holders of two-thirds of the total voting power of the Delmarva Preferred Stock and Preferred Stock—\$25 Par of all series voting as a single class is required for certain amendments to the Delmarva Charter. Although the Company Charter provides generally that amendments to the Company Charter may be approved by a majority vote of the aggregate vote that the holders of the Company's capital stock are entitled to cast on the amendment, a vote of at least 80% of the aggregate vote that the holders of the Company's capital stock are entitled to cast on the amendment is required for: (i) amendments to the provisions authorizing the Company Board to issue and fix the rights of Preferred Stock; (ii) amendments to the provisions dividing the Company Board into classes or providing that directors may be removed only for cause; (iii) amendments to the provisions limiting who may call special stockholder meetings, and the ability of stockholders to act without a meeting and providing time requirements for the submission of stockholders' proposals; and (iv) amendments that modify these supermajority voting requirements.

Adoption, Amendment and Repeal of Bylaws. The VSCA provides that a corporation's board of directors may amend or repeal bylaws except to the extent that (i) such right is reserved exclusively to the tockholders in the articles of incorporation or by statute or (ii) the stockholders, in adopting or amending a particular bylaw, provide expressly that the board of directors may not amend or repeal that bylaw. Under the VSCA, stockholders may amend or repeal bylaws even though the bylaws also may be amended or repealed by the board of directors. The DGCL provides that the power to adopt, amend, and repeal bylaws shall be in the stockholders entitled to vote; provided that any corporation may in its certificate of incorporation, confer upon the directors the nonexclusive power to adopt, amend, and repeal bylaws; provided that any bylaw made by the Delmarva Board may be altered, amended or repealed by the stockholders entitled to vote thereon at any annual or special meeting called for such purpose. The Company Charter authorizes the Company Board to make, alter, amend or repeal the Company Bylaws. The Company Charter also provides that stockholders shall have the power to amend the Company Bylaws only by the affirmative vote of 80% or more of the aggregate number of votes that the holders of the then outstanding shares of Common Stock and Preferred Stock are entitled to cast on the amendment:

Exculpation and Indemnification of Directors. Under the VSCA, in any proceeding brought by or in the name of the corporation or by or on behalf of stockholders, the damages assessed against an officer or director arising out of a single transaction, occurrence or course of conduct shall not exceed the lesser of (i) the amount, including the elimination of liability, specified in the corporation's articles of incorporation, or (if approved by the stockholders) in the bylaws, as a limitation on or elimination of the director's or officer's liability or (ii) the greater of \$100,000 or the compensation received by the director or officer from the corporation during the previous 12 months, provided that the liability of an officer or director cannot be limited if the officer or director engaged in willful misconduct or a knowing violation of a criminal law or securities law. Under the DGCL, the certificate of incorporation may include a provision eliminating or limiting the personal liability of directors for monetary damages for breach of fiduciary duty as a director, except for breach of the duty of loyalty, for acts not in good faith or involving intentional misconduct or a

knowing violation of law, for acts from which the director derived an improper personal benefit, or for the unlawful declaration of dividends or the unlawful repurchase or redemption of capital stock. The Delmarva Charter provides for the elimination of the personal liability of directors for monetary damages for breach of fiduciary duty, except for the acts enumerated in the DGCL for which no such elimination of liability is permitted. The Company Charter also provides for the elimination of the personal liability of directors for monetary damages for breach of fiduciary duty, except for such acts enumerated in the DGCL for which no such elimination of liability is permitted.

The VSCA provides that a corporation may indemnify against certain liabilities and expenses a director, officer, employee or agent of the corporation who is made a party to a proceeding by reason of his or her service in such capacity, if the director, officer, agent or employee acted in good faith and (i) he or she believed, in the case of conduct undertaken in his or her official capacity, that such conduct was in the best interests of the corporation or, in the case of other conduct, that the other conduct was not opposed to the best interests of the corporation and (ii) in the case of a criminal proceeding, he or she had no reasonable cause to believe his or her conduct was unlawful. However, under the VSCA, a corporation may not indemnify a director, officer, agent or employee if such person was found liable to a corporation in a proceeding by or in the right of the corporation or for receiving an improper personal benefit, unless the articles of incorporation or a stockholder approved bylaw provide otherwise. The VSCA permits stockholders to authorize further indemnity except with respect to willful misconduct or a knowing violation of criminal law. The Delmarva Charter provides for such indemnity. Unless limited by its articles of incorporation, the VSCA requires a corporation to indemnify a director or officer who is made a party to a proceeding by reason of his or her service in such capacity and who entirely prevails in defense of such proceeding, or if so ordered by a court. The VSCA permits a corporation to advance expenses to a director, officer, employee or agent of the corporation under certain conditions.

The DGCL provides that a corporation may indemnify against certain liabilities and expenses an officer, director, employee or agent of the corporation, or a person serving at the request of the corporation as a director, officer, employee or agent of another entity, who is made a party to certain proceedings by reason of his or her service in such capacity if such person acted in good faith and in a manner he or she reasonably believed to be in or not opposed to the best interests of the corporation and, with respect to any criminal proceeding, had no reasonable cause to believe his or her conduct was unlawful. However, under the DGCL, a corporation may not indemnify any person with respect to any claim or issue as to which such person was found liable to the corporation in a proceeding by or in the right of the corporation, unless indemnification of expenses is ordered by a court. The DGCL provides that a corporation must indemnify against reasonable expenses a director, officer, employee or agent of the corporation who is made a party to any proceeding by reason of his or her service in such capacity and who is successful, on the merits or otherwise, in the defense of any claim, issue or matter therein. The DGCL period is successful, on the merits or otherwise, to a director or officer under certain conditions.

The Delmarva Charter provides that Delmarva shall indemnify directors and officers of Delmarva to the full extent permitted by applicable law in connection with certain actions, suits or proceedings, and provides that Delmarva may indemnify employees and agents of Delmarva, as well as persons serving at the request of Delmarva as directors, officers, employees or agents of any other entity, to the full extent permitted by applicable law in connection with certain actions, suits or proceedings. The Company Charter provides that the Company shall indemnify directors or officers of the Company, as well as persons serving at the request of the Company as directors, officers, agents or employees of any other entity, to the full extent permitted by applicable law in connection with certain actions, suits or proceedings, and that the Company may, by action of the Company Board, indemnify employees or agents of the Company to the full extent permitted by applicable law in connection with certain actions, suits or proceedings. The Company may, by action of the Company Board, indemnify employees or agents of the Company to the full extent permitted by applicable law in connection with certain actions, suits or proceedings. The Company Charter also requires the Company to advance expenses to directors and officers, as well as persons serving at the request of the Company as directors, officers, agents or employees of any other entity, under certain circumstances.

### Atlantic

The rights of holders of Atlantic Common Stock are currently governed by the Restated Certificate of Incorporation of Atlantic (the "Atlantic Charter"), the Atlantic Bylaws and the NJBCA. If the Atlantic Merger is consummated, the holders of Atlantic Common Stock will become holders of Company Common Stock and Class A Common Stock, and their rights will be governed by the Company Charter, the Company Bylaws and the DGCL. The material differences between the rights of holders of Company Common Stock and the holders of Atlantic Common Stock are set forth below. This summary of the material differences between rights of stockholders does not purport to be an exhaustive list or a detailed description of the provisions discussed and is qualified in its entirety by reference to the full text of the Company Charter and the Company Bylaws attached as Annexes IV and V, respectively, to this Joint Proxy Statement/Prospectus. See also "Available Information." For a discussion of the rights of holders of Class A Common Stock, see "Description of the Company's Capital Stock" and "Risk Factors."

*Voting Power.* If the Atlantic Merger is approved, based on the number of shares outstanding as of September 30, 1996, holders of Atlantic Common Stock would hold approximately 39.5 million of the approximately 100.3 million shares of Company Common Stock that would have been outstanding at the consummation the Mergers had the Mergers been consummated on September 30, 1996. Following the Mergers, Atlantic stockholders will therefore not possess the same relative voting power on matters put to a vote to the stockholders of the Company as they possessed prior to the Atlantic Merger. Neither the Company Charter nor the Atlantic Charter permit cumulative voting in the election of directors.

*Capitalization.* The Company Charter authorizes the issuance of up to 150 million shares of Company Common Stock, 10 million shares of Class A Common Stock and 20 million shares of Company Preferred. The Company Board is authorized to issue Preferred Stock in series and to determine the relative rights and preferences of the shares of any series of Preferred Stock, which may rank prior to shares of Company Common Stock or Class A Common Stock for the payment of dividends or upon the dissolution, liquidation or winding up of the Company.

Special Stockholder Meetings; Stockholder Action Without a Meeting. The Atlantic Bylaws provide that special meetings of stockholders may be called by the Chairman of the Board, the President or the Secretary of Atlantic, by a majority of the Atlantic Board. Additionally, under the NJBCA, such a special meeting may be called by the holders of not less than 10% of all the outstanding shares of Atlantic capital stock entitled to vote at the meeting. The Company Charter provides that special meetings of its stockholders may be called only by the Chairman of the Board or by a majority of the Company Board. The DGCL does not permit a special meeting to be called by stockholders in the absence of a charter or bylaw provision granting such right.

Both the NJBCA and the DGCL permit actions that could be taken at a stockholder meeting to be taken by written action of the stockholders unless the corporation's charter requires that a meeting be held. The DGCL permits action that could be taken at a stockholder meeting to be taken by written consent of the holders of outstanding stock having not less than the minimum number of votes that would be necessary to take such action at a meeting at which all shares entitled to vote thereon were present and voted, unless the corporation's certificate of incorporation provides otherwise. The Atlantic Charter does not restrict the ability of Atlantic stockholders to act without a meeting. The Company Charter requires stockholder action to be effected at a meeting of stockholders and prohibits stockholder action by written consent.

*Board of Directors.* The Atlantic Bylaws provide for a Board of Directors consisting of not fewer than three nor more than 12 members, as determined from time to time by the Atlantic Board. The Atlantic Charter and the Atlantic Bylaws do not provide for staggered terms of directors.

The Company Charter provides for a board of not fewer than nine and not more than 18 directors, as determined from time to time by resolution of the Board of Directors. The Company Charter provides that

directors shall be divided into three classes (designated Class I, Class II and Class III) to provide for staggered terms, with each class consisting, as nearly as possible, of one-third of the total number of directors. The terms of the initial Class I directors will expire on the date of the first annual meeting following the consummation of the Mergers, the terms of the initial Class II directors will expire on the date of the terms of the initial Class III directors will expire on the date of the second annual meeting following the consummation of the Mergers, and the terms of the initial Class III directors will expire on the date of the third annual meeting following the consummation of the Mergers, and the terms of the initial Class III directors will expire on the date of the third annual meeting following the consummation of the Mergers, and thereafter the directors in each class will be elected for a term of three years.

*Removal of Directors.* The NJBCA provides for the removal of directors, with cause or unless otherwise provided in the certificate of incorporation without cause, by a vote of a majority of the shares then entitled to vote. The Atlantic Charter and Atlantic Bylaws are silent as to the issue of removal of directors.

The DGCL provides that unless a corporation's certificate of incorporation otherwise provides, directors with staggered terms may be removed only for cause and requires that the removal of directors be approved by a majority vote of the shares then entitled to vote at an election of directors. The Company Charter provides that a member of the Company Board may be removed only for cause, at an annual meeting or a special meeting of stockholders called for that purpose, by a majority vote of not less than a majority of votes entitled to be cast by the holders of all the then outstanding shares of capital stock entitled to vote in the election of directors generally.

Vacancies on the Board and Newly Created Directorships. The Atlantic Bylaws provide that newly created directorships and vacancies in the Atlantic Board may be filled by the Atlantic Board, provided that if only two directors remain, those two, or if only one director remains, that one, may elect directors so that there are at least three directors holding office until the next annual meeting of stockholders. The Company Charter provides that newly created directorships and vacancies in the Company Board of Directors shall be filled by a majority of the directors then in office, even if less than a quorum, or by the sole remaining director.

Stockholder Proposals; Nomination of Directors by Stockholders. The Company Charter provides that any stockholder proposal or nomination for election to the Board brought before the Company's annual meeting must be received by the Company not less than 60 days nor more than 90 days prior to the first anniversary of the preceding year's annual meeting. If the date of such annual meeting is advanced by more than 30 days or delayed by more than 60 days, or in the case of the Company's first annual meeting after the consummation of the Mergers, notice by the stockholder, to be timely, must be received not earlier than the ninetieth day prior to such annual meeting and not later than the close of business on the later of (i) the sixtieth day prior to such annual meeting or (ii) the tenth day following the date on which notice of the date of the annual meeting was given. The Company Charter also provides that no business may be brought before a special meeting by stockholders.

The Atlantic Bylaws require that stockholder proposals or nominations be in writing and be received not later than 90 days prior to the first anniversary of the preceding year's annual meeting. If the date of such annual meeting is advanced by more than 30 days or delayed by more than 60 days, the nomination must be received not later than the close of business on seventh day following the date on which notice of the date of the annual meeting was given. In the case of a stockholder who wishes to nominate a person for election as a director pursuant to consents in writing by stockholders without a meeting, such nomination must be received by Atlantic at least 60 days before materials soliciting such consents are first mailed to stockholders, or, if no such materials are required under applicable law, 60 days before the first such consent is executed.

*Mergers.* In the case of corporations incorporated after January 1, 1969, such as Atlantic, the NJBCA requires the vote of a majority of the votes cast by the stockholders entitled to vote to approve a merger or share exchange involving the corporation. The DGCL provides that a merger or consolidation may be

approved by a majority vote of the shares entitled to vote thereon, unless a greater percentage is required by the corporation's certificate of incorporation.

Inspection Rights. The NJBCA provides that the record of stockholders may be inspected upon five days' written notice by any stockholder that has been a stockholder for six months or any 5% stockholder or group of stockholders owning 5% of the outstanding shares in the aggregate, and only if the demand for inspection is made for a proper purpose. The DGCL provides that any stockholder shall have the right to inspect the corporation's stock ledger, a list of its stockholders and its other books and records upon written demand under oath stating the purpose thereof if the demand for inspection is made for a proper purpose.

Dividends and Repurchases of Stock. Under the NJBCA and subject to any restrictions in its articles of incorporation, a corporation generally may declare and pay dividends and redeem or repurchase shares of its common stock unless, after giving effect to such dividend or repurchase, the corporation is unable to pay its debts as they become due in the usual course of business or the corporation's assets are less than the sum of its total liabilities. Under the DGCL, the board of directors may declare and pay dividends upon shares of the corporation's capital stock either (1) out of the corporation's surplus (i.e., the excess, if any, of the net assets of the corporation over the capital of the corporation) or (2) in the case there shall be no such surplus, out of its net profits for the fiscal year in which the dividend is declared and/or the preceding fiscal year, so long as the declaration and payment of such dividends do not violate the corporation's certificate of incorporation. Under the DGCL, a corporation may repurchase shares of its stock so long as such repurchase does not violate the corporation's certificate of incorporation and so long as such purchase is not made when the capital of the corporation is impaired or if such repurchase would cause any impairment of capital, except that a corporation may repurchase shares out of capital under certain circumstances.

Certain Business Combinations and Share Purchases. Under Section 203 of the DGCL, a corporation such as the Company may not engage in any business combination (defined to include mergers, share exchanges and asset sales) with any interested stockholder (defined as a 15% stockholder) for a period of three years following the date that such stockholder became an interested stockholder, unless (i) prior to the time that such person became an interested Stockholder, the board of directors of the corporation approved either the business combination or the transaction in which the stockholder became an interested stockholder, (ii) upon consummation of the transaction by which the stockholder became an interested stockholder, the interested stockholder owned at least 85% of the voting stock of the corporation (excluding certain shares owned by officers and directors of the corporation and employee stock plans) or (iii) the business combination is approved by both the board of directors of the corporation and the affirmative vote of at least two-thirds of the outstanding voting stock of the corporation that is not owned by the interested stockholder.

The NJBCA contains a similar provision, known as the New Jersey Shareholders Protection Act (Sections 14A:10A-1 through 14A:10A-6 of the NJBCA). Under this statute, a resident domestic corporation, such as Atlantic, may not engage in any business combination (defined similarly to the Delaware provision) with any interested stockholder (a 10% stockholder) for a period of five years following the date that such stockholder became an interested stockholder, unless (i) the board of directors of the corporation approved the business combination prior to the date that the interested stockholder became such, (ii) two-thirds of the corporation's voting stock not owned by the interested stockholder approves the combination or (iii) the combination satisfies certain "fair-price" conditions, relating to the price at which the interested stockholder will purchase shares from other stockholders.

Amendment to Articles or Certificate of Incorporation. The NJBCA requires a majority vote to amend the articles of incorporation of a corporation such as Atlantic. The Atlantic Charter does not contain any provisions relating to its amendment. The DGCL provides that the certificate of incorporation of a corporation may be amended upon an affirmative vote of a majority of the outstanding shares of capital stock entitled to vote thereon and a majority of the outstanding stock of each class entitled to vote as a class thereon, unless the corporation's charter requires a larger percentage. Although the Company Charter provides generally that amendments to the Company Charter may be approved by a majority vote of the Company's stockholders, an 80% majority is required for: (i) amendments to the provisions authorizing the Company Board to issue and fix the rights of Preferred Stock; (ii) amendments to the provisions dividing the Company Board into classes or providing that directors may be removed only for cause; (iii) amendments to the provisions limiting who may call special stockholder meetings, prohibiting the stockholders from acting without a meeting, or fixing the time requirements for submitting stockholder proposals; and (iv) amendments that modify these supermajority voting requirements.

Adoption, Amendment and Repeal of Bylaws. The NJBCA provides that a corporation's board of directors shall have the power to make, alter and repeal bylaws unless such power is reserved to the stockholders in the certificate of incorporation. Bylaws made by the board may be altered and repealed, and new bylaws made, by the stockholders. The stockholders may prescribe in the bylaws that any bylaws made by them may not be altered or repealed by the board.

The DGCL provides that the power to adopt, amend and repeal bylaws shall be in the stockholders entitled to vote; provided that any corporation may in its certificate of incorporation, confer upon the directors the nonexclusive power to adopt, amend and repeal bylaws.

The Atlantic Charter is silent on the adoption, amendment and repeal of bylaws. The Atlantic Bylaws provide that they may be amended or added to by majority vote of all directors, if (i) notice of the proposed change has been sent to all the directors ten days before the meeting called to consider such change, (ii) all the directors are present at such meeting or (iii) those not present at such meeting assent in writing to such change.

The Company Charter authorizes the Company Board to make, alter, amend, or repeal the Company Bylaws. The Company Charter also provides that stockholders shall have the power to amend the Company Bylaws only by the affirmative vote of 80% or more of the aggregate number of votes that the holders of the then outstanding shares of Common Stock and Preferred Stock are entitled to cast on the amendment.

Exculpation and Indemnification of Directors. Under the NJBCA, the certificate of incorporation may provide that a director or officer shall not be personally liable, or shall be liable only to the extent therein provided, to the corporation or its stockholders for damages for breach of any duty owed to the corporation or its stockholders, except for breach of duty based upon an act or omission (a) in breach of such person's duty of loyalty to the corporation or its stockholders, (b) not in good faith or involving a knowing violation of law or (c) resulting in receipt by such person of an improper personal benefit. Under the DGCL, the certificate of incorporation may include a provision eliminating or limiting the personal liability of directors for monetary damages for breach of fiduciary duty as a director, except for breach of the duty of loyalty, for acts not in good faith or involving intentional misconduct or a knowing violation of law, for acts from which the director derived an improper personal benefit, or for the unlawful declaration of dividends or the unlawful repurchase or redemption of capital stock. The Atlantic Charter provides that directors and officers of Atlantic shall not be liable to Atlantic or its stockholders for breach of fiduciary duty as directors or officers, except for acts not in good faith or for personal benefit. The Company Charter also provides for the elimination of the personal liability of directors for monetary damages for breach of fiduciary duty, except for such acts enumerated in the DGCL for which no such elimination of liability is permitted.

The NJBCA provides that a corporation may indemnify against certain liabilities and expenses a director, officer, employee or agent of the corporation who is made a party to a proceeding by reason of his or her service in such capacity, (i) if the director, officer, agent or employee acted in good faith and in a manner he or she reasonably believed to be in or not opposed to the best interests of the corporation and (ii) in the case of a criminal proceeding, he or she had no reasonable cause to believe his or her conduct

was unlawful. However, under the NJBCA, a corporation may not indemnify a director, officer, agent or employee if such person was found liable to the corporation in a proceeding by or in the right of the corporation unless, upon proper application, the Superior Court finds such indemnification proper. The NJBCA requires a corporation to indemnify a director, officer, employee or agent who is made a party to a proceeding by reason of his or her service in such capacity to the extent that such individual has been successful in such proceeding. The NJBCA permits a corporation to advance expenses to a director, officer, employee or agent of the corporation under certain conditions.

The DGCL provides that a corporation may indemnify against certain liabilities and expenses an officer, director, employee or agent of the corporation, or a person serving at the request of the corporation as a director, officer, employee or agent of another entity, who is made a party to certain proceedings by reason of his or her service in such capacity if such person acted in good faith and in a manner he or she reasonably believed to be in or not opposed to the best interests of the corporation and, with respect to any criminal proceeding, had no reasonable cause to believe his or her conduct was unlawful. However, under the DGCL, a corporation may not indemnify any person with respect to any claim or issue as to which such person was found liable to the corporation in a proceeding by or in the right of the corporation, unless indemnification of expenses is ordered by a court. The DGCL provides that a corporation must indemnify against reasonable expenses a director, officer, employee or agent of the corporation who is made a party to any proceeding by reason of his or her service in such capacity and who is successful, on the merits or otherwise, in the defense of any claim, issue or matter therein. The DGCL permits a corporation to advance expenses to a director or officer under certain conditions.

The Atlantic Bylaws provide that Atlantic shall indemnify directors, officers and employees of Atlantic to the full extent permitted by applicable law in connection with certain actions, suits or proceedings, as well as persons serving at the request of Atlantic as directors, officers or employees of any other entity, to the full extent permitted by applicable law in connection with certain actions, suits, or proceedings. The Company Charter provides that the Company shall indemnify directors or officers of the Company, as well as persons serving at the request of the Company as directors, officers, agents or employees of any other entity, to the full extent permitted by applicable law in connection with certain actions, suits or proceedings, and that the Company may, by action of the Company Board, indemnify employees or agents of the Company to the full extent permitted by applicable law in connection with certain actions, suits, or proceedings. The Company Charter also requires the Company to advance expenses to directors and officers, as well as persons serving at the request of the Company as directors, officers, officers, agents or employees of any other employees of any other entity, under certain circumstances.

S. AN

107

· ...

1 D.

12.545 A. 1

### APPROVAL OF THE COMPANY INCENTIVE COMPENSATION PLAN

Subsequent to the execution of the Merger Agreement, Delmarva and Atlantic agreed that the Company would adopt a cash and stock incentive compensation plan to replace the Delmarva LTIP and Atlantic's Equity Incentive Plan, subject to approval by stockholders. Accordingly, the Company Plan is submitted to stockholders of the Delmarva Common Stock and the Atlantic Common Stock for approval, as more fully described below. Conditioned upon the Mergers becoming effective, the Company Plan will become effective only if approved by stockholders as described below, in which event it will become effective at the consummation of the Mergers and will terminate 10 years thereafter.

The affirmative vote of the holders of a majority of Delmarva Common Stock present in person or by proxy and entitled to vote at the Delmarva Meeting and the affirmative vote of at least a majority of all votes cast by all holders of Atlantic Common Stock entitled to vote, respectively, is required to approve the Company Plan. This stockholder approval is intended to meet the requirements of Sections 162(m) and 422 of the Code and the listing guidelines of the NYSE.

## THE BOARD OF DIRECTORS OF EACH OF DELMARVA, ATLANTIC AND THE COMPANY UNANIMOUSLY RECOMMENDS A VOTE FOR APPROVAL OF THE COMPANY PLAN.

Set forth below is a summary of certain important features of the Company Plan, which summary is qualified in its entirety by reference to the actual Company Plan document attached as Annex VII to this Joint Proxy Statement/Prospectus:

and the second second

and the second second second second

101 i.

 $(A_{i})_{ij} = A_{ij} = \frac{1}{2} \left( \frac{1}{2} \left( \frac{1}{2} \right) + \frac{1}{2} \left( \frac{1}{2} \right) \right) \left( \frac{1}{2} \left( \frac{1}{2} \right) + \frac{1}{2} \left( \frac{1}{2} \right) \right)$ 

8 . . . . . <sup>2</sup>

### Objective

The objective of the Company Plan is to support the financial performance of the Company and thereby increase stockholder value by providing annual and long-term incentives to reward officers, key employees, advisors and consultants of the Company and its subsidiaries for the continued growth, development and financial success of the Company and its subsidiaries, and to increase the ownership of Company Common Stock by such persons. The Company Plan also is designed to permit the Company and its subsidiaries to hire and retain talented and motivated executive officers and key employees

# Shares Available Under the Plan

The number of shares of Company Common Stock that may be granted to participants under the proposed Company Plan is 5,000,000. If an award lapses or the participant's rights with respect to such award otherwise terminate, any shares of Company Common Stock subject to such award will be available again for future awards under the Company Plan.

#### Termination.

The Company Plan will continue in effect until all matters relating to the payment of outstanding awards thereunder and administration thereof have been settled.

#### Administration

The Company Plan will be administered by a committee of the Company Board (the "Committee"), which shall be comprised solely of Non-Employee Directors (as such term is defined in Rule 16b-3(b)(3) under Section 16 of the Exchange Act), or which otherwise shall meet any disinterested administration or other requirements of Section 16b-3 and/or Section 162(m) of the Code, each as in effect at the applicable time. The Company Plan gives the Committee broad authority to determine the persons to whom, and the times at which, awards will be granted or lapse under the Company Plan, the types of awards to be granted, the number of shares of Company Common Stock to be covered by each award, and all other terms and conditions for awards granted under the Company Plan.

#### Participation

Each officer, key employee (whether on a regular, temporary or contract basis), consultant or advisor of the Company or its subsidiaries who, in the sole discretion of the Committee, is in a position in which his or her decisions, actions and counsel significantly affect the growth, development and financial success of the Company is an Eligible Person (as defined in the Company Plan) able to participate in the Company Plan. Nonemployee directors will not be eligible to participate in the Company Plan. Because the Company Plan includes annual as well as long-term incentives, the number of Eligible Persons will be greater than it would be in a long-term plan only. The total number of Eligible Persons will be approximately 150, including approximately 25 officers.

### Section 162(m) Compliance

Notwithstanding any other provisions in the Company Plan to the contrary, with respect to any award which is contingent upon the attainment of performance objectives, including, without limitation, performance-based restricted stock, performance units and annual incentives ("Performance-Based Awards"), granted under the Company Plan to an executive of the Company who, in the opinion of the Committee, for a given Performance Period (as defined below) is or is likely to be a "covered employee" within the meaning of Section 162(m) of the Code (a "Covered Executive"), the Committee will establish performance objectives ("Performance Goals") with respect to such Performance-Based Awards no later than the earlier of (i) 90 days after commencement of the period upon which performance will be measured (a "Performance Period") relating to the Performance-Based Award, or (ii) the date on which 25% of the Performance Period relating to the Performance-Based Award will have elapsed.

Performance Goals, in the sole discretion of the Committee, may be based on one or more business criteria that relate to the individual, groups of individuals, a product or service line, business unit, division or subsidiary of the Company or the Company as a whole, individually or in any combination (each of which business criteria may be relative to a specified goal, to historical performance of the Company or a product or service line, business unit, division or subsidiary thereof, or to the performance of any other corporation or group of corporations or a product or service line, business unit, division or subsidiary thereof). Performance Goals will be based on one or more of the following criteria: (a) gross, operating or net earnings before or after income taxes: (b) earnings per share; (c) book value per share; (d) cash flow per share; (e) return on equity; (f) return on investment; (g) return on assets, employed assets or net assets; (h) total stockholder return (expressed on a dollar or percentage basis); (i) return on cash flow; (j) internal rate of return; (k) cash flow return on investment; (l) improvements in capital structure; (m) residual income; (n) gross income, profitability or net income; (o) price of any Company security; (p) sales to customers (expressed on a dollar or percentage basis); (q) retention of customers (expressed on a dollar or percentage basis); (r) increase in the Company's or a subsidiary's customer satisfaction ratings (based on a survey conducted by an independent third party); (s) economic value added (defined to mean net operating profit minus the cost of capital); (t) market value added (defined to mean the difference between the market value of debt and equity, and economic book value); (u) market share; (v) level of expenses; (w) combined ratio; (x) payback period on investment; and (y) net present value of investment.

The Committee will certify the satisfaction of the foregoing Performance Goals prior to the payment of a Performance-Based Award. No Performance-Based Award with respect to any Covered Executive shall exceed \$3,000,000 (either in cash or in the fair market value of stock as determined on the date of grant of the Performance-Based Award, as appropriate to a given type of award) for any Performance Period. The Committee, in its sole discretion, may reduce (but not increase) the amount of any Performance-Based Award determined to be payable to a Covered Executive. No Covered Executive may receive more than 5,000,000 in the aggregate of options, SARs and shares of performance-based restricted stock for the 10-year period during which awards may be made under the Company Plan: Regardless of whether provided for in or in conjunction with the grant of an award, the Committee, in its sole discretion, may defer payment of a participant's benefit under the Company Plan if and to the extent that the sum of the participant's Company Plan benefit, plus all other compensation paid or payable to the Participant for the fiscal year in which the Company Plan benefit would otherwise be paid exceeds the maximum amount of compensation that the Company may deduct under Section 162(m) of the Code with respect to the Participant for the year. If deferred by the Committee, such award benefit will be paid in the first fiscal year of the Company in which the sum of the participant's Company Plan benefit and all other compensation paid or payable to the Participant does not exceed the maximum amount of compensation deductible by the Company under Section 162(m) of the Code.

#### Awards

Under the Company Plan, in addition to any other types of awards as the Committee deems appropriate, the following types of awards may be granted from time to time by the Committee:

*Restricted Stock.* The Committee may grant awards of Company Common Stock bearing restrictions ("Company Restricted Stock") prohibiting a participant's transfer of the Company Restricted Stock until the lapse of a restriction period. No consideration is payable by the participant as a result of the grant. The Committee may establish the terms and conditions of each grant, including the restriction period (which will be not less than one and not more than 10 years), whether dividends will be paid currently or accumulated and the form of any dividend payment, and may also condition the awards on the completion of a specified period of service or on attainment, during a performance period established by the Committee, of one or more performance objectives established by the Committee and may include, but are not limited to, the performance of the Company and/or the participant individually or collectively with any other individuals or groups, or any combination thereof. On completion of the restriction period and attainment of any performance objectives, the restrictions will expire with respect to one or more shares of Company Restricted Stock. If target performance objectives are exceeded, the Committee may award additional Company Common Stock to a participant.

*Options.* The Committee may grant either incentive stock options ("ISO"s) that are qualified under Section 422 of the Code (which may be granted only to persons employed by the Company or a subsidiary of the Company on a regular basis) or options not intended to qualify under Section 422 of the Code ("Nonqualified Options"). No consideration is payable by the participant as a result of the grant. The Committee may establish the terms and conditions of each grant; provided that an option to purchase a share of Company Common Stock may not be granted with an exercise price of less than 100% of the fair market value of a share of Company Common Stock on such date. Further, the period during which the options are exercisable will not exceed 10 years from the date of grant. In the Committee's discretion, the exercise price may be paid in cash, shares of Company Common Stock, or both.

Performance Units. The Committee may make performance awards payable in cash, Company Common Stock or both, upon attainment during a performance period established by the Committee, of one or more performance objectives established by the Committee. Performance objectives, which may vary from participant to participant, are determined by the Committee and may include, but are not limited to, the performance of the Company and/or the participant individually or collectively with any other individuals or groups, or any combination thereof.

Stock Appreciation Rights. The Committee may grant awards of stock appreciation rights ("SAR"s) in conjunction with an option or as a separate award. No consideration is payable by the participant as a result of the grant. The Committee may establish the terms and conditions of each grant; provided that the period during which the rights are exercisable will not exceed 10 years.

SARs provide the right to receive a payment in cash, Company Common Stock, or both in the Committee's discretion. If a grant is in conjunction with an option, the option must be surrendered, and the amount of the payment will be determined based on the excess of the fair market value of the Company Common Stock at the date of exercise over the option price at the date the underlying option was granted. If a grant is not in conjunction with an option, the payment will be determined based on the excess of the fair market value of the excess of the fair market value of the Company Common Stock at the date of exercise over the option price at the date of exercise over the fair market value of the excess of the fair market value of the Company Common Stock at the date of exercise over the fair market value of the fair market value of the Company Common Stock at the date of grant of the SAR.

Dividend Equivalents. The Committee may grant awards of dividend equivalents in conjunction with an option, a separately-awarded SAR or performance unit. No consideration is payable by the participant as a result of the grant. Each dividend equivalent entitles the participant to receive an amount, at such times and in a form and manner in the Committee's discretion, equal to the dividend actually paid with respect to a share of Company Common Stock on each dividend payment date from the date of grant until the dividend equivalent lapses. Dividend equivalents will lapse at a date no later than the date the underlying award lapses or is exercised.

te generation

. . . . i .

### Effect of Change in Control

After the consummation of the Mergers, unless the Committee specifies otherwise, upon the occurrence of a change in control of the Company; (a) all options and SARs granted under the Company Plan will become exercisable in full immediately; (b) all uncompleted performance periods for performance units will be deemed to have been completed, the maximum level of performance will be deemed to have been achieved and a pro rata portion of each performance unit will become payable immediately in cash: (c) any conditions to the vesting of Company Restricted Stock will be deemed to have been satisfied (assuming that maximum performance was achieved for performance-based awards), any uncompleted time periods will be deemed to have been completed and all restrictions will lapse immediately; and (d) annual incentive awards shall be treated in such manner as the Committee determines at the time of such change in control. For purposes of the Company Plan, the term "change in control" means (a) any person (within the meaning of Section 13(d) or 14(d) of the Exchange Act) is or becomes the beneficial owner (within the meaning of Rule 13d-3 promulgated under the Exchange Act) of securities of the Company representing 25% or more of either the then outstanding shares of Company Common Stock or the combined voting power of the Company's then outstanding securities; or (b) the following individuals cease for any reason to constitute a majority of the number of directors then serving: individuals who constitute the initial Board after the consummation of the Mergers and any new director (other than a director whose initial assumption of office is in connection with an actual or threatened election contest. including but not limited to a consent solicitation, relating to the election of directors) whose appointment or election by the Board or nomination for election by the Company's stockholders was approved by a vote of at least two-thirds (2/3) of the directors then still in office who either were directors of the initial Board after the consummation of the Mergers, or whose appointment, election or nomination for election was previously so approved; or (c) there is consummated a merger or consolidation of the Company with any other corporation other than (i) a merger or consolidation which would result in the voting securities of the Company outstanding immediately prior to such merger or consolidation continuing to represent at least 80% of the combined voting power of the voting securities of the Company or such surviving entity or any parent thereof outstanding immediately after such merger or consolidation, or (ii) a merger or consolidation effected to implement a recapitalization of the Company (or similar transaction) in which no person is or becomes the beneficial owner, directly or indirectly, of securities of the Company representing 25% or more of either the then outstanding shares of Company Common Stock or the combined voting power of the Company's then outstanding securities; or (d) the stockholders of the Company approve a plan of complete liquidation or dissolution of the Company or there is consummated an agreement for the sale or disposition by the Company of all or substantially all of the Company's assets, other than a sale or disposition by the Company of all or substantially all of the Company's assets to an entity, at least 80% of the combined voting power of the voting securities of which are owned by persons in substantially the same proportions as their ownership of the Company immediately prior to such sale.

· . .

-1911 -

 $(\mathcal{F}_{\lambda})$ 

### **Tax Withholding**

The Company or its subsidiaries may withhold any applicable federal, state or local taxes upon payment under an award. Subject to any applicable law, if payment under an award is to be made in Company Common Stock, the Committee may in its discretion permit or require a participant to satisfy any withholding or other taxes payable through (a) the payment of cash by the participant to the Company or its subsidiaries, (b) the retention by the Company or its subsidiaries of shares of Company Common Stock, or (c) the delivery by the participant to the Company or its subsidiaries of Company Common Stock owned by the participant. Special rules may apply to participants subject to the reporting requirements of Section 16(a) of the Exchange Act.

### **Federal Income Tax Consequences**

The federal income tax consequences of an award under the Company Plan depend on the type of award, as discussed below:

*Restricted Stock.* The grant of Company Restricted Stock does not result in taxable income to a participant or a tax deduction for the Company. At the time the restrictions expire, however, a participant will realize ordinary taxable income in an amount equal to the fair market value of the Company Common Stock on the date the restrictions expire, and the Company will be entitled to a corresponding deduction. In addition, during or after the restriction period (depending on whether the dividends are paid to the individuals or accumulated), a participant will be taxed on the dividends paid with respect to Company Restricted Stock as compensation, and the Company will be entitled to a corresponding deduction.

Incentive Stock Options. ISOs under the Plan are intended to meet the requirements of Section 422 of the Code. The grant of an ISO does not result in taxable income to the participant or a tax deduction for the Company. The exercise of an ISO will not result in ordinary taxable income to the participant (although the difference between the exercise price and the fair market value of the Company Common Stock subject to the option may result in alternative minimum tax liability to the participant) and the Company will not be allowed a deduction at any time in connection with such award, if the following conditions are met: (a) at all times during the period beginning with the date of grant and ending on the day three months before the date of exercise, the participant is an employee of the Company or of a subsidiary; and (b) the participant makes no disposition of Company Common Stock within two years from the date of grant nor within one year after the Company Common Stock is transferred to the participant. The three-month period is extended to one year in the event of disability and is waived in the event of death of the participant. If the Company Common Stock is sold by the participant after meeting these conditions, any gain realized over the exercise price ordinarily will be treated as long-term capital gain, and any loss will be treated as long-term capital loss, in the year of the sale.

If the participant fails to comply with the employment or holding period requirements discussed above, the participant will recognize ordinary taxable income in an amount equal to the lesser of (a) the excess of the fair market value of the Company Common Stock on the date of exercise over the exercise price, or (b) the excess of the amount realized upon such disposition over the exercise price. If the participant realizes ordinary taxable income on account of such a disqualifying disposition (described above), a corresponding deduction will be allowed to the Company for the same year.

Nonqualified Stock Options. The grant of a Nonqualified Option does not result in taxable income to the participant or a tax deduction for the Company. Upon exercise of a Nonqualified Stock option, the participant will realize compensation taxable as ordinary income in an amount equal to the difference between the exercise price and the fair market value of the Company Common Stock on the date of exercise, and the Company will be entitled to a corresponding deduction for the same year. The participant's basis in such shares will be the fair market value on the date income is realized, and when the participant disposes of the shares he or she will recognize capital gain or loss, either long-term or short-term, depending on the holding period of the shares.

Stock Appreciation Rights. The grant of SARs does not result in taxable income to the participant or a tax deduction for the Company. Upon exercise of an SAR, the participant will realize ordinary taxable income in an amount equal to the excess of the fair market value of the Company Common Stock or cash received over any amount paid by the participant upon exercise, and the Company will be entitled to a corresponding deduction for the same year.

Performance Units. The grant of a performance unit does not result in taxable income to the participant or a tax deduction for the Company. Upon the expiration of the applicable award cycle and receipt of the Company Common Stock distributed in payment of the award or an equivalent amount of cash, the participant will realize ordinary taxable income equal to the full fair market value of the shares delivered or the amount of cash paid. At that time, the Company generally will be allowed a corresponding tax deduction equal to the compensation taxable to participant.

Accounting principles require that restricted share awards be charged against earnings on a pro rata basis over the restriction period and will be based on the value of the stock at the date of grant. Under the terms of the Company Plan, the granting of ISOs or Nonqualified Options, without accompanying stock appreciation rights, will not require a charge against earnings. The granting of SARs, however, will require that earnings be charged over the specified award period for any appreciation in the value of the underlying stock subsequent to the date of grant.

### **New Plan Benefits**

It cannot be determined at this time what benefits or amounts, if any, will be received by or allocated to any person or group of persons under the Company Plan if the Company Plan is adopted or what benefits or amounts would have been received by or allocated to any person or group of persons for the last fiscal year if the Company Plan had been in effect. Any such determinations will be made by the Committee in its sole discretion.

# UNAUDITED PRO FORMA COMBINED FINANCIAL STATEMENTS.

The following unaudited pro forma combined financial statements combine the historical balance sheets and statements of operations of Delmarva and Atlantic to give effect to the Mergers under the purchase method of accounting and the assumptions set forth in the notes thereto. The unaudited pro forma combined balance sheet as of September 30, 1996 assumes that the Mergers were consummated on September 30, 1996. The unaudited pro forma combined statements of operations for the nine months ended September 30, 1996, and the year ended December 31, 1995 assume that the Mergers were consummated on January 1, 1995. The statements do not reflect any cost savings or other synergies anticipated by management as a result of the Mergers. In the opinion of management, all adjustments necessary to present pro forma financial statements have been made.

The following unaudited pro forma combined financial information should be read in conjunction with the historical consolidated financial statements and related notes thereto of Delmarva, Atlantic and ACE, incorporated by reference herein. See "Available Information" and "Incorporation by Reference." The following information is not necessarily indicative of the financial position or operating results that would have occurred had the Mergers been consummated on the date, or at the beginning of the periods, for which the Mergers are being given effect, nor is it necessarily indicative of future operating results or financial position.

### PRO FORMA COMBINED BALANCE SHEETS

# **SEPTEMBER 30, 1996**

# (Dollars in Thousands)

(Unaudited)

ASSETS					
	Delmarva As Adjusted(s)	Atlantic As Adjusted(s)	Pro Forma Adjustments	The Company Pro Forma	
Utility Plant, At Cost	•				
Utility Plant, At Cost Electric	\$2,944,454	\$2,497,685		\$5,442,139	
Gas	219,916			219,916	
Common	133,430			133,430	
	3,297,800	2,497,685		5,795,485	
Less: Accumulated depreciation	1,263,693	853,134	<u>.'                                    </u>	2,116,827	
Net utility plant in service	2,034,107	1,644,551		3,678,658	
Construction work-in-progress	99,008	120,075	<u> </u>	219,083	
Leased property, net	34,029	37,695	—	71,724	
Cost in excess of net assets acquired, net	75,990		\$204,476 (f)	280,466	
	2,243,134	1,802,321	204,476	4,249,931	
Investments and Nonutility Property					
Investments in leveraged leases	47,306	79,500	·	126,806	
Funds held by trustee	34,662	81,591		116,253	
Other investments and nonutility property,					
net	60,331	68,909		129,240	
	142,299	230,000	·	372,299	
Current Assets				2	
Cash and cash equivalents	28,280	15,100		43,380	
Accounts receivable:			·		
Customers	110,881	108,466		219,347	
Other	24,219	26,189		50,408	
Deferred energy costs	19,041	33,568	—	52,609	
Inventories, at average cost:		<b>a</b> 4 000		55 50 4	
Fuel (coal, oil, and gas)	31,724	24,000	. —	55,724	
Materials and supplies	35,868	40,057		75,925	
Prepayments	9,277	29,160		38,437	
Other		4,451		4,451	
	259,290	280,991	· · · · · · · · · · · · · · · · · · ·	540,281	
<b>Deferred Charges and Other Assets</b>			~4 J		
Unrecovered purchased power costs		87,507	·	87,507	
Deferred recoverable income taxes	140,983	85,858		226,841	
Unrecovered state excise taxes		57,104		57,104	
Deferred debt refinancing costs	22,018	30,493		52,511	
Other regulatory assets	31,406	59,237		90,643	
Prepaid pension cost	27,458	7,601	(7,601)(g		
Unamorfized debt expense	11,776	5,952	4,656 (k		
Other	19,747	27,636		47,383	
	253,388	361,388	(2,945)	611,831	
Total Assets	\$2,898,111	\$2,674,700	\$201,531	\$5,774,342	
				<u> </u>	

The accompanying notes to the unaudited pro forma combined balance sheet and statements of income are an integral part of this statement.

. ,

# PRO FORMA COMBINED BALANCE SHEET

# SEPTEMBER 30, 1996 (Dollars in Thousands) (Unaudited)

# **CAPITALIZATION AND LIABILITIES**

٦

	Delmarva As Adjusted(s)	Atlantic As Adjusted(s)	Pro Forma Adjustments Pro Forma
Capitalization	**		·
Common stock	\$ 136,717	\$ 565,598	\$(701,315)(a) \$ 1,000 66 (a) 66
Additional paid-in capital—common stock Additional paid-in capital—Class A common	506,680	· · · ·	940,627 (b) 1,447,307
stock	,		137,289 (b) 137,289
Retained earnings	299,108	247,222	(262,770)(d) 283,560
· · · · · · · · · · · · · · · · · · ·	942,505	812,820	113,897 1,869,222
Treasury shares, at cost	(5,020)		5,020 (e) —
Unearned compensation	(824)	(5,314)	6,138 (m) <u> </u>
Total common stockholders' equity Preferred stock not subject to mandatory	936,661	807,506	125,055 1,869,222
redemption	168,085	· · ·	(168,085)(k) —
Not subject to mandatory redemption	_	30,000	89,702 (k) 119,702
Subject to mandatory redemption	·	90,000	93,950 (k) 183,950
Long-term debt	827,242	829,799	— 1,657,041
	1,931,988	1,757,305	140,622 3,829,915
Current Liabilities			4
Short-term debt	81,187	87,700	(4,146)(k) 164,741
Preferred stock redemption requirement	—	10,000	— 10,000
Long-term debt due within one year	27,244	74,100	— 101,344
Variable rate demand bonds	86,500		— 86,500
Accounts payable	50,694	55,359	600 (k) 106,653
Taxes accrued	12,396	40,159	(5,073)(i)(m) 47,482
Interest accrued	21,336	17,186	
Dividends declared	23,288	22,598	45,886
Current capital lease obligation	12,456	688 2,438	13,144
Other	5,438 31,314	36,829	— 7,876 45,394 (h)(i) 113,537
Oulei		,	
	351,853	3,47,057	36,775 735,685
Deferred Credits and Other Liabilities		- 12 <sup>1</sup>	• • •
Deferred income taxes, net	518,785	428,309	(34,374)(n) 912,720
Deferred investment tax credits	43,141	47,211	90,352
Long-term capital lease obligations	21,711	37,006	58,717
Postretirement obligations	<u> </u>	31,581	58,508 (g) 90,089
Other	30,633	26,231	56,864
	614,270	570,338	24,134 1,208,742
Total Capitalization and Liabilities	\$2,898,111	\$2,674,700	\$ 201,531 \$5,774,342

The accompanying notes to the unaudited pro forma combined balance sheet and statements of income are an integral part of this statement.

### PRO FORMA COMBINED STATEMENTS OF INCOME

### FOR THE NINE MONTHS ENDED SEPTEMBER 30, 1996 (Dollars in Thousands, Except Per Share Amounts) (Unaudited)

(Unaudited)	ted)
-------------	------

	Delmarva As Adjusted(s)	Atlantic As Adjusted(s)	Pro Forma Adjustments	The Company Pro Forma
Operating Revenues				· · · ·
Electric	\$752,415 82,164	\$752,968 	· ·	\$1,505,383 82,164
	834,579	752,968	· · · ·	1,587,547
Operating European	001,072			
<b>Operating Expenses</b> Electric fuel and purchased energy	244,593	171,648		416,241
Gas purchased	43,844			43,844
Purchased electric capacity	25,147	146,877	· — . ´	172,024
Operation and maintenance	191,449	143,054		334,503
Depreciation and amortization	91,435	60,490	\$ 3,834 (j)	155,759
State excise taxes		80,391	(2 (10) (1)	80,391
Income taxes	62,547	30,842	(3,612)(1)	89,777
Other taxes	32,218	7,656		39,874
	691,233	640,958		1,332,413
Operating Income	143,346	112,010	(222)	255,134
Other Income				<u>i</u> ta e a co
Nonutility subsidiaries	42 104	10 572		50 767
Revenues and gains	42,194 (39,105)	10,573 (10,033)		52,767 (49,138)
Net earnings of nonutility subsidiaries Allowance for equity funds used during	3,089	540	<u></u>	3,629
construction	774	697	, <del></del>	1,471
Other income, net of income taxes	(1,202)	2,063		861
	2,661	3,300		5,961
Income Before Interest Charges	146,007	115,310	(222)	261,095
Interest Charges		_		1
Interest expense	53,589	49,303	1,176 (l)	104,068
Allowance for borrowed funds used during	(2.095)	(920)		(0,005)
construction	(2,085)	(820)		(2,905)
	51,504	48,483	1,176	101,163
Preferred stock dividend requirements of		0.177		
subsidiaries		8,475	7,050 (l)	15,525
Net income Dividends on preferred stock	94,503 7,293	58,352	(8,448) (7,293)(1)	144,407
			(1,295)(1)	,
Earnings applicable to common stock Common stock	87,210	58,352	(10,734)	134,828
Class A common stock	07,210 	J8,352	9,579 (o)	9,579
	¢ 07.010	¢ 50,250		
	<u>\$ 87,210</u>	\$ 58,352	<u>\$ (1,155)</u>	\$ 144,407
Average common shares outstanding (000):	<b>10 -1</b> 00			
Common stock	60,709	52,702	(13,175)(p)	100,236
Class A common stock Earnings per average share outstanding of:	<u> </u>	<del></del> ,	6,588´(p)	6,588
Common stock	\$1.44	\$1.11	V	\$1.35
Class A common stock	Ψ1.77	Ψ1.11	· · · · · · · · · · · · · · · · · · ·	\$1.45
Dividends declared per share of:			•	ψ1.15
Common stock	\$1.155	\$1.155		\$1.155
Class A common stock	·	, <u> </u>	. — .	\$2.40
1	· · · · ·			

The accompanying notes to the unaudited pro forma combined balance sheet and statements of income are an integral part of this statement.



# PRO FORMA COMBINED STATEMENTS OF INCOME

# FOR THE YEAR ENDED DECEMBER 31, 1995 (Dollars in Thousands, Except Per Share Amounts) (Unaudited)

· · · · · ·	Delmarva As Adjusted(s)	Atlantic As Adjusted(s)	Pro Forma Adjustments	The Company Pro Forma
Operating Revenues				
Electric	\$899,662 95,441	\$953,137 —	`	\$1,852,799 95,441
	995,103	953,137		1,948,240
Operating Expenses		<u> </u>	·	
Electric fuel and purchased energy	267,885	191,766	<del>_</del> .	459,651
Purchased electric capacity	48,615 29,116	190,570		48,615
Operation and maintenance	246.049	186,439	_	219,686 432,488
Depreciation and amortization	113,022	78,461	\$ 5,112 (j	
State excise taxes	<u> </u>	102,811	· · · ·	102,811
Income taxes	73,561	45,876	(4,927)(1	
Other taxes	. 38,449	8,677	<u> </u>	47,126
	816,697	804,600	185	1,621,482
Operating Income	178,406	148,537	(185)	326,758
Other Income Nonutility subsidiaries			ŕ	, ,
Revenues and gains	52,042	6,925	· · <u>· ·</u> ,	58,967
Expenses	(47,896)	(7,680)		(55,576)
Net earnings of nonutility subsidiaries	4,146			
Allowance for equity funds used during construction	4,146 708	(755) 817	· .	3,391
Other income, net of income taxes	557	8,996	·	1,525 9,553
	5,411	9,058		14,469
Income Before Interest Charges	183,817	157,595	(185)	, 341,227
Interest Charges				, <u></u>
Interest expense	68,395	62,879	1,886 (1	) 133,160
Allowance for borrowed funds used during construction	(2,066)		1,000 (I	(3,745)
	66,329	61,200	1,886	129,415
Preferred stock dividend requirements of subsidiary		14,627	9,618 (l	) 24,245
Net income	117,488	81,768	(11,689)	
Dividends on preferred stock	9,942		(9,942)(1	
Earnings applicable to common stock			/(/(	
Common stock	107,546	81,768	(15,578)	173,736
Class A common stock		<u> </u>	13,831 (c	
,	\$107,546	\$ 81,768		
	\$107,540	\$ 61,700	<u>(1,747)</u>	<u>    187,567</u>
Average common shares outstanding (000):	,		, i .	
Common stock	60,217	52,815	(13,288)(p	) 99,744
Class A common stock	·	· · <u> </u>	6,588´(p	o) 6,588
Earnings per average share outstanding of: Common stock	¢1.70	¢1 55	· ·	<b>61 71</b>
Class A common stock	\$1.79	\$1.55		\$1.74
Dividends declared per share of:		_	·	\$2.10
Common stock	\$1.54	\$1.54	<u> </u>	\$1.54
Class A common stock	φ1.54	φ1.5+	· <u> </u>	\$3.20
				ψJ.20

The accompanying notes to the unaudited pro forma combined balance sheet and statements of income are an integral part of this statement.

### NOTES TO UNAUDITED PRO FORMA COMBINED FINANCIAL STATEMENTS

(a) Adjustments to record the estimated par value at \$0.01 per share of Company Common Stock and Class A Common Stock to be issued and outstanding. The number of shares of Common Stock was estimated using the number of Delmarva Common Stock and Atlantic Common Stock shares outstanding as of September 30, 1996. Each outstanding share of Delmarva Common Stock was converted into one share of Company Common Stock and each outstanding share of Atlantic Common Stock was converted into 0.75 of one share of Company Common Stock plus 0.125 of one share of Class A Common Stock. The adjustments are summarized below:

	As of September 30, 1996
Common Stock:	· · · ·
Number of Atlantic Common Stock shares outstanding         Conversion Ratio	52,702,052 0.75
Number of Common Stock shares to be issued to Atlantic Common         Stockholders         Number of Common Stock shares to be issued to Delmarva Common         Stockholders (Equal to the number of Delmarva Common Stock	39,526,539
shares outstanding)	60,523,411
Total number of Common Stock shares to be issued         Par value per share	100,049,950 \$0.01
(In Thousands of Dollars): Adjusted par value of total number of Common Stock shares to be issued	\$ 1,000
Delmarva's Common Stock, as previously reported	(136,717) (565,598)
Adjustment to Common Stock	\$(701,315)
Class A Common Stock: Number of Atlantic Common Stock shares outstanding Conversion Ratio Number of Class A Common Stock shares to be issued to Atlantic	52,702,052 0.125
Common Stockholders	6,587,757 \$0.01
Par value (In Thousands of Dollars)	\$66



(b) Adjustments to record additional paid-in-capital to reflect the following:

	As of September 30, 1996
	(Dollars in Thousands)
Additional Paid-In-Capital-Common Stock:	
Cancellation of the Delmarva Treasury Stock cost in excess of par	
value	\$ (4,481)
Adjustment to par value of Delmarva Common Stock outstanding	135,572
Consideration to be paid to Atlantic's Common Stockholders in the	
form of Company Common Stock in excess of par value	810,887
Estimated registration and issuance costs	(1,750)
Redemption of preferred stock [see Note (k)]	399
	\$940,627
Additional Daid In Constal Class A Commun Starly	
Additional Paid-In-Capital—Class A Common Stock:	
Consideration to be paid to Atlantic's Common Stockholders in the	+
form of Class A Common Stock in excess of par value	<u>\$137,289</u>

(c) The total consideration to be paid to the Atlantic Common Stockholders was measured by the average daily closing market price of Atlantic's Common Stock for the ten trading days following the public announcement of the Merger Agreement on August 12, 1996.

Delmarva's Common Stockholders will receive one share of Company Common Stock for each share of Delmarva's Common Stock. Therefore, the average daily market price of Delmarva's Common Stock for the same ten day period following the public announcement of the Merger Agreement was used to measure the market value of Company Common Stock to be paid to Atlantic's Common Stockholders. Delmarva's average market price per share was multiplied by the Atlantic conversion ratio for Company Common Stock to determine the estimated market value per share of Atlantic Common Stock attributed to Company Common Stock. This market value per share was multiplied by the number of Atlantic Common Stock shares outstanding at September 30, 1996 to estimate the consideration to be paid to Atlantic Common Stockholders in the form of Company Common Stock.

The difference between the total compensation to be paid to Atlantic's Common Stockholders and the portion attributed to Company Common Stock was attributed to Class A Common Stock.

The schedules below show the calculation of the total consideration to be paid to Atlantic's Common Stockholders and the allocation of the total consideration to be paid between Company Common Stock and Class A Common Stock:

•		Amounts
Average market price per share of Atlantic Common Stock used to determine consideration to be paid Number of Atlantic Common Stock shares outstanding as of September 30,	\$	18.00
1996	52	2,702,052
Total consideration to be paid to Atlantic Common Stockholders (In         Thousands of Dollars)	\$	948,637
Average market price per share of Delmarva Common Stock for the ten trading days following the public announcement of the Merger Agreement Conversion ratio of Company Common Stock for each share of Atlantic	\$	20.525
Common Stock		0.75

and the state of the state of the september of the state of the second state of the
Estimated market value per share of Atlantic Common Stock attributed to
Company Common Stock \$ 15.39375
Number of Atlantic Common Stock shares outstanding as of September 30,
1996
Consideration to be paid to Atlantic's Common Stockholders in the form of
Company Common Stock (In Thousands of Dollars) \$ 811,282
(In Thousands of Dollars):
Total consideration to be paid to Atlantic Common Stockholders \$ 948,637
Portion of total consideration attributed to Company Common Stock
Portion of total consideration attributed to Class A Common Stock

(d) Adjustment to retained earnings to reflect the following:

: C. . .

1.1

and the second

이 가슴에 있는 것이 있는 것이 있다. 같이 같이 있는 것이 있는 것이 있는 것이 있는 것이 있는 것이 있는 것이 같이 있는 것이 같이 있는 것이 같이 있는 것이 같이 있는 것이 있는 것이 있는 것이 있는 것이 없는 것이 없는 것이 있는 것

出行 通知 とうりん

Sector States

Contraction of the second

131

4. 194

 $\{e_{ij}\}$ 

.

Stand Mr. Stand & Store

1.84

 $\{0\}, \{j\} \in$ 

414-

. . . . . . .

54 		ι, μ. τ.		n ng sa sa pina. T	Amounts -
	. · · ·			. · · · · · · · · · · · (1	Dollars in Thousands)
Eliminate retained earn	ings of Atlantic		· · · · · · · · ·	1997 - 1997 -	\$ (238,397)
Charges to expense of \$					125 <b>3</b> . 113 .
nonrecurring employe	e separation costs re	elated to Del	lmarva emj	oloyees	
and employee retrain	ing costs [see Note (	[h)]	- \$1.8°.4° स्. । • • • • • • • • •	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	(7,280)
Charges to retained ear	nings related to the	redemption of	of preferre	d stock	
[see Note (k)]	والم الم الم الم الم الم الم الم الم الم	المنتخ والمنتج والمنتج والمنتج		and the second second	(7,765)
Charge to expense to el	iminate unearned in	come [see N	ote (m)]	• • • • • • •	(3,957)
Charge to expense for A					
Note (i)] $\ldots$	• • • • • • • • • • • • • • • • • •	• • • • • • •			(5,371)
Total adjustment	,	in dia static di serie di seri Serie di serie	• • • • • •	8	\$ (262,770)

The states

 $\sim 1, f$ 

the state of the second

the second pr

化二乙酸乙酸盐酸乙酸乙酯 经运行

1.1.1.1

1 197

,

1 8 1123

1. 1. 1. 2.

Alt - pa

genere companye

· . . .

二、人口、竹、竹子

1. 1. 15.0 Tells 1. 16 me

Prior to elimination, the retained earnings of Atlantic, as reported in its Form 10-Q for the quarter ended September 30, 1996, of \$247,222,000 was reduced by \$3,454,000, which is Atlantic's after tax portion of the expense recognized that was related to employee incentive plans [see Note (m)] and \$5,371,000, which is Atlantic's after tax portion of direct costs of the acquisition [see Note (i)].

at the second states of the

· . .

العائمة المراجعة المراجع والمراجع وال

.

A MAR TO L

and the second second

the ends

(e) Adjustment to reflect the cancellation of the Delmarva treasury stock as a condition of the Mergers.

and the second second second second second second

1.124.1

i, i the e

· 1 · ·

• •

a shikiyeze menanî kir. Dartî kaza ew waliyê yeke par sê bar ji

12

in the second of the second day

(f) The schedule below shows the calculation of the cost of acquiring Atlantic and the allocation of the total acquisition cost to identifiable tangible and intangible assets and liabilities.

Cost of Acquiring Atlantic	Amounts
1 If the part was dependent of the part	(Dollars in Thousands)
Consideration to be paid to Atlantic's Common Stockholders [see Note (c)].	\$ 948,637
Add: Estimated direct costs of acquisition to be incurred by Delmarva	8,263
Less: Registration and issuance costs	(1,750)
Total acquisition cost	\$ 955,150
" Less assets acquired: "	
Electric utility plant-net	\$1,802,321
Investments and nonutility property	230,000
Current assets	280,991
Deferred debits	361,388
Total assets acquired (excluding goodwill)	\$2,674,700
Add liabilities acquired:	
Preferred stock of subsidiaries	\$ 120,000
	829,799
Current liabilities	350,568
Current liabilities	570,338
Total liabilities acquired	\$1,870,705
Costs incurred and liabilities assumed in connection with the Mergers	\$ 53,321
Cost in excess of net assets acquired	\$ 204,476

The current liabilities of Atlantic as of September 30, 1996 included in net assets acquired was adjusted to reflect transactions to be recorded by Atlantic prior to the Mergers as shown below:

The second first where the manipulation are second with the second second	As of September 30, 1996
	(Dollars in Thousands)
Current liabilities of Atlantic as adjusted [see Note (s)]	\$347,057
Accrued liability for Atlantic's portion of direct costs of the Mergers [see Note (i)]	and the second
Accrued tax benefits [see Notes (i) and (m)]	. (4,752)
Current liabilities acquired	\$350,568

The fair value of the utility assets of Atlantic is their book value due to the ratemaking process. Utility assets are recognized for ratemaking purposes at their book values in determining utility revenue requirements. Accordingly, the economic substance is that the fair value of the utility assets is their book value.

The estimated fair values of Atlantic's nonutility assets is not significantly different from their carrying values as reported by Atlantic as of September 30, 1996.

- (g) Adjustments to record additional pension and other postretirement benefit liabilities, totaling \$66.1 million, as liabilities assumed in the acquisition of Atlantic in accordance with Statements of Financial Accounting Standards (SFAS) Nos. 87 and 106.
- (h) Adjustment to record an estimated liability of \$16.9 million, which is included in the acquisition cost, for employee separation and relocation costs and facilities integration costs related to Atlantic's employees and facilities and a liability of \$11.9 million, which will be expensed, for employee separation costs related to Delmarva's employees and employee retraining costs. The Unaudited Pro

Forma Combined Statements of Income for the nine months ended September 30, 1996 and twelve months ended December 31, 1995 do not reflect the nonrecurring estimated expenses of \$11.9 million before taxes (\$7.3 million after taxes) for employee separation costs related to Delmarva's employees and employee retraining costs.

(i) Adjustments to record estimated direct costs of the Mergers of \$16.5 million.

It is estimated that these costs will be incurred equally by Atlantic and Delmarva. The Atlantic portion of these costs of \$8.3 million before taxes (\$5.4 million after taxes) will be expensed by Atlantic prior to the Mergers. The Delmarva portion of these costs will be included in the cost of acquiring Atlantic. The adjustments are summarized below:

		· · · · · ·		As of September 30, 1996
	78. i	1.6.10	<i>.</i>	(Dollars in Thousands)
Atlantic's portion of these costs: Decrease in retained earnings				
Accrued tax benefit	 			\$(2,892)
Other current liabilities	 	ار بر از	••••••	\$ 8,263
Delmarva's portion of these costs: Other current liabilities				\$ 8,263
•	 ÷.	£ . 7		· · · · ·

The Unaudited Pro Forma Combined Statements of Income for the nine months ended September 30, 1996 and twelve months ended December 31, 1995 do not reflect Atlantic's portion of the direct costs of the Mergers, which are nonrecurring.

(j) Adjustment to reflect the amortization of goodwill acquired over forty (40) years.

(k) Adjustments to record the following issuances of preferred stock by wholly owned trusts of Delmarva and an Atlantic subsidiary and redemptions of preferred stock of Delmarva and an Atlantic subsidiary: Determined Bate

	Date	Dividend Rate	Amount
• .	Issuances:       0         October 1, 1996       0         October 3, 1996       0	8.25% 8.125%	\$70,000,000 70,000,000
· .	Redemptions: October 7, 1996	\$ 7.80	37,050,000
•	October 16, 1996	. 4.00%,	1,013,400 2,012,600
:.	October 16, 1996	4.20% 4.28%	2,459,600 2,154,000
	October 16, 1996	.5.00%	3,042,900 3,147,700
	October 16, 1996	6.75% 7.75%	16,500,000 32,087,500
	October 16, 1996	Adjustable \$ 7.80	965,000 5,000,000
	November 21, 1996	\$	4,000,000 15,000,000

Also, during the third quarter of 1996, an Atlantic subsidiary redeemed \$35 million of preferred stock, including a 7.52% Series, which increased short-term debt.

λ£.

Short-term debt was adjusted to reflect the effect of the transactions listed above.

A summary of the adjustments is presented below.

	· ·	As of September 30, 1996
		(Dollars in Thousands)
	Preferred Stock: Delmarva Preferred Stock to be redeemed Delmarva Preferred Stock reclassified to Preferred Stock of Subsidiaries	(\$ 78,383) (89,702)
	Adjustment to Preferred Stock	\$(168,085)
	Preferred Stock of Subsidiaries: Not subject to mandatory redemption: Delmarva Preferred Stock reclassified	\$ 89,702
	Subject to mandatory redemption: Preferred stock to be issued Less: Atlantic subsidiary preferred stock to be redeemed	\$ 140,000 (46,050)
	Adjustment	\$ 93,950
	Paid-in-Capital: Amounts transferred to Retained Earnings	\$ 399
,	Retained Earnings:         Estimated fees to redeem preferred stock         Other costs to redeem preferred stock	\$ (1,333) (6,432)
	Adjustment to Retained Earnings	<u>\$ (7,765)</u>
	Unamortized Debt Expense: Issuance costs of wholly owned trust preferred securities	\$ 4,656
	Short-Term Debt: Net decrease as of September 30, 1996	\$ (4,146)
	Accounts Payable: Fees to be paid	\$ 600

Ac of

On October 3, 1996, Delmarva's wholly owned trust ("Delmarva Power Financing I"), formed for the purpose of issuing securities, issued \$70 million of 8.125% preferred capital securities with a liquidation amount of \$25 per preferred capital security. Delmarva's trust invested the proceeds in 8.125% subordinated debentures issued by Delmarva which are due on September 30, 2036 and may be redeemed in whole or in part on or after September 30, 2001. The preferred capital securities issued by the trust are subject to redemption upon repayment of the subordinated debentures. If (1) the trust becomes subject to federal income taxation, (2) interest paid on the debentures is no longer deductible for federal income tax purposes, or (3) the trust is required to be registered under the Investment Company Act of 1940, the debentures and preferred stock may be redeemed, or the trust may be terminated and the debentures may be distributed to holders of the preferred capital securities.

Distributions from the trust to holders of the preferred capital securities are payable quarterly at 8.125%. Delmarva may defer interest payments on the subordinated debentures for up to 20 consecutive quarters. If interest payments are deferred, distributions on the preferred capital securities are also deferred and Delmarva would not be permitted to (1) declare or pay dividends on any of its capital stock, or (2) make any payment of principal, interest or premium on debt, or reacquire debt, that is pari passu with or junior to the subordinated debentures. Interest and distributions continue to accrue, compounded quarterly, during a deferral period.

On October 1, 1996, ACE's newly formed business trust ("Atlantic Capital I"), issued \$70 million of 8.25% Cumulative Quarterly Income Preferred Securities ("QUIPS") with a stated liquidation preference of \$25 each. The trust, established for the purpose of issuing the QUIPS, invested the proceeds in 8.25% Junior Subordinated Deferrable Interest Debentures ("QUIDS") issued by ACE. The QUIDS mature on October 1, 2026, a date which may be extended not later than October 1, 2045, if certain conditions are met. On or after October 1, 2001, the QUIDS are redeemable in whole or in part at ACE's option. The QUIPS are subject to redemption upon repayment of the QUIDS. If (1) the trust becomes subject to federal income taxation, (2) interest paid on the QUIDS is no longer deductible for federal income tax purposes, or (3) the trust is required to be registered under the Investment Company Act of 1940, then the QUIDS and QUIPS may be redeemed, or the trust may be terminated and the QUIDS may be distributed to holders of the QUIPS.

Distributions from the trust to holders of the QUIDS are payable quarterly at 8.25%. ACE has the right to defer interest payments on the QUIDS for up to 20 consecutive quarters. If interest payments are deferred, distributions on the QUIPS are also deferred and ACE may not, and may not permit any subsidiary to (1) declare or pay dividends on ACE's capital stock, or (2) make any payment of principal, interest or premium on debt, or reacquire debt, that is pari passu with or junior to the QUIDS. Interest and distributions continue to accrue, compounded quarterly, during a deferral period.

(l) Adjustments to reflect the recurring effects of preferred stock issuances and redemptions and related changes in short-term debt as discussed in Note (k). A summary of the adjustments is presented below:

	Nine Months Ended September 30, 1996	Year Ended December 31, 1995
	(Dollars in Th	ousands)
Dividends on Preferred Stock:		
Dividends on preferred stock to be redeemed Remaining Delmarva dividends on preferred stock reclassified to		\$(5,391)
preferred stock dividends of Subsidiaries		(4,551)
	<u>\$ (7,293)</u>	<u>\$(9,942)</u>
Preferred Stock Dividend Requirements of Subsidiaries:		
Reclassified preferred dividends of Delmarva	\$ 3,250	\$ 4,551
Dividends on Atlantic subsidiary preferred stock to be redeemed		(6,396)
Dividends on preferred stock to be issued		11,463
Adjustment	\$ 7,050	<u>\$ 9,618</u>
Income Taxes:		
Income tax benefit of Preferred Stock to be issued by wholly		•.
owned trusts	\$ (3,612)	<u>\$(4,927)</u>
Interest Expense:		
Increase in interest expense due to a net increase in short-term		
borrowings	<b>1,176</b>	\$ 1,886

(m) Adjustments to recognize a pretax expense of \$6.1 million to eliminate unearned compensation costs payable under employee incentive plans at the time of the Mergers. The adjustments are summarized below:
 As of September 30, 1996

	As of September 30, 1996
	(Dollars in Thousands)
Decrease in retained earnings: Atlantic	
Total decrease in retained earnings	<u>\$ (3,957)</u>
Accrued tax benefit: Atlantic	\$ (1,860) (321)
Total decrease in accrued taxes	\$ (2,181)
Eliminate unearned compensation	\$ 6,138

The Unaudited Pro Forma Combined Statements of Income for the nine months ended September 30, 1996 and twelve months ended December 31, 1995 do not reflect the nonrecurring estimated expense of \$6.1 million before taxes (\$4.0 million after taxes).

(n) Adjustment to record additional deferred income taxes for the following temporary differences:

· · · ·	Temporary Differences	Deferred Income Taxes
	(Dollars in	Thousands)
Additional pension and other postretirement benefit liabilities		
[see Note (g)]	\$66,109	\$23,138
Liabilities for employee separation, relocation, and retraining costs and		
facilities integration costs [see Note (h)]	28,868	10,551
Liability for Delmarva's portion of direct acquisition costs that are		
deemed to be tax deductible [see Note (i)]	1,500	685
Total deferred income taxes		\$34,374

In accordance with SFAS No. 109, deferred income taxes were not recorded on goodwill for which the amortization is not deductible for tax purposes.

(o) Adjustment to present earnings applicable to the Class A Common Stock. The Class A Common Stock is intended to reflect the growth prospects and regulatory environment of Atlantic's regulated electric utility business. When the Mergers are consummated, the shares of Class A Common Stock received by holders of Atlantic Common Stock will represent, in aggregate, a 30% interest in any earnings of Atlantic's regulated electric utility business in excess of \$40 million per year.

The calculation of the pro forma earnings applicable to the Class A Common Stock for the nine months ended September 30, 1996 and the year ended December 31, 1995 is shown below: 2146-00

121 121

N. . . . . . .

11

191 6 44

1.11

a filos de la construcción de la co La construcción de la construcción d La construcción de la construcción d	Nine Months Ended September 30, 1996	Year Ended December 31, 1995
$\frac{1}{2} \left[ \frac{1}{2} \left$	(Dollars in	Thousands)
Atlantic City Electric Company (ACE) and		الم الأرسي عارية الم
Subsidiary Income Available for Common		
Stockholders as reported in Forms 10-Q and		the second s
10-К	\$59,915	\$84,125
ACE's portion of adjustments regarding		an an an tha an an Article. Tha an tha an
preferred stock issuances and redemptions and		an a
related changes in short-term debt [see Note		
	an a	A. S. S. S. S. S. S.
Decrease in dividends on preferred stock of	생각은 가지 않는 것 같이 같이 있는 것 같이 많이 많이 많이 있다.	
subsidiaries	466	621
Decrease in income taxes	1,749	2,443
Increase in interest expense	<u>(665</u> )	(1,205)
ACE and subsidiary balance available for	an a	
Common Stockholders—as adjusted	\$61,465	\$85,984
Add: Net Losses of Nonutility Activities		
Specifically Excluded	466	120
Less: Fixed Amount of \$40 million per year	(30,000)	(40,000)
Subtotal	\$31,931	\$46,104
Percentage Applicable to Class A Common Stock	30%	30%
Earnings Applicable to Class A Common Stock	\$ 9,579	<b>\$13,831</b>
and the second secon Second second	· · · · · · · · · · · · · · · · · · ·	

- (p) Adjustments to decrease the weighted average number of Common Stock shares outstanding based on the conversion ratio of 0.75 to 1 of Company Common Stock to be issued to holders of Atlantic Common Stock and reflect the issuance of Class A Common Stock shares to holders of Atlantic Common Stock. The number of shares of Company Common Stock and Class A Common Stock estimated to be issued to holders of Atlantic Common Stock for the acquisition were deemed to be issued and outstanding for the entire period.
  - The Merger Agreement provides, subject to certain conditions, that the dividends declared and paid on the Class A Common Stock will be maintained at a level of \$3.20 per share per annum from the Effective Date until the earlier of July 1, 2001 or the end of the twelfth calendar quarter following the calendar quarter in which the Effective Date occurs. Thereafter, it is the intention of the Company. subject to certain conditions, to pay annual dividends on the Class A Common Stock in an aggregate amount (including the amount credited to the Intergroup Interest as provided in the Company Charter) equal to 90% of the Company Net Income Attributable to the Atlantic Utility Group. The Merger Agreement further provides that if and to the extent that the annual dividends paid on the Class A Common Stock during the Initial Period (including the aforesaid amount) shall have exceeded 100% of Company Net Income Attributable to the Atlantic Utility Group during such period, the Company Board may consider such fact in determining the appropriate annual dividend rate on the Class A Common Stock following the Initial Period.

The pro forma Class A Common Stock dividends per share exceed the pro forma Class A Common Stock earnings per share for the nine months ended September 30, 1996 and twelve months December 31, 1995.

(r) The 1935 Act may require the Company to divest its gas utility operations and nonutility operations. However, in the opinion of management, divestiture is not expected. The schedule below provides the pro forma revenues, operating income and identifiable assets of the Company for the nine months ended September 30, 1996 and twelve months ended December 31, 1995.

	Nine Months Ended September 30, 1996	Year Ended December 31, 1995
	(Dollars in	Thousands)
Gas Operations: Revenues	\$ 82,164	\$ 95,441 12,492 189,339
Nonutility Operations: Revenues	52,767 3,629	68,967 3,391 364,830

(s) As necessary for fair presentation of the pro forma financial statements, amounts previously reported by Atlantic and Delmarva have been reclassified for consistency of presentation. The following schedules show the amounts reclassified.

. . \* \*

. . .

. 3

· · · · ·

· ,

t ...

and the second

 $(x,y) \in \{x,y\}$ 

. . 3.49 .

Stor March 18

...

• •

:

1. E

10 A 1

ĩ

. .

÷ ,

27

 $\cdot \phi$ 

1 1 1

, y

. i -

نه، ز

, '

.

5

. . . .

. . î.e.,

. .

۰. ب

 $\sim 1$ 

. . 4

8.0

 $^{+}$   $I_{\Lambda}$ 

കണ്

137 e

1.2

la de

e Ris Turk N

 $s \in \mathbb{C}$ 

· • ,

· · J

1 Car A Sec.

1. 10

1.

81.14

, str. t Provide States

いたい かんびきょせい

W. C. C.

i stations .

2.

3.15

· · · · ·

- 52

1.4

14 C - 2

te transfer

and a second

;

.: .,

# DELMARVA POWER & LIGHT COMPANY

# CONSOLIDATED BALANCE SHEET

SEPTEMBER 30, 1996 (Dollars in Thousands) (Unaudited)

## ASSETS

and a second	Reported Amount	Reclass Adjustments	Adjusted Amount
Utility Plant, At Cost			· · · ·
Electric	\$3,022,062 219,916	\$(77,608)(1)(2)	\$2,944,454 219,916
Common	136,236	(2,806)(2)	133,430
Less: Accumulated depreciation	3,378,214 1,266,817	(80,414) (3,124)(2)	3,297,800 1,263,693
Net utility plant in service.	2,111,397	(77,290)	2,034,107
Construction work-in-progress	99,008 32,729	1 200 (2)	99,008
Cost in excess of net assets acquired, net		1,300 (2) 75,990 (1)	34,029 75,990
		<u></u> ( ) ( ) ( ) ( ) ( ) ( ) ( ) ( ) ( ) (	
	2,243,134	0	2,243,134
Investments and Nonutility Property			and the second second
Investment in leveraged leases	47,306	- exa : h b	47,306
Funds held by trustee	34,662		34,662
Other investments and nonutility property, net	60,331	· · · · · · · · · · · · · · · · ·	60,331
"学校",这些问题了"这个人,我们不是是这个人,我们	142,299	<u> 0</u>	142,299
Current Assets			un dinina. Un antonia
Cash and cash equivalents	28,280	. (1966). 1. (1976). 1. (1976).	28,280
Customers	110,881	, <u>An en</u> de la ge€.	110,881
Customers	24,219	· · · · · · · · · · · · · · · · · · ·	24.219
Deferred energy costs	19,041	, see 13 d. (19 ≥16 Mill <del>and</del> 19 de 19	19,041
Inventories, at average cost:			
Fuel (coal, oil, and gas)	31,724	and the product of the	31,724
Materials and supplies	35,868	· · · · · · · · · · · · · · · · · · ·	35,868
Prepayments	9,277	الممراجين المحاص	9,277
	259,290	0	259,290
Deferred Charges and Other Assets			
Deferred recoverable income taxes	140,983	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	140,983
Deterred debt refinancing costs	22,018		22,018
Other regulatory assets		31,406 (3)	31,406
Prepaid pension cost.	27,458	s i <u></u> ' '., ''.	27,458
Unamortized debt expense		(21,40)	11,776
Other	51,153	(31,406)(3)	19,747
	253,388	0	253,388
Total Assets	\$2,898,111	<u>\$0</u>	\$2,898,111
	•		

The accompanying Notes to Consolidated Financial Statements are an integral part of this statement,

49 ......

 $r \sim r f_{c}$ 

t sent est

# ATLANTIC ENERGY, INC.

## CONSOLIDATED BALANCE SHEET

**SEPTEMBER 30, 1996** 

(Dollars in Thousands)

(Unaudited)

## ASSETS

· · · · · · ·	Reported Amount	Reclass Adjustments	Adjusted Amount
Electric utility plant In service	\$2,492,081	\$ 5,604 (4)	\$2,497,685
Less: Accumulated depreciation	2,492,081 853,134	5,604	2,497,685 853,134
Net utility plant in service         Construction work-in-progress         Land held for future use	1,638,947 120,075 5,604	<u>5,604</u> (5,604)(4)	1,644,551 120,075
Leased property, net	37,695 1,802,321	0	37,695
Investments and Nonutility Property Investment in leveraged leases Funds held by trustee Nonutility property and equipment-net Other investments	79,500 68,946 32,024 49,530 230,000	$\begin{array}{r} 12,645 (5) \\ (32,024)(6) \\ 19,379 (5)(6) \\ \hline 0 \end{array}$	79,500 81,591 68,909 230,000
Current Assets Cash and cash equivalents. Accounts receivable: Utility service . Miscellaneous . Allowance for doubtful accounts . Unbilled revenues Deferred energy costs . Inventories, at average cost: Fuel (coal, oil, and gas) . Materials and supplies . Working funds . Prepaid excise taxes . Other .	14,627 73,762 26,189 (3,500) 38,204 33,568 24,000 24,801 15,729 29,160 11,719 288,259	$473 (7)$ $34,704 (8)$ $\overline{3,500} (8)$ $(38,204)(8)$ $-$ $15,256 (7)$ $(15,729)(7)$ $-$ $(7,268)(9)$ $-$	15,100 $108,466$ $26,189$ $$ $33,568$ $24,000$ $40,057$ $$ $29,160$ $$ $4,451$ $$ $280,991$
Deferred Charges and Other AssetsUnrecovered purchased power costsDeferred recoverable income taxesUnrecovered state excise taxesDeferred debt refinancing costsOther regulatory assetsPrepaid pension costUnamortized debt expenseOtherTotal Assets	87,507 85,858 57,104 36,445 59,237  32,950 359,101 \$2,679,681	$(5,952)(10)$ $\overline{7,601} (9)$ $5,952 (10)$ $(5,314)(11)$ $\overline{2,287}$ $(4,981)$	87,507 85,858 57,104 30,493 59,237 7,601 5,952 27,636 <u>361,388</u> \$2,674,700

The accompanying Notes to Consolidated Financial Statements are an integral part of this statement.

## **DELMARVA POWER & LIGHT COMPANY**

## **CONSOLIDATED BALANCE SHEET**

SEPTEMBER 30, 1996 (Dollars in Thousands) (Unaudited)

## CAPITALIZATION AND LIABILITIES

·	REPORTED AMOUNT	RECLASS ADJUSTMENTS	ADJUSTED AMOUNT
Capitalization			
Ĉommon stock	\$ 136,717		\$ 136,717
Additional paid-in capital—common stock	506,680	— · ,	506,680
Retained earnings	299,108		299,108
	942,505	<b>\$ 0</b>	942,505
Treasury shares, at cost	(5,020)	·	(5,020)
Unearned compensation	(824)		(824)
. Total common stockholders' equity	936,661	0	936,661
Preferred stock:		. ,	. *
Not subject to mandatory redemption	168,085		168,085
Long-term debt	827,242		827,242
	1,931,988	. 0	1,931,988
Current Liabilities			•
Short-term debt	81,187	_	81,187
Long-term debt due within one year	27,244		27,244
Variable rate demand bonds	86,500	<u> </u>	86,500
Accounts payable	50,694	<u> </u>	50,694
Taxes accrued	12,396	· ' ' ' '	12,396
Interest accrued	21,336	<u> </u>	21,336
Dividends declared	23,288	—	23,288
Current capital lease obligation	12,456	<u> </u>	12,456
Deferred income taxes, net	5,438	—	5,438
Other	31,314		31,314
	351,853	0	351,853
Deferred Credits and Other Liabilities		,	
Deferred income taxes, net	518,785	. — .,	518,785
Deferred investment tax credits	43,141		43,141
Long-term capital lease obligations	21,711		21,711
Other	30,633	<u> </u>	30,633
	614,270	0	614,270
Total Capitalization and Liabilities	\$2,898,111	<u>\$ 0</u>	\$2,898,111

The accompanying Notes to Consolidated Financial Statements are an integral part of this statement.

# ATLANTIC ENERGY, INC.

## **CONSOLIDATED BALANCE SHEET**

# **SEPTEMBER 30, 1996**

# (Dollars in Thousands)

# (Unaudited)

# CAPITALIZATION AND LIABILITIES

	REPORTED AMOUNT	RECLASS ADJUSTED ADJUSTMENTS AMOUNT
Capitalization Common stock	\$ 565,598 247,222 812,820	$ \begin{array}{c} \\ \\ \$ & 565,598 \\ \\ 247,222 \\ \$ & 0 \\ 812,820 \end{array} $
Unearned compensation		(5,314)(11) (5,314)
Total common stockholders' equity	812,820	(5,314) 807,506
Not subject to mandatory redemption         Subject to mandatory redemption         Long-term debt	30,000 90,000 829,799	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$
	1,762,619	(3,514) 1,757,505
Current Liabilities         Short-term debt         Cumulative preferred stock redemption requirement         Long-term debt due within one year         Accounts payable         Taxes accrued         Interest accrued         Dividends declared         Current capital lease obligation         Accrued employee separation costs         Deferred income taxes, net         Other	$87,700 \\ 10,000 \\ 74,100 \\ 55,359 \\ 40,159 \\ 17,186 \\ 22,598 \\ \\ 2,683 \\ 2,438 \\ 34,501 \\ \hline 346,724$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
Deferred Credits and Other Liabilities Deferred income taxes, net Deferred investment tax credits Long-term capital lease obligations Postretirement obligations Other	428,309 47,211 37,006 57,812 570,338	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
Total Capitalization and Liabilities	\$2,679,681	<u>\$ (4,981)</u> <u>\$2,674,700</u>

The accompanying Notes to Consolidated Financial Statements are an integral part of this statement.

Ņ

ì

# DELMARVA POWER & LIGHT COMPANY CONSOLIDATED STATEMENT OF INCOME FOR THE NINE MONTHS ENDED SEPTEMBER 30, 1996 (Dollars in Thousands, Except Per Share Amounts)

(Unaudited)

Operating Revenues       \$752,415       \$752,415       \$752,415         Gas $821,164$ $821,164$ $821,164$ Gas $834,579$ $80$ $834,579$ Operating Expenses $834,579$ $834,579$ $834,579$ Electric fuel and purchased energy $244,593$ $244,593$ $244,593$ Gas purchased $43,844$ $43,844$ $43,844$ Purchased electric capacity $25,147$ $25,147$ $25,147$ Operation and maintenance $191,449$ $-191,449$ $-191,449$ Depreciation $91,435$ $91,435$ $91,435$ Income taxes $62,547$ $62,547$ $62,547$ Other taxes $32,218$ $32,218$ $32,218$ Operating Income $143,346$ $0$ $143,346$ Other Income and Expense $(39,105)$ $(39,105)$ $(39,105)$ Nonutility subsidiaries $3,089$ $0$ $3,089$ Allowance for equity funds used during construction $774$ $774$ $774$ Other income, net of income taxes $(1,202)$ $(2,085)$ $(2,085)$ </th <th></th> <th>Reported Amount</th> <th>Reclass Adjustments</th> <th>Adjusted Amount</th>		Reported Amount	Reclass Adjustments	Adjusted Amount
Gas       82,164       —       82,164         Baster       834,579       \$0       834,579         Operating Expenses       244,593       244,593         Electric fuel and purchased energy       244,593       244,593         Gas purchased       43,844       43,844         Purchased electric capacity       25,147       —       25,147         Operation and maintenance       191,449       —       191,449         Depreciation       91,435       —       91,435         Income taxes       62,547       —       62,547         Other taxes       32,218       —       32,218         Operating Income       143,346       0       143,346         Other Income and Expense       (39,105)       —       (39,105)         Nonutility subsidiaries       3,089       0       3,089         Nonucility subsidiaries       3,089       0       3,089         Allowance for equity funds used during construction       774       —       774         Other income, net of income taxes       (1,202)       (1,202)       (1,202)         Interest Charges       146,007       0       146,007         Interest expense       53,589       53,589	Operating Revenues		<b>`</b>	
Operating Expenses $834,579$ $\$0$ $834,579$ Electric fuel and purchased energy $244,593$ $244,593$ $244,593$ Gas purchased $43,844$ $-43,844$ $-43,844$ Purchased electric capacity $25,147$ $-25,147$ Operation and maintenance $191,449$ $-91,435$ $91,435$ Income taxes $62,547$ $-62,547$ $62,547$ Other taxes $32,218$ $-32,218$ $-32,213$ Operating Income $143,346$ $0$ $143,346$ Other Income and Expense $(39,105)$ $ (39,105)$ Nonutility subsidiaries $3,089$ $0$ $3,089$ Allowance for equity funds used during construction $774$ $-774$ Other income, net of income taxes $(1,202)$ $(1,202)$ $(2,661)$ Income Before Interest Charges $146,007$ $0$ $146,007$ Interest expense $53,589$ $-53,589$ $-53,589$ $-53,589$ Allowance for borrowed funds used during construction $(2,085)$ $-(2,085)$ $-(2,085)$ Net Income $94,503$	Electric	\$752,415	·	\$752,415
Operating Expenses       244,593       244,593         Electric fuel and purchased energy       244,593       244,593         Gas purchased       43,844       43,844       43,844         Purchased electric capacity       25,147       25,147         Operation and maintenance       191,449       191,449         Depreciation       91,435       91,435         Income taxes       62,547       62,547         Other taxes       32,218       32,218         Operating Income       143,346       0         Other taxes       32,218       0         Operating Income       143,346       0         Other taxes       32,218       32,218         Nonutility subsidiaries       30       0         Revenues and gains       42,194       -42,194         Expenses       (39,105)       (39,105)         Net earnings of nonutility subsidiaries       3,089       0       3,089         Allowance for equity funds used during construction       774       774         Other income, net of income taxes       (1,202)       (1,202)         Interest Charges       146,007       0       146,007         Interest Charges       146,007       0       146,00	Gas	82,164		82,164
Electric fuel and purchased energy $244,593$ $244,593$ $244,593$ Gas purchased $43,844$ $43,844$ $43,844$ Purchased electric capacity $25,147$ $-25,147$ Operation and maintenance $191,449$ $-191,449$ Depreciation $91,435$ $91,435$ Income taxes $62,547$ $62,547$ Other taxes $32,218$ $-32,218$ Operating Income $43,346$ $0$ Other taxes $32,218$ $-32,218$ Operating Income $43,346$ $0$ Other taxes $32,218$ $-32,218$ Nonutility subsidiaries $42,194$ $-42,194$ Nonutility subsidiaries $3,089$ $0$ Nonutility subsidiaries $3,089$ $0$ Nonutility subsidiaries $3,089$ $0$ Allowance for equity funds used during construction $774$ $-774$ Other income, net of income taxes $(1,202)$ $(1,202)$ Income Before Interest Charges $146,007$ $0$ Interest expense $53,589$ $53,589$ $53,589$ <td></td> <td>834,579</td> <td>\$0</td> <td>834,579</td>		834,579	\$0	834,579
Gas purchased       43,844       -       43,844         Purchased electric capacity       25,147       -       25,147         Operation and maintenance       191,449       -       191,449         Depreciation       91,435       -       91,435         Income taxes       62,547       -       62,547         Other taxes       32,218       -       32,218         Operating Income       143,346       0       143,346         Other Income and Expense       143,346       0       143,346         Nonutility subsidiaries       3,089       0       3,089         Nonutility subsidiaries       3,089       0       3,089         Allowance for equity funds used during construction       774       774       774         Other income, net of income taxes       (1,202)       (1,202)       (1,202)         Income Before Interest Charges       146,007       146,007       146,007         Interest expense       53,589       -       53,589       -       53,589         Allowance for borrowed funds used during construction       (2,085)       -       (2,085)         Interest expense       51,504       0       51,504       0       51,504 <td< td=""><td><b>Operating Expenses</b></td><td></td><td></td><td>11. 12.</td></td<>	<b>Operating Expenses</b>			11. 12.
Gas purchased       43,844       43,844         Purchased electric capacity       25,147       25,147         Operation and maintenance       191,449       191,449         Depreciation       91,435       91,435         Income taxes       62,547       62,547         Other taxes       32,218       32,218         Operating Income       143,346       0       143,346         Other Income and Expense       143,346       0       143,346         Nonutility subsidiaries       30,89       0       3,089         Nonutility subsidiaries       30,089       0       3,089         Allowance for equity funds used during construction       774       774       774         Other income, net of income taxes       (1,202)       (1,202)       (1,202)         Interest Charges       146,007       146,007       146,007         Interest expense       53,589       53,589       53,589         Allowance for borrowed funds used during construction       (2,085)       (2,085)         Nuterest Charges       146,007       146,007       146,007         Interest expense       53,589       53,589       53,589         Allowance for borrowed funds used during construction       (2,085) <td>Electric fuel and purchased energy</td> <td>244,593</td> <td></td> <td>244,593</td>	Electric fuel and purchased energy	244,593		244,593
Operation and maintenance       191,449       -       191,449         Depreciation       91,435       91,435         Income taxes       62,547       -       62,547         Other taxes       32,218       -       32,218 <b>Operating Income</b> 691,233       0       691,233 <b>Operating Income</b> 143,346       0       143,346 <b>Other Income and Expense</b> 143,346       0       143,346 <b>Nonutility subsidiaries</b> (39,105)       -       (39,105)         Net earnings of nonutility subsidiaries       3,089       0       3,089         Allowance for equity funds used during construction       774       -       774         Other income, net of income taxes       (1,202)       -       (1,202)         2,661       0       2,661       0       2,661         Income Before Interest Charges       146,007       0       146,007         Interest expense       53,589       -       53,589         Allowance for borrowed funds used during construction       (2,085)       -       (2,085)         Net Income       94,503       -       94,503       -       94,503         Dividends on preferred stock       7,2	Gas purchased	43,844	. <u></u>	43,844
Depreciation       91,435       91,435         Income taxes       62,547       62,547         Other taxes       32,218       32,218         Operating Income       143,346       0         Other Income and Expense       143,346       0         Nonutility subsidiaries       143,346       0         Revenues and gains       42,194       42,194         Expenses       (39,105)       (39,105)         Net earnings of nonutility subsidiaries       3,089       0         Allowance for equity funds used during construction       774       774         Other income, net of income taxes       (1,202)       (1,202)         Interest Charges       146,007       0       146,007         Interest Charges       146,007       0       146,007         Interest expense       53,589       53,589       53,589         Allowance for borrowed funds used during construction       (2,085)       (2,085)       (2,085)         Net Income       94,503       94,503       94,503       94,503         Dividends on preferred stock       7,293       7,293       7,293         Earnings Applicable to Common Stock       88,7210       \$0       \$ 87,210         Common Stock	Purchased electric capacity	25,147	·	25,147
Income taxes $62,547$ $ 62,547$ Other taxes $32,218$ $ 32,218$ $691,233$ $0$ $691,233$ $0$ $691,233$ <b>Operating Income</b> $143,346$ $0$ $143,346$ $0$ <b>Other Income and Expense</b> $143,346$ $0$ $143,346$ <b>Other Income and Expense</b> $(39,105)$ $ (39,105)$ Net earnings of nonutility subsidiaries $3,089$ $0$ $3,089$ Allowance for equity funds used during construction $774$ $ 774$ Other income, net of income taxes $(1,202)$ $ (1,202)$ Income Before Interest Charges $146,007$ $0$ $146,007$ Interest expense $53,589$ $ 53,589$ Allowance for borrowed funds used during construction $(2,085)$ $ (2,085)$ Interest expense $53,589$ $ 53,589$ $-$ Allowance for borrowed funds used during construction $(2,085)$ $ (2,085)$ Interest expense $51,504$ $0$ $51,504$ $0$ $51,504$		191,449	*	191,449
Other taxes $32,218$ — $32,218$ <b>691,233 0 691,233 Operating Income 143,346 0 143,346 Other Income and Expense 143,346 0 143,346 Nonutility subsidiaries 143,346 0 143,346 Nonutility subsidiaries</b> (39,105)       —       (39,105)         Net earnings of nonutility subsidiaries <b>3,089 0 3,089</b> Allowance for equity funds used during construction <b>774</b> — <b>774</b> Other income, net of income taxes       (1,202)       —       (1,202) <b>2,661 0 2,661 0 2,661 Income Before Interest Charges 146,007 0 146,007 Interest expense 53,589</b> — <b>53,589</b> Allowance for borrowed funds used during construction       (2,085)       —       (2,085) <b>Met Income 94,503</b> — <b>94,503</b> — <b>7,293</b> Dividends on preferred stock <b>7,293</b> — <b>7,293 7,293 Common Stock \$ 87,210 \$0</b>			· · · ·	91,435
<b>Operating Income</b> $691,233$ $0$ $691,233$ <b>Other Income and Expense</b> $143,346$ $0$ $143,346$ <b>Other Income and Expense</b> $143,346$ $0$ $143,346$ <b>Nonutility subsidiaries</b> $2,194$ $ 42,194$ Expenses $(39,105)$ $ (39,105)$ Net earnings of nonutility subsidiaries $3,089$ $0$ $3,089$ Allowance for equity funds used during construction $774$ $ 774$ Other income, net of income taxes $(1,202)$ $ (1,202)$ 2,661 $0$ $2,661$ $0$ $2,661$ <b>Income Before Interest Charges</b> $146,007$ $0$ $146,007$ Interest expense $53,589$ $ 53,589$ Allowance for borrowed funds used during construction $(2,085)$ $ (2,085)$ Met Income $94,503$ $ 94,503$ $ 94,503$ Dividends on preferred stock $7,293$ $ 7,293$ $ 7,293$ Earnings Applicable to Common Stock $$87,210$ $$0$ $$87,210$ <td>Income taxes</td> <td></td> <td></td> <td>62,547</td>	Income taxes			62,547
Operating Income $143,346$ 0 $143,346$ Other Income and ExpenseNonutility subsidiaries $42,194$ $ 42,194$ Nonutility subsidiaries $(39,105)$ $ (29,105)$ Net earnings of nonutility subsidiaries $3,089$ 0 $3,089$ Allowance for equity funds used during construction $774$ $ 774$ Other income, net of income taxes $(1,202)$ $(1,202)$ $(1,202)$ Income Before Interest Charges $146,007$ 0 $146,007$ Interest Charges $146,007$ 0 $146,007$ Interest Charges $53,589$ $ 53,589$ Allowance for borrowed funds used during construction $(2,085)$ $ (2,085)$ Net Income $94,503$ $ 94,503$ $-$ Dividends on preferred stock $7,293$ $ 7,293$ $-$ Common Stock $\$ 87,210$ $\$ 0$ $\$ 87,210$ $\$ 0$ Average shares outstanding (000) $60,709$ $ 60,709$ $=$ Earnings per average share $\$ 1.44$ $\$ 1.44$ $=$ $\$ 1.44$	Other taxes	32,218	·	32,218
Other Income and ExpenseNonutility subsidiariesRevenues and gainsRevenues and gainsAllowance for equity funds used during constructionRevenues and funds used during constructionRevenues and funds used during constructionRevenues and funds used during construction(2,085)Revenues and funds usedRevenues and fundsRevenues and funds </td <td></td> <td>691,233</td> <td>0</td> <td>691,233</td>		691,233	0	691,233
Nonutility subsidiaries $42,194$ $ 42,104$ $ 42,105$ $  14,007$ $ 146,007$ $ 146,007$ $ 146,007$ $ 146,007$ $ 146,007$ $ 146,007$ $ 146,007$ $ 146,007$ $ 146,007$ $ 146,007$ $ 146,007$ $ 146,007$ $ 146,007$ $ 146,007$ $-$ <th< td=""><td>Operating Income</td><td>143,346</td><td>. 0</td><td>143,346</td></th<>	Operating Income	143,346	. 0	143,346
Revenues and gains $42,194$ $ 42,194$ Expenses $(39,105)$ $ (39,105)$ Net earnings of nonutility subsidiaries $3,089$ $0$ $3,089$ Allowance for equity funds used during construction $774$ $ 774$ Other income, net of income taxes $(1,202)$ $ (1,202)$ 2,661 $0$ $2,661$ $0$ $2,661$ Income Before Interest Charges $146,007$ $0$ $146,007$ Interest Charges $53,589$ $ 53,589$ Allowance for borrowed funds used during construction $(2,085)$ $ (2,085)$ Met Income $94,503$ $ 94,503$ $ 94,503$ Dividends on preferred stock $7,293$ $ 7,293$ $ 7,293$ Earnings Applicable to Common Stock $\$87,210$ $\$0$ $\$87,210$ $\$0$ $\$8,72,10$ Common Stock $4verage$ shares outstanding $(000)$ $60,709$ $ 60,709$ $ 60,709$	Other Income and Expense		· ····	1
Expenses(39,105)—(39,105)Net earnings of nonutility subsidiaries $3,089$ 0 $3,089$ Allowance for equity funds used during construction $774$ — $774$ Other income, net of income taxes(1,202)—(1,202) $2,661$ 0 $2,661$ 0 $2,661$ Income Before Interest Charges146,0070146,007Interest Charges146,0070146,007Interest charges53,589—53,589Allowance for borrowed funds used during construction(2,085)—51,504051,5040Net Income94,503—94,503Dividends on preferred stock7,293—7,293Earnings Applicable to Common Stock\$ 87,210\$0\$ 87,210Common Stock60,709—60,709—Average shares outstanding (000)60,709—60,709Earnings per average share\$ 1.44—\$ 1.44		د د -		· ·
Net earnings of nonutility subsidiaries $3,089$ $0$ $3,089$ Allowance for equity funds used during construction $774$ $ 774$ Other income, net of income taxes $(1,202)$ $ (1,202)$ $2,661$ $0$ $2,661$ $0$ $2,661$ Income Before Interest Charges $146,007$ $0$ $146,007$ Interest Charges $146,007$ $0$ $146,007$ Interest expense $53,589$ $ 53,589$ Allowance for borrowed funds used during construction $(2,085)$ $ 51,504$ $0$ $51,504$ $0$ Net Income $94,503$ $ 94,503$ Dividends on preferred stock $7,293$ $ 7,293$ Earnings Applicable to Common Stock $\$ 87,210$ $\$0$ $\$ 87,210$ Common Stock $\$ 1.44$ $\$ 1.44$ $\$ 1.44$	Revenues and gains	42,194	· _ ·	42,194
Allowance for equity funds used during construction $774$ $ 774$ Other income, net of income taxes $(1,202)$ $ (1,202)$ $2,661$ $0$ $2,661$ Income Before Interest Charges $146,007$ $0$ Interest Charges $146,007$ $0$ Interest expense $53,589$ $-$ Allowance for borrowed funds used during construction $(2,085)$ $ 51,504$ $0$ $51,504$ $0$ Net Income $94,503$ $ 94,503$ Dividends on preferred stock $7,293$ $ 7,293$ Earnings Applicable to Common Stock $\$ 87,210$ $\$ 0$ $\$ 87,210$ Common Stock $\$ 87,210$ $\$ 0$ $\$ 87,210$ Average shares outstanding (000) $60,709$ $ 60,709$ Earnings per average share $\$ 1.44$ $ \$ 1.44$	Expenses	(39,105)		(39,105)
Allowance for equity funds used during construction $774$ $ 774$ Other income, net of income taxes $(1,202)$ $ (1,202)$ $2,661$ $0$ $2,661$ Income Before Interest Charges $146,007$ $0$ Interest Charges $146,007$ $0$ Interest expense $53,589$ $-$ Allowance for borrowed funds used during construction $(2,085)$ $ 51,504$ $0$ $51,504$ $0$ Net Income $94,503$ $ 94,503$ Dividends on preferred stock $7,293$ $ 7,293$ Earnings Applicable to Common Stock $\$ 87,210$ $\$ 0$ $\$ 87,210$ Common Stock $\$ 87,210$ $\$ 0$ $\$ 87,210$ Average shares outstanding (000) $60,709$ $ 60,709$ Earnings per average share $\$ 1.44$ $ \$ 1.44$	Net earnings of nonutility subsidiaries	3,089	0	3.089
Income Before Interest Charges $2,661$ $0$ $2,661$ Interest Charges $146,007$ $0$ $146,007$ Interest Charges $116,007$ $0$ $146,007$ Interest expense $53,589$ $ 53,589$ Allowance for borrowed funds used during construction $(2,085)$ $ (2,085)$ $ (2,085)$ $51,504$ $0$ $51,504$ Net Income $94,503$ $ 94,503$ $ 94,503$ Dividends on preferred stock $7,293$ $ 7,293$ $ 7,293$ Earnings Applicable to Common Stock $\$87,210$ $\$0$ Some Stock $\$7,210$ $\$0$ $\$87,210$ Average shares outstanding (000) $60,709$ $ 60,709$ Earnings per average share $\$1.44$ $\$1.44$		,	·	,
Income Before Interest Charges $2,661$ $0$ $2,661$ Interest Charges $146,007$ $0$ $146,007$ Interest Charges $116,007$ $0$ $146,007$ Interest expense $53,589$ $ 53,589$ Allowance for borrowed funds used during construction $(2,085)$ $ (2,085)$ $ (2,085)$ $51,504$ $0$ $51,504$ Net Income $94,503$ $ 94,503$ $ 94,503$ Dividends on preferred stock $7,293$ $ 7,293$ $ 7,293$ Earnings Applicable to Common Stock $\$87,210$ $\$0$ Some Stock $\$7,210$ $\$0$ $\$87,210$ Average shares outstanding (000) $60,709$ $ 60,709$ Earnings per average share $\$1.44$ $\$1.44$	Other income, net of income taxes	(1,202)	_	(1,202)
Interest Charges $53,589$ $53,589$ Interest expense $53,589$ $53,589$ Allowance for borrowed funds used during construction $(2,085)$ $(2,085)$ S1,504       0 $51,504$ 0         Net Income $94,503$ $ 94,503$ Dividends on preferred stock $7,293$ $ 7,293$ Earnings Applicable to Common Stock $\$87,210$ $\$0$ $\$87,210$ Common Stock $60,709$ $ 60,709$ Earnings per average share $\$1.44$ $\$1.44$		2,661	0	´
Interest Charges $53,589$ $53,589$ Interest expense $53,589$ $53,589$ Allowance for borrowed funds used during construction $(2,085)$ $(2,085)$ S1,504       0 $51,504$ 0         Net Income $94,503$ $ 94,503$ Dividends on preferred stock $7,293$ $ 7,293$ Earnings Applicable to Common Stock $\$87,210$ $\$0$ $\$87,210$ Common Stock $60,709$ $ 60,709$ Earnings per average share $\$1.44$ $\$1.44$	Income Before Interest Charges	146,007	. 0	146.007
Interest expense       53,589       -       53,589         Allowance for borrowed funds used during construction       (2,085)       -       (2,085)         S1,504       0       51,504       0       51,504         Net Income       94,503       -       94,503       94,503         Dividends on preferred stock       7,293       -       7,293       7,293         Earnings Applicable to Common Stock       \$87,210       \$0       \$87,210       \$0       \$87,210         Common Stock       60,709       -       60,709       60,709       60,709         Earnings per average share       \$1.44       \$1.44       \$1.44       \$1.44	• · · · · · · · · · · · · · · · · · · ·		,	
Allowance for borrowed funds used during construction       (2,085)       —       (2,085)         Net Income $51,504$ 0 $51,504$ 0 $51,504$ Net Income       94,503       —       94,503       —       94,503         Dividends on preferred stock       7,293       —       7,293       —       7,293         Earnings Applicable to Common Stock       \$87,210       \$0       \$87,210       \$0       \$87,210         Common Stock       60,709       —       60,709       —       60,709       —         Earnings per average share       \$1.44       \$1.44       \$1.44       \$1.44       \$1.44		53 589	<u> </u>	53,589
Net Income94,503 $-$ 94,503Dividends on preferred stock7,293 $-$ 7,293Earnings Applicable to Common Stock\$ 87,210\$0\$ 87,210Common Stock\$ 87,210\$0\$ 87,210Common Stock $60,709$ $ 60,709$ Earnings per average share\$ 1.44 $-$ \$ 1.44		,		
Dividends on preferred stock $7,293$ $ 7,293$ Earnings Applicable to Common Stock $$87,210$ $$0$ $$87,210$ Common Stock $$60,709$ $ 60,709$ Earnings per average share $$1.44$ $ $1.44$	n an	51,504	. 0	51,504
Dividends on preferred stock $7,293$ $ 7,293$ Earnings Applicable to Common Stock $$87,210$ $$0$ $$87,210$ Common Stock $$60,709$ $ 60,709$ Earnings per average share $$1.44$ $ $1.44$	Net Income	94,503		94 503
Earnings Applicable to Common Stock       \$ 87,210         Common Stock       \$ 0         Average shares outstanding (000)       60,709         Earnings per average share       \$ 1.44		•	· · ·	
Average shares outstanding (000)       60,709       60,709         Earnings per average share       \$1.44       \$1.44	Earnings Applicable to Common Stock	\$ 87,210	\$0	\$ 87,210
Average shares outstanding (000)       60,709       60,709         Earnings per average share       \$1.44       \$1.44				
Earnings per average share \$1.44 — \$1.44		60 709	·	60 700
	Earnings per average share	,		,
			<u> </u>	

The accompanying Notes to Consolidated Financial Statements are an integral part of this statement.

# ATLANTIC ENERGY, INC.

## CONSOLIDATED STATEMENT OF INCOME

## FOR THE NINE MONTHS ENDED SEPTEMBER 30, 1996 (Dollars in Thousands, Except Per Share Amounts) (Unaudited)

	Reported Amount	Reclass Adjustments	Adjusted. Amount
Operating Revenues		· .	
Electric	\$752,968		\$752,968
· · ·	752,968	\$ 0	752,968
Operating Expenses			· · · · · ·
Electric fuel and purchased energy	171,648		171,648
Purchased electric capacity	146,877		146,877
Operation and maintenance	143,054	—	143,054
Depreciation and amortization	60,490	`~	60,490
State excise taxes	80,391		80,391
Federal income taxes	30,842	<u> </u>	30,842
Other taxes	7,656	—	7,656
	640,958	0	640,958
Operating Income	112,010	0	112,010
Other Income and Expense			. ,
Nonutility subsidiaries		· , ·	
Revenues and gains	—	10,573 (14)	
Expenses		(10,033)(14)	(10,033)
Net earnings of nonutility subsidiaries		540	540
Allowance for equity funds used during construction	697	,—	697
Other income, net of income taxes	2,603	(540)(14)	2,063
	3,300	0	3,300
Income Before Interest Charges	115,310	0	115,310
Interest Charges	*	i.	ĥ
Interest expense	49,303	—	49,303
Allowance for borrowed funds used during construction	(820)		(820)
	48,483	0	48,483
Less preferred stock dividend requirements of subsidiary	8,475		8,475
Net Income	\$ 58,352	\$ 0	\$ 58,352
Common Stock	· · · · · · · · · · · · · · · · · · ·	,	
Average shares outstanding (000)	52,702		52,702
Earnings per average share	\$1.11		\$1.11
- Dividends declared	\$1.155	<u> </u>	\$1.155

The accompanying Notes to Consolidated Financial Statements are an integral part of this statement.

# **DELMARVA POWER & LIGHT COMPANY** CONSOLIDATED STATEMENT OF INCOME FOR THE YEAR ENDED DECEMBER 31, 1995 (Dollars in Thousands, Except Per Share Amounts)

· · · · · · · · · · · · · · · · · · ·	Reported Amount	Reclass Adjustments	Adjusted Amount
Operating Revenues		1	
Electric	\$899,662	_	\$899,662
Gas	95,441		95,441
	995,103	<b>\$</b> . 0	995,103
Operating Expenses	,	·	
Electric fuel and purchased energy	267,885	<u> </u>	267,885
Gas purchased	48,615	<u> </u>	48,615
Purchased electric capacity		29,116(15)	29,116
Operation and maintenance	275,165	(29,116)(15)	246,049
Depreciation	113,022	<u> </u>	113,022
Income taxes	73,561	—	73,561
Other taxes			38,449
	816,697	2 <u></u>	.816,697
Operating Income	178,406	0	178,406
Other Income and Expense			
Nonutility subsidiaries	•		
Revenues and gains	52,042		52,042
Expenses	<u>(47,896)</u>	· · · ·	<u>(47,896)</u>
Net earnings of nonutility subsidiaries	4,146		4,146
Allowance for equity funds used during construction	708	—	708
Other income, net of income taxes	557	· · · · · · · · · · · · · · · · · · ·	557
	5,411	0	5,411
Income Before Interest Charges	183,817	0	183,817
Interest Charges			
Interest expense	68,395		68,395
Allowance for borrowed funds used during construction	(2,066)	<del></del>	(2,066)
	66,329	0	66,329
Net Income	117,488	0	117,488
Dividends on preferred stock	9,942		9,942
Earnings Applicable to Common Stock	\$107,546	\$ 0	\$107,546
Common Stock			•
Average shares outstanding (000)	60,217	•	60,217
Earnings per average share	\$1.79	_	\$1.79
Dividends declared	\$1.54	•	\$1.54

The accompanying Notes to Consolidated Financial Statements are an integral part of this statement.

# ATLANTIC ENERGY, INC.

# CONSOLIDATED STATEMENT OF INCOME

# FOR THE YEAR ENDED DECEMBER 31, 1995

(Dollars in Thousands, Except Per Share Amounts)

	Reported Amount	Reclass Adjustments	Adjusted Amount
Operating Revenues			
Electric	\$953,137		\$953,137
	953,137	\$ 0	953,137
Operating Expenses			
Electric fuel and purchased energy	191,766		191,766
Purchased electric capacity	190,570	_	190,570
Operation and maintenance	186,439	—	186,439
Depreciation and amortization	78,461	—	78,461
State excise taxes	102,811	—	102,811
Federal income taxes	45,876		45,876
Other taxes	8,677		8,677
· · · ·	804,600	0	804,600
Operating Income	148,537	0	148,537
Other Income and Expense Nonutility subsidiaries			() ( )25
Revenues and gains	_	6,925 (1 (7,680)(1	
Net earnings of nonutility subsidiaries		(755)	(755)
Allowance for equity funds used during construction	817		817
Other income, net of income taxes	8,241	755 (1	4) 8,996
	9,058	· 0	9,058
Income Before Interest Charges	157,595	0	157,595
Interest Charges			
Interest expense	62,879		62,879
Allowance for borrowed funds used during construction	(1,679)		(1,679)
	61,200	0	61,200
Less preferred stock dividend requirements of subsidiary	14,627		14,627
Net Income	\$ 81,768	\$ 0	\$ 81,768
Common Stock			
Average shares outstanding (000)	52,815		52,815
Earnings per average share	\$1.55		\$1.55
Dividends declared	\$1.54		\$1.54

The accompanying Notes to Consolidated Financial Statements are an integral part of this statement.

. ' )

۰,

.

1

## NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

- (1) Transfer goodwill from Electric plant to "Cost in excess of net assets acquired, net."
- (2) Transfer capital leases, net to "Leased property, net."
- (3) Transfer regulatory assets from "Other" to "Other regulatory assets."
- (4) Transfer "Land held for future use" to "Electric utility plant in service."
- (5) Transfer \$12,645 for Investment in Bond Escrow Trust from "Other investments" to "Funds held by trustee."
- (6) Transfer "Nonutility property and equipment-net" to "Other investments."
- (7) Transfer "Working funds" to "Cash" and to "Materials & supplies", as appropriate.
- (8) Transfer "Unbilled revenues" and "Allowance for doubtful accounts" to "Accounts receivable Utility service."
- (9) Transfer prepaid pension cost to a separate line.
- (10) Transfer unamortized debt costs from "Deferred debt refinancing costs" to "Unamortized debt expense."
- (11) Transfer unearned compensation from other deferred debits to a separate deduction from commonequity.
- (12) Transfer "Accrued employee separation costs" to "Other current liabilities" and transfer current capital lease obligation from "Other current liabilities" to a separate line.
- (13) Transfer other postretirement benefits from "Other deferred credits" to a separate line.
- (14) Transfer "Net earnings of nonutility subsidiaries" from "Other income" to a separate line.
- (15) Transfer purchased electric capacity from "Operation and maintenance" to a separate line.

To the Board of Directors and Stockholders Conectiv, Inc. Wilmington, Delaware

We have audited the accompanying consolidated balance sheet of Conectiv, Inc. (formerly DS, Inc.) as of September 30, 1996. This consolidated financial statement is the responsibility of the Company's management. Our responsibility is to express an opinion on this consolidated financial statement based on our audit.

We conducted our audit in accordance with generally accepted auditing standards. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the consolidated balance sheet is free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the consolidated balance sheet. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall consolidated balance sheet presentation. We believe that our audit provides a reasonable basis for our opinion.

In our opinion, the consolidated balance sheet referred to above presents fairly, in all material respects, the consolidated financial position of Conectiv, Inc. as of September 30, 1996, in conformity with generally accepted accounting principles.

COOPERS & LYBRAND L.L.P.

2400 Eleven Penn Center Philadelphia, Pennsylvania October 1, 1996

## CONECTIV, INC.

# CONSOLIDATED BALANCE SHEET **SEPTEMBER 30, 1996**

	х. ,	ASSETS				
Cash	••••••••••••	ABBEIS	· · · · · · · · · · · · · · · · · · ·	n and a sign and a sign and a sign	· · · · · · · · · ·	\$ 50
25 - A - A - A - A - A - A - A - A - A -	·	· · · ·		Merida a Ag		
	LIABILITIES AN	ND STOCKHOI	LDERS' EQU	ITY		
Liabilities Stockholders' equity Common stock, \$0,0 1,000 shares author Additional paid-in ca	1 par value (Note prized; 1,000 share	1): s issued and ou	tstanding	n an an Arthon an S Airtín Staite an Anna An An Staite an Anna Anna An	۳ ۲	10
	s and stockholders	s' equity	· · · · · · · · · · · · · · · · · · ·		· · · · · · · · · · · · · · · · · · ·	<u>\$ 50</u>

Sector and the sector of the and the manufacture of the second states and the second states and the second states and the second states and the star was the start of the start الميكن المراجعين في المراجع ال 1- المراجع المر 1- مراجع المراجع المراجع

All a compared · · . . .

.

139

, .÷

## CONECTIV, INC.

# NOTE TO CONSOLIDATED FINANCIAL STATEMENT SEPTEMBER 30, 1996

#### 1. Formation and Organization (information subsequent to October 1, 1996 is unaudited)

Conectiv, Inc. (the "Company") was incorporated under the laws of the State of Delaware on August 8, 1996 as DS, Inc. with 50% of its outstanding \$0.01 par value common stock owned by Delmarva Power & Light Company ("Delmarva") and 50% owned by Atlantic Energy, Inc. ("Atlantic"). DS, Inc.'s name was changed to Conectiv, Inc. on December 24, 1996.

Delmarva and Atlantic entered into the Merger Agreement, dated as of August 9, 1996 and amended and restated as of December 26, 1996, providing for a business combination of Delmarva and Atlantic as peer firms in a merger of equals (the "Transaction"). As a result of the Transaction, the Company will become the parent company of Delmarva and its direct and indirect subsidiaries and the direct and indirect subsidiaries of Atlantic, including Atlantic City Electric Company, which currently is Atlantic's regulated utility subsidiary. The Company will be a holding company registered under the Public Utility Holding Company Act of 1935, as amended. The Transaction is expected to close shortly after all of the conditions to the consummation of the Transaction, including obtaining applicable regulatory approvals, are met or waived. The regulatory approval process is expected to take approximately 12 to 18 months from the date of the Merger Agreement.

Under the terms of the Merger Agreement, the Company's subsidiary, DS Sub, Inc., will be merged with and into Delmarva with Delmarva being the surviving corporation in the merger (the "Delmarva Merger"). Simultaneously, Atlantic will be merged with and into the Company with the Company being the surviving corporation in the merger (the "Atlantic Merger"). Each share of the Company Common Stock issued and outstanding immediately prior to the effective time of the Mergers will be canceled, and no consideration shall be delivered in exchange for such stock.

Each outstanding share of Delmarva Common Stock, par value \$2.25 per share, will be exchanged for one share of the Company Common Stock, par value \$0.01 per share. Each share of Atlantic Common Stock will be exchanged for 0.75 of one share of the Company Common Stock and 0.125 of one share of Class A Common Stock, par value \$0.01 per share. [For more information, see Note (a) of the Notes To Unaudited Pro Forma Combined Financial Statements.]

The corporate headquarters of the Company will be located in Wilmington, Delaware, with a significant presence in New Jersey. The Board of Directors of the Company after the Mergers will consist of 18 directors, ten of whom will be nominated by Delmarva and eight of whom will be nominated by Atlantic.

#### SELECTED INFORMATION CONCERNING DELMARVA AND ATLANTIC

The following is a brief description of Delmarva and Atlantic and their respective businesses. This information does not purport to be complete. For information that is complete in all material respects, the reader should refer to the additional information contained in the Delmarva 1995 Form 10-K, Delmarva 1996 Forms 10-Q and Atlantic 1995 Form 10-K and Atlantic 1996 Forms 10-Q which are incorporated herein by reference. See "Incorporation by Reference."

## **Business of Delmarva**

Delmarva is predominantly a public utility that provides electric and gas service. Delmarva provides electric service to approximately 437,500 retail and wholesale customers in an area encompassing about 6,000 square miles in Delaware, ten primarily Eastern Shore counties in Maryland, and the Eastern Shore area of Virginia. Delmarva provides gas service to approximately 98,000 retail and transportation customers in an area consisting of about 275 square miles in northern Delaware, including the City of Wilmington.

Delmarva's wholly owned subsidiaries are engaged in nonutility activities and represent approximately 4% of the consolidated book value of Delmarva and its subsidiaries as of June 30, 1996. The direct subsidiaries include Delmarva Energy Company, Delmarva Industries, Inc., Delmarva Services Company, Delmarva Capital Investments, Inc. and Service Confidence, Inc. For more information on the subsidiaries, see the Delmarva 1995 Form 10-K, which is incorporated herein by reference.

Information concerning the names, ages, positions and business experiences of the executive officers and directors of Delmarva, along with additional information, including executive compensation, security ownership of certain beneficial owners and management, and certain relationships and related transactions is incorporated by reference herein from Items 1, 10, 11, 12 and 13 of the Delmarva 1995 Form 10-K (which incorporates by reference portions of Delmarva's Definitive Proxy Statement for its Annual Meeting of Stockholders held on May 30, 1996).

#### Business of Atlantic

Atlantic is a public utility holding company that is exempt under Section 3(a)(1) of the 1935 Act pursuant to Rule 2 thereunder. Atlantic has three wholly owned subsidiaries, ACE, Atlantic Energy Enterprises, Inc., a New Jersey corporation ("AEE"), and Atlantic Energy International, Inc., a Delaware corporation ("AEII"). ACE is a public utility holding company that is exempt under Section 3(a)(2) of the 1935 Act pursuant to Rule 2 thereunder. ACE is primarily engaged in the generation, transmission, distribution and sale of electric energy to over 473,000 customers in an area consisting of 2,700 square miles in southern New Jersey. ACE is Atlantic's principal subsidiary.

AEE is a holding company that manages investments in several nonutility subsidiaries, including Atlantic Generation, Inc., Atlantic Southern Properties, Inc., ATE Investment, Inc., Atlantic Thermal Systems, Inc., CoastalComm, Inc. and Atlantic Energy Technology, Inc. For more information about these subsidiaries, see the Atlantic 1995 Form 10-K, which is incorporated herein by reference. AEII was formed in July, 1996 to provide utility consulting services and equipment sales to international markets.

Information about the names, ages, positions and business experience of the executive officers and directors of Atlantic, along with additional information, including executive compensation, security ownership of certain beneficial owners and management, and certain relationships and related transactions is incorporated by reference herein from Items 1, 10, 11, 12 and 13 of Atlantic's 1995 Form 10-K (which incorporates by reference portions of its Definitive Proxy Statement for its 1996 Annual Meeting of Stockholders held on April 24, 1996).

#### Prior Relationship Between Delmarva and Atlantic

Delmarva and Atlantic, as owners of adjacent electric utility systems, have had a long history of close operational coordination and participation in joint projects. As members of the Pennsylvania-New Jersey-Maryland Interconnection Association, the generation and transmission facilities of both companies are operated on an integrated basis with those of six other utilities in Pennsylvania, New Jersey, Maryland and the District of Columbia. In addition, Delmarva and Atlantic have common ownership interests in Peach Bottom and Salem nuclear generation stations, Keystone and Conemaugh coal-fired generation stations and certain transmission facilities.

#### Salem Nuclear Generating Station

After consummation of the Mergers, the Company will own 14.82% of Salem Nuclear Generating Station ("Salem"), which consists of two pressurized water nuclear reactors ("PWR") and is operated by Public Service Electric & Gas Company ("PSE&G"). As of September 30, 1996, the Company's pro forma net investment in plant in service for Salem was approximately \$230 million. Salem represents approximately 4% of the Company's pro forma total assets and approximately 6% of the Company's pro forma installed electric generating capacity.

Salem Units 1 and 2 were removed from operation by PSE&G on May 16, 1995 and June 7, 1995, respectively, due to operational problems and maintenance concerns. Their return dates are subject to completion of the requirements of their respective restart plans to the satisfaction of PSE&G and the NRC, which encompasses a substantial review and improvement of personnel, process and equipment issues.

With respect to Unit 1, PSE&G informed Delmarva and Atlantic in early 1996 that inspections of the steam generators using a new testing technology indicated degradation in a significant number of tubes. After evaluating several options, in May 1996 replacement steam generators from the unfinished Seabrook Unit 2 nuclear power plant in New Hampshire were purchased from Northeast Utilities Service Compan for installation in Salem Unit 1. The replacement steam generators arrived on site in October 1996 and are scheduled for installation by early 1997. By using these steam generators, PSE&G expects to return Unit 1 to service in mid-1997. The Company's pro forma share of the costs to be capitalized for the steam generators, including installation, will range from approximately \$22 million to \$26 million.

With respect to Unit 2, PSE&G also informed Delmarva and Atlantic in early 1996 that inspections of the steam generators using the new testing technology confirmed that the condition of the generators is within current repair limits. On July 22, 1996, PSE&G announced that although substantial progress has been made, the outage at Unit 2 would continue well into the fourth quarter of 1996. PSE&G recently advised the Company that Unit 2 currently is expected to return to service early in the first quarter of 1997.

For the nine-month period ended September 30, 1996, the Company incurred and expensed higher than expected pro forma operation and maintenance costs at Salem of approximately \$16 million.

The Company incurs pro forma replacement power costs while the units are out of service of approximately \$1,450,000 per month, per unit. Such amounts vary, however, depending on the cost and availability of other Company owned generation and the cost of purchased energy. Replacement power costs typically are not incurred for routine refueling and maintenance outages, and the recovery of replacement power costs is subject to approval by the regulatory commissions having jurisdiction over Delmarva and Atlantic.

For Delmarva, from the inception of the Salem unit outages through September 30, 1996, approximately one-half of the current estimated replacement power costs of \$18 million has been expensed and the remaining portion has been deferred on the Company's Consolidated Balance Sheet in expectation of future recovery. For Atlantic, a \$12 million provision for revenue credits to customers has been recorded in accordance with a stipulation agreement dated October 22, 1996, as discussed below.

The actual costs to be incurred by the Company may vary from the foregoing estimates, since the periods projected by PSE&G during which the Salem units will be out of service, the extent of the maintenance that will be required, and the costs of replacement power and the extent of its recovery may be different from those set forth above.

In May 1996, Delmarva filed an application with the VSCC for increased fuel rates effective July 1996. In June 1996, Delmarva filed an application with the MPSC for increased fuel rates effective August 1996. In both filings, the Delmarva proposed that one-half of the replacement power costs associated with the Salem outage be permitted on an interim basis until a full review of the outage is made at a future time. The VSCC and MPSC approved Delmarva's filings, with rates subject to refund.

In October 1996, Delmarva filed a proposal with the DPSC to address the recovery of replacement power costs and other fuel costs. In that filing, Delmarva asserted a belief that it should be permitted to recover all replacement power costs associated with this outage, and requested an interim treatment that would permit recovery of approximately 50% of replacement power costs, beginning in January 1997, until the DPSC renders a final decision on this issue. Delmarva's filing also provides that any proceeds from litigation concerning replacement power costs will be credited to customers, net of litigation costs, if full recovery is authorized by the DPSC. On December 10, 1996, the DPSC suspended the portion of the interim rates related to the Salem replacement power costs until the earlier of June 1, 1997 or the end of the case. If the suspended interim rates go into effect prior to the conclusion of the case, they would go into effect subject to refund pending the final decision.

With regard to Atlantic's regulatory treatment of the Salem outage, the NJBPU by an order dated March 14, 1996, initiated an investigation of the ongoing outage at Salem. By its order the NJBPU declared the base rates associated with ownership in Salem Unit 1 interim and subject to refund pending a earing as to whether Salem Unit 1 is currently used and useful. The NJBPU also, in an order dated June 6, 1996, declared the base rates associated with ownership in Salem Unit 2 interim and subject to refund. The NJBPU voted on July 31, 1996 to include Unit 2 in the hearings scheduled for October 1996 to determine if both units were still considered used and useful. On October 22, 1996, the New Jersey Ratepayer Advocate and the staff of the NJBPU signed a stipulation settling the ongoing outage of Salem. Under the terms of the stipulation, Atlantic will provide credits to customers totaling \$12 million. The credits will be made during January and February 1997 and will be based on customer usage between January 1, 1996 and October 31, 1996. The stipulation also provides that replacement power costs incurred, up to the agreed upon return-to-service dates (June 30, 1997 for Unit 1 and December 31, 1996 for Unit 2), will be recoverable in the next annual levelized energy clause revenue proceeding. Should either unit not return to service by its agreed upon return-to-service date, replacement power costs incurred after such dates will not be recoverable by Atlantic. The performance of the Salem Units will not be included in the calculation of the Nuclear Performance Standard for the period each unit was taken out of service to each unit's respective return-to-service date. As such, Atlantic will not be subject to a penalty or reward under the Nuclear Performance Standard for 1995 or 1996. In regard to the litigation pending between Atlantic and PSE&G, discussed below, the stipulation provides that the first \$8 million of any net recovery, if any, by Atlantic as the result of the litigation (the gross amount of recovery less litigation expenses) will not be subject to any claim on behalf of the customers. Also, any settlement amounts received for damages incurred after the agreed upon return-to-service dates will not be subject to refund to customers, since Atlantic will bear the costs of all replacement power costs incurred.

The above described stipulation has been submitted to the NJBPU for acceptance and approval. A public hearing was held to review the stipulation on November 25, 1996 and take public comments on this matter. On December 18, 1996, the NJBPU approved the stipulation of settlement among ACE, the New Jersey Division of the Ratepayer Advocate, and the Staff of the NJBPU.

Notwithstanding current discussions with regulators concerning deregulation of the generation portion of the business, management of each of Delmarva and Atlantic believes it is reasonable to assume that rates will be set at levels that will recover the current and anticipated costs, including the costs needed to return the Salem units to operating status, of the joint investments in the Salem plant, and such rates can be charged to and collected from customers.

On February 27, 1996, the co-owners of Salem, including Delmarva and Atlantic, filed a complaint in the United States District Court for the District of New Jersey against Westinghouse Electric Corporation ("Westinghouse"), the designer and manufacturer of the Salem steam generators. The complaint, which seeks to recover from Westinghouse the costs associated with and resulting from the cracks discovered in Salem's steam generators and with replacing such steam generators, alleges violations of federal and New Jersey Racketeer Influenced and Corrupt Organizations Acts, fraud, negligent misrepresentation and breach of contract. The estimated replacement cost of such generators is between \$150 million and \$170 million. The Salem co-owners contend that the recently discovered degradation of the steam generators will prevent the steam generators from operating for a design life of 40 years. The lawsuit asserts that the Salem steam generators require replacement and these costs should be borne by Westinghouse and not the customers and shareholders of the Salem co-owners. Westinghouse filed an answer and a \$2.5 million counterclaim for unpaid work on April 30, 1996. On June 17, 1996, the court ordered the parties to mediate their claims rather than proceeding to litigation, taking the position that Westinghouse's involvement in steam generator lawsuits throughout the county, involving substantially similar issues as are involved in the Salem litigation, should enable the parties to resolve their dispute efficiently in mediation. Mediation is nonbinding on the parties and for the purpose of enabling the mediator to evaluate the parties' respective positions and the parties' settlement discussions. The mediation occurred on October 29 and 30, 1996. The parties were unable to resolve their claims in the mediation. It is expected that the parties will proceed to litigation. Delmarva and Atlantic cannot predict the outcome of this lawsuit.

On March 5, 1996, Delmarva and PECO Energy Company ("PECO") filed a complaint in the United States District Court for the Eastern District of Pennsylvania against Public Service Enterprise Group, Ind ("Enterprise") and PSE&G. On the same day, Atlantic filed a complaint in Superior Court of New Jerses against Enterprise and PSE&G. The lawsuits allege that the defendants failed to heed numerous citations, warnings, notices of violations and fines by the NRC as well as repeated warnings from the Institute of Nuclear Power Operations about performance, safety and management problems at Salem and to take appropriate corrective action. The suits contend that as a result of these actions and omissions, the Salem units were forced to shut down in 1995. The suits ask for compensatory damages for breach of contract, negligence and punitive damages, in amounts to be specified. Delmarva and Atlantic cannot predict the outcome of these lawsuits.

PSE&G has informed Delmarva and Atlantic that in August 1996, the NRC conducted an inspection of the Physical Security Program for Salem and Hope Creek and identified six apparent violations that are being considered for escalated enforcement. These apparent violations include the failure to: (1) control photo badge key cards; (2) properly search an individual prior to entrance to the protected area; (3) notify the nuclear shift supervisor of a potential threat event; (4) deactivate photo badges for individuals who no longer require site access; (5) complete training for security supervisors prior to assignment of duties; and (6) test an intrusion detection system in accordance with procedures. On September 3, 1996, PSE&G met with the NRC to discuss these issues and provide specific corrective actions. On November 14, 1996, a predecisional enforcement conference was held to address these apparent violations. At the conference, PSE&G presented their corrective actions, including a change in security management. On December 11, 1996, the NRC issued its written report to PSE&G. Based on the NRC's review of the inspection findings and information provided during the enforcement conference, PSE&G was cited for the aforementioned six violations and a civil penalty of \$100,000 was imposed. The Company's pro forma combined share of the penalty is \$14,820. Delmarva and Atlantic cannot predict what other actions the NRC may take on this matter.

## THE COMPANY FOLLOWING THE MERGERS

#### Management of the Company

The Delmarva Board will be entitled to nominate ten members to serve on the Company Board and the Atlantic Board will be entitled to nominate eight members to serve on the Company Board. The Delmarva Board and the Atlantic Board will each take all action necessary to cause each member of the Delmarva Board and each member of the Atlantic Board serving in such capacity immediately prior to the Effective Time to have the opportunity to serve as a member of the Company Board. The Company Board will be divided into three classes so that each class, to the extent possible, has the same proportion of directors nominated by each of the Delmarva Board and the Atlantic Board. In addition, at the consummation of the Mergers, the Audit Committee of the Company Board will consist of an equal number of directors nominated by the Delmarva Board and the Atlantic Board.

At the consummation of the Mergers, Howard E. Cosgrove will be the Chief Executive Officer and Chairman of the Company Board, Jerrold L. Jacobs (who will retire from active employment after the consummation of the Mergers) will be Vice Chairman of the Company Board and Michael J. Chesser will be the President and Chief Operating Officer of the Company. Jerrold L. Jacobs will serve as Vice Chairman of the Company Board until the second anniversary of the consummation of the Mergers and during his term as Vice Chairman, will be a member of the Executive Committee of the Company Board.

Following the consummation of the Mergers, the Audit Committee of the Company Board will be charged with the responsibility of advising the Company Board with respect to certain intercompany transactions and other fiduciary matters that may relate to the Class A Common Stock.

Under Delaware law, the Company Board has a duty to act with due care and in the best interests of all of the Company's stockholders, including the holders of Company Common Stock and Class A Common Stock. The existence of the Company Common Stock and the Class A Common Stock may give ise to occasions when the interests of the holders of Company Common Stock and Class A Common tock may diverge or appear to diverge. Examples include, among others, determinations by the Company Board to (i) convert each outstanding share of Class A Common Stock into shares of Company Common Stock, (ii) approve the disposition of all or substantially all of the properties and assets of the Atlantic Utility Group, (iii), allocate resources and financial support to or pursue business opportunities or operational strategies through one Group instead of the other Group, (iv) allocate the proceeds of issuances of Class A Common Stock either to the Company in a reduction in the Intergroup Interest or to the Atlantic Utility Group, (v) pay or omit to pay dividends on Company Common Stock or Class A Common Stock or (vi) approve transactions involving the transfer of funds or assets from one Group to the other Group or make other operational or financial decisions with respect to one Group that could be considered to be detrimental to the other Group. In such instances, the Company Board will be required to act on behalf of the Company and its stockholders taken as a whole. It is anticipated that the regulatory environment in which Delmarva and ACE will be conducting their respective operations following the consummation of the Mergers will help to ensure dealings between the Groups and among the Company and the Groups will be appropriate under the foregoing standard. For this reason, and because Delmarva and ACE have not as yet operated as a combined enterprise, Delmarva and Atlantic have determined to defer the adoption of management policies with respect to cash management, corporate expenses, allocations of assets and liabilities and inter-Group transactions, except to provide that the Audit Committee of the Company Board will advise the Company Board with respect to certain intercompany transactions and other fiduciary matters that may relate to the Class A Common Stock of the Company, At and after the consummation of the Mergers, the Company Board will exercise from time to time its judgment, as to how best to obtain information regarding the divergence (or potential divergence) of interests, under what circumstances to seek the assistance of outside advisers and how to assess which available alternative is in the best interests of the Company and all of its stockholders. The Company

Board will evaluate the need for and establish such management policies, if any, as it deems appropriate in its sole discretion.

The Company and its subsidiaries and affiliates will be subject to extensive federal and state regulation governing dealings among their utility and nonutility operations. Accordingly, any management policies adopted by the Company Board must adhere to any procedural, substantive, record keeping, accounting and other requirements imposed by such regulations. Such regulations may include the following:

1935 Act. The Company will become a registered holding company under the 1935 Act. The 1935 Act includes a number of restrictions on dealings between registered holding companies and their subsidiaries.

Section 12(a) of the 1935 Act prohibits a registered holding company from borrowing from any of its subsidiaries or from receiving from them any extension or credit or indemnity. Section 12(b) of the 1935 Act prohibits a registered holding company and its subsidiaries from lending or in any manner extending its credit to or indemnifying any company in the same holding company system except as authorized by the SEC.

Under Section 13(a) of the 1935 Act, registered holding companies are prohibited from entering into service, sales or construction contracts with an associated utility. Section 2(a)(19) of the 1935 Act defines "service contract" to include any "contract, agreement, or understanding whereby a person undertakes to sell or furnish, for a charge, any managerial, financial, legal, engineering, purchasing, marketing, auditing, statistical, advertising, publicity, tax, research, or any other service, information, or data." Although the restrictions of Section 13(a) of the 1935 Act do not apply to transactions involving special or unusual circumstances or not in the ordinary course of business if the SEC exempts them, no exemptions generally applicable to registered holding companies have been granted by the SEC to date.

Under Section 13(b) of the 1935 Act, no subsidiary of a registered holding company may render an services to any associate (subsidiary) company except pursuant to rules or order of the Commission "t insure that such contracts are performed economically and efficiently for the benefit of such associate companies at cost, fairly and equitably allocated among such companies." Rule 91 issued by the SEC under the 1935 Act provides that "a transaction shall be deemed to be performed at not more than cost if the price (taking into account all charges) does not exceed a fair and equitable allocation of expenses (including the price paid for goods) plus reasonable compensation for necessary capital procured through the issuance of capital stock (or similar securities of an unincorporated company)." Rule 91(b) specifies standards for allocating direct charges and other elements of cost, including taxes, interest, other overhead and compensation for the use of capital procured by the issuance of capital stock (or similar securities of an unincorporated company), on a fair and equitable basis. Rule 93 requires the service company to file an annual report with the SEC.

*Federal Power Act.* Under Section 203 of the Federal Power Act, no public utility may sell, lease or otherwise dispose of any part of its Federal Power Act-jurisdictional facilities having a value in excess of \$50,000 without first having secured an order of the FERC authorizing it to do so. The application for such an order must set forth the fact relied upon to show that the proposed disposition will be consistent with the public interest.

*New Jersey.* Under Section 48:3-7.1 of the New Jersey Statutes, no management, advisory service, construction or engineering contract of a public utility with any of its affiliates involving the expenditure of more than \$25,000 by the public utility is valid or effective until approved by the NJBPU, which is required to disapprove any such contract whose price exceeds the fair price for the property, work or services provided or that is contrary to the public interest. Under Section 48:3-7.2 of the New Jersey Statutes, no public utility may loan money or property to any affiliate without the prior approval of the NJBPU. Section 48:3-7.8 of the New Jersey Statutes requires every public utility to provide the NJBPU access to all of its\_

records, books and accounts. The NJBPU has required ACE to adopt certain cost allocation procedures for affiliate transactions. Furthermore, under Sections 48:2-29.1 and 48:2-29.2 of the New Jersey Statutes, the NJBPU may prohibit or limit the payment of dividends by a public utility to insure compliance with orders issued by the NJBPU and with the requirements of New Jersey corporate law.

*Delaware.* Under Section 215 of the Delaware Public Utilities Act, no public utility may dispose of any essential part of its property, or issue any stock or indebtedness payable in more than one year, without the prior approval of the DPSC. Section 207 of the Delaware Public Utilities Act provides that the DPSC shall have access to all books, accounts and records of public utilities. Under DPSC standards, in order for a public utility to have a transaction expense with an affiliated entity allowed for purposes of fixing public utility rates, the expense must be reasonable in that it must be the product of fair dealing and at a fair price.

*Maryland.* Under Section 56 of the Maryland Public Service Commission Law, the MPSC is granted general authority to supervise and regulate public utilities operating in the State of Maryland to assure their operation in the interest of the public and to promote adequate, economical and efficient delivery of utility services.

*Virginia.* Section 56-77 of the Code of Virginia requires the approval of or exemption by the VSCC for all contracts and arrangements for the furnishing of management, supervisory, construction, engineering, accounting, legal, financial or similar services or for the purchase, sale, lease or exchange of any other property, right or thing between a public service company and an affiliate. An application requesting approval or exemption of every such contract or arrangement must be filed with the VSCC regardless of the amount involved. Under Section 56-77 of the Code of Virginia, any such proposed transaction may be excluded in whole or in part from the accounts of a public service company in any proceeding involving its rates or practices if the transaction is not consistent with the public interest, and any payment or compensation may be disapproved or disallowed by the VSCC unless satisfactory proof is submitted thereto of the cost to the affiliated interest rendering the service or furnishing the property or service in question. Also, under Section 56-80 of the Code of Virginia, the VSCC retains continuing supervisory control over the terms and conditions of affiliate contracts and any modifications or amendments thereto and may revise or amend the terms of such contracts as necessary to protect and promote the public interest. Section 56-82 of the Code of Virginia requires a public utility to obtain the approval of the VSCC to loan money to, or assume any obligation or liability of, an affiliate, and to extend or renew any such loan or assumption of an obligation or liability. An important aspect of the public interest is assurance that an affiliated company of a regulated utility does not receive unjust benefits to the detriment of the utility's customers.

#### **Operations of the Company**

The Merger Agreement provides that at and after the consummation of the Mergers, the Company will maintain (i) its corporate headquarters and principal executive offices in Wilmington, Delaware and (ii) a significant presence in New Jersey.

Following the consummation of the Mergers, the Company and its subsidiaries will honor all prior contracts, agreements, collective bargaining agreements and commitments with current or former employees or current or former directors of Delmarva or Atlantic and their respective subsidiaries, in accordance with the respective terms of such contracts, agreements and commitments, subject to the Company's right to enforce them in accordance with their terms (including any reserved right to amend, modify, suspend, revoke or terminate them). The Company will take all action necessary so that after the consummation of the Mergers, the Dividend Reinvestment and Common Share Purchase Plan, the Savings and Thrift Plan and the Long-Term Incentive Plan of Delmarva and the Equity Incentive Plan, the Directors' Restricted Stock Plan, the Employee Stock Purchase Plan and the Dividend Reinvestment Plan of Atlantic will be



terminated, replaced or amended to provide for the issue and sale of Company Common Stock in place of Delmarva Common Stock or Atlantic Common Stock, as the case may be, under such plans.

After the consummation of the Mergers, the Company will provide charitable contributions and community support within the service areas of Delmarva and Atlantic and each of their respective subsidiaries at levels substantially comparable to the historical levels of charitable contribution and community support provided by Delmarva, Atlantic and their respective subsidiaries within their service areas.

#### **Dividend Policies**

It is anticipated that following the Mergers the Company will initially pay dividends on the Company Common Stock at the rate of \$1.54 per annum, subject to evaluation from time to time by the Company Board based on the Company's results of operations, financial condition, capital requirements and other relevant considerations. However, no assurance can be given that such dividend rate will be in effect or will remain unchanged, and the Company reserves the right to increase or decrease the dividend on the Company Common Stock as may be required by law or contract or as may be determined by the Company Board, in its discretion, to be advisable. For a description of certain restrictions on the Company's ability to pay dividends on the Company Common Stock, see "Description of the Company's Capital Stock."

It is anticipated that, subject to declaration by the Company Board and the obligations of the Company Board to react to the financial condition and regulatory environment of the Company and its results of operations, the dividends declared and paid on the Class A Common Stock will be maintained at a level of \$3.20 per share per annum until the earlier of July 1, 2001, or the end of the twelfth calendar quarter following the calendar quarter in which the Mergers are consummated.

Thereafter, the Company intends, subject to declaration by the Company Board and the obligation of the Company Board to consider the financial condition and regulatory environment of the Company and the results of its operations, to pay annual dividends on the Class A Common Stock in an amount (such amount to include the amount credited to the Intergroup Interest as described under "Description of the Company's Capital Stock—Intergroup Interest—Adjustments in Connection with Various Transactions—Dividends") equal to 90% of the Company Net Income (Loss) Attributable to the Atlantic Utility Group, subject to the fact that if annual dividends on the Class A Common Stock during the period referred to above exceed 100% of the Company Net Income (Loss) Attributable to the Atlantic Utility Group during such period, the Company Board may consider such fact in determining the amount of future dividends, if any. There can be no assurance that the Company's earnings attributable to the Atlantic Utility Group will be sufficient to cover dividends on the Class A Common Stock during the Initial Period. Dividends on the Class A Common Stock during the Initial Period. Dividends on the Class A Common Stock during the Initial Period. Dividends on the Class A Common Stock during the Initial Period.

#### **Financial Statements**

Both the holders of Company Common Stock and the holders of Class A Common Stock will receive the consolidated financial statements of the Company. Since upon consummation of the Mergers the financial results of ACE will be substantially identical to the financial results for the Atlantic Utility Group, the notes to the consolidated financial statements of the Company will at such time include condensed financial information of ACE, including a reconciliation of ACE's Income Available to common stockholders to the Earnings Applicable for Class A Common Stock. Complete financial statements of ACE will continue to be filed under the Exchange Act and will be available to stockholders upon request.

#### Name of the Company

Upon consummation of the Mergers, the name of the Company will be changed to "Conectiv."

#### EXPERTS

The consolidated balance sheets and statements of capitalization of Delmarva Power & Light Company as of December 31, 1995 and 1994 and the consolidated statements of income, changes in common stockholders' equity, and cash flows for each of the three years in the period ended December 31, 1995, incorporated by reference in this joint proxy statement/prospectus, have been incorporated herein in reliance on the report of Coopers & Lybrand L.L.P., independent accountants, given on the authority of that firm as experts in accounting and auditing. The consolidated balance sheet of Conectiv, Inc. as of September 30, 1996, included in this joint proxy statement/prospectus, has been included herein in reliance on the report of Coopers & Lybrand L.L.P., independent accountants, given on the authority of that firm as experts in accounting and auditing.

The consolidated financial statements of Atlantic Energy, Inc. and Atlantic City Electric Company incorporated herein by reference from Atlantic Energy, Inc.'s and Atlantic City Electric Company's Annual Reports on Form 10-K for the year ended December 31, 1995 have been audited by Deloitte & Touche LLP, independent auditors, as stated in their report, which are incorporated herein by reference, and have been so incorporated in reliance upon the reports of such firm given upon their authority as experts in accounting and auditing.

## LEGAL MATTERS

LeBoeuf, Lamb, Greene & MacRae, L.L.P. and Simpson Thacher & Bartlett (a partnership which includes professional corporations) will pass upon the legality of the shares of Company Common Stock and the Class A Common Stock issued in connection with the Mergers on behalf of Delmarva and Atlantic, respectively.

#### STOCKHOLDER PROPOSALS

For proposals of holders of Delmarva Common Stock intended to be presented at the Annual Meeting of Stockholders to be held in 1997 to be considered for inclusion in the Delmarva proxy statement and form of proxy relating to that meeting, such proposals must be received by Delmarva on or before December 27, 1996. Proposals should be sent to Donald P. Connelly, Secretary, Delmarva Power & Light Company, 800 King Street, Wilmington, Delaware 19899.

The November 15, 1996 deadline for receipt by Atlantic of proposals from holders of Atlantic Common Stock intended to be presented at the Annual Meeting of Stockholders to be held in 1997 to be considered for inclusion in the Atlantic proxy statement and form of proxy relating to that meeting has expired. For proposals of holders of Atlantic Common Stock intended to be presented at the Annual Meeting of Stockholders to be held in 1998 to be considered for inclusion in the Atlantic proxy statement and form of proxy relating to that meeting, such proposals must be received by Atlantic on or before November 19, 1997. Proposals should be sent to James E. Franklin II, Secretary, Atlantic Energy, Inc., 6801 Black Horse Pike, Egg Harbor Township, New Jersey 08234.

# AGREEMENT AND PLAN OF

### MERGER

dated as of August 9, 1996

as

amended and restated

as of December 26, 1996

by and among

# **DELMARVA POWER & LIGHT COMPANY**

# ATLANTIC ENERGY, INC.

# CONECTIV, INC.

and

DS SUB, INC.

## AGREEMENT AND PLAN OF MERGER

AGREEMENT AND PLAN OF MERGER, dated as of August 9, 1996 as amended and restated in its entirety as of December 26, 1996 (this "Agreement"), by and among Delmarva Power & Light Company, a corporation formed under the laws of the State of Delaware and the Commonwealth of Virginia ("Delmarva"), Atlantic Energy, Inc., a corporation formed under the laws of the State of Delaware, 50% of whose outstanding capital stock is owned by Delmarva and 50% of whose capital stock is owned by Atlantic (the "Company"), and DS Sub, Inc., a corporation formed under the laws of the State of Delaware and a wholly owned subsidiary of the Company ("DS Sub").

WHEREAS, Delmarva and Atlantic have determined to engage in a business combination as peer firms in a merger of equals;

WHEREAS, in furtherance thereof, the respective Boards of Directors of Delmarva, Atlantic, the Company and DS Sub have approved the consummation of a reorganization provided for in this Agreement, pursuant to which Atlantic will merge with and into the Company and DS Sub will merge with and into Delmarva, in each case upon the terms and subject to the conditions set forth in this Agreement (such transactions being referred to herein respectively as the Atlantic Merger and the Delmarva Merger, each individually as a "Merger" and together as the "Mergers"), as a result of which the common shareholders of Delmarva and Atlantic will together own all of the outstanding shares of common stock of the Company, including, in the case of the common shareholders of Atlantic, the Letter Stock (as defined in Section 2.1(b)), and each share of each other class of capital stock, if any, of Delmarva and Atlantic shall be unaffected and remain outstanding; and

WHEREAS, for federal income tax purposes, it is intended that (i) the Delmarva Merger will qualify as an exchange under the provisions of Section 351 of the United States Internal Revenue Code of 1986, as amended (the "Code"), and/or as a reorganization under the provisions of Section 368(a) of the Code and (ii) the Atlantic Merger will qualify as a reorganization under the provisions of Section 368(a) of the Code.

NOW THEREFORE, in consideration of the premises and the representations, warranties, covenants and agreements contained herein, the parties hereto, intending to be legally bound, hereby agree as follows:

## ARTICLE I

#### THE MERGERS

Section 1.1 The Mergers. Upon the terms and subject to the conditions of this Agreement:

ţ

(a) At the Effective Time (as defined in Section 1.3), Atlantic shall be merged with and into the Company (the "Atlantic Merger") in accordance with the applicable provisions of the laws of the State of Delaware and New Jersey. The Company shall be the surviving corporation in the Atlantic Merger and shall continue its corporate existence under the laws of the State of Delaware. The effects and the consequences of the Atlantic Merger shall be as set forth in Section 1.2(a). Throughout this Agreement, the term "Company" shall refer to the Company prior to the Atlantic Merger or to the Company in its capacity as the surviving corporation in the Atlantic Merger, as the context requires.

(b) At the Effective Time, DS Sub shall be merged with and into Delmarva (the "Delmarva Merger") in accordance with the laws of the State of Delaware and the Commonwealth of Virginia. Delmarva shall be the surviving corporation in the Delmarva Merger and shall continue its corporate existence under the laws of the State of Delaware and the Commonwealth of Virginia. The effects and the consequences of the Delmarva Merger shall be as set forth in Section 1.2(b).

## Section 1.2 Effects of the Mergers.

(a) At the Effective Time, (i) the certificate of incorporation of the Company, as in effect at the Effective Time (which shall be amended and restated pursuant to Section 7.16), shall be the certificate of incorporation of the surviving corporation in the Atlantic Merger until thereafter amended as provided by law and such certificate of incorporation, and (ii) the bylaws of the Company, as in effect immediately prior to the Effective Time (which shall be amended and restated pursuant to Section 7.16), shall be the bylaws of the surviving corporation in the Atlantic Merger until thereafter amended as provided by law, the certificate of incorporation in the Atlantic Merger until thereafter amended as provided by law, the certificate of incorporation of the surviving corporation in the Atlantic Merger and such bylaws. Subject to the foregoing, the additional effects of the Atlantic Merger shall be as provided in the applicable provisions of the Delaware General Corporation Law (the "DGCL") and the New Jersey Business Corporation Act (the "NJBCA").

(b) At the Effective Time, (i) the certificate and articles of incorporation of Delmarva, as in effect immediately prior to the Effective Time, shall be the certificate and articles of incorporation of the surviving corporation in the Delmarva Merger until thereafter amended as provided by law and such certificate and articles of incorporation, and (ii) the bylaws of Delmarva, as in effect immediately prior to the Effective Time, shall be the bylaws of the surviving corporation in the Delmarva Merger until thereafter amended as provided by law. Merger until thereafter amended as provided by law, the certificate and articles of incorporation of the surviving corporation in the Delmarva Merger and such bylaws. Subject to the foregoing, the additional effects of the Delmarva Merger shall be as provided in the applicable provisions of the DGCL and the Virginia Stock Corporation Act (the "VSCA").

Section 1.3 *Effective Time of the Mergers.* On the Closing Date (as defined in Section 3.1): (a) a certificate of merger with respect to the Delmarva Merger shall be executed and filed by Delmarva with the Secretary of State of the State of Delaware pursuant to Section 251 of the DGCL and articles of merger with respect to the Delmarva Merger shall be executed and filed by Delmarva with the State Corporation Commission of Virginia (the "Virginia Commission") pursuant to Section 13.1-720 of the VSCA; and (b) a certificate of merger with respect to the Atlantic Merger shall be executed and filed by the Company with the Secretary of State of the State of Delaware pursuant to Section 252 of the DGCL and a certificate of merger with respect to the Atlantic Merger shall be executed and filed by Atlantic and the Company with the Secretary of State of the State of New Jersey pursuant to Section 14A:10-4.1 of the NJBCA. The Mergers shall become effective simultaneously and at the time that Delmarva and Atlantic shall agree, which time shall be specified in the respective certificates and articles of merger for the Mergers (such time being herein referred to as the "Effective Time").

#### ARTICLE II

### TREATMENT OF SHARES

Section 2.1 *Effect of Mergers on Capital Stock*. At the Effective Time, by virtue of the Mergers and without any action on the part of any holder of any capital stock of Delmarva, Atlantic, the Company or DS Sub:

(a) Cancellation of Certain Common Stock. Each share of (i) Delmarva common stock, par value \$2.25 ("Delmarva Common Stock"), that is owned by Delmarva as treasury stock or by Atlantic or by any wholly owned subsidiary (as defined in Section 4.1) of Delmarva or Atlantic, (ii) Atlantic common stock, no par value ("Atlantic Common Stock"), that is owned by Atlantic as treasury stock or by Delmarva or by any wholly owned subsidiary of Atlantic or Delmarva and (iii) Company Common Stock, par value \$.01 per share ("Company Common Stock"), that is owned by Delmarva, Atlantic or any wholly owned subsidiary of Delmarva or Atlantic, in each case shall be cancelled and shall cease to exist, and no consideration shall be delivered in exchange therefor.

I-2

(b) Conversion of Certain Common Stock. Each share of Delmarva Common Stock issued and outstanding immediately prior to the Effective Time (other than shares cancelled pursuant to Section 2.1(a)) shall be converted into the right to receive one fully paid and nonassessable share of Company Common Stock (the "Delmarva Conversion Ratio"), and each share of Atlantic Common Stock issued and outstanding immediately prior to the Effective Time (other than shares cancelled pursuant to Section 2.1(a)) shall be converted into the right to receive 0.75 fully paid and nonassessable share(s) of Company Common Stock and 0.125 fully paid and nonassessable share(s) of Company Common Stock and 0.125 fully paid and nonassessable share(s) of Class A common stock, par value \$0.01 per share ("Letter Stock") of the Company, the provisions of which are set forth in Article IV of the form of certificate of incorporation of the Company attached hereto as Exhibit A (the "Atlantic Conversion Ratio"). Upon such conversions as provided for herein, each holder of a certificate formerly representing any such shares of Delmarva Common Stock or Atlantic Common Stock shall cease to have any rights with respect thereto, except the right to receive the shares of Company Common Stock and, in the case of holders of Atlantic Common Stock, the shares of Letter Stock to be issued in consideration therefor (and cash in lieu of fractional shares) upon the surrender of such certificate in accordance with Section 2.2.

(c) Conversion of DS Sub Common Stock. Each share of DS Sub Common Stock, par value \$.01 per share, issued and outstanding immediately prior to the Effective Time shall be converted into and become one share of Common Stock, par value \$2.25, of the surviving corporation in the Delmarva Merger.

(d) Delmarva Preferred Stock Unchanged. Each share of Delmarva Preferred Stock and each share of Delmarva Preferred Stock — \$25 par (together, "Delmarva Preferred Stock"), shall be unchanged in and shall remain outstanding after the Mergers.

(e) Shares of Dissenting Holders. Any issued and outstanding shares of Delmarva stock which is neither (i) listed on a national securities exchange or designated as a national market system security on an interdealer quotation system by the National Association of Security Dealers, Inc., nor (ii) held of record by more than 2,000 holders, and which are held by a person who objects to the Merger and who has demanded appraisal for such shares in accordance with all applicable provisions of the DGCL or the VSCA concerning the right of such person to dissent from the Mergers and demand appraisal of such shares (each, a "Dissenting Holder") shall, if the DGCL or the VSCA provides for appraisal rights with respect to such shares in the Mergers, from and after the Effective Time represent only the right to receive such consideration as may be determined to be due to such Dissenting Holder with respect to such shares pursuant to the DGCL or the VSCA, as applicable; provided, however, that shares outstanding immediately prior to the Effective Time and held by a Dissenting Holder who shall withdraw the demand for appraisal, or lose the right of appraisal of such shares, pursuant to the DGCL or the VSCA, as applicable; shall be unchanged in and remain outstanding after the Mergers, without interest.

Section 2.2 Exchange of Common Stock Certificates.

(a) Deposit with Exchange Agent. On or before the Effective Time, the Company shall deposit with a bank, trust company or other agent selected by Delmarva and Atlantic (the "Exchange Agent") certificates representing shares of Company Common Stock and Letter Stock required to effect the conversion of Delmarva Common Stock or Atlantic Common Stock, as the case may be, into Company Common Stock and, in the case of Atlantic Common Stock, Letter Stock as described in Section 2.1(b).

(b) Exchange Procedures. As soon as practicable after the Effective Time, the Exchange Agent shall mail to each holder of record of a certificate or certificates that immediately prior to the Effective Time represented issued and outstanding shares of Delmarva Common Stock or Atlantic Common Stock (collectively, "Certificates") that were converted ("Converted Shares") into the right to receive shares of Company Common Stock and, in the case of Atlantic Common Stock, Letter Stock (collectively, "Company Shares") pursuant to Section 2.1(b), (i) a letter of transmittal (which shall specify that delivery shall be effected, and risk of loss and title to the Certificates shall pass, only upon actual delivery of the Certificates to the Exchange Agent) and (ii) instructions for use in effecting the exchange of Certificates for certificates representing Company Shares. Upon delivery of a Certificate to the Exchange Agent for

I-3

exchange, together with a duly executed letter of transmittal and such other documents as the Exchange Agent shall require, the holder of such Certificate shall be entitled to receive in exchange therefor a certificate representing that number of whole Company Shares and the amount of cash in lieu of fractional share interests that such holder has the right to receive pursuant to the provisions of this Article II. In the event of a transfer of ownership of Converted Shares that is not registered in the transfer records of Delmarva or Atlantic, as the case may be, a certificate representing the proper number of Company Shares may be issued to a transferee if the Certificate representing such Converted Shares is presented to the Exchange Agent, accompanied by all documents required to evidence and effect such transfer and by evidence satisfactory to the Exchange Agent that any applicable stock transfer taxes have been paid. Until delivered as contemplated by this Section 2.2, each Certificate shall be deemed at any time after the Effective Time to represent only the right to receive upon such delivery the certificate representing Company Shares and cash in lieu of any fractional shares of Company Common Stock as contemplated by this Section 2.2.

(c) Distributions with Respect to Unexchanged Shares. No dividends or other distributions declared or made after the Effective Time with respect to Company Shares with a record date after the Effective Time shall be paid to the holder of any undelivered Certificate with respect to the Company Shares represented thereby, and no cash payment in lieu of fractional shares shall be paid to any such holder pursuant to Section 2.2(d), until the holder of record of such Certificate (or a transferee as described in Section 2.2(b)) shall have delivered such Certificate as contemplated in Section 2.2(b). Subject to the effect of unclaimed property, escheat and other applicable laws, following delivery of any such Certificate there shall be paid to the record holder (or transferee) of the certificates representing whole Company Shares issued in exchange therefor, without interest, (i) at the time of such delivery, the amount of any cash payable in lieu of a fractional share of Company Shares to which such holder (or transferee) is entitled pursuant to Section 2.2(d) and the amount of dividends or other distributions with a record date after the Effective Time theretofore paid with respect to such whole Company Shares and (ii) at the appropriate payment date, the amount of dividends or other distributions with a record date after the Effective Time but prior to such delivery and a payment date subsequent to such delivery payable with respect to such whole Company Shares.

(d) *No Fractional Shares.* (i) No certificates or scrip representing fractional shares of Company Shares shall be issued upon the delivery for exchange of Certificates, and such fractional share interests shall not entitle the owner thereof to vote or to any rights of a shareholder of the Company.

(ii) As promptly as practicable following the Effective Time, the Exchange Agent shall determine the excess of (x) the number of full shares of Company Shares delivered to the Exchange Agent by the Company pursuant to Section 2.2(a) over (y) the aggregate number of full shares of Company Common Stock and Letter Stock to be distributed to holders of Delmarva Common Stock and Atlantic Common Stock, respectively, pursuant to Section 2.1 (such excess being hereinafter referred to as the "Excess Shares"). As soon after the Effective Time as practicable, the Exchange Agent, as agent for the holders of Atlantic Common Stock, shall sell the Excess Shares at then prevailing prices on the New York Stock Exchange (the "NYSE"), all in the manner provided in Section 2.2(d)(iii).

(iii) The sale of the Excess Shares by the Exchange Agent shall be executed on the NYSE through one or more member firms of the NYSE and shall be executed in round lots to the extent practicable. Until the proceeds of such sale or sales have been distributed to the holders of Atlantic Common Stock, the Exchange Agent shall, until remitted pursuant to Section 2.2(f), hold such proceeds in trust for the holders of Atlantic Common Stock (the "Common Shares Trust"). The Company shall pay all commissions, transfer taxes and other out-of-pocket transaction costs, including the expenses and compensation, of the Exchange Agent incurred in connection with such sale of the Excess Shares. The Exchange Agent shall determine the portion of the proceeds comprising the Common Shares Trust to which each holder of Atlantic Common Stock shall be entitled, if any, by multiplying the amount of the aggregate proceeds comprising the Common Shares Trust by a fraction the numerator of which is the amount of the fractional share interest to which such holder of Atlantic Common Stock is entitled and the denominator of which is the aggregate amount of fractional share interests to which all holders of Atlantic Common Stock are entitled.

(iv) As soon as practicable after the sale of Excess Shares pursuant to clause (iii) above and the determination of the amount of cash, if any, to be paid to holders of Atlantic Common Stock in lieu of any fractional share interests, the Exchange Agent shall distribute such amounts to holders of Atlantic Common Stock who have theretofore delivered Certificates for Atlantic Common Stock for exchange pursuant to this Article II.

(e) Closing of Transfer Books. From and after the Effective Time, the stock transfer books of Delmarva with respect to shares of Delmarva Common Stock, and of Atlantic with respect to shares of Atlantic Common Stock, issued and outstanding prior to the Effective Time shall be closed and no transfer of any such shares shall thereafter be made. If after the Effective Time Certificates are presented to the Company, they shall be cancelled and exchanged for certificates representing the appropriate number of whole Company Shares and cash in lieu of fractional shares of Company Common Stock as provided in this Section 2.2.

(f) Termination of Exchange Agent. Any certificates representing Company Shares deposited with the Exchange Agent pursuant to Section 2.2(a) and not exchanged within one year after the Effective Time pursuant to this Section 2.2 shall be returned by the Exchange Agent to the Company, which shall thereafter act as Exchange Agent. All funds held by the Exchange Agent for payment to the holders of undelivered Certificates and unclaimed at the end of one year from the Effective Time shall be remitted to the Company, after which time any holder of undelivered Certificates shall look as a general creditor only to the Company for payment of such funds to which such holder may be due, subject to applicable law. The Company shall not be liable to any person for such shares or funds delivered to a public official pursuant to any applicable abandoned property, escheat or similar law.

#### ARTICLE III

#### THE CLOSING

Section 3.1 *Closing*. The closing (the "Closing") of the Mergers shall take place at a place to be mutually agreed upon by the parties hereto at 10:00 A.M., local time, on the second business day immediately following the date on which the last of the conditions set forth in Article VIII is fulfilled or waived, or at such other time and date as Delmarva and Atlantic shall mutually agree (the "Closing Date").

#### • ARTICLE IV

#### **REPRESENTATIONS AND WARRANTIES OF DELMARVA**

Delmarva represents and warrants to Atlantic as follows:

Section 4.1 Organization and Qualification. Except as disclosed in Section 4.1 of the Delmarva Disclosure Schedule (as defined in Section 7.6(ii)), (i) Delmarva is a corporation duly organized, validly existing and in good standing under the laws of its jurisdictions of incorporation, (ii) each of Delmarva's subsidiaries is a corporation duly organized, validly existing and in good standing under the laws of its jurisdiction of incorporation and (iii) each of Delmarva and each of its subsidiaries has all requisite corporate power and authority, and has been duly authorized by all necessary regulatory approvals and orders, to own, lease and operate its assets and properties and to carry on its business as it is now being conducted and is duly qualified and in good standing to do business in each jurisdiction in which the nature of its business or the ownership or leasing of its assets and properties makes such qualification necessary other than in such jurisdictions where the failure to be so qualified and in good standing will not, when

taken together with all other such failures, have a material adverse effect on the business, operations, properties, assets, condition (financial or otherwise), prospects or results of operations of Delmarva and its subsidiaries taken as a whole or on the consummation of this Agreement (any such material adverse effect being hereinafter referred to as a "Delmarva Material Adverse Effect"). As used in this Agreement the term "subsidiary" with respect to any person shall mean any corporation or other entity (including partnerships and other business associations) in which such person directly or indirectly owns outstanding capital stock or other voting securities having the power, under ordinary circumstances, to elect a majority of the directors or similar members of the governing body of such corporation or other entity or otherwise to direct the management and policies of such corporation or other entity.

Section 4.2 Subsidiaries. Section 4.2 of the Delmarva Disclosure Schedule contains a description as of the date hereof of all subsidiaries and joint ventures of Delmarva, including the name of each such entity, the state or jurisdiction of its incorporation, a brief description of the principal line or lines of business conducted by each such entity and Delmarva's interest therein. Except as disclosed in Section 4.2 of the Delmarva Disclosure Schedule, none of such entities is a "public utility company," a "holding company," a "subsidiary company" or an "affiliate" of any public utility company within the meaning of Section 2(a)(5), 2(a)(7), 2(a)(8) and 2(a)(11) of the Public Utility Holding Company Act of 1935, as amended (the "1935 Act"), respectively. Except as disclosed in Section 4.2 of the Delmarva Disclosure Schedule, all of the issued and outstanding shares of capital stock of each subsidiary of Delmarva are validly issued, fully paid, nonassessable and free of preemptive rights and are owned directly or indirectly by Delmarva free and clear of any liens, claims, encumbrances, security interests, equities, charges and options of any nature whatsoever, and there are no outstanding subscriptions, options, calls, contracts, voting trusts, proxies or other commitments, understandings, restrictions, arrangements, rights or warrants, including any right of conversion or exchange under any outstanding security, instrument or other agreement, obligating any such subsidiary to issue, deliver or sell, or cause to be issued, delivered or sold, additional shares of its capital stock or obligating it to grant, extend or enter into any such agreement or commitment. As used in this Agreement, the term "joint venture" with respect to any person shall mean any corporation or other entity (including partnerships and other business associations and joint ventures) in which such person of one or more of its subsidiaries owns an equity interest that is less than a majority of any class of the outstanding voting securities or equity, other than equity interests held for passive investment purposes that are less than 5% of any class of the outstanding voting securities or equity.

Section 4.3 *Capitalization*. The authorized capital stock of Delmarva consists of 90,000,000 shares of Delmarva Common Stock, 1,800,000 shares of Delmarva Preferred Stock, 3,000,000 shares of Delmarva Preferred Stock—\$25 Par, and 10,000,000 shares of Delmarva Preferred Stock—\$1 Par. As of the close of business on June 30, 1996, 60,697,635 shares of Delmarva Common Stock, 1,280,850 shares of Delmarva Preferred Stock and 1,600,000 shares of Delmarva Preferred Stock—\$25 Par were issued and outstanding and no shares of Delmarva Preferred Stock—\$1 Par were issued and outstanding and no shares of the capital stock of Delmarva are validly issued, fully paid, nonassessable and free of preemptive rights. Except as disclosed in Section 4.3 of the Delmarva Disclosure Schedule, as of the date hereof, there are no outstanding subscriptions, options, calls, contracts, voting trusts, proxies or other commitments, understanding security, instrument or other agreement, obligating Delmarva or any of its subsidiaries to issue, deliver or sell, or cause to be issued, delivered or sold, additional shares of the capital stock or other voting securities of Delmarva or obligating Delmarva or any of its subsidiaries to grant, extend or enter into any such agreement or commitment.

#### Section 4.4 Authority; Noncontravention; Statutory Approvals; Compliance:

(a) Authority. Delmarva has all requisite power and authority to enter into this Agreement and, subject to the Delmarva Shareholders' Approval (as defined in Section 4.13) and the Delmarva Required Statutory Approvals (as defined in Section 4.4(c)), to consummate the transactions contemplated hereby. The execution and delivery of this Agreement and the consummation by Delmarva of the transactions

contemplated hereby have been duly authorized by all necessary corporate action on the part of Delmarva, subject to obtaining the Delmarva Shareholders' Approval. This Agreement has been duly and validly executed and delivered by Delmarva and, assuming the due authorization, execution and delivery of this Agreement by the other signatories hereto constitutes the legal, valid and binding obligation of Delmarva enforceable against Delmarva in accordance with its terms.

(b) Noncontravention. Except as disclosed in Section 4.4(b) of the Delmarva Disclosure Schedule, the execution and delivery of this Agreement by Delmarva do not, and the consummation of the transactions contemplated hereby will not, violate, conflict with or result in a breach of any provision of, or constitute a default (with or without notice or lapse of time or both) under, or result in the termination of, or accelerate the performance required by, or result in a right of termination, cancellation or acceleration of any obligation under or the loss of a material benefit under, or result in the creation of any lien, security interest, charge or encumbrance upon any of the properties or assets (any such violation, conflict, breach, default, right of termination, cancellation or acceleration, loss or creation, a "Violation") of Delmarva or any of its subsidiaries or, to the best knowledge of Delmarva, any of its joint ventures, under any provisions of (i) the certificate or articles of incorporation, bylaws or similar governing documents of Delmarva or any of its subsidiaries or joint ventures, (ii) subject to obtaining the Delmarva Required Statutory Approvals and the receipt of the Delmarva Shareholders' Approval, any statute, law, ordinance, rule, regulation, judgment, decree, order, injunction, writ, permit or license of any court, governmental or regulatory body (including a stock exchange or other self-regulatory body) or authority, domestic or foreign (each, a "Governmental Authority") applicable to Delmarva or any of its subsidiaries or joint ventures or any of their respective properties or assets or (iii) subject to obtaining the third-party consents or other approvals set forth in Section 4.4(b) of the Delmarva Disclosure Schedule (the "Delmarva Required Consents"), any note, bond, mortgage, indenture, deed of trust, license, franchise, permit, concession, contract, lease or other instrument, obligation or agreement of any kind to which Delmarva or any of its subsidiaries or joint ventures is now a party or by which any of them or any of their respective properties or assets may be bound or affected, excluding from the foregoing clauses (ii) and (iii) such Violations as, in the aggregate, are not reasonably likely to have a Delmarva Material Adverse Effect.

(c) Statutory Approvals. Except as disclosed in Section 4.4(c) of the Delmarva Disclosure Schedule, no declaration, filing or registration with, or notice to or authorization, consent, finding by or approval of, any Governmental Authority is necessary for the execution and delivery of this Agreement by Delmarva or the consummation by Delmarva of the transactions contemplated hereby, the failure to obtain which, in the aggregate, is reasonably likely to have a Delmarva Material Adverse Effect (the "Delmarva Required Statutory Approvals"), it being understood that references in this Agreement to "obtaining" such Delmarva Required Statutory Approvals shall mean making such declarations, filings or registrations, giving such notice, obtaining such authorizations, consents or approvals and having such waiting periods expire as are, in each case, necessary to avoid a violation of law.

(d) Compliance. Except as disclosed in Section 4.4(d), 4.7, 4.10 or 4.11 of the Delmarva Disclosure Schedule or as disclosed in the Delmarva SEC Reports (as defined in Section 4.5), neither Delmarva nor any of its subsidiaries nor, to the best knowledge of Delmarva, any of its joint ventures is in violation of or under investigation with respect to, or has been given notice or been charged with any violation of, any law, statute, order, rule, regulation, ordinance or judgment (including without limitation any applicable Environmental Laws (as defined in Section 4.11(g)(ii)) of any Governmental Authority, except for violations that, in the aggregate, do not have, and to the best knowledge of Delmarva are not reasonably likely to have, a Delmarva Material Adverse Effect. Except as disclosed in Section 4.4(d) or 4.11 of the Delmarva Disclosure Schedule, Delmarva, its subsidiaries and, to the best knowledge of Delmarva, its joint ventures have all permits, licenses, franchises and other governmental authorizations, consents and approvals necessary to conduct their respective businesses as currently conducted (collectively, "Permits"), except those the failure to obtain which, in the aggregate, are not reasonably likely to have a Delmarva Disclosure Schedule, neither Delmarva nor any of its subsidiaries nor, to the best knowledge of Delmarva, any of its subsidiaries and, the Delmarva fractions, consents and approvals necessary to conduct their respective businesses as currently conducted (collectively, "Permits"), except those the failure to obtain which, in the aggregate, are not reasonably likely to have a Delmarva Disclosure Schedule, neither Delmarva nor any of its subsidiaries nor, to the best knowledge of Delmarva, any of its

joint ventures, is in breach or violation of or in default in the performance or observance of any term or provision of, and no event has occurred which, with lapse of time or action by a third party, could result in a default under (i) its certificate or articles of incorporation or its bylaws or (ii) except where such breaches, violations and defaults, in the aggregate, are not reasonably likely to have a Delmarva Material Adverse Effect, any contract, commitment, agreement, indenture, mortgage, loan agreement, note, lease, bond, license, approval or other instrument to which it is a party or by which it is bound or to which any of its property is subject.

Section 4.5 Reports and Financial Statements. Except as disclosed in Section 4.5 of the Delmarva Disclosure Schedule, the filings required to be made by Delmarva and its subsidiaries since January 1, 1991 under the Securities Act of 1933, as amended (the "Securities Act"), the Securities Exchange Act of 1934, as amended (the "Exchange Act"), applicable Delaware, Maryland, Virginia and Pennsylvania laws and regulations, the Federal Power Act, as amended (the "Power Act"), the 1935 Act and the Atomic Energy Act of 1954, as amended (the "Atomic Energy Act"), have been filed with the Securities and Exchange Commission (the "SEC"), the Delaware Public Service Commission (the "Delaware Commission"), the Maryland Public Service Commission (the "Maryland Commission"), the Virginia Commission, the Pennsylvania Public Utility Commission (the "Pennsylvania Commission"), the Federal Energy Regulatory Commission (the "FERC") and the Nuclear Regulatory Commission (the "NRC"), respectively, including all forms, statements, reports, agreements (oral or written) and all documents, exhibits, amendments and supplements appertaining thereto, and, as of their respective dates, complied in all material respects with all applicable requirements of the appropriate act and the rules and regulations thereunder. Delmarva has made available to Atlantic a true and complete copy of each report, schedule, registration statement and definitive proxy statement filed by Delmarva with the SEC since January 1, 1991 and through the date hereof (as such documents have since the time of their filing been amended, the "Delmarva SEC Reports"). The Delmarva SEC Reports, including without limitation any financial statements or schedules included therein, at the time filed, and any forms, reports or other documents filed by Delmarva with the SEC after the date hereof, did not and will not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. The audited consolidated financial statements and unaudited interim financial statements of Delmarva included in the Delmarva SEC Reports (collectively, the "Delmarva Financial Statements") have been prepared, and will be prepared, in accordance with generally accepted accounting principles applied on a consistent basis ("GAAP") (except as may be indicated therein or in the notes thereto and except with respect to unaudited statements as permitted by Form 10-Q) and fairly present the consolidated financial position of Delmarva as of the respective dates thereof or the consolidated results of operations and cash flows for the respective periods then ended, as the case may be, subject, in the case of the unaudited interim financial statements, to normal, recurring audit adjustments. True, accurate and complete copies of the certificate and articles of incorporation and the bylaws of Delmarva, as in effect on the date hereof, have been delivered to Atlantic.

Section 4.6 Absence of Certain Changes or Events; Absence of Undisclosed Liabilities. (a) Except as disclosed in the Delmarva SEC Reports filed prior to the date hereof or as disclosed in Section 4.6 of the Delmarva Disclosure Schedule, from December 31, 1995 through the date hereof each of Delmarva and each of its subsidiaries has conducted its business only in the ordinary course of business consistent with past practice and no event has occurred which has had, and no fact or condition exists that would have or, to the best knowledge of Delmarva, is reasonably likely to have, a Delmarva Material Adverse Effect.

(b) Neither Delmarva nor any of its subsidiaries has any liabilities or obligations (whether absolute, contingent, accrued or otherwise) of a nature required by GAAP to be reflected in a consolidated corporate balance sheet, except liabilities, obligations or contingencies that are accrued or reserved against in the consolidated financial statements of Delmarva or are reflected in the notes thereto for the year ended December 31, 1995 or that were incurred after December 31, 1995 in the ordinary course of business and are not reasonably likely to have a Delmarva Material Adverse Effect.



Section 4.7 Litigation. Except as disclosed in the Delmarva SEC Reports filed prior to the date hereof or as disclosed in Section 4.7, 4.9, 4.10 or 4.11 of the Delmarva Disclosure Schedule, (i) there are no claims, suits, actions or proceedings pending or to the best knowledge of Delmarva threatened, nor are there any investigations or reviews pending or, to the best knowledge of Delmarva, threatened against, relating to or affecting Delmarva or any of its subsidiaries, (ii) there have not been any developments since December 31, 1995 with respect to any such disclosed claims, suits, actions, proceedings, investigations or review and (iii) there are no judgments, decrees, injunctions, rules or orders of any court, governmental department, commission, agency, instrumentality or authority or any arbitrator applicable to Delmarva or any of its subsidiaries that, in the aggregate, are reasonably likely to have a Delmarva Material Adverse Effect.

Section 4.8 Registration Statement and Proxy Statement. None of the information supplied or to be supplied by or on behalf of Delmarva for inclusion or incorporation by reference in (i) the registration statement on Form S-4 to be filed with the SEC by the Company in connection with the issuance of shares of Company Common Stock and Letter Stock in the Mergers (the "Registration Statement") will, at the time the Registration Statement becomes effective under the Securities Act, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading and (ii) the joint proxy in definitive form, relating to the meetings of the shareholders of Delmarva and Atlantic to be held in connection with the Mergers (the "Joint Proxy Statement") will, at the date mailed to such shareholders and, as the same may be amended or supplemented, at the times of such meetings, contain any untrue statement of a material fact or omit to state meetings, contain any untrue statement of a material fact or omit to such shareholders and, as the same may be amended or supplemented, at the times of such meetings, contain any untrue statement of a material fact or omit to state any material fact necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading. The Registration Statement and the Joint Proxy Statement will comply as to form in all material respects with the provisions of the Securities Act and the Exchange Act and the rules and regulations thereunder.

Section 4.9 Tax Matters. "Taxes", as used in this Agreement, means any federal, state, county, local or oreign taxes, charges, fees, levies, or other assessments, including all net income, gross income, sales and use, ad valorem, transfer, gains, profits, excise, franchise, real and personal property, gross receipts, capital stock, production, business and occupation, disability, employment, payroll, license, estimated, stamp, custom duties, severance or withholding taxes or charges imposed by any governmental entity, and includes any interest and penalties (civil or criminal) on or additions to any such taxes and any expenses incurred in connection with the determination, settlement or litigation of any tax liability. "Tax Return", as used in this Agreement, means a report, return or other information required to be supplied to a governmental entity with respect to Taxes including, where permitted or required, combined or consolidated returns for any group of entities that includes Delmarva or any of its subsidiaries, on the one hand, or Atlantic or any of its subsidiaries, on the other hand.

(a) *Filing of Timely Tax Returns*. Except as disclosed in Section 4.9(a) of the Delmarva Disclosure Schedule, Delmarva and each of its subsidiaries have filed all Tax Returns required to be filed by each of them under applicable law. All Tax Returns were in all material respects (and, as to Tax Returns not filed as of the date hereof, will be) true, complete and correct and filed on a timely basis.

(b) *Payment of Taxes.* Except as disclosed in Section 4.9(b) of the Delmarva Disclosure Schedule, Delmarva and each of its subsidiaries have, within the time and in the manner prescribed by law, paid (and until the Closing Date will pay within the time and in the manner prescribed by law) all Taxes that are currently due and payable except for those contested in good faith and for which adequate reserves have been taken.

(c) *Tax Reserves.* Delmarva and its subsidiaries have established (and until the Closing Date will maintain) on their books and records reserves adequate to pay all Taxes and reserves for deferred income taxes in accordance with GAAP.



(d) Tax Liens. There are no Tax liens upon the assets of Delmarva or any of its subsidiaries except liens for Taxes not yet due.

(e) Withholding Taxes. Delmarva and each of its subsidiaries have complied (and until the Closing Date will comply) in all material respects with the provisions of the Code relating to the payment and withholding of Taxes, including without limitation the withholding and reporting requirements under Code §§ 1441 through 1464, 3401 through 3606, and 6041 and 6049, as well as similar provisions under any other laws, and have within the time and in the manner prescribed by law withheld from employee wages and paid over to the proper governmental authorities all amounts required.

(f) *Extensions of Time for Filing Tax Returns*. Except as disclosed in Section 4.9(f) of the Delmarva Disclosure Schedule, neither Delmarva nor any of its subsidiaries has requested any extension of time within which to file any Tax Return which Tax Return has not since been filed.

(g) *Waivers of Statute of Limitations*. Except as disclosed in Section 4.9(g) of the Delmarva Disclosure Schedule, neither Delmarva nor any of its subsidiaries has executed any outstanding waivers or comparable consents regarding the application of the statute of limitations with respect to any Taxes or Tax Returns.

(h) *Expiration of Statute of Limitations*. Except as disclosed in Section 4.9(h) of the Delmarva Disclosure Schedule, the statute of limitations for the assessment of all Taxes has expired for all applicable Tax Returns of Delmarva and each of its subsidiaries or those Tax Returns have been examined by the appropriate taxing authorities for all periods through the date hereof, and no deficiency for any Taxes has been proposed, asserted or assessed against Delmarva or any of its subsidiaries that has not been resolved and paid in full.

(i) Audit, Administrative and Court Proceedings. Except as disclosed in Section 4.9(i) of the Delmarva Disclosure Schedule, no audits or other administrative proceedings or court proceedings are presently pending with regard to any Taxes or Tax Returns of Delmarva or any of its subsidiaries.

(j) *Powers of Attorney*. Except as disclosed in Section 4.9(j) of the Delmarva Disclosure Schedule, no power of attorney currently in force has been granted by Delmarva or any of its subsidiaries concerning an Tax matter.

(k) Tax Rulings. Except as disclosed in Section 4.9(k) of the Delmarva Disclosure Schedule, neither Delmarva nor any of its subsidiaries has received a Tax Ruling (as defined below) or entered into a Closing Agreement (as defined below) with any taxing authority that would have a continuing adverse effect after the Closing Date. "Tax Rulings", as used in this Agreement, shall mean a written ruling of a taxing authority relating to Taxes. "Closing Agreement", as used in this Agreement, shall mean a written and legally binding agreement with a taxing authority relating to Taxes.

(1) Availability of Tax Returns. Delmarva and its subsidiaries have made available to Atlantic complete and accurate copies, covering all open years, of (i) all Tax Returns, and any amendments thereto, filed by Delmarva or any of its subsidiaries, (ii) all audit reports received from any taxing authority relating to any Tax Return filed by Delmarva or any of its subsidiaries and (iii) any Closing Agreements entered into by Delmarva or any of its subsidiaries with any taxing authority.

(m) Tax Sharing Agreements. Except as disclosed in Section 4.9(m) of the Delmarva Disclosure Schedule, no agreements relating to the allocation or sharing of Taxes exist between or among Delmarva and any of its subsidiaries.

(n) Code § 341(f). Neither Delmarva nor any of its subsidiaries has filed a consent pursuant to Code § 341(f) or has agreed to have Code § 341(f)(2) apply to any disposition of a subsection (f) asset (as such term is defined in Code § 341(f)(4)) owned by Delmarva or any of its subsidiaries.

(o) Code § 168. Except as disclosed in Section 4.9(o) of the Delmarva Disclosure Schedule, no property of Delmarva or any of its subsidiaries is property that Delmarva or any such subsidiary or any

party to this transaction is or will be required to treat as being owned by another person pursuant to the provisions of Code § 168(f)(8) (as in effect prior to its amendment by the Tax Reform Act of 1986) or is tax-exempt use property within the meaning of Code § 168.

(p) Code § 481 Adjustments. Except as disclosed in Section 4.9(p) of the Delmarva Disclosure Schedule, neither Delmarva nor any of its subsidiaries is required to include in income any adjustment pursuant to Code § 481(a) by reason of a voluntary change in accounting method initiated by Delmarva or any of its subsidiaries, and, to the best of the knowledge of Delmarva, the Internal Revenue Service (the "IRS") has not proposed any such adjustment or change in accounting method.

(q) Code §§ 6661 and 6662. Except as disclosed in Section 4.9(q) of the Delmarva Disclosure Schedule, all transactions that could give rise to an understatement of federal income tax (within the meaning of Code § 6661 for Tax Returns filed on or before December 31, 1990, and within the meaning of Code § 6662 for tax returns filed after December 31, 1990) that could reasonably be expected to result in a Delmarva Material Adverse Effect have been adequately disclosed (or, with respect to Tax Returns filed following the Closing, will be adequately disclosed) on the Tax Returns of Delmarva and its subsidiaries in accordance with Code § 6661(b)(2)(B) for Tax Returns filed on or prior to December 31, 1990, and in accordance with Code § 6662(d)(2)(B) for Tax Returns filed after December 31, 1990.

(r) Code § 280G. Except as disclosed in Section 4.9(r) of the Delmarva Disclosure Schedule, neither Delmarva nor any of its subsidiaries is a party to any agreement, contract, or arrangement that could reasonably be expected to result, on account of the transactions contemplated hereunder, separately or in the aggregate, in the payment of any "excess parachute payment" within the meaning of Code § 280G.

(s) *NOLS*. As of December 31, 1994, Delmarva and its subsidiaries had net operating loss carryovers available to offset future income as disclosed in Section 4.9(s) of the Delmarva Disclosure Schedule. Section 4.9(s) of the Delmarva Disclosure Schedule discloses the amount of and year of expiration of each company's net operating loss carryovers.

(t) Credit Carryover. As of December 31, 1994, Delmarva and its subsidiaries had tax credit carryovers available to offset future tax liability as disclosed in Section 4.9(t) of the Delmarva Disclosure Schedule. Section 4.9(t) of the Delmarva Disclosure Schedule discloses the amount and year of expiration of each company's tax credit carryovers.

(u) Code § 338 Elections. Except as disclosed in Section 4.9(u) of the Delmarva Disclosure Schedule, no election under Code § 338 (or any predecessor provision) has been made by or with respect to Delmarva or any of its subsidiaries or any of their respective assets or properties.

(v) Acquisition Indebtedness. Except as disclosed in Section 4.9(v) of the Delmarva Disclosure Schedule, no indebtedness of Delmarva or any of its subsidiaries is "corporate acquisition indebtedness" within the meaning of Code § 279(b).

(w) Intercompany Transactions. Except as disclosed in Section 4.9(w) of the Delmarva Disclosure Schedule, neither Delmarva nor any of its subsidiaries have engaged in any intercompany transactions within the meaning of Treasury Regulations § 1.1502-13 for which any income or gain will remain unrecognized as of the close of the last taxable year prior to the Closing Date.

(x) Liability for Others. Except as disclosed in Section 4.9(x) of the Delmarva Disclosure Schedule, neither Delmarva nor any of its subsidiaries has any liability for Taxes of any person other than Delmarva and its subsidiaries (i) under Treasury Regulations Section 1.1502-6 (or any similar provision of state, local or foreign law) as a transferee or successor, (ii) by contract or (iii) otherwise.

#### Section 4.10 Employee Matters; ERISA.

(a) Benefit Plans. Section 4.10(a) of the Delmarva Disclosure Schedule contains a true and complete list of: (i) each employee benefit plan, program or arrangement covering employees, former employees or directors of Delmarva (or any of its subsidiaries) or any of their dependents or beneficiaries, or providing benefits to such persons in respect of services provided to any such entity, including, but not limited to, any "employee benefit plan" within the meaning of Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended ("ERISA") (whether or not terminated, if Delmarva or any of its subsidiaries could have statutory or contractual liability with respect thereto on or after the date hereof); (ii) each management, employment, deferred compensation, severance (including any payment, right or benefit resulting from a change in control), bonus or other contract for personal services with or covering any current officer, key employee or director or any consulting contract with any person who prior to entering into such contract was a director or officer of Delmarva or any of its subsidiaries (whether or not terminated, if Delmarva or any of its subsidiaries could have statutory or contractual liability with respect thereto on or after the date hereof); (iii) each "employee pension benefit plan" (within the meaning of ERISA § 3(2)) subject to Title IV of ERISA or the minimum funding requirements of Code § 412 maintained or contributed to by Delmarva or any entity required to be aggregated therewith pursuant to Code § 414(b) or (c) (each, a "Delmarva ERISA Affiliate") at any time during the seven-year period immediately preceding the date hereof (collectively, the "Delmarva Benefit Plans") and (iv) with respect to each Delmarva Benefit Plan, the source or sources of benefit payments under the plan (including, where applicable, the identity of any trust (whether or not a grantor trust), insurance contract, custodial account, agency agreement, or other arrangement that holds the assets of, or serves as a funding vehicle or source of benefits for, such Delmarva Benefit Plan).

(b) Contributions. Except as disclosed in Section 4.10(b) of the Delmarva Disclosure Schedule, all material contributions and other payments required to have been made by Delmarva or any of its subsidiaries pursuant to any Delmarva Benefit Plan (or to any person pursuant to the terms thereof) have been timely made or the amount of such payment or contribution obligation has been reflected in the Delmarva Financial Statements.

(c) *Qualification; Compliance.* Except as disclosed in Section 4.10(c) of the Delmarva Disclosure Schedule, each Delmarva Benefit Plan that is intended to be "qualified" within the meaning of Code § 401(a) has been determined by the IRS to be so qualified, and, to the best knowledge of Delmarva, no event or condition exists or has occurred that could reasonably be expected to result in the revocation of any such determination. Delmarva and each of its subsidiaries are in compliance with, and each Delmarva Benefit Plan is and has been operated in compliance with, all applicable laws, rules and regulations governing such plan, including without limitation ERISA and the Code, except for violations that could not reasonably be expected to have a Delmarva Material Adverse Effect. To the best knowledge of Delmarva, no individual or entity has engaged in any transaction with respect to any Delmarva Benefit Plan as a result of which Delmarva or any of its subsidiaries could reasonably expect to be subject to liability pursuant to ERISA § 409 or § 502, or subject to an excise tax pursuant to Code § 4975. To the best knowledge of Delmarva, (i) no Delmarva Benefit Plan is subject to any ongoing audit, investigation, or other administrative proceeding of the Internal Revenue Service, the Department of Labor, or any other federal, state or local governmental entity and (ii) no Delmarva Benefit Plan is the subject of any pending application for administrative relief under any voluntary compliance program of any governmental entity (including without limitation the IRS's Voluntary Compliance Resolution Program or Walk-in Closing Agreement Program, or the Department of Labor's Delinquent Filer Voluntary Compliance Program).

(d) Liabilities. With respect to the Delmarva Benefit Plans, individually and in the aggregate, no termination or partial termination of any Delmarva Benefit Plan or other event has occurred, and, to the best knowledge of Delmarva, there exists no condition or set of circumstances, that could subject Delmarva or any of its subsidiaries to any liability arising under the Code, ERISA or any other applicable law (including without limitation any liability to or under any such plan or to the Pension Benefit Guaranty

Corporation (the "PBGC"), or under any indemnity agreement to which Delmarva, any of its subsidiaries or any Delmarva ERISA Affiliate is a party, which liability, excluding liability for benefit claims and funding obligations payable in the ordinary course and liability for PBGC insurance premiums payable in the ordinary course, is reasonably likely to have a Delmarva Material Adverse Effect.

(e) Welfare Plans. Except as disclosed in Section 4.10(e) of the Delmarva Disclosure Schedule, no Delmarva Benefit Plan that is a "welfare plan" (within the meaning of ERISA § 3(1)) provides benefits for any retired or former employees (other than as required pursuant to ERISA § 601).

(f) Documents Made Available. Delmarva has made available to Atlantic a true and correct copy of each collective bargaining agreement to which Delmarva is a party or under which Delmarva has obligations and, with respect to each Delmarva Benefit Plan, as applicable (i) the current plan document (including all amendments adopted since the most recent restatement) and its most recently prepared summary plan description and all summaries of material modifications prepared since the most recent summary plan description, (ii) the most recently prepared annual report (IRS Form 5500 Series) including financial statements, (iii) each related trust agreement, insurance contract, service provider or investment management agreement (including all amendments to each such document), (iv) the most recent IRS determination letter with respect to the qualified status under Code § 401(a) of such plan and a copy of any application for an IRS determination letter filed since the most recent IRS determination letter was issued and (v) the most recent actuarial report or valuation.

(g) Payments Resulting from Mergers. Other than as disclosed in Section 4.10(g) of the Delmarva Disclosure Schedule, the consummation or announcement of any transaction contemplated by this Agreement will not (either alone or upon the occurrence of any additional or further acts or events) result in any (i) payment (whether of severance pay or otherwise) becoming due from the Company or Delmarva or any of its subsidiaries under any applicable Delmarva Benefit Plans to any officer, employee, former employee or director thereof or to the trustee under any "rabbi trust" or similar arrangement, or (ii) benefit under any Delmarva Benefit Plan being established or becoming accelerated, vested or payable, except for a payment or benefit that would have been payable under the same terms and conditions without regard to the transactions contemplated by this Agreement.

(h) Funded Status of Plans. Except as disclosed in Section 4.10(h) of the Delmarva Disclosure Schedule, each Delmarva Benefit Plan that is subject to either or both of the minimum funding requirements of ERISA § 302 or to Title IV of ERISA has assets that, as of the date hereof, have a fair market value equal to or exceeding the present value of the accrued benefit obligations thereunder on a termination basis, as of the date hereof based on the actuarial methods, tables and assumptions thereto-fore utilized by such plan's actuary in preparing such plan's most recently prepared actuarial valuation report, except to the extent that applicable law would require the use of different actuarial assumptions if such plan was to be terminated as of the date hereof. No Delmarva Benefit Plan subject to the minimum funding requirements of ERISA § 302 has incurred any "accumulated funding deficiency" (within the meaning of ERISA § 302).

(i) Multiemployer Plans. Except as disclosed in Section 4.10(i) of the Delmarva Disclosure Schedule, no Delmarva Benefit Plan is or was a "multiemployer plan" (within the meaning of ERISA § 4001(a)(3)), a multiple employer plan described in Code § 413(c), or a "multiple employer welfare arrangement" (within the meaning of ERISA § 3(40)); and none of Delmarva, any subsidiary thereof or any Delmarva ERISA Affiliate has been obligated to contribute to, or otherwise has or has had any liability with respect to, any multiemployer plan, multiple employer plan, or multiple employer welfare arrangement. With respect to any Delmarva Benefit Plan that is listed in Section 4.10(i) of the Delmarva Disclosure Schedule as a multiemployer plan, Delmarva and its subsidiaries have not made or incurred a "complete withdrawal" or a "partial withdrawal," as such terms are defined in ERISA § 4203 and 4205, therefrom at any time

during the five calendar year period immediately preceding the date of this Agreement and the transactions contemplated by the Agreement will not, in and of themselves, give rise to such a "complete withdrawal" or "partial withdrawal."

(j) Modification or Termination of Plans. Except as disclosed in Section 4.10(j) of the Delmarva Disclosure Schedule: (i) neither Delmarva nor any subsidiary of Delmarva is subject to any legal, contractual, equitable or other obligation to establish as of any date any employee benefit plan of any nature, including without limitation any pension, profit sharing, welfare, post-retirement welfare, stock option, stock or cash award, nonqualified deferred compensation or executive compensation plan, policy or practice; and (ii) to the best knowledge of Delmarva, after review of all Delmarva Benefit Plan documents, the Company, Delmarva or one or more of its subsidiaries may, in any manner, and without the consent of any employee, beneficiary or dependent, employees' organization or other person, terminate, modify or amend any Delmarva Benefit Plan or any other employee benefit plan, policy, program or practice) at any time sponsored, maintained or contributed to by Delmarva or any of its subsidiaries, effective as of any date before, on or after the Effective Time except to the extent that any retroactive amendment would be prohibited by ERISA § 204(g) or would deprive a plan participant of a benefit in which such participant has a vested right.

(k) Reportable Events; Claims. Except as disclosed in Section 4.10(k) of the Delmarva Disclosure Schedule, (i) no event constituting a "reportable event" (within the meaning of ERISA § 4043(b)) for which the 30-day notice requirement has not been waived by the PBGC has occurred with respect to any Delmarva Benefit Plan and (ii) no liability, claim, action or litigation has been made, commenced or, to the best knowledge of Delmarva, threatened, by or against Delmarva or any of its subsidiaries with respect to any Delmarva Benefit Plan (other than for benefits or PBGC premiums payable in the ordinary course) that is reasonably likely to have a Delmarva Material Adverse Effect.

(1) Labor Agreements. Except as disclosed in Section 4.10(1) of the Delmarva Disclosure Schedule, to the best knowledge of Delmarva, as of the date hereof, there is no current labor union representation question involving employees of Delmarva or any of its subsidiaries, nor does Delmarva or any of its subsidiaries know of any activity or proceeding of any labor organization (or representative thereof) or employee group (or representative thereof) to organize any such employees. Except as disclosed in the Delmarva SEC Reports or as disclosed in Section 4.10(1) of the Delmarva Disclosure Schedule: (i) neither Delmarva nor any of its subsidiaries is a party to any collective bargaining agreement or other labor agreement with any union or labor organization; (ii) there are no unfair labor practice charges or grievances arising out of a collective bargaining agreement or other grievance procedure against Delmarva or any of its subsidiaries pending, or to the best knowledge of Delmarva threatened, that, in the aggregate, are reasonably likely to have a Delmarva Material Adverse Effect; (iii) there are no complaints, lawsuits or proceedings in any forum or forums by or on behalf of any present or former employees, any applicants for employment or classes of the foregoing alleging breach of any express or implied contract of employment, any law or regulation governing employment or the termination thereof or other discriminatory, wrongful or tortious conduct in connection with the employment relationship against Delmaiva or any of its subsidiaries pending, or to the best knowledge of Delmarva threatened, that, in the aggregate, are reasonably likely to have, a Delmarva Material Adverse Effect; (iv) there are no strikes, disputes, slowdowns, work stoppages or lockouts pending, or to the best knowledge of Delmarva threatened, against or involving Delmarva or any of its subsidiaries that, in the aggregate, are reasonably likely to have a Delmarva Material Adverse Effect; (v) Delmarva and each of its subsidiaries are in compliance with all applicable laws respecting employment and employment practices, terms and conditions of employment, wages, hours of work and occupational safety and health, except for noncompliance that, in the aggregate, is not reasonably likely to have a Delmarva Material Adverse Effect; and (vi) there is no proceeding, claim, suit, action or governmental investigation pending, or to the best knowledge of Delmarva threatened, in respect to which any director, officer, employee or agent of Delmarva or any of its subsidiaries is or may be entitled to claim indemnification from Delmarva or any of its subsidiaries pursuant to their respective certificates or articles of incorporation or bylaws or as provided in the indemnification agreements listed on Section 4.10(l) of the Delmarva Disclosure Schedule.

### Section 4.11 Environmental Protection.

(a) Compliance. Except as disclosed in Section 4.11(a) of the Delmarva Disclosure Schedule or as disclosed in the Delmarva SEC Reports, Delmarva and each of its subsidiaries are and have been in material compliance with all applicable Environmental Laws (as defined in Section 4.11(g)), except where the failure to be or to have been so in material compliance, in the aggregate, is not reasonably likely to have a Delmarva Material Adverse Effect. Except as disclosed in Section 4.11(a) of the Delmarva Disclosure Schedule, neither Delmarva nor any of its subsidiaries has received any written notice from any person or Governmental Authority that alleges that Delmarva or any of its subsidiaries is not or has not been in material compliance with applicable Environmental Laws, except where the failure to be or to have been so in material compliance, in the aggregate, is not reasonably likely to have a Delmarva Material Adverse Effect.

(b) Environmental Permits. Except as disclosed in Section 4.11(b) of the Delmarva Disclosure Schedule or as disclosed in the Delmarva SEC Reports, Delmarva and each of its subsidiaries have obtained or have applied for all material environmental, health and safety permits and authorizations (collectively, "Environmental Permits") necessary for the construction of their facilities and the conduct of their operations, and all such Environmental Permits are in good standing or, where applicable, a renewal application has been timely filed and is pending agency approval, and Delmarva and its subsidiaries are in material compliance with all terms and conditions of all such Environmental Permits and are not required to make any material expenditures in connection with any renewal application pending agency approval, except where the failure to obtain or be in such compliance and the requirement to make such expenditures, in the aggregate, is not reasonably likely to have a Delmarva Material Adverse Effect.

(c) Environmental Claims. Except as disclosed in Section 4.11(c) of the Delmarva Disclosure Schedule or as disclosed in the Delmarva SEC Reports, to the best knowledge of Delmarva, there are no Environmental Claims (as hereinafter defined in Section 4.11(g)) pending, or to the best knowledge of Delmarva threatened, (i) against Delmarva or any of its subsidiaries or joint ventures, (ii) against any person or entity whose liability for any Environmental Claim Delmarva or any of its subsidiaries or joint ventures has or may have retained or assumed either contractually or by operation of law or (iii) against any real or personal property or operations that Delmarva or any of its subsidiaries or joint ventures owns, leases or manages, in whole or in part, that, if adversely determined, are in the aggregate reasonably likely to have a Delmarva Material Adverse Effect.

(d) *Releases.* Except as disclosed in Section 4.11(c) or 4.11(d) of the Delmarva Disclosure Schedule or as disclosed in the Delmarva SEC Reports, to the best knowledge of Delmarva, there has been no Release (as hereinafter defined in Section 4.11(g)) of any Hazardous Material (as hereinafter defined in Section 4.11(g)) that would be reasonably likely to form the basis of any Environmental Claim against Delmarva or of any subsidiary or joint venture of Delmarva, or against any person or entity whose liability for any Environmental Claim Delmarva or any subsidiary or joint venture of Delmarva has or may have retained or assumed either contractually or by operation of law, except for Releases of Hazardous Materials the liability for which is not in the aggregate reasonably likely to have a Delmarva Material Adverse Effect.

(e) *Predecessors.* Except as disclosed in Section 4.11(e) of the Delmarva Disclosure Schedule, or as disclosed in the Delmarva SEC Reports, to the best knowledge of Delmarva, with respect to any predecessor of Delmarva or of any subsidiary or joint venture of Delmarva, there are no Environmental Claims pending or threatened, or any Releases of Hazardous Materials that would be reasonably likely to form the basis of any Environmental Claims, that are reasonably likely to have, in the aggregate, a Delmarva Material Adverse Effect.



(f) Disclosure. To the best knowledge of Delmarva, Delmarva has disclosed to Atlantic all material facts that Delmarva reasonably believes are likely to form the basis of a material Environmental Claim or to require material expenditures in order to comply with current or future applicable Environmental Laws arising from (i) the cost of pollution control equipment currently required or known to be required in the future, (ii) current investigatory, removal, remediation or response costs or investigatory, removal, remediation or response costs known to be required in the future, in each case, both on-site and off-site and (iii) any other environmental matters affecting Delmarva or its subsidiaries.

(g) As used in this Agreement:

(i) "Environmental Claim" means any and all administrative, regulatory or judicial actions, suits, demands, demand letters, directives, claims, liens, investigations, proceedings or notices of noncompliance or violation by any person or entity (including without limitation any Governmental Authority) alleging potential liability (including without limitation potential liability for enforcement costs, investigatory costs, cleanup costs, response costs, removal costs, remedial costs, natural resources damages, property damages, personal injuries, fines or penalties) arising out of, based on or resulting from (A) the presence, or Release or threatened Release, of any Hazardous Materials at any location, whether or not owned, operated, leased or managed by Delmarva or any of its subsidiaries or joint ventures (for purposes of Section 5.11 only), (B) circumstances forming the basis of any violation, or alleged violation, of any Environmental Law or (C) any and all claims by any third party seeking damages, contribution, indemnification, cost recovery, compensation or injunctive relief resulting from the presence or Release of any Hazardous Materials.

(ii) "Environmental Laws" means all federal, state and local laws, rules and regulations relating to pollution or protection of human health or the environment (including without limitation ambient air, surface water, groundwater, land surface or subsurface strata), including without limitation laws and regulations relating to Releases or threatened Releases of Hazardous Materials or otherwise relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of Hazardous Materials.

(iii) "Hazardous Materials" means (A) any petroleum or petroleum products or petroleum wastes (including crude oil or any fraction thereof); nuclear fuel or waste or other radioactive materials, friable asbestos or friable asbestos-containing material, urea formaldehyde foam insulation, and transformers or other equipment that contain dielectric fluid containing polychlorinated biphenyls, (B) any chemicals, materials or substances which are now defined as or included in the definition of "hazardous substances", "hazardous wastes", "hazardous materials", "extremely hazardous wastes", "restricted hazardous wastes", "toxic substances", "toxic pollutants", or words of similar import, under any Environmental Law and (C) any other chemical, material, substance or waste, exposure to which is now prohibited, limited or regulated under any Environmental Law in a jurisdiction in which Delmarva or any of its subsidiaries or joint ventures operates (for purposes of this Section 4.11 only) or in which Atlantic or any of its subsidiaries or joint ventures operates (for purposes of Section 5.11 only).

(iv) "Release" means any release, spill, emission, leaking, injection, deposit, disposal, discharge, dispersal, leaching or migration into the atmosphere, soil, surface water, groundwater or property (indoors or outdoors).

Section 4.12 *Regulation as a Utility*. Delmarva is regulated as a public utility in the States of Delaware and Maryland and in the Commonwealth of Virginia and in no other state. Except as disclosed in Section 4.12 of the Delmarva Disclosure Schedule, neither Delmarva nor any "subsidiary company" or "affiliate" (as such terms are defined in the 1935 Act) of Delmarva is subject to regulation as a public utility or public service company (or similar designation) by any other state in the United States, by the United States or any agency or instrumentality of the United States or by any foreign country. Delmarva is not a holding company under the 1935 Act.

Section 4.13 *Vote Required.* The approval of the Delmarva Merger by more than two-thirds of all votes entitled to be cast by all holders of Delmarva Common Stock (or such lesser percentage as may be required as a result of any change of applicable law) (the "Delmarva Shareholders' Approval") is the only vote of the holders of any class or series of the capital stock of Delmarva required to approve this Agreement, the Mergers and the other transactions contemplated hereby.

Section 4.14 *Opinion of Financial Advisor*. Delmarva has received the opinion of Merrill Lynch, Pierce, Fenner & Smith Incorporated, dated the date hereof, to the effect that, as of the date hereof, the consideration to be received by the holders of Delmarva Common Stock are fair from a financial point of view to the holders of Delmarva Common Stock.

Section 4.15 *Insurance*. Except as disclosed in Section 4.15 of the Delmarva Disclosure Schedule, each of Delmarva and each of its subsidiaries is, and has been continuously since January 1, 1991, insured with financially responsible insurers in such amounts and against such risks and losses as are customary for companies conducting the respective businesses conducted by Delmarva and its subsidiaries during such time period. Except as disclosed in Section 4.15 of the Delmarva Disclosure Schedule, neither Delmarva nor any of its subsidiaries has received any notice of cancellation or termination with respect to any material insurance policy thereof. All material insurance policies of Delmarva and its subsidiaries are valid and enforceable policies.

Section 4.16 Applicability of Certain Delaware and Virginia Law. Assuming the accuracy of the representation of Atlantic set forth in Section 5.17, none of the control share acquisition restrictions of Section 13.1-728.1 et seq. of the VSCA, the business combination restrictions of Section 203 of the DGCL or the affiliated transactions restrictions of Section 13.1-725 et seq. of the VSCA is applicable to the transactions contemplated by this Agreement.

Section 4.17 *Ownership of Atlantic Common Stock*. Delmarva does not "beneficially own" (as such term is defined in Rule 13d-3 under the Exchange Act) any shares of Atlantic Common Stock.

Section 4.18 *Operations of Delmarva Nuclear Power Plants.* Except as set forth in Section 4.18 of the Delmarva Disclosure Schedule, to the best knowledge of Delmarva, the operations of the Salem and Peach Bottom Nuclear Generating Stations owned by Delmarva (together with Atlantic) (collectively, the "Delmarva Facilities") are and have at all times been conducted in compliance with applicable health, safety, regulatory and other legal requirements, except where the failure to be in compliance in the aggregate does not and insofar as can reasonably be foreseen would not have a Delmarva Material Adverse Effect. To the best knowledge of Delmarva, each of the Delmarva Facilities maintains emergency plans designed to respond to an unplanned release therefrom of radioactive materials into the environment and customary liability insurance consistent with industry practice and consistent with Delmarva's view of the risks inherent in the operation of a nuclear power facility currently exists with respect to such Delmarva Facilities and for the short-term storage of spent nuclear fuel conform with the requirements of applicable law, and such plans have at all times been funded consistently with reasonable budget projections for such plans.

## ARTICLE V

## **REPRESENTATIONS AND WARRANTIES OF ATLANTIC**

Atlantic represents and warrants to Delmarva as follows:

Section 5.1 Organization and Qualification. Except as disclosed in Section 5.1 of the Atlantic Disclosure Schedule (as defined in Section 7.6(i)), (i) Atlantic is a corporation duly organized, validly existing and in good standing under the laws of New Jersey, (ii) each of Atlantic's subsidiaries is a corporation or partnership duly organized, validly existing and in good standing under the laws of its jurisdiction of organization and (iii) each of Atlantic and its subsidiaries has all requisite corporate or partnership power and authority, and has been duly authorized by all necessary regulatory approvals and orders, to own, lease and operate its assets and properties and to carry on its business as it is now being conducted, and is duly qualified and in good standing to do business in each jurisdiction in which the nature of its business or the ownership or leasing of its assets and properties makes such qualification necessary other than in such jurisdictions where the failure to be so qualified and in good standing will not, when taken together with all other such failures, have a material adverse effect on the business, operations, properties, assets, condition (financial or otherwise), prospects or results of operations of Atlantic and its subsidiaries taken as a whole or on the consummation of this Agreement (any such material adverse effect being hereinafter referred to as a "Atlantic Material Adverse Effect").

Section 5.2 Subsidiaries. Section 5.2 of the Atlantic Disclosure Schedule contains a description as of the date hereof of all subsidiaries and joint ventures of Atlantic, including the name of each such entity, the state or jurisdiction of its organization, a brief description of the principal line or lines of business conducted by each such entity and Atlantic's interest therein. Except as disclosed in Section 5.2 of the Atlantic Disclosure Schedule, none of such entities is a "public utility company," a "holding company," a "subsidiary company" or an "affiliate" of any public utility company within the meaning of Section 2(a)(5), 2(a)(7), 2(a)(8) and 2(a)(11) of the 1935 Act, respectively. Except as disclosed in Section 5.2 of the Atlantic Disclosure Schedule, all of the issued and outstanding shares of capital stock of each subsidiary of Atlantic are validly issued, fully paid, nonassessable and free of preemptive rights and are owned directly or indirectly by Atlantic free and clear of any liens, claims, encumbrances, security interests, equities, charges and options of any nature whatsoever, and there are no outstanding subscriptions, options, calls, contracts, voting trusts, proxies or other commitments, understandings, restrictions, arrangements, rights or warrants, including any right of conversion or exchange under any outstanding security, instrument or other agreement, obligating any such subsidiary to issue, deliver or sell, or cause to be issued, delivered or sold, additional shares of its capital stock or obligating it to grant, extend or enter into any such agreement or commitment.

Section 5.3 *Capitalization*. The authorized capital stock of Atlantic consists of 75,000,000 shares of Atlantic Common Stock. As of the close of business on June 30, 1996, 52,702,052 shares of Atlantic Common Stock were issued and outstanding. All of the issued and outstanding shares of the capital stock of Atlantic are validly issued, fully paid, nonassessable and free of preemptive rights. Except as disclosed in Section 5.3 of the Atlantic Disclosure Schedule, as of the date hereof, there are no outstanding subscriptions, options, calls, contracts, voting trusts, proxies or other commitments, understandings, restrictions, arrangements, rights or warrants, including any right of conversion or exchange under any outstanding security, instrument or other agreement, obligating Atlantic or any of its subsidiaries to issue, deliver or sell, or cause to be issued, delivered or sold, additional shares of the capital stock or other voting securities of Atlantic or obligating Atlantic or any of its subsidiaries to grant, extend or enter into any such agreement or commitment.

## Section 5.4 Authority; Noncontravention; Statutory Approvals; Compliance.

(a) Authority. Atlantic has all requisite power and authority to enter into this Agreement and, subject to the Atlantic Shareholders' Approval (as defined in Section 5.13) and the Atlantic Required Statutory Approvals (as defined in Section 5.4(c), to consummate the transactions contemplated hereby. The execution and delivery of this Agreement and the consummation by Atlantic of the transactions contemplated hereby have been duly authorized by all necessary corporate action on the part of Atlantic, subject to obtaining the Atlantic Shareholders' Approval. This Agreement has been duly and validly executed and delivered by Atlantic and, assuming the due authorization, execution and delivery of this Agreement by the other signatories hereto, constitutes the legal, valid and binding obligation of Atlantic enforceable against Atlantic in accordance with its terms.

(b) Noncontravention. Except as disclosed in Section 5.4(b) of the Atlantic Disclosure Schedule, the execution and delivery of this Agreement by Atlantic do not, and the consummation of the transactions contemplated hereby will not result in any Violation by Atlantic or any of its subsidiaries or, to the best knowledge of Atlantic, any of its joint ventures under any provisions of (i) the certificate or articles of incorporation, bylaws or similar governing documents of Atlantic or any of its subsidiaries or joint ventures, (ii) subject to obtaining the Atlantic Required Statutory Approvals and the receipt of the Atlantic Shareholders' Approval, any statute, law, ordinance, rule, regulation, judgment, decree, order, injunction, writ, permit or license of any Governmental Authority applicable to Atlantic or any of its subsidiaries or joint ventures or any of their respective properties or assets, or (iii) subject to obtaining the third-party consents or other approvals set forth in Section 5.4(b) of the Atlantic Disclosure Schedule (the "Atlantic Required Consents"), any note, bond, mortgage, indenture, deed of trust, license, franchise, permit, concession, contract, lease or other instrument, obligation or agreement of any kind to which Atlantic or any of its subsidiaries or joint ventures is now a party or by which any of them or any of their respective properties or assets may be bound or affected, excluding from the foregoing clauses (ii) and (iii) such Violations as, in the aggregate, are not reasonably likely to have an Atlantic Material Adverse Effect.

(c) Statutory Approvals. Except as disclosed in Section 5.4(c) of the Atlantic Disclosure Schedule, no declaration, filing or registration with, or notice to or authorization, consent, finding by or approval of, any Governmental Authority is necessary for the execution and delivery of this Agreement by Atlantic or the consummation by Atlantic of the transactions contemplated hereby, the failure to obtain which, in the aggregate, is reasonably likely to have an Atlantic Material Adverse Effect (the "Atlantic Required Statutory Approvals"), it being understood that references in this Agreement to "obtaining" such Atlantic Required Statutory Approvals shall mean making such declarations, filings or registrations, giving such notice, obtaining such authorizations, consents or approvals and having such waiting periods expire as are, in each case, necessary to avoid a violation of law.

(d) Compliance. Except as disclosed in Section 5.4(d), 5.7, 5.10 or 5.11 of the Atlantic Disclosure Schedule or as disclosed in the Atlantic SEC Reports (as defined in Section 5.5), neither Atlantic nor any of its subsidiaries nor, to the best knowledge of Atlantic, any of its joint ventures is in violation of or under investigation with respect to, or has been given notice or been charged with any violation of, any law, statute, order, rule, regulation, ordinance or judgment (including without limitation any applicable Environmental Laws) of any Governmental Authority, except for violations that, in the aggregate, do not have, and to the best knowledge of Atlantic are not reasonably likely to have, an Atlantic Material Adverse Effect. Except as disclosed in Section 5.4(d) or 5.11 of the Atlantic Disclosure Schedule, Atlantic, its subsidiaries and, to the best knowledge of Atlantic, its joint ventures have all Permits, except those the failure to obtain which, in the aggregate, are not reasonably likely to have an Atlantic Material Adverse Effect. Except as set forth in Section 5.4(d) or 5.11 of the Atlantic Disclosure Schedule, neither Atlantic nor any of its subsidiaries nor, to the best knowledge of Atlantic, any of 5.11 of the Atlantic Disclosure Schedule, neither Atlantic nor any of its subsidiaries nor, to the best knowledge of Atlantic, any of its joint ventures, is in breach or violation of or in default in the performance or observance of any term or provision of, and no event has occurred which, with lapse of time or action by a third party, could result in a default under (i) its certificate or articles of incorporation or its bylaws or (ii) except where such breaches, violations and

defaults, in the aggregate, are not reasonably likely to have an Atlantic Material Adverse Effect, any contract, commitment, agreement, indenture, mortgage, loan agreement, note, lease, bond, license, approval or other instrument to which it is a party or by which it is bound or to which any of its property is subject.

Section 5.5 Reports and Financial Statements. Except as disclosed in Section 5.5 of the Atlantic Disclosure Schedule, the filings required to be made by Atlantic and its subsidiaries since January 1, 1991 under the Securities Act, the Exchange Act, applicable New Jersey and Pennsylvania laws and regulations, the Power Act, the 1935 Act and the Atomic Energy Act have been filed with the SEC, the New Jersey Board of Public Utilities (the "New Jersey Commission"), the Pennsylvania Commission, the FERC and the NRC, respectively, including all forms, statements, reports, agreements (oral or written) and all documents, exhibits, amendments and supplements appertaining thereto, and, as of their respective dates, complied in all material respects with all applicable requirements of the appropriate act and the rules and regulations thereunder. Atlantic has made available to Delmarva a true and complete copy of each report, schedule, registration statement and definitive proxy statement filed by Atlantic with the SEC since January 1, 1991 and through the date hereof (as such documents have since the time of their filing been amended, the "Atlantic SEC Reports"). The Atlantic SEC Reports, including without limitation any financial statements or schedules included therein, at the time filed, and any forms, reports or other documents filed by Atlantic with the SEC after the date hereof, did not and will not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. The audited consolidated financial statements and unaudited interim financial statements of Atlantic included in the Atlantic SEC Reports (collectively, the "Atlantic Financial Statements") have been prepared, and will be prepared, in accordance with GAAP (except as may be indicated therein or in the notes thereto and except with respect to unaudited statements as permitted by Form 10-Q) and fairly present the consolidated financial position of Atlantic as of the respective dates thereof or the consolidated results of operations and cash flows for the respective periods then ended, as the case may be, subject, in the case of the unaudited interim financial statements, to normal, recurring audit adjustments. True accurate and complete copies of the certificate of incorporation and bylaws of Atlantic, as in effect on the date hereof, have been delivered to Delmarva.

Section 5.6 Absence of Certain Changes or Events; Absence of Undisclosed Liabilities. (a) Except as disclosed in the Atlantic SEC Reports filed prior to the date hereof or as disclosed in Section 5.6 of the Atlantic Disclosure Schedule, from December 31, 1995 through the date hereof, each of Atlantic and each of its subsidiaries has conducted its business only in the ordinary course of business consistent with past practice and no event has occurred which has had, and no fact or condition exists that would have or, to the best knowledge of Atlantic, is reasonably likely to have, an Atlantic Material Adverse Effect.

(b) Neither Atlantic nor any of its subsidiaries has any liabilities or obligations (whether absolute, contingent, accrued or otherwise) of a nature required by GAAP to be reflected in a consolidated corporate balance sheet, except liabilities, obligations or contingencies that are accrued or reserved against in the consolidated financial statements of Atlantic or are reflected in the notes thereto for the year ended December 31, 1995 or that were incurred after December 31, 1995 in the ordinary course of business and are not reasonably likely to have an Atlantic Material Adverse Effect.

Section 5.7 *Litigation*. Except as disclosed in the Atlantic SEC Reports filed prior to the date hereof or as disclosed in Section 5.7, 5.9, 5.10 or 5.11 of the Atlantic Disclosure Schedule, (i) there are no claims, suits, actions or proceedings pending or, to the best knowledge of Atlantic, threatened, nor are there any investigations or reviews pending or, to the best knowledge of Atlantic, threatened against, relating to or affecting Atlantic or any of its subsidiaries, (ii) there have not been any developments since December 31, 1995 with respect to any such disclosed claims, suits, actions, proceedings, investigations or reviews and (iii) there are no judgments, decrees, injunctions, rules or orders of any court, governmental department,

commission, agency, instrumentality or authority or any arbitrator applicable to Atlantic or any of its subsidiaries that, in the aggregate, are reasonably likely to have an Atlantic Material Adverse Effect.

Section 5.8 Registration Statement and Proxy Statement. None of the information supplied or to be supplied by or on behalf of Atlantic for inclusion or incorporation by reference in (i) the Registration Statement will, at the time the Registration Statement becomes effective under the Securities Act, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading and (ii) the Joint Proxy Statement will, at the date mailed to the shareholders of Atlantic and Delmarva and, as the same may be amended or supplemented, at the times of the meetings of such shareholders to be held in connection with the Mergers, contain any untrue statement of a material fact or omit to state any material fact necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading. The Registration Statement and the Joint Proxy Statement will comply as to form in all material respects with the provisions of the Securities Act and the Exchange Act and the rules and regulations thereunder.

Section 5.9 Tax Matters.

(a) Filing of Timely Tax Returns. Except as disclosed in Section 5.9(a) of the Atlantic Disclosure Schedule, Atlantic and each of its subsidiaries have filed all Tax Returns required to be filed by each of them under applicable law. All Tax Returns were in all material respects (and, as to Tax Returns not filed as of the date hereof, will be) true, complete and correct and filed on a timely basis.

(b) Payment of Taxes. Except as disclosed in Section 5.9(b) of the Atlantic Disclosure Schedule, Atlantic and each of its subsidiaries have, within the time and in the manner prescribed by law, paid (and until the Closing Date will pay within the time and in the manner prescribed by law) all Taxes that are currently due and payable except for those contested in good faith and for which adequate reserves have been taken.

(c) Tax Reserves. Atlantic and its subsidiaries have established (and until the Closing Date will maintain) on their books and records reserves adequate to pay all Taxes and reserves for deferred income taxes in accordance with GAAP.

(d) Tax Liens. There are no Tax liens upon the assets of Atlantic or any of its subsidiaries except liens for Taxes not yet due.

(e) *Withholding Taxes*. Atlantic and each of its subsidiaries have complied (and until the Closing Date will comply) in all material respects with the provisions of the Code relating to the payment and withholding of Taxes, including without limitation the withholding and reporting requirements under Code §§ 1441 through 1464, 3401 through 3606, and 6041 and 6049, as well as similar provisions under any other laws, and have within the time and in the manner prescribed by law withheld from employee wages and paid over to the proper governmental authorities all amounts required.

(f) *Extensions of Time for Filing Tax Returns*. Except as disclosed in Section 5.9(f) of the Atlantic Disclosure Schedule, neither Atlantic nor any of its subsidiaries has requested any extension of time within which to file any Tax Return which Tax Return has not since been filed.

(g) Waivers of Statute of Limitations. Except as disclosed in Section 5.9(g) of the Atlantic Disclosure Schedule, neither Atlantic nor any of its subsidiaries has executed any outstanding waivers or comparable consents regarding the application of the statute of limitations with respect to any Taxes or Tax Returns.

(h) *Expiration of Statute of Limitations.* Except as disclosed in Section 5.9(h) of the Atlantic Disclosure Schedule, the statute of limitations for the assessment of all Taxes has expired for all applicable Tax Returns of Atlantic and each of its subsidiaries or those Tax Returns have been examined by the appropriate taxing authorities for all periods through the date hereof, and no deficiency for any Taxes has been proposed, asserted or assessed against Atlantic or any of its subsidiaries that has not been resolved and paid in full.



(i) Audit, Administrative and Court Proceedings. Except as disclosed in Section 5.9(i) of the Atlantic Disclosure Schedule, no audits or other administrative proceedings or court proceedings are presently pending with regard to any Taxes or Tax Returns of Atlantic or any of its subsidiaries.

(j) Powers of Attorney. Except as disclosed in Section 5.9(j) of the Atlantic Disclosure Schedule, no power of attorney currently in force has been granted by Atlantic or any of its subsidiaries concerning any Tax matter.

(k) Tax Rulings. Except as disclosed in Section 5.9(k) of the Atlantic Disclosure Schedule, neither Atlantic nor any of its subsidiaries has received a Tax Ruling or entered into a Closing Agreement with any taxing authority that would have a continuing adverse effect after the Closing Date.

(1) Availability of Tax Returns. Atlantic and its subsidiaries have made available to Delmarva complete and accurate copies, covering all open years, of (i) all Tax Returns, and any amendments thereto, filed by Atlantic or any of its subsidiaries, (ii) all audit reports received from any taxing authority relating to any Tax Return filed by Atlantic or any of its subsidiaries and (iii) any Closing Agreements entered into by Atlantic or any of its subsidiaries with any taxing authority.

(m) Tax Sharing Agreements. Except as disclosed in Section 5.9(m) of the Atlantic Disclosure Schedule, no agreements relating to the allocation or sharing of Taxes exist between or among Atlantic and any of its subsidiaries.

(n) Code § 341(f). Neither Atlantic nor any of its subsidiaries has filed a consent pursuant to Code § 341(f) or has agreed to have Code § 341(f)(2) apply to any disposition of a subsection (f) asset (as such term is defined in Code § 341(f)(4)) owned by Atlantic or any of its subsidiaries.

(o) Code § 168. Except as disclosed in Section 5.9(o) of the Atlantic Disclosure Schedule, no property of Atlantic or any of its subsidiaries is property that Atlantic or any such subsidiary or any party to this transaction is or will be required to treat as being owned by another person pursuant to the provisions of Code § 168(f)(8) (as in effect prior to its amendment by the Tax Reform Act of 1986) or is tax-exempt use property within the meaning of Code § 168.

(p) Code § 481 Adjustments. Except as disclosed in Section 5.9(p) of the Atlantic Disclosure Schedule, neither Atlantic nor any of its subsidiaries is required to include in income any adjustment pursuant to Code § 481(a) by reason of a voluntary change in accounting method initiated by Atlantic or any of its subsidiaries, and to the best of the knowledge of Atlantic, the IRS has not proposed any such adjustment or change in accounting method.

(q) Code §§ 6661 and 6662. Except as disclosed in Section 5.9(q) of the Atlantic Disclosure Schedule, all transactions that could give rise to an understatement of federal income tax (within the meaning of Code § 6661 for Tax Returns filed on or before December 31, 1990, and within the meaning of Code § 6662 for tax returns filed after December 31, 1990) that could reasonably be expected to result in an Atlantic Material Adverse Effect have been adequately disclosed (or, with respect to Tax Returns filed following the Closing will be adequately disclosed) on the Tax Returns of Atlantic and its subsidiaries in accordance with Code § 6661(b)(2)(B) for Tax Returns filed on or prior to December 31, 1990, and in accordance with Code § 6662(d)(2)(B) for Tax Returns filed after December 31, 1990.

(r) Code § 280G. Except as disclosed in Section 5.9(r) of the Atlantic Disclosure Schedule, neither Atlantic nor any of its subsidiaries is a party to any agreement, contract, or arrangement that could reasonably be expected to result; on account of the transactions contemplated hereunder, separately or in the aggregate, in the payment of any "excess parachute payment" within the meaning of Code § 280G.

(s) NOLS. As of December 31, 1994, Atlantic and its subsidiaries had net operating loss carryovers available to offset future income as disclosed in Section 5.9(s) of the Atlantic Disclosure Schedule. Section 5.9(s) of the Atlantic Disclosure Schedule discloses the amount of and year of expiration of each company's net operating loss carryovers.

(t) Credit Carryover. As of December 31, 1994, Atlantic and its subsidiaries had tax credit carryovers available to offset future tax liability as disclosed in Section 5.9(t) of the Atlantic Disclosure Schedule. Section 5.9(t) of the Atlantic Disclosure Schedule discloses the amount and year of expiration of each company's tax credit carryovers.

(u) Code § 338 Elections. Except as disclosed in Section 5.9(u) of the Atlantic Disclosure Schedule, no election under Code § 338 (or any predecessor provision) has been made by or with respect to Atlantic or any of its subsidiaries or any of their respective assets or properties.

(v) Acquisition Indebtedness. Except as disclosed in Section 5.9(v) of the Atlantic Disclosure Schedule, no indebtedness of Atlantic or any of its subsidiaries is "corporate acquisition indebtedness" within the meaning of Code § 279(b).

(w) Intercompany Transactions. Except as disclosed in Section 5.9(w) of the Atlantic Disclosure Schedule, neither Atlantic nor any of its subsidiaries have engaged in any intercompany transactions within the meaning of Treasury Regulations § 1.1502-13 for which any income or gain will remain unrecognized as of the close of the last taxable year prior to the Closing Date.

(x) Liability for Others. Except as disclosed in Section 5.9(x) of the Atlantic Disclosure Schedule, neither Atlantic nor any of its subsidiaries has any liability for Taxes of any person other than Atlantic and its subsidiaries (i) under Treasury Regulations Section 1.1502-6 (or any similar provision of state, local or foreign law) as a transferee or successor, (ii) by contract or (iii) otherwise.

#### Section 5.10 Employee Matters; ERISA.

(a) Benefit Plans. Section 5.10(a) of the Atlantic Disclosure Schedule contains a true and complete list of: (i) each employee benefit plan, program or arrangement covering employees, former employees or directors of Atlantic (or any of its subsidiaries) or any of their dependents or beneficiaries, or providing benefits to such persons in respect of services provided to any such entity, including, but not limited to, any "employee benefit plan" within the meaning of ERISA § 3(3) (whether or not terminated, if Atlantic or any of its subsidiaries could have statutory or contractual liability with respect thereto on or after the date hereof); (ii) each management, employment, deferred compensation, severance (including any payment, right or benefit resulting from a change in control), bonus or other contract for personal services with or covering any current officer, key employee or director or any consulting contract with any person who prior to entering into such contract was a director or officer of Atlantic or any of its subsidiaries (whether or not terminated, if Atlantic or any of its subsidiaries could have statutory or contractual liability with respect thereto on or after the date hereof); (iii) each "employee pension benefit plan" (within the meaning of ERISA § 3(2)) subject to Title IV of ERISA or the minimum funding requirements of Code § 412 maintained or contributed to by Atlantic or any entity required to be aggregated therewith pursuant to Code § 414(b) or (c) (each, an "Atlantic ERISA Affiliate") at any time during the seven-year period immediately preceding the date hereof (collectively, the "Atlantic Benefit Plans") and (iv) with respect to each Atlantic Benefit Plan, the source or sources of benefit payments under the plan (including, where applicable, the identity of any trust (whether or not a grantor trust), insurance contract, custodial account, agency agreement, or other arrangement that holds the assets of, or serves as a funding vehicle or source of benefits for, such Atlantic Benefit Plan).

(b) *Contributions.* Except as disclosed in Section 5.10(b) of the Atlantic Disclosure Schedule, all material contributions and other payments required to have been made by Atlantic or any of its subsidiaries pursuant to any Atlantic Benefit Plan (or to any person pursuant to the terms thereof) have been timely made or the amount of such payment or contribution obligation has been reflected in the Atlantic Financial Statements.

(c) *Qualification; Compliance.* Except as disclosed in Section 5.10(c) of the Atlantic Disclosure Schedule, each Atlantic Benefit Plan that is intended to be "qualified" within the meaning of Code § 401(a) has been determined by the IRS to be so qualified, and, to the best knowledge of Atlantic, no

event or condition exists or has occurred that could reasonably be expected to result in the revocation of any such determination. Atlantic and each of its subsidiaries are in compliance with, and each Atlantic Benefit Plan is and has been operated in compliance with, all applicable laws, rules and regulations governing such plan, including without limitation ERISA and the Code, except for violations that could not reasonably be expected to have an Atlantic Material Adverse Effect. To the best knowledge of Atlantic, no individual or entity has engaged in any transaction with respect to any Atlantic Benefit Plan as a result of which Atlantic or any of its subsidiaries could reasonably expect to be subject to liability pursuant to ERISA § 409 or § 502, or subject to an excise tax pursuant to Code § 4975. To the best knowledge of Atlantic, (i) no Atlantic Benefit Plan is subject to any ongoing audit, investigation, or other administrative proceeding of the Internal Revenue Service, the Department of Labor, or any other federal, state or local governmental entity and (ii) no Atlantic Benefit Plan is the subject of any pending application for administrative relief under any voluntary compliance program of any governmental entity (including without limitation the IRS's Voluntary Compliance Resolution Program or Walk-in Closing Agreement Program, or the Department of Labor's Delinquent Filer Voluntary Compliance Program).

(d) Liabilities. With respect to the Atlantic Benefit Plans, individually and in the aggregate, no termination or partial termination of any Atlantic Benefit Plan or other event has occurred, and, to the best knowledge of Atlantic, there exists no condition or set of circumstances, that could subject Atlantic or any of its subsidiaries to any liability arising under the Code, ERISA or any other applicable law (including without limitation any liability to or under any such plan or to the PBGC), or under any indemnity agreement to which Atlantic, any of its subsidiaries or any Atlantic ERISA Affiliate is a party, which liability, excluding liability for benefit claims and funding obligations payable in the ordinary course and liability for PBGC insurance premiums payable in the ordinary course is reasonably likely to have an Atlantic Material Adverse Effect.

(e) Welfare Plans. Except as disclosed in Section 5.10(e) of the Atlantic Disclosure Schedule, no Atlantic Benefit Plan that is a "welfare plan" (within the meaning of ERISA § 3(1)) provides benefits for any retired or former employees (other than as required pursuant to ERISA § 601).

(f) Documents Made Available. Atlantic has made available to Delmarva a true and correct copy of each collective bargaining agreement to which Atlantic is a party or under which Atlantic has obligations and, with respect to each Atlantic Benefit Plan, as applicable (i) the current plan document (including all amendments adopted since the most recent restatement) and its most recently prepared summary plan description and all summaries of material modifications prepared since the most recent summary plan description, (ii) the most recently prepared annual report (IRS Form 5500 Series) including financial statements, (iii) each related trust agreement, insurance contract, service provider or investment management agreement (including all amendments to each such document), (iv) the most recent IRS determination letter with respect to the qualified status under Code § 401(a) of such plan and a copy of any application for an IRS determination letter filed since the most recent IRS determination letter was issued and (v) the most recent actuarial report or valuation.

(g) Payments Resulting from Mergers. Other than as disclosed in Section 5.10(g) of the Atlantic Disclosure Schedule, the consummation or announcement of any transaction contemplated by this Agreement will not (either alone or upon the occurrence of any additional or further acts or events) result in any (i) payment (whether of severance pay or otherwise) becoming due from the Company or Atlantic or any of its subsidiaries under any applicable Atlantic Benefit Plans to any officer, employee, former employee or director thereof or to the trustee under any "rabbi trust" or similar arrangement, or (ii) benefit under any Atlantic Benefit Plan being established or becoming accelerated, vested or payable, except for a payment or benefit that would have been payable under the same terms and conditions without regard to the transactions contemplated by this Agreement.

(h) Funded Status of Plans. Except as disclosed in Section 5.10(h) of the Atlantic Disclosure Schedule, each Atlantic Benefit Plan that is subject to either or both of the minimum funding requirements of



ERISA § 302 or to Title IV of ERISA has assets that, as of the date hereof, have a fair market value equal to or exceeding the present value of the accrued benefit obligations thereunder on a termination basis, as of the date hereof based on the actuarial methods, tables and assumptions theretofore utilized by such plan's actuary in preparing such plan's most recently prepared actuarial valuation report, except to the extent that applicable law would require the use of different actuarial assumptions if such plan was to be terminated as of the date hereof. No Atlantic Benefit Plan subject to the minimum funding requirements of ERISA § 302 has incurred any "accumulated funding deficiency" (within the meaning of ERISA § 302).

(i) Multiemployer Plans. Except as disclosed in Section 5.10(i) of the Atlantic Disclosure Schedule, no Atlantic Benefit Plan is or was a "multiemployer plan" (within the meaning of ERISA § 4001(a)(3)), a multiple employer plan described in Code § 413(c), or a "multiple employer welfare arrangement" (within the meaning of ERISA § 3(40)); and none of Atlantic, any subsidiary thereof or any Atlantic ERISA Affiliate has been obligated to contribute to, or otherwise has or has had any liability with respect to, any multiemployer plan, multiple employer plan, or multiple employer welfare arrangement. With respect to any Atlantic Benefit Plan that is listed in Section 5.10(i) of the Atlantic Disclosure Schedule as a multiemployer plan, Atlantic and its subsidiaries have not made or incurred a "complete withdrawal" or a "partial withdrawal," as such terms are defined in ERISA § 4203 and 4205, therefrom at any time during the five calendar year period immediately preceding the date of this Agreement and the transactions contemplated by the Agreement will not, in and of themselves, give rise to such a "complete withdrawal."

(j) Modification or Termination of Plans. Except as disclosed in Section 5.10(j) of the Atlantic Disclosure Schedule: (i) neither Atlantic nor any subsidiary of Atlantic is subject to any legal, contractual, equitable or other obligation to establish as of any date any employee benefit plan of any nature, including without limitation any pension, profit sharing, welfare, post-retirement welfare, stock option, stock or cash award, nonqualified deferred compensation or executive compensation plan, policy or practice; and (ii) to the best knowledge of Atlantic after review of all Atlantic Benefit Plan documents, the Company, Atlantic or one or more of its subsidiaries may, in any manner, and without the consent of any employee, beneficiary or dependent, employees' organization or other person, terminate, modify or amend any Atlantic Benefit Plan or any other employee benefit plan, policy, program or practice (or its participation in any such Atlantic Benefit Plan or other employee benefit plan, policy, program or practice) at any time sponsored, maintained or contributed to by Atlantic or any of its subsidiaries, effective as of any date before, on or after the Effective Time except to the extent that any retroactive amendment would be prohibited by ERISA § 204(g) or would deprive a plan participant of a benefit in which such participant has a vested right.

(k) *Reportable Events; Claims.* Except as disclosed in Section 5.10(k) of the Atlantic Disclosure Schedule, (i) no event constituting a "reportable event" (within the meaning of ERISA § 4043(b)) for which the 30-day notice requirement has not been waived by the PBGC has occurred with respect to any Atlantic Benefit Plan and (ii) no liability, claim, action or litigation has been made, commenced or, to the best knowledge of Atlantic, threatened, by or against Atlantic or any of its subsidiaries with respect to any Atlantic Benefit Plan (other than for benefits or PBGC premiums payable in the ordinary course) that is reasonably likely to have an Atlantic Material Adverse Effect.

(1) Labor Agreements. Except as disclosed in Section 5.10(1) of the Atlantic Disclosure Schedule, to the best knowledge of Atlantic, as of the date hereof, there is no current labor union representation question involving employees of Atlantic or any of its subsidiaries, nor does Atlantic or any of its subsidiaries know of any activity or proceeding of any labor organization (or representative thereof) or employee group (or representative thereof) to organize any such employees. Except as disclosed in the Atlantic SEC Reports or as disclosed in Section 5.10(1) of the Atlantic Disclosure Schedule: (i) neither Atlantic nor any of its subsidiaries is a party to any collective bargaining agreement or other labor agreement with any union or labor organization; (ii) there are no unfair labor practice charges or grievances arising out of a collective bargaining agreement or other grievance procedure against Atlantic

or any of its subsidiaries pending, or to the best knowledge of Atlantic threatened, that, in the aggregate, are reasonably likely to have an Atlantic Material Adverse Effect; (iii) there are no complaints, lawsuits or proceedings in any forums by or on behalf of any present or former employees, any applicants for employment or classes of the foregoing alleging breach of any express or implied contract of employment, any law or regulation governing employment or the termination thereof or other discriminatory, wrongful or tortious conduct in connection with the employment relationship against Atlantic or any of its subsidiaries pending, or to the best knowledge of Atlantic threatened, that, in the aggregate, are reasonably likely to have an Atlantic Material Adverse Effect; (iv) there are no strikes, disputes, slowdowns, work stoppages or lockouts pending, or to the best knowledge of Atlantic threatened, against or involving Atlantic or any of its subsidiaries that, in the aggregate, are reasonably likely to have an Atlantic Material Adverse Effect; (v) Atlantic and each of its subsidiaries are in compliance with all applicable laws respecting employment and employment practices, terms and conditions of employment, wages, hours of work and occupational safety and health, except for noncompliance that, in the aggregate, is not reasonably likely to have an Atlantic Material Adverse Effect; and (vi) there is no proceeding, claim, suit, action or governmental investigation pending, or to the best knowledge of Atlantic threatened, in respect to which any director, officer, employee or agent of Atlantic or any of its subsidiaries is or may be entitled to claim indemnification from Atlantic or any of its subsidiaries pursuant to their respective certificates or articles of incorporation or bylaws or as provided in the indemnification agreements listed on Section 5.10(1) of the Atlantic Disclosure Schedule.

## Section 5.11 Environmental Protection.

(a) Compliance. Except as disclosed in Section 5.11(a) of the Atlantic Disclosure Schedule or as disclosed in the Atlantic SEC Reports, Atlantic and each of its subsidiaries are and have been in material compliance with all applicable Environmental Laws, except where the failure to be or to have been so in material compliance, in the aggregate, is not reasonably likely to have an Atlantic Material Adverse Effect. Except as disclosed in Section 5.11(a) of the Atlantic Disclosure Schedule, neither Atlantic nor any of its subsidiaries has received any written notice from any person or Governmental Authority that alleges that Atlantic or any of its subsidiaries is not or has not been in material compliance with applicable Environmental Laws, except where the failure to be or to have been so in material compliance, in the aggregate, is not reasonably likely to have an Atlantic Material Compliance, in the aggregate, is not reasonably likely to have been so in material compliance, in the aggregate, is not reasonably likely to have Effect.

(b) Environmental Permits. Except as disclosed in Section 5.11(b) of the Atlantic Disclosure Schedule or as disclosed in the Atlantic SEC Reports, Atlantic and each of its subsidiaries have obtained or have applied for all material Environmental Permits necessary for the construction of their facilities and the conduct of their operations, and all such Environmental Permits are in good standing or, where applicable, a renewal application has been timely filed and is pending agency approval, and Atlantic and its subsidiaries are in compliance with all terms and conditions of all such Environmental Permits and are not required to make any material expenditures in connection with any renewal application pending agency approval, except where the failure to obtain or be in such compliance and the requirement to make such expenditures, in the aggregate, is not reasonably likely to have an Atlantic Material Adverse Effect.

(c) Environmental Claims. Except as disclosed in Section 5.11(c) of the Atlantic Disclosure Schedule or as disclosed in the Atlantic SEC Reports, to the best knowledge of Atlantic, there are no Environmental Claims (as defined in Section 4.11(g)) pending, or to the best knowledge of Atlantic threatened, (i) against Atlantic or any of its subsidiaries or joint ventures, (ii) against any person or entity whose liability for any Environmental Claim Atlantic or any of its subsidiaries or joint ventures has or may have retained or assumed either contractually or by operation of law or (iii) against any real or personal property or operations that Atlantic or any of its subsidiaries or joint ventures owns, leases or manages, in whole or in part, that, if adversely determined are in the aggregate reasonably likely to have an Atlantic Material Adverse Effect. (d) *Releases.* Except as disclosed in Section 5.11(c) or 5.11(d) of the Atlantic Disclosure Schedule or as disclosed in the Atlantic SEC Reports, to the best knowledge of Atlantic, there has been no Release of any Hazardous Material that would be reasonably likely to form the basis of any Environmental Claim against Atlantic or any subsidiary or joint venture of Atlantic, or against any person or entity whose liability for any Environmental Claim Atlantic or any subsidiary or joint venture of Atlantic, we can be available of the transformed either contractually or by operation of law, except for Releases of Hazardous Materials the liability for which is not in the aggregate reasonably likely to have an Atlantic Material Adverse Effect.

(e) *Predecessors.* Except as disclosed in Section 5.11(e) of the Atlantic Disclosure Schedule or as disclosed in the Atlantic SEC Reports, to the best knowledge of Atlantic with respect to any predecessor of Atlantic or of any subsidiary or joint venture of Atlantic, there are no Environmental Claims pending or threatened, or any Releases of Hazardous Materials that would be reasonably likely to form the basis of any Environmental Claims, that are reasonably likely to have, in the aggregate, an Atlantic Material Adverse Effect.

(f) Disclosure. To the best knowledge of Atlantic, Atlantic has disclosed to Delmarva all material facts that Atlantic reasonably believes are likely to form the basis of a material Environmental Claim or to require material expenditures in order to comply with current or future applicable Environmental Laws arising from (i) the cost of pollution control equipment currently required or known to be required in the future, (ii) current investigatory, removal, remediation or response costs or investigatory, removal, remediation or response costs whom to be required in the future, in each case, both on-site and off-site and (iii) any other environmental matters affecting Atlantic or its subsidiaries.

Section 5.12 Regulation as a Utility. One of Atlantic's wholly owned subsidiaries is regulated as a public utility in the State of New Jersey and by the FERC. Except as disclosed in Section 5.12 of the Atlantic Disclosure Schedule, neither Atlantic nor any "subsidiary company" or "affiliate" (as such terms are defined in the 1935 Act) of Atlantic is subject to regulation as a public utility or public service company (or similar designation) by any other state in the United States, by the United States or any agency or instrumentality of the United States or by any foreign country. Atlantic is a holding company exempt from all provisions of the 1935 Act.

Section 5.13 *Vote Required*. The approval of the Atlantic Merger by a majority of all votes cast by the holders of Atlantic Common Stock (the "Atlantic Shareholders' Approval"), is the only vote of the holders of any class or series of the capital stock of Atlantic required to approve this Agreement, the Mergers and the other transactions contemplated hereby.

Section 5.14 *Opinion of Financial Advisor*. Atlantic has received the opinion of Morgan Stanley & Co. Incorporated, dated the date hereof, to the effect that, as of the date hereof, the Atlantic Conversion Ratio, taking into account the Delmarva Conversion Ratio, is fair from a financial point of view to the holders of Atlantic Common Stock.

Section 5.15 *Insurance*. Except as disclosed in Section 5.15 of the Atlantic Disclosure Schedule, each of Atlantic and each of its subsidiaries is, and has been continuously since January 1, 1991, insured with financially responsible insurers in such amounts and against such risks and losses as are customary for companies conducting the respective businesses conducted by Atlantic and its subsidiaries during such time period. Except as disclosed in Section 5.15 of the Atlantic Disclosure Schedule, neither Atlantic nor any of its subsidiaries has received any notice of cancellation or termination with respect to any material insurance policy thereof. All material insurance policies of Atlantic and its subsidiaries are valid and enforceable policies.

Section 5.16 Applicability of Certain New Jersey Law. Assuming the accuracy of the representation of Delmarva set forth in Section 4.17, the New Jersey Shareholders Protection Act (Section 14A:10A-1 et seq. of the NJBCA) is not applicable to the transactions contemplated by this Agreement.

Section 5.17 Ownership of Delmarva Common Stock. Atlantic does not "beneficially own" (as such term is defined in Rule 13d-3 under the Exchange Act) any shares of Delmarva Common Stock.

Section 5.18 Operations of Atlantic Nuclear Power Plants. Except as set forth in Section 5.18 of the Atlantic Disclosure Schedule, to the best knowledge of Atlantic, the operations of the Hope Creek Nuclear Generating Station owned by Atlantic and the Salem and Peach Bottom Nuclear Generating Stations owned by Atlantic together with Delmarva (collectively, the "Atlantic Facilities") are and have at all times been conducted in compliance with applicable health, safety, regulatory and other legal requirements, except where the failure to be in compliance in the aggregate does not and insofar as can reasonably be foreseen would not have an Atlantic Material Adverse Effect. To the best knowledge of Atlantic, each of the Atlantic Facilities maintains emergency plans designed to respond to an unplanned release therefrom of radioactive materials into the environment and customary liability insurance consistent with industry practice and consistent with Atlantic's view of the risks inherent in the operation of a nuclear power facility currently exists with respect to such Atlantic Facilities and for the short-term storage of spent nuclear fuel conform with the requirements of applicable law, and such plans have at all times been funded consistently with reasonable budget projections for such plans.

#### **ARTICLE VI**

### CONDUCT OF BUSINESS PENDING THE MERGERS

Delmarva and Atlantic have each delivered to the other a budget for the years 1996 through 1998 (respectively, the "Delmarva Budget" and the "Atlantic Budget"), which Delmarva or Atlantic, as the case may be, may update or otherwise modify in writing for purposes of this Article VI only with the consent in writing of Atlantic or Delmarva, as the case may be. After the date hereof and prior to the Effective Time or earlier termination of this Agreement, each of Delmarva and Atlantic agrees as to itself and its subsidiaries, except as expressly contemplated or permitted in this Agreement, or to the extent Atlantic or Delmarva, as the case may be, shall otherwise consent in writing, as follows:

Section 6.1 Ordinary Course of Business. Each of Delmarva and Atlantic shall, and each shall cause its respective subsidiaries to, carry on its and their respective businesses in the usual, regular and ordinary course consistent with past practice and use all commercially reasonable efforts to preserve intact their present business organizations and goodwill, preserve the goodwill and relationships with customers, suppliers and others having business dealings with them and, subject to prudent management of their workforces, including (without limitation) ongoing or planned activities, benefits, programs, practices, and policies related to effecting their business strategies and downsizing, re-engineering and similar matters, keep available the services of their present officers and employees, to the end that their goodwill and ongoing businesses shall not be impaired in any material respect at the Effective Time. Neither Delmarva nor Atlantic shall, nor shall either permit any of its subsidiaries to, (i) enter into a new line of business that is not described, in the case of Delmarva, in Section 6.1 of the Delmarva Disclosure Schedule, or, in the case of Atlantic, in Section 6.1 of the Atlantic Disclosure Schedule (in each case, a "Listed Activity") involving any material investment of assets or resources or any material exposure to liability or loss (including without limitation any loans or capital contributions to, and the undertaking of any guarantees in favor of or any "keepwell" or other agreements to maintain the financial condition of, another person), in the case of Delmarva, to Delmarva and its subsidiaries taken as a whole, and, in the case of Atlantic, to Atlantic and its subsidiaries taken as a whole, (ii) make aggregate investments (including without limitation any loans or capital contributions to, and the undertaking of any guarantees in favor of or any "keep well" or other agreements to maintain the financial condition of, another person) during the period from the date of this Agreement to the Effective Time in or in respect of Listed Activities in excess of \$250,000,000 in the aggregate, or (iii) make any such investment in or in respect of any Listed Activity, other than an investment by Atlantic in thermal (heating and cooling) services, in excess of \$7,500,000 per year in any Listed Activity without consultations with each other concerning the investment. For purposes of this Article VI, "consistent with past practice" as applied to Atlantic and its subsidiaries shall mean among other things that, unless otherwise required by applicable law or regulation, the allocation of benefits and burdens to Atlantic City Electric Company ("Atlantic Utility") in comparison with those allocated to Atlantic and its subsidiaries other than Atlantic Utility shall be on a basis not more favorable to Atlantic Utility than has previously been the case.

Section 6.2 Dividends. Neither Delmarva nor Atlantic shall, nor shall either permit any of its subsidiaries to: (a) declare or pay any dividends on or make other distributions in respect of any of their capital stock other than (i) to such party or its wholly-owned subsidiaries, (ii) dividends required to be paid on any Delmarva Preferred Stock or any preferred stock issued by Atlantic City Electric Company ("Atlantic Subsidiary Preferred Stock") in accordance with their respective terms, (iii) regular quarterly dividends on Delmarva Common Stock with usual record and payment dates at an annual rate not in excess of \$1.54 per share and (iv) regular quarterly dividends on Atlantic Common Stock with usual record and payment dates at an annual rate not in excess of \$1.54 per share; (b) split, combine or reclassify any of their capital stock or issue or authorize or propose the issuance of any other securities in respect of, in lieu of, or in substitution for, shares of its capital stock; or (c) redeem, repurchase or otherwise acquire any shares of their capital stock other than (i) redemptions, repurchases and other acquisitions of shares of capital stock in the ordinary course of business consistent with past practice including without limitation, (w) repurchases, redemptions and other acquisitions in connection with the administration of employee benefit and dividend reinvestment plans as in effect on the date hereof in the ordinary course of the operation of such plans, (x) redemptions, purchases or acquisitions required by the respective terms of any series of Delmarva Preferred Stock or Atlantic Subsidiary Preferred Stock, (y) in connection with the refunding of Delmarva Preferred Stock or Atlantic Subsidiary Preferred Stock through the issuance of additional Delmarva Preferred Stock or Atlantic Utility Preferred Stock or indebtedness, as the case may be, at a lower cost of funds (calculating such cost on an aggregate after-tax basis) or through the issuance of long-term indebtedness as permitted under Section 6.7 and (z) open-market repurchases of Delmarva Common Stock or Atlantic Common Stock, as the case may be, utilized to fund up to \$50,000,000 in any fiscal year (when aggregated with common stock so utilized by Delmarva or Atlantic, as the case may be, pursuant to clause (c) of Section 6.3) of the cost of any acquisitions permitted under Section 6.5 and (ii) intercompany acquisitions of capital stock.

Section 6.3 *Issuance of Securities*. Neither Delmarva nor Atlantic shall, nor shall either permit any of its subsidiaries to, issue, deliver or sell, or authorize or propose the issuance, delivery or sale of, any shares of their capital stock of any class or any securities convertible into or exchangeable for, or any rights, warrants or options to acquire, any such shares or convertible or exchangeable securities, except as provided in the Delmarva Budget or the Atlantic Budget, as the case may be, and except for (a) the issuance of common stock or stock appreciation or similar rights, as the case may be, pursuant to (i) the Dividend Reinvestment and Common Share Purchase Plan, the Savings and Thrift Plan, and the Long-Term Incentive Plan of Delmarva or (ii) the Employee Incentive Plan, the Directors' Restricted Stock Plan, the Employee Stock Purchase Plan and the Dividend Reinvestment Plan of Atlantic, in each case consistent with past practice in kind and amount and in the ordinary course of business under such plans substantially in accordance with their present terms, (b) the issuance by a wholly owned subsidiary of shares of its capital stock to its parent and (c) the issuance of common stock by Delmarva or Atlantic, as the case may be, utilized to fund up to \$50,000,000 in any fiscal year (when aggregated with common stock so utilized by Delmarva or Atlantic, as the case may be, pursuant to clause (c)(i)(z) of Section 6.2) of the cost of any acquisitions permitted under Section 6.5.

Section 6.4 *Charter Documents*. Neither Delmarva nor Atlantic shall amend or propose to amend its certificate or articles of incorporation or its bylaws, except as necessary to provide for the issuance of securities as permitted pursuant to Section 6.3 or as required by law.

Section 6.5 Acquisitions. Except as permitted by Section 6.1 in respect of Listed Activities or as may be disclosed in Section 6.5 of the Delmarva Disclosure Schedule or the Atlantic Disclosure Schedule, and except for acquisitions not exceeding \$20,000,000 in the aggregate during any fiscal year, or more than



\$40,000,000 in the aggregate during the period from the date of this Agreement to the Effective Time, in the case of, on the one hand, Delmarva and its subsidiaries and, on the other hand, Atlantic and its subsidiaries, neither Delmarva nor Atlantic shall, nor shall either permit any of its subsidiaries to, acquire or agree or publicly propose to acquire, by merging or consolidating with, or by purchasing a substantial equity interest in or a substantial portion of the assets of, or by any other manner, any business or any corporation, partnership, association or other business organization or division thereof, or otherwise acquire or agree to acquire any material amount of assets other than in the ordinary course of business consistent with past practice.

Section 6.6 *No Dispositions*. Except as disclosed in Section 6.6 of the Delmarva Disclosure Schedule or the Atlantic Disclosure Schedule, and other than (a) dispositions not exceeding \$5,000,000 during any fiscal year in the case of, on the one hand, Delmarva and its subsidiaries and, on the other hand, Atlantic and its subsidiaries, (b) as may be required by law to consummate the transactions contemplated hereby or (c) in the ordinary course of business consistent with past practice, neither Delmarva nor Atlantic shall, nor shall either permit any of its subsidiaries to, sell, lease, license, encumber or otherwise dispose of, any of its assets that are material, individually or in the aggregate, to such party and its subsidiaries taken as a whole.

Section 6.7 Indebtedness. Except as disclosed in Section 6.7 of the Delmarva Disclosure Schedule or the Atlantic Disclosure Schedule and except as provided in the Delmarva Budget and the Atlantic Budget, as the case may be, neither Delmarva nor Atlantic shall, nor shall either permit any of its subsidiaries to, incur or guarantee any indebtedness (including any debt borrowed or guaranteed or otherwise assumed, including without limitation the issuance of debt securities or warrants or rights to acquire debt) or enter into any "keepwell" or other agreement to maintain the financial condition of another person or enter into arrangements having the effect of any of the foregoing other than (a) short-term indebtedness in the ordinary course of business consistent with past practice, (b) long-term indebtedness in connection with the refinancing of existing indebtedness either at its stated maturity or at a lower cost of funds (calculating such cost on an aggregate after-tax basis), (c) long-term indebtedness in connection with the refunding of Delmarva Preferred Stock or Atlantic Preferred Stock at a lower cost of funds (calculated as aforesaid), and (d) additional indebtedness in any fiscal year not exceeding \$25,000,000 more than the amount provided therefor in the Delmarva Budget for such fiscal year with respect to Delmarva and its subsidiaries and in the Atlantic Budget for such fiscal year with respect to Atlantic and its subsidiaries.

Section 6.8 *Capital Expenditures*. Except as permitted by Section 6.1 in respect of Listed Activities or as may be disclosed in Section 6.8 of the Delmarva Disclosure Schedule or the Atlantic Disclosure Schedule or as required by law, neither Delmarva nor Atlantic shall, nor shall either permit any of its subsidiaries to, make any capital expenditures, other than (a) capital expenditures incurred in connection with the construction of new facilities, (b) capital expenditures to repair or replace facilities destroyed or damaged due to casualty or accident (whether or not covered by insurance) and (c) additional capital expenditures in any year of not more than 10% of the amount provided therefor in the Delmarva Budget for that year with respect to Delmarva and its subsidiaries and in the Atlantic Budget for that year with respect to Atlantic and its subsidiaries.

Section 6.9 Compensation, Benefits. Except as disclosed in Section 6.9 of the Delmarva Disclosure Schedule or the Atlantic Disclosure Schedule, neither Delmarva nor Atlantic shall, nor shall either permit any of its subsidiaries to, (i) enter into, adopt or amend (except as may be required by applicable law), or increase the amount or accelerate the payment or vesting of any benefit or amount payable under, any employee benefit plan or other contract, agreement, commitment, arrangement, plan or policy maintained by, contributed to or entered into by such party or any of its subsidiaries, or increase, or enter into any contract, agreement, commitment to increase in any manner, the compensation or fringe benefits, or otherwise to extend, expand or enhance the engagement, employment or any related rights, of any director, officer or other employee of such party or any of its subsidiaries, except pursuant to binding legal commitments and except for normal (including incentive) increases, extensions, expansions, enhancements, amendments or adoptions in the ordinary course of business consistent with past practice that, in the aggregate, do not result in a material increase in benefits or compensation expense to such party and its subsidiaries taken as a whole or (ii) enter into or amend any employment, severance, special pay arrangement with respect to termination of employment or other similar contract, agreement or arrangement with any director or officer other than in the ordinary course of business consistent with past practice. Notwithstanding the foregoing, (i) Atlantic shall terminate its retirement plan for non-employee directors and pay in cash all accrued benefits thereunder as described in Section 6.9 of the Atlantic Disclosure Schedule prior to the Effective Time, and (ii) each of Delmarva and Atlantic may, with the agreement of the other, modify the benefits payable under its severance arrangements with management employees in a manner which makes the aggregate level of severance benefits received by such management employees comparable to the aggregate level of severance benefits which are, on the date of this Agreement, provided to comparable employees of the other.

Section 6.10 1935 Act. Atlantic shall not, nor shall it permit any of its subsidiaries to, except as required or contemplated by this Agreement, engage in any activities that would cause a change in its status, or that of its subsidiaries, under the 1935 Act, that would impair the ability of Atlantic to claim an exemption from all provisions of the 1935 Act except Section 9(a)(2) under Section 3(a)(1) of the 1935 Act or that would impair the ability of Atlantic Utility to claim an exemption from all provisions of the 1935 Act except Section 9(a)(2) under Section 9(a)(2) under Section 3(a)(2) of the 1935 Act, other than (i) the application to the SEC under the 1935 Act contemplated by this Agreement for approval to the extent required of the transactions contemplated hereby and (ii) the registration of the Company pursuant to the 1935 Act.

Section 6.11 Accounting. Neither Delmarva nor Atlantic shall, nor shall either permit any of its subsidiaries to, make any changes in their accounting methods, except as required by law, rule, regulation or GAAP.

Section 6.12 *Tax-Free Status*. No party shall, nor shall any party permit any of its subsidiaries to, take any actions that would, or would be reasonably likely to, adversely affect the status of the Mergers as a tax-free transaction (except as to dissenters' rights and fractional shares) under Code § 368(a) and/or Code § 351, and each party shall use all commercially reasonable efforts to achieve such result.

Section 6.13 Discharge of Liabilities. Neither Delmarva nor Atlantic shall, nor shall either permit any of its subsidiaries to, pay, discharge or satisfy any material claims, liabilities or obligations (absolute, accrued, asserted or unasserted, contingent or otherwise), other than (i) the payment, discharge or satisfaction, in the ordinary course of business consistent with past practice (which includes the payment of final and unappealable judgments and the refinancing of existing indebtedness for borrowed money either at its stated maturity or at a lower cost of funds) or in accordance with their terms, of liabilities reflected or reserved against in, or contemplated by, the most recent consolidated financial statements (or the notes thereto) of such party included in such party's reports filed with the SEC, or incurred in the ordinary course of business consistent with past practice or, or pursuant to Section 6.7 or as disclosed in Section 6.7 of the Delmarva Disclosure Schedule or the Atlantic Disclosure Schedule, or (ii) as part of or pursuant to any settlement of any rate filings before the public utility commission of any state or the FERC pending on the date of this Agreement.

Section 6.14 *Cooperation, Notification.* Each of Delmarva and Atlantic shall: (a) confer on a regular and frequent basis with one or more representatives of the other to discuss, subject to applicable law, material operational matters and the general status of its ongoing operations; (b) promptly notify the other of any significant changes in its business (including the entry into material new lines of business or material extensions or expansions of existing lines of business), properties, assets, condition (financial or other), prospects or results of operations; (c) advise the other of any change or event that has had or, insofar as reasonably can be foreseen, is reasonably likely to result in, a Delmarva Material Adverse Effect or an Atlantic Material Adverse Effect, as the case may be; and (d) promptly provide the other with copies of all filings made by it or any of its subsidiaries with any state or federal court, administrative agency, commission or other Governmental Authority in connection with this Agreement and the transactions contemplated hereby.



Section 6.15 *Rate Matters*. Other than currently pending rate filings and matters related to the recovery of costs associated with outages at the Salem Nuclear Generating Station, each of Delmarva and Atlantic shall, and shall cause its subsidiaries to, discuss with the other any changes in its or its subsidiaries' regulated rates or charges, standards of service or accounting from those in effect on the date hereof and consult with the other parties prior to making any filing (or any amendment thereto), or effecting any agreement, commitment, arrangement or consent with governmental regulators, whether written or oral, formal or informal, with respect thereto, and neither shall make any filing to change its rates or charges on file with the public utility commission of any state or FERC that would have a material adverse effect on the benefits associated with the Mergers.

Section 6.16 *Third-Party Consents*. Delmarva shall, and shall cause its subsidiaries to, use all commercially reasonable efforts to obtain all Delmarva Required Consents. Delmarva shall promptly notify Atlantic of any failure or anticipated failure to obtain any such consents and, if requested by Atlantic, shall provide copies of all Delmarva Required Consents obtained by Delmarva to Atlantic. Atlantic shall, and shall cause its subsidiaries to, use all commercially reasonable efforts to obtain all Atlantic Required Consents. Atlantic shall promptly notify Delmarva of any failure or anticipated failure to obtain any such consents and, if requested by Delmarva, shall provide copies of all Atlantic Required Consents obtained by Atlantic to Delmarva.

Section 6.17 No Breach, Etc. No party shall, nor shall any party permit any of its subsidiaries to, take any action that would or is reasonably likely to result in a material breach of any covenant or agreement of this Agreement or in any of its representations and warranties set forth in this Agreement being untrue on and as of the Closing Date.

Section 6.18 *Tax-Exempt Status*. No party shall, nor shall any party permit any subsidiary to, take any action that would, or would be reasonably likely to, jeopardize the qualification of the outstanding revenue bonds issued for the benefit of Delmarva (or any subsidiary thereof) or for the benefit of Atlantic (or any subsidiary thereof) that qualify on the date hereof under Code § 142(a) as "exempt facility bonds" or as tax-exempt industrial development bonds under Section 103(b)(4) of the Internal Revenue Code of 1954, as amended prior to the Tax Reform Act of 1986.

Section 6.19 *Transition Management*. Delmarva and Atlantic shall create a special transition management task force (the "Task Force") to be headed by Howard E. Cosgrove (or an individual designated by him who shall be reasonably satisfactory to the other Task Force head) and Michael J. Chesser (or an individual designated by him and reasonably satisfactory to the other Task Force head). The Task Force shall examine various alternatives regarding the manner in which best to organize and manage the business of the Company after the Effective Time, subject to applicable law. From time to time, the Task Force shall report its findings to the Board of Directors of each of Delmarva and Atlantic.

Section 6.20 *Insurance*. Each of Delmarva and Atlantic shall, and shall cause its subsidiaries to, maintain with financially responsible insurance companies insurance in such amounts and against such risks and losses as are customary for companies engaged in the utility industry and employing methods of generating electric power and fuel sources similar to those methods employed and fuels used by such party or such party's subsidiaries.

Section 6.21 *Permits*. Delmarva and Atlantic shall, and shall each cause its subsidiaries to, use all reasonable efforts to maintain in effect all existing Permits (as defined in Section 4.4) pursuant to which it or its subsidiaries operate.

Section 6.22 *Contracts; Agreements*. Neither Delmarva nor Atlantic shall, nor shall either permit any of its subsidiaries to, except in the ordinary course of business consistent with past practice, modify, amend, terminate, renew or fail to use reasonable business efforts to renew any material contract or agreement to which such party or any subsidiary of such party is a party or waive, release or assign any material rights or claims. Neither Delmarva nor Atlantic shall, nor shall either permit any of its subsidiaries to, agree in writing to take any action not permitted by this Article VI.

### **ARTICLE VII**

### ADDITIONAL AGREEMENTS

Section 7.1 Access to Information. Upon reasonable notice and during normal business hours, each party shall, and shall cause its subsidiaries to, afford to the officers, directors, employees, accountants, counsel, investment banker, financial advisor and other representatives of the other (collectively, "Representatives") reasonable access, during normal business hours throughout the period prior to the Effective Time, to all of its properties, books, contracts, commitments and records (including, but not limited to, Tax Returns) and, during such period, each party shall, and shall cause its subsidiaries to, furnish promptly to the other (i) access to each report, schedule and other document filed or received by it or any of its subsidiaries pursuant to the requirements of federal or state securities laws or filed with the SEC, the FERC, the NRC, the Department of Justice, the Federal Trade Commission, the Delaware Commission, the Maryland Commission, the Virginia Commission, the New Jersey Commission, the Pennsylvania Commission or any other federal or state regulatory agency or commission, and (ii) access to all information concerning themselves, their subsidiaries, directors, officers and shareholders and such matters as may be reasonably requested by the other party in connection with any filings, applications or approvals required or contemplated by this Agreement. All documents and information furnished pursuant to this Section 7.1 shall be subject to the confidentiality agreement dated April 10, 1996 between Delmarva and Atlantic, as it may be amended from time to time (the "Confidentiality Agreement"). The party requesting copies of any documents from any other party hereto shall be responsible for all out-of-pocket expenses incurred by the party to whom such request is made in complying with such request, including any cost of reproducing and delivering any required information.

## Section 7.2 Joint Proxy Statement and Registration Statement.

(a) Preparation and Filing. As promptly as reasonably practicable after the date hereof, the parties shall prepare and file with the SEC the Registration Statement and the Joint Proxy Statement (together, the "Joint Proxy/Registration Statement"). The parties shall take such actions as may be reasonably required to cause the Registration Statement to be declared effective under the Securities Act as promptly as practicable after such filing. The parties shall also take such action as may be reasonably required to cause the shares of Company Common Stock and Letter Stock issuable in connection with the Mergers to be registered or to obtain an exemption from registration under applicable state "blue sky" or securities laws; provided, however, that none of the Company, Atlantic or Delmarva shall be required to register or qualify as a foreign corporation or to take any other action that would subject it to general service of process in any jurisdiction in which the Company will not, following the Mergers, be so subject. Each of the parties shall furnish all information concerning itself that is required or customary for inclusion in the Joint Proxy/Registration Statement. No representation, covenant or agreement contained in this Agreement is made by any party with respect to information supplied by any other party for inclusion in the Joint Proxy/ Registration Statement. The Joint Proxy/Registration Statement shall comply as to form in all material respects with the Securities Act and the rules and regulations thereunder. The parties shall take such action as may be reasonably required to cause the shares of Company Common Stock and Letter Stock to be issued in the Mergers to be approved for listing on the NYSE and any other stock exchanges agreed to by the parties, each upon official notice of issuance.

(b) Letter of Delmarva's Accountants. Following receipt by Coopers & Lybrand, Delmarva's independent auditors, of an appropriate request from Atlantic pursuant to SAS No. 72, Delmarva shall use best efforts to cause to be delivered to the Company and Atlantic a letter of Coopers & Lybrand, dated a date within two business days before the effective date of the Registration Statement, and addressed to the Company and Atlantic, in form and substance reasonably satisfactory to the Company and Atlantic and customary in scope and substance for "cold comfort" letters delivered by independent public accountants in connection with registration statements and proxy statements similar to the Joint Proxy/Registration Statement. (c) Letter of Atlantic's Accountants. Following receipt by Deloitte & Touche, LLP, Atlantic's independent auditors, of an appropriate request from Delmarva pursuant to SAS No. 72, Atlantic shall use best efforts to cause to be delivered to the Company and Delmarva a letter of Deloitte & Touche, LLP, dated a date within two business days before the effective date of the Registration Statement, and addressed to the Company and Delmarva, in form and substance satisfactory to the Company and Delmarva and customary in scope and substance for "cold comfort" letters delivered by independent public accountants in connection with registration statements and proxy statements similar to the Joint Proxy/Registration Statement.

(d) Fairness Opinions. It shall be a condition to the mailing of the Joint Proxy Statement to the shareholders of Atlantic and Delmarva that (i) Delmarva shall have received an opinion from Merrill Lynch, Pierce, Fenner & Smith Incorporated, dated the date of the Joint Proxy Statement, to the effect that, as of the date thereof, the Delmarva Conversion Ratio is fair to the holders of Delmarva Common Stock, and (ii) Atlantic shall have received an opinion from Morgan Stanley & Co. Incorporated, dated the date of the Joint Proxy Statement, to the effect that; as of the date thereof, the Atlantic Conversion Ratio is fair to the holders of Atlantic Conversion Ratio is fair to the holders of Atlantic Conversion Ratio is fair to the holders of Atlantic Conversion Ratio is fair to the holders of Atlantic Conversion Ratio is fair to the holders of Atlantic Common Stock.

## Section 7.3 Regulatory Matters.

(a) HSR Filings. Each party shall file or cause to be filed with the Federal Trade Commission and the Department of Justice any notifications required to be filed by their respective "ultimate parent" companies under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "HSR Act"), and the rules and regulations promulgated thereunder with respect to the transactions contemplated hereby. The parties shall use all commercially reasonable efforts to make such filings promptly and shall respond promptly to any requests for additional information made by either of such agencies.

(b) Other Regulatory Approvals. Each party shall cooperate and use its best efforts to promptly prepare and file all necessary documentation, to effect all necessary applications, notices, petitions, filings and other documents, and to use all commercially reasonable efforts to obtain all necessary permits, consents, approvals and authorizations of all Governmental Authorities and all other persons necessary or advisable to consummate the transactions contemplated by this Agreement, including without limitation the Delmarva Required Statutory Approvals and the Atlantic Required Statutory Approvals. Atlantic shall have the right to review and approve in advance all characterizations of the information relating to Atlantic, on the one hand, and Delmarva shall have the right to review and approve in advance all characterizations of the information relating to Delmarva, on the other hand, in either case which appear in any filing made in connection with the transactions contemplated by this Agreement. Delmarva and Atlantic shall each consult with the other with respect to the obtaining of all such necessary or advisable permits, consents, approvals and authorizations of Governmental Authorities.

# Section 7.4 Shareholder Approvals.

(a) Approval of Atlantic Shareholders. Atlantic shall, as promptly as reasonably practicable after the date hereof (i) take all steps reasonably necessary to call, give notice of, convene and hold a special meeting of its shareholders (the "Atlantic Special Meeting") for the purpose of securing the Atlantic Shareholders' Approval, (ii) distribute to its shareholders the Joint Proxy Statement in accordance with applicable federal and state law and its articles of incorporation and bylaws, (iii) subject to the fiduciary duties of its Board of Directors, recommend to its shareholders the approval of the Atlantic Merger, this Agreement and the transactions contemplated hereby and (iv) cooperate and consult with Delmarva with respect to each of the foregoing matters.

(b) Approval of Delmarva Shareholders. Delmarva shall, as promptly as reasonably practicable after the date hereof (i) take all steps reasonably necessary to call, give notice of, convene and hold a special meeting of its shareholders (the "Delmarva Special Meeting") for the purpose of securing the Delmarva Shareholders' Approval, (ii) distribute to its shareholders the Joint Proxy Statement in accordance with applicable federal and state law and its articles of incorporation and bylaws, (iii) subject to the fiduciary duties of its Board of Directors, recommend to its shareholders the approval of the Delmarva Merger, this Agreement and the transactions contemplated hereby and (iv) cooperate and consult with Atlantic with respect to each of the foregoing matters.

(c) *Meeting Date*. The Delmarva Special Meeting and the Atlantic Special Meeting shall be held on the same day unless otherwise agreed by Delmarva and Atlantic.

(d) *Fairness Opinions Not Withdrawn*. It shall be a condition to the obligation of Delmarva to hold the Delmarva Special Meeting that the opinion of Merrill Lynch, Pierce, Fenner & Smith Incorporated referred to in Section 7.2(d) shall not have been withdrawn, and it shall be a condition to the obligation of Atlantic to hold the Atlantic Special Meeting that the opinion of Morgan Stanley & Co. Incorporated referred to in Section 7.2(d) shall not have been withdrawn.

ς,

# Section 7.5 Directors' and Officers' Indemnification.

(a) Indemnification. To the extent, if any, not provided by an existing right of indemnification or other agreement or policy, from and after the Effective Time, the Company shall, to the fullest extent not prohibited by applicable law, indemnify, defend and hold harmless the present and former directors, officers and management employees of the parties hereto and their respective subsidiaries (each, an "Indemnified Party" and, collectively, the "Indemnified Parties") against (i) all losses, expenses (including reasonable attorneys' fees and expenses), claims, damages, costs, liabilities, judgments or (subject to the proviso of the next succeeding sentence) amounts that are paid in settlement of or in connection with any claim, action, suit, proceeding or investigation based in whole or in part on or arising in whole or in part out of the fact that such person is or was a director, officer or management employee of such party or any subsidiary thereof, whether pertaining to any matter existing or occurring at or prior to or after the Effective Time and whether asserted or claimed prior to, at or after the Effective Time and (ii) all liabilities based in whole or in part on, or arising in whole or in part out of, or pertaining to this Agreement or the transactions contemplated hereby. In the event of any such loss, expense, claim, damage, cost, liability, judgment or settlement (whether or not arising before the Effective Time), (x) the Company shall pay the reasonable fees and expenses of counsel selected by the Indemnified Parties, which counsel shall be reasonably satisfactory to the Company, promptly after statements therefor are received, and otherwise advance to the Indemnified Parties upon request reimbursement of documented expenses reasonably incurred, in either case to the extent not prohibited by applicable law, (y) the Company shall cooperate in the defense of any such matter and (z) any determination required to be made with respect to whether an Indemnified Party's conduct complies with the standards under applicable law or as set forth in the Company's certificate of incorporation or bylaws shall be made by independent counsel mutually acceptable to the Company and the Indemnified Party; provided, however, that the Company shall not be liable for any settlement effected without its written consent (which consent shall not be unreasonably withheld or delayed). The Indemnified Parties as a group may retain only one law firm (other than local counsel) with respect to each related matter except to the extent there is, in the sole opinion of counsel to an Indemnified Party, under applicable standards of professional conduct, a conflict on any significant issue between positions of any two or more Indemnified Parties, in which case each Indemnified Party with a conflicting position on a significant issue shall be entitled to separate counsel. In the event any Indemnified Party is required to bring any action to enforce rights or to collect moneys due under this Agreement and is successful in such action, the Company shall reimburse such Indemnified Party for all of its expenses in bringing and pursuing such action. Each Indemnified Party shall be entitled to the advancement of expenses to the full extent contemplated in this Section 7.5(a) in connection with any such action.

(b) *Insurance.* For a period of six years after the Effective Time, the Company shall cause to be maintained in effect the policies of directors' and officers' liability insurance maintained by Delmarva and Atlantic; provided that the Company may substitute therefor policies of at least the same coverage containing terms that are no less advantageous with respect to matters occurring at or prior to the Effective Time to the extent such liability insurance can be maintained annually at a cost to the Company not greater than 200 percent of the current annual premiums for the policies currently maintained by Delmarva and Atlantic for their directors' and officers' liability insurance; provided further that if such

insurance cannot be so maintained or obtained at such cost, the Company shall maintain or obtain a policy providing the best coverage available, as determined by the Board of Directors of the Company, for a premium not exceeding 200 percent of the aggregate annual premiums currently paid by Delmarva and Atlantic for their directors' and officers' liability insurance and other indemnity agreements.

(c) *Successors*. In the event the Company or any of its successors or assigns (i) consolidates with or merges into any other person and shall not be the continuing or surviving corporation or entity of such consolidation or merger or (ii) transfers all or substantially all of its properties and assets to any person, then and in either such case proper provision shall be made so that the successors and assigns of the Company shall assume the obligations set forth in this Section 7.5.

(d) Survival of Indemnification. To the fullest extent not prohibited by law, from and after the Effective Time, all rights to indemnification now existing in favor of the employees, agents, directors or officers of Delmarva, Atlantic and their respective subsidiaries with respect to their activities as such prior to or at the Effective Time, as provided in their respective articles of incorporation or bylaws or indemnification agreements in effect on the date of such activities or otherwise in effect on the date hereof, shall survive the Mergers and shall continue in full force and effect for a period of not less than six years from the Effective Time; provided that in the event any claim or claims are asserted or made within such six-year period, all such rights to indemnification in respect of such claim or claims shall continue until the final disposition thereof.

Section 7.6 Disclosure Schedules. On or before the date of this Agreement, (i) Atlantic has delivered to Delmarva a schedule (the "Atlantic Disclosure Schedule") accompanied by a certificate signed by the chief financial officer of Atlantic stating that the Disclosure Schedule is being delivered pursuant to this Section 7.6(i) and (ii) Delmarva has delivered to Atlantic a schedule (the "Delmarva Disclosure Schedule") accompanied by a certificate signed by the chief financial officer of Delmarva by a certificate signed by the chief financial officer of Delmarva stating that the Delmarva Disclosure Schedule is being delivered pursuant to this Section 7.6(ii). The Atlantic Disclosure Schedule and the Delmarva Disclosure Schedule are collectively referred to herein as the "Disclosure Schedules". The Disclosure Schedules constitute an integral part of this Agreement and modify the respective representations, warranties, covenants or agreements expressly refer to the Disclosure Schedules. Any and all statements, representations, warranties or disclosures set forth in the Disclosure Schedules shall be deemed to have been made on and as of the date of this Agreement.

Section 7.7 *Public Announcements*. Delmarva and Atlantic shall cooperate with each other in the development and distribution of all news releases and other public information disclosures with respect to this Agreement or any of the transactions contemplated hereby and, subject to each party's disclosure obligations imposed by law or any applicable national securities exchange, shall not issue any public announcement or statement prior to consultation with the other party.

Section 7.8 *Rule 145 Affiliates*. Atlantic shall identify in a letter to Delmarva, and Delmarva shall identify in a letter to Atlantic, all persons who are and, to such persons' best knowledge who will be, at the Closing Date, "affiliates" of Atlantic and Delmarva, respectively, as such term is used in Rule 145 under the Securities Act. Atlantic and Delmarva shall use their respective best efforts to cause their respective affiliates to deliver to the Company on or prior to the Closing Date a written agreement substantially in the form attached hereto as Exhibit C (each, an "Affiliate Agreement").

# Section 7.9 Employee Agreements and Workforce Matters.

(a) Certain Employee Agreements. Subject to Section 7.10 and Section 7.14, the Company and its subsidiaries shall honor, without modification, all contracts, agreements, collective bargaining agreements and commitments of the parties that apply to any current or former employees or current or former directors of the parties hereto; provided, however, that this undertaking is not intended to prevent the Company from enforcing such contracts, agreements, collective bargaining agreements and commitments in accordance with their terms or from exercising any right to amend, modify, suspend, revoke or terminate any such contract, agreement, collective bargaining agreement.

(b) Workforce Matters. Subject to applicable collective bargaining agreements, for a period of two years following the Effective Time, any reductions in workforce in respect of employees of the Company shall be made on a fair and equitable basis, in light of the circumstances and the objectives to be achieved without regard to whether employment was with Delmarva or its subsidiaries or Atlantic or its subsidiaries, and any employees whose employment is terminated or jobs are eliminated by the Company or any of its subsidiaries during such period shall be entitled to participate on a fair and equitable basis in the job opportunity and employment placement programs offered by the Company or any of its subsidiaries. Any workforce reductions carried out following the Effective Time by the Company and its subsidiaries shall be done in accordance with all applicable collective bargaining agreements, and all laws and regulations governing the employment relationship thereof including without limitation the Worker Adjustment and Retraining Notification Act and regulations promulgated thereunder, and any comparable state or local law. However, no provision contained in this Section 7.9 shall be deemed to constitute an employment contract between the Company and any individual, be construed as conferring any rights on the collective bargaining representative of any employee, or be considered a waiver of the Company's right, subject to the provisions of any applicable collective bargaining agreement, to discharge any employee at any time, with or without cause.

(c) Subject to applicable collective bargaining agreements, the Company shall develop employee benefit programs and policies for all of its employees which minimize the differences between the employee benefit programs and policies maintained by Delmarva and Atlantic prior to the Effective Time and which are fair and equitable and appropriate to the business environment in which the Company will operate.

Section 7.10 Employee Benefit Plans. Each of the Delmarva Benefit Plans and Atlantic Benefit Plans (other than plans specifically provided for in Section 7.11), in effect on the date hereof (or as amended in accordance with or as permitted by this Agreement) shall be maintained in effect with respect to the employees or former employees of Delmarva and any of its subsidiaries and of Atlantic and any of its subsidiaries, respectively, who are covered by such plans immediately prior to the Closing Date until the Company determines otherwise on or after the Effective Time; provided, however, that nothing herein contained, other than the provisions of Section 6.9, shall limit any reserved right contained in any such Delmarva Benefit Plan or Atlantic Benefit Plan to amend, modify, suspend, revoke or terminate any such plan. Without limiting the foregoing, each participant in any Delmarva Benefit Plan or Atlantic Benefit Plan shall receive credit for purposes of eligibility to participate, vesting and eligibility to receive benefits under any benefit plan of the Company or any of its subsidiaries or affiliates for service credited for the corresponding purpose under any such benefit plan; provided, however, that such crediting of service shall not operate to duplicate any benefit to any such participant or the funding for any such benefit. However, no provision contained in this Section 7.10 shall be deemed to constitute an employment contract between the Company and any individual, or a waiver of the Company's right to discharge any employee at any time, with or without cause.

Section 7.11 *Incentive, Stock and Other Plans.* With respect to each of the Dividend Reinvestment and Common Share Purchase Plan, the Savings and Thrift Plan and the Long-Term Incentive Plan of Delmarva, and the Employee Incentive Plan, the Directors' Restricted Stock Plan, the Employee Stock Purchase Plan and the Dividend Reinvestment Plan of Atlantic, and each other employee benefit plan, program or arrangement under which the delivery of Delmarva Common Stock, Atlantic Common Stock or Company Common Stock, as the case may be, is required to be used for purposes of the payment of benefits, grant of awards or exercise of options (each, a "Stock Plan"), (i) Delmarva and Atlantic shall take such action as may be necessary so that, after the Effective Time, such Stock Plan shall provide for the issuance only of Company Common Stock and (ii) the Company shall (x) take all corporate action necessary or appropriate to obtain shareholder approval with respect to such Stock Plan to the extent such approval is required for purposes of the Code or other applicable law, or, to the extent the Company deems it desirable, to enable such Stock Plan to comply with Rule 16b-3 promulgated under the Exchange Act, (y) reserve for issuance under such Stock Plan or otherwise provide a sufficient number of shares of

Company Common Stock for delivery upon payment of benefits, grants of awards or exercise of options under such Stock Plan and (z) as soon as practicable after the Effective Time, file one or more registration statements under the Securities Act with respect to the shares of Company Common Stock subject to such Stock Plan to the extent such filing is required under applicable law and use its best efforts to maintain the effectiveness of such registration statement(s) (and the current status of the prospectuses contained therein or related thereto) so long as such benefits, grants or awards remain payable or such options remain outstanding, as the case may be. With respect to those individuals who subsequent to the Mergers will be subject to the reporting requirements under § 16(a) of the Exchange Act, the Company shall administer the Stock Plans, where applicable, in a manner that complies with Rule 16b-3 under the Exchange Act. Each of Delmarva and Atlantic shall obtain any shareholder approvals that may be necessary for the deduction of any compensation payable under any Stock Plan or other compensation arrangement.

Section 7.12 No Solicitations. From and after the date hereof, Delmarva and Atlantic shall not, and shall not authorize or permit any of their respective Representatives to, directly or indirectly, solicit, initiate or encourage (including by way of furnishing information) or take any other action to facilitate knowingly any inquiries or the making of any proposal that constitutes or may reasonably be expected to lead to an Acquisition Proposal (as defined herein) from any person, or engage in any discussion or negotiations relating thereto or accept any Acquisition Proposal; provided, however, that notwithstanding any other provision hereof, Delmarva or Atlantic may (i) at any time prior to the time at which the Delmarva Shareholders' Approval, in the case of Delmarva, or the Atlantic Shareholders' Approval, in the case of Atlantic, has been obtained, engage in discussions or negotiations with a third party who (without any solicitation, initiation, encouragement, discussion or negotiation, directly or indirectly, by or with Delmarva or Atlantic, as the case may be, or its Representatives after the date hereof) seeks to initiate such discussions or negotiations and may furnish such third party information concerning itself and its business, properties and assets if, and only to the extent that, (A) (x) such third party shall first have made an Acquisition Proposal that is financially superior to the Mergers and have demonstrated that financing for such Acquisition Proposal is reasonably likely to be obtained (as determined in good faith in each case) by the Board of Directors of Delmarva or Atlantic, as the case may be, after consultation with its financial advisors) and (y) the Board of Directors of Delmarva or Atlantic, as the case may be, shall have concluded in good faith on the basis of a written opinion of outside counsel that such action is necessary for such Board of Directors to act in a manner consistent with its fiduciary duties under applicable law and (B) prior to furnishing such information to or entering into discussions or negotiations with such third party, Delmarva or Atlantic, as the case may be, (x) provides prompt notice to Atlantic or Delmarva, as the case may be, to the effect that it is furnishing information to or entering into discussions or negotiations with such third party and (y) receives from such third party an executed confidentiality agreement in reasonably customary form on terms not in the aggregate materially more favorable to such third party than the terms contained in the Confidentiality Agreement, (ii) comply with Rule 14e-2 promulgated under the Exchange Act with regard to a tender or exchange offer, and (iii) accept an Acquisition Proposal from a third party, provided Delmarva or Atlantic, as the case may be, terminates this Agreement pursuant to Section 9.1(g) or 9.1(h), as the case may be. Each party shall immediately cease and terminate any existing solicitation, initiation, encouragement, activity, discussion or negotiation with any parties conducted heretofore by such party or its Representatives with respect to the foregoing. Atlantic and Delmarva shall each notify the other orally and in writing of any such inquiries, offers or proposals (including without limitation the terms and conditions of any such proposal and the identity of the person making it) within 24 hours of the receipt thereof, shall keep the other informed of the status and details of any such inquiry, offer or proposal, and shall give the other five days' advance notice of any agreement to be entered into with, or any information to be supplied to, any person making such inquiry, offer or proposal. As used herein, "Acquisition Proposal" shall mean a proposal or offer (other than by another party hereto) for a tender or exchange offer, merger, consolidation or other business combination involving a party or any material subsidiary of such party or any proposal to acquire in any manner a

ŝ

substantial equity interest in or a substantial portion of the assets of such party or any material subsidiary of such party.

Section 7.13 Company Board of Directors. The Board of Directors of Delmarva shall be entitled to nominate 10 members and the Board of Directors of Atlantic shall be entitled to nominate 8 members to serve on the Board of Directors of the Company at the Effective Time and shall each take such action as may be necessary to cause each director of Delmarva and Atlantic serving as such immediately prior to the Effective Time to have the opportunity to serve as a director of the Company. The directors nominated by the Delmarva Board and directors nominated by the Atlantic Board shall be divided between the classes of directors of the Company so that each class, to the extent possible, has the same proportion of directors nominated by each of the Delmarva Board and the Atlantic Board. At the Effective Time, the Audit Committee of the Board of Directors of the Company shall consist of an equal number of directors nominated by the Delmarva Board and the Atlantic Board.

Section 7.14 *Company Officers*. At the Effective Date, Howard E. Cosgrove shall be the Chief Executive Officer and Chairman of the Board of Directors of the Company, Jerrold L. Jacobs shall be the Vice Chairman of the Board of Directors of the Company and Michael J. Chesser shall be the President and Chief Operating Officer of the Company. Jerrold L. Jacobs shall serve as Vice Chairman of the Company until the second anniversary of the Effective Date and, during his term as Vice Chairman, shall be a member of the Executive Committee of the Board of Directors of the Company. The other officers of the Company at the Effective Time shall be such officers as may be designated by the Board of Directors of the Company.

Section 7.15 Location of Corporate Offices and Operations; Company Name. At and after the Effective Time, the corporate headquarters and principal executive offices of the Company shall be located in Wilmington, Delaware, and the Company shall maintain a significant presence in New Jersey. After the Effective Time, the Company shall provide charitable contributions and community support within the service areas of Delmarva and Atlantic and each of their respective subsidiaries at levels substantially comparable to the historical levels of charitable contributions and community support provided by Delmarva and Atlantic and their respective subsidiaries within their service areas. The Company's name shall be as agreed upon by the Board of Directors of Delmarva and the Board of Directors of Atlantic following the completion of marketing studies which will give serious consideration to names containing "Atlantic" or variations of the names of Atlantic and its subsidiaries.

Section 7.16 Company Certificate of Incorporation and Bylaws. Delmarva, Atlantic and the Company shall take all actions necessary so that (i) at or prior to the Effective Time, the certificate of incorporation of the Company shall be amended and restated so that, at the Effective Time, such certificate of incorporation shall read in its entirety substantially in the form attached hereto as Exhibit A (the "Company's Restated Charter") and (ii) at or prior to the Effective Time, the bylaws of the Company shall be amended and restated so that, at the Effective Time, the bylaws of the Company shall be amended and restated so that, at the Effective Time, such bylaws shall read in their entirety substantially in the form attached hereto as Exhibit B.

Section 7.17 *Expenses*. Subject to Section 7.1 and Section 9.3, all costs and expenses incurred in connection with this Agreement and the transactions contemplated hereby shall be paid by the party incurring such expenses, except that those expenses incurred in connection with printing the Joint Proxy/ Registration Statement, as well as the filing fee relating thereto, shall be shared equally by Delmarva and Atlantic.

Section 7.18 Letter Stock Dividend Policy. Subject to declaration by the Board of Directors of the Company and the obligation of the Board of Directors of the Company to react to the financial condition and regulatory environment of the Company and its results of operations, the dividends declared and paid on the Letter Stock shall be maintained at a level of \$3.20 per share per annum until the earlier of July 1, 2001 or the end of the twelfth calendar quarter following the calendar quarter in which the Effective Date occurs. Thereafter, it is the intention of the Company, subject to declaration by the Board of Directors of the Company and the obligation of the Board of Directors of the Company to react to the financial

condition and regulatory environment of the Company and the results of its operations, to pay annual dividends on the Letter Stock (the amount of such dividends to include the amount credited to the Intergroup Interest as provided in the Company's Restated Charter) at a rate equal to 90% of Net Income (Loss) Attributable to the Atlantic Utility Group; provided that if, and to the extent that, the annual dividends on the Letter Stock paid during the period referred to in the preceding sentence exceeds 100% of Net Income (Loss) Attributable to the Atlantic Utility Group during such period, the Board of Directors of the Company may consider such fact in determining the appropriate annual dividend rate on the Letter Stock thereafter. Following the Effective Time, the Audit Committee of the Board of Directors of the Company shall be charged with the responsibility of advising the Board of Directors of the Company with respect to certain intercompany transactions and other fiduciary matters that may relate to the Letter Stock.

## Section 7.19 Further Assurances.

(a) Each of Atlantic and Delmarva shall, and shall cause its subsidiaries to, execute such further documents and instruments and take such further actions as may reasonably be requested by the other in order to consummate the Mergers and the other transactions contemplated by this Agreement, and to use its best efforts to take or cause to be taken all actions, and to do or cause to be done all things, necessary, proper or advisable under applicable laws and regulations to consummate and make effective the Mergers and the other transactions contemplated hereby (subject to the votes of its shareholders described in Sections 4.13 and 5.13, respectively), including fully cooperating with the other in obtaining the Atlantic Required Statutory Approvals, the Delmarva Required Statutory Approvals and all other approvals and authorizations of any Governmental Authorities necessary or advisable to consummate the transactions contemplated hereby.

(b) Atlantic and Delmarva, respectively, shall be responsible for the taking of any action necessary or advisable to obtain the Atlantic Required Statutory Approvals and to obtain the Delmarva Required Statutory Approvals. Atlantic and Delmarva agree to cooperate in obtaining the necessary approvals from the NRC, the FERC and the SEC under the 1935 Act, the Securities Act and the Exchange Act and from the applicable state authorities under state "blue sky", securities or utility regulatory laws. Atlantic and Delmarva shall each provide the other with copies of any filings made with any Governmental Authorities in connection with the foregoing.

(c) It may be preferable to effectuate a business combination between Delmarva and Atlantic by means of an alternative structure in light of the conditions set forth in Sections 8.1(e), 8.2(f) and 8.3(f). Accordingly, if the only conditions to the parties' obligations to consummate the Mergers that are not satisfied or waived are receipt of any one or more of the Delmarva Required Consents, Delmarva Statutory Approvals, Atlantic Required Consents and Atlantic Statutory Approvals, and the adoption of an alternative structure (that otherwise substantially preserves for Delmarva and Atlantic the economic benefits of the Mergers) would result in such conditions being satisfied or waived, then the parties shall use their respective best efforts to effect a business combination among themselves by means of a mutually agreed upon structure other than the Mergers that so preserves such benefits; provided that prior to closing any such restructured transaction, all material third party and Governmental Authority declarations, filings, registrations, notices, authorizations, consents or approvals necessary for the effectuation of such alternative business combination shall have been obtained and all other conditions to the parties' obligations to consummate the Mergers, as applied to such alternative business combination, shall have been satisfied or waived.

### **ARTICLE VIII**

### CONDITIONS

Section 8.1 Conditions to Each Party's Obligation to Effect the Merger to Which it is Party. The respective obligations of each party to effect the Merger to which it is party shall be subject to the satisfaction on or prior to the Closing Date of the following conditions, except, to the extent permitted by applicable law, that such conditions may be waived in writing pursuant to Section 9.5:

(a) *Shareholder Approvals*. The Atlantic Shareholders' Approval and the Delmarva Shareholders' Approval shall have been obtained.

(b) *No Injunction.* No temporary restraining order or preliminary or permanent injunction or other order by any federal or state court preventing consummation of either or both of the Mergers shall have been issued and continuing in effect, and the Mergers and the other transactions contemplated hereby shall not have been prohibited under any applicable federal or state law or regulation.

(c) *Registration Statement.* The Registration Statement shall have become effective in accordance with the provisions of the Securities Act, and no stop order suspending such effectiveness shall have been issued and remain in effect.

(d) *Listing of Shares.* The shares of Company Common Stock and the shares of Letter Stock issuable in the Mergers pursuant to Article II shall have been approved for listing on the NYSE upon official notice of issuance.

(e) Statutory Approvals. The Delmarva Required Statutory Approvals and the Atlantic Required Statutory Approvals shall have been obtained at or prior to the Effective Time, such approvals shall have become Final Orders (as hereinafter defined), and no Final Order shall impose terms or conditions that would have, or would be reasonably likely to have, a material adverse effect on the business, operations, properties, assets, condition (financial or otherwise), prospects or results of operations of Delmarva as if it were organized as a separate subsidiary of the Company or a material adverse effect on the business, operations, properties, assets, condition (financial or other), prospects or results of operations of Atlantic as if it were organized as a separate division of the Company, or which would be materially inconsistent with the agreements of the parties contained herein. A "Final Order" means action by the relevant regulatory authority that has not been reversed, stayed, enjoined, set aside, annulled or suspended, with respect to which any waiting period prescribed by law before the transactions contemplated hereby may be consummated has expired, and as to which all conditions to the consummation of such transactions prescribed by law, regulation or order have been satisfied, and as to which all opportunities for rehearing are exhausted (whether or not any appeal thereof is pending).

Section 8.2 Conditions to Obligation of Atlantic to Effect the Atlantic Merger. The obligation of Atlantic to effect the Atlantic Merger shall be further subject to the satisfaction, on or prior to the Closing Date, of the following conditions, except as may be waived by Atlantic in writing pursuant to Section 9.5:

(a) *Performance of Obligations of Delmarva*. Delmarva shall have performed in all material respects its agreements and covenants contained in or contemplated by this Agreement required to be performed by it at or prior to the Effective Time.

(b) *Representations and Warranties.* The representations and warranties of Delmarva set forth in this Agreement qualified as to materiality shall be true in all respects and those not so qualified shall be true and correct in all material respects as of the date hereof and as of the Closing Date as if made on and as of the Closing Date, except as otherwise contemplated by this Agreement.

(c) *Closing Certificates.* Atlantic shall have received a certificate signed by the Chief Executive Officer and Chief Financial Officer of Delmarva, dated the Closing Date, to the effect that, to the best





of each such officer's knowledge, the conditions set forth in Section 8.2(a) and Section 8.2(b) have been satisfied.

(d) Delmarva Material Adverse Effect. No Delmarva Material Adverse Effect shall have occurred, and there shall exist no fact or circumstance that is reasonably likely to have a Delmarva Material Adverse Effect.

(e) Tax Opinion. Atlantic shall have received an opinion of Simpson Thacher & Bartlett, in form and substance satisfactory to Atlantic, dated the Closing Date, which opinion may be based on appropriate representations of Delmarva, Atlantic and the Company that are in form and substance reasonably satisfactory to such counsel, to the effect that the Atlantic Merger will be treated as a reorganization described in Code § 368(a).

(f) Delmarva Required Consents. The material Delmarva Required Consents shall have been obtained.

(g) Affiliate Certificates. The Company shall have received Affiliate Agreements, duly executed by each "affiliate" of Delmarva, substantially in the form of Exhibit C, as provided in Section 7.8.

Section 8.3 Conditions to Obligation of Delmarva to Effect the Delmarva Merger. The obligation of Delmarva to effect the Delmarva Merger shall be further subject to the satisfaction, on or prior to the Closing Date, of the following conditions, except as may be waived by Delmarva in writing pursuant to Section 9.5:

(a) *Performance of Obligations of Atlantic.* Atlantic shall have performed in all material respects its agreements and covenants contained in or contemplated by this Agreement required to be performed by it at or prior to the Effective Time.

(b) *Representations and Warranties.* The representations and warranties of Atlantic set forth in this Agreement qualified as to materiality shall be true in all respects and those not so qualified shall be true and correct in all material respects as of the date hereof and as of the Closing Date as if made on and as of the Closing Date, except as otherwise contemplated by this Agreement.

(c) *Closing Certificates.* Delmarva shall have received a certificate signed by the Chief Executive Officer and Chief Financial Officer of Atlantic, dated the Closing Date, to the effect that, to the best of each such officer's knowledge, the conditions set forth in Section 8.3(a) and Section 8.3(b) have been satisfied.

(d) Atlantic Material Adverse Effect. No Atlantic Material Adverse Effect shall have occurred, and there shall exist no fact or circumstance that is reasonably likely to have an Atlantic Material Adverse Effect.

(e) Tax Opinion. Delmarva shall have received an opinion of LeBoeuf, Lamb, Greene & MacRae, L.L.P., in form and substance satisfactory to Delmarva, dated the Closing Date, which opinion may be based on appropriate representations of Delmarva, Atlantic and the Company that are in form and substance reasonably satisfactory to such counsel, to the effect that (i) the Delmarva Merger, taken together with the Atlantic Merger, will be treated as a nontaxable exchange described in Code § 351 and/or (ii) the Delmarva Merger will be treated as a nontaxable reorganization described in Section 368.

(f) Atlantic Required Consents. The material Atlantic Required Consents shall have been obtained.

(g) Affiliate Certificates. The Company shall have received Affiliate Agreements, duly executed by each "affiliate" of Atlantic, substantially in the form of Exhibit C, as provided in Section 7.7.

## ARTICLE IX

### TERMINATION, AMENDMENT AND WAIVER

Section 9.1 *Termination*. This Agreement may be terminated at any time prior to the Closing Date, whether before or after approval by the shareholders of the respective parties contemplated by this Agreement:

(a) by mutual written consent of the Boards of Directors of Delmarva and Atlantic;

(b) by Delmarva or Atlantic, by written notice to the other, if the Effective Time shall not have occurred on or before 18 months from signing; provided, however, that such date shall automatically be extended to 30 months from signing if, on 18 months from signing: (i) the condition set forth in Section 8.1(e) has not been satisfied or waived; (ii) the other conditions to the consummation of the transactions contemplated hereby are then capable of being satisfied; and (iii) any approvals required by Section 8.1(e) that have not yet been obtained are being pursued with diligence; provided further that the right to terminate this Agreement under this Section 9.1(b) shall not be available to any party whose failure to fulfill any obligation under this Agreement has been the cause of, or resulted in, the failure of the Effective Time to occur on or before the termination date;

(c) by Delmarva or Atlantic, by written notice to the other, if the Delmarva Shareholders' Approval shall not have been obtained at a duly held Delmarva Special Meeting, including any adjournments thereof, or the Atlantic Shareholders' Approval shall not have been obtained at a duly held Atlantic Special Meeting, including any adjournments thereof;

(d) by Delmarva or Atlantic, if any state or federal law, order, rule or regulation is adopted or issued that has the effect, as supported by the written opinion of outside counsel for such party, of prohibiting either or both of the Mergers, or if any court of competent jurisdiction in the United States or any State shall have issued an order, judgment or decree permanently restraining, enjoining or otherwise prohibiting either or both of the Mergers and such order, judgment or decree shall have become final and nonappealable;

(e) by Delmarva, upon five days' prior notice to Atlantic, if, as a result of an Acquisition Proposal by a party other than Atlantic or any of its affiliates, the Board of Directors of Delmarva determines in good faith on the basis of a written opinion of outside counsel that acceptance of such Acquisition Proposal is necessary for the Board of Directors to act consistent with its fiduciary duties under applicable law; provided, however, that (i) the Board of Directors of Delmarva shall have been advised by outside counsel that, notwithstanding a binding commitment to consummate an agreement of the nature of this Agreement entered into in the proper exercise of their applicable fiduciary duties, and notwithstanding all concessions that may be offered by Atlantic in negotiations entered into pursuant to clause (ii) below, such fiduciary duties would also require the directors to reconsider such commitment as a result of such Acquisition Proposal and (ii) prior to any such termination, Delmarva shall, and shall cause its respective financial and legal advisors to, negotiate with Atlantic to make such adjustments in the terms and conditions of this Agreement as would enable Delmarva to proceed with the transactions contemplated herein; provided further that Delmarva and Atlantic acknowledge and affirm that, notwithstanding anything in this Section 9.1(e) to the contrary, Delmarva and Atlantic intend this Agreement to be an exclusive agreement and, accordingly, nothing in this Agreement is intended to constitute a solicitation of an Acquisition Proposal, it being acknowledged and agreed that any such proposal would interfere with the strategic advantages and benefits that Delmarva and Atlantic expect to derive from the Mergers and the other transactions contemplated hereby;

(f) by Atlantic, upon five days' prior notice to Delmarva, if, as a result of an Acquisition Proposal by a party other than Delmarva or any of its affiliates, the Board of Directors of Atlantic determines in good faith on the basis of written advice of outside counsel that acceptance of such Acquisition Proposal is necessary for the Board of Directors to act consistent with its fiduciary duties



under applicable law; provided, however, that (i) the Board of Directors of Atlantic shall have been advised by outside counsel that, notwithstanding a binding commitment to consummate an agreement of the nature of this Agreement entered into in the proper exercise of their applicable fiduciary duties, and notwithstanding all concessions that may be offered by Delmarva in negotiations entered into pursuant to clause (ii) below, such fiduciary duties would also require the directors to reconsider such commitment as a result of such Acquisition Proposal and (ii) prior to any such termination, Atlantic shall, and shall cause its respective financial and legal advisors to, negotiate with Delmarva to make such adjustments in the terms and conditions of this Agreement as would enable Atlantic to proceed with the transactions contemplated herein; provided further that Delmarva and Atlantic acknowledge and affirm that, notwithstanding anything in this Section 9.1(f) to the contrary, Delmarva and Atlantic intend this Agreement to be an exclusive agreement and, accordingly, nothing in this Agreement is intended to constitute a solicitation of an Acquisition Proposal, it being acknowledged and agreed that any such proposal would interfere with the strategic advantages and benefits that Delmarva and Atlantic expect to derive from the Mergers and the other transactions contemplated hereby;

(g) by Delmarva, by written notice to Atlantic, if (i) there shall have been any material breach of any representation or warranty, or any material breach of any covenant or agreement, of Atlantic hereunder, and such breach shall not have been remedied within 20 days after receipt by Atlantic of notice in writing from Delmarva, specifying the nature of such breach and requesting that it be remedied, or (ii) the Board of Directors of Atlantic or any committee thereof (A) shall withdraw or modify in any manner adverse to Delmarva its approval or recommendation of this Agreement or the Atlantic Merger, (B) shall fail to reaffirm such approval or recommendation upon Delmarva's request, (C) shall approve or recommend any Acquisition Proposal with respect to Atlantic by a party other than Delmarva or any of its affiliates or (D) shall resolve to take any of the actions specified in clause (A), (B) or (C) above.

(h) by Atlantic, by written notice to Delmarva, if (i) there shall have been any material breach of any representation or warranty, or any material breach of any covenant or agreement, of Delmarva hereunder, and such breach shall not have been remedied within 20 days after receipt by Delmarva of notice in writing from Atlantic, specifying the nature of such breach and requesting that it be remedied, or (ii) the Board of Directors of Delmarva or any committee thereof (A) shall withdraw or modify in any manner adverse to Atlantic its approval or recommendation of this Agreement or the Delmarva Merger, (B) shall fail to reaffirm such approval or recommendation upon Atlantic's request, (C) shall approve or recommend any Acquisition Proposal with respect to Delmarva by a party other than Atlantic or any of its affiliates or (D) shall resolve to take any of the actions specified in clause (A), (B) or (C) above.

(i) by either Delmarva or Atlantic, by written notice to the other party, if (i) a third party acquires securities representing greater than 50% of the voting power of the outstanding voting securities of Atlantic or Delmarva, as the case may be, or (ii) individuals who as of the date hereof constitute the Board of Directors of Atlantic or Delmarva, as the case may be (together with any new directors whose election by such Board of Directors or whose nomination for election by the shareholders of such party was approved by a vote of a majority of the directors of such party then still in office who are either directors as of the date hereof or whose election or nomination for election was previously so approved), cease for any reason to constitute a majority of the Board of Directors of Atlantic or Delmarva, as the case may be, then in office.

Section 9.2 *Effect of Termination*. In the event of termination of this Agreement by either Delmarva or Atlantic pursuant to Section 9.1, there shall be no liability hereunder on the part of either Delmarva or Atlantic or their respective officers or directors, except that (i) Section 7.17 (expenses), Section 9.3 (termination fee) and the agreement contained in the second to the last sentence of Section 7.1 (confidentiality) shall survive any such termination and (ii) no such termination shall relieve any party from liability by reason of any willful breach of any agreement, representation, warranty or covenant contained in this Agreement.

### Section 9.3 Certain Damages, Payments and Expenses.

(a) Damages Payable Upon Termination for Breach or Withdrawal of Approval. If this Agreement is terminated pursuant to Section 9.1(e) or (f) (fiduciary out), Section 9.1(g)(i) or (ii) or Section 9.1(h)(i) or (ii) (breach or change of recommendation) or Section 9.1(i) (acquisition of voting power or change of board), then the breaching party or party whose board has exercised its fiduciary out or changed its recommendation or whose voting stock has been acquired or whose board has changed, as the case may be, shall promptly (but not later than five business days after receipt of notice that the amount is due from the other party) pay to the other party, as liquidated damages, an amount in cash equal to the of out-of-pocket expenses and fees incurred by the other party arising out of, in connection with or related to the Mergers and the other transactions contemplated by this Agreement not in excess of \$10 million ("Out-of-Pocket Expenses"); provided, however, that if this Agreement is terminated by a party as a result of a willful breach of a representation, warranty, covenant or agreement by the other party, the nonbreaching party may pursue any remedies available to it at law or in equity and shall, in addition to the amount of Out-of-Pocket Expenses set forth above, be entitled to recover such additional amounts as such nonbreaching party may be entitled to receive at law or in equity.

(b) Other Termination Payments. If (i) this Agreement is terminated pursuant to (A) Section 9.1(b) (expiration date), (B) Section 9.1(e) or (f) (fiduciary out), (C) Section 9.1(c) (failure to obtain shareholder approval), (D) as a result of a breach of Section 7.4 (approval of shareholders) or (E) pursuant to Section 9.1(g)(i) or (ii) or 9.1(h)(i) or (ii) (breach or change of recommendation); and (ii) at the time of such termination (or in the case of clause (i)(C) above, prior to the meeting of such party's shareholders) there shall have been an Acquisition Proposal involving the Delmarva or Atlantic (as the case may be, the "Target Party") or any of its affiliates which, at the time of such termination (or such meeting, as the case may be) shall not have been (x) rejected by the Target Party and its Board of Directors and (y) withdrawn by the third party; and (iii) within two and one-half years of any such termination described in clause (i) above, the Target Party or any of its affiliates becomes a subsidiary of such offeror or a subsidiary of an affiliate of such offeror or accepts a written offer or enters into a written agreement to consummate or consummates an Acquisition Proposal with such offeror or an affiliate thereof, then such Target Party (jointly and severally with its affiliates), upon the signing of a definitive agreement relating to such Acquisition Proposal, or, if no such agreement is signed, then at the closing (and as a condition to the closing) of such Target Party becoming such a subsidiary or of such Acquisition Proposal, shall pay Atlantic or Delmarva, as the case may be, a termination fee equal to \$30 million plus Out-of-Pocket Expenses. If this Agreement is terminated by Delmarva or Atlantic pursuant to Section 9.1(i) (third party acquisition of voting power or change of board), then Atlantic or Delmarva, as the case may be, shall pay the terminating party a termination fee equal to \$30 million plus Out-of-Pocket Expenses.

(c) *Expenses.* Delmarva and Atlantic agree that the agreements contained in this Section 9.3 are an integral part of the transactions contemplated by this Agreement and constitute liquidated damages and not a penalty. If one party fails to promptly pay to the other any fees due hereunder, such defaulting party shall pay the costs and expenses (including legal fees and expenses) in connection with any action, including the filing of any lawsuit or other legal action, taken to collect payment, together with interest on the amount of any unpaid fee at the publicly announced prime rate of Citibank, N.A. in effect from time to time from the date such fee was required to be paid.

(d) Limitation of Fees. Notwithstanding anything herein to the contrary, the aggregate amount payable by Delmarva and its affiliates pursuant to Sections 9.3(a) and 9.3(b) shall not exceed \$40 million and the aggregate amount payable by Atlantic and its affiliates pursuant to Sections 9.3(a) and 9.3(b) shall not exceed \$40 million.

Section 9.4 *Amendment*. This Agreement may be amended by the parties hereto or thereto pursuant to action of the respective Boards of Directors of each of Delmarva and Atlantic, the Company and DS

Sub at any time before or after approval hereof by the shareholders of Delmarva, Atlantic and prior to the Effective Time, but after such approvals no such amendment shall (a) alter or change the amount or kind of shares, securities or cash to be received or exchanged for or on conversion of any class or series of capital stock of either Delmarva or Atlantic as provided in Article II, (b) alter or change any of the terms and conditions of this Agreement if any of the alterations or changes, alone or in the aggregate, would adversely affect the rights of holders of any class or series of stock of Delmarva, except for alterations or change any term of the certificate of incorporation of the Company or Delmarva, except for alterations or changes that could otherwise be adopted by the Board of Directors of the Company, without the further approval of such shareholders, as applicable. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the parties hereto or thereto.

Section 9.5 *Waiver*. At any time prior to the Effective Time, the parties may (a) extend the time for the performance of any of the obligations or other acts of the other parties, (b) waive any inaccuracies in the representations and warranties contained herein or in any document delivered pursuant hereto and (c) waive compliance with any of the agreements or conditions contained herein. Any agreement to any such extension or waiver shall be valid only if set forth in an instrument in writing signed by a duly authorized officer of each party.

#### ARTICLE X

### **GENERAL PROVISIONS**

Section 10.1 Nonsurvival of Representations, Warranties, Covenants and Agreements. All representations, warranties, covenants and agreements in this Agreement shall not survive the Mergers, except the covenants and agreements contained in this Section 10.1 and in Article II (Treatment of Shares), the second to the last sentence of Section 7.1 (Confidentiality), Section 7.5 (Directors' and Officers' Indemnification), Section 7.9 (Employee Agreements and Workforce Matters), Section 7.10 (Employee Benefit Plans), Section 7.11 (Incentive, Stock and Other Plans), Section 7.13 (Company Board of Directors) Section 7.14 (Company Officers), Section 7.15 (Location of Corporate Offices and Operations; Company Name), Section 7.17 (Expenses), Section 7.18 (Letter Stock Dividend Policy) and Section 10.7 (Parties in Interest), each of which shall survive in accordance with its terms.

Section 10.2 *Brokers*. Delmarva represents and warrants that, except for Merrill Lynch, Pierce, Fenner & Smith Incorporated, its investment banking firm, no broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission in connection with the Mergers or the transactions contemplated by this Agreement based upon arrangements made by or on behalf of Delmarva. Atlantic represents and warrants that, except for Morgan Stanley & Co. Incorporated, its investment banking firm, no broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission in connection with the Mergers or the transactions in connection with the Mergers or the transactions contemplated by this Agreement banker is entitled to any brokerage, finder's or other fee or commission in connection with the Mergers or the transactions contemplated by this Agreement based upon arrangements made by or on behalf of Atlantic.

Section 10.3 *Notices*. All notices and other communications hereunder shall be in writing and shall be deemed given (a) if delivered personally, or (b) if sent by overnight courier service (receipt confirmed in writing), or (c) if delivered by facsimile transmission (with receipt confirmed), or (d) five days after being

. .

mailed by registered or certified mail (return receipt requested) to the parties, in each case to the following addresses (or at such other address for a party as shall be specified by like notice):

(i) If to Delmarva, to:

Delmarva Power & Light Company 800 King Street Wilmington, Delaware 19899 Attention: Barbara S. Graham

with a copy to:

LeBoeuf, Lamb, Greene & MacRae, L.L.P. 125 West 55th Street New York, New York 10019 Attention: Douglas W. Hawes, Esq. Steven H. Davis, Esq.

and a copy to:

Potter Anderson & Corroon 350 Delaware Trust Building P.O. Box 951 Wilmington, Delaware 19899 Attention: Robert K. Payson, Esq. Michael B. Tumas, Esq.

(ii) If to Atlantic, to:

Atlantic Energy, Inc. 6801 Black Horse Pike Pleasantville, NJ 08232 Attention: Michael J. Barron

with a copy to:

Simpson Thacher & Bartlett 425 Lexington Avenue New York, New York 10017 Attention: James M. Cotter Vincent Pagano, Jr.

Section 10.4 Entire Agreement; Assignment; Governing Law, Waiver of Jury Trial; Etc. This Agreement (including the documents and instruments referred to herein) constitutes the entire agreement and supersedes all other prior agreements and understandings, both written and oral, among the parties, or any of them, with respect to the subject matter hereof other than the Confidentiality Agreement. This Agreement shall not be assigned by operation of law or otherwise. This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware applicable to contracts executed in and to be fully performed in such state, without giving effect to its conflicts of laws statutes, rules or principles. Each party hereto acknowledges and agrees that any controversy that may arise under this Agreement is likely to involve complicated and difficult issues, and therefore each such party hereby irrevocably and unconditionally waives any right such party may have to a trial by jury in respect of any litigation directly or indirectly arising out of or relating to this Agreement or the transactions contemplated hereby. Each party certifies and acknowledges that (i) no representative, agent or attorney of any other party has represented, expressly or otherwise, that such other party would not, in the event of litigation, seek to enforce the foregoing waiver, (ii) each such party understands and has considered the implications of this waiver, (iii) each such party makes this waiver voluntarily, and (iv) each such party has been induced to enter into this Agreement by, among other things, the mutual waivers and certifications of this Section 10.4. The invalidity or unenforceability of any provision of this Agreement shall not affect the validity or enforceability of any other provision of this Agreement, which shall remain in full force and effect. The parties hereto shall negotiate in good faith to replace any provision of this Agreement so held invalid or unenforceable with a valid provision that is as similar as possible in substance to the invalid or unenforceable provision.

Section 10.5 Interpretation. When reference is made in this Agreement to Articles, Sections or Exhibits, such reference shall be to an Article, Section or Exhibit of this Agreement, as the case may be, unless otherwise indicated. The table of contents and headings contained in this Agreement are for reference purposes and shall not affect in any way the meaning or interpretation of this Agreement. Whenever the words "include", "includes", or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation." Whenever "or" is used in this Agreement it shall be construed in the nonexclusive sense.

Section 10.6 *Counterparts; Effect.* This Agreement may be executed in one or more counterparts, each of which shall be deemed to be an original, but all of which shall constitute one and the same agreement.

Section 10.7 *Parties in Interest*. This Agreement shall be binding upon and inure solely to the benefit of each party hereto, and, except for rights of Indemnified Parties as set forth in Section 7.5 (Directors' and Officers' Indemnification), nothing in this Agreement, express or implied, is intended to confer upon any person any rights or remedies of any nature whatsoever under or by reason of this Agreement.

Section 10.8 Specific Performance. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties hereto shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court of the United States or any state having jurisdiction, this being in addition to any other remedy to which they are entitled at law or in equity.

IN WITNESS WHEREOF, Delmarva, Atlantic, the Company and DS Sub have caused this Agreement, as amended and restated as of December 26, 1996 to be signed by their respective officers thereunto duly authorized as of the date first above written.

## **DELMARVA POWER & LIGHT COMPANY**

By /s/ HOWARD E. COSGROVE

Name: Howard E. Cosgrove Title: President and Chairman

## ATLANTIC ENERGY, INC.

By

/s/ Jerrold L. Jacobs

Name: Jerrold L. Jacobs Title: President and Chief Executive Officer

# CONECTIV, INC.

By

/s/ BARBARA S. GRAHAM

Name: Barbara S. Graham Title: President

And By /s/ MICHAEL J. BARRON

Name: Michael J. Barron Title: Vice President

DS SUB, INC.

By ·

/s/ Barbara S. Graham

Name: Barbara S. Graham Title: President

ANNEX II

Investment Banking

Corporate and Institutional Client Group

World Financial Center North Tower New York, New York 10281-1330 212 449 1000

December 26, 1996

Merrill Lynch

Board of Directors Delmarva Power & Light Company 800 King Street Wilmington, DE 19899

Directors:

Delmarva Power & Light Company, a corporation formed under the laws of the State of Delaware and the Commonwealth of Virginia ("Delmarva"), Atlantic Energy, Inc., a corporation formed under the laws of the State of New Jersey ("Atlantic"), Conectiv, Inc., a Delaware corporation whose capital stock is 50% owned by each of Delmarva and Atlantic, respectively (the "Company"), and DS Sub, Inc., a Delaware corporation and wholly owned subsidiary of the Company ("DS Sub"), entered into an Agreement and Plan of Merger, dated as of August 9, 1996, as amended and restated as of December 26, 1996 (the "Agreement"), pursuant to which Atlantic will merge with and into the Company (the "Atlantic Merger") and DS Sub will merge with and into Delmarva (the "Delmarva Merger" and, together with the Atlantic Merger, the "Mergers"). In the Delmarva Merger, each issued and outstanding share (other than shares canceled pursuant to section 2.1(a) of the Agreement) of Delmarva common stock, par value \$2.25 per share ("Delmarva Common Stock"), shall be converted into the right to receive one share of Company common stock, par value \$0.01 per share ("Company Common Stock"), and in the Atlantic Merger, each issued and outstanding share (other than shares canceled pursuant to section 2.1(a) of the Agreement) of Atlantic common stock, no par value ("Atlantic Common Stock"), shall be converted into the right to receive 0.750 shares of Company Common Stock and 0.125 shares of Company Class A common stock, par value \$0.01 per share. The ratio at which Delmarva Common Stock is converted into Company Common Stock, in accordance with the Agreement, is referred to herein as the "Delmarva Conversion Ratio." Consummation of the Mergers is subject to the terms and conditions set forth in the Agreement.

You have asked us whether, in our opinion, the Delmarva Conversion Ratio contemplated by the Agreement is fair to Delmarva and its shareholders from a financial point of view.

In arriving at the opinion set forth below, we have, among other things:

(1) Reviewed Delmarva's Annual Reports, Forms 10-K and related financial information for the three fiscal years ended December 31, 1995 and Delmarva's Form 10-Q and the related unaudited financial information for the quarterly periods ended September 30, 1996;



- (2) Reviewed Atlantic's Annual Reports, Forms 10-K and related financial information for the three fiscal years ended December 31, 1995 and Atlantic's Form 10-Q and the related unaudited financial information for the quarterly periods ended September 30, 1996;
- (3) Reviewed certain other filings with the Securities and Exchange Commission made by Delmarva and Atlantic, including proxy statements, Forms 8-K and registration statements, during the last three years;
- (4) Reviewed certain information, including financial forecasts, relating to the business, earnings, cash flow, assets and prospects of Delmarva and Atlantic, furnished to us by Delmarva and Atlantic, respectively;
- (5) Conducted discussions with members of senior management of Delmarva and Atlantic concerning their respective businesses, regulatory environments, prospects and strategic objectives and possible operating, administrative and capital synergies which might be realized for the Company following the Mergers;
- (6) Reviewed the historical market prices and trading activity for the Delmarva Common Stock and the Atlantic Common Stock;
- (7) Compared the results of operations of Delmarva and Atlantic with that of certain companies which we deemed to be reasonably similar to Delmarva and Atlantic, respectively;
- (8) Compared the proposed financial terms of the Mergers with the financial terms of certain other mergers and acquisitions which we deemed to be relevant;
- (9) Considered the pro forma effect of the Mergers, in terms of net income available to common shareholders, dividends per common share, book value per common share and capitalization, on the Delmarva Common Stock;
- (10) Reviewed the Agreement; and
- (11) Reviewed such other financial studies and analyses and performed such other investigations and took into account such other matters as we deemed necessary or appropriate.

In preparing our opinion, we have relied on the accuracy and completeness of all information supplied or otherwise made available to us by Delmarva and Atlantic, and we have not independently verified such information, conducted a physical inspection of the properties or facilities of Delmarva or Atlantic, or undertaken an evaluation or independent appraisal of the assets or liabilities (contingent or otherwise) of Delmarva or Atlantic. We have assumed that the financial forecasts and projected synergies furnished by Delmarva and Atlantic have been reasonably prepared and reflect the best currently available estimates and judgment of Delmarva's and Atlantic's management as to the expected future financial performance of Delmarva and Atlantic, respectively, and as to the expected future projected outcomes of various legal, regulatory and other contingencies. We have also assumed that the estimates relating to the impact of Atlantic's non-utility generation contracts, furnished by Delmarva and Atlantic, have been reasonably prepared and reflect the best currently available estimates and judgment of Delmarva's and Atlantic's management as to the expected future projected outcomes relating to such contracts. We have also assumed that the Mergers will be free of Federal tax to Delmarva, the Company, DS Sub, Atlantic and the respective holders of Delmarva Common Stock and Atlantic Common Stock. Our opinion is based upon general economic, market, monetary and other conditions as they exist and can be evaluated, and the information made available to us, as of the date hereof.

We have, in the past, provided financial advisory and financing services to Delmarva and Atlantic and have received fees for the rendering of such services. In addition, in the ordinary course of our securities business, we may actively trade debt and equity securities of Delmarva and Atlantic for our own account and the accounts of our customers, and we therefore may from time to time hold a long or short position in such securities.

This opinion has been prepared for the confidential use of Delmarva's Board of Directors and may not be reproduced, summarized, described or referred to without Merrill Lynch's prior written consent.

On the basis of, and subject to the foregoing, we are of the opinion that the Delmarva Conversion Ratio contemplated by the Agreement is fair to Delmarva and its shareholders from a financial point of view.

Very truly yours,

MERRILL LYNCH, PIERCE, FENNER & SMITH INCORPORATED

 $i^{(1)}$ 

#### ANNEX III

# MORGAN STANLEY

MORGAN STANLEY & CO. INCORPORATED 1585 BROADWAY NEW YORK, NEW YORK 10036 (212) 761-4000

December 26, 1996

Board of Directors Atlantic Energy, Inc. 6801 Black Horse Pike Egg Harbor Township, NJ 08232-4130

Members of the Board of Directors:

We understand that Atlantic Energy, Inc. ("AEI" or the "Company"), Delmarva Power & Light Company ("Delmarva"), Conectiv, Inc., a corporation whose outstanding capital stock is 50% owned by AEI and 50% owned by Delmarva ("Newco"), and Newco Sub, a wholly owned subsidiary of Newco ("Newco Sub") have entered into an Agreement and Plan of Merger, dated as of August 9, 1996, as amended and restated as of December 26, 1996 (the "Merger Agreement"), which provides, among other things, for the merger of AEI with and into Newco (the "AEI Merger") and the merger of Newco Sub with and into Delmarva (the "Delmarva Merger" and together with the AEI Merger, the "Merger"). Pursuant to the AEI Merger, Newco will be the surviving corporation and each issued and outstanding share of common stock, no par value per share, of the Company (the "Company Common Stock"), other than shares held in treasury or held by Delmarva or the subsidiaries of the Company or Delmarva, will be converted into the right to receive (i) .75 shares of common stock, par value \$.01 per share, of Newco (the "Newco Common Stock") and (ii) .125 shares of Class A Common Stock, par value \$.01 per share (the "Letter Stock"), of Newco (collectively, the "AEI Conversion Ratio"). Pursuant to the Delmarva Merger, Delmarva will be the surviving corporation and will become a wholly owned subsidiary of Newco and each issued and outstanding share of common stock, par value \$2.25 per share, of Delmarva (the "Delmarva Common Stock"), other than shares held in treasury or held by the Company or the subsidiaries of Delmarva or the Company will be converted into the right to receive one share (the "Delmarva Conversion Ratio") of Newco Common Stock. The terms and conditions of the Merger are more fully set forth in the Merger Agreement.

You have asked for our opinion as to whether the AEI Conversion Ratio, taking into account the Delmarva Conversion Ratio, is fair from a financial point of view to the holders of shares of the Company Common Stock.

For purposes of the opinion set forth herein, we have:

(i) analyzed certain publicly available financial statements and other information of the Company and Delmarva, respectively;



- (ii) analyzed certain internal financial statements and other financial and operating data concerning the Company and Delmarva prepared by their respective managements;
- (iii) analyzed certain financial projections of the Company and Delmarva prepared by their respective managements;
- (iv) discussed the past and current operations and financial condition and the prospects of the Company and Delmarva with senior executives of the Company and Delmarva, respectively;
- (v) reviewed the reported prices and trading activity of both the Company Common Stock and Delmarva Common Stock;
- (vi) compared the financial performance of the Company and Delmarva and the prices and trading activity of the Company Common Stock and Delmarva Common Stock with that of certain other comparable publicly traded companies and their securities;
- (vii) reviewed the financial terms, to the extent publicly available, of certain comparable merger or acquisition transactions;
- (viii) analyzed the pro forma financial impact of the Merger on the Company;
  - (ix) participated in discussions and negotiations among representatives of the Company and Delmarva and their respective financial and legal advisors;
  - (x) reviewed the Merger Agreement and certain related documents;
  - (xi) reviewed and discussed with the Company and Delmarva an analysis prepared by the Company and Delmarva with the assistance of a third party consultant regarding estimates of the amount and timing of the cost savings estimated to be derived from the Merger; and
- (xii) performed such other analyses and examinations and considered such other factors as we have deemed appropriate.

We have assumed and relied upon without independent verification the accuracy and completeness of the information reviewed by us for the purposes of this opinion. With respect to the financial projections and estimates of the cost savings expected to be derived in the Merger, we have assumed that they have been reasonably prepared on bases reflecting the best currently available estimates and judgments of the future financial performance of Newco, the Company and Delmarva, respectively. We have not made any independent valuation or appraisal of the assets or liabilities of the Company or Delmarva. In addition, we have assumed that the Merger will be consummated in accordance with the terms set forth in the Merger Agreement, including, among other things that the Merger will be treated as a tax-free reorganization and/ or exchange, each pursuant to the Internal Revenue Code of 1986. Our opinion is necessarily based on economic, market and other conditions as in effect on, and the information made available to us as of, the date hereof.

In arriving at our opinion, we have assumed that in connection with the receipt of all the necessary regulatory and governmental approvals for the proposed Merger, no restriction will be imposed that would have a material adverse effect on the contemplated benefits expected to be derived in the proposed Merger. In addition, we were not authorized to solicit, and did not solicit, interest from any party with respect to a merger with or other business combination transaction involving the Company or any of its assets, nor did we have discussions or negotiations with any parties, other than Delmarva, in connection with the Merger.

We have acted as financial advisor to the Board of Directors of the Company in connection with this transaction and will receive a fee for our services. In the past, Morgan Stanley & Co. Incorporated and its affiliates have provided financial advisory and financing services for the Company and Delmarva and have received fees for the rendering of these services.

It is understood that this letter is for the information of the Board of Directors of the Company and may not be used for any other purpose without our prior written consent, except that this opinion may be included in its entirety in any filing made by the Company with the Securities and Exchange Commission with respect to the Merger and the transactions related thereto. In addition, we express no recommendation as to how the shareholders of the Company should vote at the shareholders' meeting held in connection with the Merger.

Based on the foregoing, we are of the opinion on the date hereof that the AEI Conversion Ratio, taking into account the Delmarva Conversion Ratio, is fair from a financial point of view to the holders of shares of the Company Common Stock.

Very truly yours,

## MORGAN STANLEY & CO. INCORPORATED

By: <u>/s/ Jeffrey R. Holzschuh</u> Jeffrey R. Holzschuh Managing Director

### FORM OF

## **RESTATED CERTIFICATE OF INCORPORATION OF CONECTIV**

#### Pursuant to Sections 242 and 245 of the

# General Corporation Law of the State of Delaware

Conectiv, a corporation organized and existing under the laws of the State of Delaware, hereby certifies as follows:

1. The name of the corporation is Conectiv (the "Corporation"). The Corporation was originally incorporated under the name DS, Inc., which name was changed to "Conectiv, Inc." on December 24, 1996 and to "Conectiv" on . The original Certificate of Incorporation was filed with the Secretary of State of the State of Delaware on August 8, 1996.

2. This Restated Certificate of Incorporation restates and further amends the Certificate of Incorporation of the Corporation and has been adopted and approved in accordance with Sections 242 and 245 of the General Corporation Laws of the State of Delaware.

3. The text of the Certificate of Incorporation as heretofore amended is hereby amended and restated to read in its entirety as follows:

#### ARTICLE I.

#### NAME

The name of the Corporation is Conectiv.

#### ARTICLE II.

#### **REGISTERED OFFICE AND AGENT**

The address of the registered office of the Corporation is 800 King Street, Wilmington, New Castle County, Delaware 19899, and the name of the registered agent at such office is the Corporation itself.

## **ARTICLE III.**

#### PURPOSE

The Corporation is organized for the purpose of engaging in any lawful act or activity for which corporations may be organized under the General Corporation Law of the State of Delaware (the "GCLD").

#### ARTICLE IV.

# **CAPITAL STOCK**

SECTION I. Authorization. The aggregate number of shares of stock which the Corporation shall have authority to issue is one hundred eighty million (180,000,000) shares, of which one hundred fifty million (150,000,000) shares shall be shares of a class of common stock designated as "Common Stock," having a par value of \$0.01 per share (the "Company Common Stock"), ten million (10,000,000) shares shall be shares of a class of common Stock", having a par value of \$0.01 per share (the "Company Common Stock"), ten million (10,000,000) shares shall be shares of a class of common stock designated as "Class A Common Stock," having a par value of \$0.01 per share (the "Class A Common Stock"), and twenty million (20,000,000) shares shall be shares of a class of preferred stock having a par value of \$.01 per share (the "Preferred Stock") and issuable in one or more series as hereinafter provided. The Company Common Stock and the Class A Common Stock shall hereinafter collectively be called "Common Stock" and either shall sometimes be called a class of Common Stock. Certain capitalized terms used in this Article IV shall have the meanings set forth in Section II.7 of this Article IV. For purposes of this Article IV, the Class A Common Stock, when issued,

shall be considered issued in respect of the Atlantic Utility Group and the Company Common Stock, when issued, shall be considered issued in respect of the Residual Group, in each case upon the terms and subject to the conditions of this Article IV.

SECTION II. *Common Stock.* The voting powers, preferences and relative, participating, optional or other special rights of the Common Stock, and the qualifications and restrictions thereon, shall be as follows in this Section II.

1. *Dividends.* Subject to any preferences and relative, participating, optional or special rights of any outstanding series of Preferred Stock and any qualifications or restrictions on the Common Stock or any class thereof created thereby, dividends may be declared and paid upon the Company Common Stock and the Class A Common Stock, upon the terms with respect to each such class, and subject to the limitations provided for below in this subsection 1, as the Board of Directors may determine.

1.1. *Limitation on Dividends on Company Common Stock.* Dividends on Company Common Stock may be declared and paid only out of the lesser of (i) the funds of the Corporation legally available therefor and (ii) the Residual Group Available Dividend Amount.

1.2. Limitation on Dividends on Class A Common Stock. Dividends on Class A Common Stock may be declared and paid only out of the lesser of (i) the funds of the Corporation legally available therefor and (ii) the Atlantic Utility Group Available Dividend Amount.

1.3. Discrimination in Dividends Between Classes of Common Stock. The Board of Directors, subject to the provisions of subsections 1.1 and 1.2 of this Section II, may at any time declare and pay dividends exclusively on Company Common Stock, exclusively on Class A Common Stock or on both such classes in equal or unequal amounts, notwithstanding the relative amounts of the Residual Group Available Dividend Amount and the Atlantic Utility Group Available Dividend Amount, the amount of dividends previously declared on each class, the respective voting or liquidation rights of each class or any other factor.

1.4. Share Distributions. Subject to subsections 1.1 and 1.2, as the case may be, of this Section II, and except as permitted by subsection 4.1 of this Section II, the Board of Directors may declare and pay dividends or distributions of shares of Common Stock (or Convertible Securities convertible into or exchangeable or exercisable for shares of Common Stock) on shares of Common Stock or shares of Preferred Stock only as follows:

(A) dividends or distributions of shares of Company Common Stock (or Convertible Securities convertible into or exchangeable or exercisable for shares of Residual Common Stock) on shares of Company Common Stock or shares of Preferred Stock attributed to the Residual Group;

(B) dividends or distributions of shares of Class A Common Stock (or Convertible Securities convertible into or exchangeable or exercisable for shares of Class A Common Stock) on shares of Class A Common Stock or shares of Preferred Stock attributed to the Atlantic Utility Group; and

(C) dividends or distributions of shares of Class A Common Stock (or Convertible Securities convertible into or exchangeable or exercisable for shares of Class A Common Stock) on shares of Company Common Stock or shares of Preferred Stock attributed to the Residual Group, but only if the sum of (1) the number of shares of Class A Common Stock to be so issued (or the number of such shares which would be issuable upon conversion, exchange or exercise of any Convertible Securities to be so issued, as the case may be) and (2) the number of shares of Class A Common Stock which are issuable upon conversion, exchange or exercise of any Convertible Securities then outstanding that are attributed in accordance with this Article IV to the Residual Group is less than or equal to the Number of Shares Issuable with Respect to the Intergroup Interest.

For purposes of this subsection 1.4, any outstanding Convertible Securities that are convertible into or exchangeable or exercisable for any other Convertible Securities which are themselves convertible into or exchangeable or exercisable for Company Common Stock or Class A Common Stock (or other Convertible Securities that are so convertible, exchangeable or exercisable) shall be deemed to have been converted, exchanged or exercised in full for such Convertible Securities.

2. Voting Powers. Except as otherwise provided by law or by the terms of any outstanding series of Preferred Stock or any provision of the certificate of incorporation of the Corporation restricting the power to vote on a specified matter to other stockholders, the entire voting power of the stockholders of the Corporation shall be vested in the holders of Common Stock of the Corporation, who shall be entitled to vote on any matter on which the holders of stock of the Corporation shall, by law or by the provisions of the certificate of incorporation or bylaws of the Corporation, be entitled to vote, and each class of Common Stock shall vote thereon together as though one class. On each matter to be voted on by the holders of all classes of Common Stock voting together as one class, (i) each outstanding share of Company Common Stock shall have one vote and (ii) each outstanding share of Class A Common Stock may be increased or decreased by the affirmative vote of a majority of the outstanding shares of Common Stock voting as a single class.

3. Liquidation Rights. In the event of the voluntary or involuntary dissolution of the Corporation or the liquidation and winding up of the Corporation, after payment or provision for payment of the debts and other liabilities of the Corporation and the full preferential amounts (including any accumulated and unpaid dividends) to which the holders of Preferred Stock are entitled (regardless of the Group to which such shares of Preferred Stock were attributed in accordance with this Article IV), unless otherwise provided in respect of a series of preferred stock by the resolution of the Board of Directors fixing the liquidation rights and preferences of such series of preferred stock, the holders of the outstanding shares of Common Stock shall be entitled to receive the remaining assets of the Corporation, regardless of the Group to which such assets are attributed in accordance with this Article IV, divided among the holders of Common Stock in accordance with the per share "Liquidation Units" attributable to each class of Common Stock. Each share of Company Common Stock is hereby attributed one "Liquidation Unit" and each share of Class A Common Stock is hereby attributed one "Liquidation Unit," in the case of each such class of Common Stock subject to adjustment as determined by the Board of Directors to be appropriate to reflect any subdivision (by stock split or otherwise) or combination (by reverse stock split or otherwise) of such class of Common Stock or any dividend or other distribution of shares of such class of Common Stock to holders of shares of such class of Common Stock. None of the merger or consolidation of the Corporation into or with any other company, the merger or consolidation of any other company into or with the Corporation, or the sale, transfer or lease of all or any part of the assets of the Corporation, shall be deemed a liquidation or winding up of the Corporation, or cause the dissolution of the Corporation, for purposes of this subsection 3.

4. Conversion or Redemption of Common Stock. The Class A Common Stock is subject to conversion or redemption upon the terms provided below in this subsection 4.

4.1. Conversion or Redemption of Class A Common Stock.

(A) In the event of the Disposition, in one transaction or a series of related transactions, by the Corporation and/or its subsidiaries of all or substantially all of the properties and assets attributed to the Atlantic Utility Group to one or more persons or entities (other than (1) the Disposition by the Corporation of its properties and assets in one transaction or a series of related transactions in connection with the dissolution or the liquidation and winding up of the Corporation and the distribution of assets to stockholders as referred to in subsection 3 of this Section II, (2) the Disposition of the properties and assets of the Atlantic Utility Group to all holders of shares of Class A Common Stock and to the Corporation or subsidiaries thereof, divided among such holders and the Corporation or subsidiaries thereof on a *pro rata* basis in accordance with the number of shares of Class A Common Stock outstanding and the Number of Shares Issuable with Respect to the Intergroup Interest, (3) to any person or entity controlled (as determined by the Board of Directors) by the Corporation or (4) pursuant to a Related Business Transaction), the Corporation shall, on or prior to the 85th Trading Day after the date of consummation of such Disposition (the "Atlantic Utility Group Disposition Date"), pay a dividend on the Class A Common Stock or redeem some or all of the Class A Common Stock or convert Class A Common Stock into Company Common Stock

IV-3

(or another class or series of common stock of the Corporation), all as provided by the following subparagraphs (1) and (2) of this paragraph (A) and, to the extent applicable, by subsection 4.3 of this Section II, as the Board of Directors shall have selected among such alternatives:

(1) provided that there are funds of the Corporation legally available therefor:

(a) pay to the holders of the shares of Class A Common Stock a dividend, as the Board of Directors shall have declared subject to compliance with subsection 1, in cash and/or in securities (other than a dividend of Common Stock) or other property having a Fair Value as of the Atlantic Utility Group Disposition Date in the aggregate equal to the product of the Outstanding Atlantic Utility Fraction as of the record date for determining holders entitled to receive such dividend multiplied by the Fair Value of the Net Proceeds of such Disposition; or

(b) (i) subject to the last sentence of this paragraph (A), if such Disposition involves all (not merely substantially all) of the properties and assets attributed to the Atlantic Utility Group, redeem as of the Redemption Date provided by paragraph (C) of subsection 4.3 of this Section II, all outstanding shares of Class A Common Stock in exchange for cash and/or for securities (other than Common Stock) or other property having a Fair Value as of the Atlantic Utility Group Disposition Date in the aggregate equal to the product of the Outstanding Atlantic Utility Fraction as of such Redemption Date multiplied by the Fair Value of the Net Proceeds of such Disposition; or

(ii) subject to the last sentence of this paragraph (A), if such Disposition involves substantially all (but not all) of the properties and assets attributed to the Atlantic Utility Group, redeem as of the Redemption Date provided by paragraph (D) of subsection 4.3 of this Section II such number of whole shares of Class A Common Stock (which may be all of such shares outstanding) as have in the aggregate an average Market Value during the period of ten consecutive Trading Days beginning on the sixteenth Trading Day immediately succeeding the Atlantic Utility Group Disposition Date closest to the product of the Outstanding Atlantic Utility Fraction as of the date such shares are selected for redemption multiplied by the Fair Value as of the Atlantic Utility Group Disposition Date of the Net Proceeds of such Disposition (but in no event more than all the shares of Class A Common Stock then outstanding), in consideration for cash and/or securities (other than Common Stock) or other property having a Fair Value in the aggregate equal to such product; or

(2) declare that each outstanding share of Class A Common Stock shall be converted as of the Conversion Date provided by paragraph (E) of subsection 4.3 of this Section II into a number of fully paid and nonassessable shares of Company Common Stock (or, if the Company Common Stock is not Publicly Traded at such time and shares of another class or series of common stock of the Corporation (other than Class A Common Stock) are then Publicly Traded, of such other class or series of common stock as has the largest Market Capitalization as of the close of business on the Trading Day immediately preceding the date of the notice of such conversion required by such paragraph (E) equal to 110% of the ratio, expressed as a decimal fraction rounded to the nearest five decimal places, of the average Market Value of one share of Class A Common Stock over the period of ten consecutive Trading Days beginning on the sixteenth Trading Day following the Atlantic Utility Group Disposition Date to the average Market Value of one share of Company Common Stock (or such other class or series of common stock) over the same ten Trading Day period.

Notwithstanding the foregoing provisions of this paragraph (A), the Corporation shall redeem Class A Common Stock as provided by subparagraph (1)(b)(i) or (1)(b)(ii) of this paragraph (A) only if the amount to be paid in redemption of such stock is less than or equal to the sum of (i) the Atlantic Utility Group Available Dividend Amount as of the Redemption Date and (ii) the amount determined to be capital in respect of the shares to be redeemed in accordance with applicable corporation law as of the Redemption Date.

# (B) For purposes of this subsection 4.1:

(1) as of any date, "substantially all of the properties and assets" attributed to the Atlantic Utility Group shall mean a portion of such properties and assets (x) that represents at least 80% of the Fair Value of the properties and assets attributed to the Atlantic Utility Group as of such date, or (y) from which were derived at least 80% of the aggregate revenues for the immediately preceding twelve fiscal quarterly periods of the Corporation (calculated on a pro forma basis to include revenues derived from any of such properties and assets acquired during such period) derived from the properties and assets of the Atlantic Utility Group as of such date;

(2) in the case of a Disposition of the properties and assets attributed to the Atlantic Utility Group in a series of related transactions, such Disposition shall not be deemed to have been consummated until the consummation of the last of such transactions; and

(3) the Board of Directors may pay any dividend or redemption price referred to in paragraph (A) of this subsection 4.1 in cash, securities (other than Common Stock) or other property, regardless of the form or nature of the proceeds of the Disposition.

(C) After the payment of the dividend or the redemption price with respect to the Class A Common Stock provided for by subparagraph (1) of paragraph (A) of this subsection 4.1, the Board of Directors may declare that each share of Class A Common Stock remaining outstanding shall be converted, but only as of a Conversion Date (determined as provided by paragraph (E) of subsection 4.3 of this Section II) prior to the first anniversary of the payment of such dividend or redemption price, into a number of fully paid and nonassessable shares of Company Common Stock (or, if the Company Common Stock is not Publicly Traded at such time and shares of any other class or series of common stock of the Corporation (other than Class A Common Stock) are then Publicly Traded, of such other class or series of common stock as has the largest Market Capitalization as of the close of business on the Trading Day immediately preceding the date of the notice of such conversion required by such paragraph (E).

(D) The Board of Directors may declare that each outstanding share of Class A Common Stock shall be converted, as of the Conversion Date provided by paragraph (E) of subsection 4.3 of this Section II, into the number of fully paid and nonassessable shares of Company Common Stock (or, if the Company Common Stock is not Publicly Traded at such time and shares of any other class or series of common stock of the Corporation (other than Class A Common Stock) are then Publicly Traded, of such other class or series of common stock as has the largest Market Capitalization as of the close of business on the Trading Day immediately preceding the date of the notice of conversion required by such paragraph (E) of subsection 4.3) equal to the applicable percentage, on the Conversion Date, set forth below of the Market Value Ratio of the Class A Common Stock to the Company Common Stock as of the fifth Trading Day prior to the date of the notice of such conversion required by such paragraph (E):

12 Month Period Prior to the Applicable Anniversary of the Effective Date	Percentage of Market Value Ratio	0'
First	125%	
Second		
Third	115%	
Fourth and Thereafter	110%	

(E) If the Corporation consummates (i) a tender offer made by the Corporation for all of the outstanding shares of Class A Common Stock at an all cash price of at least 110% of the Time-Weighted Market Price of a share of Class A Common Stock as of the Trading Day immediately preceding the date of such offer or (ii) an exchange offer by the Corporation to exchange each outstanding share of Class A Common Stock into a number of shares of Company Common Stock (or,

if the Company Common Stock is not Publicly Traded at such time and shares of any other class or series of common stock of the Corporation (other than Class A Common Stock) are then Publicly Traded, of such other class or series of common stock as has the largest Market Capitalization as of the close of business on the Trading Day immediately preceding the date of such offer) equal to at least 110% of the Market Value Ratio of the Class A Common Stock to the Company Common Stock as of the Trading Day immediately preceding the date of such offer, which, in either case, is accepted by the holders of greater than 50% of the outstanding shares of Class A Common Stock, then the Board of Directors may either (x) provided that there are funds of the Corporation legally available therefor, redeem as of the Redemption Date provided by paragraph (F) of subsection 4.3 of this Section II each share of Class A Common Stock remaining outstanding in exchange for cash in an amount equal to the highest cash price paid per share by the Corporation pursuant to such tender offer or to the product of the highest number of shares of Company Common Stock (or such other class or series of common stock of the Corporation) per share issued in exchange for any share of Class A Common Stock pursuant to such exchange offer and the Time-Weighted Market Price of a share of Company Common Stock (or such other class or series of common stock of the Corporation) as of the Trading Day immediately preceding the date of such exchange offer, as the case may be, or (y) declare that each share of Class A Common Stock remaining outstanding shall be converted as of the Conversion Date provided by paragraph (E) of subsection 4.3 of this Section II into a number of fully paid and nonassessable shares of Company Common Stock (or, if the Company Common Stock is not Publicly Traded at such time and shares of any other class or series of common stock of the Corporation (other than Class A Common Stock) are then Publicly Traded, of such other class or series of common stock as has the largest Market Capitalization as of the close of business on the Trading Day immediately preceding the date of the notice of such conversion required by such paragraph (E) equal to the quotient of the highest cash price paid per share by the Corporation pursuant to such tender offer and the Time-Weighted Market Price of a share of Company Common Stock (or such other class or series of common stock of the Corporation) as of the Trading Day immediately preceding the date of the notice of such conversion required by such paragraph (E) or to the highest number of shares of Company Common Stock (or such other class or series of common stock of the Corporation) per share issued in exchange for any share of Class A Common Stock pursuant to such exchange offer, as the case may be.

(F) If any person (including the Corporation) makes a tender offer to purchase shares of Company Common Stock (or, if the Company Common Stock is not Publicly Traded at such time and shares of any other class or series of common stock of the Corporation (other than Class A Common Stock) are then Publicly Traded, of such other class or series of common stock as has the largest Market Capitalization as of the close of business on the Trading Day immediately preceding the date of such offer) for cash, property or other securities, the holders of shares of Class A Common Stock will be entitled to convert each and any such share of Class A Common Stock into the number of fully paid and nonassessable shares of Company Common Stock (or such other class or series of common stock, as the case may be) equal to 100% of the Market Value Ratio of the Class A Common Stock to the Company Common Stock as of the Trading Day immediately preceding the date of such tender offer; provided, that (x) any election by such holder to make such conversion may be fully revoked by such holder with respect to any such share of Class A Common Stock by giving written notice to the Corporation prior to the consummation of such tender offer and (y) such conversion will only be effective with respect to such shares of Company Common Stock (or such other class or series of common stock of the Corporation) issuable upon such conversion which are actually accepted for purchase pursuant to such tender offer.

(G) If any person (including the Corporation) consummates a tender offer for all of the outstanding shares of Company Common Stock (or, if the Company Common Stock is not Publicly Traded at such time and shares of any other class or series of common stock of the Corporation (other than Class A Common Stock) are then Publicly Traded, of such other class or series of common stock as has the largest Market Capitalization as of the close of business on the Trading Day immediately

preceding the date of such offer) at an all cash price which is accepted by the holders of greater than 50% of the outstanding shares of Company Common Stock (or such other class or series of common stock of the Corporation), then the Board of Directors may either (x) redeem as of the Redemption Date provided by paragraph (F) of subsection 4.3 of this Section II each share of Class A Common Stock outstanding in exchange for cash in an amount equal to the product of the highest cash price paid per share by such person pursuant to such tender offer and the Market Value Ratio of the Class A Common Stock to the Company Common Stock as of the fifth Trading Day prior to the date of such tender offer or (y) declare that each share of Class A Common Stock outstanding shall be converted as of the Conversion Date provided by paragraph (E) of subsection 4.3 of this Section II into a number of fully paid and nonassessable shares of Company Common Stock (or, if the Company Common Stock is not Publicly Traded at such time and shares of any other class or series of common stock of the Corporation (other than Class A Common Stock) are then Publicly Traded, of such other class or series of common stock as has the largest Market Capitalization as of the close of business on the Trading Day immediately preceding the date of the notice of such conversion required by such paragraph (E) equal to the quotient of the highest cash price paid per share by such person pursuant to such tender offer and the Time-Weighted Market Price of a share of Company Common Stock (or such other class or series of common stock of the Corporation) as of the Trading Day immediately preceding the date of the notice of such conversion required by such paragraph (E).

4.2. Treatment of Convertible Securities. After any Conversion Date or Redemption Date on which all outstanding shares of Class A Common Stock were converted or redeemed, any share of Class A Common Stock that is to be issued on conversion, exchange or exercise of any Convertible Securities shall, immediately upon such conversion, exchange or exercise and without any notice from or to, or any other action on the part of, the Corporation or its Board of Directors or the holder of such Convertible Security:

(A) in the event the shares of Class A Common Stock outstanding on such Conversion Date were converted into shares of Company Common Stock (or another class or series of common stock of the Corporation) pursuant to subparagraph (A)(2) or paragraph (C), (D), (E), (F) or (G) of subsection 4.1 of this Section II, be converted into the amount of cash and/or the number of shares of the kind of capital stock and/or other securities or property of the Corporation that the number of shares of Class A Common Stock that were to be issued upon such conversion, exchange or exercise would have received had such shares been outstanding on such Conversion Date; or

(B) in the event the shares of Class A Common Stock outstanding on such Redemption Date were redeemed pursuant to subparagraph (A)(1)(b) of subsection 4.1 of this Section II, be redeemed, to the extent of funds of the Corporation legally available therefor, for \$.01 per share in cash for each share of Class A Common Stock that otherwise would be issued upon such conversion, exchange or exercise.

The provisions of the immediately preceding sentence shall not apply to the extent that other adjustments in respect of such conversion, exchange or redemption of Class A Common Stock are otherwise made pursuant to the provisions of such Convertible Securities.

4.3. Notice and Other Provisions.

(A) Not later than the tenth Trading Day following the consummation of a Disposition referred to in paragraph (A) of subsection 4.1 of this Section II, the Corporation shall announce publicly by press release (1) the Net Proceeds of such Disposition, (2) the number of shares outstanding of the Class A Common Stock, (3) the number of shares of Class A Common Stock into or for which Convertible Securities are then convertible, exchangeable or exercisable and the conversion, exchange or exercise price thereof and (4) the Outstanding Atlantic Utility Fraction on the date of such notice. Not earlier than the 26th Trading Day and not later than the 30th Trading Day following the consummation of such Disposition, the Corporation shall announce publicly by press release which of the actions specified in paragraph (A) of such subsection 4.1 it has irrevocably determined to take in respect of such Disposition.

(B) If the Corporation determines to pay a dividend on shares of Class A Common Stock pursuant to subparagraph (A)(1)(a) of subsection 4.1 of this Section II, the Corporation shall, not later than the 30th Trading Day following the consummation of the Disposition referred to in such subparagraph, cause notice to be given to each holder of shares of Class A Common Stock and to each holder of Convertible Securities that are convertible into or exchangeable or exercisable for shares of Class A Common Stock (unless alternate provision for such notice to the holders of such Convertible Securities is made pursuant to the terms of such Convertible Securities), setting forth (1) the record date for determining holders entitled to receive such dividend, which shall be not earlier than the 40th Trading Day and not later than the 50th Trading Day following the consummation of such Disposition. (2) the anticipated payment date of such dividend (which shall not be more than 85 Trading Days following the consummation of such Disposition), (3) the type of property to be paid as such dividend in respect of the outstanding shares of Class A Common Stock, (4) the Net Proceeds of such Disposition, (5) the Outstanding Atlantic Utility Fraction on the date of such notice, (6) the number of outstanding shares of Class A Common Stock and the number of shares of Class A Common Stock into or for which outstanding Convertible Securities are then convertible, exchangeable or exercisable and the conversion, exchange or exercise price thereof and (7) in the case of notice to be given to holders of Convertible Securities, a statement to the effect that a holder of such Convertible Securities shall be entitled to receive such dividend only if such holder properly converts, exchanges or exercises such Convertible Securities on or prior to the record date referred to in clause (1) of this sentence. Such notice shall be sent by first-class mail, postage prepaid, to each such holder at such holder's address as the same appears on the transfer books of the Corporation.

(C) If the Corporation determines to redeem Class A Common Stock pursuant to subparagraph (A)(1)(b)(i) of subsection 4.1 of this Section II, the Corporation shall, not earlier than the 35th Trading Day and not later than the 45th Trading Day prior to the Redemption Date, cause notice to be given to each holder of shares of Class A Common Stock, and to each holder of Convertible Securities convertible into or exchangeable or exercisable for shares of Class A Common Stock (unless alternate provision for such notice to the holders of such Convertible Securities is made pursuant to the terms of such Convertible Securities), setting forth (1) a statement that all shares of Class A Common Stock outstanding on the Redemption Date shall be redeemed, (2) the Redemption Date (which shall not be more than 85 Trading Days following the consummation of such Disposition), (3) the type of property in which the redemption price for the shares to be redeemed is to be paid, (4) the Net Proceeds of such Disposition, (5) the Outstanding Atlantic Utility Fraction on the date of such notice, (6) the place or places where certificates for shares of Class A Common Stock, properly endorsed or assigned for transfer (unless the Corporation waives such requirement), are to be surrendered for delivery of cash and/or securities or other property, (7) the number of outstanding shares of Class A Common Stock and the number of shares of Class A Common Stock into or for which such outstanding Convertible Securities are then convertible, exchangeable or exercisable and the conversion, exchange or exercise price thereof, (8) in the case of notice to be given to holders of Convertible Securities, a statement to the effect that a holder of such Convertible Securities shall be entitled to participate in such selection for redemption only if such holder properly converts, exchanges or exercises such Convertible Securities on or prior to the Redemption Date referred to in clause (2) of this sentence and a statement as to what, if anything, such holder will be entitled to receive pursuant to the terms of such Convertible Securities or, if applicable, this subsection 4 if such holder thereafter converts, exchanges or exercises such Convertible Securities and (9) a statement to the effect that, except as otherwise provided by paragraph (I) of this subsection 4.3, dividends on such shares of Class A Common Stock shall cease to be paid as of such Redemption Date. Such notice shall be sent by firstclass mail, postage prepaid, to each such holder at such holder's address as the same appears on the transfer books of the Corporation.

(D) If the Corporation determines to redeem Class A Common Stock pursuant to subparagraph (A)(1)(b)(ii) of subsection 4.1 of this Section II, the Corporation shall, not later than the 30th Trading Day following the consummation of the Disposition referred to in such subparagraph, cause notice to



be given to each holder of shares of Class A Common Stock and to each holder of Convertible Securities that are convertible into or exchangeable or exercisable for shares of Class A Common Stock (unless alternate provision for such notice to the holders of such Convertible Securities is made pursuant to the terms of such Convertible Securities) setting forth (1) a date not earlier than the 40th Trading Day and not later than the 50th Trading Day following the consummation of the Disposition in respect of which such redemption is to be made on which shares of Class A Common Stock shall be selected for redemption, (2) the anticipated Redemption Date (which shall not be more than 85 Trading Days following the consummation of such Disposition), (3) the type of property in which the redemption price for the shares to be redeemed is to be paid, (4) the Net Proceeds of such Disposition, (5) the Outstanding Atlantic Utility Fraction, (6) the number of shares of Class A Common Stock outstanding and the number of shares of Class A Common Stock into or for which outstanding Convertible Securities are then convertible, exchangeable or exercisable and the conversion, exchange or exercise price thereof, (7) in the case of notice to be given to holders of Convertible Securities, a statement to the effect that a holder of such Convertible Securities shall be eligible to participate in such selection for redemption only if such holder properly converts, exchanges or exercises such Convertible Securities on or prior to the record date referred to in clause (1) of this sentence, and a statement as to what, if anything, such holder will be entitled to receive pursuant to the terms of such Convertible Securities or, if applicable, this subsection 4 if such holder thereafter converts, exchanges or exercises such Convertible Securities and (8) a statement that the Corporation will not be required to register a transfer of any shares of Class A Common Stock for a period of 15 Trading Days next preceding the date referred to in clause (1) of this sentence. Promptly following the date referred to in clause (1) of the preceding sentence, but not earlier than 40 Trading Days nor later than 50 Trading Days following the consummation of such Disposition, the Corporation shall cause a notice to be given to each holder of record of shares of Class A Common Stock to be redeemed setting forth (1) the number of shares of Class A Common Stock held by such holder to be redeemed, (2) a statement that Class A Common Stock shall be redeemed, (3) the Redemption Date, (4) the kind and per share amount of cash and/or securities or other property to be received by such holder with respect to each share of Class A Common Stock to be redeemed, including details as to the calculation thereof, (5) the place or places where certificates for shares of Class A Common Stock, properly endorsed or assigned for transfer (unless the Corporation shall waive such requirement), are to be surrendered for delivery of such cash and/or securities or other property, (6) if applicable, a statement to the effect that the shares being redeemed may no longer be transferred on the transfer books of the Corporation after the Redemption Date and (7) a statement to the effect that, except as otherwise provided by paragraph (I) of this subsection 4.3, dividends on such shares of Class A Common Stock shall cease to be paid as of the Redemption Date. Such notices shall be sent by firstclass mail, postage prepaid, to each such holder at such holder's address as the same appears on the transfer books of the Corporation.

(E) If the Corporation determines to convert the Class A Common Stock into Company Common Stock (or another class or series of common stock of the Corporation) pursuant to subparagraph (A)(2) or paragraph (C), (D), (E) or (G) of subsection 4.1 of this Section II, the Corporation shall, not earlier than the 35th Trading Day and not later than the 45th Trading Day prior to the Conversion Date, cause notice to be given to each holder of shares of Class A Common Stock and to each holder of Convertible Securities that are convertible into or exchangeable or exercisable for shares of Class A Common Stock (unless alternate provision for such notice to the holders of such Convertible Securities is made pursuant to the terms of such Convertible Securities) setting forth (1) a statement that all outstanding shares of Class A Common Stock shall be converted, (2) the Conversion Date (which, in the case of a conversion after a Disposition, shall not be more than 85 Trading Days following the consummation of such Disposition and, in the case of a conversion after a tender or exchange offer pursuant to paragraph (E) or (G) of subsection 4.1 of this Section II, shall not be less than 35 or more than 85 Trading Days following the consummation of such offer), (3) the per share number of shares of Class A Common Stock or another class or series of common stock of the

Corporation, as the case may be, to be received with respect to each share of Class A Common Stock, including details as to the calculation thereof, (4) the place or places where certificates for shares of Class A Common Stock, properly endorsed or assigned for transfer (unless the Corporation shall waive such requirement), are to be surrendered for delivery of certificates for shares of Class A Common Stock, (5) the number of outstanding shares of Class A Common Stock and the number of shares of Class A Common Stock into or for which outstanding Convertible Securities are then convertible, exchangeable or exercisable and the conversion, exchange or exercise price thereof, (6) a statement to the effect that, except as otherwise provided by paragraph (I) of this subsection 4.3, dividends on such shares of Class A Common Stock shall cease to be paid as of such Conversion Date and (7) in the case of notice to holders of such Convertible Securities, a statement to the effect that a holder of such Convertible Securities shall be entitled to receive shares of common stock upon such conversion only if such holder properly converts, exchanges or exercises such Convertible Securities on or prior to such Conversion Date and a statement as to what, if anything, such holder will be entitled to receive pursuant to the terms of such Convertible Securities or, if applicable, this subsection 4 if such holder thereafter converts, exchanges or exercises such Convertible Securities. Such notice shall be sent by first-class mail, postage prepaid, to each such holder at such holder's address as the same appears on the transfer books of the Corporation.

(F) If the Corporation determines to redeem Class A Common Stock pursuant to subparagraph (E) or (G) of subsection 4.1 of this Section II, the Corporation shall, not earlier than the 35th Trading Day and not later than the 45th Trading Day prior to the Redemption Date, cause notice to be given to each holder of shares of Class A Common Stock, and to each holder of Convertible Securities convertible into or exchangeable or exercisable for shares of Class A Common Stock (unless alternate provision for such notice to the holders of such Convertible Securities is made pursuant to the terms of such Convertible Securities), setting forth (1) a statement that all shares of Class A Common Stock outstanding on the Redemption Date shall be redeemed, (2) the Redemption Date (which shall not be less than 35 or more than 85 Trading Days following the consummation of the applicable tender or exchange offer), (3) the redemption price for the shares, (4) the place or places where certificates for shares of Class A Common Stock, properly endorsed or assigned for transfer (unless the Corporation waives such requirement), are to be surrendered for delivery of cash, (5) the number of outstanding shares of Class A Common Stock and the number of shares of Class A Common Stock into or for which such outstanding Convertible Securities are then convertible, exchangeable or exercisable and the conversion, exchange or exercise price thereof, (6) in the case of notice to be given to holders of Convertible Securities, a statement to the effect that a holder of such Convertible Securities shall be entitled to participate in such selection for redemption only if such holder properly converts, exchanges or exercises such Convertible Securities on or prior to the Redemption Date referred to in clause (2) of this sentence and a statement as to what, if anything, such holder will be entitled to receive pursuant to the terms of such Convertible Securities or, if applicable, this subsection 4 if such holder thereafter converts, exchanges or exercises such Convertible Securities and (7) a statement to the effect that, except as otherwise provided by paragraph (I) of this subsection 4.3, dividends on such shares of Class A Common Stock shall cease to be paid as of such Redemption Date. Such notice shall be sent by first-class mail, postage prepaid, to each such holder at such holder's address as the same appears on the transfer books of the Corporation.

(G) If less than all of the outstanding shares of Class A Common Stock are to be redeemed pursuant to subparagraph (A)(1) of subsection 4.1 of this Section II, the shares to be redeemed by the Corporation shall be selected from among the holders of shares of Class A Common Stock outstanding at the close of business on the record date for such redemption on a pro rata basis among all such holders or by lot or by such other method as may be determined by the Board of Directors to be equitable.

(H) The Corporation shall not be required to issue or deliver fractional shares of any capital stock or of any other securities to any holder of Class A Common Stock upon any conversion, redemption, dividend or other distribution pursuant to this subsection 4. If more than one share of Class A Common Stock shall be held at the same time by the same holder, the Corporation may aggregate the number of shares of any capital stock that shall be issuable or any other securities or property that shall be distributable to such holder upon any conversion, redemption, dividend or other distribution (including any fractional shares). If there are fractional shares of any capital stock or of any other securities remaining to be issued or distributed to the holders of Class A Common Stock, the Corporation shall, if such fractional shares are not issued or distributed to the holder, pay cash in respect of such fractional shares in an amount equal to the Fair Value thereof on the fifth Trading Day prior to the date such payment is to be made (without interest).

(I) No adjustments in respect of dividends shall be made upon the conversion or redemption of any shares of Class A Common Stock; provided, however, that if the Conversion Date or Redemption Date, as the case may be, with respect to any shares of Class A Common Stock shall be subsequent to the record date for the payment of a dividend or other distribution thereon or with respect thereto, the holders of Class A Common Stock at the close of business on such record date shall be entitled to receive the dividend or other distribution payable on or with respect to such shares on the date set for payment of such dividend or other distribution, in each case without interest, notwithstanding the subsequent conversion or redemption of such shares.

(J) Before any holder of Class A Common Stock shall be entitled to receive any cash payment and/or certificates or instruments representing shares of any capital stock and/or other securities or property to be distributed to such holder with respect to Class A Common Stock pursuant to this subsection 4, such holder shall surrender at such place as the Corporation shall specify certificates for Class A Common Stock, properly endorsed or assigned for transfer (unless the Corporation shall waive such requirement). The Corporation shall as soon as practicable after receipt of certificates representing shares of Class A Common Stock deliver to the person for whose account such shares were so surrendered, or to such person's nominee or nominees, the cash and/or the certificates or instruments representing the number of whole shares of the kind of capital stock and/or other securities or property to which such person shall be entitled as aforesaid, together with any payment in respect of fractional shares contemplated by paragraph (H) of this subsection 4.3, in each case without interest. If less than all of the shares of Class A Common Stock represented by any one certificate are to be redeemed or converted, the Corporation shall issue and deliver a new certificate for the shares of Class A Common Stock not redeemed.

(K) From and after any applicable Conversion Date or Redemption Date, as the case may be, all rights of a holder of shares of Class A Common Stock that were converted or redeemed shall cease except for the right, upon surrender of the certificates representing such shares as required by paragraph (J) of this subsection 4.3, to receive the cash and/or the certificates or instruments representing shares of the kind of capital stock and/or other securities or property for which such shares were converted or redeemed, together with any payment in respect of fractional shares contemplated by paragraph (H) of this subsection 4.3 and rights to dividends as provided in paragraph (I) of this subsection 4.3, in each case without interest. No holder of a certificate that immediately prior to the applicable Conversion Date represented shares of Class A Common Stock shall be entitled to receive any dividend or other distribution or interest payment with respect to shares of any kind of capital stock or other security or instrument for which Class A Common Stock was converted until the surrender as required by this subsection 4 of such certificate in exchange for a certificate or certificates or instrument or instruments representing such capital stock or other security. Upon such surrender, there shall be paid to the holder the amount of any dividends or other distributions (without interest) which theretofore became payable on any class or series of capital stock of the Corporation as of a record date after the Conversion Date, but that were not paid by reason of the

foregoing, with respect to the number of whole shares of the kind of capital stock represented by the certificate or certificates issued upon such surrender. From and after a Conversion Date, the Corporation shall, however, be entitled to treat the certificates for Class A Common Stock that have not yet been surrendered for conversion as evidencing the ownership of the number of whole shares of the kind or kinds of capital stock of the Corporation for which the shares of Class A Common Stock represented by such certificates shall have been converted, notwithstanding the failure to surrender such certificates.

(L) The Corporation shall pay any and all documentary, stamp or similar issue or transfer taxes that may be payable in respect of the issuance or delivery of any shares of capital stock and/or other securities upon conversion or redemption of shares of Class A Common Stock pursuant to this subsection 4. The Corporation shall not, however, be required to pay any tax that may be payable in respect of any transfer involved in the issuance or delivery of any shares of capital stock and/or other securities in a name other than that in which the shares of Class A Common Stock so converted or redeemed were registered, and no such issuance or delivery shall be made unless and until the person requesting such issuance or delivery has paid to the Corporation the amount of any such tax or has established to the satisfaction of the Corporation that such tax has been paid.

(M) Neither the failure to mail any notice required by this subsection 4.3 to any particular holder of Class A Common Stock or of Convertible Securities nor any defect therein shall affect the sufficiency thereof with respect to any other holder of outstanding shares of Class A Common Stock or of Convertible Securities or the validity of any such conversion or redemption.

(N) The Board of Directors may establish such rules and requirements to facilitate the effectuation of the transactions contemplated by this subsection 4 as the Board of Directors shall determine to be appropriate.

5. Mergers and Consolidations. In the event of a merger or consolidation to which the Corporation is a party and pursuant to which the holders of common stock of the Corporation are entitled to receive securities or other consideration or pursuant to which shares of common stock of the Corporation are converted into securities or other consideration, the holder of each outstanding share of Class A Common Stock shall be entitled to receive, or to have such share converted into, as the case may be, the securities or other consideration attributable pursuant to such merger or consolidation to the number of shares of Company Common Stock (or, if the Company Common Stock is not Publicly Traded at such time and shares of any other class or series of common stock of the Corporation (other than Class A Common Stock) are then Publicly Traded, of such other class or series of common stock as has the largest Market Capitalization as of the close of business on the Trading Day immediately preceding the date of the first public announcement of such merger or consolidation) equal to the Market Value Ratio of the Class A Common Stock to the Company Common Stock as of the Trading Day immediately preceding the date of the first public announcement of such merger or consolidation, subject to adjustment as determined by the Board of Directors to be appropriate to reflect any subdivision (by stock split or otherwise) or combination (by reverse stock split or otherwise) of either class of Common Stock or any dividend or other distribution of shares of such class of Common Stock to holders of shares of such class of Common Stock after such date and prior to the consummation of such merger or consolidation.

6. *Board Determinations Binding.* Subject to applicable law, any determinations made in good faith by the Board of Directors of the Corporation under any provision of this Article IV, and any determinations with respect to any Group or the rights of the holders of any class of Common Stock made pursuant to or in furtherance of this Article IV, shall be final and binding on all stockholders.

7. Certain Definitions. As used in this Article IV, the following terms shall have the following meanings (with terms defined in the singular having comparable meaning when used in the plural and vice versa), unless the context otherwise requires. As used in this Article IV, references to the "Board of Directors" shall refer to the Board of Directors of the Corporation. As used in this subsection 7, a "contribution" or

"transfer" of assets or properties from one Group to another shall refer to the reattribution of such assets or properties from the contributing or transferring Group to the other Group and correlative phrases shall have correlative meanings.

7.1. Atlantic Utility Group shall mean, as of any date from and after the Effective Date:

(A) the interest of the Corporation on such date in Atlantic City Electric Company, a New Jersey corporation (the "Atlantic Utility Group Company"), and any successor companies, and solely those lines of business in which the Atlantic Utility Group Company was engaged as of August 9, 1996, and the assets and liabilities attributable to those lines of business, and which meet all of the criteria listed in clause (i) below, as of August 9, 1996, specifically excluding those businesses, lines of business, and products and services listed in clause (ii) below, and specifically including those businesses, lines of business and products and services listed in clause (iii) below:

(i) only those businesses meeting all of the following criteria as of August 9, 1996 shall be included in the Atlantic Utility Group: (a) price regulated by the New Jersey Board of Public Utilities, including, without limitation, off-tariff agreements; (b) directly related to the supply of electricity (generation and purchase of electricity) or the delivery of electricity (transmission and distribution of electricity); and (c) a line of business for which Atlantic Utility Group Company has a franchise;

(ii) specifically excluded from the Atlantic Utility Group are the following businesses, lines of business and products and services, which list is not intended to be inclusive: (a) Appliance Shield program, (b) lighting upgrade programs, (c) water heater service business, (d) thermostat sales, (e) telecommunications business and investments, (f) energy services and consulting, (g) utility services and consulting (e.g. plant services and electrical equipment O&M services), (h) Atlantic Energy International, and (i) Atlantic Energy Enterprises and its subsidiaries;

(iii) specifically included in the Atlantic Utility Group is the Deepwater Operating Company.

(B) all assets and liabilities of the Corporation and its subsidiaries (other than capital stock of a subsidiary) and liabilities relating to the notional obligation of the Atlantic Utility Group to the Residual Group described in clause (iii) of the definition of Company Net Income (Loss) Attributable to the Atlantic Utility Group on such date attributed by the Board of Directors to any of the Atlantic Utility Group Company or the businesses thereof, whether or not such assets or liabilities are or were also assets and liabilities of the Atlantic Utility Group Company, including, without limitation, the assets and liabilities as of the Effective Date specified in the schedules filed with the records of the actions of the Board of Directors (a copy of which shall be made available to any stockholder of the Corporation upon written request therefor);

(C) all properties and assets transferred to the Atlantic Utility Group from the Residual Group (other than a transaction pursuant to paragraph (D) of this subsection 7.1) after the Effective Date pursuant to transactions in the ordinary course of business of both the Residual Group and the Atlantic Utility Group or otherwise as the Board of Directors may have directed;

(D) all properties and assets transferred to the Atlantic Utility Group from the Residual Group in connection with an increase in the Number of Shares Issuable with respect to the Intergroup Interest; and

(E) the interest of the Corporation or any of its subsidiaries in any business or asset acquired and any liabilities assumed by the Corporation or any of its subsidiaries outside of the ordinary course of business and attributed to the Atlantic Utility Group, as determined by the Board of Directors;

provided that (1) from and after the payment date of any dividend or other distribution with respect to shares of Class A Common Stock (other than a dividend or other distribution payable in shares of Class A Common Stock, with respect to which adjustment shall be made as provided in paragraph (A) of



IV-13

subsection 7.16, or in securities of the Corporation attributed to the Atlantic Utility Group, for which provision shall be made as set forth in clause (2) of this proviso), the Atlantic Utility Group shall no longer include an amount of assets or properties previously attributed to the Atlantic Utility Group of the same kind as so paid in such dividend or other distribution with respect of shares of Class A Common Stock as have a Fair Value on the record date for such dividend or distribution equal to the product of (a) the Fair Value on such record date of the aggregate of such dividend or distribution to holders of shares of Class A Common Stock declared multiplied by (b) a fraction the numerator of which is equal to the Intergroup Interest Fraction in effect on the record date for such dividend or distribution and the denominator of which is equal to the Outstanding Atlantic Utility Fraction in effect on the record date for such dividend or distribution, (2) if the Corporation shall pay a dividend or make some other distribution with respect to shares of Class A Common Stock payable in securities of the Corporation that are attributed to the Atlantic Utility Group for purposes of this Article IV (other than Class A Common Stock), there shall be excluded from the Atlantic Utility Group an interest in the Atlantic Utility Group equivalent to the number or amount of such securities that is equal to the product of the number or amount of securities so distributed to holders of Class A Common Stock multiplied by the fraction specified in clause 1(b) of this proviso (determined as of the record date for such distribution) (and such interest in the Atlantic Utility Group shall be attributed to the Residual Group) and, to the extent interest is or dividends are paid on the securities so distributed, the Atlantic Utility Group shall no longer include a corresponding ratable amount of the kind of assets paid as such interest or dividends as would have been paid in respect of the securities equivalent to such interest in the Atlantic Utility Group deemed held by the Residual Group if the securities equivalent to such interest were outstanding (and in such eventuality such assets as are no longer included in the Atlantic Utility Group shall be attributed to the Residual Group) and (3) from and after any transfer of any assets or properties from the Atlantic Utility Group to the Residual Group, the Atlantic Utility Group shall no longer include such assets or properties so contributed or transferred. The Corporation may also, to the extent a dividend or distribution on the Class A Common Stock has been paid in Convertible Securities that are convertible into or exchangeable or exercisable for Class A Common Stock, cause such Convertible Securities as are deemed to be held by the Residual Group in accordance with the third to last sentence of subsection 7.21 and clause (2) of the proviso to the immediately preceding sentence to be deemed to be converted, exchanged or exercised as provided in the penultimate sentence of subsection 7.21, in which case such Convertible Securities shall no longer be deemed to be held by the Residual Group.

# 7.2. Atlantic Utility Group Available Dividend Amount, on any date, shall mean either:

(i) (x) the amount equal to the product of (1) the Outstanding Atlantic Utility Fraction as of such date multiplied by (2) an amount equal to the fair market value of the total assets attributed to the Atlantic Utility Group less the total amount of the liabilities attributed to the Atlantic Utility Group (provided that preferred stock shall not be treated as a liability), in each case as of such date and determined on a basis consistent with that applied in determining Company Net Income (Loss) Attributable to the Atlantic Utility Group, minus (y) the aggregate par value of, or any greater amount determined to be capital in respect of, all outstanding shares of Class A Common Stock and shares of each class or series of Preferred Stock attributed to the Atlantic Utility Group, plus (z) the amount, as of such date, of amortization of goodwill during the period from the Effective Date through such date arising from the mergers (together, the "Mergers") of Atlantic Energy, Inc., a New Jersey corporation, with and into the Corporation, and of DS Sub, Inc., a Delaware corporation, with and into Delmarva Power & Light Company, a Delaware corporation, with respect to the Atlantic Utility Group, or

(ii) in case the total amount calculated pursuant to clause (i) above is not a positive number, an amount equal to Company Net Income (Loss) Attributable to the Atlantic Utility Group (if positive) for the fiscal year in which the dividend is declared and/or the preceding fiscal year.

Notwithstanding the foregoing provisions of this subsection 7.2, and consistent with subsection 7.6, at any time when there are not outstanding both (i) one or more shares of Residual Common Stock or Convertible Securities convertible into or exchangeable or exercisable for Residual Common Stock and (ii) one or more shares of Class A Common Stock or Convertible Securities convertible into or exchangeable or exercisable for Class A Common Stock, the "Available Dividend Amount," on any calculation date during such time period, with respect to the Residual Common Stock or Convertible Securities convertible securities convertible into or exchangeable or exercisable for Class A Common Stock, the "Available Dividend Amount," on any calculation date during such time period, with respect to the Residual Common Stock or Convertible Securities convertible into or exchangeable or exercisable for such classes of Common Stock is outstanding), shall mean the amount available for the payment of dividends on such Common Stock in accordance with law.

7.3. Company Net Income (Loss) Attributable to the Atlantic Utility Group, for any period through any date, shall mean (i) the net income or loss of the Atlantic Utility Group for such period (or in respect of the fiscal periods of the Corporation commencing prior to the Effective Date, the pro forma net income or loss of the Atlantic Utility Group for such period as if the Effective Date had been the first day of such period) determined in accordance with generally accepted accounting principles in effect at such time, reflecting income and expense of the Corporation attributed to the Atlantic Utility Group on a basis substantially consistent with attributions of income and expense made in the calculation of Company Net Income (Loss) Attributable to the Residual Group, including, without limitation, corporate administrative costs, net interest and other financial costs and income taxes, increased (or, in the case of a loss, reduced) by (ii) the amount of amortization of goodwill arising from the Mergers with respect to the Atlantic Utility Group to the extent but only to the extent such goodwill would otherwise decrease the Company Net Income Attributable to the Atlantic Utility Group or increase the Company Net Loss Attributable to the Atlantic Utility Group (such amount calculated for fiscal periods of the Corporation commencing prior to the Effective Date on a pro forma basis as if the Effective Date had been the first day of the relevant period), determined in accordance with generally accepted accounting principles in effect at such time applied on a basis substantially consistent with that applied in determining Company Net Income (Loss) Attributable to the Atlantic Utility Group and reduced (or, in the case of a loss, increased) by (iii) an amount equal to \$40 million per fiscal year and by (iv) the amount of dividends paid in such period (or, in respect of fiscal periods of the Corporation commencing prior to the Effective Date, the pro forma amount of dividends paid in such period as if the Effective Date had been the first day of such period) with respect to shares of preferred and preference stock of Atlantic Utility Group Company.

7.4. Company Net Income (Loss) Attributable to the Residual Group, for any period through any date, shall mean (i) the net income or loss of the Residual Group for such period (or in respect of fiscal periods of the Corporation commencing prior to the Effective Date, the pro forma net income or loss of the Residual Group for such period as if the Effective Date had been the first day of such period) determined in accordance with generally accepted accounting principles in effect at such time, reflecting income and expense of the Corporation attributed to the Residual Group on a basis substantially consistent with attributions of income and expense made in the calculation of Company Net Income (Loss) Attributable to the Atlantic Utility Group, including, without limitation, corporate administrative costs, net interest and other financial costs and income taxes, reduced (or, in the case of a loss, increased) by (ii) the amount of amortization of goodwill arising from the Mergers with respect to the Atlantic Utility Group to the extent but only to the extent such goodwill results in a decrease (increase) in the Company Net Income (Loss) Attributable to the Atlantic Utility Group pursuant to clause (ii) of subsection 7.3 of this Article IV (such amount calculated for fiscal periods of the Corporation commencing prior to the Effective Date on a pro forma basis as if the Effective Date had been the first day of the relevant period), determined in accordance with generally accepted accounting principles in effect at such time applied on a basis substantially consistent with that applied in determining Company Net Income (Loss) Attributable to the Atlantic Utility Group (excluding the portion thereof, if any, already applied to reduce net income or increase net loss of the Residual Group for such period by virtue of the Intergroup Interest Fraction) and increased (or, in the case of a loss, decreased) by (iii) the amount described in clause (iii) of the definition of Company Net Income (Loss) Attributable to the Atlantic Utility Group.

7.5. Conversion Date shall mean the date fixed by the Board of Directors as the effective date for the conversion of shares of Class A Common Stock into shares of Company Common Stock (or another class or series of common stock of the Corporation) as shall be set forth in the notice to holders of shares of Class A Common Stock and to holders of any Convertible Securities that are convertible into or exchangeable or exercisable for shares of Class A Common Stock required pursuant to paragraph (E) of subsection 4.3 of this Section II.

7.6. *Convertible Securities* at any time shall mean any securities of the Corporation or of any subsidiary thereof (other than shares of Common Stock), including warrants and options, outstanding at such time that by their terms are convertible into or exchangeable or exercisable for or evidence the right to acquire any shares of any class of Common Stock, whether convertible, exchangeable or exercisable at such time or a later time or only upon the occurrence of certain events, but in respect of antidilution provisions of such securities only upon the effectiveness thereof.

7.7. *Disposition* shall mean a sale, transfer, assignment or other disposition (whether by merger, consolidation, sale or contribution of assets or stock or otherwise) of properties or assets (including stock, other securities and goodwill).

7.8. *Effective Date* shall mean the date on which the merger of Atlantic Energy, Inc., a New Jersey corporation, with and into the Corporation, and the merger of DS Sub, Inc., a Delaware corporation, with and into Delmarva Power & Light Company, a Delaware corporation, shall become effective.

7.9. Fair Value shall mean, in the case of equity securities or debt securities of a class or series that has previously been Publicly Traded for a period of at least 15 months, the Market Value thereof (if such value, as so defined, can be determined) or, in the case of an equity security or debt security that has not been Publicly Traded for at least such period, shall mean the fair value per share of stock or per other unit of such other security, on a fully distributed basis, as determined by an independent investment banking firm experienced in the valuation of securities selected in good faith by the Board of Directors, or, if no such investment banking firm is, as determined in the good faith judgment of the Board of Directors, available to make such determination, in good faith by the Board of Directors; provided, however, that in the case of property other than securities, the "Fair Value" thereof shall be determined in good faith by the Board of Directors based upon such appraisals or valuation reports of such independent experts as the Board of Directors shall in good faith determine to be appropriate in accordance with good business practice. Any such determination of Fair Value shall be described in a statement filed with the records of the actions of the Board of Directors.

7.10. *Group* shall mean, as of any date, the Residual Group or the Atlantic Utility Group, as the case may be.

7.11. Intergroup Interest Fraction as of any date shall mean a fraction the numerator of which shall be the Number of Shares Issuable with Respect to the Intergroup Interest on such date and the denominator of which shall be the sum of (A) such Number of Shares Issuable with Respect to the Intergroup Interest and (B) the aggregate number of shares of Class A Common Stock outstanding on such date. A statement setting forth the Intergroup Interest Fraction as of the record date for any dividend or distribution on any class of Common Stock, as of the effective date of any conversion, exchange or exercise of Convertible Securities into or for shares of Class A Common Stock and as of the end of each fiscal quarter of the Corporation shall be filed by the Secretary of the Corporation in the records of the Board of Directors of the Corporation not later than ten days after such date.

7.12. *Market Capitalization* of any class or series of common stock on any date shall mean the product of (i) the Market Value of one share of such class or series of common stock on such date and (ii) the number of shares of such class or series of common stock outstanding on such date.

7.13. *Market Value* of a share of any class or series of capital stock of the Corporation on any day shall mean the average of the high and low reported sales prices regular way of a share of such class or series on such Trading Day or, in case no such reported sale takes place on such Trading Day, the average of the

reported closing bid and asked prices regular way of a share of such class or series on such Trading Day, in either case as reported on the New York Stock Exchange Composite Tape or, if the shares of such class or series are not listed or admitted to trading on such Exchange on such Trading Day, on the principal national securities exchange in the United States on which the shares of such class or series are listed or admitted to trading or, if not listed or admitted to trading on any national securities exchange on such Trading Day, on the NASDAQ National Market or, if the shares of such class or series are not listed or admitted to trading on any national securities exchange or quoted on such National Market System on such Trading Day, the average of the closing bid and asked prices of a share of such class or series in the over-the-counter market on such Trading Day as furnished by any New York Stock Exchange member firm selected from time to time by the Corporation or, if such closing bid and asked prices are not made available by any such New York Stock Exchange member firm on such Trading Day, the Fair Value of a share of such class or series; provided that, for purposes of determining the market value of a share of any class or series of capital stock for any period, (i) the "Market Value" of a share of capital stock on any day prior to any "ex-dividend" date or any similar date occurring during such period for any dividend or distribution (other than any dividend or distribution contemplated by clause (ii)(B) of this sentence) paid or to be paid with respect to such capital stock shall be reduced by the Fair Value of the per share amount of such dividend or distribution and (ii) the "Market Value" of any share of capital stock on any day prior to (A) the effective date of any subdivision (by stock split or otherwise) or combination (by reverse stock split or otherwise) of outstanding shares of such class or series of capital stock occurring during such period or (B) any "ex-dividend" date or any similar date occurring during such period for any dividend or distribution with respect to such capital stock to be made in shares of such class or series of capital stock or Convertible Securities that are convertible, exchangeable or exercisable for such class or series of capital stock shall be appropriately adjusted, as determined by the Board of Directors, to reflect such subdivision, combination, dividend or distribution.

7.14. Market Value Ratio of the Class A Common Stock to the Company Common Stock as of any date shall mean the fraction (which may be greater than 1/1), expressed as a decimal (rounded to the nearest five decimal places), of a share of Company Common Stock (or another class or series of common stock of the Corporation, if so provided by subsection 4.1 or subsection 5 of this Section II because Company Common Stock is not then Publicly Traded) to be issued in respect of a share of Class A Common Stock upon a conversion of Class A Common Stock into Company Common Stock (or another class or series of common stock of the Corporation) in accordance with subsection 4.1 of this Section II or to determine the consideration attributable to a share of Class A Common Stock in accordance with subsection 5 of this Section II, based on the ratio of the market value of a share of Class A Common Stock to the market value of a share of Company Common Stock (or such other common stock) as of such date, determined by the fraction the numerator of which shall be the sum of (A) four times the average Market Value of one share of Class A Common Stock over the period of five consecutive Trading Days ending on such date, (B) three times the average Market Value of one share of Class A Common Stock over the period of five consecutive Trading Days ending on the fifth Trading Day prior to such date, (C) two times the average Market Value of one share of Class A Common Stock over the period of five consecutive Trading Days ending on the tenth Trading Day prior to such date and (D) the average Market Value of one share of Class A Common Stock over the period of five consecutive Trading Days ending on the fifteenth Trading Day prior to such date and the denominator of which shall be the sum of (A) four times the average Market Value of one share of Company Common Stock (or such other common stock) over the period of five consecutive Trading Days ending on such date, (B) three times the average Market Value of one share of Company Common Stock (or such other common stock) over the period of five consecutive Trading Days ending on the fifth Trading Day prior to such date, (C) two times the average Market Value of one share of Company Common Stock (or such other common stock) over the period of five consecutive Trading Days ending on the tenth Trading Day prior to such date and (D) the average Market Value of one share of Company Common Stock (or such other common stock) over the period of five consecutive Trading Days ending on the fifteenth Trading Day prior to such date.

7.15. Net Proceeds shall mean, as of any date with respect to any Disposition of any of the properties and assets attributed to the Atlantic Utility Group an amount, if any, equal to what remains of the gross proceeds of such Disposition after payment of, or reasonable provision is made as determined by the Board of Directors for, (A) any taxes payable by the Corporation (or which would have been payable but for the utilization of tax benefits attributable to the Residual Group) in respect of such Disposition or in respect of any resulting dividend or redemption pursuant to subparagraph (A)(1)(a) or (b) of subsection 4.1 of this Section II, (B) any transaction costs, including, without limitation, any legal, investment banking and accounting fees and expenses, (C) any liabilities (contingent or otherwise) of or attributed to the Atlantic Utility Group, including, without limitation, any liabilities for deferred taxes or any indemnity or guarantee obligations of the Corporation incurred in connection with the Disposition or otherwise, and any liabilities for future purchase price adjustments and any preferential amounts plus any accumulated and unpaid dividends in respect of Preferred Stock attributed to the Atlantic Utility Group and (D) a capitalized amount (as determined by the Board of Directors) of the notional obligation of the Atlantic Utility Group to the Residual Group described in clause (iii) of the definition of Company Net Income (Loss) Attributable to the Atlantic Utility Group. For purposes of this definition, any properties and assets attributed to the Atlantic Utility Group remaining after such Disposition shall constitute "reasonable provision" for such amount of taxes, costs, liabilities (contingent or otherwise) and capitalized amount as the Board of Directors determines can be expected to be supported by such properties and assets.

7.16. *Number of Shares Issuable with Respect to the Intergroup Interest* shall as of the Effective Date be \_\_\_\_\_\_\_^(1) provided, however, that such number shall from time to time thereafter be:

(A) adjusted as determined by the Board of Directors to be appropriate to reflect any subdivision (by stock split or otherwise) or combination (by reverse stock split or otherwise) of the Class A Common Stock or any dividend or other distribution of shares of Class A Common Stock to holders of shares of Class A Common Stock or any reclassification of Class A Common Stock;

(B) decreased (but to not less than zero) by action of the Board of Directors by (1) the number of shares of Class A Common Stock issued or sold by the Corporation that, immediately prior to such issuance or sale, were included (as determined by the Board of Directors pursuant to paragraph (C) of this subsection 7.16) in the Number of Shares Issuable with Respect to the Intergroup Interest, (2) the number of shares of Class A Common Stock issued upon conversion, exchange or exercise of Convertible Securities that, immediately prior to the issuance or sale of such Convertible Securities, were included in the Number of Shares Issuable with Respect to the Intergroup Interest, (3) the number of shares of Class A Common Stock issued by the Corporation as a dividend or other distribution (including in connection with any reclassification or exchange of shares) to holders of Company Common Stock, (4) the number of shares of Class A Common Stock issued upon the conversion, exchange or exercise of any Convertible Securities issued by the Corporation as a dividend or other distribution (including in connection with any reclassification or exchange of shares) to holders of Company Common Stock, or (5) the number (rounded, if necessary, to the nearest whole number) equal to the quotient of (a) the aggregate Fair Value as of the date of contribution of properties or assets (including cash) transferred from the Atlantic Utility Group to the Residual Group in consideration for a reduction in the Number of Shares Issuable with Respect to the Intergroup Interest divided by (b) the Market Value of one share of Class A Common Stock as of the date of such transfer; and

(C) increased by (1) the number of outstanding shares of Class A Common Stock repurchased by the Corporation for consideration that is attributed as provided by subsection 7.21 to the Residual Group and (2) the number (rounded, if necessary, to the nearest whole number) equal to the quotient of (a) the Fair Value of properties or assets (including cash) theretofore attributed as provided by

<sup>(1)</sup> This number equals 7/3 times the number of Class A Common Stock to be issued pursuant to the Mergers.

subsection 7.21 to the Residual Group that are contributed to the Atlantic Utility Group in consideration of an increase in the Number of Shares Issuable with Respect to the Intergroup Interest, divided by (b) the Market Value of one share of Class A Common Stock as of the date of such contribution and (3) the number of shares of Class A Common Stock into or for which Convertible Securities are deemed converted, exchanged or exercised pursuant to the penultimate sentence of the definition of "Residual Group" in subsection 7.21 of this Section II.

7.17. Outstanding Atlantic Utility Fraction, as of any date, means the fraction (which may simplify to 1/1) the numerator of which shall be the number of shares of Class A Common Stock outstanding on such date and the denominator of which shall be the sum of the number of shares of Class A Common Stock outstanding on such date and the Number of Shares Issuable with Respect to the Intergroup Interest on such date. A statement setting forth the Outstanding Atlantic Utility Fraction as of the record date for the payment of any dividend or distribution on any class of Common Stock and as of the end of each fiscal quarter of the Corporation shall be filed by the Secretary of the Corporation in the records of the actions of the Board of Directors not later than ten days after such date.

7.18. *Publicly Traded* with respect to any security shall mean (i) registered under Section 12 of the Securities Exchange Act of 1934, as amended (or any successor provision of law), and (ii) listed for trading on the New York Stock Exchange or the American Stock Exchange (or any national securities exchange registered under Section 7 of the Securities Exchange Act of 1934, as amended (or any successor provision of law), that is the successor to either such exchange) or quoted in the National Association of Securities Dealers Automation Quotation System (or any successor system).

7.19. *Redemption Date* shall mean the date fixed by the Board of Directors as the effective date for a redemption of shares of Class A Common Stock, as set forth in a notice to holders thereof required pursuant to paragraph (C), (D) or (F) of subsection 4.3 of this Section II.

7.20. Related Business Transaction means any Disposition of all or substantially all the properties and assets attributed to the Atlantic Utility Group in a transaction or series of related transactions that result in the Corporation receiving in consideration of such properties and assets primarily equity securities (including, without limitation, capital stock, debt securities convertible into or exchangeable for equity securities or interests in a general or limited partnership or limited liability company, without regard to the voting power or other management or governance rights associated therewith) of (1) any entity which (i) acquires such properties or assets or succeeds (by merger, formation of a joint venture or otherwise) to the business conducted with such properties or assets or controls such acquiror or successor and (ii) is primarily engaged or proposes to engage primarily in one or more businesses similar or complementary to the businesses conducted by the Atlantic Utility Group prior to such Disposition, as determined by the Board of Directors.

7.21. Residual Group shall mean, as of any date from and as of the Effective Date:

(A) the interest of the Corporation or any of its subsidiaries on such date in all of the assets, liabilities and businesses of the Corporation or any of its subsidiaries (and any successor companies), other than any assets, liabilities and businesses attributed in accordance with this Article IV to the Atlantic Utility Group;

(B) a proportionate undivided interest in each and every business, asset and liability attributed to the Atlantic Utility Group equal to the Intergroup Interest Fraction as of such date;

(C) all properties and assets transferred to the Residual Group from the Atlantic Utility Group (other than pursuant to paragraph (D) of this subsection 7.19) after the Effective Date pursuant to transactions in the ordinary course of business of both the Residual Group and the Atlantic Utility Group or otherwise as the Board of Directors may have directed;

(D) all properties and assets transferred to the Residual Group from the Atlantic Utility Group in connection with a reduction of the Number of Shares Issuable with Respect to the Intergroup Interest;

(E) the interest of the Corporation or any of its subsidiaries in any business or asset acquired and any liabilities assumed by the Corporation or any of its subsidiaries outside the ordinary course of business and attributed to the Residual Group, as determined by the Board of Directors; and

(F) from and after the payment date of any dividend or other distribution with respect to shares of Class A Common Stock (other than a dividend or other distribution payable in shares of Class A Common Stock, with respect to which adjustment shall be made as provided in paragraph (A) of subsection 7.16, or in securities of the Corporation attributed to the Atlantic Utility Group, for which provision shall be made as set forth in the third to last sentence of this definition), an amount of assets or properties previously attributed to the Atlantic Utility Group of the same kind as were paid in such dividend or other distribution with respect to shares of Class A Common Stock as have a Fair Value on the record date for such dividend or distribution equal to the product of (1) the Fair Value on such record date of the aggregate of such dividend or distribution to holders of shares of Class A Common Stock declared multiplied by (2) a fraction the numerator of which is equal to the Intergroup Interest Fraction in effect on the record date for such dividend or distribution and the denominator of which is equal to the Outstanding Atlantic Utility Fraction in effect on the record date for such dividend or distribution; provided that from and after any transfer of any assets or properties from the Residual Group to the Atlantic Utility Group, the Residual Group shall no longer include such assets or properties so transferred (other than as reflected in respect of such a transfer by the Intergroup Interest Fraction, as provided by paragraph (B) of this subsection 7.21).

If the Corporation shall pay a dividend or make some other distribution with respect to shares of Class A Common Stock payable in securities of the Corporation that are attributed to the Atlantic Utility Group for purposes of this Article IV (other than Class A Common Stock), the Residual Group shall be deemed to hold an interest in the Atlantic Utility Group equivalent to the number or amount of such securities that is equal to the product of the number or amount of securities so distributed to holders of Class A Common Stock multiplied by the fraction specified in clause (2) of paragraph (F) of this subsection 7.21 (determined as of the record date for such distribution) and, to the extent interest is or dividends are paid on the securities so distributed, the Residual Group shall include, and there shall be transferred thereto out of the Atlantic Utility Group, a corresponding ratable amount of the kind of assets paid as such interest or dividends as would have been paid in respect of such securities so deemed to be held by the Residual Group if such securities were outstanding. The Corporation may also, to the extent the securities so paid as a dividend or other distribution to the holders of Class A Common Stock are Convertible Securities and at the time are convertible into or exchangeable or exercisable for shares of Class A Common Stock, treat such Convertible Securities as are so deemed to be held by the Residual Group to be deemed to be converted, exchanged or exercised, and shall do so to the extent such Convertible Securities are mandatorily converted, exchanged or exercised (and to the extent the terms of such Convertible Securities require payment of consideration for such conversion, exchange or exercise, the Residual Group shall then no longer include an amount of the kind of properties or assets required to be paid as such consideration for the amount of Convertible Securities deemed converted, exchanged or exercised (and the Atlantic Utility Group shall be attributed such properties or assets)), in which case, from and after such time, the securities into or for which such Convertible Securities so deemed to be held by the Residual Group were so considered converted, exchanged or exercised shall be deemed held by the Residual Group (as provided in clause (3) of paragraph (C) of subsection 7.16 of this Section II) and such Convertible Securities shall no longer be deemed to be held by the Residual Group. A statement setting forth the election to effectuate any such deemed conversion, exchange or exercise of Convertible Securities so deemed to be held by the Residual Group and the properties or assets, if any, to be attributed to the Atlantic Utility Group in consideration of such conversion, exchange or exercise (if any) shall be filed in

the records of the actions of the Board of Directors and, upon such filing, such deemed conversion, exchange or exercise shall be effectuated.

7.22. Residual Group Available Dividend Amount, on any date, shall mean either:

(i) (x) the amount equal to the fair market value of the total assets attributed to the Residual Group less the total amount of the liabilities attributed to the Residual Group (provided that preferred stock shall not be treated as a liability), in each case as of such date and determined on a basis consistent with that applied in determining Company Net Income (Loss) Attributable to the Residual Group, minus (y) the aggregate par value of, or any greater amount determined to be capital in respect of, all outstanding shares of Company Common Stock and shares of each class or series of Preferred Stock attributed to the Residual Group, minus (z) the amount, if any, as of such date, of amortization of goodwill during the period from the Effective Date through such date arising from the Mergers with respect to the Atlantic Utility Group (determined as set forth in clause (ii) of the definition of Company Net Income (Loss) Attributable to the Residual Group and, as in such clause (ii), excluding the portion thereof, if any, already applied to reduce net income or increase net loss of the Residual Group for such period by virtue of the Intergroup Interest Fraction), or

(ii) in case the total amount calculated pursuant to clause (i) above is not a positive number, an amount equal to Company Net Income (Loss) Attributable to the Residual Group (if positive) for the fiscal year in which the dividend is declared and/or the preceding fiscal year.

Notwithstanding the foregoing provisions of this subsection 7.22, at any time when there are not outstanding both (i) one or more shares of Company Common Stock or Convertible Securities convertible into or exchangeable or exercisable for Company Common Stock and (ii) one or more shares of Class A Common Stock or Convertible Securities convertible into or exchangeable or exercisable for Class A Common Stock, the "Available Dividend Amount," on any calculation date during such time period, with respect to the Company Common Stock or Convertible Securities convertible Securities convertible Securities convertible into or exchangeable or exercisable for Class A Common Stock, the "Available Dividend Amount," on any calculation date during such time period, with respect to the Company Common Stock or the Class A Common Stock, as the case may be (depending on which of such classes of Common Stock or Convertible Securities convertible into or exchangeable or exercisable for the payment of dividends on such Common Stock in accordance with law.

7.23. Time-Weighted Market-Price as of any date with respect to any class of Common Stock shall mean an amount equal to (i) the sum of (A) four times the average Market-Value of one share of such class of Common Stock over the period of five consecutive Trading Days ending on such date, (B) three times the average Market Value of one share of such class of Common Stock over the period of five consecutive Trading Days ending on the fifth Trading Day prior to such date, (C) two times the average Market Value of one share of such class of Common Stock over the period of five consecutive Trading Days ending Day prior to such date and (D) the average Market Value of one share of such class of Common Stock over the period of five consecutive Trading Days ending on the tenth Trading Day prior to such date and (D) the average Market Value of one share of such class of Common Stock over the period of five consecutive Trading Days ending on the fifteenth Trading Day prior to such date, divided by (ii) ten (10).

7.24. *Trading Day* shall mean each weekday other than any day on which the relevant class of common stock of the Corporation is not traded on any national securities exchange or quoted in the NASDAQ National Market or in the over-the-counter market.

SECTION III. *Preferred Stock.* The Preferred Stock may be issued from time to time in one or more series. The Board of Directors is authorized, by resolution adopted and filed in accordance with law, to fix the number of shares in each series, the designation thereof, the voting powers, preferences and relative participating, optional or other special rights thereof, and the qualifications or restrictions thereon, of each series and the variations in such voting powers and preferences and rights as between series. Any shares of any class or series of Preferred Stock purchased, exchanged, converted or otherwise acquired by the Corporation, in any manner whatsoever shall be retired and cancelled promptly after the acquisition thereof. All such shares shall upon their cancellation become authorized but unissued shares of Preferred Stock, without designation as to series, and may be reissued as part of any series of Preferred Stock created by resolution or resolutions of the Board of Directors, subject to the conditions and restrictions on issuance set forth in this certificate of incorporation or in such resolution or resolutions.

## ARTICLE V.

#### **BOARD OF DIRECTORS**

SECTION I. *Powers*. The business and affairs of the Corporation shall be managed by, or under the direction of, a Board of Directors, which shall exercise all of the powers of the Corporation except as are by law or by this Certificate of Incorporation or the Bylaws of the Corporation conferred upon or reserved to the stockholders of the Corporation.

## SECTION II. Number, Tenure and Qualifications of Directors.

1. *Number of Directors.* The Board of Directors shall consist of no fewer than 9 and no more than 18 directors, as determined from time to time by resolution of the Board of Directors

2. Terms of Directors. The directors shall be divided into three classes for the purpose of providing for staggered director terms, to be designated Class I, Class II and Class III. Each class shall consist, as nearly as possible, of one-third of the total number of directors constituting the entire Board of Directors. Class I directors shall be elected for a term expiring on the first annual meeting of stockholders following the Effective Date, Class II directors shall be elected for a term expiring on the second annual meeting of stockholders following the Effective Date, and Class III directors shall be elected for a term expiring on the third annual meeting of stockholders following the Effective Date. At each succeeding annual meeting of stockholders, successors to the class of directors whose term expires at that annual meeting shall be elected for three-year terms. If the number of directors is changed, any increase or decrease shall be apportioned among the classes so as to maintain the number of directors in each class as nearly equal as possible, and any additional director of any class elected to fill a vacancy resulting from an increase in such class shall hold office for a term that shall coincide with the remaining term of that class, but in no case will a decrease in the number of directors shorten the term of any incumbent director. A director shall hold office until the annual meeting of stockholders for the year in which his or her term expires and until his or her successor shall be elected and shall qualify, subject, however, to prior death, resignation, retirement, disqualification or removal from office. Except as otherwise required by law or in this Certificate of Incorporation, any vacancy on the Board of Directors that results from an increase in the number of directors and any other vacancy occurring in the Board of Directors shall be filled by a majority of the directors then in office, even if less than a quorum, or by the sole remaining director. Any director elected to fill a vacancy not resulting from an increase in the number of directors shall have the same remaining term as that of his or her predecessor.

3. *Removal of Directors.* Any director, or the entire Board of Directors, may be removed from office only for cause and only by the affirmative vote of not less than a majority of the votes entitled to be cast by the holders of all the then outstanding shares of capital stock of the Corporation of any class or series entitled to vote in the election of directors generally, voting together as one class at an annual meeting or at a special meeting of the stockholders called for such purpose.

4. Class Votes for Directors. Notwithstanding the foregoing, whenever the holders of any one or more classes or series of stock issued by the Corporation shall have the right, voting separately by class or series, to elect directors, the election, term of office, filling of vacancies and other features of such directorships shall be governed by the terms of this Certificate of Incorporation applicable thereto, as amended, and such directors so elected shall not be divided into classes pursuant to this Article V, Section II unless expressly provided by such terms.

SECTION III. Additional Authority of Board. In furtherance and not in limitation of the powers conferred by statute, the Board of Directors is expressly authorized to make, alter, amend or repeal the Bylaws of the Corporation.

SECTION IV. Nomination and Election of Directors. Subject to the rights of holders of any class or series of stock having a preference over the Common Stock as to dividends or upon liquidation, dissolution

or winding up of the Corporation, nominations for the election of directors shall be made by a nominating committee of the Board of Directors if then constituted pursuant to the Bylaws of the Corporation, or if no nominating committee has been constituted, by the Board of Directors. In addition, any stockholder entitled to vote in the election of directors generally may nominate one or more persons for election as directors at an annual meeting of stockholders, but only if written notice of such stockholder's intent to make such nomination or nominations has been received by the Secretary of the Corporation not less than sixty nor more than ninety days prior to the first anniversary of the preceding year's annual meeting of stockholders. In the event that the date of the annual meeting of stockholders is advanced by more than thirty days or delayed by more than sixty days from such anniversary or in the case of the Corporation's first annual meeting of stockholders after the Effective Date, notice by the stockholder to be timely must be received not earlier than the ninetieth day prior to such annual meeting and not later than the close of business on the later of (a) the sixtieth day prior to such annual meeting or (b) the tenth day following the day on which notice of the date of the annual meeting was mailed or public disclosure thereof was made by the Corporation, whichever first occurs. Each such notice by a stockholder shall set forth: (a) the name and address of the stockholder who intends to make the nomination and of the person or persons to be nominated; (b) a representation that the stockholder is a holder of record of stock of the Corporation entitled to vote at such meeting and intends to appear in person or by proxy at a meeting to nominate the person or persons specified in the notice; (c) a description of all arrangements or understandings between the stockholder or any person that directly or indirectly through one or more intermediaries controls, or is controlled by, or is under common control with, such stockholder (an "Affiliate" of such stockholder) and each nominee and any other person or persons (naming such person or persons) relating to the nomination or nominations; (d) the class and number of shares of the Corporation which are beneficially owned by such stockholder and the person to be nominated as of the date of such stockholder's notice and by any other stockholders known by such stockholder to be supporting such nominees as of the date of such stockholder's notice: (e) such other information regarding each nominee proposed by such stockholder as would be required to be included in a proxy statement filed pursuant to the proxy rules of the Securities and Exchange Commission; and (f) the written consent of each nominee to serve as a director of the Corporation if so elected. The stockholder shall also comply with all applicable requirements of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and the rules and regulations thereunder, with respect to the matters-set-forth-in this-Article-V-Section-IV.

In addition, in the event the Corporation calls a special meeting of stockholders for the purpose of electing one or more directors, any stockholder entitled to vote in the election of directors generally may nominate one or more persons for election as directors at a special meeting only if written notice of such stockholder's intent to make such nomination or nominations, setting forth the information and complying with the form described in the inmediately preceding paragraph, has been received by the Secretary of the Corporation not earlier than the ninetieth day prior to such special meeting or (ii) the tenth day following the day on which notice of the date of the special meeting was mailed or public disclosure thereof was made by the Corporation, whichever comes first. The stockholder shall also comply with all applicable requirements of the Exchange Act and the rules and regulations thereunder with respect to the matters set forth in this Article V, Section IV.

No person shall be eligible for election as a director of the Corporation unless nominated in accordance with the procedures set forth in this Article V, Section IV. The presiding officer of the meeting shall, if the facts warrant, determine and declare to the meeting that a nomination was not made in accordance with the procedures prescribed by this Article V, Section IV, and if he or she should so determine, the defective nomination shall be disregarded.

Elections of directors shall be by written ballot.

## ARTICLE VI.

#### STOCKHOLDERS

SECTION I. *Meetings of Stockholders; Books.* Meetings of the stockholders may be held within or without the State of Delaware, as the Bylaws may provide. Any action required or permitted to be taken by the stockholders of the Corporation must be effected at a duly called annual or special meeting of such stockholders and may not be effected by a consent in writing by any such holders. Subject to the rights of holders of any class or series of stock having a preference over the Common Stock as to dividends or upon liquidation, dissolution or winding up of the Corporation, special meetings of the stockholders of the Corporation may be called only by the Chairman of the Board or by the Board of Directors pursuant to a resolution approved by a majority of the entire Board of Directors. The books of the Corporation may be kept (subject to any provision of law) outside the State of Delaware at such place or places as may be designated from time to time by the Board of Directors or in the Bylaws of the Corporation.

Except as otherwise required by law or by this Certificate of Incorporation, the holders of not less than a majority in voting power of the shares entitled to vote at any meeting of stockholders, present in person or by proxy, shall constitute a quorum, and in all matters other than the election of directors the act of the holders of a majority in voting power of the shares present in person or by proxy and entitled to vote on the subject matter shall be deemed the act of the stockholders. Directors shall be elected by a plurality of the vote of the shares present in person or represented by proxy at a meeting of stockholders and entitled to vote in the election of directors. If a quorum shall fail to attend any meeting, the presiding officer may adjourn the meeting to another place, date or time.

SECTION II. Proposals of Stockholders. At any meeting of the stockholders, only such business shall be conducted as shall have been properly brought before such meeting. To be properly brought before an annual meeting of stockholders, business must be (a) specified in the notice of meeting (or any supplement thereto) given by or at the direction of the Board of Directors, (b) otherwise properly brought before the meeting by or at the direction of the Board of Directors or (c) otherwise properly brought before the meeting by a stockholder. For business to be properly brought before an annual meeting by a stockholder, the stockholder must have given timely notice thereof in writing to the Secretary of the Corporation. To be timely, a stockholder's notice must be received no less than sixty days nor more than ninety days prior to the first anniversary of the preceding year's annual meeting of stockholders; provided, however, that in the event that the date of the annual meeting is advanced by more than thirty days or delayed by more than sixty days from such anniversary or in the case of the Corporation's first annual meeting of stockholders after the Effective Date, notice by the stockholder to be timely must be received not earlier than the ninetieth day prior to such annual meeting of stockholders and not later than the close of business on the later of (a) the sixtieth day prior to such annual meeting or (b) the tenth day following the date on which notice of the date of the annual meeting was mailed or public disclosure thereof was made, whichever first occurs. Each such notice shall set forth as to each matter the stockholder proposes to bring before the annual meeting of stockholders: (a) a brief description of the business desired to be brought before the annual meeting of stockholders and the reasons for conducting such business at such meeting, (b) the name and address, as they appear on the Corporation's books, of the stockholder proposing such business, (c) the class, series and number of shares of the Corporation which are beneficially owned by the stockholder, and (d) any material interest of the stockholder or any Affiliate of the stockholder in such business. The stockholder shall also comply with all applicable requirements of the Exchange Act and the rules and regulations thereunder with respect to the matters set forth in this Article VI, Section II. To be properly brought before a special meeting, business must be (a) specified in the notice of meeting (or any supplement thereto) given by or at the direction of the Board of Directors or (b) otherwise properly brought before the meeting by or at the direction of the Board of Directors. No business may be brought before a special meeting by stockholders.

No business shall be conducted at any meeting of the stockholders except in accordance with the procedures set forth in this Article VI, Section II. The presiding officer of the meeting shall, if the facts warrant, determine and declare to the meeting that business was not properly brought before the meeting and in accordance with the provisions of this Article VI, Section II, and if he or she should so determine, any such business not properly brought before the meeting shall not be transacted. Nothing herein shall be deemed to affect any rights of stockholders to request inclusion of proposals in the Corporation's proxy statement pursuant to Rule 14a-8 under the Exchange Act or any successor provision.

SECTION III. *Power to Amend Bylaws.* The stockholders shall have the power and authority to amend the Bylaws of the Corporation only by the affirmative vote of 80% or more of the aggregate number of votes that the holders of the then outstanding shares of Common Stock and Preferred Stock are entitled to cast on the amendment.

# ARTICLE VII.

#### AMENDMENTS

The Corporation reserves the right to amend, alter, change or repeal any provision contained in this Certificate of Incorporation, in the manner now or hereafter prescribed by statute, and all rights conferred upon stockholders herein are granted subject to this reservation; provided, however, that except with respect to the designation of the rights and preferences of series of Preferred Stock pursuant to Article IV, Section III, which is delegated to the Board of Directors, and notwithstanding any other provisions of this Certificate of Incorporation or the Bylaws of the Corporation (and notwithstanding the fact that a lesser percentage or separate class vote may be specified by law, this Certificate of Incorporation or the Bylaws of the Corporation), any lawful amendment of this Certificate of Incorporation may be made by affirmative vote by at least the proportion specified below of the aggregate number of votes which the holders of the then outstanding shares of Common Stock and Preferred Stock are entitled to cast on the amendment and, if the shares of one or more classes or series are entitled under this Certificate of Incorporation or otherwise by law to vote thereon as a class, affirmative vote by the same proportion of the aggregate number of votes which the holders of the then outstanding shares of such one or more classes or series are entitled to cast on the amendment. The proportion referred to above in this Article VII shall be 80% in the case of any amendment of the provisions set forth in Section III of Article IV, Article V and Article VI of this Certificate of Incorporation and this Article VII and shall be a majority in all other cases.

### ARTICLE VIII.

# LIMITATION ON DIRECTOR LIABILITY AND INDEMNIFICATION OF DIRECTORS AND OFFICERS

SECTION I. *Limited Liability*. A person who is or was a director of the Corporation shall not be personally liable to the Corporation or its stockholders for monetary damages for breach of fiduciary duty as a director, except for liability (a) for any breach of the director's duty of loyalty to the Corporation or its stockholders, (b) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (c) under Section 174 of the GCLD, or (d) for any transaction from which the director derived an improper personal benefit. If the GCLD is amended to authorize corporate action further eliminating or limiting the personal liability of directors, then the liability of the directors of the Corporation and limitation of liability provided herein shall continue after a director has ceased to occupy such position as to acts or omissions occurring during such director's term or terms of office; and no amendment, repeal or modification of this Article IX shall apply to or have any effect on the liability or

alleged liability of any director of the Corporation for or with respect to any acts or omissions of such director occurring prior to such amendment, repeal or modification.

# SECTION II. Right to Indemnification.

1. Each person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (hereinafter a "proceeding"), by reason of the fact that he or she, or the person of whom he or she is the legal representative, is or was a director or officer of the Corporation or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust or other enterprise, including service with respect to employee benefit plans, shall be indemnified and held harmless by the Corporation to the fullest extent authorized by the GCLD, as the same exists or may hereafter be amended (but, in the case of any such amendment, only to the extent that such amendment permits the Corporation to provide broader indemnification rights than said law permitted the Corporation to provide prior to such amendment), against all expenses, liability and loss (including attorneys' fees, judgments, fines, ERISA excise taxes or penalties and amounts paid or to be paid in settlement) reasonably incurred or suffered by such person in connection therewith and such indemnification shall continue as to a person who has ceased to be a director, officer, employee or agent. and shall inure to the benefit of his or her heirs, executors and administrators; provided, however, that, except as provided in this Article VIII, Section II, the Corporation shall indemnify any such person seeking indemnification in connection with a proceeding (or part thereof) initiated by such person only if authorized by the Board of Directors of the Corporation. Any indemnification under this Article VIII, Section II (unless ordered by a court) shall be made by the Corporation only as authorized in the specific case upon a determination that indemnification of the director, officer, employee or agent is proper in the circumstances because he or she has met the applicable standard set forth in the GCLD. Such a determination shall be made (a) by a majority vote of the directors who are not parties to such action, suit or proceeding, even though less than a quorum; (b) by independent legal counsel (compensated by the Corporation) in a written opinion; (c) by the stockholders; or (d) in any other manner permitted by the GCLD. In addition to the right to indemnification conferred in this Article VIII, Section II, each of the above persons shall have the right to be paid by the Corporation the expenses incurred in defending any such proceeding in advance of its final disposition; provided, however, that, if the GCLD requires, the payment of such expenses incurred by a director or officer in his or her capacity as a director or officer of the Corporation (and not in any other capacity in which service was or is rendered by such person while a director or officer, including, without limitation, service to an employee benefit plan) in advance of the final disposition of a proceeding, shall be made only upon delivery to the Corporation of an undertaking, by or on behalf of such director or officer, to repay all amounts so advanced if it shall ultimately be determined that such director or officer is not entitled to be indemnified under this Section II or otherwise. The Corporation may, by action of its Board of Directors, provide indemnification to employees and agents of the Corporation with the same scope and effect as the foregoing indemnification of directors and officers. The right to indemnification and to an advancement of expenses conferred in this Article VIII, Section II, shall be a contract right.

2. If a claim under paragraph 1 of this Section II is not paid in full by the Corporation within 30 days after a written claim has been received by the Corporation, the claimant may at any time thereafter bring suit against the Corporation to recover the unpaid amount of the claim and, if successful in whole or in part, the claimant shall be entitled to be paid also the expense of prosecuting such claim (including attorneys' fees). It shall be a defense to any such action (other than an action brought to enforce a claim for expenses incurred in defending any proceeding in advance of its final disposition where the required undertaking, if any is required, has been tendered to the Corporation) that the claimant has not met the standard of conduct which makes it permissible under the GCLD for the Corporation to indemnify the claimant for the amount claimed, but the burden of proving such defense shall be on the Corporation. Neither the failure of the Corporation (including its Board of Directors, independent legal counsel, or its

IV-26

stockholders) to have made a determination prior to the commencement of such action that indemnification of the claimant is proper in the circumstances because he or she has met the applicable standard of conduct set forth in the GCLD, nor an actual determination by the Corporation (including its Board of Directors, independent legal counsel, or its stockholders) that the claimant has not met such applicable standard of conduct, shall be a defense to the action or create a presumption that the claimant has not met the applicable standard of conduct. In any suit brought by the claimant to enforce a right to indemnification or to an advancement of expenses hereunder, or brought by the Corporation to recover an advancement of expenses pursuant to the terms of an undertaking, the burden of proving that the claimant is not entitled to be indemnified, or to such advancement of expenses, under this Article VIII or otherwise shall be on the Corporation.

3. The rights to indemnification and the payment of expenses incurred in defending a proceeding in advance of its final disposition conferred in this Article VIII, Section II, shall not be exclusive of any other right which any person may have or hereafter acquire under any statute, provision of this Certificate of Incorporation, bylaw, agreement, vote of stockholders or disinterested directors, or otherwise.

4. The Corporation may maintain insurance, at its expense, to protect itself and any director, officer, employee or agent of the Corporation or another corporation, partnership, joint venture, trust or other enterprise against any such expense, liability or loss, whether or not the Corporation would have the power to indemnify such person against such expense, liability or loss under the GCLD.

5. The Corporation may enter into an indemnity agreement with any director, officer, employee or agent of the Corporation, or of another corporation, partnership, joint venture, trust or other enterprise, upon terms and conditions that the Board of Directors deems appropriate, as long as the provisions of the agreement are not impermissible under applicable law.

6. Any amendment or repeal of this Article VIII, Section II, shall not be retroactive in effect.

7. In case any provision in this Article VIII, Section II, shall be determined at any time to be unenforceable in any respect, the other provisions shall not in any way be affected or impaired thereby, and the affected provision shall be given the fullest possible enforcement in the circumstances, it being the intention of the Corporation to afford indemnification and advancement of expenses to the persons indemnified hereby to the fullest extent permitted by law.

8. The Corporation may, by action of the Board of Directors, authorize one or more officers to grant rights to indemnification and advancement of expenses to employees or agents of the Corporation on such terms and conditions as such officer or officers deem appropriate under the circumstances.

IN WITNESS WHEREOF, the undersigned has caused this Restated Certificate of Incorporation to be executed in its corporate name by its this day of , 199.

Name: Title:

# FORM OF BYLAWS OF CONECTIV

1.1.1

# 1. OFFICES.

1.1 Offices. In addition to its registered office in the State of Delaware, the Corporation shall have a corporate office in Wilmington, Delaware and a significant presence in New Jersey, and such other offices, either within or without the State of Delaware, at such locations as the Board of Directors may from time to time determine or the business of the Corporation may require.

# 2. SEAL.

2.1 Seal. The Corporation shall have a seal, which shall have inscribed thereon its name and year of incorporation and the words, "Corporate Seal Delaware."

#### 3. MEETINGS OF STOCKHOLDERS.

3.1 Annual Meetings. The annual meeting of stockholders of the Corporation shall be held on such date, at such time and at such place within or without the State of Delaware as shall be determined by the Board of Directors from time to time.

3.2 Special Meetings. Special meetings of the stockholders of the Corporation shall be held on such date, at such time and at such place within or without the State of Delaware as the Board of Directors may designate.

3.3 Notice of Meetings. (a) Notices of meetings of stockholders shall be in writing and shall state the place, date and hour of the meeting, and, in the case of a special meeting, the purpose or purposes for which a meeting is called. No business other than that specified in the notice thereof shall be transacted at any special meeting.

(b) Such notice shall either be delivered personally or mailed, postage prepaid, to each stockholder entitled to vote at such meeting not less than 10 nor more than 60 days before the date of the meeting. If mailed, the notice shall be directed to the stockholder at his or her address as it appears on the records of the Corporation. Personal delivery of any such notice to any officer of a corporation or association or to any member of a partnership shall constitute delivery of such notice to such corporation, association or partnership.

(c) Notice of any meeting of stockholders need not be given to any stockholder if waived by such stockholder in writing, whether before or after such meeting is held, or if such stockholder shall sign the minutes or attend the meeting, except that if such stockholder attends a meeting for the express purpose of objecting at the beginning of the meeting to the transaction of any business because the meeting is not lawfully called or convened, such stockholder shall not be deemed to have waived notice of such meeting.

3.4 Adjourned Meetings. When a meeting is adjourned to another time or place, unless otherwise provided by these Bylaws, notice need not be given of the adjourned meeting if the time and place thereof are announced at the meeting at which the adjournment is taken. At the adjourned meeting the stockholders may transact any business which might have been transacted at the original meeting. If an adjournment is for more than 30 days, or if after an adjournment, a new record date is fixed for the adjourned meeting, a notice of the adjourned meeting shall be given to each stockholder entitled to vote at the meeting.

3.5 Quorum and Adjournment. Except as otherwise provided by law, by the Certificate of Incorporation of the Corporation or by these Bylaws, the presence, in person or by proxy, of the holders of a majority of the aggregate voting power of the stock issued and outstanding, entitled to vote thereat, shall constitute a quorum for the transaction of business at all meetings of stockholders. If such majority shall not be present or represented at any meeting of stockholders, the stockholders present, although less than a quorum, shall have the power to adjourn the meeting.

3.6 Vote Required. Except as otherwise provided by law or by the Certificate of Incorporation:

(a) Directors shall be elected by a plurality of the votes present in person or represented by proxy at a meeting of stockholders and entitled to vote in the election of directors, and

(b) whenever any corporate action other than the election of Directors is to be taken, it shall be authorized by a majority in voting power of the shares present in person or by proxy at a meeting of stockholders and entitled to vote on the subject matter.

3.7 *Manner of Voting*. At each meeting of stockholders, each stockholder having the right to vote shall be entitled to vote in person or by proxy. Proxies need not be filed with the Secretary of the Corporation until the meeting is called to order, but shall be filed before being voted. Each stockholder shall be entitled to vote each share of stock having voting power registered in his name on the books of the Corporation on the record date fixed for determination of stockholders entitled to vote at such meeting. All elections of Directors by stockholders shall be by written ballot.

3.8 *Proxies*. (a) At any meeting of stockholders, any stockholder may be represented and vote by proxy or proxies appointed by a written form of proxy. In the event that any form of proxy shall designate two or more persons to act as proxies, a majority of such persons present at the meeting or, if only one shall be present, then that one shall have and may exercise all of the powers conferred by the form of proxy upon all of the persons so designated unless the form of proxy shall otherwise provide.

(b) The Board of Directors may, in advance of any annual or special meeting of the stockholders, prescribe additional regulations concerning the manner of execution and filing of proxies and the validation of the same, which are intended to be voted at any such meeting.

3.9 Presiding Officer and Secretary. The Chairman of the Board shall act as chairman of all meetings of the stockholders. In the absence of the Chairman of the Board, the Vice Chairman of the Board or, in his or her absence, the Chief Executive Officer or, in his or her absence, the President or, in his or her absence, any Vice President designated by the Board of Directors shall act as chairman of the meeting.

The Secretary of the Corporation shall act as secretary of all meetings of the stockholders, but, in the absence of the Secretary, the Assistant Secretary designated in accordance with Section 5.11(b) of these Bylaws shall act as secretary of all meetings of the stockholders, but in the absence of a designated Assistant Secretary, the chairman of the meeting may appoint any person to act as secretary of the meeting.

3.10 *Procedure*. At each meeting of stockholders, the chairman of the meeting shall fix and announce the date and time of the opening and the closing of the polls for each matter upon which the stockholders will vote at the meeting and shall determine the order of business and all other matters of procedure. Except to the extent inconsistent with any such rules and regulations as adopted by the Board of Directors, the chairman of the meeting may establish rules, which need not be in writing, to maintain order and safety and for the conduct of the meeting. Without limiting the foregoing, he or she may:

(a) restrict attendance at any time to bona fide stockholders of record and their proxies and other persons in attendance at the invitation of the chairman;

(b) restrict dissemination of solicitation materials and use of audio or visual recording devices at the meeting;

(c) adjourn the meeting without a vote of the stockholders, whether or not there is a quorum present; and

(d) make rules governing speeches and debate, including time limits and access to microphones.

The chairman of the meeting acts in his or her absolute discretion and his or her rulings are not subject to appeal.

# 4. DIRECTORS.

4.1 *Powers*. The Board of Directors shall exercise all of the powers of the Corporation except such as are by law, or by the Certificate of Incorporation of this Corporation or by these Bylaws conferred upon or reserved to the stockholders of any class or classes.

4.2 *Resignations*. Any Director may resign at any time by giving written notice to the Board of Directors or the Secretary. Such resignation shall take effect at the date of receipt of such notice or at any later time specified therein. Acceptance of such resignation shall not be necessary to make it effective.

4.3 Presiding Officer and Secretary. The Chairman of the Board shall act as chairman of all meetings of the Board of Directors. In the absence of the Chairman of the Board, the Vice Chairman of the Board, or in his absence, the Chief Executive Officer or other person designated by the Board of Directors shall act as chairman of the meeting.

The Secretary of the Corporation shall act as secretary of all meetings of the Board of Directors, but, in the absence of the Secretary, the Assistant Secretary designated in accordance with Section 5.11(b) of these Bylaws shall act as secretary of all meetings of the stockholders, but in the absence of a designated Assistant Secretary, the chairman of the meeting may appoint any person to act as secretary of the meeting.

4.4 Annual Meetings. The Board of Directors shall meet each year immediately following the annual meeting of stockholders, at the place where such meeting of stockholders has been held, or at such other place as shall be fixed by the person presiding over the meeting of the stockholders, for the purpose of election of officers and consideration of such other business as the Board of Directors considers relevant to the management of the Corporation.

4.5 Regular Meetings. Regular meetings of the Board of Directors shall be held on such dates and at such times and places, within or without the state of Delaware, as shall from time to time be determined by the Board of Directors. In the absence of any such determination, such meetings shall be held at such times and places, within or without the State of Delaware, as shall be designated by the Chairman of the Board on not less than twelve hours notice to each Director, given verbally or in writing either personally, by telephone (including by message or recording device), by facsimile transmission, by telegram or by telex or on not less than three (3) calendar days' notice to each Director given by mail.

4.6 Special Meetings. Special meetings of the Board of Directors shall be held at the call of the Chairman of the Board at such times and places, within or without the State of Delaware, as he or she shall designate, on not less than twelve hours notice to each Director, given verbally or in writing either personally, by telephone (including by message or recording device), by facsimile transmission, by telegram or by telex or on not less than three (3) calendar days' notice to each Director given by mail. Special meetings shall be called by the Secretary on like notice at the written request of a majority of the Directors then in office.

4.7 Quorum and Powers of a Majority. At all meetings of the Board of Directors and of each committee thereof, a majority of the members shall be necessary and sufficient to constitute a quorum for the transaction of business, and the act of a majority of the members present at any meeting at which a quorum is present shall be the act of the Board of Directors or such committee, unless by express provision of law, of the Certificate of Incorporation or these Bylaws, a different vote is required, in which case such express provision shall govern and control. In the absence of a quorum, a majority of the members present at any

meeting may, without notice other than announcement at the meeting, adjourn such meeting from time to time until a quorum is present.

4.8 Waiver of Notice. Notice of any meeting of the Board of Directors, or any committee thereof, need not be given to any member if waived by him or her in writing, whether before or after such meeting is held, or if he or she shall sign the minutes or attend the meeting, except that if such Director attends a meeting for the express purpose of objecting at the beginning of the meeting to the transaction of any business because the meeting is not lawfully called or convened, then such Director shall not be deemed to have waived notice of such meeting.

4.9 *Manner of Acting.* (a) Members of the Board of Directors, or any committee thereof, may participate in any meeting of the Board of Directors or such committee by means of conference telephone or similar communications equipment by means of which all persons participating therein can hear each other, and participation in a meeting by such means shall constitute presence in person at such meeting.

(b) Any action required or permitted to be taken at any meeting of the Board of Directors or any committee thereof may be taken without a meeting if all members of the Board of Directors or such committee, as the case may be, consent thereto in writing, and the writings are filed with the minutes of proceedings of the Board of Directors or such committee.

4.10 *Compensation*. (a) The Board of Directors, by a resolution or resolutions, may fix, and from time to time change, the compensation of Directors.

(b) Each Director shall be entitled to reimbursement from the Corporation for his or her reasonable expenses incurred with respect to duties as a member of the Board of Directors or any committee thereof.

(c) Nothing contained in these Bylaws shall be construed to preclude any Director from serving the Corporation in any other capacity and from receiving compensation from the Corporation for service rendered to it in such other capacity.

4.11 Committees. The Board of Directors may designate one or more committees, each committee to consist of one or more Directors, which to the extent provided in said resolution or resolutions shall have and may exercise the powers and authority of the Board of Directors in the management of the business and affairs of the corporation; provided, however, that no such committee shall have the power or authority in reference to the following matters: (i) approving or adopting, or recommending to the stockholders, any action or matter expressly required by the General Corporation Law of Delaware (the "GCLD") to be submitted to stockholders for approval or (ii) adopting, amending, or repealing any bylaw of the corporation. The Board of Directors may designate one or more directors as alternate members of any committee, who may replace any absent or disqualified member of the committee. In the absence or disqualification of a member of a committee, the member or not such member of members constitute a quorum, may unanimously appoint another member of the board of directors to act as at the meeting in place of such absent of disqualified director.

4.12 Committee Procedure, Limitations of Committee Powers. (a) Except as otherwise provided by these Bylaws, each committee shall adopt its own rules governing the time, place and method of holding its meetings and the conduct of its proceedings and shall meet as provided by such rules or by resolution of the Board of Directors. Unless otherwise provided by these Bylaws or any such rules or resolutions, notice of the time and place of each meeting of a committee shall be given to each member of such committee as provided in Section 4.6 of these Bylaws with respect to notices of special meetings of the Board of Directors.

(b) Each committee shall keep regular minutes of its proceedings and report the same to the Board of Directors when required.

(c) Any member of any committee may be removed from such committee either with or without cause, at any time, by the Board of Directors at any meeting thereof. Any vacancy in any committee shall be filled by the Board of Directors in the manner prescribed by the Certificate of Incorporation or these Bylaws for the original appointment of the members of such committee.

1. E

## 5. OFFICERS.

5.1 Number. (a) The officers of the Corporation shall include a Chief Executive Officer, a President, one or more Vice Presidents (including one or more Executive Vice Presidents and one or more Senior Vice Presidents if deemed appropriate by the Board of Directors), a Secretary and a Treasurer. The Board of Directors shall also elect a Chairman of the Board and may elect a Vice Chairman of the Board. The Board of Directors may also elect such other officers as the Board of Directors may from time to time deem appropriate or necessary. Except for the Chairman of the Board, the Vice Chairman of the Board and the Chief Executive Officer, none of the officers of the Corporation need be a director of the Corporation. Any two or more offices may be held by the same person to the extent permitted by the GCLD.

(b) The Board of Directors may delegate to the Chief Executive Officer or President the power to appoint one or more employees of the Corporation as divisional or departmental vice presidents and fix the duties of such appointees. However, no such divisional or departmental vice president shall be considered as an officer of the Corporation, the officers of the Corporation being limited to those officers elected by the Board of Directors.

5.2 *Election of Officers, Qualification and Term.* The officers of the Corporation shall be elected from time to time by the Board of Directors and, except as may otherwise be expressly provided in a contract of employment duly authorized by the Board of Directors or the Merger Agreement, shall hold office at the pleasure of the Board of Directors.

5.3 *Removal*. Except as otherwise expressly provided in the Merger Agreement, any officer elected by the Board of Directors may be removed, either with or without cause, by the Board of Directors at any meeting thereof, or to the extent delegated to the Chairman of the Board or the Chief Executive Officer, by the Chairman of the Board or the Chief Executive Officer.

5.4 *Resignations*. Any officer of the Corporation may resign at any time by giving written notice to the Board of Directors or to the Chairman of the Board or to the Chief Executive Officer. Such resignation shall take effect at the date of the receipt of such notice or at any later time specified therein and, unless otherwise specified therein, the acceptance of such resignation shall not be necessary to make it effective.

5.5 *Salaries*. The salaries of all officers of the Corporation shall be fixed by the Board of Directors from time to time, and no officer shall be prevented from receiving such salary by reason of the fact that he is also a Director of the Corporation.

5.6 The Chairman of the Board. The Chairman of the Board shall have the powers and duties customarily and usually associated with the office of the Chairman of the Board. The Chairman of the Board shall preside at meetings of the stockholders and of the Board of Directors.

5.7 Vice Chairman of the Board. The Vice Chairman of the Board shall have the powers and duties customarily and usually associated with the office of the Vice Chairman of the Board.

5.8 Chief Executive Officer. The Chief Executive Officer shall have, subject to the supervision, direction and control of the Board of Directors, the general powers and duties of supervision, direction and management of the affairs and business of the Corporation usually vested in the chief executive officer of a corporation, including, without limitation, all powers necessary to direct and control the organizational and reporting relationships within the Corporation. If at any time the office of the Chairman of the Board and the Vice Chairman of the Board shall not be filled, or in the event of the temporary absence or disability of

the Chairman of the Board and the Vice Chairman of the Board, the Chief Executive Officer shall have the powers and duties of the Chairman of the Board.

5.9 *The President*. The President shall serve as chief operating officer and shall have such other powers and perform such other duties as may be delegated to him or her from time to time by the Board of Directors or the Chief Executive Officer.

5.10 *The Vice Presidents*. Each Vice President shall have such powers and perform such duties as may from time to time be assigned to him or her by the Board of Directors, the Chief Executive Officer or the President.

5.11 *The Secretary and the Assistant Secretary*. (a) The Secretary shall attend meetings of the Board of Directors and meetings of the stockholders and record all votes and minutes of all such proceedings in a book kept for such purpose. He or she shall have all such further powers and duties as generally are incident to the position of Secretary or as may from time to time be assigned to him or her by the Board of Directors, the Chief Executive Officer or the President.

(b) Each Assistant Secretary shall have such powers and perform such duties as may from time to time be assigned to him or her by the Board of Directors, the Chief Executive Officer, the President or the Secretary. In case of the absence or disability of the Secretary, the Assistant Secretary designated by the Chief Executive Officer (or, in the absence of such designation, by the Secretary) shall perform the duties and exercise the powers of the Secretary.

5.12 The Treasurer and the Assistant Treasurer. (a) The Treasurer shall have custody of the Corporation's funds and securities and shall keep full and accurate accounts of receipts and disbursements in books belonging to the Corporation and shall deposit or cause to be deposited moneys or other valuable effects in the name and to the credit of the Corporation in such depositories as may be designated by the Board of Directors. The Treasurer shall also maintain adequate records of all assets, liabilities and transactions of the Corporation and shall see that adequate audits thereof are currently and regularly made. The Treasurer shall have such other powers and perform such other duties that generally are incident to the position of Treasurer or as may from time to time be assigned to him or her by the Board of Directors, the Chief Executive Officer or the President.

(b) Each Assistant Treasurer shall have such powers and perform such duties as may from time to time be assigned to him or her by the Board of Directors, the Chief Executive Officer, the President or the Treasurer. In case of the absence or disability of the Treasurer, the Assistant Treasurer designated by the Chief Executive Officer (or, in the absence of such designation, by the Treasurer) shall perform the duties and exercise the powers of the Treasurer.

## 6. STOCK.

6.1 Certificates. Certificates for shares of stock of the Corporation shall be issued under the seal of the Corporation, or a facsimile thereof, and shall be numbered and shall be entered in the books of the Corporation as they are issued. Each certificate shall bear a serial number, shall exhibit the holder's name and the number of shares evidenced thereby, and shall be signed by the Chairman of the Board or a Vice Chairman, if any, or the Chief Executive Officer or the President or any Vice President, and by the Secretary or an Assistant Secretary or the Treasurer or an Assistant Treasurer. Any or all of the signatures on the certificate may be a facsimile. In case any officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate shall have ceased to be such officer, transfer agent or registrar before such certificate is issued, it may be issued by the corporation with the same effect as if such person or entity were such officer, transfer agent or registrar at the date of issue.

6.2 Transfers. Transfers of stock of the Corporation shall be made on the books of the Corporation only upon surrender to the Corporation of a certificate (if any) for the shares duly endorsed or accompanied by proper evidence of succession, assignment or authority to transfer, provided such

succession, assignment or transfer is not prohibited by the Certificate of Incorporation, these Bylaws, applicable law or contract. Thereupon, the Corporation shall issue a new certificate (if requested) to the person entitled thereto, cancel the old certificate (if any) and record the transaction upon its books.

6.3 Lost, Stolen or Destroyed Certificates. Any person claiming a certificate of stock to be lost, stolen or destroyed shall make an affidavit or an affirmation of that fact, and shall give the Corporation a bond of indemnity in satisfactory form and with one or more satisfactory sureties, whereupon a new certificate (if requested) may be issued of the same tenor and for the same number of shares as the one alleged to be lost, stolen or destroyed.

6.4 *Registered Stockholders*. The Corporation shall be entitled to recognize the exclusive right of a person registered on its books as the owner of shares as the person entitled to exercise the rights of a stockholder and shall not be bound to recognize any equitable or other claim to or interest in any such shares on the part of any other person, whether or not it shall have express or other notice thereof, except as otherwise expressly provided by the GCLD.

6.5 Additional Powers of the Board. (a) In addition to those powers set forth in Section 4.1, the Board of Directors shall have power and authority to make all such rules and regulations as it shall deem expedient concerning the issue, transfer and registration of certificates for shares of stock of the Corporation, including the use of uncertificated shares of stock subject to the provisions of the GCLD.

(b) The Board of Directors may appoint and remove transfer agents and registrars of transfers, and may require all stock certificates to bear the signature of any such transfer agent and/or any such registrar of transfers.

#### 7. MISCELLANEOUS.

7.1 *Place and Inspection of Books*. (a) The books of the Corporation other than such books as are required by law to be kept within the State of Delaware shall be kept in such place or places either within or without the State of Delaware as the Board of Directors may from time to time determine.

(b) At least ten days before each meeting of stockholders, the officer in charge of the stock ledger of the Corporation shall prepare a complete list of the stockholders entitled to vote at the meeting, arranged in alphabetical order and showing the address of each stockholder and the number of shares registered in the name of each stockholder. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting, during ordinary business hours, for a period of at least ten days prior to the meeting, either at a place within the city where the meeting is to be held, which place shall be specified in the notice of the meeting, or, if not specified, at the place where the meeting is to be held. The list shall also be produced and kept at the time and place of the meeting during the whole time thereof, and may be inspected by any stockholder who is present.

(c) The Board of Directors shall determine from time to time whether and, if allowed, when and under what conditions and regulations the accounts and books of the Corporation (except such as may be by law specifically open to inspection or as otherwise provided by these Bylaws) or any of them shall be open to the inspection of the stockholders and the stockholders' rights in respect thereof.

7.2 Voting Shares in Other Corporations. The Chief Executive Officer, the President or any other officer of the Corporation designated by the Board of Directors may vote any and all shares held by the Corporation in any other corporation.

7.3 Fiscal Year. The fiscal year of the Corporation shall be such fiscal year as the Board of Directors from time to time by resolution shall determine.

7.4 Gender/Number. As used in these Bylaws, the masculine, feminine or neuter gender, and the singular or plural number, shall each include the others whenever the context so indicates.





7.5 Paragraph Titles. The titles of the paragraphs have been inserted as a matter of reference only and shall not control or affect the meaning or construction of any of the terms and provisions hereof.

7.6 Amendment. These Bylaws may be altered, amended or repealed by (a) the affirmative vote of 80% or more of the aggregate number of votes that the holders of the then outstanding shares of Common Stock and Preferred Stock are entitled to cast on the amendment, or (b) by resolution adopted by the affirmative vote of not less than a majority of the Directors in office, at any annual or regular meeting of the Board of Directors or at any special meeting of the Board of Directors if notice of the proposed alteration, amendment or repeal be contained in written notice of such special meeting. Notwithstanding the foregoing, the amendment of any provision of these Bylaws requiring an affirmative vote in excess of a majority of the Directors in office shall require the affirmative vote of at least the number of directors the affirmative vote of whom is required by such provision.

7.7 Certificate of Incorporation. Notwithstanding anything to the contrary contained herein, if any provision contained in these Bylaws is inconsistent with or conflicts with a provision of the Certificate of Incorporation, such provision of these Bylaws shall be superseded by the inconsistent provision in the Certificate of Incorporation to the extent necessary to give effect to such provision in the Certificate of Incorporation.

## ILLUSTRATION OF CERTAIN CLASS A COMMON STOCK TERMS

The following illustrations demonstrate the method of calculating the Intergroup Interest in the Atlantic Utility Group and certain other items following the issuance of Class A Common Stock to the holders of Atlantic Common Stock pursuant to the Merger Agreement and, subsequent thereto, in a range of transactional contexts. The illustrations use 40 million shares as the number of authorized shares of Class A Common Stock, 6 million shares as the number of shares of Class A Common Stock issued to the holders of Atlantic Common Stock pursuant to the Atlantic Merger, and 14 million shares as the Number of Shares Issuable with Respect to the Intergroup Interest in the Atlantic Utility Group upon consummation of the Mergers. Unless otherwise specified, each illustration below should be read independently as if none of the other transactions illustrated in this Annex VI had occurred. Capitalized terms used herein have the respective meanings assigned to them in the Joint Proxy Statement/Prospectus.

The following are illustrations only and are not intended to reflect either the results of the Mergers or transactions that are anticipated to occur following the Mergers. Nor do the illustrations indicate all of the consequences of the transactions that are illustrated. The illustrations are qualified in their entirety by the more detailed information contained in the Joint Proxy Statement/Prospectus and the other Annexes thereto, including Annex IV, which sets forth the terms of the Class A Common Stock.

At any given time, the Outstanding Atlantic Utility Fraction, which is the percentage interest in the equity value of the Company attributable to the Atlantic Utility Group that is represented by the outstanding shares of Class A Common Stock, would equal:

OUTSTANDING SHARES OF CLASS A COMMON STOCK

OUTSTANDING SHARES OF CLASS A COMMON STOCK + NUMBER OF SHARES ISSUABLE WITH RESPECT TO THE INTERGROUP INTEREST

The balance of the equity value of the Company attributable to the Atlantic Utility Group is represented by the Residual Group's Intergroup Interest in the Atlantic Utility Group. At any given time, the Intergroup Interest Fraction, which is the percentage interest in the equity value of the Company attributable to the Atlantic Utility Group that is retained by the Residual Group, would equal:

NUMBER OF SHARES ISSUABLE	WITH RESPECT TO THE INTERGROUP INTEREST
OUTSTANDING SHARES	NUMBER OF SHARES ISSUABLE WITH RESPECT
OF CLASS A COMMON STOCK +	TO THE INTERGROUP INTEREST

The sum of the Outstanding Atlantic Utility Fraction and the Intergroup Interest Fraction will at all times equal 100%.

#### **Consummation of the Mergers**

The following illustration shows how the foregoing items are calculated after giving effect to the initial issuance by the Company of Class A Common Stock to the holders of Atlantic Common Stock pursuant to the Merger Agreement.

• The Number of Shares Issuable with Respect to the Intergroup Interest will be 14 million immediately following the consummation of the Mergers. As a result, the Outstanding Atlantic Utility Fraction will be 30%, calculated as follows:

6 million

6 million + 14 million

- The Intergroup Interest Fraction would accordingly represent an interest of 70% in the Atlantic Utility Group.
- In the event of a dividend or other distribution on the Class A Common Stock (other than a dividend or other distribution payable in shares of Class A Common Stock or in securities of the Company attributed to the Atlantic Utility Group), the Residual Group would be credited, and the Atlantic Utility Group would be charged, with 2<sup>1</sup>/<sub>3</sub> (representing the ratio of the Intergroup Interest Fraction (70%) to the Outstanding Atlantic Utility Fraction (30%)) times the amount with respect to such distribution. If, for example, a dividend of \$1 per share were declared and paid on the 6 million shares of Class A Common Stock outstanding (an aggregate of \$6 million), the Residual Group would be credited with \$14 million, and the Atlantic Utility Group would be charged with that amount in addition to the \$6 million dividend on the outstanding shares of Class A Common Stock (a total of \$20 million). An example of the effects of a distribution of Class A Common Stock is set forth below under the caption "Class A Common Stock Dividends."
- Immediately after the consummation of the Mergers, the Company would have 34 million authorized and unissued shares of Class A Common Stock remaining (40 million minus 6 million issued and outstanding).

## Repurchases of Class A Common Stock

The following two illustrations reflect the repurchase by the Company of 2 million issued and outstanding shares of Class A Common Stock, which are then retired or otherwise cease to be outstanding following their purchase.

#### Repurchase with Atlantic Utility Group Funds

Assume all of the repurchased shares are identified by the Company Board were purchased with funds attributed to the Atlantic Utility Group, with the Atlantic Utility Group being charged with the consideration paid for such shares.

Shares of Class A Common Stock Outstanding	
Shares previously issued and outstanding	6 million
Shares purchased	2 million
Total shares issued and outstanding after purchase	4 million

- The Number of Shares Issuable with Respect to the Intergroup Interest would not be changed by the purchase of Class A Common Stock with funds attributed to the Atlantic Utility Group but would continue to be 14 million.
- The Outstanding Atlantic Utility Fraction would be reduced to approximately 22.2%, calculated as follows:

# $\frac{4 \text{ million}}{4 \text{ million} + 14 \text{ million}}$

- The Intergroup Interest Fraction would accordingly represent an interest of approximately 77.8% in the Atlantic Utility Group.
- In the event of a dividend or other distribution on the Class A Common Stock immediately following the repurchase (other than a dividend or other distribution payable in shares of Class A Common Stock or in securities of the Company attributed to the Atlantic Utility Group), the Residual Group would be credited, and the Atlantic Utility Group would be charged, with an

amount equal to 3.5 (representing the ratio of the Intergroup Interest Fraction (77.8%) to the Outstanding Atlantic Utility Fraction (22.2%)) times the amount of such distribution.

• The Company would have 36 million authorized and unissued shares of Class A Common Stock (40 million minus 4 million issued and outstanding).

## Repurchase with Residual Group Funds

Assume all of the repurchased shares identified by the Company Board were purchased with funds attributed to the Residual Group, with the Residual Group being charged with the consideration paid for such shares.

	· ·	1 ·	· · · ·	·
• Shares previously issued and o	utstanding	••••••	••••••••••	6 million
Shares purchased	• • • • • • • • • • • • • • • • • •	•••••	•••••••••••••••••••••••••••••••••••••••	2 million
Total shares issued and outstar	nding after purchase		· · · · · · · · · · · · · · · · · · ·	4 million

• The Number of Shares Issuable with Respect to the Intergroup Interest would be increased by the number of shares of Class A Common Stock so purchased because the purchase was effected with funds attributed to the Residual Group.

Number of Shares Issuable with Respect to the Intergroup Interest prior to	
purchase	14 million
Shares purchased	
Number of Shares Issuable with Respect to the Intergroup Interest after purchase	16 million

• The Outstanding Atlantic Utility Fraction would be reduced to 20%, calculated as follows:

4 million

## 4 million + 16 million

- The Intergroup Interest Fraction would accordingly represent an interest of 80% in the Atlantic Utility Group.
- In the event of a dividend or other distribution on the Class A Common Stock immediately following the repurchase (other than a dividend or other distribution payable in shares of Class A Common Stock or in securities of the Company attributed to the Atlantic Utility Group), the Residual Group would be credited, and the Atlantic Utility Group would be charged, with an amount equal to 4.0 (representing the ratio of the Intergroup Interest Fraction (80%) to the Outstanding Atlantic Utility Fraction (20%)) times the amount of such distribution.

• The Company would have 36 million authorized and unissued shares of Class A Common Stock (40 million minus 4 million issued and outstanding).

#### Transfers of Assets Between the Residual Group and the Atlantic Utility Group

Contribution of Assets from the Residual Group to the Atlantic Utility Group

The following illustration reflects the contribution, by the Residual Group to the Atlantic Utility Group with respect to the Intergroup Interest, of \$20 million of assets previously attributed to the Residual Group, on a date on which the Market Value of the Class A Common Stock is \$20 per share.

ø	Shares previously issued and outstanding    6 million      Newly issued shares    0
	Total shares issued and outstanding after contribution
•	The Number of Shares Issuable with Respect to the Intergroup Interest would be increased to reflect the contribution to the Atlantic Utility Group of assets theretofore attributed to the Residual Group.
	Number of Shares Issuable with Respect to the Intergroup Interest prior to contribution
	Adjustment to reflect contribution to the Atlantic Utility Group of assets attributed         to the Residual Group (\$20 million divided by \$20)
; +	Number of Shares Issuable with Respect to the Intergroup Interest after contribution       15 million
•	The Outstanding Atlantic Utility Fraction would be reduced to approximately 28.6%, calculated as follows:

6 million

#### 6 million + 15 million

- The Intergroup Interest Fraction would accordingly represent an interest of approximately 71.4% in the Atlantic Utility Group.
- In the event of a dividend or other distribution on the Class A Common Stock immediately following the contribution (other than a dividend or other distribution payable in shares of Class A Common Stock or in securities of the Company attributed to the Atlantic Utility Group), the Residual Group would be credited, and the Atlantic Utility Group would be charged, with an amount equal to 2.5 (representing the ratio of the Intergroup Interest Fraction (approximately 71.4%) to the Outstanding Atlantic Utility Fraction (approximately 28.6%)) times the amount of such distribution.
- The Company would have 34 million authorized and unissued shares of Class A Common Stock (40 million minus 6 million issued and outstanding).

#### Transfer of Assets from the Atlantic Utility Group to the Residual Group

The following illustration reflects the transfer, from the Atlantic Utility Group to the Residual Group in reduction of the Intergroup Interest, of \$20 million of assets attributed to the Atlantic Utility Group, on a date on which the Market Value of the Class A Common Stock is \$20 per share.

• Shares previously issued and outstanding	6 million
Shares repurchased	0
Total shares issued and outstanding after transfer	6 million

• The Number of Shares Issuable with Respect to the Intergroup Interest would be decreased to reflect the contribution to the Residual Group of assets theretofore attributed to the Atlantic Utility Group.

Number of Shares Issuable with Respect to the Intergroup Interest prior to transfer ... 14 million Adjustment to reflect transfer to the Residual Group of assets attributed to the Atlantic Utility Group (\$20 million divided by \$20) ..... 1 million

Number of Shares Issuable with Respect to the Intergroup Interest after transfer ... 13 million

• The Outstanding Atlantic Utility Fraction would be approximately 31.6%, calculated as follows:

6 million

6 million + 13 million

- The Intergroup Interest Fraction would accordingly represent an interest of approximately 68.4% in the Atlantic Utility Group.
- In the event of a dividend or other distribution on the Class A Common Stock immediately following the transfer (other than a dividend or other distribution payable in shares of Class A Common Stock or in securities of the Company attributed to the Atlantic Utility Group), the Residual Group would be credited, and the Atlantic Utility Group would be charged, with an amount equal to 2½ (representing the ratio of the Intergroup Interest Fraction (approximately 68.4%) to the Outstanding Atlantic Utility Fraction (approximately 31.6%)) times the amount of such distribution.
- The Company would have 34 million authorized and unissued shares of Class A Common Stock (40 million minus 6 million issued and outstanding).
- The Atlantic Utility Group may not make transfers of assets to the Residual Group in reduction of the Intergroup Interest if the effect would be to reduce the Number of Shares Issuable with Respect to the Intergroup Interest to less than zero. The Atlantic Utility Group cannot have an interest in the Residual Group corresponding to the Intergroup Interest.

#### **Future Offerings of Class A Common Stock**

The following illustrations reflect the issuance and sale by the Company of 6 million shares of Class A Common Stock on a date on which the Number of Shares Issuable with Respect to the Intergroup Interest is 6 million.

#### Offering for the Atlantic Utility Group

Assume all the issued and sold shares are identified as issued and sold for the account of the Atlantic Utility Group, with the net proceeds credited to the Atlantic Utility Group.

Shares previously issued and outstanding	 6 million
Newly issued shares	 6 million
Total shares issued and outstanding after offering	 12 million

• The Number of Shares Issuable with Respect to the Intergroup Interest would not be changed by the issuance of any shares of Class A Common Stock for the account of the Atlantic Utility Group and would remain equal to 14 million.

• The Outstanding Atlantic Utility Fraction would be approximately 46.2%, calculated as follows:

12 million 12 million + 14 million

- The Intergroup Interest Fraction would accordingly represent an interest of approximately 53.8% in the Atlantic Utility Group.
- The Company would have 28 million authorized and unissued shares of Class A Common Stock remaining (40 million minus 12 million issued and outstanding).

## Offering for the Residual Group

يعقن المراجع المراجع

Assume all of the shares issued and sold are identified as issued and sold for the account of the Residual Group with respect to the Intergroup Interest, with the net proceeds credited to the Residual Group.

St. . . .

·	Shares previously issued and outstanding	6 million
	Newly issued shares	<u>6 million</u>
	Total shares issued and outstanding after offering	<u>12 million</u>

- The Number of Shares Issuable with Respect to the Intergroup Interest would decrease by the number of shares of Class A Common Stock issued for the account of the Residual Group.
  - Number of Shares Issuable with Respect to the Intergroup Interest prior to offering
     14 million

     Shares issued in offering
     6 million

     Number of Shares Issuable with Respect to the Intergroup Interest after offering
     8 million
- The Outstanding Atlantic Utility Fraction would be 60%, calculated as follows:

12 million

12 million + 8 million

- The Intergroup Interest Fraction would accordingly represent an interest of 40% in the Atlantic Utility Group.
- The Company would have 28 million authorized and unissued shares of Class A Common Stock remaining (40 million minus 12 million issued and outstanding).

#### **Class A Common Stock Dividends**

The following illustrations reflect dividends of Class A Common Stock on outstanding Class A Common Stock and outstanding Company Common Stock, respectively.

## Class A Common Stock Dividend on Class A Common Stock

Assume the Company declares a dividend of half a share of Class A Common Stock on each outstanding share of Class A Common Stock.

Shares previously issued and outstanding	6 million
Newly issued shares	3 million
Total shares issued and outstanding after dividend	9 million

• The Number of Shares Issuable with Respect to the Intergroup Interest would be increased proportionately to reflect the stock dividend payable in shares of Class A Common Stock to holders of Class A Common Stock (*i.e.*, by an amount equal to (i) the ratio of the Intergroup Interest Fraction to the Outstanding Atlantic Utility Fraction times (ii) the number of shares distributed to the Class A Common Stock holders.

Number of Shares Issuable with Respect to the Intergroup Interest prior to	
dividend	14 million
Adjustment to reflect dividend of shares on outstanding shares of Class A Common	
Stock (2 <sup>1</sup> / <sub>3</sub> times 3 million shares)	7 million
Number of Shares Issuable with Respect to the Intergroup Interest after dividend	21 million

• The Outstanding Atlantic Utility Fraction would continue to be 30%, calculated as follows:

9 million

9	million	÷	21	million

- The Intergroup Interest Fraction would accordingly continue to represent an interest of 70% in the Atlantic Utility Group.
- The Company would have 31 million authorized and unissued shares of Class A Common Stock remaining (40 million minus 9 million issued and outstanding).

## Class A Common Stock Dividend on Company Common Stock

Assume the Company declares a dividend of 0.1 shares of Class A Common Stock on each outstanding share of Company Common Stock, where 30 million shares of Company Common Stock are outstanding.

Shares previously issued and outstanding	6 million
Newly issued shares	3 million
Total shares issued and outstanding after dividend	9 million

• Any dividend of shares of Class A Common Stock on the outstanding shares of Company Common Stock will be treated as reducing the Number of Shares Issuable with Respect to the Intergroup Interest on a share-for-share basis.

Number of Shares Issuable with Respect to the Intergroup Interest prior to

dividend	14 million
Shares distributed on outstanding shares of Company Common Stock	3 million
Number of Shares Issuable with Respect to the Intergroup Interest after dividend	11 million

• The Outstanding Atlantic Utility Fraction would be 45%, calculated as follows:

9 million	
9 million + 11 million	

- The Intergroup Interest Fraction would accordingly represent an interest of 55% in the Atlantic Utility Group.
- The Company would have 31 million authorized and unissued shares of Class A Common Stock remaining (40 million minus 9 million issued and outstanding).

## Dividend, Redemption or Conversion upon a Disposition of Substantially All Assets of Atlantic Utility Group

Assume that a Disposition of all or substantially all of the properties and assets of the Atlantic Utility Group occurs and the Net Proceeds (which includes an adjustment for a capitalized amount of the Notional Fixed Charge) from such Disposition equal \$500 million.

• If the Company elected to pay a dividend to the holders of Class A Common Stock, the aggregate amount of such dividend would be \$150 million (representing the product of the Outstanding Atlantic Utility Fraction and the Net Proceeds of such Disposition), calculated as follows:

## 6 million

 $\frac{\text{o million}}{6 \text{ million} + 14 \text{ million}} \times \$500 \text{ million}$ 

In this case, the Residual Group would be credited, and the Atlantic Utility Group would be charged, with \$350 million, an amount equal to 21/3 (representing the ratio of the Intergroup Interest Fraction (70%) and the Outstanding Atlantic Utility Fraction (30%)), times the amount of such dividend.

• If such Disposition involved all (not merely substantially all) of the properties and assets of the Atlantic Utility Group and the Company elected to redeem all outstanding shares of Class A Common Stock, the aggregate redemption price would be \$150 million (representing the product of the Outstanding Atlantic Utility Fraction and the Net Proceeds of such Disposition), calculated as follows:

## 6 million

-- × \$500 million 6 million + 14 million

In this case, each outstanding share of Class A Common Stock would be redeemed in exchange for \$25 per share (representing the quotient of the aggregate redemption price (\$150 million) and the number of outstanding shares of Class A Common Stock (6 million)).

• If such Disposition involved substantially all (but not all) of the properties and assets of the Atlantic Utility Group and the Company elected to redeem shares of Class A Common Stock, the aggregate

redemption price would be \$150 million (representing to the product of the Outstanding Atlantic Utility Fraction and the Net Proceeds of such Disposition), calculated as follows:

 $\frac{6 \text{ million}}{6 \text{ million} + 14 \text{ million}} \times $500 \text{ million}$ 

In this case, assuming that the average Market Value of one share of Class A Common Stock during the 10-Trading Day period following the consummation of such Disposition is \$30, an aggregate of 5 million (equal to the quotient of the aggregate redemption price and such average Market Value) shares of Class A Common Stock would be redeemed at a redemption price of \$30 per share.

• If the Company elected to convert the Class A Common Stock into Company Common Stock assuming that the Market Value of one share of Class A Common Stock on each Trading Day during the 10-Trading Day period referred to in the preceding paragraph is \$30 and the Market Value of one share of Company Common Stock on each Trading Day during such period is \$40, the Class A Common Stock would be converted into Company Common Stock at a ratio of 0.83 (representing 110% of the average daily ratio during such period of the Market Value of one share of Class A Common Stock to the Market Value of one share of Company Common Stock) shares of Company Common Stock for each share of Class A Common Stock.

#### CONECTIV, INC. INCENTIVE COMPENSATION PLAN

1. Objective. The objective of the Conectiv, Inc. Incentive Compensation Plan (this "Plan") is to support the financial performance of the Company (as defined below) and thereby increase stockholder value. This Plan provides both annual and long-term incentives to reward those officers, key employees, consultants and advisors of the Company and its subsidiaries who, in the opinion of the Committee (as defined below), are responsible for the continued growth, development and financial success of the Company and its subsidiaries. This Plan also is designed to permit the Company and its subsidiaries to hire and retain talented and motivated executive officers and key employees and to increase their ownership in Company Common Stock.

2. *Definitions.* All singular terms defined in this Plan will include the plural and vice versa. As used herein, the following terms will have the meaning specified below:

*"Annual Incentive"* means any Award in cash, Stock or in any other form or type under this Plan, which is based on the performance of the Company and/or the Participant individually or collectively with any other individuals or groups, or any combination thereof, as the Committee, in its sole discretion, deems appropriate, over the prior fiscal year.

"Atlantic" means Atlantic Energy, Inc., a New Jersey corporation.

"Award" means, individually, collectively or in any combination, Annual Incentives, Restricted Stock, Options, Performance Units, Stock Appreciation Rights, Dividend Equivalents, or such other forms or types of Awards as the Committee, in its sole discretion, deems appropriate.

"Beneficial Owner" means "Beneficial Owner" as defined in Rule 13d-3 under the Exchange Act.

"Board" means the Board of Directors of the Company.

"Change in Control" means the occurrence of an event in any one of the following paragraphs:

(a) Any Person is or becomes the Beneficial Owner, directly or indirectly, of securities of the Company (not including in the securities beneficially owned by such Person any securities acquired directly from the Company or its subsidiaries) representing 25% or more of either the then outstanding shares of Stock of the Company or the combined voting power of the Company's then outstanding securities; or

(b) The following individuals cease for any reason to constitute a majority of the number of directors then serving: individuals who constitute the initial Board after the Effective Time and any new director (other than a director whose initial assumption of office is in connection with an actual or threatened election contest, including but not limited to a consent solicitation, relating to the election of directors) whose appointment or election by the Board or nomination for election by the Company's stockholders was approved by a vote of at least two-thirds ( $\frac{2}{3}$ ) of the directors then still in office who either were directors of the initial Board after the Effective Time, or whose appointment, election or nomination for election was previously so approved; or

(c) There is consummated a merger or consolidation of the Company with any other corporation other than (i) a merger or consolidation which would result in the voting securities of the Company outstanding immediately prior to such merger or consolidation continuing to represent (either by remaining outstanding or by being converted into voting securities of the surviving entity or any parent thereof) at least 80% of the combined voting power of the voting securities of the Company or such surviving entity or any parent thereof outstanding immediately after such merger or consolidation, or (ii) a merger or consolidation effected to implement a recapitalization of the Company (or similar transaction) in which no Person is or becomes the Beneficial Owner, directly or indirectly, of securities of the Company (not including in the securities Beneficially Owned by such Person any securities acquired directly from the Company or its subsidiaries) representing 25% or more of either the then outstanding shares of Stock of the Company or the combined voting power of the Company's then outstanding securities; or

(d) The stockholders of the Company approve a plan of complete liquidation or dissolution of the Company or there is consummated an agreement for the sale or disposition by the Company of all or substantially all of the Company's assets, other than a sale or disposition by the Company of all or substantially all of the Company's assets to an entity, at least 80% of the combined voting power of the voting securities of which are owned by Persons in substantially the same proportions as their ownership of the Company immediately prior to such sale.

"Code" means the Internal Revenue Code of 1986, as amended. Reference in this Plan to any section of the Code will be deemed to include any amendments or successor provisions to such section and any regulations promulgated thereunder.

"Committee" means the committee of the Board which has been designated by the Board to administer this Plan or similar plans, which shall be comprised solely of Non-Employee Directors (as such term is defined in Rule 16b-3(b)(3) of the Section 16 Rules), or which otherwise shall meet any disinterested administration or other requirements of the Section 16 Rules and/or Section 162(m) of the Code, each as in effect at the applicable time.

"Company" means Conectiv, Inc., whose name formerly was DS, Inc. and whose name will be changed to Conectiv as of the Effective Time.

"Date of Grant" means the date on which the granting of an Award is authorized by the Committee or such later date as may be specified by the Committee in such authorization.

"Date of Retirement" means the date of Retirement or Early Retirement applicable to a Participant.

"Delmarva" means Delmarva Power & Light Company, a Delaware and Virginia corporation.

"Disability" means the determination that a Participant is "disabled" under the Company disability plan in effect at that time.

"Dividend Equivalent" means an Award granted under Section 14 hereof.

"Early Retirement" means retirement prior to the Normal Retirement Date.

. \* . .

*"Earned Performance Award"* means an actual Award of a specific number of Performance Units (or shares of Restricted Stock, as the context requires) which the Committee has determined have been earned and are payable (or, in the case of Restricted Stock, earned and with respect to which restrictions will lapse) for a particular Performance Period.

"Effective Time" means the Effective Time as defined in the Merger Agreement.

*"Eligible Person"* means any person employed by the Company or a subsidiary of the Company on a regular, contract or temporary basis, or any consultant or advisor, who satisfies all of the requirements of Section 7 hereof.

"Exchange Act" means the Securities Exchange Act of 1934, as amended from time to time.

"Exercise Period" means the period or periods during which a Stock Appreciation Right is exercisable as described in Section 13(b) hereof.

*"Fair Market Value"* means the average of the highest and lowest price at which the Stock was sold in the typical manner on the New York Stock Exchange—Composite Transactions on a specified date, or if such Stock was not traded on the New York Stock Exchange—Composite Transactions on such specified date, on the last date on which such Stock was so traded.

*"Incentive Stock Option"* means an incentive stock option within the meaning of Section 422 of the Code.

"Merger Agreement" means the Agreement and Plan of Merger, dated as of August 9, 1996, as amended and restated as of December 26, 1996, among the Company, Delmarva Power & Light Company, Atlantic Energy, Inc. and DS Sub, Inc., as it shall be amended from time to time prior to the Effective Time.

"Normal Retirement Date" means the earliest, as described in the Pension Plan, when a Participant is entitled to an unreduced retirement benefit under such Pension Plan.

"Option" or "Stock Option" means either a nonqualified stock option or an Incentive Stock Option granted under Section 11 hereof.

"Option Period" means the period or periods during which an Option is exercisable as described in Section 11(e) hereof.

"Participant" means any Eligible Person who has been granted an Award under this Plan.

*"Pension Plan"* means the applicable qualified or nonqualified defined benefit retirement or pension plan of the Company or its subsidiaries, as it may be amended from time to time, which covers the applicable Participant.

"Performance-Based" means that, in determining the amount of a Restricted Stock Award payout, the Committee will take into account the performance of the Company and/or the Participant individually or collectively with any other individuals or groups, or any combination thereof, as the Committee, in its sole discretion, deems appropriate.

"Performance Period" means a period of time, established by the Committee at the time the Award is granted, during which corporate and/or individual performance is measured.

"Performance Unit" means a unit of measurement equivalent to such amount or measure as the Committee, in its sole discretion, deems appropriate at the time the Performance Unit is granted, which may include, but is not limited to, dollars or market value shares.

"Person" means "Person" as defined in Section 3(a)(9) of the Exchange Act, as modified and used in Sections 13(d) and 14(d) thereof (except that such term shall not include (i) the Company or any of its subsidiaries, (ii) a trustee or other fiduciary holding securities under an employee benefit plan of the Company or any of its subsidiaries, (iii) an underwriter temporarily holding securities pursuant to an offering of such securities, or (iv) a corporation owned, directly or indirectly, by the stockholders of the Company in substantially the same proportions as their ownership of stock of the Company).

"Plan Administrator" means the Committee, as set forth in Section 6 hereof.

"Potential Change in Control" means the occurrence of an event in any one of the following paragraphs:

(a) The Company enters into an agreement, the consummation of which would result in the occurrence of a Change in Control; or

(b) The Company or any Person publicly announces an intention to take or to consider taking actions which, if consummated, would constitute a Change in Control; or

(c) Any Person becomes the Beneficial Owner, directly or indirectly, of securities of the Company (not including in the securities Beneficially Owned by such Person any securities acquired directly from the Company or its subsidiaries) representing 15% or more of either the then outstanding shares of Stock of the Company or the combined voting power of the Company's then outstanding securities; or

(d) The Board adopts a resolution to the effect that, for purposes of this Plan, a Potential Change in Control has occurred.

"Reporting Person" means a Participant who is subject to the reporting requirements of Section 16(a) of the Exchange Act.

"Restricted Stock" means an Award, granted conditionally and subject to forfeiture, under Section 10 hereof.

"Retirement" means retirement on or after the Normal Retirement Date.

"Section 16 Rules" means the rules and regulations promulgated by the Securities and Exchange Commission (or any successor) under Section 16 of the Exchange Act. Reference in this Plan to the Section 16 Rules will be deemed to include any amendments or successor provisions thereto.

"Service-Based" means that, in determining the amount of a Restricted Stock Award payout, the Committee will take into account only the period of time that the Participant performed services for the Company or its subsidiaries since the Date of Grant.

"Stock" means the Common Stock, \$.01 par value, of the Company.

"Stock Appreciation Right" means an Award granted under Section 13 hereof.

*"Target Performance Award"* means a targeted Award of a specified number of Performance Units (or shares of Restricted Stock, as the context requires) which may be earned and payable (or, in the case of Restricted Stock, earned and with respect to which restrictions will lapse) based upon the performance objectives for a particular Performance Period, all as determined by the Committee, in its sole discretion. The Target Performance Award will be a factor in the Committee's ultimate determination of the Earned Performance Award.

*"Termination"* means resignation or discharge from employment with the Company or any of its subsidiaries, except in the event of death, Disability, Retirement or Early Retirement.

3. Effective Date and Stockholder Approval. Subject to the approval of this Plan by the affirmative vote of the holders of a majority of Common Stock of Delmarva present in person or by proxy and entitled to vote and a majority of all votes cast by holders of Common Stock of Atlantic entitled to vote at a meeting of its stockholders and conditioned on the consummation of the transactions contemplated by the Merger Agreement, this Plan will be effective at the Effective Time.

4. Period for Grants of Awards. Awards may be made as provided herein for a period of 10 years after the Effective Time.

5. *Termination*. This Plan will continue in effect until all matters relating to the payment of outstanding Awards and administration of this Plan shall have been settled.

6. *Plan Administration.* The Committee is the Plan Administrator and has sole authority (except as otherwise specified herein) to determine all questions of interpretation and application of this Plan, or of the terms and conditions pursuant to which Awards are granted, exercised or forfeited under Plan provisions, and, in general, to make all determinations advisable for the administration of the Plan to achieve its stated objectives. Such determinations by the Committee shall be final and not subject to further appeal.

7. *Eligibility.* Persons eligible to participate shall be limited to those officers and other key employees of the Company and its subsidiaries, and consultants and advisors who, in the sole discretion of the Committee, are in positions in which their decisions, actions and counsel significantly affect the growth, development and financial success of the Company. The eligibility of persons to be Participants shall be in the sole discretion of the Committee and no officer or employee of the Company or its subsidiaries shall have any right to receive any Award under this Plan.

8. Shares of Stock Subject to Awards. The Committee may, from time to time, grant Awards to one or more Eligible Persons; provided that: (a) subject to any adjustment pursuant to Section 24 hereof, the aggregate number of shares of Stock subject to Awards under this Plan may not exceed 5,000,000; (b) to the extent that an Award lapses, is forfeited or the rights of the Participant to whom it was granted otherwise terminate, any shares of Stock subject to such Award again shall be available for the grant of an Award under this Plan; and (c) shares of Stock delivered by the Company under this Plan may be authorized and unissued Stock, Stock held in treasury of the Company or Stock purchased on the open market (including private purchases) in accordance with applicable securities laws.

## 9. Section 162(m) Compliance.

(a) Performance-Based Awards; Covered Executives. Notwithstanding any provisions herein to the contrary, with respect to any Award which is contingent upon the attainment of performance objectives, including, without limitation, Performance-Based Restricted Stock, Performance Units and Annual Incentives (for purposes of this Section 9, "Performance-Based Awards"), granted to an executive of the Company who, in the opinion of the Committee, for a given Performance Period is or is likely to be a "covered employee" within the meaning of Section 162(m) of the Code (for purposes of this Section 9, a "Covered Executive"), the Committee shall establish performance objectives (for purposes of this Section 9, "Performance Goals") with respect to such Awards no later than the earlier of (i) 90 days after commencement of the Performance Period relating to the Performance-Based Award, or (ii) the date on which 25% of the Performance Period relating to the Performance-Based Award will have elapsed.

(b) Performance Criteria: Performance Goals, in the sole discretion of the Committee, may be based on one or more business criteria that relate to the individual, groups of individuals, a product or service line, business unit, division or subsidiary of the Company or the Company as a whole, individually or in any combination (each of which business criteria may be relative to a specified goal, to historical performance of the Company or a product or service line, business unit, division or subsidiary thereof, or to the performance of any other corporation or group of corporations or a product or service line, business unit, division or subsidiary thereof). Performance Goals will be based on one or more of the following criteria: (i) gross, operating or net earnings before or after income taxes; (ii) earnings per share; (iii) book value per share; (iv) cash flow per share; (v) return on equity; (vi) return on investment; (vii) return on assets, employed assets or net assets; (viii) total stockholder return (expressed on a dollar or percentage basis); (ix) return on cash flow; (x) internal rate of return; (xi) cash flow return on investment; (xii) improvements in capital structure; (xiii) residual income; (xiv) gross income, profitability or net income; (xv) price of any Company security; (xvi) sales to customers (expressed on a dollar or percentage basis); (xviii) retention of customers (expressed on a dollar or percentage basis); (xviii) retention of customers (expressed on a dollar or percentage basis); (xviii) retention of customers (expressed on a dollar or percentage basis); (xviii) retention of customers (expressed on a dollar or percentage basis); (xviii) retention of customers (expressed on a dollar or percentage basis); (xviii) retention of customers (expressed on a dollar or percentage basis); (xviii) increase in the Company's or a subsidiary's customer satisfaction ratings (based on a survey conducted by an independent third party);

(xix) economic value added (defined to mean net operating profit minus the cost of capital); (xx) market value added (defined to mean the difference between the market value of debt and equity, and economic book value); (xxi) market share; (xxii) level of expenses; (xxiii) combined ratio; (xxiv) payback period on investment; and (xxv) net present value of investment.

(c) Certification; Maximum Award and Committee Discretion. The Committee shall certify the satisfaction of the foregoing Performance Goals prior to the payment of a Performance-Based Award. No Performance-Based Award with respect to any Covered Executive shall exceed \$3,000,000 (either in cash or in Fair Market Value of Stock as determined on the Date of Grant, as appropriate to a given type of Award) for any Performance-Based Award determined to be payable to a Covered Executive. No Covered Executive may receive more than 5,000,000 in the aggregate of Options, Stock Appreciation Rights and shares of Performance-Based Restricted Stock for the 10-year period during which Awards may be made pursuant to Section 4 hereof.

(d) Deferral of Payment. Regardless of whether provided for in or in conjunction with the grant of the Award, the Committee, in its sole discretion, may defer payment of a Participant's benefit under this Plan if and to the extent that the sum of the Participant's Plan benefit, plus all other compensation paid or payable to the Participant for the fiscal year in which the Plan benefit would otherwise be paid exceeds the maximum amount of compensation that the Company may deduct under Section 162(m) of the Code with respect to the Participant for the year. If deferred by the Committee, such Award benefit and all other compensation paid or payable to the Participant does not exceed the maximum amount of compensation deductible by the Company under Section 162(m).

10. Restricted Stock Awards.

(a) Grants of Restricted Stock. One or more shares of Restricted Stock may be granted conditionally, subject to forfeiture, as determined by the Committee, in its sole discretion, to any person who is an Eligible Person. Certificates evidencing the conditional grant of the Restricted Stock will be issued to the Participant on the Date of Grant without the payment of consideration by the Participant. Such certificates will be issued in the name of the Participant and will bear a restrictive legend prohibiting sale, transfer, pledge or hypothecation of the Restricted Stock until the expiration of the restriction period.

The Committee also may impose such other restrictions and conditions on the Restricted Stock as it, in its sole discretion, deems appropriate, and will designate an Award as Service-Based or Performance-Based.

Upon the issuance to the Participant of the Restricted Stock, the Participant will have the right to vote the Restricted Stock, and if so determined by the Committee, in its sole discretion, to receive cash dividends distributable with respect to such shares, with such dividends treated as compensation to the Participant. The Committee, in its sole discretion, may direct the accumulation and payment of dividends on the Restricted Stock to the Participant at such times, and in such form and manner, as determined by the Committee.

(b) Service-Based Award.

(i) *Restriction Period.* At the time a Service-Based Restricted Stock Award is granted, the Committee will establish a restriction period applicable to such Award which will be not less than one year and not more than 10 years. Each Restricted Stock Award may have a different restriction period, as determined in the sole discretion of the Committee.

(ii) Forfeiture or Payout of Award. Unless the Committee, in its sole discretion, specifies otherwise at the time of the grant of the Award, in the event that a Participant ceases employment during a restriction period, a Restricted Stock Award is subject to forfeiture or payout (*i.e.*, removal of restrictions) as follows: (A) Termination—the Restricted Stock Award is forfeited entirely; (B) Retirement, Disability or death—payout of the Restricted Stock Award is prorated for service during the period; or (C) Early Retirement—if at the Participant's request, the payout or forfeiture of the Restricted Stock Award is determined in the sole discretion of the Committee, or if at the Company's request, payout of the Restricted Stock Award is prorated for service during the period; provided that the Committee may modify the above at the time the Participant ceases employment, if it determines, in its sole discretion, that special circumstances warrant such modification. In the event of forfeiture, the Participant shall not be entitled to any payment or consideration by the Company, and neither the Participant nor any successors, heirs, assigns or personal representatives of such Participant shall thereafter have any further rights or interest in such shares.

Except as set forth above in this Section 10(b)(ii), upon completion of the restriction period, all Award restrictions will lapse and certificates representing full ownership of the Award in Stock will be issued (the payout) to the Participant or his or her legal representative, without the restrictive legend described in Section 10(a) hereof.

## (c) Performance-Based Award.

(i) *Restriction Period.* At the time a Performance-Based Restricted Stock Award is granted, the Committee will establish a restriction period applicable to such Award which will be not less than one year and not more than ten years. Each Restricted Stock Award may have a different restriction period, as determined in the sole discretion of the Committee. The Committee also will establish a Performance Period.

(ii) *Performance Objectives*. Prior to the Date of Grant, the Committee will determine the number of shares of Restricted Stock for each Target Performance Award that will be issued on the Date of Grant. No later than 90 days after the commencement of each Performance Period, the Committee will determine the performance objectives for each Participant's Target Performance Award. Performance objectives may vary from Participant to Participant and will be based upon such performance criteria or combination of factors as the Committee, in its sole discretion, deems appropriate, which may include, but not be limited to, the performance of the Participant, the Company, one or more subsidiaries or any combination thereof. Performance Periods may overlap and Participants may participate simultaneously with respect to Performance-Based Restricted Stock Awards for which different Performance Periods are prescribed.

If, during the course of a Performance Period, significant events occur, as determined in the sole discretion of the Committee, which the Committee expects to have a substantial effect on any performance objective during the Performance Period, the Committee may revise such objective.

(iii) *Forfeiture or Payout of Award*. As soon as practicable after the end of each Performance Period, the Committee will determine whether the performance objectives and other material terms of the Award were satisfied. The Committee's determination of all such matters will be final and conclusive.

As soon as practicable after the later of (A) the date the Committee makes the above determination, or (B) the completion of the restriction period, the Committee will determine the Earned Performance Award for each Participant. Such determination may result in forfeiture of all or some shares of Restricted Stock (if Target Performance Award performance objectives were not attained) or the issuance of additional shares of Stock (if Target Performance Award performance objectives were exceeded), and will be based upon such factors as the Committee determines, in its sole discretion, but including the Target Performance Award performance objectives.

Unless the Committee, in its sole discretion, specifies otherwise at the time of the grant of the Award, in the event that a Participant ceases employment during a restriction period, a Restricted Stock Award is subject to forfeiture or payout (*i.e.*, removal of restrictions) as follows: (A) Termination—the Restricted Stock Award is forfeited entirely; (B) Retirement or Disability—the amount of the Earned Performance Award is determined in accordance with the preceding two paragraphs at the end of the Performance

Period and a pro rata portion of the resulting Earned Performance Award is received by the Participant based on service during the period; (C) death—payout of the Restricted Stock Award is prorated for service during the period, without regard to the performance objectives; or (D) Early Retirement—if at the Participant's request, the payout or forfeiture of the Restricted Stock Award is determined in the sole discretion of the Committee, or if at the Company's request, the amount of the Earned Performance Award is determined in accordance with the preceding two paragraphs at the end of the Performance Period and a pro rata portion of the resulting Earned Performance Award is received by the Participant based on service during the period; provided that the Committee may modify the above at the time the Participant ceases employment, if it determines, in its sole discretion, that special circumstances warrant such modification. In the event of forfeiture, the Participant nor any successors, heirs, assigns or personal representatives of such Participant shall thereafter have any further rights or interest in such shares.

With respect to shares of Restricted Stock for which restrictions lapse, certificates representing full ownership of the Award in Stock will be issued (the payout) without the restrictive legend described in Section 10(a) hereof. Certificates also will be issued for additional Stock, if any, awarded to the Participant because Target Performance Award performance objectives were exceeded.

(d) Waiver of Section 83(b) Election. If directed by the Committee, as a condition of receiving an Award of Restricted Stock, a Participant must waive in writing the right to make an election under Section 83(b) of the Code to report the value of the Restricted Stock as income on the Date of Grant.

#### 11. Stock Options.

(a) *Grants of Options.* One or more Options may be granted to any person who is an Eligible Person on the Date of Grant (except that Incentive Stock Options may be granted only to persons employed by the Company or a subsidiary of the Company on a regular basis) without the payment of consideration by the Participant.

(b) Stock Option Agreement. Each Option granted under the Plan will be evidenced by a "Stock Option Agreement" between the Company and the Participant containing provisions determined by the Committee, in its sole discretion, including, without limitation, provisions to qualify Incentive Stock Options as such under Section 422 of the Code if directed by the Committee at the Date of Grant; provided that each Incentive Stock Option Agreement must include the following terms and conditions: (i) that the Options are exercisable, either in whole or in part, with a partial exercise not affecting the exerciseability of the balance of the Option; (ii) every share of Stock purchased through the exercise of an Option will be paid for in full at the time of the exercise; (iii) each Option will cease to be exercisable, as to any share of Stock, at the earliest of (A) the Participant's purchase of the Stock to which the Option relates, (B) the Participant's exercise of a related Stock Appreciation Right, or (C) the lapse of the Option; (iv) Options will not be transferable by the Participant except by will or the laws of descent and distribution and will be exercisable during the Participant's lifetime only by the Participant or by the Participant's guardian or legal representative; and (v) notwithstanding any other provision, in the event of a public tender for all or any portion of the Stock or in the event that any proposal to merge or consolidate the Company with another company is approved by the stockholders of the Company, the Committee, in its sole discretion, may declare any previously granted Option to be immediately exercisable.

(c) Option Price. The Option price per share of Stock will be set by the grant, but will be not less than 100% of the Fair Market Value at the Date of Grant.

(d) *Form of Payment.* At the time of the exercise of the Option, the Option price will be payable in cash or in other shares of Stock or in a combination of cash and other shares of Stock, in a form and manner as required by the Committee, in its sole discretion. When Stock is used in full or partial payment of the Option price, it will be valued at the Fair Market Value on the date the Option is exercised.

(e) Other Terms and Conditions. The Option will become exercisable in such manner and within such Option Period or Periods, not to exceed 10 years from its Date of Grant, as set forth in the Stock Option Agreement upon payment in full. Except as otherwise provided in this Plan or in the Stock Option Agreement, any Option may be exercised in whole or in part at any time, but in no event beyond the Option Period set by the grant.

(f) Lapse of Options. An option will lapse upon the earlier of: (i) 10 years from the Date of Grant, or (ii) at the expiration of the Option Period set by the grant. Unless the Committee, in its sole discretion, specifies otherwise in the Stock Option Agreement, if the Participant ceases employment for any reason within the Option Period and prior to the lapse of the Option, the Option will lapse as follows: (i) Termination—if Termination occurs prior to the Option being exercisable, the Option will lapse on the effective date of the Termination, or if Termination occurs after the Option becomes exercisable and within the Option Period, the Option will lapse if not exercised within 90 days of the effective date of the Termination; or (ii) Retirement, Early Retirement or Disability-the Option will lapse at the expiration of the Option Period set by the grant; provided that the Committee may modify the above if it determines, in its sole discretion, that special circumstances warrant such modification; and provided further that, notwithstanding the above, for Incentive Stock Options, if not yet exercisable, the Option will become exercisable immediately upon Retirement, Early Retirement or Disability, but will lapse if not exercised within (A) 90 days of the effective date of the Retirement or Early Retirement or (B) one year of the effective date of the Disability. Unless the Committee, in its sole discretion, specifies otherwise in the Stock Option Agreement, if the Participant dies within the Option Period and prior to the lapse of the Option, the Option will lapse at the expiration of the Option Period set by the grant unless it is exercised before such time by the Participant's legal representative(s) or by the person(s) entitled to do so under the Participant's will or, if the Participant fails to make testamentary disposition of the Option or dies intestate, by the person(s) entitled to receive the Option under the applicable laws of descent and distribution; provided that notwithstanding the above, for Incentive Stock Options, the Option, if not yet exercisable, will become exercisable immediately upon death of a Participant who is an employee of the Company at the time of death, but will lapse if not exercised within one year of the date of death by any such person.

(g) Individual Limitation. In the case of an Incentive Stock Option, the aggregate Fair Market Value of the Stock for which Incentive Stock Options (whether under this Plan or another arrangement) in any calendar year are first exercisable will not exceed \$100,000 with respect to such calendar year (or such other individual limit as may be in effect on the Code on the Date of Grant).

12. Performance Units.

(a) *Performance Units*. One or more Performance Units may be earned by an Eligible Person based on the achievement of preestablished performance objectives during a Performance Period.

(b) Performance Period and Performance Objectives. Prior to the Date of Grant, the Committee will determine a Performance Period and the number of Performance Units subject to each Target Performance Award. No later than 90 days after the commencement of each Performance Period, the Committee will determine the performance objectives for each Participant's Target Performance Award. Performance objectives may vary from Participant to Participant and will be based upon such performance criteria or combination of factors as the Committee, in its sole discretion, deems appropriate, which may include, but not be limited to, the performance of the Company and/or the Participant individually or collectively with any other individuals or groups, or any combination thereof. Performance Periods may overlap and Participants may participate simultaneously with respect to Performance Units for which different Performance Periods are prescribed.

VII-9

If, during the course of a Performance Period, significant events occur, as determined in the sole discretion of the Committee, which the Committee expects to have a substantial effect on a performance objective during the Performance Period, the Committee may revise such objective.

(c) *Forfeiture or Payout of Award.* As soon as practicable after the end of each Performance Period, the Committee will determine whether the performance objectives and other material terms of the Award were satisfied. The Committee's determination of all such matters will be final and conclusive.

As soon as practicable after the date the Committee makes the above determination, the Committee will determine the Earned Performance Award for each Participant. Such determination may result in an increase or decrease in the number of Performance Units payable based upon such Participant's Target Performance Award, and will be based upon such factors as the Committee determines, in its sole discretion, but including the Target Performance Award performance objectives.

In the event a Participant ceases employment during a Performance Period, unless otherwise determined by the Committee, in its sole discretion, all Performance Units granted to a Participant shall terminate and be forfeited by such Participant to the Company without payment or any consideration by the Company, and neither the Participant nor any successors, heirs, assigns or personal representatives of such Participant shall thereafter have any further rights or interest in such Performance Units.

(d) Form and Timing of Payment. Each Performance Unit is payable in cash or shares of Stock or in a combination of cash and Stock, as determined by the Committee, in its sole discretion. Such payment will be made as soon as practicable after the Earned Performance Award is determined.

#### 13. Stock Appreciation Rights.

(a) Grants of Stock Appreciation Rights. Stock Appreciation Rights may be granted under this Plan in conjunction with an Option either at the Date of Grant or by amendment or may be separately granted. Stock Appreciation Rights will be subject to such terms and conditions not inconsistent with this Plan as the Committee may impose.

(b) Right to Exercise; Exercise Period. A Stock Appreciation Right issued pursuant to an Option will be exercisable to the extent the Option is exercisable; both such Stock Appreciation Right and the Option to which it relates will not be exercisable during the six months following their respective Dates of Grant except in the event of the Participant's Disability or death. A Stock Appreciation Right issued independent of an Option will be exercisable pursuant to such terms and conditions established in the grant. Notwithstanding such terms and conditions, in the event of a public tender for all or any portion of the Stock or in the event that any proposal to merge or consolidate the Company with another company is approved by the stockholders of the Company, the Committee, in its sole discretion, may declare any previously granted Stock Appreciation Right immediately exercisable.

(c) *Failure to Exercise*. If on the last day of the Option Period, in the case of a Stock Appreciation Right granted pursuant to an Option, or the specified Exercise Period, in the case of a Stock Appreciation Right issued independent of an Option, the Participant has not exercised a Stock Appreciation Right, then such Stock Appreciation Right will be deemed to have been exercised by the Participant on the last day of the Option Period or Exercise Period.

(d) *Payment.* An exercisable Stock Appreciation Right granted pursuant to an Option will entitle the Participant to surrender unexercised the Option or any portion thereof to which the Stock Appreciation Right is attached, and to receive in exchange for the Stock Appreciation Right payment (in cash or Stock or a combination thereof as described below) equal to the excess of the Fair Market Value of one share of Stock at the date of exercise over the Option price, times the number of shares called for by the Stock Appreciation Right not granted pursuant to an Option, the Participant will receive for each Stock Appreciation Right payment (in cash or Stock or a combination thereof) which is so surrendered. Upon exercise of a Stock Appreciation Right not granted pursuant to an Option, the Participant will receive for each Stock Appreciation Right payment (in cash or Stock or a combination thereof as described below) equal to the

excess of the Fair Market Value of one share of Stock at the date of exercise over the Fair Market Value of one share of Stock at the Date of Grant of the Stock Appreciation Right, times the number of shares called for by the Stock Appreciation Right.

The Committee may direct the payment in settlement of the Stock Appreciation Right to be in cash or Stock or a combination thereof. Alternatively, the Committee may permit the Participant to elect to receive cash in full or partial settlement of the Stock Appreciation Right; provided that (i) the Committee must consent to or disapprove such election and (ii) unless the Committee directs otherwise, the election and the exercise must be made during the period beginning on the third business day following the date of public release of quarterly or year-end earnings and ending on the twelfth business day following the date of public release of quarterly or year-end earnings. The value of the Stock to be received upon exercise of a Stock Appreciation Right shall be the Fair Market Value of the Stock on the trading day preceding the date on which the Stock Appreciation Right is exercised. To the extent that a Stock Appreciation Right issued pursuant to an Option is exercised, such Option shall be deemed to have been exercised, and shall not be deemed to have lapsed.

(e) *Nontransferability.* A Stock Appreciation Right will not be transferable by the Participant except by will or the laws of descent and distribution and will be exercisable during the Participant's lifetime only by the Participant or by the Participant's guardian or legal representative.

(f) Lapse of a Stock Appreciation Right. A Stock Appreciation Right will lapse upon the earlier of: (i) 10 years from the Date of Grant; or (ii) at the expiration of the Exercise Period as set by the grant. Unless the Committee, in its sole discretion, specifies otherwise at the time of the grant of the Award, if the Participant ceases employment within the Exercise Period and prior to the lapse of the Stock Appreciation Right, the Stock Appreciation Right will lapse as follows: (i) Termination---if Termination occurs prior to the Stock Appreciation Right being exercisable, the Stock Appreciation Right will lapse on the effective date of the Termination, or if Termination occurs after the Stock Appreciation Right becomes exercisable and within the Exercise Period, the Stock Appreciation Right will lapse if not exercised within 90 days of the effective date of the Termination; or (ii) Retirement, Early Retirement or Disability-the Stock Appreciation Right will lapse at the expiration of the Exercise Period set by the grant; provided that the Committee may modify the above at the time the Participant ceases employment, if it determines, in its sole discretion, that special circumstances warrant such modification. Unless the Committee, in its sole discretion, specifies otherwise at the time of the grant of the Award, if the Participant dies within the Exercise Period and prior to the lapse of the Stock Appreciation Right, the Stock Appreciation Right will lapse at the expiration of the Exercise Period set by the grant unless it is exercised before such time by the Participant's legal representative(s) or by the person(s) entitled to do so under the Participant's will or, if the Participant fails to make testamentary disposition of the Stock Appreciation Right or dies intestate, by the person(s) entitled to receive the Stock Appreciation Right under the applicable laws of descent and distribution.

14. Dividend Equivalents.

(a) Grants of Dividend Equivalents. Dividend Equivalents may be granted under this Plan in conjunction with an Award of an Option or a separately-awarded Stock Appreciation Right, at the Date of Grant or by amendment, without consideration by the Participant. Dividend Equivalents also may be granted under this Plan in conjunction with Performance Units, at any time during the Performance Period, without consideration by the Participant.

(b) *Payment.* Each Dividend Equivalent will entitle the Participant to receive an amount equal to the dividend actually paid with respect to a share of Stock on each dividend payment date from the Date of Grant to the date the Dividend Equivalent lapses as set forth in Section 14(d) hereof. The Committee, in its sole discretion, may direct the payment of such amount at such times and in such form and manner as determined by the Committee.

(c) Nontransferability. A Dividend Equivalent will not be transferable by the Participant.

(d) Lapse of a Dividend Equivalent. Each Dividend Equivalent will lapse on the earlier of: (i) the date of the lapse of the related Option or Stock Appreciation Right; (ii) the date of the exercise of the related Option or Stock Appreciation Right; (iii) the end of the Performance Period (or, if earlier, the date the Participant ceases employment) of the related Performance Units; or (iv) the lapse date established by the Committee on the Date of Grant of the Dividend Equivalent.

15. Annual Incentive Awards. Annual Incentive Awards may be granted to any person who is an Eligible Person on the Date of Grant without the payment of consideration by the Participant, in such amounts and on such terms and conditions as the Committee, in its sole discretion, deems appropriate. In the event a Participant ceases employment for any reason prior to receiving an Annual Incentive Award, unless otherwise determined by the Committee, in its sole discretion, such Award shall terminate and be forfeited by such Participant without payment or any consideration by the Company, and neither the Participant nor any successors, heirs, assigns or personal representatives of such Participant shall thereafter have any further rights or interest in such Award.

16. Effect of Change in Control. Notwithstanding anything in this Plan to the contrary, unless the Committee, in its sole discretion, specifies otherwise at the time of the grant of an Award or at any time thereafter prior to the exercise, vesting, lapse or forfeiture of such Award, as applicable, upon the occurrence of a Change in Control: (a) all Options and Stock Appreciation Rights granted under this Plan that are outstanding at the date of such Change of Control shall become exercisable in full immediately, without regard to the years that have elapsed from the Date of Grant; (b) with respect to Performance Units that are outstanding at the date of such Change in Control, all uncompleted Performance Periods at the date of such Change in Control shall be deemed to have been completed, the maximum level of performance set forth under the respective performance objectives shall be deemed to have been achieved and a pro rata portion (based on the number of full and partial months that have elapsed with respect to such Performance Period) of each such outstanding Award shall become payable immediately in cash to each Participant, with the remainder of each such outstanding Award being canceled for no value; (c) and conditions to the vesting of the Restricted Stock (including, but not limited to, attainment of performance objectives, if any, assuming that maximum performance was achieved) shall be deemed to have been satisfied, any uncompleted time periods (whether Service-Based or Performance-Based, or otherwise) at the date of such Change in Control shall be deemed to have been completed and all restrictions with respect to outstanding shares of Restricted Stock shall lapse immediately, and such shares shall be fully vested and nonforfeitable; and (d) Annual Incentive Awards shall be treated in such manner as the Committee, in its sole discretion, shall determine at the time of such Change in Control; provided that, this Section 16 shall not apply with respect to Participants who have retired (whether Early Retirement or Normal Retirement) prior to the occurrence of a Change in Control, whose Awards shall vest, lapse or be forfeited in the same manner as if a Change in Control had not occurred.

17. General Restrictions. This Plan and each Award under this Plan shall be subject to the requirement that, if at any time the Committee shall determine that: (a) the listing, registration or qualification of the shares of Stock subject or related thereto upon any securities exchange or under any state or federal law; or (b) the consent or approval of any governmental regulatory body; or (c) an agreement by the recipient of an Award with respect to the disposition of shares of Stock; is necessary or desirable as a condition of, or in connection with this Plan or the granting of such Award or the issuance or acquisition of shares of Stock thereunder, this Plan will not be effective and the Award may not be consummated in whole or in part unless such listing, registration, qualification, consent, approval or agreement, as the case may be, shall have been effected or obtained free of any conditions not acceptable to the Committee.

18. Amendment of Plan. The Board or the Committee may at any time and from time to time alter, amend, suspend or terminate this Plan in whole or in part, except: (a) no amendment that requires

stockholder approval in order for (i) the shares of Stock subject to this Plan to continue to be eligible for listing under the New York Stock Exchange (or its successor or any other exchange on which the Committee determines the Stock should be listed) rules and guidelines for such listing, or (ii) this Plan to continue to be eligible to satisfy the requirements for exemption of acquisition of securities under this Plan pursuant to the Section 16 Rules, or (iii) this Plan to continue to be eligible to comply with Section 162(m) of the Code, or any amendment of or substitute for any of them, shall be effective unless the same shall be approved by the requisite vote of the stockholders of the Company; and (b) during the period commencing on the date of a Potential Change in Control and ending on the earlier of (i) six months after the later of (A) the expiration of six months following the occurrence of such Potential Change in Control, or (B) the Board's adoption of a resolution certifying that a Potential Change in Control ceases to exist, or (ii) the date of the occurrence of a Change in Control, the Plan may not be amended in any manner adverse or detrimental to any person who is a Participant immediately prior to such amendment or to such Participant's beneficiaries (including, but not limited to, any adverse amendment to this sentence) without the consent of at least two-thirds (3) of Participants who were Participants immediately prior to such amendment. Notwithstanding the foregoing, the Committee may amend this Plan as desirable or in the discretion of the Committee to address any issues concerning Section 162(m) of the Code, maintaining an exemption under the Section 16 Rules, or maintaining the listing of the Stock subject to the Plan on any stock exchange, to the extent that such amendment does not require stockholder action for the purposes described in clause (a) above.

19. Rights of a Stockholder. The recipient of any Award under this Plan shall have no rights as a stockholder with respect thereto unless and until legended certificates for shares of Stock are issued.

20. *Rights to Terminate Employment.* Nothing in this Plan or in any agreement entered into pursuant to this Plan shall constitute a contract or employment or confer upon any Participant the right to continue in the employment of the Company or its subsidiaries or affect any right that the Company or its subsidiaries may have to terminate the employment of such Participant, or be deemed to be consideration for, or a condition of, continued employment of any person.

21. Nontransferability. No benefit provided under this Plan shall be subject to alienation or assignment by a Participant (or by any person entitled to such benefit pursuant to the terms of this Plan), nor shall it be subject to attachment or other legal process, except: (a) to the extent specifically mandated and directed by applicable state or federal statute; and (b) as requested by the Participant (or by any persons entitled to such benefit pursuant to the terms of this Plan), and approved by the Committee, to satisfy tax withholding.

22. Tax Withholding. The Company or a subsidiary may withhold any applicable federal, state or local taxes at such time and upon such terms and conditions as are required by law or determined by the Company or such subsidiary. Subject to compliance with any requirements of applicable law, the Committee may permit or require a Participant to have any portion of any withholding or other taxes payable in respect to a distribution of Stock satisfied through the payment of cash by the Participant to the Company or a subsidiary, the retention by the Company or such subsidiary of shares of Stock, or delivery of previously-owned shares of the Participant's Stock, having a Fair Market Value equal to the withholding amount. A Reporting Person may elect to have a sufficient number of shares of Stock withheld to fulfill such withholding obligation (the "Withholding Election") only if the Withholding Election is made during any period in which a Withholding Election may be made to ensure an exemption with respect to such transaction pursuant to the Section 16 Rules. Any fractional share of Stock required to satisfy such withholding obligations shall be disregarded and the amount due with respect thereto shall be paid in cash by the Participant.

23. Non-Uniform Determinations. The Committee's determinations under this Plan (including, but not limited to, determinations of the persons eligible to receive Awards, the form, amount and timing of Awards, the terms and provisions of Awards and the agreements evidencing Awards, and the establishment

of values and performance targets and objectives) need not be uniform and may be made by it selectively among persons who receive, or are eligible to receive, Awards under this Plan, whether or not such persons are similarly situated, except as such determinations may be limited by applicable law.

24. Adjustments. In the event that the Committee shall determine, in its sole discretion, that any dividend or other distribution (whether in the form of cash, Stock or other property), recapitalization, Stock split, reverse Stock split, reorganization, merger, consolidation, spin-off, combination, repurchase or share exchange or other similar corporate transaction or event, affects the Stock such that an adjustment is appropriate in order to prevent dilution or enlargement of the rights of Participants, then the Committee shall make such equitable changes or adjustments as it deems necessary or appropriate to any or all of (a) the number and kind of shares of Stock that may thereafter be issued in connection with Awards, whether in the aggregate or to any individual, (b) the number and kind of shares of Stock issued or issuable in respect of outstanding Awards, and (c) the exercise price, grant price or purchase price relating to any Award; and in any such event, the Committee may, in its sole discretion, terminate any Award and make equitable cash payment for the value of such Award, as such value shall be determined by the Committee in its sole discretion; provided that, with respect to Incentive Stock Options, such adjustment or termination shall be made in accordance with Section 424 of the Code or any amendment thereof or substitute therefor or regulations thereunder.

25. *Fractional Shares.* Any fractional shares concerning Awards shall be eliminated at the time of payment or payout by rounding down for fractions of less than one-half and rounding up for fractions of equal to or greater than one-half. No cash settlements shall be made with respect to fractional shares eliminated by rounding.

26. Government and Other Regulations. The obligation of the Company to make payment of Awards in Stock or otherwise shall be subject to all applicable laws, rules and regulations, and to such approvals by any governmental agencies as may be required. The Company shall be under no obligation to register under the Securities Act of 1933, as amended (the "Act"), any of the shares of Stock issued, delivered or paid in settlement under this Plan. If Stock awarded under this Plan in certain circumstances may be exempt from registration under the Act, the Company may restrict the transfer of such Stock in such manner as it deems advisable to ensure such exempt status.

27. Indemnification. Each person who is, or at any time serves as, a member of the Committee (and each person to whom the Committee or any member thereof has delegated any of its authority or power under this Plan), to the fullest extent of applicable law, shall be indemnified and held harmless by the Company against and from: (a) any loss, cost, liability or expense that may be imposed upon or reasonably incurred by such person in connection with or resulting from any claim, action, suit or proceeding to which such person may be a party or in which such person may be involved by reason of any action or failure to act under this Plan; and (b) any and all amounts paid by such person in satisfaction of judgment in any such action, suit or proceeding relating to this Plan. As a condition to the Company's indemnification obligation hereunder, each person covered by this indemnification shall give the Company an opportunity, at the Company's expense, to handle and defend the same before such person undertakes to handle and defend it on such person's own behalf. The foregoing right of indemnification shall not be exclusive of any other rights of indemnification to which such persons may be entitled under the Restated Certificate of Incorporation or By-Laws of the Company or any of its subsidiaries, as a matter of law, or otherwise, or any power that the Company may have to indemnify such person or hold such person harmless.

28. *Reliance on Reports.* Each member of the Committee (and each person to whom the Committee or any member thereof has delegated any of its authority or power under this Plan) shall be justified fully in relying or acting in good faith upon any report made by the independent public accountants of the Company and its subsidiaries and upon any other information furnished in connection with this Plan. In no event shall any person who is or shall have been a member of the Committee be liable for any determination made or other action taken or any omission to act in reliance upon any such report or information or for any action taken, including the furnishing of information, or failure to act, if in good faith.

29. *Effect on Other Plans.* Participation in this Plan shall not affect a Participant's eligibility to participate in any other benefit or incentive plan of the Company and any Awards made pursuant to this Plan shall not be considered compensation for purposes of determining benefits under any pension, profit sharing or other retirement or welfare plan, or for any other general employee benefit plan, unless specifically provided otherwise.

30. *Funding of this Plan.* This Plan shall be unfunded. The Company shall not be required to establish any special or separate fund or to make any other segregation of assets to assure the payment of any Award and payment of Awards shall be subordinate to the claims of the Company's general creditors. In no event shall interest be paid or accrued on any Award, including any unpaid installments of Awards.

31. *Expenses.* The expenses of administering this Plan shall be borne by the Company and its subsidiaries.

32. Governing Law. All matters relating to this Plan or to Awards granted hereunder shall be governed by the laws of the State of Delaware, without regard to the principles of conflict of laws.

33. *Titles and Headings*. The titles and headings of the sections in this Plan are for convenience or reference only, and in the event of any conflict, the text of this Plan, rather than such titles or headings, shall control.